



AFFILIATE "SS "ATOMPROJECTENGINEERING" JSC "NNEGC "ENERGOATOM"

<u>5333</u>

(case registration number of the impact assessment on the environment of the planned activity)

6 Simi Prakhovykh St., Kyiv, 01033 phone number: +380 44 206 97 87 e-mail: office_api@direkcy.atom.gov.ua URL: https://energoatom.com.ua EDRPOU: 36469530

ENVIRONMENTAL IMPACT ASSESSMENT REPORT

of the planned activity "Construction of power units 5 and 6 with AP1000 reactor unit at the Khmelnytskyi NPP site"

P.46.02/23-01-EIAR

Acting General Director

Deputy General Director for Production Management

Tetiana AMOSOVA

Iryna MITICHKINA

Head of the Design and Development Division

Dmytro KASIAN

2024

CONTENTS

ACRONYMS LIST						
INTRODUCTION						
1	1 THE PLANNED ACTIVITY DESCRIPTION					
	1.1	De	scription of the location of the planned activity			
	1.2	Ob	jectives of the planned activity14			
	1.3 constr disma prepa	Der ructi intlin rato	scription of the characteristics of the activity during preparatory and ion works and implementation of the planned activity, including (if necessary) ng works, and the need (restrictions) on the use of land plots during ry and construction works and implementation of the planned activity 14			
	1.3	.1	Worksite			
	1.3	.2	Duration of work performance15			
	1.3	.3	Content, main assets and technologies of work performance15			
	1.3	.4	Supply chain for the planned activities			
	1.3	.5	Planned temporary structures, transportation and utility networks			
	1.3 of t	.6 these	Natural resource needs for preparatory and construction works and sources e needs			
	1.3 (wa acti	.7 astev iviti	Management of construction and other solid waste, as well as liquid waste water and sewage sludge) generated during preparatory and construction es			
	1.3 per	.8 mis	Compliance with air quality standards (hygienic standards) and maximum sible noise levels in relation to the nearest residential buildings			
	1.4 manu (wate	De fact r, la	scription of the main characteristics of the planned activity (including uring processes), e.g., the type and quantity of materials and natural resources nd, soil, biodiversity) to be used			
	1.4	.1	Characteristics of the NPP site area			
	1.4	.2	Technological characteristics of the planned activity			
	1.4	.3	Water and energy supply of the construction performance			
	1.4	.4	Provision of resources during operation			
	1.5 Assessment of the expected wastes, emissions (discharges), water, air, soil and subsoil pollution, noise, vibration, light, heat and radiation pollution by type and amount, as well as radiation caused by the preparatory and construction activities and the planned activity					
	1.5	.1	Contamination assessment			
	1.5	.2	Assessment of non-radiation pollution			

2. DESCRIPTION OF THE REASONABLE ALTERNATIVES TO THE PLANNE ACTIVITY, THE MAIN REASONS FOR SELECTING THE PROPOSED OPTION CONSIDERING THE ENVIRONMENTAL CONSEQUENCES	D N, 57
2.1 Construction conditions at the South Ukrainian NPP site	57
2.1.1 Brief site description	57
2.1.2 Scheme of the NPP situational plan. A disposed land area	58
2.1.3 Power output	70
2.1.4 Service water supply system	70
2.1.5 Transport connection	70
2.1.6 Current condition of the land plot	70
2.2 Construction conditions at the Rivne NPP site	71
2.2.1 Brief site description	71
2.2.2 Scheme of the NPP situational plan. A disposed land area	72
2.2.3 Power output	72
2.2.4 Service water supply system	72
2.2.5 Transport connection	74
2.2.6 The current land plot state	74
2.3 Construction conditions at the KhNPP site	74
2.3.1 Brief site description	74
2.3.2 Scheme of the NPP situational plan. A disposed land area	75
2.3.3 Power output	77
2.3.4 Service water supply system	77
2.3.5 Transport connection	78
2.3.6 Current condition of the land plot	78
2.4 Conclusion on the selection of a site for the new power units construction	78
3 A DESCRIPTION OF THE CURRENT ENVIRONMENTAL STATE (BAS SCENARIO) AND A DESCRIPTION OF ITS POSSIBLE CHANGE WITHOUT TH PLANNED ACTIVITY TO THE EXTENT THAT NATURAL CHANGES FROM TH BASE SCENARIO CAN BE ESTIMATED BASED ON AVAILABL ENVIRONMENTAL INFORMATION AND SCIENTIFIC KNOWLEDGE	SE IE IE LE 80
3.1 Geographical location of the site	30
3.2 Terrain and landscapes	30
3.3 Soils	31
3.4 Climate	32
3.5 Hydrological characteristics	33
3.5.1 Surface water	34

3.5.2 Groundwater
3.6 Generalized characteristics of flora and fauna
3.6.1 Floristic complexes
3.6.2 Faunal complexes
3.7 Characterization of the nature reserve fund objects
3.7.1Emerald Network 102
3.8 Characterization of the distribution of all negative factors in the area of influence of the planned activity
3.8.1 Natural hazards
3.8.2 Hazards of anthropogenic origin
4 DESCRIPTION OF THE ENVIRONMENTAL FACTORS LIKELY TO BE AFFECTED BY THE PLANNED ACTIVITY AND ITS ALTERNATIVES, INCLUDING POPULATION HEALTH, FAUNA, FLORA, BIODIVERSITY, LAND (INCLUDING LAND ACQUISITION), SOIL WATER, AIR, CLIMATIC FACTORS (INCLUDING CLIMATE CHANGE AND GREENHOUSE GAS EMISSIONS), PROPERTY, INCLUDING ARCHITECTURAL, ARCHAEOLOGICAL AND CULTURAL HERITAGE, LANDSCAPE, SOCIO-ECONOMIC CONDITIONS AND INTERACTIONS BETWEEN THESE FACTORS
4.1 Population health
4.2 Condition of fauna 109
4.3 Condition of flora
4.4 Land resources
4.5 Aquatic (water) environment 113
4.6 Geological environment
4.7 Atmospheric air
4.7.1 Radiation impact 115
4.7.2 Non-radiation impact 120
4.8 Climate effects
4.9 Architectural, historical, and cultural monuments
4.10 Socio-economic factors
4.11 Technogenic environment
5 DESCRIPTION AND THE POSSIBLE ENVIRONMENTAL IMPACT ASSESSMENT OF THE PLANNED ACTIVITY, INCLUDING THE MAGNITUDE AND SCOPE OF SUCH IMPACT (AREA AND POPULATION THAT MAY BE AFFECTED), NATURE (IF ANY - TRANSBOUNDARY), INTENSITY AND COMPLEXITY, PROBABILITY, EXPECTED ONSET, DURATION, FREQUENCY AND INEVITABILITY OF THE IMPACT (INCLUDING DIRECT AND ANY

TERM AN	D LONG-TERM, PERMANENT AND TEMPORARY, POSITIVE AND EIMPACT)			
5.1 Pre including	paratory and construction works and the planned activity implementation, (if necessary) dismantling works after completion of such activity 131			
5.1.1	Preparatory and construction works			
5.1.2	Implementation of planned activities			
5.1.3	Works during the decommissioning			
5.1.4	Conclusion			
5.2 Use planned a	e of natural resources, including land, soil, water and biodiversity during the ctivity			
5.2.1	Soils136			
5.2.2	Water resources			
5.2.3	Biodiversity			
5.2.4	Conclusions 150			
5.3 Pol radiation	lutant emissions and discharges, noise, vibration, light, heat, contamination, and other impact factors, as well as waste management operations 150			
5.3.1	Pollutant exposure			
5.3.2	Thermal effects			
5.3.3	Wastewater impact 156			
5.3.4	Noise exposure			
5.3.5	Vibration effect			
5.3.6	Ultrasound exposure			
5.3.7	Radiation effect			
5.3.8	Transboundary transfer			
5.3.9	Calculation of the sanitary protection zone and supervised area			
5.3.10	Waste treatment			
5.4 Rist through the function of the function	ks to human health, cultural heritage sites, and the environment, including he possibility of emergencies			
5.4.1	Risks to population health and the environment			
5.4.2	Risks to Cultural Heritage Sites			
5.5 The cumulative impact of other existing facilities, planned activities, and facilities for which a decision on the planned activity has been received, taking into account all existing environmental issues related to areas of special environmental protection significance that may be affected or where natural resources may be used				

199

5.7 Technology and Su	ıbstances Used	
5.7.1 Handling chem	icals and solutions	
5.7.2 Nuclear Fuel H	landling	
6 A DESCRIPTION OF ENVIRONMENTAL IMPA FORECASTING, AS WEL	THE FORECASTING ACTS AND THE ASSUI L AS THE USED ENVIR	METHODS USED TO ASSESS MPTIONS UNDERLYING SUCH CONMENTAL DATA
7 A DESCRIPTION C PRECLUDE, AVOID, F ENVIRONMENTAL IMP. POSSIBLE)	OF THE MEASURES REDUCE, ELIMINATE ACT, INCLUDING COM	ENVISAGED TO PREVENT, A SIGNIFICANT ADVERSE MPENSATORY MEASURES (IF
7.1 Description of tech hazardous substances, en	nnical solutions to preven sure fire and explosion sa	t accidents and localize releases of fety
7.2 Justification for the	e acceptability of the desig	gn solution package 219
7.3 Protection of the en	nvironment from ionizing	radiation 221
7.4 Measures to reduce	e non-radiation environme	ental impact 224
7.5 Comprehensive me	easures to reduce negative	impact on the air environment 227
7.6 Remediation measurement	ures	
7.6.1 Technical and	biological reclamation	
7.6.2 Normalization	of vegetation, fauna and r	ature reserves 229
7.7 Compensation mea	asures	
8 DESCRIPTION OF ENVIRONMENTAL AC VULNERABILITY TO MITIGATE THE DISA EMERGENCY RESPONSI	THE EXPECTED TIVITY IMPACT CA DISASTER RISKS, M STER IMPACT ON E MEASURES	SIGNIFICANT ADVERSE USED BY THE PROJECT'S EASURES TO PREVENT OR THE ENVIRONMENT AND 232
8.1 AP1000 design		
8.2 Probabilistic Safety	y Assessment	
8.3 Maximum design b conditions)	asis accident. Beyond des	ign basis accident (design extension
8.4 Dose load assessm	ent of an accident at the A	P1000 unit
8.4.1 Design basis ac	ccident at the KhNPP Unit	ts 5-6 with AP1000 reactors 240
8.4.2 Beyond Design	Basis Accident at the Kh	NPP AP1000 Units 5-6 245
8.5 The Ensuring Safe	ty Principles of the NPP	
8.6 Measures to protec	t the population and the e	nvironment254
8.7 Emergency prepar emergency	edness and emergency re	sponse in the event of a radiation
8.7.1 Radiation mon	itoring	

International commitments on early notification of a nuclear accident ... 260 8.7.2 IDENTIFICATION OF ALL DIFFICULTIES (TECHNICAL DEFICIENCIES, 9 LACK OF SUFFICIENT TECHNICAL MEANS OR KNOWLEDGE) IDENTIFIED IN PROCESS OF THE ENVIRONMENTAL IMPACT THE PREPARATION 10 ALL COMMENTS AND SUGGESTIONS RECEIVED BY THE AUTHORIZED CENTRAL BODY AFTER THE PUBLISHED NOTIFICATION OF THE PLANNED ACTIVITY, AND A TABLE WITH INFORMATION ON FULL CONSIDERATION. PARTIAL ONE OR JUSTIFICATION FOR COMMENTS REJECTION AND 11 A SUMMARY OF THE MONITORING AND CONTROL PROGRAMS FOR ENVIRONMENTAL IMPACTS DURING THE IMPLEMENTATION OF THE PLANNED ACTIVITY, AS WELL AS (IF NECESSARY) POST-PROJECT 12 SUMMARY OF NON-TECHNICAL INFORMATION INTENDED FOR A WIDE APPENDIX A Comments and proposals from the public on the planned activities, scope APPENDIX B Conditions for the scope of studies and level of information detail..... 334 APPENDIX C Calculations of air pollutant dispersion during construction and operation 340 Calculation of pollutant dispersion from construction equipment during the C.1 Calculation of pollutant dispersion during the KhNPP units 5 and 6 operation **C**.2 386 **H.1** H.2 H.3 H.4 H.5 H.6

APPENDIX J Information on the dam	481
APPENDIX K Comments and proposals on the planned activity, scope of resear the affected countries	ch from 482
APPENDIX L Calculations of the pollutant releases rate from the designed so	urces of
units 5 and 6 with referenced methods	512
APPENDIX M Characterization of the projected sources of non-radioactive p	ollutant
emissions	524

ACRONYMS LIST

AMS	—	aeronautical meteorological station (civil)
BDBA	_	beyond design basis accident
BDPP	_	backup diesel power plant
CA	_	controlled area
CAW	_	construction and assembly works
CDF	_	core damage frequency
ChNPP	_	Chernobyl Nuclear Power Plant
CMU	_	The Cabinet of Ministers of Ukraine
СР	_	cooling pond
CRDM	_	control rod drive mechanism
CST	_	condensate storage tank
CV	_	containment vessel
CWT	_	chemical water treatment
DE	_	design earthquake
DS	_	dead storage
DSL	_	dead storage level
Dnipro PS	_	Dnipro Power System
ECG	_	emergency crews and groups
EGP	_	exogenous geological process
EIA	_	environmental impact assessment
EPRS	_	emergency preparedness and response systems
FA	_	fuel assembly
Gy	_	Gray – unit of absorbed dose
HRO	_	high-risk object
HVL	_	high-voltage lines
I&C	_	instrumentation and control
IPS of Ukraine	_	integrated power system PS of Ukraine
JSC "NNEGC"	_	Joint Stock Company National Nuclear Energy Generating
"Energoatom"		Company Energoatom
KhNPP	_	Khmelnytskyi Nuclear Power Plant
LRW	_	liquid radioactive waste
MCP	_	main coolant pump
MCR	_	main control room
MDA	_	maximum design basis accident
MDE	_	maximum design basis earthquake
MEDF	_	maximum emergency discharge frequency
MPC	_	maximum permissible concentration
NF	_	nuclear fuel
NI	_	nuclear installation
NHL	_	normal headwater level
NPP	_	nuclear power plant

_	NPP shift supervisor
_	nuclear reactor core
_	natural reserve fund
_	open switchgear
_	permissible discharges
_	personal protective equipment
—	permissible releases
—	power system
—	probabilistic safety analysis
—	pumped-storage hydroelectricity
—	reactor facility
—	radioactive inert gas
—	radiation monitoring
—	radiation monitoring group
—	radioactive noble gases
—	radiation safety shop
—	radiation safety standards of Ukraine
—	radioactive waste
—	supervised area
—	start-up boiler house
_	site emergency response supervisor
_	steam generator
_	seismic hazard deterministic analysis
—	spent nuclear fuel
_	spray pool
_	sanitary protection zone
—	solid radioactive waste
—	solid radioactive waste system
—	separate subdivision
—	stake (marking measure, distance between stakes is of 100 m)
_	South Ukrainian Nuclear Power Plant
_	Sievert – unit of equivalent and effective dose
_	The United States of America
—	water and chemical regime

INTRODUCTION

The necessity of new nuclear power units' construction to replace NPP capacity that will be decommissioned after 2030 was determined by the Energy Strategy [1, 2].

On August 31, 2021, in Washington, D.C., with the participation of President of Ukraine Volodymyr Zelenskyy, Energoatom CEO Petro Kotin and Westinghouse CEO Patrick Fragman signed a Memorandum of Understanding (hereinafter referred to as the Memorandum) [3], which provides for the construction of five new nuclear power units using AP1000 technology in Ukraine.

The pilot project is the construction of two new AP1000 power units at the Khmelnytskyi NPP site. On June 2, 2022, an Agreement was signed at the KhNPP site to start the practical implementation of a joint project between NNEGC Energoatom and Westinghouse Electric Company for the construction of AP1000 power units at the KhNPP site.

In accordance with the Law of Ukraine "On Environmental Impact Assessment" [4], the environmental impact assessment is mandatory in the decision-making process for the implementation of planned activity. The construction of NPPs is subject to the EIA procedure along with the assessment of transboundary environmental impacts, including the Convention on Environmental Impact Assessment in a Transboundary Context (Espoo Convention).

The EIA is conducted to identify and describe environmental factors that are likely to be affected by the planned activity, including public health, socio-economic conditions, fauna, flora, biodiversity, soil, water, etc., and to assess all possible significant environmental and social impacts of the proposed project.

The main methodological approach and content of the study were determined by Ukrainian legislation, primarily the Law of Ukraine "On Environmental Impact Assessment" [4], which establishes the legal and organizational framework for the EIA, as well as the Order of the Ministry of Environment [5].

In preparing the EIA report, scientific literature, official reports and Internet information have been also used.

The measures to prevent, avoid, reduce, eliminate a significant adverse environmental impact, including (if possible) compensatory measures are developed in case of significant potential impacts. These measures are also aimed at enhancing the potentially favorable effects of the project implementation.

The EIA also includes monitoring and control activities regarding the environmental impacts during the implementation of the planned activity.

1 THE PLANNED ACTIVITY DESCRIPTION

The Environmental Impact Assessment Report has been developed in accordance with the requirements of paragraph 2 of Article 6 of the Law of Ukraine "On Environmental Impact Assessment" [4] in compliance with environmental, sanitary and hygienic, fire protection, urban planning and territorial restrictions in accordance with current regulatory documents.

According to paragraph 2 of part 2 of Article 3 of the Law of Ukraine "On Environmental Impact Assessment", the project object - "Construction of power units 5, 6 with the AP1000 reactor unit at the KhNPP site" – belongs to the first category of planned activity and facilities that may have a significant impact on the environment and are subject to environmental impact assessment.

The planned activity is the construction of power units No. 5 and No. 6 at the KhNPP site using the technical characteristics of the AP1000 reactor unit of the Westinghouse Electric Company for further operation and electricity generation. The AP1000 is a proven Generation III+ reactor with passive safety systems, modular standard design, high availability and load monitoring capability licensed by the U.S. Nuclear Regulatory Commission (NRC).

1.1 Description of the location of the planned activity

New construction of power units No. 5, 6 with AP1000 reactor at the KhNPP site in Netishyn, Netishyn City Territorial Community, Shepetivka District, and Khmelnytskyy Region.

The KhNPP site is located in the northwest of Slavuta district of Khmelnytskyy region of Ukraine, 18 km west of the district center of Slavuta, 100 km north of the regional center of Khmelnytskyy, near Netishyn (NPP town).

The KhNPP site and the boundaries of its observation area are shown in Fig. 1.1.

The KhNPP supervised area includes the territories of Khmelnytskyy region (lands of Izyaslav, Slavuta, Bilohirsk and Shepetivka districts) and Rivne region (lands of Ostroh, Hoshcha and Zdolbuniv districts).

Currently, power units 1 and 2 are in operation at the plant. The planned activity envisages the placement of power units No. 5 and No. 6 on the territory of the operating KhNPP (preliminary designated location).

The area of the main industrial site of the NPP (fenced) is 90.2 hectares. The area of the land plot allocated for the main site, ISP-110, ISP-330, ISP-750 and the supply and discharge canals between the main site and the ISP territory is 147.0804 hectares. The

site is located within the lands of Netishyn City Council and Slavuta State Forestry.



Figure 1.1 – Khmelnytskyi NPP location

Measures related to the disposal, reclamation and compensation of land acquisition costs were implemented during the commissioning of Unit 1. It should be noted that the previously allocated lands were characterized as low-productive for agricultural production.

The sufficiency of the land allocated for the placement of power units 5 and 6 and a set of facilities for their maintenance will be determined during the development of the working project "Construction of KhNPP Units 5 and 6".

The KhNPP site area has a well-developed network of roads and railways.



There is no water transport.

There are 47 protected areas of varying degrees of protection on the territory of KhNPP, with an area of more than 3000 hectares. Seven of the 47 protected areas have the status of national importance, and the other 40 are of local importance. According to the Decree of the President of Ukraine [6], the Small Polissia National Nature Park was established in Khmelnytskyy Oblast. River valleys and the Netishyn cooling pond conditionally lay out the boundaries of the national park (about 8762.7 hectares). In the north, there is the Goryn River and the cooling pond; in the east, the Goryn River; in the northwest, the Vilia River; in the south, the tributaries of the Goryn River and the Vilia River. Most of the southern and southeastern zone of the NPP is included in the territory of the mentioned national park.

1.2 Objectives of the planned activity

The objective of the planned activity is the construction of power units 5 and 6 with the AP1000 reactor unit at the KhNPP site.

The planned activity means the construction of power units No. 5 and No. 6 at the KhNPP site using the technical characteristics of the AP1000 type reactor plant of the Westinghouse Electric Company for the purpose of further operation and generation of electricity.

In accordance with the law, the decision to carry out this planned activity will be made after the Verkhovna Rada adopts the Law of Ukraine "On Siting, Design and Construction of Khmelnytskyy NPP Units 5 and 6".

1.3 Description of the characteristics of the activity during preparatory and construction works and implementation of the planned activity, including (if necessary) dismantling works, and the need (restrictions) on the use of land plots during preparatory and construction works and implementation of the planned activity

The preparatory and construction works were characterized in terms of the location, duration, content, fixed assets and work technologies, planned temporary structures, transport and utility networks, natural resource requirements, management of construction and other solid waste, as well as liquid waste, compliance with air quality standards (hygiene standards) and maximum permissible noise levels in relation to the nearest residential development [5], taking into account the relevant regulations [7, 8].

1.3.1 Worksite

The KhNPP site is located in Shepetivka district of Khmelnytskyy region on the left bank of the Goryn River. The town of Netishyn is located to the north of the site . To the northwest of the site is the city of Rivne (regional center of Rivne oblast).

The construction site of KhNPP Units 5 and 6 was selected on the territory adjacent to the existing KhNPP industrial site.



KhNPP site area and the boundaries of its supervised areas are shown in Figure 1.1.

1.3.2 Duration of work performance

The target period for the construction of power units 5, 6 [10, 11] with the selected set of drive machines before the startup of the second stage (power unit 5) is 108 months (9 years) based on the following operating mode: number of shifts -2; shift duration -8.0 hours; number of working days per month -21.0.

The recommended construction duration is determined taking into account the combination of periods and stages, including:

- preparatory period 12 months;
- construction of the first stage 102 months (unit No 6);
- construction of the second stage -102 months (unit No. 5).

The construction of the first stage is combined with the preparatory period and begins 6 months after its start.

The introduction of the second stage of construction -6 months.

Units 6 and 5 will be commissioned after setting up and adjustment works that last 30 months.

1.3.3 Content, main assets and technologies of work performance

The following preparatory work must be completed before the start of the main construction and installation works [11]:

- restoration of the geodetic basis for the construction site;
- allocation of land plots that are not available;

- relocation of construction, installation and other specialized organizations, staffed with the required number of specialists of various profiles and construction equipment, to the place of work;

- ensuring that construction equipment can move and transport construction cargoes of maximum weight and dimensions on existing off-site and on-site roads and railways;

- construction of additional temporary on-site roads required for construction;

- providing construction with energy resources, water, heat, and communications (installation of temporary utility networks and their connection to the networks of the corresponding name after obtaining the necessary technical specifications);

– providing construction with temporary buildings and structures in the required quantity and placing them in specially designated areas.

Excavations for pits of individual buildings and structures, vertical layout, as well as backfilling of pit sinuses and vertical mounds, under roads and other structures represent earthworks.

Excavation of pits for buildings and structures should be carried out in accordance with [12, 13] using backhoe excavators with a bucket capacity of 0.50-1.50 m³, loaded into dump trucks with a carrying capacity of 10-20 tons and transported to the worked quarries at a distance of up to 3 km and to a temporary dump at a distance of up to 1 km.

It is advisable to excavate trenches and swamps using backhoe excavators with a bucket capacity of $0.25-0.30 \text{ m}^3$.

Bulldozers are used for cutting, moving, unloading and placing soil in zero-cycle operations.

Various compaction methods are used, which can be used individually or in combination with each other during performing soil compaction. Similar operations are required when creating a sand cushion for the foundation and a crushed stone base for roads. Combined compaction methods include rolling with vibration or tamping with vibration. The choice of roller for the respective application depends on many factors, such as the required compaction ratio of the road mixture, compaction speed, quality and type of material used.

The list of machines and mechanisms for earthworks is given in Table 1.1.

Item	Model	Quantity	Note
Backhoe excavator	EO-4121	8	Bucket capacity 1.0 m ³
Backhoe excavator	EO -4321	5	Bucket capacity 0,65 m ³
Universal excavator	EO -2621	4	Bucket capacity 0,25 m ³
Bulldozer	DZ-53 or DZ-54	5	Engine power 82 kW
Bulldozer	DZ 3-9 or DZ -33	4	Engine power 129 kW
Bulldozer	DZ -34	2	Engine power 228 kW
Front loader	TO-18A	1	Payload capacity 3.0 tons
Trailed roller on pneumatic tires	DU-30	1	Weight 12.5 t
Static pneumatic wheeled roller	DU -55	1	Weight 20,0 t
Sidewalk roller	BV-76	2	
Scraper	DZ-33	2	Bucket capacity 3.0 m ³
Motor grader	DZ-143	1	
Motor grader	DZ-122A	1	
Motor grader	DZ-180	1	
Asphalt paver	DS-143	1	

Table 1.1 - List of main machines and mechanisms for earthworks and road construction

Item	Model	Quantity	Note
Asphalt paver	DS-195	1	
Asphalt sprayer	DS-396	2	
Bitumen truck	DS-138	2	
Bitumen truck	DS-39Б	2	
Hinged rammers		3	Weight 1,3 t
Tractor-trailer	T-100	2	

All main buildings and structures are made of monolithic and precast reinforced concrete. Reinforcement frames and grids are manufactured in the reinforcement shop located on the construction base. Delivery is carried out by motor vehicle.

Concreting of foundation slabs, walls, and floors is carried out using concrete pumping equipment and tubs.

The main large processes in the construction of the KhNPP units 5 and 6 buildings are the assembly and concreting of the containment and shield building structural modules.

Concreting is carried out mainly [14] with the use of modern concrete pumping equipment with concrete spreading booms, which ensures the intensity of concrete mix placement of 80-100 m³/hour at a feeding height of up to 100 m.

To produce extra-heavy concrete with an average density of 2950, 3350 and 4000 kg/m3, aggregates from magnetite or hematite iron ores are used. Sand made from metallurgical scale, a waste product of the steel industry, can also be used as a fine aggregate.

The concrete mix is placed by gravity feeding from the hopper through a vertical pipe (concrete pipe) extending above the formwork surface to the bottom of the structure. As the concrete flows out of the bottom of the pipe, more concrete is added to the hopper, so that the pipe is constantly filled with new mix. The main purpose of this method of concreting is to place the mixture in its final position with as little interference as possible. A concrete mix with sufficient mobility is required to be easily fed from a pipe whose end is immersed in the mass. Gradually, the mixture flows to the edges, filling the molds, and as the concrete accumulates, the pipe is raised so that its outlet end is recessed by about 1 meter.

Extra heavy concrete is compacted with flexible shaft vibrators.

Specialized vehicles based on truck chassis - concrete mixer trucks - are used to transport ready-mixed concrete at a distance of more than 1 km from mixing plants and factories to the construction site, equipped with technological equipment to prevent losses and maintain the quality of the mixtures on the road.

The construction of the steel cylindrical containment vessel and the shield building is carried out using assembly blocks [15] (bottom, rings, and dome part). The assembly of the blocks is carried out on special stands, after which they are transported to the area of the assembly crane by multi-axis heavy-duty conveyors. Before lifting and installing the blocks in the design position, transition platforms (in the area of horizontal joints), hinged ladders (in the area of vertical joints), fencing and formwork guides are attached to them, reducing the main and secondary work operations performed directly at height as much as possible. The choice of the size and weight of the assembly units is related to the adopted mechanization scheme and transportation capabilities.

The finished tier of the shield building is concreted after completion of the installation and welding work around the entire perimeter of the shell. The installation of elements or blocks of the next tier is started after a technological break to gain concrete strength.

A large lift capacity LTL 3000 Transi-lift crane will carry out the installation of the main building modules CA, CB, CH (Table 1.2).

Module / Item	Weight, t	Required workspace (length/width/height), m
CA-01	1069,353	30,5x21,4x21,4
CA-02	49,885	9,2x6x12
CA-03	213,145	9,2x19x12
CA-05	75,281	15,3x15,3x7,7
CA-20	887,953	30,5x21,4x21,4
CA-22	16,326	17x5x0,6
CA-31	34,466	8x7,5x0,6
CA-32	2,721	3,5x2,5x2
CA-33	22,675	12,5x6,7x1,2
CA-34	10,884	4,8x7,6x0,9
CA-35	31,745	6,7x6,7x0,9
CA-36	9,07	3,7x3x1,2
CA-37	39,001	10,7x10,7x1,2
CA-41	11,791	4,8x11,6x0,6
CA-42	10,884	4,8x11,6x0,6
CA-44	11,791	4,8x11,6x0,6
CA-45	11,791	4,8x11,6x0,6
CA-51	27,21	10,4x11,3x0,6
CA-52	19,047	7x12,8x0,6
CA-55	76,188	12,2x12,2x0,6
CA-56	13,605	11x5,5x0,6
CA-57	19,954	10,7x9,2x0,9
CA-58	13,605	9,2x9,2x0,9
CB-11	1,814	6x2,5x0,9
CB-12	1,814	6x2,2x0,9
CB-20	294,775	21,4x21,4x10,7
CH-80	362,8	16,2x10,4x24
CH-81A	145,12	16,2x0,6x24
CH-81B	145,12	16,2x0,6x24
CH-81C	145,12	16,2x0,6x24
CH-82	390,01	16,2x12,2x24

Table 1.2 – Dimensions and weights of AP1000 components

Module / Item	Weight, t	Required workspace (length/width/height), m
Schield Building Roof	1121,052	Diameter 44,2
Shield Building Steel Panel Assemblies	19,95 - 48,97	12 panels / rings
Condenser Lower Bodies	580,48	14,3x9,2x12,2
Condenser Upper Bodies	126,98 14,3x9,5x7	
TG Roof 1-CH-85	263,03	36,5x32,9x5,8
TG Roof 2-CH-86	176,865	36,5x19,5x5,8
TG Pedestal Deck-CA-81	268,472	16,2x36x4,6
Steam generator	680,25	24x6,4x5,5
CV Ring 1	703,832	Diameter 48,8
CV Ring 2	675,715	Diameter 48,8
CV Ring 3	694,762	Diameter 48,8
CVTH (Top Head)	585,922	Diameter 48,8
CVBH (Bottom Head)	601,341	Diameter 48,8
Gen Stator 60 Hz	410,871	11x4,3x4,5
Reactor cover assembly (IHP)	235,82	7,6x7,6x15,3
Deaerator	273,007	44,2x5,2x5,5
CR-10	390,01	Diameter 48,8
Reactor Vessel	331,055	10x6,4x6,4
Moisture Separator	260,309	13,7x1,5x1,5
Pressurizer (MV20)	107,026	13,7x2,7x3,3
Main transformer 230 Kv	175,051	7,9x4,2x5,8
Polar crane	107,026 37,8x12,8x8,5	

Pre-assembly of the modules is carried out from steel structures of submodules (of the original grades) in a temporary building equipped with three overhead cranes with a lifting capacity of 50 tons (2 pcs.) and 30 tons (1 pc.). For the consolidation assembly, a cantilever lift is used to bring the submodules into a vertical position before placing them on the assembly platform, which is a:

– a template to ensure proper alignment of the submodules with the required tolerances during the assembly process;

– a means of interaction with multi-axis heavy-duty conveyors when the module assembly is completed.

The assembled modules are transported to the crane platform area by heavy-duty transporters via a special road for cargo weighing 1500 tons with a width of 50 meters, after which the lifting loops are connected to the rigging, which is a traverse beam (crossbeams).

The structural modules are precisely positioned in the containment on the base plates (embedded parts) using guide pins and hydraulic jacks.

The structural modules installation is completed by making the connections to the base plates.

Two heavy-lifting cranes are required for the construction of the reactor and turbine islands [16]. The heavy components include the containment vessel cover and rings, large

steel modular structures, large building frame modules, equipment modules, primary circuit components, and turbine generator components.

The heavy-duty cranes will have T-platforms or ring rails on the west side of the containment building from which they will be able to deliver heavy loads to the reactor building and turbine building.

The number and the leading cranes capacity required to support construction work on the critical path is determined based on research and is shown in Table 1.3. The list of auxiliary machines and mechanisms is given in Table 1.4 [16].

Vendor	Quantity (Units)	Model	Location	Lift capacity	Notes
Lampson	2	LTL 3000 Transi-lift	Heavy components of units 5 and 6	3000 t	Heavy lift crane
Demag	1	CC-2800 Superlift	North end of turbine building	660 t	Crawler crane serving both units
Manitowoc	2	4100 Crawler	MAB, CV, shield wall assembly	230 t	Crawler crane
Demag	1	Truck crane AC 100	Workshops, warehouse	100 t	Crawler crane - site- specific (additional cranes may be required)
Liebherr	8	280 EC H12 Litronics (70 model)	Cooling towers, CV, special building, turbine compartment	3 t with a max. outreach of 70 m	Tower Crane – Max lift 13 t: six 60 m reach, two 70 m reach
Liebherr	3	Telescopic truck crane LTM series	As needed	75 – 100 t	Mobile Crane
Sarens	6	Goldhofer SPMT	Various	15000 t per axle	Self-propelled Modular Transporter (SPMT)

Table 1.3 – List of the main lifting mechanisms

Table 1.4 – List of main vehicles and specialized vehicles

Item	Title	Quantity	Note
Dump trucks		50	Load capacity 8,0 - 27 t
Flatbed vehicles		40	Load capacity 5,0 - 12 t
Tractor truck		1	Load capacity 130 t
Trailer		1	Load capacity 130 t
Tractor truck		10	Load capacity 8,0 - 70 t
Trailer		10	Load capacity 8,0 - 70 t
Fuel truck		3	
Tanker trucks		22	
Drilling and crane machine	BM-305A	2	Tractor-based ДТ-75
Cement trucks			Load capacity 8,0 - 20 t
Concrete mixers		15	Сарасіту 4,0-6,0 м3
Concrete pumps		4	Productivity 80 м3/ч

Diesel shunting locomotive	4	Railway transportation
Platforms	14	Rail transport, load capacity 60.0 to
		200.0 t
Multi-axle heavy-duty conveyor		Load capacity 1000 t
Buses	30	For transportation of employees, with
		60 seats

1.3.4 Supply chain for the planned activities

1.3.4.1 Hydraulic facilities

The service water supply scheme of the designed power units No. 5 and 6 is reversed with the cooling towers as radiators. The KhNPP cooling pond in the adopted cooling and process water supply scheme for power units 5 and 6 is a source of water intake for cooling tower feeding, blowdown, and the other process system needs; it is also used for blowdown water discharge. The water supply scheme is unified and separate for each power unit.

The designed systems involved in the transportation and cooling of equipment at power units 5 and 6 using AP 1000 technology include SWS, CWS, and RWS systems:

- the pumps of the power units 5 and 6 pumping station of the Raw Water System (RWS) are used to intake and transport water for blowdown and makeup of the SWS and CWS systems, as well as for the needs of other systems;

- the Service Water System (SWS) removes heat energy from the primary circuit equipment and selected secondary plant circuit equipment, which is transferred from the heat exchangers of the Component Cooling Water system (CCS) to the atmosphere through fan cooling towers;

- the Circulation Water System (CWS) supplies water to the main condenser, the heat exchangers of the Turbine Building Closed Cooling Water System (TCS), and the sealing water heat exchangers of the Condenser Air Removal System (CMS) vacuum pump under various plant power load conditions and design weather conditions.

Hydraulic facilities that can be used during the power unit operation with AP 1000:

– a cooling pond;

– an earthen water retention dam with a flood spillway;

- drainage channel for filtration water discharge and filtration water return pumping station (FWRPS);

cooling water supply and discharge canals;

- water intake canal from the Goryn River and supplementary water pumping station (SWPS).

A cooling pond

The KhNPP cooling pond is a combined type pond – a flood pond from the Goryn River and a channel pond on the Hnylyi Rih River. It is 5 km south of Netishyn, Netishyn Territorial Community, and Khmelnytskyy Oblast. The dam is located three kilometers from the mouth of the Hnylyi Rih River.



To ensure the design temperature in the cooling pond and to increase the efficiency of cooling the circulating water, the construction of a 1.300 m jet-directing dam is planned, which should be carried out at the stage of construction of units No. 3, 4 according to a separate project.

Enclosure dam

Enclosure dam of the cooling pond - sand, soil, alluvial, trapezoidal profile. The dam is located three kilometers from the mouth of the Hnylyi Rih River. The dam is 7124.0 meters long, 8.0 meters wide along the crest, and the crest elevation is 206.0 meters. Monolithic reinforced concrete slabs 200 mm thick support the upstream slope of the dam, while the downstream slope is sown with perennial grasses. The dam's downstream slope is protected by a 600 mm thick sloping drainage made of stone rubble. A drainage channel is installed in the downstream area to intercept and organize the water drainage that filters through the dam body and base.

A flood spillway is on the Sta 14+00 to purge the reservoir and water discharge through the bottom outlet. The water is partially released during floods of exceptional recurrence, coinciding with periods of full reservoir filling. The flood spillway consists of two shafts, three spillway tunnels with a cross-section of 3.0×3.0 m, a spillway in the lower reaches, and a breakwater in the upper reaches. The flow rate of the flood spillway at 0.01% of the capacity is 110.0 m³/s.

A piezometric network consisting of 15 piezometric shafts with 72 piezometers controls the filtration regime in the body and base of the dam.

During the EIA report development, the observation materials presented in the letter No. 4292 dated 09.10.2023 to the Khmelnytskyi NPP subdivision (see Appendix J) were used to analyze the filtration process in the body and base of the cooling pond enclosure dam. The reservoir operation is completely safe if the dam's operational condition is assessed as normal (reliable), i.e. the value of the water level marks in the piezometer does not exceed the maximum operational level. When comparing the provided actual water levels values in piezometers in 2021 and 2022 with the maximum permissible

values determined in the work "Clarification of permissible operational values of the depression curve position in the earthen dam of Khmelnytskyi NPP UDC 626.862.3:532.001", performed by the National University of Water and Environmental Engineering (NUWEE), Khmelnytskyy, Ukraine. Rivne in 2014, it can be noted that water levels are increased:

in piezometers No. 46, No. 50, No. 51 located on the downstream slope of the dam;

in piezometers No. 5, No. 40, No. 69 located behind the drainage channel.

There is partial siltation of the drainage channel and damage to the dam's sloping drainage. The cause of this damage is the shrub root system spread and the digging animals activities.

To recover the dam's filtration capacity, it is necessary to repair the reinforced concrete screen of the dam's upstream slope, clear the drainage channel, and periodically cut down shrubs.

The soil dam filtration regime monitoring and its sole should be carried out according to the Operating Manual "System VU". Hydraulic facilities of the circulating water supply system of KhNPP. The flow rates and filtration volumes are recorded based on the flow meters of the filtration water pumping station. In general, the dam's condition is satisfactory.

Supply and discharge open channel

The water supply channel delivers water for cooling from the reservoir to the unit pumping station (UPS). The trapezoidal channel with a capacity of 266.1 m³/s is 2.8 km long, 10.0 to 30.0 m wide along the bottom, 9.0 m deep, and has a 1:3 slope; monolithic reinforced concrete slabs support the slopes. The supply water channel bottom is 194.0 m.

The discharge canal with facilities has a trapezoidal cross-sectional shape. The channel's capacity is 266.1 m³/s. The length of the canal is 3.3 km, the width along the bottom is from 9.0 m to 23.0 m, the depth is from 5.5 to 6.5 m, and the slopes are 1:3. The discharge channel bottom is 199.5 m.

Drainage channel and filtration water pumping station (FWPS)

The drainage channel intercepts the reservoir filtration water that passes through the dam body in depression curves. Part of the drainage channel's filtration water is intercepted by the filtration water pumping station (FWPS) and directed back to the reservoir. The inclined drainage is used to support the drainage channel and berm.

The filtration water pumping station (FWPS) is a reinforced concrete submersible. The main four pumps are D2000-21-2 for pumping filtration water into the reservoir with a flow rate of 2000 m^3/h .

Water supply channel from the Goryn River

The water supply channel from the Goryn River delivers the water to the Supplemental Water Pumping Station. The channel length is 2.4 km, the width is $16.0\div40.0$ m, and the maximum water velocity is 0.32 m/s.

Supplemental water pumping station (SWPS)

It recharges the cooling pond to replenish irrecoverable losses in the service water consumption system. It is located in the lower pool near the dam at Sta 65. The maximum design flow rate is $30.0 \text{ m}^3/\text{s}$.

The supplemental water pumping station (SWPS) is equipped with pumps:

- OV2-110K (OB2-110K) with a flow rate of 18000.0 m³/h, 5 pcs;
- D5000-32A (Д5000-32A) with a flow rate of 3200.0 m³/h, 2 pcs;
- 550D-22 (550Д-22) with a flow rate of 2000.0 m³/h, 2 pcs.

1.3.4.2 Transport network

The territory of the KhNPP's industrial area and the construction site is densely covered with roads and railways, which will be partially used during the commissioning and construction of units 5 and 6 using the AP1000 technology.

The width of the roads is 6-8 meters. The project provides for the further use of approximately 40% of the existing road transport network of the KhNPP construction site. Roads on the territory of the industrial site of power units 1-4 are not used in the operation of power 5-6, their route is not disturbed. The proposed roads are mostly in satisfactory condition with minor defects: crumbling edges, potholes, and pits in the places of concrete joints.

Rail transport is represented by railway lines serving the industrial area of power units 1-4 and those used for loading and unloading operations at the construction yard.

The lines related to the industrial area are mostly in satisfactory condition and require minor repairs. In general, these lines are not used in the operation of units 5 and 6, but they are preserved and continue to maintain units 1-4.

The loading and unloading of the construction yard lines are in poor condition. There is no direct connection with the external railroad network due to the dismantling of the connecting line. Significant sections of tracks, signaling, and switching equipment are missing. Wooden sleepers are destroyed or affected by natural decomposition. The railroad network is not expected to be used, and its restoration is economically unviable due to the dismantling of the warehouses they served.

1.3.5 Planned temporary structures, transportation and utility networks

An indicative list of temporary buildings and structures required for the construction of power units 5 and 6 is given below [16].

1.3.5.1 Building materials laydown areas

Building materials laydown areas include the following buildings:

– Warehouse. Three buildings measuring 40 x 20-meter. They provide storage of ready-to-use materials. Depending on the purpose, they can be prefabricated metal or metal-frame awning sheds;

– Enclosed shop for concreting tools and devices. The 40 x20-meter building. This building is used for storage, repair and modernization of equipment. Power, lighting, water, sewerage and compressed air must be provided;

- *Carpentry laydown area.* A 60 x 20-meter site, along with a carpentry shop for storing the manufactured structures;

– Formwork laydown area. A 60 x 40-meter site area to store formwork elements and equipment for its assembly;

– Rigging storage. A 28 x 20-meter warehouse building for storing, testing and calibrating equipment;

- *Small equipment and tools storage*. A 28 x 20-meter warehouse building for storing, testing and calibrating equipment;

– Laydown Areas. A complex of 16 sites made of compacted gravel with a slope towards the reservoir with a total area of 51.63 hectares for storing any equipment.

1.3.5.2 Fabrication shops

The fabrication shops will include the following ones:

- *Carpentry shop.* A 30 x 20-meter carpentry shop is set up at the construction site to manufacture additional scaffolding and building structures;

- Formwork wash area. A 15 x 10-meter site area for cleaning the formwork elements after use;

- Paint shop. A 26 x 13-meter building for storing, mixing and calibrating equipment;

– Duct fabrication shop. A building measuring 26 x 13-meter for the manufacture of air ducts for construction;

- Insulation fabrication shop. A building measuring 26 x 13-meter;

- Small diameter piping shop. A building measuring 50 x 25-meter for the manufacture of pipelines for construction;

– Large diameter piping shop. A 50 x 25-meter building for the manufacture of pipelines supporting the construction;

- *Maintenance shop.* A building measuring 18x12-meter for equipment maintenance;

– Fab and storage. A 26 x 13-meter building for the manufacture and storage of air and pipelines for construction;

- *Precast concrete panel fabrication area.* An area measuring 63x25-meter, where reinforcement, concrete placement, vibration and temporary storage take place;

- Weld test facility. A 27 x 25-meter site area where welded joints are tested.

1.3.5.3 Sanitary and administrative rooms

Sanitary and administrative rooms will include:

- *Subcontractors' offices.* A 40 x 24-meter building, where contractors place their engineering and technical staff;

- *Subcontractors' field offices*. Six buildings measuring 49x13 meters, in which contractors place their engineering and technical staff and construction management;

- *Craft training*. The 24 x 23-meter building and is used for occupational safety and health briefings, ongoing training of construction workers and organizational meetings;

– New hire training building. A 29 x 14-meter building housing administrative staff responsible for hiring and training new employees;

- *Construction orientation building*. A 29 x 14-meter building, where new employees are trained with an emphasis on industrial safety, compliance with procedures, and the principles of a safe working environment specific to the facility;

– First aid / Medical Facility. The 29 x 14-meter building and is used for medical examinations of employees and first aid;

- *Canteen for construction workers*. The 50 x 25-meter building and is equipped with kitchen accessories, places for eating and a sanitary unit;

– Craft Change / Lunch Facilities. The 50 x 25-meter building to provide a place for gathering the workforce, receiving daily work instructions before work, changing into work clothes and personal protective equipment, and can also serve as a place for lunch breaks and heating (protection from solar radiation) for workers;

- The 50x25-meter shower building includes a room for women's personal hygiene and a drying room for clothes and shoes;

- *Craft toilet trailer (men) is* a complex of 6-10 portable restrooms measuring 13 x 6 meters. Such complexes are located on the site territory, taking into account the saturation of each zone with workers;

- *Craft toilet trailer (women)* is a complex of 6-10 portable toilets measuring 13 x 6 meters

- *Construction administration office.* The 40x24-meter building and houses the construction administration, offices of the general contractor and general designer;

- Time alley. A 29×14-meter building that houses turnstiles, an access control point and time readers;

- Refueling station in the construction zone. The 21 x 16-meter building is designed to refuel construction vehicles. It contains two 55,000-liter tanks with gasoline and diesel fuel. The filling station is to be provided with power supply, fire protection systems, lighting, ventilation, sewerage and water supply. Upon completion of construction, the gas station will be completely dismantled;

– Truck parking. The 138x64-meter site is intended for the storage of construction vehicles. It is located next to a maintenance station and a wash facility;

- *Truck wash facility.* A 24x15-meter area designed to wash construction vehicles after completion of work. It is located next to the maintenance facility site and truck parking;

- *Construction site maintenance facility.* The 29 x 14-meter building is intended for the maintenance of construction vehicles and equipment;

- *Heavy maintenance vehicle parking*. A 29x30-meter area designed for the parking of construction vehicles being repaired at the maintenance facility site. It is located adjacent to the maintenance facility site.

1.3.5.4 Batch plant

Production area of 3.3 hectares. It needs to be provided with power, sewerage, water supply and communications. The batch plant includes the following buildings:

- test laboratories for testing concrete and reinforcement;
- rock breaker;
- concrete plant water tanks;
- concrete installation site;
- concrete plant sump;
- unit to produce ice for concrete;
- quality control building.

1.3.5.5 Assembly pads sites

Assembly pads include:

- Containment Vessel (CV) module assembly pad. The main site is located next to the road for heavy vehicles, that is a complex of five round sites with a total size of 325x82 meters;

- CV plate laydown area. The covered-with-compacted gravel site size is 325x75 meters. It is placed near the containment assembly area;

- CA Sub-assembly Storage Area. Two compacted gravel sites measuring 80x46 m and 97x56 m are located next to the MAB and are the main storage/preparation area for submodules arriving on site awaiting final assembly inside the MAB or at the external final assembly area;

- CA20 Assembly area. The site measures 66x25 meters. It will be covered with compacted gravel. It is located next to the module assembly building. It is used for assembly and storage of finished CA20 modules;

- Module Assembly building - MAB. The building measures 101x47 meters and its height is 32 meters. The floor is to be covered with a 1.2 m thick concrete slab. It is used for the final assembly of large structural modules. It is located near the module assembly site;

- CA01 Assembly area. The site is 80x58 meters in size. The site is located next to the MAB module assembly building. The site is intended for pre-assembly/preparation of the CA01 module before its delivery to the module assembly building;

– Shield building fabrication / assembly. The site is 277x92 meters in size. It is located next to the main heavy hault road;

- Turbine building module assembly. The building measures 151x75 m. It is located next to the main heavy hault road. This building is used for assembling the turbine building components and preparing them for installation;

- CR10 laydown. The site measures 70x60 meters. It is located next to the main heavy hault road. This site is used to assemble the CR-10, CA-03 modules, the roof of the shield building and prepare for their installation. The site is to be covered with concrete;

- condenser laydown;
- gas industry (sector).

A special 50-meter-wide road for 1,500-ton loads is being built to supply goods for installation by multi-axle heavy-duty transporters, adjacent to the industrial site.

1.3.6 Natural resource needs for preparatory and construction works and sources of these needs

The demand for structures, semi-finished products and materials can be satisfied from geographically close areas, taking into account the transportation schemes of the following enterprises:

- prefabricated concrete structures - local reinforced concrete plants, Kyiv reinforced concrete plants, Svitlovodsk Reinforced-Concrete Wares Plant LLC, and Pivdenoukrainskyi HBC;

– metal-rolled products - Zaporizhstal PJSC, Zaporizhzhia; Dnipro Metallurgical Plant PJSC; Spetssplav LLC, Kryvyi Rih.

– Rebar – ArcelorMittal Kryvyi Rih Public Joint Stock Company (ArcelorMittal Kryvyi Rih PJSC ArcelorMittal Kryvyi Rih) KAMET STEEL, Kamianske town.

- special reinforced concrete and special steel structures - production facilities for the manufacture of non-standard products, enlargement of heavy equipment and structures at Khmelnytskyy NPP (after their reconstruction);

– PrJSC Dickergoff Cement Ukraine (Volyn-Cement branch office, Zdolbuniv; Yugcement branch office, Olshanske); Cemark Ukraine (JSC Podilsky Cement, Humentsi, Khmelnytskyy region, LLC Cement, Odesa; PrJSC Mykolaivcement, Mykolaiv, Lviv region);

- crushed stone and granite - Shepetivka Granite Quarry PRONEX LLC (Sudylkove village, Shepetivka district, Khmelnytskyy region), Shepetivka Granite Quarry LLC (Rudnia-Novenka village, Shepetivka district, Khmelnytskyy region), Polonne Mining Plant LLC (Polonne, Khmelnytskyy region), Korosten Crushed Stone Plant ALC, Zhytomyr region, JSC Malynskyi Stone Crushing Plant (Zhytomyr region), LLC Hnivanskyi Granite Quarry (Vinnytsia region);

- sand for concrete, backfill and quality fill – Slavutskyi Sand Quarry, Khmelnytskyy region, Khmelnytskyy NPP Construction Management PJSC, Starytsia-2 quarry (Netishyn, Khmelnytskyy region), quarries in Kyiv region, Ukrtransbud Corporation, Kyiv and Chernihiv regions;

- concrete and mortars - it is planned to build a temporary concrete plant for the construction period, located at the construction and assembly base.

The construction energy and water needs are shown in Table 1.5.

Table 1.5 - List of energy and water needs for the construction of KhNPP Units 5 and 6 $\,$

Name	Value
Electricity (installed capacity of current collectors), kVA	6400
Hot water supply, Gcal/hour	37
Technical steam, t/h	11,5
Potable water, m ^{3/h}	48
Service water, m ^{3/h}	24
Oxygen, m ³ /hour	130
Acetylene, m ^{3/h}	16
Propane-butane, m ^{3 /hour}	32
Argon,m ³ / unit	80000
Carbon dioxide, ^{m3} / unit	94000
Freon, ^{m3} / unit	Freon,m ³ / unit
Compressed air, m ^{3/min}	71,0

1.3.7 Management of construction and other solid waste, as well as liquid waste (wastewater and sewage sludge) generated during preparatory and construction activities

The project provides for the following measures to reduce the negative impact of construction production on the environment [17].

Cutting and uprooting of trees falling into the construction zone must be compensated by planting new trees in the same number in the adjacent areas. During assembly, the movement of goods must be performed at a distance of at least 0.5 m from the crown and trunks of trees. The destruction of tree and shrub vegetation and the filling of root necks and tree trunks with soil not provided for by the project are not allowed on the territory of the construction site.

Waste concrete, bricks, insulation, polymeric materials, asphalt, etc. must be separated by type and disposed of after crushing and fractionation.

Regularly remove construction waste from the construction site to a specially designated landfill – solid domestic waste landfill of the public utility Netishyn City Council "Housing and Utility Community".

Soil dumping shall be arranged exclusively in specially designated areas.

Drain fuel and lubricants in specially designated and equipped areas.

During construction and assembly activities, it is prohibited to discharge wastewater, as well as untreated household or industrial wastewater generated at or near the construction site. It is envisaged to transfer the wastewater for further treatment to the existing wastewater treatment facilities.

Use mobile toilet cabins (MTC) with a storage capacity of no more than 250 liters. In the case of pumping at the construction site, the locations of liquid waste removal must be established in the work performance plan before the start of work. The distance between the bottom of the watertight tank of the MTC and the groundwater level should be at least 0.5 m.

Serviceable technical means are used to transport concrete and mortar, that prevent their loss on the road.

1.3.8 Compliance with air quality standards (hygienic standards) and maximum permissible noise levels in relation to the nearest residential buildings

All construction work will be carried out in compliance with the requirements for the prevention of dust formation and air basin pollution [17]:

- construction waste will be loaded into closed containers with preliminary irrigation, preventing dusting of the territory;

- loading, transportation and storage of bulk dusting materials to be carried out using special mechanisms, machines and closed containers;

– in summer, all roads and road-type sites will be regularly watered;

- at the exit from the construction site, set up a wheel washing station and build a paved area to clean the wheels of vehicles and construction machines from dirt.

The following actions are envisaged to reduce the noise level during construction and assembly activities:

- ensure that vehicle engines are turned off while on the construction site;

exclude loudspeaker communication;

- to exclude the performance of works accompanied by noise over the permissible norm;

- to exclude the operation of equipment that has a noise and vibration level exceeding the permissible standards;

- to soundproof construction machinery engines, use protective covers and hoods with multilayer coating;

- to isolate local sources, use temporary noise protection screens, noise curtains, and tents (for example, place compressors in a sound-absorbing tent).

1.4 Description of the main characteristics of the planned activity (including manufacturing processes), e.g., the type and quantity of materials and natural resources (water, land, soil, biodiversity) to be used

The planned activity is characterized in terms of the presence of hazardous substances that may be used or manufactured, processed, stored, or transported, the need for raw materials, fuel, and other materials, water in appropriate units of measurement, etc.

1.4.1 Characteristics of the NPP site area

The NPP site and off-site facilities are located in the Shepetivka district of Khmelnytskyy region of Ukraine.

Within the 30-kilometer zone of the KhNPP site are the 157-kilometer-long Goryn River and its tributaries with a total length of 450 kilometers. The main ones are the Vilia River, the upper reaches of the Ustia River, and the Hnylyi Rih River. There are many reservoirs, ponds, and canals within the 30-kilometer observation zone of the KhNPP. These artificial reservoirs were created under conditions of uneven flow to regulate it and provide water for the needs of the national economy and population.

In total, there are 216 ponds within the 30-kilometer zone of the KhNPP,

Three reservoirs (Myslyatyn, Izyaslav and KhNPP cooling pond) and 1290 canals. The sources of service water supply for the NPP are the Goryn and Hnylyi Rih rivers. An artificial cooling pond is directly adjacent to the industrial site and is used to cool the main process equipment of the NPP.

Groundwater is widespread on the territory.

The territory taken up for NPP includes the following facilities:

- industrial site;
- construction yard.

Cooling towers and water intake facilities are located on the western side of the site. The NPP site is connected to off-site facilities by a network of paved roads.

Key indicators for the master plan (for the built-up area of the industrial site):

_	land plot area	51.1 ha;
_	building area	10.26 hectares;
_	building density	30%;
_	road area	18.06 hectares.

According to the technical report on engineering and geological surveys at the facility "Construction of power units 5, 6 with AP1000 reactor unit at the KhNPP site." No. 1509/4254 dated 2024, performed by Geobest Engineering Center LLC, the works site based on the set of factors specified in SBC A.2.1-1-2008 belongs to the III category of complexity of engineering and geological conditions. Soil strata are represented by siltstone-argillite strata covered with loams. Alluvial sands represent topsoil.

Karst phenomena and landslides do not characterize the design activity area.

In addition, during the preparation of the territories and the construction of power units No. 1 and No. 2, peat-containing soils were replaced with alluvial sand throughout the territory.

The planned activity will provide additional stabilization and site protection from water and wind erosion by installing coverings, sowing soil areas with lawn grasses, organizing surface water drainage into a closed sewerage network, and planting trees. After finishing the landscaping works, no unstable slopes are expected within the direct impact area.

There are no existing buildings on the territory for the KhNPP units 5&6, therefore, the report of existing buildings survey is not attached.

Maps are provided in Appendix H.

1.4.2 Technological characteristics of the planned activity

The AP1000 is a double-circuit pressurized water reactor (PWR) that uses a simplified, innovative and efficient approach to safety. The AP1000 design provides clear benefits, including high safety, economic competitiveness, and improved and more efficient operation. See also Figure 1.2.



Figure 1.2 – AP1000 power unit

The AP1000 power unit site consists of five main building structures:

- nuclear island (containment, shield building and auxiliary building);
- turbine island (turbine building, the first bay);
- radioactive waste management building;
- diesel generator building;

- annex building (divided into four zones, 1-3 zones are technological premises,

4 zone is an office building).

1.4.2.1 Technological characteristics of the nuclear island

The layout of the main facilities and components of the AP1000 power unit is shown in Figure 1.3.



Figure 1.3 – The layout of the AP1000 unit main facilities and components

The reactor coolant system (RCS) consists of:

- Reactor;
- Steam Generator;
- Pressurizer;
- Main Coolant Pumps;
- Primary Coolant Pipelines.

The AP1000 RCS is designed to circulate the coolant to transfer heat generated in the reactor core to the secondary side.

The AP1000 RCS provides heat removal by the coolant from the core in all operating modes of the reactor unit.

The AP1000 RCS has two circuits with a single hot and two cold legs for each circuit, two steam generators (SG), to both of which 2 main coolant pumps (MCP) are directly connected (See Figure 1.4).

A cooling water reservoir is planned to be used as a source of service water for KhNPP Units 5 and 6.

The SWS and CWS equipment cooling at the KhNPP units 5&6 is planned with the construction of cooling towers – one tower-type cooling tower for the CWS and one two-section fan cooling tower for each power unit.

The nuclear island systems of the AP1000 unit include the following:

1) Reactor coolant system (RCS)

- 2) Nuclear island process systems:
 - Chemical and Volume Control System (CVS)
 - Primary Sampling System (PSS)
 - Radio-chemistry Laboratory System (RLS)
 - Spent Fuel Pool Cooling System (SFS)
 - Normal Residual Heat Removal System (RNS)
 - Containment Leak Rate Test System (VUS)
 - Steam Generator Blowdown System (BDS)
 - Component Cooling Water System (CCS)
 - Service Water System (SWS)
 - Special Process Heat Tracing System (EHS)
- 3) Safety systems
 - Passive Containment Cooling System (PCS)
 - Passive Core Cooling System (PXS)
 - Containment Hydrogen Control System (VLS)
 - Emergency Habitability System (VES)
 - Containment Isolation System (CNS)
 - Spent fuel pool cooling system (SFS)
- 4) Protection and Safety Monitoring System (PMS)
- 5) Class 1E DC system (IDS)

The main characteristics of the AP1000 power unit are shown in Table 1.6.

Table 1.6 – Main characteristics of	f the AP1000 power unit
-------------------------------------	-------------------------

Option	Values for the AP1000 power unit
Rated electric power	1117 - 1145 MW (the value is variable for each
	site, depending on the turbine unit used, cooling
	configuration, planned load, etc.)
Design operating lifetime of the power unit	60 years
Rated thermal power	3400 MW
Operating pressure in the reactor unit RU	15,51 MPa
Hot loop temperature	321,11 °C
Calculated pressure in the steam generator	8,27 MPa
Temperature of the main feedwater	226,67 °C
C	ore
The number of fuel assemblies (FA), units.	157
Core height	4,267 mm
Fuel assembly (FA) array	17 x 17
Number of test assemblies	53
Number of rods with a burnup absorber in the FA	16

1.4.2.2 Reactor coolant system (RCS)

Safety-Related Functions

The RCS performs and/or supports the following safety-related functions: maintaining the reactor coolant pressure boundary, core cooling and reactivity control,

process monitoring, automatic depressurization, and reactor vessel head emergency letdown/venting. These are described below:

• Reactor coolant pressure boundary (RCPB) – serves as a pressure boundary for containing the reactor coolant, soluble boron, limiting radiation releases (by limiting coolant leakage) to the containment and between the primary and non-radioactive secondary circuit except for plant conditions that postulate a failure of the RCPB. The RCS provides relief capability to prevent the reactor coolant system from overpressure during all plant operating conditions.

• Core cooling and reactivity control – circulates the coolant to remove the generated and decay heat produced in the reactor after it is shutdown, provide an adequately-uniform temperature distribution, and maintain the chemical homogeneity of the reactor coolant when chemical adjustments occur. The RCS provides the coolant circulation and decay heat removal required during the transition from forced circulation to natural circulation. The RCS, in conjunction with the reactor system (RXS), and the passive/active core cooling systems (PXS) contain the soluble neutron poison, which supplements the negative reactivity inserted by the control rods to provide the reactor shutdown subcriticality margin.

• Process monitoring – monitoring of the process parameters within the reactor coolant pressure boundary. The RCS supplies the signals required by the Protection and Safety Monitoring System (PMS) to provide automatic reactor trip and actuation of the passive safety systems.

• Automatic depressurization – the safety-related automatic depressurization system (ADS) function is to automatically depressurize the RCS so that the Passive Core Cooling System can adequately cool the core during small-break loss of coolant accidents (LOCAs).

• Emergency letdown – an emergency letdown system alows to control pressurizer level during accident events associated with an increase in pressurizer water level.

• The RCS provides the capability to vent non-condensable gases that might collect in the pressurizer and reactor vessel head in order to support core cooling capability in accident scenarios in accordance with the United States Nuclear Regulatory Commission (NRC) guidance.

• The RCS provides the capability to depressurize the system to an extent necessary to support the PXS during a non-design basis external event.

• Any of the six ADS 1-3 subsystem flow paths in the RCS provides a manuallyopened vent path large enough, 27.8 cm² (4.15 in²), to relieve water from the RCS to prevent reactor vessel low temperature over-pressurization in the event that the RNS relief valves are not operable when the reactor coolant is less than 135°C (275°F).

RCS Description and Design Data

The RCS consists of two heat transfer circuits, each with a steam generator, two reactor coolant pumps (RCPs), a single hot leg and two cold legs, for circulating reactor coolant between the reactor and the steam generators. In addition, the system includes a pressurizer, interconnecting piping, and valves and instrumentation necessary for

operational control and safeguards actuation. All system equipment is located in the reactor containment.

The RCPs circulate pressurized water through the reactor vessel and the steam generators. The pressurized water, which serves as coolant, moderator, and solvent for boric acid (used for chemical shim control), is heated as it passes through the core. The coolant then flows to the steam generators where the heat is transferred to the secondary side, and then is returned to the reactor by the reactor coolant pumps to repeat the process.

The steam generators have a vertical shell and u-tube configuration with integral moisture separating equipment. The reactor coolant pumps are enhanced-inertia, high-reliability, lowmaintenance, canned-motor pumps and are integrated into the steam generator channel heads in an inverted position. Instrumentation is provided to monitor the following primary loop process parameters as required by the Plant Control System (PLS) and the PMS:

• Loop flow, wide range temperature, and narrow range temperature for each RCS cold leg.

• Level, wide range pressure, wide range temperature, and narrow range temperature for each RCS hot leg.



Figure 1.4 – RCS View

The reactor vessel has two direct vessel injection nozzles entering the downcomer below the level of the hot leg nozzles and cold leg nozzles. Each nozzle is connected to
an accumulator tank, a core makeup tank, and to one of the two branch lines from the common discharge of the RNS.

The reactor coolant loop has connections to the Passive Residual Heat Removal Heat Exchanger, which is part of the PXS. This connection, along with the two core makeup tanks results in three safetyrelated, natural circulation flow paths to the reactor core through the RCS loop piping and PXS components.

Natural circulation would be expected to continue through until plant or equipment conditions change (such as depletion of safety-related batteries) and require actuation of the ADS valves and transition from natural circulation core cooling to safety injection and containment sump recirculation core cooling.

1.4.2.3 Technological characteristics of the turbine island

The final choice of the turbine unit and generator, their manufacturer and supplier will be determined at the project stage.

Currently, the use of a complete turbine island equipment package for KhNPP Units 5 and 6 is envisaged to use the previously obtained initial data of the complete turbine island equipment supply for General Electric. However, this information will be specified at the design stage.

The main equipment of the turbine building for KhNPP Units 5 and 6 is a steam turbine unit of the ARABELLE 1000 type manufactured by General Electric. The turbine unit and generator are assembled on a common foundation.

The main process equipment and systems of the second circuit of the power unit with the AP1000 reactor:

- 1) Turbine unit:
 - Complete Turbine Generator;
 - Gland Seal system (GSS);
 - Generator Hydrogen and CO2 Systems (HCS);
 - Hydrogen Seal Oil System (HSS);
 - Main Turbine and Generator Lub Oil System (LOS);
 - Main Turbine System (MTS);
 - Main Turbine Control and Diagnostic System (TOS);
 - Main Steam System (MSS);
 - Main and Sturtup Feedwater System (FWS).
- 2) Auxiliary systems of the turbine unit:
 - Auxiliary Steam Supply system (ASS);
 - Condensate System (CDS);
 - Condenser Tube Cleaning System (CES);
 - Turbine Island Chemical Feed System (CFS);
 - Condenser Air Removal System (CMS);
 - Condenser Polishing System (CPS);
 - Circulating Water System (CWS);
 - Demineralized Water Treatment System (DTS);
 - Heater Drain system (HDS);

- Secondary Sampling System (SSS);
- Turbine Building Closed Cooling Water System (TCS);
- Turbine Island Vents, Drains and Relief System (TDS);
- Turbine Building Ventilation System (VTS).

1.4.3 Water and energy supply of the construction performance

Quantitative data on the energy and water needs for construction are shown in Table 1.7, are preliminary, and will be refined at the "project" stage.

Table 1.7 – Energy and water needs for the construction of KhNPP Units 5 and 6

Name	Value
Electricity (current collectors installed capacity), kVA	6400
Hot water supply, Gcal/hour	37
Technical steam, t/h	11,5
Potable water, m 3 /h	48
Service water, m 3 /h	24
Oxygen, m 3 /hour	130
Acetylene, m 3 /h	16
Propane-butane, m 3 /h	32
Argon,m3 / unit	80000
Carbon dioxide,m3 / unit	94000
Freon,m3 / unit	23600
Compressed air, m 3 /min	71,0*

Notes. *The specified amount of compressed air corresponds to the compressed air demand at the construction site of power units 5 and 6

The KhNPP industrial site (construction of Units 5 and 6), industrial site (construction yard), and construction site will be supplied with electricity via 10 kV networks from the 10 kV outdoor switchgear at Substation 35/10/6.

The existing utility networks will provide heat supply, water supply and sewage of the KhNPP industrial site, industrial site (construction yard) and construction site.

The demand for oxygen, acetylene, propane-butane and compressed air will be covered by the respective temporary construction site facilities.

1.4.3.1 Preliminary water treatment

The main technological solution for the water pretreatment system is the technology of membrane ultrafiltration; it means that lime is not used. This technology is used to remove suspended solids, colloidal impurities, some organic contaminants from the treated water, as well as to remove bacteria, algae and other microorganisms. The ultrafiltration pretreatment technology has the following advantages compared to the coagulation method with liming in clarifiers and treatment of treated water using mechanical filters:

• High quality of clarified water: suspended solids less than 0.1 mg/l; turbidity less than 0.2 NTU; iron less than 0.1 mg/l; manganese less than 0.1 mg/l;

• No need for lime farming - when operating ultrafiltration plants, only periodic hypochlorite and acid flushing of membrane modules is required;

• No need for precise observance of technological parameters (pH, water flow), as required by the operation of clarifiers - whereas, the quality of water treatment remains consistently high;

• Absence of sludge;

• Purification efficiency is $95.0 \div 96.0\%$, that is comparable to the existing lighting technology;

- Significant reduction of production space for the main and auxiliary equipment;
- Easy operation, high degree of automation of the water treatment process;
- Lifetime of ultrafiltration membrane modules is 8-10 years.

1.4.4 Provision of resources during operation

This section provides information on the provision of power units No. 5 and 6 with the main resources for operation, including water, power and fuel.

1.4.4.1 Provision of service water

The existing KhNPP units 1 and 2 service water supply plan is recirculated with a combined type cooling pond: bulk water from the Goryn River and channel water for the Hnylyi Rih River. The service water supply plan for corresponding consumers is a recirculating one with spray ponds. The main facilities include: a cooling pond, supply and discharge canals, block pumping stations, circulation water pipelines and a supplementary water pumping station.

The cooling pond was created by constructing a water retention dam in the valley of the Hnylyi Rih River. The Khmelnytskyy NPP service water reservoir was designed based on the permissible cooling water temperature (no more than 33°C) to replenish losses and remove heat from NPP equipment, taking into account repair schedules.

Units 1 and 2 are currently in operation. In the existing KhNPP service water supply system, fresh water is used to cool the main and auxiliary equipment, as well as to replenish irrecoverable losses in the plant's process cycles. It is discharged to the reservoir:

- from domestic wastewater treatment plants;
- from the neutralization unit (NU);
- from the oily wastewater treatment plant (OWTP);
- from the sludge pond;
- from spray ponds;
- stormwater;
- from the wash water tank;
- drainage water from the industrial site.

The service water supply system of power units No. 5 and 6 is reversed with the use of cooling towers as radiators and is designed using existing hydraulic facilities. The KhNPP cooling pond is the source of water intake for cooling tower feeding and blowdown and for the needs of other process systems in the adopted cooling and service water supply system for KhNPP Units 5 and 6. The water supply system is unified and separate for each power unit.

The designed systems involved in the transportation and cooling of equipment at Units 5 and 6 using AP1000 technology include the SWS, CWS and RWS systems. The SWS and CWS are used to supply cooling water to the components of the AP1000 reactor and turbine units. The intake from the cooling pond and transportation to the SWS, CWS and technological processes of the Westinghouse Electric Company AP1000 plant is performed using the RWS pumping station pumps of Units 5&6.

To assess the availability and possibility of using the water resources of the Goryn River and the pond under different conditions of water availability, the affiliate SS "Atomprojectengineering", JSC "NNEGC "Energoatom", in 2024, performed a water management calculation to determine the need for service water for cooling the main and auxiliary equipment of units 1-4 and feeding the circulation system of two designed units 5 and 6 with AP1000 reactors [83].

Water management balances have been developed for average and low-water years of 50%, 95% and 97% supply at the current level of manufacturing development with the prospective commissioning of units 3&4 with VVER 1000 reactors and units 5&6 with AP1000 reactors.

As a result of calculations of the water balance of the KhNPP cooling pond in a lowwater year at 95% supply, it was found:

• The annual intake of circulating water for units 1-6 from the cooling pond is 5514.633 million m³, including:

– for cooling the main and auxiliary equipment of units 1-4 and replenishment of irrecoverable losses - 5467.34 million m³;

- for feeding the CWS and chemical treatment system of the AP1000 units 5 and 6 - 47.23 million m^3 ;

• The total annual cooling pond losses are 5607.60 million m³;

• The annual water inflow to the cooling pond is 5559.60 million m³;

• There is a need for additional water intake from the Goryn River for the years of 95% supply in the amount of 30.95 million m^3 according to the authorized limit for water intake from the Goryn River for the KhNPP needs – 33.72 mln.m³.

To substantiate the possibility of meeting the needs of KhNPP for service water supply during the operation of units 5 and 6 under design, in view of the hydrological state and surface water bodies regime located in the area of influence of the planned activity, Vinnytsia National University performed a research work in 2024: "Construction and forecasting of the water balance of the water management site M5.1.4.45 of the Goryn River from the source to the border of Khmelnytskyi and Rivne regions" [84]. The main purpose of this work is to assess the amount of the inflow and release parts of the Goryn River water balance at the current level and at the perspective levels of the KhNPP production activity during the operation of units 1-6, as well as to draw up the water balance of the studied water management area for 50%, 75%, 95% of supply, using the initial data provided by the State Agency of Water Resources of Ukraine, the Central Geophysical Observatory, the Ukrainian Hydrometeorological Center, and the Khmelnytskyi NPP.

Based on the results of these two calculations, we can conclude that the availability of water reserves is sufficient to meet the needs of KhNPP Units 5 and 6 for service additional water from the Goryn River. In a year of 95% availability, the pond is operated according to the scheme of filling the cooling pond in winter and spring with operation in summer months.

To prevent the impact of the planned activities of power units 5 and 6 on the hydrological regime of the Goryn River, it is proposed to use cooling towers with optimal cooling water evaporation losses, regulate water intake from the Goryn River, including the rational operation of the KhNPP units 5 and 6 main equipment, discharge treated industrial, rain and domestic wastewater from power units 5 and 6 into the supply channel for reuse in the NPP cycle, which will reduce the need for additional fresh service water.

1.4.4.2 Provision of potable water

The city of Netishyn and KhNPP are supplied with water through a centralized potable water supply system that also meets the city's firefighting needs. Water is supplied from a water intake located in the northeastern suburb of the town. The source of water intake is non-pressure groundwater. Centrifugal well pumps are used to lift water from artesian wells. Raw water is supplied from artesian wells with a depth of 200-240 meters to the water demineralization station via four collectors. Each artesian well is connected to the main water supply system using water pipes.

The water intake capacity, according to the approved reserves and data provided by the KhNPP utility, is 18 000 m³/day. The actual average load is from 8000 to 10000m³/day. According to the Feasibility Study "Construction of Units 3&4", water consumption for the household and potable needs of Units 3&4 is 317.25 m³/day. According to preliminary calculations, the water consumption for the household and potable needs of units 5&6 is 126.58 m³/day.

The total water consumption for the household and potable needs of the designed KhNPP Units 3-6 is $444.64m^3/day$.

Thus, the design water intake capacity is sufficient to meet the needs of existing and projected consumers.

Commercial accounting of potable water supply is carried out at the off-site demineralization station, which supplies water to the city of Netishyn and through separate water pipelines to the KhNPP site.

Water meters are installed in the designed buildings and structures of power units 5 and 6 at each building inlet where household and potable water supply is provided.

Netishyn is located within the eastern part of the Volyn-Podilskyi artesian basin, in the area of its connection with the Ukrainian crystalline massif.

The existing water intake in Netishyn exploits the Olchedayivska aquifer of the Upper Proterozoic, which is confined to the sandstones of the Horbashivska Formation of the Volynian Series and to the sandstones of the Polissia Series of the Upper Proterozoic.

The closest water intake that uses the same aquifer for centralized water supply is the water intake of Slavutych.

In the area of the wells' location, the Horbashivska Formation aquifer lies at a depth of 150 meters. The thickness of the aquifer is about 35 meters.

Three sanitary protection zones are planned around the wells: Zone I is a "strict" zone; Zones II and III are restricted zones.

Water intake wells are located on the floodplain of the Goryn River. The geological structure of the floodplain consists of modern fourth-quarter alluvial deposits, represented by sands of various sizes, underlain by clays of the Sarmatian layer of the Lower Neogene. The waters of the Gorbashivska Formation aquifer meet the requirements [20] for groundwater sources by the key indicators.

The degree of participation of surface waters in the formation of operational reserves of the water intake aquifer is less than 20%. Thus, as a result of natural filtration (the Horbashivska Formation aquifer is reliably protected by a thick tuff layer of overlying sediments), surface water from the Goryn River does not affect groundwater quality. Operational reserves of 18,000m3/day are secured. Thus, there will be no changes in the surrounding hydrological environment with the increase in water intake.

The aquifers feeding the mine water intakes of the surrounding villages are hydraulically distant from the main aquifer of the Horbashivska Formation feeding the water intake in Netishyn. Therefore, the increase in the flow rate of the water intake in Netishyn will not have an impact on the groundwater and surface water of the surrounding villages.

1.4.4.3 Provision of demineralized water

A chemical water treatment system was designed, which includes a water pretreatment system and the Demineralized Water Treatment System (DTS), to replenish the losses of water, steam and condensate in the steam-water cycle of Khmelnytskyy NPP Units 5 and 6.

Membrane ultrafiltration technology was chosen as the main technological solution for the water pretreatment system. This technology is used to remove suspended solids, colloidal impurities, some organic contaminants from the treated water, as well as to remove bacteria, algae and other microorganisms. The membrane ultrafiltration unit was chosen as a common unit for the two power units. The maximum water flow rate to the pretreatment system is 400.0 m³/hour. After pretreatment, the water is supplied to a separate demineralized water treatment system (DTS) for each power unit with a flow rate of 125.0 m³/h per unit.

The DTS system receives the prepared water after pretreatment, treats it to reduce ionic impurities and sends it to the Demineralized Water Transfer or Storage System (DWS) for further distribution. The quality of the demineralized water is brought in line with industry standards.

After water treatment in the DTS system, the water is transported to the demineralized water storage tank (DWST) of the Demineralized Water Transfer and

Storage System (DWS), which provides supply and distribution of demineralized water throughout the plant. Water is supplied to demineralized water consumers in the containment, containment building, auxiliary systems building, turbine building, radwaste storage facility, diesel generator building, as well as in the condensate storage tank (CST) in all plant operating modes.

To ensure normal operation of the power units and meet the needs for the demineralized water of the Chemical Feed System (CFS), the Condensate Polishing System (CPS), the Chemical and Volume Control System (CVS), the Rheochemical Laboratory System (RLS), the following are installed: the Demineralized Water Storage Tank (DWST) with a capacity of 477.0m³ and the Condensate Storage Tank (CST) with a capacity of 2483m³.

The amount and reserve of demineralized water is sufficient to ensure the plant in all operating modes, including: startup, hot shutdown, shutdown and operation at capacity.

1.4.4.4 Provision of energy resources for NPP operation

To provide power units No. 5 and 6 with the main necessary resources, the following are in operation:

Diesel oil storage tanks (DOS-MT-01-A and DOS-MT-01-B). The Diesel Oil Storage tanks of the DOS system are steel. Each tank has a useful working capacity of approximately 279.0 m³ of diesel oil for seven days of continuous operation of the diesel generator. The useful working capacity includes an additional 18.9 m³. The nominal oil consumption of the diesel generator (Caterpillar C280-16) at 5903 kW is 1.54 m³/hour. Based on the expected load sequence and fuel consumption at demand, the total fuel consumption for a period of seven days is less than 208.0 m³.

Boric Acid Tank. The Boric Acid Storage Tank (BAST) is designed to store up to 302.8 m³ of boric acid solution with a concentration of 2.5 weight percent.

Plant gas storage tanks (PGS). The liquid nitrogen storage tank has a volume of 11.36 m³. It is used to clean the gaseous radioactive waste disposal system, liquid radioactive waste system, and central chilled water system; consumption is 0.46 m^3 /hour. Three liquid hydrogen storage tanks with a total capacity of 5.68 m³ are used to reserve accumulative tanks for liquid radioactive waste treatment.

The main reagents required for the normal power units operation, namely, for pretreatment, as well as for the chemical water treatment system of power units 5 and 6, consisting of a pretreatment system, the Demineralized Water Treatment System (DTS, DWS), a Chemical Feed System for the turbine hall (CFS),the Condensate Purification System (CPS), the Chemical and Volume Control System (CVS), to maintain the water-chemical regimes of the first and second circuits of the power units, there are chemicals that maintain the pH in the system, control dissolved oxygen, control dissolved carbon dioxide, chlorine, biological fouling, disperse trapped sludge, limit scale formation and control corrosion.

Chemicals used for the operation of AP1000 units 5 and 6 are stored in the nuclear island, turbine hall, and NPP site.

The reagent storage facility is designed for receiving, storing and dispensing reagents. The reagent storage building is combined with a pretreatment unit (one for two blocks) with dimensions of 61.0×109.0 m. According to the general plan, the reagent storage facility is located at a distance of 22.0 m from the checkpoint and 45.0 m from the control room. According to the master plan, there is a railroad and road access to the storage facility with an unloading area.

The Chemical and Volume Control System (CVS), and the radiochemical laboratory (RLS) add the reagents of the required concentration to maintain the water-chemical regime of the primary circuit. The CVS system provides control of the reactor cooling system (removal of fission products and undesirable impurities, control of hydrogen, lithium hydroxide, hydrazine, hydrogen peroxide, and zinc acetate concentration), provides control of boron concentration (both for adding boric acid and for removing boron to introduce the gray control rod and T_{avg}/T_{cep}). The main components of the CVS system are located in the Auxiliary Building, the Annex, inside the containment, and inside the Containment Vessel. The chemicals used in the extended nuclear island are shown in Table 1.8.

Preparation of reagents for the needs of the Demineralized Water Treatment System (DTS, DWS) is carried out in the reagent farm of the chemical feed system for the turbine island (CFS). The functions of the CFS system using the main reagents are shown in Table 1.9.

Tank	Volume, M ³	Chemical reagents	Form and concentration	System/function	Location
Boric acid solution tank (BAST)	302,8	Boric acid	Liquid solution (2.5 wt. %)	CVS / Neutron absorber and pH buffer for the first circuit coolant	BAST is on the site, at the east end of the annex. The boric acid solution preparation tank is located in the annex.
Hydrogen (liquid)	5,7	Hydrogen	Liquid (100%)	CVS / Oxygen absorber in the first circuit coolant system	Plant gas storage system (PGS) - on the site to the west of the northern end of the turbine building
Lithium hydroxide tank	0,019	Lithium hydroxide 7	Liquid solution (12 wt. %)	CVS / Monitoring the pH of the first circuit coolant system	Annexe building
Hydrazine tank	0,019	Hydrazine	Liquid solution (35 wt. %)	CVS / Oxygen absorber in the first circuit coolant system / AVT	Annexe building, (100 kg), and turbine building, (1 t)
Zinc injection kit	0,76	Zinc acetate	Liquid solution	CVS / injection of zinc acetate into the RCS	Turbine building room

Table 1.8 – Chemical reagents used in the expanded nuclear island

Table 1.9 – The water-chemical regime parameters for systems supplied by the Turbine Island Chemical Feed System (CFS)

System	Chemical agent	Normal values during operation at capacity
Condensate System (CDS)	Oxygen absorber	\leq 0,01 mg/L oxygen
	Chemical for pH control	> 9,0
Main and Startup and	Oxygen absorber	\leq 0,002 mg/L oxygen
Feedwater System (FWS)	Chemical for pH control	> 9,5
Steam Generator	Oxygen absorber	< 0.1 mg/L oxygen (cold shutdown / moist
Blowdown System (BDS)		parking mode)
	Chemical for pH control	9.8 - 10.5 (cold shutdown / moist parking mode)
Auxiliary Steam Supply	Oxygen absorber	\leq 0,002 mg/L oxygen
System (ASS)	Chemical for pH control	> 9,5
Service water system	Chemical for pH control	6,0-8,5
(SWS)	Scale inhibitor	Langelier index 0.0 - 1.0
	Sludge disperser	150 mg/l of suspended solids
	Biocide	0.0005 - 0.001 mg/l residual chlorine
	Non-oxidizing biocide	< 10 CFU/ml Legionella
Demineralized Water	Caustic (CO2 control)	> pH 9,0
Treatment System (DTS)	Chlorine control	<0.001 mg/l residual chlorine
	Scale inhibitor	Langellier index 0.0 - 1.0

Reagents for maintaining the water-chemical regime of the second circuit are prepared at each power unit in a separate building.

Each demineralization unit uses a mixture of lithiated cationic and anionic resin. Both forms of resin remove fission and corrosion products.

The quality of the filter materials (ion exchange resins) used to maintain the waterchemical regime (WCR) must meet the requirements of ISO-NNEGC 1.013:2014 [21].

All the necessary reagents in the form of concentrated solutions or dry form are supplied to the CWT reagent storage and then sent to the reagent preparation unit. Table 1.10 shows the consumption of basic reagents for one power unit.

Table 1.10 – List of basic reagent needs for one power unit (indicative)

Chemical reagent	Annual demand for one power unit	Type of delivery/transportation	
Boric acid H3BO3 grade A	20,0	By railroad cars	
Aqueous ammonia technical _{NH3} grade A, high grade, 25%	29,5	By railroad tankers or special vehicles in glass bottles	
Carbohydrazide, 10%	15,5	By rail or road in barrels made of corrosion- resistant steel	
Hydrazine hydrate _{N2H4} , 35 %	2,0	By rail or road in barrels made of corrosion- resistant steel	
Sodium hydroxide 20%	10,0 ÷ 25,0	By rail tankers with boilers made of corrosion- resistant steel, by rail or road in tall or polyethylene barrels, in 15 kg canisters	
Sodium hypochlorite	$20,0 \div 60,0$	By rail in special rubberized tanks	
Hydrochloric acid solution 14%	5,0÷15,0	By rail or road in steel "sulfuric acid" tanks, barrels or glass bottles	
Sulfuric acid contact technical H2SO4, 92%	250,0	By rail or road in steel "sulfuric acid" tanks, barrels or glass bottles	

Chemical reagent	Annual demand for one power unit	Type of delivery/transportation
Morpholine, 41%	2,5	By rail or road in barrels or canisters of 200 kg
Coogulant	60.0	By road or rail in tanks with a corrosion-
Coaguiant	00,0	resistant coating

1.4.4.5 Fuel provision

The AP1000 core consists of 157 Class C fuel assemblies, 53 Class B rod cluster control assemblies (RCCAs), 16 gray rod cluster assemblies (GRCAs), and 4 (2 primary and 2 secondary) neutron source assemblies. Neutron sources are usually required only for the initial cycle. Burnup absorbers are typically used to compensate for some installed excess reactivity and to ensure that the moderator temperature coefficient is maintained at a negative value.

Fuel rod. The rods consist of ceramic uranium dioxide pellets contained in tubes of ZIRLOTM alloy, which are plugged and welded at the ends to seal the fuel.





Fuel cycle duration and fuel batch size. The standard fuel cycle duration of the AP1000 project is 18 months, which is optimal in terms of fuel efficiency. The requirements for the cycle 1 fuel loading batch size are 157 new assemblies. The cycle 1 reactor load is designed to provide full capacity during 465 effective days.

New nuclear fuel storage. The fuel storage facility provides on-site storage of a sufficient number of new fuel assemblies to be replaced at each outage for fuel cycles up to 18 months. This is usually about 40% of the fuel assemblies.

1.5 Assessment of the expected wastes, emissions (discharges), water, air, soil and subsoil pollution, noise, vibration, light, heat and radiation pollution by type and amount, as well as radiation caused by the preparatory and construction activities and the planned activity

The operation of any large industrial facility involves certain environmental impacts. NPP is a source of radiation, chemical, thermal, electromagnetic and noise impacts.

The key component of NPP environmental impact is related to the specifics of electricity generation using energy released as a result of nuclear reactions – radiation impact. The NPP radiation environmental impact is determined by releases and discharges of radioactive substances both during normal NPP operation and in the event of accidents. The main sources of radioactive substances at NPP with VVER reactors are uranium-235 fission products during core fuel neutron irradiation, neutron activation of structural materials, primary coolant impurities and air in the reactor space.

The amount of radioactive substances released to the environment, under other equivalent conditions, during normal operation directly depends on the primary circuit activity and during accidents on the fuel activity, the activity under fuel cladding and the primary circuit activity.

KhNPP under operation may have the following impact on the air environment and microclimate:

- radioactive gaseous discharge;
- non-radioactive releases;

heat and moisture releases from the cooling system ponds (cooling and spray ponds);

– noise, electromagnetic radiation.

1.5.1 Contamination assessment

The construction of power units using the Westinghouse's AP1000 technology at KhNPP is carried out in compliance with the current regulations on environmental protection against radioactive contamination in Ukraine, as well as radiological protection and radiation safety of personnel and the public [22, 23, 24].

In accordance with the requirements of [22], radiation safety and protection in relation to practical activities are based on the following basic principles:

- any practical activity that involves exposure of people shall not be carried out unless it brings greater benefit to the exposed persons or society as a whole compared to the harm it causes (the principle of justification);

- exposure levels from all significant practical activities shall not exceed the established dose limits (the principle of non-exceedance);

– individual dose levels and/or the number of exposed persons in relation to each radiation source shall be as low as reasonably achievable including economic and social factors (optimization principle).

In accordance with the provisions of regulatory documents, the limits are set with the following criteria:

– internal and external exposure of personnel and the public;

– maximum permissible emissions and discharges of radioactive substances into the environment.

According to the Ukrainian legislation on radiation safety, critical groups are divided into the principle of direct work with radiation sources, by gender and reference age.

In accordance with the provisions of [22] the following categories of persons exposed to radiation are established:

- Category A (personnel) – persons who permanently or temporarily work directly with sources of ionizing radiation.

- Category B (personnel) – persons who are not directly involved in work with sources of ionizing radiation, but due to the workplace location in the facility rooms and industrial sites with radiation and nuclear technologies may receive additional exposure.

Category V – the entire population.

In terms of gender, Ukrainian NPPs have established restrictions on the exposure of pregnant women and women of reproductive age.

In addition, by reference age, the distribution is divided into the following age groups: 3 months, 1 year, 5 years, 10 years, 15 years, and "Adult".

The numerical values of the received dose limits are set at levels that exclude the possibility of deterministic exposure effects and at the same time guarantee such a low probability of stochastic exposure effects that it is acceptable for both individuals and society as a whole. Dose limits are given in Table 1.12.

	Category of the exposed persons		
	A ^{a) b)}	B ^{a)}	V ^{a)}
DL _E (effective dose limit)	20 в)	2	1
Equivalent external dose limits:			
DL _{lens} (for the eye lens)	150	15	15
DL _{skin} (for the skin)	500	50	50
DL _{extrim} (for the hands and feet)	500	50	-

Table 1.12 – Radiation dose limits (mSv/year)

Notes:

a) radiation dose distribution during a calendar year is not regulated;

b) for the childbearing age women (up to 45 years) and for pregnant women, the restrictions of Clause 5.6, NRBU-97 apply;

c) on average for any consecutive 5 years, but not more than 50 mSv in a single year (DL_{max}).

Public exposure is limited through regulation and monitoring:

- radionuclide intake through the respiratory system (ALI_{inhal} V) and digestion (ALI_{ingest});

- radionuclide concentration in air (PC_{inhal} V) and drinking water (PC_{ingest});
- permissible environmental discharge and release.

1.5.1.1 Radiation exposure during construction

No additional radiation impact is expected during the construction of AP1000 units, given the low values of the construction site contamination (within the background contamination values of the adjacent territory).

The AP1000 construction is carried out at the existing plant site, which has been in operation since the start of KhNPP construction, so no compensation measures are required.

The data on the radiation situation give the following grounds:

- not to limit the time spent by personnel at workplaces;
- construction wastes are not be classified as radioactive waste.

1.5.1.2 Radiation exposure during normal AP1000 units operation

For the relevant radiation and nuclear facilities, a dose limit quota is set for exposure of category V persons due to radioactive releases and discharges [22]. Dose limit quotas are given in Table 1.13.

Table 1.13 –	Annual d	dose lin	nit quotas	used to	o set	permissible	discharges	and
permissible emissio	ns							

Source	DL quota dose for pathways fr	due to all cmation com releases	Discha DL quota du wate	arges: ue to critical r use	Total DL quota for an individual enterprise	
	%	μSv	%	μSv	%	μSv
NPP, NCP, NHP	4	40	1	10	8	80

The permissible discharges and releases are set based on the dose limits quota for each individual NPP. The development and approval of permissible discharges and releases values are performed in accordance with the procedure established by the Ministry of Health of Ukraine.

When establishing the permissible discharges and releases values, the migration of radionuclides in the environment and in food chains are considered, as well as land use structure and actual water bodies use (recreation, fishing and fish farming, irrigated agriculture, water bodies for livestock, presence of flooded meadows, etc.)

Exceeding the permissible releases and discharges under normal NPP operation is not allowed.

The release limit is set for the entire NPP, regardless of the operating units number.

The radiation impact from units 5 and 6 is additional to the operating NPP impact.

Liquid, solid and gaseous radioactive waste is inevitably generated during NPP operation. The sources of radioactive waste generation are NPP equipment and systems containing gaseous, liquid, and solid media.

GRW – volatile emissions (aerosols) of the primary coolant that occur during maintenance, small leaks, and fugitive coolant leaks.

LRW – generated during operation and maintenance of the reactor main and auxiliary equipment, balance of plant containing radioactive media or that is contaminated.

SRW includes spent reactor internals, pump parts, pipelines, valves, thermal insulation, ventilation system filters, wiping materials, cotton and film overalls, used PPE, various construction waste, etc.

The AP1000 unit provides a radioactive waste treatment and management system to support the expected waste generation from plant operation, and provides the flexibility and space to install mobile systems during emergency situations and special waste treatment systems that may be required in the event of such situations. Below is a list of the AP1000 radioactive waste treatment systems:

- Gaseous Radwaste System (WGS);
- off-site Liquid Radwaste System (WLS);
- Radioactive Waste Drain System (WRS);
- Solid Radwaste System (WSS).

1.5.1.3 Gaseous radwaste sources

The AP1000 reactor is designed to accept all potentially hydrogen-containing and radioactive gases generated during plant operation. The largest amount of gaseous waste is generated during plant operation at the end of the fuel cycle. Gaseous radioactive waste includes:

– gas blowdowns, non-condensable gases and volatile aerosols from equipment operating on radioactive media;

- air that is removed from workspaces.

Gases that accumulate in the primary circuit during operation are removed from it. This leads to the formation of a gaseous radwaste flow. Gaseous emissions can also be generated as a result of ventilation of volatile emissions of the primary coolant arising from small leaks, controlled and fugitive leaks. Gaseous waste is treated before being released to the atmosphere.

Expected radionuclide releases during normal and abnormal operation modeled on the basis of input data obtained from PWR NPP operation experience are presented in Table 1.14 and are conservative.

	Activity Release (GBq/yr) of Airborne Gases							
Nuclide	Fuel Handling	Containment	Auxiliary	Turbine	Condenser Air	Total		
	Area	Building	Building	Building	Removal System	Release		
I-131	7.4E-03	1.9E-02	1.8E-01	2.4E-03	9.6E-04	2.1E-01		
I-133	1.1E-02	7.4E-02	2.6E-01	7.4E-04	3.0E-03	3.5E-01		
Total Airborne Radioiodine								

Table 1.14 – Expected iodine release during normal and abnormal operation

Note: Values less than 3.7E-5 GBq/yr are considered to be negligible.

At the design stage, such data will be revised using the baseline data for Ukraine and recalculated using software tools (calculation codes) to substantiate the safety of nuclear power plants approved in accordance with the Company's procedure. The expected radioactive noble gases emissions under normal and abnormal operation are shown in Table 1.15.

Table 1.15 – Expected discharges of radioactive noble gases under normal and abnormal operation

	Activity Release (GBq/yr) of Airborne Gases						
Nuclide	Waste Gas	Containment	Auxiliary	Turbine	Condenser Air	Total	
	System	Building	Building	Building	Removal System	Release	
Kr-85m	4.6E-01	1.4E-01	1.6E+01	8.5E-04	7.8E+00	2.4E+01	
Kr-85	3.0E+03	1.1E+01	5.2E+01	2.9E-03	2.6E+01	3.1E+03	
Kr-87	negligible	4.4E-02	1.7E+01	2.6E-04	2.2E+00	1.9E+01	
Kr-88	6.7E-03	1.0E-01	1.8E+01	9.6E-04	8.5E+00	2.7E+01	
Xe-131m	1.1E+03	3.1E+01	1.8E+02	9.3E-03	8.1E+01	1.4E+03	
Xe-133m	3.6E-02	6.7E+00	7.4E+01	4.1E-03	3.5E+01	1.1E+02	
Xe-133	2.4E+02	8.9E+01	6.3E+02	3.3E-02	2.9E+02	1.3E+03	
Xe-135m	negligible	6.7E-02	1.3E+02	7.0E-03	5.9E+01	1.9E+02	
Xe-135	negligible	3.1E+00	1.7E+02	2.9E-02	2.6E+02	4.4E+02	
Xe-137	negligible	negligible	3.4E+01	1.8E-03	1.6E+01	4.8E+01	
Xe-138	negligible	2.9E-02	5.9E+01	3.3E-03	2.9E+01	8.9E+01	
Total Noble	e Gas					6.7E+03	

Note: Values less than 3.7E-5 GBq/yr are considered to be negligible.

The expected aerosol emissions under normal and abnormal operation are shown in Table 1.16.

Table 1.16 – Gaseous discharge of particulates during normal and abnormal operation

		Activity Release (GBq/yr)							
Nuclide	Waste Gas System	Containment	Auxiliary	Fuel Handling	Total				
	Wuble Gus System	Building	Building	Area	Release				
Cr-51	negligible	negligible	1.2E-04	6.7E-05	2.3E-04				
Mn-54	negligible	negligible	negligible	1.1E-04	1.6E-04				
Co-57	negligible	negligible	negligible	negligible	negligible				
Co-58	negligible	9.3E-05	7.0E-04	7.8E-03	8.5E-03				
Co-60	negligible	negligible	1.9E-04	3.0E-03	3.2E-03				
Fe-59	negligible	negligible	negligible	negligible	negligible				
Sr-89	negligible	4.8E-05	2.8E-04	7.8E-04	1.1E-03				
Sr-90	negligible	negligible	1.1E-04	3.0E-04	4.4E-04				
Zr-95	negligible	negligible	3.7E-04	negligible	3.7E-04				
Nb-95	negligible	negligible	negligible	8.9E-04	9.3E-04				
Ru-103	negligible	negligible	negligible	negligible	negligible				
Ru-106	negligible	negligible	negligible	negligible	negligible				
Sb-125	negligible	negligible	negligible	negligible	negligible				
Cs-134	negligible	negligible	2.0E-04	6.3E-04	8.5E-04				
Cs-136	negligible	negligible	negligible	negligible	negligible				
Cs-137	negligible	negligible	2.7E-04	1.0E-03	1.3E-03				
Ba-140	negligible	negligible	1.5E-04	negligible	1.6E-04				
Ce-141	negligible	negligible	negligible	negligible	negligible				
Total Particu	lates				1.7E-02				

Note: Values less than 3.7E-5 GBq/yr are considered to be negligible.

1.5.1.4 Liquid radwaste sources

The system for handling liquid contaminated media includes subsystems that can be used to treat and dispose of liquids containing radioactive material.

The main sources of liquid radioactive waste generation at the AP1000 facilities are:

- the primary coolant;

- decontamination water generated during decontamination activities at NPP equipment;

controlled and fugitive leaks from equipment containing liquid radioactive media;

- steam generator blowdown water;
- drainage water coming from the Radioactive Waste Drain System (WRS);
- others.

Before liquid waste is discharged and throughout its management, it is mandatory to monitor and control the compliance of radiation parameters of the process environment.

The AP1000 plant is designed to provide adequate capacity to treat the anticipated wastes. The AP1000 plant also includes provisions for a connection to mobile equipment in the event that liquid radwaste production exceeds the anticipated volume, or for other situations such as unusual impurity content or prolonged equipment unavailability.

The projected radwaste flows in the WLS system under normal operation are shown in Table 1.17.

Collection Tank and Sources	Expected Input Rate	xpected Input Rate Activity		
	Effluent Holdup	Tanks		
CVS Letdown	601.9 m ³ /yr	100% of Reactor Coolant		
Leakage Inside Containment (to Reactor Coolant Drain Tank)	0.04 m ³ /day	167% of Reactor Coolant	Degassed, filtered,	
Leakage Outside Containment (Relief Valve Leakoffs Piped to Effluent Holdup Tanks)	0.30 m ³ /day	100% of Reactor Coolant	monitored, and discharged.	
Sampling Drains	$0.76 \text{ m}^3 / \text{day}$	100% of Reactor Coolant		
	Waste Holdup T	anks		
Reactor Containment Cooling	1.89 m ³ /day	0.1% of Reactor Coolant	Degassed, filtered,	
Spent Fuel Pool Liner Leakage	0.10 m ³ /day	0.1% of Reactor Coolant	demineralized as	
Floor Drains	2.56 m ³ /day	0.1% of Reactor Coolant	required, monitored, and discharged.	
	Chemical Wastes (Deter	rgent Waste)		
Hot Shower	0 m^3 /day	10 ⁻⁷ μCi/g	Filtered,	
Hand Wash	0.76 m ³ /day	10 ⁻⁷ µCi/g	monitored, and	
Equipment and Area Decontamination	$0.15 \text{ m}^3/\text{day}$	0.1% of Reactor Coolant	discharged. If necessary,	
Laundry	Laundry waste	e is treated offsite	processed with	
Chemical wastes	0.01 m^3 /day	\leq reactor coolant	mobile equipment.	

Table 1.17 – The projected liquid radwaste flows for the radwaste management system under normal operation conditions

1.5.1.5 Solid radwaste sources

The main solid radioactive wastes from the AP1000 reactors for which the system is designed can be identified as follows:

- spent ion exchange resins;
- spent filter cartridges;

- dry active wastes and mixed wastes generated as a result of plant equipment operation, containing contamination;

– other.

The Solid Radwaste System collects spent ion exchange media and filters from the Chemical and Volume Control System, the Spent Fuel Pool Cooling System, and the Liquid Radioactive waste System. The spent ion exchange media is stored within two spent resin tanks in the Auxiliary Building. See Table 1.18 for details on the spent resin tank. The data on solid radwaste during normal operation and anticipated operational occurrences are presented in Table 1.19.

Table 1.18 –	Spent Resin	Tanks
--------------	-------------	-------

Tank Name	Tag Number	Building and Location	Usable Volume
Spent Resin Tank A	WSS-MV-01A	Auxiliary Building Room 12373	7.0 m ³ (250 ft ³)
Spent Resin Tank B	WSS-MV-01B	Auxiliary Building Room 12373	7.0 m ³ (250 ft ³)

T 1 1 1	10 10	C 1	ODIT	CT	1	A D1000	1 .
Table 1.	19 - The	forecasted	SRW	flows	at the	AP1000	plant

RAW Description	RAW Classification	Frequency	Average volume per unit (m ³)	Maximum volume per unit (m ³)	Emissions for the entire period of operation (m ³)
Ion exchange resins	intermediate level	40%/18 months	7,8	15,6	561
Gray rod clusters	intermediate level	Annually	1,7	-	5,1
Control rod clusters	intermediate level	1 time/20 years	5,6	1,1	16,9
Wet granular carbon sorbent	intermediate level	1 time/20 years	0,6	0,4	41
Metal filter element cartridge	intermediate level	Annually	0,2	206	13,7
Pressed paper, films, clothing, plastics, elastomers	low level	Annually	135	1,06	8924
Non-sealed metal objects, glass, wood	low level	Annually	6,6		455
HVAC filters - unsealed fiberglass/metal	low level	Miscellaneous		7,7	761
Spent ion exchange resins from the condensate treatment system	low level	Annually	3,9	3,3	69,3
Dry granular carbon sorbent	low level	Annually	0,3	-	54,3
HVAC system filters - granular coal	low level	1 time/10 years	4,9		29,1
Compressed rigid plastic - gaskets, valve seals, insulation	low level	Miscellaneous	-	-	7,6
Electrodeionization unit - resin/membrane module	low level	1 time/12 years	1,7	-	10,8
Heat exchanger insulation	low level	1 time/60 years	8,4	-	8,4

1.5.2 Assessment of non-radiation pollution

According to the Constitution of Ukraine [25], the state policy of Ukraine is aimed at ensuring environmental safety and maintaining the ecological balance in the territory of Ukraine. For that purpose Ukraine pursues an environmental policy on its territory aimed at preserving a safe habitat for wildlife and inorganic nature, protecting the life and health of the population from the negative impact caused by environmental pollution, achieving harmonious interaction between society and nature, and protecting, rationally using and restoring natural resources [25, 26].

1.5.2.1 Air emissions assessment

The assessment of air pollutant emissions is carried out both during the preparatory and construction activities and during operation. When operating mobile and stationary fugitive emission sources, the assessment is carried out in accordance with the methodology for calculating air pollutant concentrations from emissions (according to OND-86) for surface emission sources. At the same time, the emission capacity of a mobile source is determined by the calculation method in accordance with the calculation methodologies.

Environmental impact assessment during preparatory and construction activities During the preparatory and construction activities, some air pollution will be associated with the operation of construction and transport equipment (exhaust emissions from internal combustion engines). Prior to the start of the main construction and installation activities, the preparatory activities specified in clause 1.3.3 shall be performed.

The following works cause the negative environmental impact during the construction of power units:

• soil development to install the technological equipment, structures, pipelines, and other engineering networks;

- road surface arrangement;
- installation works.

The need for construction and transport equipment for the construction and installation works is given in clause 1.3.3 in the Tables 1.1, 1.3, 1.4.

Construction and transport equipment that operates simultaneously during the facility construction is shown in Table 1.20.

Name	Model	Quantity	Notes	
Backhoe excavator	EO-4121	1	A bucket capacity of 1.0 m ³	
Backhoe excavator	EO-4321	2	A bucket capacity of 0.65 m ³	
Universal excavator	EO-2621	2	A bucket capacity of 0.25 m ³	
Bulldozer	DZ-53 or DZ-54	2	Engine power 82 kW	
Bulldozer	DZ-9 or DZ-33	1	Engine power 129 kW	
Autocropo	LTM series	2	Load consists 75 100 tons	
Autocrane	telescopic autocrane	3		
Self-propelled modular transporter	Goldhofer SPMT	2	Load capacity 15000 tons per axle	
Dump truck		5	Load capacity 8.0 - 27 tons	
Drop-side truck		5	Load capacity 5.0 - 12 tons	
Tractor truck		2	Load capacity 8.0 - 70 tons	
Truck mixer		5	Capacity 4.0-6.0 m ³	
Truck mounted concrete pump		2	Capacity 80 m ³ /h	
Total		32		

T 11 1 O O	T ! 0	•	•	• •	
$T_{0}h = 1/20$	List of	anninnant	onoroting	011111	tonoouoly
1000 - 1.20 - 1.20			operating	SIIIIU	
-		1 1	1 0		

The calculation of pollutant emissions from transport and construction equipment during the construction was performed in accordance with the Methodology for Calculating Pollutant Emissions from Mobile Sources, 1999, published by UkrNTEK OJSC.

The average diesel fuel consumption of 8 liters per hour was used in the calculation. The density of diesel fuel -0.85 kg/l.

The calculation results of pollutant emissions from machinery are shown in Table 1.21.

Pollutant	SpecificCoefficient ofemissions,equipment technicalkg/tcondition		Fuel consumption*, t/h	Pollutant emissions, g/s*			
	Diesel fuel						
301 Nitrogen dioxide	32.8	0.95	0.007	0.061			
328 Soot	3.85	1.8	0.007	0.013			
330 Sulfur dioxide	5.0	1.0	0.007	0.010			
337 Carbonic oxide	32.0	1.5	0.007	0.093			
2754 Hydrocarbons C12-C19	5.65	1.4	0.007	0.015			

Table 1.21 – Pollutant emissions from transport and construction equipment

Note*: fuel consumption and pollutant emissions are given for one piece of equipment (excavator, bulldozer, crane, dump truck, etc.).

Gross emissions of pollutants from all equipment used during the facility construction (tons/construction period) will be calculated at the project development stage.

The calculation of pollutant dispersion in the surface air layer for the facility construction period was performed according to the OND-86 methodology, using the EOL 2000 [h] software package for all simultaneously operating construction and transport equipment.

Ground-level pollutant concentrations were determined in a calculated square with a side size of 8000 m and a step of 250 m. The control points were selected as the points of the sanitary protection area boundary and on the border of the nearest city of Netishyn.

The calculation was performed for the warm season, including the background concentrations in Netishyn, issued by the Khmelnytskyy Regional State Administration on June 02, 2023.

The records for calculating the airborne pollutant dispersion for the construction period is in Appendix C.1.

The calculation analysis of the surface pollutant emission concentrations shows that during construction activities the pollutant emission impact is limited to the KhNPP industrial site.

Assessment of the airborne pollutant emissions for the facility operation period

The main KhNPP operating units for the existing power units 1 and 2, which have sources of air pollutants, are as follows:

• a hydroengineering shop, consisting of a start-up and backup boiler house, an oil, fuel oil and diesel facility, and repair workshops;

- a power repair unit with repair and mechanical workshops and shops;
- a turbine shop turbine halls of units 1 and 2.

During the operation of units 5 and 6, the air will be polluted by the releases from the designed diesel fuel tanks, buildings: turbine building, diesel generators, repair and mechanical workshop, internal passage of vehicles (during arrival and departure from the parking lot).

The calculations of the polluted releases rate from the designed sources (No. 1001-1051) with referenced methods are in Appendix L. The location of the designed pollutant sources is shown on the general plan of units 5 and 6 in Appendix H.6.

The calculation of the pollutant dispersion in the surface air layer for the period of the facility's operation was performed using the EOL 2000h v 4.0 program, which implements the OND-86 methodology.

The calculation was performed for the designed equipment of units 5 and 6, including the existing main pollutant sources of emissions from units 1 and 2, the designed pollutany sources from units 3 and 4, as well as background pollutant concentrations in the facility impact area (Netishyn). The calculation includes the simultaneous operation of power and technological equipment.

The values of background pollutant concentrations in the facility impact area (Netishyn), issued by the Khmelnytskyi Regional State Administration dd 02.06.2023, are given in Appendix E.

The pollutant parameters are given in Appendix M.

During the calculation, the ground-level air pollutant concentrations were determined at two control points selected in the direction of Netishyn - on the SPZ border and the settlement.

The calculation results of the air pollutant dispersion for the period of facility operation are presented in Appendix C.2.

Based on the dispersion calculation, no exceedances of pollutant concentrations are expected on the SPZ border and the nearest settlement (Netishyn) during the KhNPP operation.

1.5.2.2 Assessment of discharges and water pollution

The preparatory and construction activities. No additional negative impact sources on the hydrosphere are expected during the construction of units No. 5 and 6. Temporary residential and storage facilities at the construction site have water supply and sewerage systems. Domestic drinking water for these temporary facilities is provided through

existing utilities. Wastewater is discharged into existing sewage systems. No wastewater is expected to be generated or discharged into the environment.

Operation. Production activities may have an impact on surface and groundwater.

Surface water. The impact of KhNPP production activities on surface water is in places of direct contact of NPP process components and facilities with surface water bodies of public use, i.e. with the cooling pond.

The KhNPP cooling pond was built for the plant process needs based on the permissible water cooling temperature of 33 °C [27], to remove heat from the 4000 MW NPP equipment (4 power units), including outage schedule. Under the most unfavorable "hot" climatic conditions and the most unfavorable wind situations, the thermal calculation of the KhNPP cooling pond, including the construction of AP1000 units 5 and 6, performed by the affiliate SS "Atomprojectengineering" in 2023, showed the following:

– during the operation of two units in summer, the water temperature in the underwater channel (at the turbine condenser inlet) is 24.4-25.5 °C;

- during the operation of three units in summer, it is 30.0-30.8 °C, which is less than the regulatory temperature value of 33.0 °C [27];

during the operation of four units in summer, the water temperature is 33.0-33.2 °C and will already reach the critical temperature under process conditions of 33,0°C [27];

- during the operation of units 5 and 6 – according to the AP1000 process flow diagram, developed by Westinghouse, the purge water from the tower cooling basin of the CWS system is combined with the purge flow from the SWS system, taken from the pump discharge header and sent to the purge sump of the wastewater collection and disposal system for further discharge into the KhNPP cooling pond. Based on the operating conditions of the cooling and blowdown water removal scheme, it can be concluded that the cooling pond receives water already cooled to ambient temperature. That is, the operation of power units 5 and 6 will not affect the value of additional evaporation from the reservoir and the temperature regime in the reservoir as a whole.

When units 3 and 4 are commissioned at the permissible cooled water limit temperature of 33.0 °C, the cooling pond can provide NPP capacity of no more than 3240 MW. Additional cooling of the circulating water is required to ensure NPP capacity of 4000 MW. To ensure NPP capacity of 4000 MW or more, additional cooling of the circulating water is required. When developing the feasibility study of units 3 and 4 [28], the UkrNIIEP Institute designed a 1300.0 m long flow control dam.

The dam height is from 5.0 m to 9.0 m, depending on the depth, with the crest mark at 205.00 m. The width of the dam crest is 12.0 meters. The slope of the dam to the 203.00 m mark is 1:3. The slope of the dam in the underwater part is 1:5. To avoid washing away of the alluvial soil, it is planned to fix the alluvial slopes with 1.0 m thick stone riprap.

The length of the flow control dam of 1300.0 m is minimal to ensure full utilization of the reservoir area and cooling of the circulating water. Cooling efficiency is increased in this case by redistributing water flows and increasing the core of the transit flow and whirlpool. This allows cooling the circulating water during the operation of three units to 27.7 °C in the conditions of hot decade meteorological factors and to 29.6 °C during the operation of four units.

The dam will be constructed under the units 3&4 construction project implementation and will allow to improve the efficiency of cooling the recycled water in the pond and guarantee the necessary temperature conditions for the operation of six units even in the most unfavorable "hot" hydrometeorological conditions.

Groundwater. The stationary groundwater monitoring network includes 189 wells. The purpose of hydrogeological monitoring is to control the stability of groundwater regime criteria (level, temperature, chemical composition) and assess the impact of technogenic factors on groundwater.

During the operation of units 1 and 2, the NPP technogenic impact had almost no effect on the groundwater level, but affected its chemical composition and temperature as a result of the industrial water infiltration into the soil due to leaks from water supply lines. It should be emphasized that, according to the monitoring data, chemical and thermal contamination of groundwater and hydraulically connected in the upper part of the Upper Proterozoic horizon is localized only within the NPP site; the background (i.e., unaffected by technogenesis) values of groundwater chemical composition and temperature were recorded on the site periphery. At the same time, within the industrial site, there is not a continuous field of technogenically polluted groundwater, but rather disconnected local areas where groundwater is characterized by increased mineralization and temperature.

Seasonal fluctuations in the groundwater level averaged 0.5m in 2017, 0.56m in 2016, 0.7m in 2015, 0.52m in 2014, 0.72m in 2013, and 0.73m in 2012. Groundwater levels are stable and respond only to seasonal climate changes.

The difference in groundwater temperature within the industrial site is 10.0 °C. The background temperature is between 9.0 °C and 10.0 °C. The water temperature of the Upper Proterozoic aquifer, as well as the groundwater temperature, is also higher than the background temperature in some areas. The background temperature is 10 °C.

The range of temperature fluctuations within the industrial site is 10.5 °C, with a maximum of 20.5 °C.

The chemical composition of groundwater has been gradually changing in recent years. The mineralization changing during recent years is shown in Table 1.22:

	2012	2013	2014	2015	2016	2017	2019	2020	2021	2022	2023
Ground water, mg/l	342.36	406.42	423.72	552.91	541.19	605.26	676.98	-	996.40	901.71	869.11
Upper Proterozoic aquifer, mg/l	401.11	475.4	487.38	564.97	620.99	586.68	624.84	644.78	724.84	717.92	610.37

Table 1.22 – The mineralization changing during recent years

It is not possible to fully predict quantitative changes in the chemical composition and temperature of groundwater associated with the technogenic impact of NPP facilities, as there are no input data for such a quantitative forecast (possible losses of chemically contaminated and hot water are unpredictable).

1.5.2.3 Assessment of the expected wastes

In the course of *production activities*, certain non-radioactive wastes may be generated (see also Table 1.23), namely:

• saturated ion exchange resins (damaged or spent);

• waste materials used for water disinfection and purification (anthracite chips, sulfonated carbon);

• activated carbon (faulty or spent);

• waste materials used in water extraction, treatment and distribution (waste slaked lime);

• process waste and waste from distribution of electricity, gas, steam and hot water, not otherwise designated (waste heat insulation);

• mixed waste from the construction and demolition of buildings and structures (roofing material waste, broken concrete, broken bricks, glass, etc;)

• fragments of porcelain and semi-porcelain;

• waste from the wastewater treatment plants operation, not otherwise designated (sand);

• other industrial and process waste, not otherwise designated, or waste from combined processes of drinking or service water preparation;

- lumpy wood waste;
- crushed artificial sand from volcanic slag (perlite sand);
- crushed stone and sand from industrial waste.

Waste Code as per the National Waste List	Waste Name as per the National Waste List	Production Waste Name	Amount of Waste per year, tons
20 01 21*	Fluorescent lamps and other mercury-containing waste	Fluorescent lamps	2.220 (7398 pcs)
06 04 04*	Mercury-containing waste	Instruments containing mercury (manometers, oscilloscopes, normalizers, thermometers, etc.)	0.007
16 06 01*	Lead batteries	Spent lead batteries and lead waste	39.276
13 02 05*	Mineral oils, non-chlorinated motor, transmission and lubricating oils	Used mineral turbine oils	4.031

Table 1.23 – List of the expected wastes

Waste Code as			Amount of
per the National Waste List	Waste Name as per the National Waste List	Production Waste Name	Waste per year, tons
20 01 35*	Electrical and electronic equipment waste other than those specified in codes 20 01 21 and 20 01 23 containing hazardous components	Spent paper insulation polluted with oil products	3.241
17 06 03*	Other insulating materials consisting of or containing hazardous substances	Spent silica gel polluted with oil products	2.070
06 05 02*	Sediments (sludges) from wastewater treatment at an enterprise containing hazardous substances	Sediment, sludges polluted with oil products	29.253
05 01 03*	Bottom sludge (sediment, silt) at the tank bottom	Sludge from tank cleaning polluted with oil products	8.000
20 01 34	Batteries and accumulators other than those specified in code 20 01 33	Used batteries	0.185
13 07 03*	Other fuels (including mixtures)	Spent blends (ethanol-gasoline, gasoline, diesel fuel, oils, etc.)	1.871
16 01 07*	Oil filters	Used automotive fuel and oil filters	1.000
15 02 02*	Absorbents, filter media (including oil filters not otherwise specified), wiping rags and protective clothing polluted with hazardous substances	Wiping materials and overalls polluted with oil products	5.750
2000.3.1.16 (DK 005-96)	Off grade ties for railway or tram tracks	_	20.977
16 02 14	Equipment waste other than specified in codes from 16 02 09 to 16 02 13	Waste electronic equipment: monitors, laptops, printers, servers, keyboards, mice, telephones, etc.	0.423
16 01 03	Used tires	Used tires	13.454
03 01 04*	Sawdust, shavings, scraps, wood, chipboard and veneer containing hazardous substances	Oiled sawdust, filters from car wash treatment facilities	0.728
16 02 09*	Transformers and condensers containing polychlorinated biphenyls (PCBs) or polychlorinated terephthalates (PCTs)	Used transformers and condensers	0.014
16 02 15*	Hazardous components removed from waste equipment	Spent cartridges and filters for cleaning cartridges	0.604
20 01 32	Medicines other than those specified in code 20 01 31	Drugs and medicinal substances expired	0.002

Waste Code as per the National Waste List	Waste Name as per the National Waste List	Production Waste Name	Amount of Waste per year, tons
16 02 12*	Waste equipment containing free asbestos	Used asbestos, ebonite, textolite, fiberglass, paronite parts or materials	2.817
10 01 26	Waste from cooling water treatment (purification)	Sludge from water desalination	195.328
19 08 06*	Saturated or used ion exchange resins	Used filtration material polluted with oil products: sipron, birch charcoal, ion exchange resins, anthracite chips, sulfonated carbon	35.625
17 06 04	Insulating materials other than those specified in codes 17 06 01 and 17 06 03	Waste thermal insulation material	26.000
17 09 04	Mixed construction and demolition waste other than that specified in codes 17 09 01, 17 09 02, 17 09 03	Construction waste	999.636
10 13 04	Burning and lime slaking waste	Waste slaked lime	70.000
19 08 05	Sludge from municipal wastewater treatment	Sludge from municipal wastewater treatment	75.468
19 08 02	Waste from sand removal	Spent sand from wastewater treatment	32.304
17 02 01	Wood	Lumpy wood waste	112.126
20 03 01	Mixed domestic waste	Domestic waste	1202.990
06 10 99	Other waste in this subgroup	Spent extinguishing powder	0.970
16 01 18	Non-ferrous metals	Non-ferrous scrap	30.705
16 01 17	Ferrous metals	Ferrous scrap including electrodes	515.297
15 01 04	Metal packaging	Polluted metal containers	0.293
08 02 99	Other waste in this subgroup	Spent porcelain insulators and fragments of porcelain, semi- porcelain	39.623
11 01 99	Other waste in this subgroup	Spent membranes and filter elements (osmosis elements, filters for water purification made of foamed polypropylene)	0.102
19 09 04	Spent activated carbon	Spent activated carbon	4.200
20 01 01	Paper and cardboard	Used paper and cardboard (waste paper)	29.056
20 01 10	Clothes	Worn overalls and footwear	4.313
07 02 13	Waste plastics	Polluted plastic and plastic containers	5.544
10 12 03	Small residues and dust	Waste perlite sand, zeolite, silica gel	1.500
19 09 99	Other waste in this subgroup	Crushed stone, waste sand from drinking water treatment	1.999
15 01 07	Glass containers (packaging)		2.091
17 02 02	Glass	Waste technical glass	8.089

Waste Code as per the National Waste List	Waste Name as per the National Waste List	Production Waste Name	Amount of Waste per year, tons
20 03 99	Other domestic waste of this subgroup	Polyethylene waste (clean)	0.102
16 02 16	Components removed from waste equipment other than that specified in 16 02 15	—	0.314
19 12 04	Plastics and rubber	Waste rubber	4.283
12 01 21	Spent abrasion bodies and abrasion materials other than those specified in code 12 01 20	Waste abrasives, grinding wheels, belts, sandpaper	0.022
07 02 13	Waste plastics	Polluted plastic and plastic containers	4.662

Waste generated *during construction* is mainly generated in buildings and structures of the site facilities and during welding operations. Wastes generated at site facilities include:

- unsorted ferrous metal scrap;
- concrete waste in lump form;
- scrap of reinforced concrete products;
- wiping material polluted with oil products;
- metal containers polluted with paint.

Welding operations generate the following types of waste:

- remnants and fragments of steel welding electrodes;
- cardboard containers for electrodes;
- welding slag;
- domestic waste.

During the construction of industrial facilities, the generated waste is stored at a specially designated construction waste dump territory.

All solid industrial and domestic waste unsuitable for further use will be disposed at the landfill of the Municipal Enterprise for Public Utilities and Housing in Netishyn as it accumulates and upon completion of construction. Metal containers are used for domestic waste collection.

1.5.2.4 Assessment of noise, vibration and ultrasonic pollution

Noise impact assessment. The assessment is carried out including:

• impact of additional noise sources that appear with the commissioning of units 5 and 6;

• absence of permanent workplaces for service personnel at the industrial site, outside of production buildings and structures, and noise impact assessment is appropriate only inside these buildings;

- absence of any buildings and structures within the sanitary and protection area with permanent presence of people who are not NPP personnel.

In the case of construction activities at units 3&4, the noise impact assessment shall include the noise of construction equipment.

At the industrial site, the noise sources are power transformers and backup diesel power unit. As for the turbine building, the noise sources are the turbine generator, feed pumps, control pumps, hydraulic lift pumps, ejectors, and as for the NPP buildings and structures – ventilation and air conditioning systems, pumping units.

Transformers are operated without the constant presence of personnel and, as a result, no noise requirements are imposed on transformers. Transformer operation is monitored periodically (1-2 times per shift) by operating personnel for 15-20 minutes.

The operating personnel who are constantly at their workplaces in the respective buildings maintain the turbine building equipment (turbine unit, pumping units, and ejectors) and the backup diesel power unit equipment (diesel generator, compressors, pumps) around the clock.

The noise level in the turbine hall and backup diesel power unit buildings may exceed the permissible sound pressure level [29] in the production facilities of the enterprises. In this regard, the project provides for acoustically protected rooms for the turbine shop and backup diesel power unit operating personnel in the turbine hall and backup diesel power unit buildings. When the operating personnel carry out mandatory visual inspection of the equipment at the place of its installation, it is assumed that the personnel use appropriate personal protective equipment (headphones, earplugs, antiphons).

The following measures are provided to reduce noise and vibrations from operating ventilation units to values that do not exceed the permissible sound pressure levels in the premises [29]:

- installation of fans on vibration isolators;
- connection of air handling units with air ducts through flexible inserts;

• limiting the air velocity in the air ducts, which ensures that the noise levels generated by the control and air distribution devices in the serviced premises are within the permissible limits;

• installation of air handling units with the lowest specific sound power levels;

• installation of sound attenuators in ventilation systems serving premises with a permanent presence of people;

• installation of sound-absorbing structures on the walls and ceilings of ventilation chambers.

Vibration impact assessment. No vibration impact exceeding the established standards is expected during the construction and operation of KhNPP units 5 and 6.

The main sources of vibration are fans of ventilation systems. Noise and vibration from ventilation and air conditioning equipment after the use of noise attenuators and measures against noise and vibration impact comply with sanitary and hygienic standards

for the working area in accordance with regulatory documents [29, 30]. The fans of ventilation and air conditioning systems are isolated from the unit casing by means of spring-type vibration isolators.

Ultrasound exposure assessment. The ultrasound impact from operating thermal and mechanical equipment during the operation of KhNPP units is not envisaged.

A single, short-term, local, ultrasonic exposure is possible during ultrasonic quality control of welded butt joints performed under repair.

1.5.2.5 Electromagnetic radiation assessment

There are no sources of electromagnetic radiation exceeding the limit values established by regulatory documents at KhNPP units. New sources are not expected during the construction and operation of units 5 and 6.

Electrical equipment installed in NPP facilities is not a source of harmful emissions, radio interference and noise.

According to the sanitary standards [31], protection of the public from the electric field of aerial power lines with a voltage of 220 kV and below is not required.

330 kV and 750 kV aerial power lines from the KhNPP switchyard are designed in accordance with the sanitary standards requirements [31].

1.5.2.6 Thermal pollution assessment

A nuclear power plant is a source of significant heat emissions. Approximately twothirds of the thermal energy generated by the reactor cannot be used to generate electricity and is discharged into the environment.

The service water supply scheme for units 5 and 6 was adopted as a reversed, using cooling towers as radiators. In the adopted scheme, the KhNPP cooling pond is a source of water intake for cooling tower basins feeding and blowdown and is used for the needs of other process systems, as well as for blowdown water discharge. The water supply scheme is unified, separate for each unit. In the NPP cooling systems, heat emissions are discharged into the atmosphere through cooling towers.

For the circulating water system (CWS), a naturally ventilated cooling tower is adopted to receive the heated circulating water from the main condensers and support the heat exchangers. The circulating water is cooled by evaporation and returned to the cooling tower basin.

The service water system (SWS) is cooled by a fan cooling tower. The cooling tower is a straight, counterflow tower with a suction fan and a clog-resistant film fill.

The total heat emissions to the atmosphere from the cooling towers according to [32, 33], are shown in Table 1.24.

Table 1.24 – Heat emissions to the atmosphere from units 5 and 6 cooling towers (at 7200 hours of units operation)

Cooling tower type	The amount of heat, maximum/minimum

	Cooling tower number, pcs.	Gcal/h	Gcal/day	Gcal/year
CWS cooling towers	2	3810.0/0	91440.0/0	27.4 x 10 ⁶ /0
SWS fan cooling towers	Two double-chamber cooling towers	174.6/37.74	4190.4/905.76	1.26 x 10 ⁶ / 0.74 x 10 ⁶
Total	4	3 984.6/37.74	95 630.4/905.76	28.66 x 10 ⁶ / 1.74 10 ⁶

2. DESCRIPTION OF THE REASONABLE ALTERNATIVES TO THE PLANNED ACTIVITY, THE MAIN REASONS FOR SELECTING THE PROPOSED OPTION, CONSIDERING THE ENVIRONMENTAL CONSEQUENCES

The main requirement for the selection of sites for new power units with the AP1000 reactor should ensure nuclear and radiation safety of the population and environmental protection according to the current legislation at all stages of the nuclear facility life cycle, as well as maximum efficiency of electricity generation for supplying electricity to consumers under the best conditions.

According to international practice, the most acceptable option for the deployment of new facilities is to build them at the sites of operating NPPs to efficiently use previously invested funds and reduce the construction cost of new units by using readymade infrastructure and qualified personnel.

The Energy Strategy of Ukraine until 2050 envisages the construction of new power units at the sites of operating NPPs, including KhNPP. Favorable factors include:

- a developed road and railway networks;
- general plant facilities, including auxiliary facilities and water supply facilities;

 construction facilities, construction industry enterprises, and construction and installation organizations;

– staffing, etc.

This section provides information on the current status of the selected sites for the new power units construction (information according to the State Land Cadaster on ownership and property rights to the land plot) and construction conditions at these sites.

According to preliminary calculations, the following areas are required for the construction of two power units with an AP 1000 reactor:

- for the industrial site -45 ha for power units with cooling towers
 - 38 ha for power units without cooling towers;
- 100 ha for the construction site.

The cadastral information for each site is provided only for the area to be occupied by the NPP industrial site.

2.1 Construction conditions at the South Ukrainian NPP site

2.1.1 Brief site description

The South Ukrainian NPP site is located in the Voznesensk district of Mykolaiv region, on the left bank of the middle reaches of the Southern Bug River. The distance from the river shoreline to the NPP site is 3 km.

The South Ukrainian NPP is located in the central part of the IPS of Ukraine on the territory of the Dnipro power system. A complex closed network of 330 kV and 750 kV has been formed in this area.

There are three WWER-1000 power units in operation at the South Ukrainian NPP:

- power unit No.1 with the V-302 reactor;
- power unit No. 2 with the V-338 reactor;
- power unit No. 3 with the V-320 reactor.

The total capacity of the NPP is 3000 MW. Table 2.1 shows the general characteristics of the power units and their lifetime according to the licenses to carry out activities at the life cycle stage "operation of nuclear facilities of South Ukrainian NPP power units" issued to the operating organization, National Nuclear Energy Generating Company Energoatom JSC by the State Nuclear Regulatory Inspectorate of Ukraine.

Power Unit number	Reactor type	Commissioning	Estimated lifetime end	Installed capacity, MW
Power unit No. 1	WWER - 1000/V-302	31.12.1982	The lifetime is extended until 02.12.2033	1000
Power unit No. 2	WWER - 1000/V 338	09.01.1985	The lifetime is extended until 31.12.2025	1000
Power unit No. 3	WWER - 1000/V -320	20.09.1989	The lifetime is extended until 10.02.2030	1000

Table 2.1 – General characteristics of SUNPP power units

The site has the infrastructure to support NPP operations. Possibility of deploying operating, construction, and assembly specialists in the existing city of power engineers. There is a production of construction materials and a well-developed road and rail transportation network.

2.1.1.1 Demographic conditions

The South Ukraine NPP site is located in Mykolaiv Oblast, on the left bank of the Pivdennyi Buh River. The district center is Voznesensk town.



Currently, the 3 operating power units are located on a plot of 317.96 hectares. The industrial site can be used for the construction of two new power units, the existing 43.2-hectare; and an additional 6.93 hectares of privately owned land for private farming can be used..

The site layout is shown in Figure 2.1.



Figure 2.1 – Situation layout scheme of new power units at South Ukrainian NPP

2.1.3 Power output

South Ukrainian NPP together with Tashlyk PSP and Oleksandrivska HPP make up the entire South Ukrainian power complex.



2.1.4 Service water supply system

The operating South Ukrainian NPP is supplied with water using the specially created Tashlyk cooling reservoir in the Tashlyk Gully, which is fed from the Southern Bug River.

Due to the construction of the Tashlyk PSPP, a dam separates the deepest part of the cooling pond. This causes an additional burden on the rest of the reservoir used for the operation of existing SUNPP units.

The Tashlyk Reservoir provides for the operation of the NPP at a nominal capacity of 3000 MW only in the cold season. In the summer, as a result of the temperature increase in the circulating water, the plant's capacity is limited to 1800 MW, and on particularly hot days - to 1500 MW.

The commissioning of five spray pools will reduce the circulating water temperature by 7-8°C and allow the SUNPP to operate at its nominal capacity of 3000 MW.

It is necessary to envisage the recycled service water supply system for the new SUNPP units with water cooling at cooling towers and spray pools.

2.1.5 Transport connection

The new NPP units are to be supplied with road access from the existing access roads of the South Ukraine NPP.

The national highway Ulyanivka - Mykolaiv runs from south to north near the NPP industrial site and the city of Yuzhnoukrainsk. It connects the NPP with the network of other highways. The Oleksiivka-Arbuzynka-Yelanets highway runs in the eastern direction. Most settlements within the 30-kilometer zone have paved access roads.

The new NPP units are to be supplied with railroad service from the existing single-track access railroad that serves the South Ukrainian NPP.

2.1.6 Current condition of the land plot

The existing power units are located on the site

the area is 317.96 hectares, and the type of ownership is stateowned. The 43.2-hectare plot of this area can be used for the construction of a new power unit, but there is a shortage of 100 hectares of land at the SUNPP site for the construction of temporary buildings and facilities for the construction and assembly base.

2.2 Construction conditions at the Rivne NPP site

2.2.1 Brief site description

The Rivne NPP is located in the northern part of the Western region of the IPS of Ukraine. The 110, 220, 330, and 750 kV grids have been developed in this region.

Four power units are in operation at the Rivne NPP:

- Units 1 and 2 with the V-213 (WWER-440) reactor;
- Units 3 and 4 with the V-320 (WWER-1000) reactor.

The total capacity of the NPP is 2835 MW. Table 2.2 shows the general characteristics of power units and their lifetime according to the licenses to carry out activities at the life cycle stage "operation of nuclear facilities of Rivne NPP power units" issued to the operating organization - National Nuclear Energy Generating Company Energoatom JSC by the State Nuclear Regulatory Inspectorate of Ukraine.

Power Unit number	Reactor type	Commissioning	Estimated lifetime end	Installed capacity, MW
Power unit No 1	WWER- 400/ V-213	22.12.1980	The lifetime is extended until 22.12.2030	420
Power unit No 2	WWER -440/ V-213	22.12.1981	The lifetime is extended until 22.12.2031	415
Power unit No 3	WWER- 1000/V -320	22.12.1986	The lifetime is extended until 11.12.2037	1000
Power unit No 4	WWER -1000/ V-320	10.10.2004	07.06.2035	1000

Table 2.2 – General characteristics of the RNPP power units

The site has the infrastructure to support NPP operations. There is a possibility of deploying operational and construction and assembly specialists in the existing satellite city and the availability of construction materials production. It is the developed road and rail transportation network.

2.2.1.1 Demographic conditions

The RNPP site supervised area (30 km) in administrative terms includes most of the Varash district of the Rivne region and part of the Kamen-Kashyr district of the Volyn region. The total area is 2826 km². There are 106 settlements with a total population of 130 thousand people on this territory.

2.2.2 Scheme of the NPP situational plan. A disposed land area

Currently, the four operating power units are located on a plot of about 218 hectares of which 12 hectares can be used for new power units. To accommodate two new power units with a capacity of 1100 MW each at the Rivne NPP site, an additional area of about 138 hectares will be required near the NPP industrial site.

The site layout is shown in Figure 2.2.

2.2.3 Power output

The Rivne NPP connection scheme is aimed at supplying power to the backbone grid of the Western region of the IPS of Ukraine.

2.2.4 Service water supply system

The main source of water for RNPP is the Styr River. Water intake is performed from the river itself without the use of any retaining structures. The service water supply system at RNPP is a reverse system with circulating water cooled by six tower cooling towers with a capacity of up to 100,000 m³/h each, designed to provide cooling water for turbine condensers and other heat exchangers in the NPP's engine and reactor buildings.

A supplementary water pumping station with bucket intake is located on the Styr River to replenish water losses in the service water supply system.

Currently, the estimated consumption of additional water to the service water supply system and for the needs of other consumers is 10,660 m³/h based on the following:

– evaporation losses in cooling towers and from the canals` water surface - $8,700 \text{ m}^3/\text{h}$; system blowdown - $1,715 \text{ m}^3/\text{h}$;

- removal and filtration $655 \text{ m}^3/\text{h}$;
- for service water supply to other consumers $690 \text{ m}^3/\text{h}$.

To analyze the current situation and prospective water supply for the new power unit construction, JSC KIEP (Kyiv Research and Design Institute Energoproject) performed the "Calculation of the water balance for the Styr River" 381817.216.001(4, 5).3T00

According to this calculation, if based on the sanitary water flow rate below the RNPP – $12.1m^3/s$ (sanitary flow rate determined by the minimum average annual flow rate of 95% availability at the Lutsk station), there is a small water shortage in the sections from the source to the Lutsk station, from the Lutsk station to RNPP, and from RNPP to the Mlynok station in low-water years. At the same time, given the approved sanitary flow rate of $8.8m^3/s$ and the maximum water intake limit for RNPP ($2.86m^3/s$), which correspond to the special water use permit No. 87/RV/49d-23 dated 14.07.2023, there are no water shortages. In any case, even under conditions of abnormally low water content of the Styr River, water supply to the new RNPP units will be guaranteed.


Figure 2.2 – Situation layout scheme of new power units at Rivne NPP

2.2.5 Transport connection

The new NPP unit is to be supplied with railroad service from the existing single-track access railroad that serves the RNPP.

The new NPP units are to be supplied with road access from the existing RNPP access roads.

The connection with the general road network is provided by the station access road.

2.2.6 The current land plot state

The existing power units are located on the site the type of ownership is state-owned, and the area is 217.895 hectares, of which 12 hectares can be used for the new power units placement.

This area is only sufficient to locate the main buildings of the nuclear and turbine sites, but not enough to site all buildings of the power unit technological complex and to locate temporary buildings and structures of the construction and assembly base. Therefore, the space shortage at the RNPP site is approximately 138 hectares. It is necessary to allocate additional land plots to implement the project for the new power units construction.

The soils in the Rivne NPP area are prone to sufosion and karst services. Rocks that are easily dissolved by water, such as gypsum and limestone; cover the area and there is a large amount of precipitation, which leads to soil leaching.

The RNPP site is located in unfavorable soil conditions, which can lead to instability of buildings and structures constructed on these soils, subsidence, and uneven settlement of buildings, which can cause cracks, deformations, and other damage. These conditions can be challenging for nuclear power plant construction due to possible risks of groundwater contamination.

To provide for RNPP safety, dams and other facilities have been built to protect the plant from floods, soil reinforcement has been performed; and the soil condition and RNPP buildings are constantly monitored. Despite these measures, sufosion and karst services remain a serious threat to NPPs, which is a serious problem for the new power unit construction and requires further technical measures.

2.3 Construction conditions at the KhNPP site

2.3.1 Brief site description

KhNPP is located in the northern part of the South-Western region of the IPS of Ukraine.

KhNPP operates two power units (No. 1 and No. 2) with a V-320 (WWER-1000) reactor. The total capacity of the NPP is 2000 MW. Basic information about the power units, including their lifetime, according to the licenses to carry out activities at the life cycle stage "operation of nuclear facilities of Khmelnytskyy NPP" issued to the operating organization, JSC National Nuclear Energy Generating Company Energoatom by the State Nuclear Regulatory Inspectorate of Ukraine, is presented in Table 2.3.

Power Unit number	Reactor type	Commissioning	Estimated lifetime end	Installed capacity, MW
Power unit No 1	WWER -1000/ V-320	22.12.1987	The lifetime is extended until 13.12.2028	1000
Power unit No 2	WWER -1000/ V-320	07.08.2004	07.09.2035	1000

Table 2.3 –	General	characteristics	of the	KhNPP	units
	~~~~		· · · · · ·		

The site already has a well-developed infrastructure to support NPP operations. There is a possibility of deploying operating and construction and assembly specialists in the existing satellite town. Besides, there is a production of construction materials and a developed road and rail transportation network.

2.3.1.1 Demographic conditions

The KhNPP supervised area includes a part of Khmelnytskyy (Shepetivka district)

and Rivne regions (Rivne district).

The supervised area is 2826 km², with 1024 km² being the territory of the Rivne region and 1802 km² of the Khmelnytskyy region.

The area includes 207 settlements with a population of over 195 thousand people. The majority of the population lives in the Khmelnytskyy region - about 136 thousand people.

## 2.3.2 Scheme of the NPP situational plan. A disposed land area

Currently, two operating units 1 and 2, and the uncompleted units 3 and 4 are located on a site with a total area of 147.0804 hectares, and the new units 5 and 6 will be located on the same site. The area will be additionally required to construct the temporary construction and assembly facilities.

The site layout is shown in Figure 2.3.



Figure 2.3 – Situation layout scheme of the KhNPP Units 5 and 6

# 2.3.3 Power output

The connection scheme of KhNPP is focused on the power plant's output to the backbone grid of the South-Western region of the IPS of Ukraine.



## 2.3.4 Service water supply system

The Khmelnytskyi NPP is fed by water from a cooling pond created on the Hnylyi Rih River, which is a tributary of the Vilia River, which in turn is a tributary of the Horyn River.

To fill the reservoir and recharge it during the spring flood (March-April), a pumping station was built to supply water from the Horyn River. According to the permit for special water use, the cooler reservoir water recharge regime is carried out only during the spring flood (March-April). Otherwise, KhNPP service water supply is supposed to be provided by using the useful volume of the reservoir. The amount of water recharge to the KhNPP service water supply system at the 4000 MW plant capacity will be 54.2 million  $m^3/year$ .

The designed service water supply plan for power units 5 and 6 is reverse with cooling towers as radiators and is designed using existing hydraulic structures. The KhNPP cooling pond is a source of water intake for cooling tower feeding and purging and for the needs of other technological systems in the adopted cooling and process water supply plan for power units 5 and 6. The water supply scheme is unified and separate for each power unit.

In 2023, affiliate Atomprojectengineering performed a thermal calculation of the KhNPP cooling pond [34], considering the construction of power units 5 and 6 with the AP1000 reactor unit, which resulted in determining the temperature of the cooled water in the pond during operation of one, two, three, four and six power units with a capacity of 1000 MW each:

- during the operation of two power units in summer, the water temperature in the underwater channel (at the inlet to the turbine condensers) is 24.4-25.5 °C;

during the operation of three power units in summer, the water temperature is 30.0-30.8 °C, which is less than the standard temperature value of 33.0 °C;

- during operation of four power units, the water temperature in summer months is 33.0-33.20 °C and will already reach the critical temperature of 33.0 °C under technological conditions [27];

- during the operation of power units 5 and 6 – in accordance with the AP1000 process flow diagram, developed by Westinghouse, the purge water of the cooling tower basin of the CWS system is combined with the purge flow from the SWS system, taken

from the pump discharge header and directed to the purge sump of the wastewater collection and disposal system for further discharge into the KhNPP cooling pond. Based on the operating conditions of this cooling and blowdown water discharge plan, it can be concluded that the cooling pond receives water already cooled to ambient temperature. Thus, the operation of power units 5 and 6 will not affect the value of additional evaporation from the pond and the temperature in the pond as a whole;

It is envisaged to construct a flow control dam for the operation of units 3 and 4.

The cooling system for KhNPP units 5 and 6 involves the use of cooling towers:

- for the CWS system of units 5 and 6, a naturally ventilated cooling tower is adopted to receive heated circulating water from the main condensers and support the heat exchangers. The circulating water is cooled by evaporation and returned to the cooling tower basin. One cooling tower is provided for each of power units 5 and 6.

- the SWS service water of power units 5 and 6 is cooled by a fan cooling tower. The cooling tower is a straight, counterflow tower with a suction fan and a clog-resistant film fill. The cooling tower is divided into two chambers. Each chamber utilizes a single propeller-type fan located at the top of the chamber to draw air upward through the fill against the downward water flow.

## 2.3.5 Transport connection

The new NPP units are to be supplied with road transport from the existing KhNPP access roads.

The KhNPP is located in an area with a developed road network.

The railroad connection of the new NPP units is planned to be provided from the existing access railroad.

## 2.3.6 Current condition of the land plot

Currently, two operating units 1 and 2, and uncompleted units 3 and 4 are located on a site with a total area of 147.0804 hectares; the new units 5 and 6 will be located on a site with an area of 56.8 hectares. The area will be additionally required to construct the temporary facilities for the construction and assembly base.

## 2.4 Conclusion on the selection of a site for the new power units construction

Hydrogeological conditions at the Rivne NPP site require special measures to prevent possible karst formation under the influence of natural and technogenic factors. The water supply of Rivne NPP is sufficient for the 6 units with installed capacity more than 5000 MWt.

The water supply of the South Ukrainian NPP is sufficient in the cold season for the capacity of 3000 MWt, which is exhausted by the power units 1-3 that are in operation. New units can be commissioned subject to the implementation of additional technical measures for a water supply, such as the use of tower cooling towers and spray pools, but it should be taken into account that these measures lead to an increase in the share of consumption for own needs up to 10-15%.

Following the site selection studies, construction of new power units at KhNPP is more optimal than at SUNPP and RNPP.

The Feasibility Study confirms the suitability of the KhNPP site for the construction of Units 5 and 6 following the requirements of the current regulatory documents.

Since new power units cannot be less reliable and safe compared to operating power units, the environmental impact is a conservative estimate of the possible new units impact. The actual impact of the RNPP, KhNPP, and NPP units during their normal operation is orders of magnitude less than the established threshold levels.

## 3 A DESCRIPTION OF THE CURRENT ENVIRONMENTAL STATE (BASE SCENARIO) AND A DESCRIPTION OF ITS POSSIBLE CHANGE WITHOUT THE PLANNED ACTIVITY TO THE EXTENT THAT NATURAL CHANGES FROM THE BASE SCENARIO CAN BE ESTIMATED BASED ON AVAILABLE ENVIRONMENTAL INFORMATION AND SCIENTIFIC KNOWLEDGE

The description of the current environmental state is an integral part of the EIA report, and provides an overview of the main environmental characteristics, such as water sources, atmospheric air, terrain, climate, soils, etc. Information from the KhNPP Unit 2 Final Safety Analysis Report was used in this section [35].

#### 3.1 Geographical location of the site

The KhNPP site is located in Shepetivka district of Khmelnytskyi region on the left bank of the Horyn River, 30 km southwest of the district center – Shepetivka. The town of Netishyn is located 3 km north of the site. To the northwest of the site, at a distance of 50 km, is the city of Rivne (the regional center of Rivne region). The city of Slavuta is located 15 km to the east.

#### 3.2 Terrain and landscapes

The territory and site of the KhNPP are located on the Volyn-Podolian plate, which is situated between the western slopes of the Ukrainian Shield and the Carpathian Alpine geosyncline. The plate consists of an Archean-Middle Proterozoic metamorphic basement and an Upper Proterozoic-Paleozoic sedimentary cover. The surface of the basement gently dips to the west and northwest. The basement dips in step faults of meridional and submeridional strike. The basement structure of the plate is divided into separate blocks by a system of faults. Due to tectonic movements, the blocks moved in relation to each other in vertical and horizontal directions, which led to uneven accumulation of sedimentary complexes.

Most of the KhNPP supervised area belongs to the Horyn River basin. The eastern spurs of the Volyn Upland (Goshcha Plateau) represent the right bank of the Horyn basin. In the left-bank part of the basin, there is an alternation of wavy and low-wavy plains of different levels: Podolian (in the south area) and Volyn Upland dissected by the valleys of the left tributaries of the Horyn. The Rivne Plateau in the north and the Ostroh Plateau in the southeast of the Upland represent the Volyn Upland on the left bank of the Horyn. The Podolian and Volyn Uplands are separated by the lower plains of Male Polissia, almost entirely located in the left-bank part of the 30-kilometer area, and the Mizotskyi Ridge.

The modern Volyn-Podolian terrain plate is represented by plains of different genesis and morphological design. Karst landforms are on the cretaceous deposits of the Slavuta Plain, within the Volyn Upland, on the second and third terraces of the Horyn

River. They are mainly represented by sinkholes located in separate groups or chains. The main karst blocks are located to the west of Slavuta, near the urban area of Radoshivka, in the upper reaches of the Hnylyi Rih River, and urban area of Storonyche. A network of ravines and gullies represents the gully terrain. This terrain is mostly within denudation and structural-denudation plains. The gully network reaches its greatest density in the southern part of the territory and is confined to the slopes of the Horyn River valley and Podolian Upland. The largest number of active ravines is observed within the Podolian Upland.

Dunes and ridges of various configurations are found in large quantities within the interfluves. Their relative height reaches from 5 to 25 m, and they are oriented in the western and southwestern directions.

The landscapes of the Volyn physical and geographical region are dominated by hilly forest plains with erosive terrain - gullies, ravines, and water galls - that affect the processes of redistribution of chemicals with water flows and lead to differentiation of vegetation and soil cover. The terrain fragmentation in the Volyn physiographic region contributes to intensive rain and meltwater runoff.

There are 15 physical and geographical (landscape) districts within the Volyn region: Novoselskyi, Novomalynskyi, Hoshchanskyi, Zavozovskyi, Mirotynskyi, Hremyachivskyi, Myliatynskyi, Bochanetskyi, Annopolskyi, Badovskyi, Prykorchytskyi, Ostrohskyi, Netishynskyi, Mizotskyi-Bushevskyi and Mynkovetskyi.

The NPP construction site is located in the Malopilska lowland, on the first floodplain terrace of the Horyn River. The NPP facilities occupy alienated lands of the Slavuta and Iziaslav State Forestry Funds, collective farms, and the territory of residential development in Dorohohoscha village. The NPP sanitary protection area includes the lands of Slavuta State Forestry, Izyaslav State Forestry, collective farms and the territory of Siltse village, whose population was settled out in accordance with the requirements for facilities that can be located in the sanitary protection area.

## 3.3 Soils

Most of the soil cover in the region around NPP consists of acidic sod-podzolic soils of light composition, which are highly acidic and permeable.

The absorption complex of these soils is unsaturated, which leads to the fact that colloidal complexes in the form of exchangeable state absorb some elements in the upper soil layer. To increase the fertility of these soils, reduce migration processes and increase resistance to anthropogenic impacts, it is necessary to liming and applying organic and mineral fertilizers.

A significant area of the controlled territory includes deep chernozems, dark gray, podzolized soils, and podzolized, meadow chernozem and meadow soils, which are highly resistant to anthropogenic loads [36]. This is facilitated by such soil properties as a close to neutral reaction of the environment, a much higher humus content, high absorption capacity, etc. These properties cause low mobility and almost complete fixation of heavy metals and radionuclides by the absorbing complex.

The development of exogenous geological processes is not expected at the KhNPP site, since the redeposited chalk was removed and replaced with sand at the location of the main structures (main buildings of power units 1-6), which created conditions for infiltration of precipitation and thus prevented waterlogging.

#### 3.4 Climate

The KhNPP site is located in the northwestern part of Ukraine in the Volyn Polissya region, in a moderate continental climate area with a positive moisture balance. Relatively high temperatures and low relative humidity characterize this type of climate in summer and low temperatures, high humidity and snow cover in winter.

According to the climatic zoning [37], the territory under consideration is located in the second climatic region (sub-region I I-B).

The climate of the region is formed under the influence of both sea and continental air masses. The nature and intensity of the main climate-forming factors varies significantly by season.

*Winter.* Cyclonic activity is most pronounced. Short-term warming, intense snowfall, strong winds and blizzards often accompany the passage of western and northwestern cyclones. The coldest winter month is January (average monthly temperature is minus  $5.2^{\circ}$ C).

*Spring.* Given a significant fluctuation in heat and cold, the spring transition is characterized by an active rise in temperature, intense snowmelt, and increased evaporation from the soil and water surface. In April and May, there is often a return of cold weather caused by the intrusion of Arctic air, which causes sharp cold snaps and frosts.

*Summer*. A summer season is characterized by a significant increase in air temperature due to the warming of the earth's surface, a high frequency of clear days, rare fogs, increased precipitation and active thunderstorm. The summer season begins in mid-May. The highest temperature in summer is observed mainly in July, with an average temperature of  $+19.5^{\circ}$ C, an average maximum of  $+25.4^{\circ}$ C, and an average minimum of  $+14.1^{\circ}$ C. In 35-40% of cases, the highest air temperatures are observed in August or June. Summer is characterized by heavy rainfall.

*Autumn*. Solar radiation decreases and the air begins to cool. The first half of autumn is characterized by mostly dry, warm weather without precipitation. The second half of autumn is characterized by a general deterioration of the weather, a large number of cloudy days, heavy precipitation and prolonged fog. The passage of western cyclones during this period is often accompanied by increased winds and ice. The atmospheric circulation in the second half of autumn is close to the winter season.

The average annual air temperature within the KhNPP supervised area is 7.1°C, the absolute maximum is 36.6°C, and the absolute minimum is minus 33.6°C.

The annual average monthly air temperatures in the KhNPP area is characterized by the highest values in July (18.4 °C) and the lowest in January (minus 5.2 °C).

The frost-free period lasts an average of 169 days.

The average annual relative humidity is 79%; the average annual partial pressure of water vapor is 8.9 hPa; the saturation deficit is 3.3 hPa.

The average annual cloudiness within the area under consideration is 6.7 points for total cloudiness and 4.9 points for partial cloudiness. In terms of precipitation, the area under consideration belongs to the area of sufficient moisture. The annual precipitation is 667 mm. About 70% of the annual precipitation falls during the warm season.

The average number of days with precipitation within the area is 163 days.

The number of days with snow cover averages 90 days. The average ten-day snow cover is 12 cm, the maximum is 52-53 cm.

The annual amount of evaporation from the land surface (total evaporation) is 538 mm, the highest monthly amount (106 mm) is in July. In the winter months, the total evaporation has the lowest values of 2-7 mm.

Prevailing wind direction:

- for the whole year: westerly direction 20.4%; a calm is 11.9%;
- in the warm period: westerly direction 19.9%; a calm is 13.9%;
- in the cold period: westerly direction 21.5%; a calm is 8.0%.

The average annual wind speed is 2.7 m/s. The lowest average monthly wind speed is observed in the summer months and ranges from 1.9 to 2.2 m/s. In winter, the average monthly wind speeds range from 3.1 to 3.4 m/s.

The maximum wind speeds recorded by the aeronautical meteorological stations closest to the KhNPP site (Shepetivka, Khmelnytskyy, Rivne and Lutsk) reached 28, 34, 38 and 40 m/s [38].

During the year, there are on average 62 days with fog and 31 days with thunderstorms in the KhNPP area.

*Natural meteorological phenomena* on the territory bordering the KhNPP site, within a radius of 200 km (Khmelnytskyi, Rivne, Volyn, Ternopil and partially Vinnytsia and Lviv regions) include: strong winds, squalls, hurricane winds, tornadoes (from zero to second intensity class), heavy rain, heavy blizzards, heavy snowfall, heavy ice. These natural meteorological phenomena are considered particularly dangerous. In some cases, these phenomena led to catastrophic consequences and caused significant damage to the agriculture. However, during KhNPP operation, natural meteorological phenomena that occurred in the territories adjacent to the plant did not create emergencies at NPP.

The aerometeorological conditions in the KhNPP area are characterized by an increased probability of a stably stratified atmosphere at surface and low elevated inversions and low-power mixing layers (especially in winter), which weakens the mechanism of natural self-purification of atmospheric air in this region.

#### 3.5 Hydrological characteristics

Hydrological characterization is a key tool in understanding the water cycle and water resources. It provides comprehensive information on various aspects of the water environment. It is important to provide a hydrological characterization of surface and groundwater in the EIA report.

# 3.5.1 Surface water

The Horyn River basin, as well as lakes, ponds, reservoirs, and a reclamation network of canals represent the hydrographic network in the 30-kilometer area of KhNPP impact.

Within the 30-kilometer area of the KhNPP, 37 tributaries ranging in length from 10 to 80 km and approximately 220 tributaries less than 10 km long flow into the river.

The Horyn and Hnylyi Rih rivers are the sources of service water supply for KhNPP power units.

The Horyn River is one of the largest tributaries of the Pripyat River, which in turn is the largest tributary of the Dnipro River. According to the Hydrometeorological Service, the river is 659 km long, with a basin area of 27.7 thousand km² and an average slope of 29 cm/km.

The Horyn River catchment in the upstream is located within the Volyn-Podolian Upland and in the midstream and downstream within the Polissia Lowland. The Horyn River has its source in Ternopil Oblast, northwest of Volytsia village.

The largest left tributaries of the Horyn river are the Vilia (catchment area of 1815 km²) and the Ustia (762 km²), with the Vyrka (261 km²) being much smaller. The largest right tributary of the Horyn River is the Sluch River, with a catchment area of 13.9 thousand km², or almost half of the entire Horyn catchment. These rivers merge a few tens of kilometers before the border with Belarus. The right tributaries include the Tsvitokha River with a catchment area of 368 km².

It is in the Vilia River basin that the KhNPP cooling pond was created, more precisely, on its right tributary, the Hnylyi Rih River. The length of this small river is 28 km and the catchment area is 201 km². The Hnylyi Rih River is a right-bank tributary of the Vilia River and flows into it one kilometer from the mouth. It originates in the north-east of Mokrets village at an altitude of 230 m above sea level and flows through the Volyn Upland. The river basin is swampy. The river valley and floodplain do not have distinct boundaries.

The reservoir was filled due to the flow of the Hnylyi Rih River and intake from the Horyn River. The inlet channel site from this river is located 204.0 km from the source (455.0 km from the mouth). The total annual water intake into the reservoir (including the volume supplied from the Horyn River) is determined in accordance with the current Special Water Use Permit. The main hydrological characteristics of the Hnylyi Rih River and the Horyn River are shown in Table 3.1. The dynamics of KhNPP water consumption during the operation of Units 1 and 2 is shown in Table 3.2.

Catchment basin	Area of the catchment basin up to the	Alimentation type	Annual runo (mln. m ^a provisi	off volume ³ ) with on of	Flooding
	hydropower plant site, km ²		50%	75%	pened
the Hnylyi Rih river	201.0	Surface runoff and groundwater	24.12	21.1	mid-March
the Horyn river	3830.0	Surface runoff and groundwater	501.4	400.0	mid-March

Table 3.1 – Main hydrological characteristics of the Hnylyi Rih and Horyn rivers

Supervisory control of the Hnylyi Rih river flow is not performed. The Hnylyi Rih river flow is completely accumulated in the reservoir. Flood releases from the Hnylyi Rih River after the reservoir is filled to the NWL level are carried out automatically through the flood spillway.

As a result of the reservoir measurements carried out by LvivODRES in 2007, the average reservoir depth was 6.1 m and the maximum depth was 18.2 m. On the non-pressure reservoir side (left bank), there is a shallow water area 100-300 meters wide and up to two meters deep. The water volume in the reservoir at the water level of 202.800 meters was 115.55 million m³. This volume of water corresponds to the design data, which concludes that the reservoir is slightly silted up, despite its long operational period.

Table 3.2 – Dynamics of KhNPP water consumption during operation of units 1&2

No	Water supply source	Water used, thousand m ³					
JI⊡	water suppry source	2023	2022	2021	2020	2019	
1	Artesian	999.6	1060.6	1378.3	1277.7	1471.4	
2	Service	29017.9	22882.8	39660.8	30664.0	14471.8	
2.1	the Hnylyi Rih river	14901.1	15986.4	16253.1	7889.2	10909.2	
2.2	the Horyn river	14116.8	6896.4	23407.7	22774.8	3562.6	

**Notes:** The table shows water use values for KhNPP only (excluding losses and transfer of water resources to the population or other enterprises).

The total number of lakes in the KhNPP area (in river basins with a length of  $\geq 10$  km) is 111, with a total water surface area of 5.92 km². The largest number of lakes is located in the basins of the Vilia (28–1.55 km²) and Tsvitoha (22–1.02 km²) rivers.

The total number of reservoirs in the KhNPP supervised area is three. The largest is the NPP cooling pond.

## 3.5.2 Groundwater

The diversity of groundwater distribution and formation conditions, its chemical composition, feeding and discharge are determined by the geological structure, geomorphological and climatic factors.

In terms of hydrology, KhNPP units 5&6 are located within the Volyn-Podolian fractured water basin. The NPP area is characterized by predominantly planned groundwater flows with the general direction of movement to the regional drain – the Pripyat River, which is due to a significant predominance of their lateral extent compared to the power and relatively low hydrographic terrain.

Currently, two aquifers have been identified [39]:

1. The first groundwater aquifer from the surface was recorded at a depth of 1.40-3.50 meters. The water-bearing soils are medium-sized sands of anthropogenic origin and partially light and heavy loams. The conditional water-resistant layer is a thickness of siltstone and mudstone rocks, the roof of which lies at a depth of 3.00-7.90 m. The aquifer is recharged by losses from water supply systems and infiltration of precipitation, as well as by upward flow from the pressure aquifer. The aquifer is discharged in the southwestern direction towards the KhNPP cooling pond;

2. Lower Neogene pressure water horizon. The depth of the piezometric level of pressure water is 1.40 - 4.20 m. The appearance of water in the pressure aquifer was recorded at a depth of 11.30 - 15.90 m. The pressure water horizon is confined mainly to fractured sandstones. The upper and lower water-resistant layer is made up of siltstone and mudstone. The water masses migrate mainly through fractured sandstones and through the area of siltstone-argillite rocks.

The regime of the first aquifer from the surface is unstable and depends on climatic and anthropogenic factors. The level of the aquifer directly depends on the amount of infiltration losses and accidental leaks.

The pressure water regime is characterized as closed and not subject to seasonal fluctuations.

The level regime of the first groundwater aquifer from the surface is in a stable state.

The chemical composition of the groundwater of the first horizon is of the bicarbonate-calcium type and is not aggressive to all types of concrete.

The water of the pressure horizon is of the sulfate-chloride-natrium-magnesium type. They are highly aggressive to Portland cement concrete.

The main changes in hydrodynamic conditions are localized in the upper area of the superficially developed system of three aquifer complexes: sedimentary complex of Quaternary, Neogene, Paleogene and Upper Cretaceous rocks; Upper Proterozoic (Vendian) rocks of the Mohyla-Podilska and Kanylivska series; rocks of the Volynska series (Gorbashevska Formation) and Polissya series of the Upper Proterozoic.

The regional distribution of piezometric surfaces of the second and third aquifers is similar to the distribution of levels of the first aquifer (regional direction of movement, dynamics of slopes, etc.).

The radiation status of groundwater, including the Netishyn water intake, is satisfactory, i.e. below the limit level regulated by regulatory documents.

# 3.6 Generalized characteristics of flora and fauna

A generalized characterization of flora and fauna is an important step in the study of biological diversity and ecosystems of a particular territory and ecosystems of a territory.

## **3.6.1 Floristic complexes**

The KhNPP supervised area is located at the intersection of three geobotanical districts. The northern part belongs to the southern edge of the Volyn Upland, the territory of which is heavily plowed. The natural vegetation consists of oak, hornbeam-oak forests and meadow steppes. Typical features of Polissya nature (Male Polissya) characterize the central and eastern parts, where oak-pine, hornbeam-oak-pine and pine forests prevail, with mesotrophic bogs interspersed in the lower parts. The southern part is similar to the terrain, soils and vegetation of the Volyn forest plateau and is located on the spurs of the Podolian Upland. All three areas are crossed by water streams (e.g., the Horyn river and its tributaries), where azonal hydrophilic vegetation (aquatic, marsh, meadow) is formed.

The anthropogenic factor has influenced the structure and distribution of vegetation cover. The diversity of the plant world is determined by the complexity of the terrain, uneven moisture supply and the diversity of quaternary sediments.

The supervised area flora belongs to the flora of the migratory type and consists of various species that have arisen as a result of the influence of different development centers. This area is dominated by forests, in particular pine forests, which occupy the upper parts of the slopes and peaks, and are replaced by oak-pine forests in the lowlands. The following main groups of associations characterize pine forests: lichen, green lichen, blueberry-green lichen, and molinia. Oak-pine forests are widespread throughout the territory and are the main subformation. Hornbeam-oak and hornbeam-pine-oak forests cover smaller areas, while alder and birch forests occur in relatively small areas.

In 2024, the National Museum of Natural History of the National Academy of Sciences of Ukraine assessed the impact of the planned activity on construction of new units at KhNPP on populations of rare plant species of international, national and regional protection status, rare plant communities, habitats of the Emerald Network and nature reserve sites [77].

The work was carried out to identify, inventory and characterize the rare phytodiversity of plant species of national, regional and international conservation status, rare plant communities and habitats of the Emerald Network in the KhNPP affected area.

The phytodiversity of vascular plants of summer vegetation within the KhNPP construction site, the dam of the KhNPP cooling pond and its natural coastline was studied in detail. Particular attention was paid to the study of rare species of state, regional and international protection, rare coenoses listed in the Green Book of Ukraine, habitats of the Emerald Network and objects of the nature reserve network of Ukraine.

In the KhNPP impact area and adjacent protected areas, 22 plant species from the Red Book of Ukraine grow, of which 4 species are included in Appendix I of the Bern Convention; 20 species, mostly of the LC category, are included in the World Red List.

During floristic surveys in the KhNPP construction site area, 133 species of higher vascular plants were identified, of which 96 were native species and 37 (27.8%) were adventitious.

Along the KhNPP cooling pond water pressure dam (10m from the road and on concrete slopes), 108 species of higher vascular plants were found, including 85 (78.7%) native species and 23 (21.3%) adventitious species.

Field studies are in progress, which will result in the establishment of population phytomonitoring for model rare species, rare plant communities, and invasive adventitious species.

№	Ukrainian species name	Latin species name	Presence on the National Nature Park territory	Sozological status	Family	The authors of the article in the Red Book of Ukraine (RBU)
1.	Fir clubmoss	<i>Huperzia selago</i> (L.) Bernh. ex Schrank et Mart.	Male Polissia, Dermansko- Ostrozkyi	RBU(2009)- <b>Endangered</b> IUCN (2024) – (Global, EU) LC	Lycopsids – Huperziaceae	S.M. Panchenko, I.I. Chornei, N.M. Sychak, V.I. Melnyk (p. 19).
2.	Sword-leaved helleborine	<i>Cephalanthera longifolia</i> (L.) Fritsch.	Dermansko- Ostrozkyi	CITES(2015) RBU(2009)- <b>Rare</b> IUCN (2024) – (EU) LC	Orchid – Orchidaceae	I.A. Timchenko, V.I. Goncharenko, O.O. Orlov, S.M. Panchenko, (p. 159).
3.	Red helleborine	<i>Cephalanthera rubra</i> (L.) Rich.	Dermansko- Ostrozkyi	CITES(2015) RBU (2009)- <b>Rare</b> IUCN (2024) – (EU) LC	Orchid – Orchidaceae	I.A. Timchenko, V.I. Goncharenko, O.O. Orlov, S.M. Panchenko (p. 160).
4.	Swamp willow	Salix myrtilloides L.	Male Polissia, Dermansko- Ostrozkyi	RBU (2009)- Vulnerable	Willow – Salicaceae	T.L. Andrienko, O.O. Kahalo (p. 586).
5.	Garland flower (spurge flax, cepe)	Daphne cneorum L.	Dermansko- Ostrozkyi	RBU (2009)- Vulnerable	Flowering plants – Thymelaeaceae	Y.P. Didukh, (p. 608).
6.	Water caltrop	Trapa natans L. s.l.	_	RBU (2009)- <b>Endangered</b> BK (2002) IUCN (2024) – (Global) LC	Water chestnut – Trapaceae	D.V. Dubina, G.A. Chorna (p. 612).
7.	Bird's-nest orchid	<i>Neottia nidus-avis</i> (L.) Rich.	Dermansko- Ostrozkyi	RBU (2009)- <b>Endangered</b> CITES(2015) IUCN (2024) – (Global, EU) LC	Orchid – Orchidaceae	I.A. Timchenko, O.O. Orlov, S.M. Panchenko (p. 196).
8.	Yellow widelip orchid	<i>Liparis loeselii</i> (L.) Rich.	Dermansko- Ostrozkyi	RBU (2009)- <b>Vulnerable</b> BK (2002) CITES(2015)	Orchid i – Orchidaceae	Orlov O.O., Chornei I.I., Goncharenko V.I. (p. 190).
9.	Eggleaf twayblade	Listera ovata (L.) R.Br.	Dermansko- Ostrozkyi	RBU (2009)- <b>Endangered</b> CITES(2015) IUCN (2024) –(EU) LC	Orchid – Orchidaceae	I.A. Timchenko, O.O. Orlov, O.T. Kuzyarin,

Table 3.3 – List of state and internationally protected plant species predicted to be within the KhNPP impact area.

№	Ukrainian species name	Latin species name	Presence on the National Nature Park territory	Sozological status	Family	The authors of the article in the Red Book of Ukraine (RBU)
						S.M. Panchenko (p. 192)
10.	Early marsh-orchid	Dactylorhiza incarnata (L.) Soy s.l.	Male Polissia, Dermansko- Ostrozkyi	RBU (2009)- <b>Vulnerable</b> CITES(2015) IUCN (2024) – (EU) LC	Orchid – Orchidaceae	O.T. Kuzyarin, M.S. Kozyr I.A. Timchenko, O.V. Lukash S.M. Panchenko, O.M. Bayrak (p. 168).
11.	Heath spotted-orchid (moorland spotted orchid)	<i>Dactylorhiza maculata</i> (L.) Soó s.l.	Male Polissia	RBU (2009)- <b>Vulnerable</b> CITES(2015) IUCN (2024) – (EU) LC	Orchid – Orchidaceae	I.A. Timchenko, O.O. Orlov, S.M. Panchenko (p. 169).
12.	Broad-leaved marsh orchid	<i>Dactylorhiza majalis</i> (Rchb.) P.F.Hunt et Summerhayes s.l.	Dermansko- Ostrozkyi	RBU (2009)- <b>Rare</b> CITES(2015) IUCN (2024) – (EU) LC	Orchid – Orchidaceae	V.V. Protopopova (c. 170).
13.	Marsh helleborine	<i>Epipactis palustris</i> (L.) Crantz	Dermansko- Ostrozkyi	RBU (2009)- <b>Vulnerable</b> CITES(2015) IUCN (2024) – (Global, EU) LC	Orchid – Orchidaceae	I.A. Timchenko, O.T. Kuzyarin (p. 179).
14.	Dark-red helleborine	<i>Epipactis atrorubens</i> (Hoffm. ex Bernh.) Besser	Male Polissia, Dermansko- Ostrozkyi	RBU (2009)- <b>Vulnerable</b> CITES(2015) IUCN (2024) – (EU) LC	Orchid – Orchidaceae	I.A. Timchenko (p. 176).
15.	Broad-leaved helleborine	<i>Epipactis helleborine</i> (L.) Crantz	Male Polissia, Dermansko- Ostrozkyi	RBU (2009)- <b>Endangered</b> CITES(2015) IUCN (2024) – (EU) LC	Orchid – Orchidaceae	I.A. Timchenko, O.O. Orlov, S.M. Panchenko, O.M. Bayrak (p. 177).
16.	Martagon lily	Lilium martagon L.	Male Polissia, Dermansko- Ostrozkyi	RBU (2009)- <b>Endangered</b> IUCN (2024) – (EU) LC	Lily – Liliaceae	T.L. Andrienko (p. 141).

№	Ukrainian species name	Latin species name	Presence on the National Nature Park territory	Sozological status	Family	The authors of the article in the Red Book of Ukraine (RBU)
17.	Lesser butterfly-orchid	<i>Platanthera bifolia</i> (L.)	Dermansko-	RBU (2009)- Endangered	Orchid –	V.V. Protopopova,
		Rich.	Ostrozkyi	IUCN (2024) – (EU) LC	Orchidaceae	O. Oflov (p. 212).
18.	Stiff clubmoss	Lycopodium annotinum L.	Male Polissia, Dermansko- Ostrozkyi	RBU (2009)- <b>Vulnerable</b> IUCN (2024) – (EU) LC	Vascular plants – Lycopodiaceae	T.L. Andrienko, O.I. Pryadko (p. 18).
19.	Marsh clubmoss	<i>Lycopodiella inundata</i> (L.) Holub	Male Polissia	RBU (2009)- <b>Vulnerable</b> IUCN (2024) –(Global, EU) LC	Vascular plants – Lycopodiaceae	O. Pryadko (p. 17).
20.	Floating watermoss	Salvinia natans (L.) All.	_	RBU (2009)- <b>Endangered</b> BK (2002) IUCN (2024) – (Global) – LC, (EU) – NT	Heterosporous ferns – Salviniaceae	D.V. Dubina (p. 38).
21.	Eastern pasqueflower (cutleaf anemone)	<i>Pulsatilla patens</i> (L.) Mill. s.l.	Dermansko- Ostrozkyi	RBU (2009)- <b>Endangered</b> BK (2002) IUCN (2024) – (EU) – DD	Buttercup – Ranunculaceae	O.O. Kagalo, I.A. Korotchenko, O.V. Lukash (p. 565).
22.	Wild garlic	Allium ursinum L.	Dermansko- Ostrozkyi	RBU (2009)- <b>Endangered</b> IUCN (2024) –(EU) LC	Herbaceous – Alliaceae	T.L. Andrienko (p. 60).

# 3.6.2 Faunal complexes

The fauna is characterized by a high level of species diversity and rare species for Ukraine and Europe.

In 2024, the I.I. Schmalhausen Institute of Zoology of the National Academy of Sciences of Ukraine assessed the impact of the planned activities for the construction of new power units at KhNPP on the species of animals listed in Resolution 6 (1998) of the Convention on the Conservation of Wild Flora and Fauna and Natural Habitats (Bern Convention) [78].

The available data on animals as per Resolution 6 were collected and analysed in the KhNPP 30-km area and field surveys were conducted.

The analysis of available data within KhNPP 30-km area revealed the presence of 84 animal species as per Resolution 6: 1 species of molluscs, 7 species of insects, 3 species of fish, 3 species of amphibians and reptiles, 66 species of birds, and 4 species of mammals.

# 3.6.2.1 Terrestrial and aquatic molluscs (Mollusca)

The analysis of the Resolution 6 species list revealed that out of 33 species of molluscs in Ukraine, 6 species are found in Ukraine, and only one of them, Vertigo angustior Jeffreys, 1830, has been identified in the KhNPP 30-km area.

# 3.6.2.2 Insects (Insecta)

Based on the analysis of the species list from the Annex to Resolution 6 of the Standing Committee of the Berne Convention and information from the National Biodiversity Information Network (UkrBIN), the Global Biodiversity Information System (GBIF), the iNaturalist network, as well as all available literature data, it was found that out of 99 insect species included in the Annex to Resolution 6, 51 species live in Ukraine, and 7 species occur in the KhNPP 30-km area, namely:

- Ophiogomphus cecilia (Fourcroy, 1785) included in the Red Book of Ukraine;
- Lucanus cervus (Linnaeus, 1758) included in the Red Book of Ukraine;
- Cucujus cinnaberinus (Scopoli, 1763) included in the Red Book of Ukraine;
- Lycaena dispar (Haworth, 1802);
- Phengaris teleius (Bergsträsser, 1779);
- Phengaris nausithous (Bergsträsser, 1779);
- Euplagia quadripunctaria (Poda, 1761).
- 3.6.2.3 Lampreys and ray fishes (Petromyzonti and Actinopterygii)

There are about 30 species of lampreys and fish in Ukraine, and three species of fish from this international list have been reliably recorded within 30 kilometers around KhNPP, namely:

- Misgurnus fossilis (Linnaeus, 1758);

- Cobitis taenia (Linnaeus, 1758);

- Rhodeus amarus (Bloch, 1782).

3.6.2.4 Amphibians (Amphibia) and reptiles (Reptilia)

The analysis of the Resolution 6 species list showed that out of 4 reptile and 6 amphibian species listed in Resolution 6 for the territory of Ukraine, 1 and 2 species occur within KhNPP 30-km area, respectively, namely:

1. Reptiles – Emys orbicularis (Linnaeus, 1758)

2. Amphibians – Triturus cristatus (Laurenti, 1768), Bombina bombina (Linnaeus, 1761)

3.6.2.5 Birds (Aves)

Based on the analysis of available ornithological data, at least 66 species of birds listed in Annex 6 of the Bern Convention occur in the study area during all periods of the annual bird cycle, see Table 3.4.

<b>1</b>	1
Order	Species name
GAVIIFORMES	Gavia arctica
GAVIIFORMES	Gavia stellata
PODICIPEDIFORMES	Podiceps auritus
CICONIIF ORMES	Ardea purpurea
CICONIIF ORMES	Botaurus stellaris
CICONIIF ORMES	Casmerodius albus (Egretta alba)
CICONIIF ORMES	Egretta garzetta
CICONIIF ORMES	Ixobrychus minutus
CICONIIF ORMES	Nycticorax nycticorax
CICONIIF ORMES	Ciconia nigra
CICONIIF ORMES	Ciconia ciconia
CHARADRIIFORMES	Chlidonias hybridus
CHARADRIIFORMES	Chlidonias leucopterus
CHARADRIIFORMES	Chlidonias niger
CHARADRIIFORMES	Larus minutus
CHARADRIIFORMES	Sterna albifrons
CHARADRIIFORMES	Sterna caspia (Hydroprogne caspia)
CICONIIF ORMES CICONIIF ORMES CHARADRIIFORMES CHARADRIIFORMES CHARADRIIFORMES CHARADRIIFORMES CHARADRIIFORMES CHARADRIIFORMES	Ciconia nigra Ciconia ciconia Chlidonias hybridus Chlidonias leucopterus Chlidonias niger Larus minutus Sterna albifrons Sterna caspia (Hydroprogne caspia)

Table 3.4 – List of birds predicted within the KhNPP impact zone.

Species name
Sterna hirundo
Asio flammeus
Strix uralensis
Caprimulgus europaeus
Alcedo atthis
Dendrocopos leucotos
Dendrocopos médius
Dendrocopos syriacus
Dryocopus martius
Picoides tridactylus
Picus canus
Lullula arborea
Anthus campestris
Lanius collurio
Lanius minor
Luscinia svecica (Cyanosylvia svecica)
Sylvia nisoria
Ficedula albicollis
Ficedula parva

## 3.6.2.6 Mammals (Mammalia)

The Ukrainian list of mammals of the fauna included in Resolution 6 of the Bern Convention includes 24 species from 6 orders. Based on the analysis of available data (prior to the field surveys), the species known to occur within the KhNPP 30-km study area were identified, namely:

- 1. Order Chiroptera. Chiropterans
  - Myotis dasycneme (Boie, 1825)
  - Barbastella barbastellus (Schreber, 1774)
- 2. Order Carnivora. Raptors
  - Lutra lutra (Linnaeus, 1758)
- 3. Order Rodentia. Rodents
  - Castor fiber (Linnaeus, 1758)

The field research is in progress, which will result in the development of recommendations for the conservation of animal species of Resolution 6 of the Bern Convention.

# **3.7** Characterization of the nature reserve fund objects

Two national nature parks are located in the KhNPP supervised area:

- "Derman-Ostroh" (partially);

- "Male Polissia".

**The Derman-Ostroh National Nature Park** is located in the southern part of Rivne region, in the Zdolbuniv and Ostroh districts.

The park is based on 18 existing objects of the nature reserve fund, including the Bushchansky botanical reserve of national importance, botanical reserves of local importance Tract "Bir", "Kruglyak swamp", "Zbytynka river floodplain", landscape reserves of local importance "Pivdenno-Mostivsky" and "Pivnichno-Mostivsky", and a forest reserve of local importance "Olkhava", hydrological reserve of local significance "Zbytynsky", geological reserve of local significance "Kynva Spring", protected tracts "Gurby", "Mostivske", "Budky", "Zinkiv Stone", "Peklo", "Turova Grave".

The park's vegetation is dominated by forest vegetation. Other vegetation types include swamps, marshes, and peaty meadows, which were formed on the site of drained swamps. Swamps are found mainly in and adjacent to the territory of Male Polissia. They are located in the valleys of small rivers. The bswamps are peaty, almost exclusively eutrophic. Forest vegetation is represented by deciduous, mixed and coniferous forests.



Figure 3.1 – National natural parks within the KhNPP 30-kilometer supervised area

The Nature Park "Male Polissia" is located in the Izyaslav and Slavuta districts of Khmelnytskyi region.

The park's boundaries are conditionally drawn along the valleys of the rivers and the KhNPP cooling pond. In the north – the Goryn River and the cooling pond, the source of the Hnylyi Rih River; in the east – the Goryn River; in the northwest – the Vilia River; in the south – the tributaries of the Goryn and Vilia Rivers.

The park has well-preserved forest vegetation. On the massifs of thick sands, the largest areas are occupied by green moss and blueberry-green moss pine forests. Oak-pine and oak-hornbeam forests do not occupy significant areas. Their main areas are concentrated in the SE "Slavutske Forestry". Scots pine is found in natural stands in fragments. Black cedar forests occupy small areas in the foothills of the Goryn River floodplain and are represented by fern, pike and sedge cedars with pointed sedge.

The swampy floodplains are occupied by peat bogs and marshy meadows. The area is characterized by large tracts of marshes. All types of matshes of Male Polissia are represented here, and there are few lakes.

The KhNPp 10-kilometer area includes such protected areas as the Pralis reserve located between the KhNPP fire station and the KhNPP training center, the Dorohoshcha reserve near the KhNPP fish farm, the Vilshyna reserve on the outskirts of Netishyn, the May Lily of the Valley botanical reserve, the Klynovetske hydrological reserve and parts of the Male Polissya National Park.



Figure 3.2 – Nature protected areas within the KhNPP10-km area

**"Dorohoshcha"** is a botanical reserve of local importance located in the Netishyn forestry, SE Slavutske Forestry, near the KhNPP cooling pond.

The protected area is slightly hilly, covered with an almost continuous massif of medieval and mature forest. This area is dominated by a sparsely herbaceous hornbeampine forest with oak, birch, and alder in the lower elevations. There is an admixture of sharp-leaved maple and sycamore in the stand. The undergrowth is poorly detected, it is represented by hazel, warty larch.

The grass stand is sparse, dominated by yellow archangel, and European bentgrass. There are many rare and uncommon species in the grass stand. The fir clubmoss was found here. It grows in two large clumps in the natural slope of alder-hornbeam-pine forest of sparse herbs. The growth of species listed in the Red Book of Ukraine are identified in the reserve – forest lily and broad-leaved helleborine. In the lowlands, large patches of prickly pear form a boreal species from the Red Book of Ukraine, which is located at the southern border of its range. There are large populations of Melittis sarmatica and common wolfberry. These species are included in the list of protected species in Khmelnytskyi region.

**"Pralis"** is a forest reserve of local importance located in the Kryvynske forestry of the SE Slavutske Forestry.

The protected area is an old oak-hornbeam forest. The age of the preserved old oaks, which form the basis of the stand, is about 200 years, their height reaches 22-24 m, the average trunk diameter is 90-92 cm, and there are some oak trees with a trunk diameter of 110 cm. In addition to oak, the first tier contains old pines and some birches. Hornbeam forms a dense second tier 14-16 meters high. Numerous undergrowth is provided by hornbeam and sharp-leaved maple. Undergrowth is not developed. The herb layer is sparse. Yellow archangel and common quatrefoil dominate. Occasionally there is a forest starflower, May lily. On the edges of the reserve, the Wood cow-wheat, yellow foxglove, tall cinquefoil, and crownvetch. The growth of plant species listed in the Red Book of Ukraine was detected: forest lily, cinquefoil, dark red cinquefoil and rare species of noble liverwort, Austrian fern, May rose, medicinal and multi-leaved hawthorn.

The preserved area of the old hornbeam-oak forest is of great scientific value. This forest plantation is a model of virgin forests, which do not exist in this region. Such forests are listed in the Green Book of Ukraine.

"Lily of the Valley" is a botanical reserve of local importance, located in the Pluzhnyansky forestry, SE Izyaslavske Forestry, near the village of Pluzhne. The reserve was transferred to the jurisdiction of the Male Polissia National Park.

The territory of the reserve is characterized by a leveled relief, sod-podzolic sandy soils, medieval pine forest with an admixture of oak. The grass cover, dominated by lily of the valley, is uneven. In the reserve, lily of the valley blooms profusely, has a high vitality, and is distributed over the entire area, forming spots.

Along with the lily of the valley, there are mainly northern (boreal) species blueberries, club-leaved plantain, heather, lingonberries, and Koeleria glauca. All of these are mostly Central Polissia species, which are also common in the Male Polissia area. A rare species of this group is the narrow-leaved gerbil. Among the Red Book plants, a relict plant of the Tertiary period, wolf's berry, was found on the territory of the reserve. **"The Klynovetsky tract"** is a hydrological reserve of local importance located in the Pluzhnyansky forestry, SE Izyaslavske Forestry.

The ecotope, which is typical for Male Polissia, is protected and consists of a lowland forest tract with alder-birch and pine forests with an admixture of maple, ash and spruce, as well as pine plantations.

A special rarity of the reserve's dendroflora is a specimen of the common oak over 400 years old. Several left tributaries of the Vilia River (a tributary of the Horyn River) originate and flow in the reserve. The herbaceous cover includes typical forest plants (e.g., ground elder, hairy sedge, European bindweed, male and lady fern), meadow and wetland plants (water horsetail, water chickweed, and alternate-leaved golden-saxifrage), and wetland plants (swamp horsetail reeds, forest reed, double-leaved mint, water marshmallow). There are areas with different types of willow: brittle, purple, white, and ash willow. A plant of the Red Book of Ukraine, wolf berries, was found, as well as a rare sedge in Khmelnytskyi region.

The KhNPP supervised area includes 10 protected tracts. Only the protected tract "Vilshyna" was observed not far from KhNPP.

**"Vilshyna"** is a protected tract of local importance located in the Kryvynske forestry, SE Slavutske Forestry in the eastern part of Netishyn.

The protected area is a part of the foothills of the Goryn River floodplain, which is clearly distinguished in the relief, occupying a natural depression. The site is characterized by rather significant moisture, especially in the area adjacent to the peaty meadow. Black alder grows in the low-lying, waterlogged part of the tract. In the part of the tract adjacent to the road, elevations are observed, resulting in old pine trees among the alder plantations. The majority of the tract is forested, the smaller part is occupied by peat meadows.

The predominant fern alder forest is dominated by marsh thelipteris, and in the higher parts - by female shieldwort and Chartres shieldwort. Male shieldwort also occurs here. Typical alder companion plants grow in alder, such as false and elongated sedges, sweet-bitter nightshade, marsh wolfsbane, and marsh cleavers. In the cereal-fern alder, you can see thin broom, blueberry clumps. Along the edge of the tract, large spots (several tens of square meters) are formed by rough blackberries. This is a Carpathian species, which is located here at the eastern border of its range. There is also a species of blackberry called bearberry (Nesian blackberry), as well as a Central European species called small-flowered dog rose.

In the southern part of the tract, a peaty meadow has formed on a flat elevation on the site of a reduced alder tree. There are some specimens of overgrown alder on it. The meadows are pike-small sedge meadows with soddy pike, yellow, black, and prostrate sedges.

Large populations are formed by river thistle, willow weeping grass, and marsh puddleweed. Also, populations of two species of meadow-marsh orchids listed in the Red Book of Ukraine - meat-red and spotted finger-root orchids - were found. They bloom

and bear fruit. Less common species include the marsh trident, four-winged St. John's wort, and tall valerian, a valuable medicinal plant.

During the faunal field surveys, the primary focus was on the KhNPP impact zone and the areas immediately adjacent to this zone. Surveys were also conducted in more remote areas. The surveys and observations were conducted within the territories of the Nature Reserve Fund (NRF) of Ukraine (NNP "Male Polissia", forest reserve of local importance "Pralis", botanical reserve of local importance "Dorohoshcha", landscape reserve of local importance "Myslyatynskyi", park-monument of landscape art of local importance "Kryvynskyi", on the border of the protected tract of local importance "Vilshyna"). Surveys within the territory of the Izyaslavsko-Slavutskyi Emerald Network were also perfomed. In addition, observations were conducted within settlements, on agricultural areas, near water bodies, within exploited forest areas, etc.

*Birds.* Bird surveys were conducted on road routes within the 30-km zone of the KhNPP in combination with short-term spot and walking surveys. A total of 92 locations were surveyed.



Figure 3.3 – Locations of bird surveys

Based on the analysis of the field survey data, a total of 87 species of birds were recorded, of which 18 are included in the species list of Resolution 6 of the Bern Convention:

- Botaurus stellaris (Linnaeus, 1758) The great bittern
- Ixobrychus minutus (Linnaeus, 1766) The little bittern
- Casmerodius albus (Egretta alba) ((Linnaeus, 1758) Great egret
- Ciconia nigra (Linnaeus, 1758) Black stork
- Ciconia ciconia (Linnaeus, 1758) White stork
- Pernis apivorus (Linnaeus, 1758) European honey buzzard

- Milvus migrans (Boddaert, 1783) Black-crowned Falcon
- Circus aeruginosus (Linnaeus, 1758) Harrier
- Circaetus gallicus (Gmelin, 1788) Snake-eater
- Haliaeetus albicilla (Linnaeus, 1758) White-tailed eagle
- Sterna albifrons (Pallas, 1764) Little tern
- Sterna hirundo (Linnaeus, 1758) Common tern
- Caprimulgus europaeus (Linnaeus, 1758) Redshank
- Alcedo atthis (Linnaeus, 1758) Common kingfisher
- Dryocopus martius (Linnaeus, 1758) Black woodpecker
- Anthus campestris (Linnaeus, 1758) Tawny pipit
- Lanius collurio (Linnaeus, 1758) Red-backed shrike
- Ficedula albicollis (Temminck, 1815) Collared flycatcher

# Mammals

1. *Bats.* Currently, the list of species confirmed to be present within the study area includes 10 species:

- Barbastella barbastellus (Schreber, 1774) barbastelle bat
- Eptesicus serotinus (Schreber, 1774) serotine bat
- Myotis dasycneme (Boie, 1825) pond bat
- Myotis daubentonii (Kuhl, 1817) Daubenton's bat
- Myotis nattereri (Kuhl, 1817) Natterer's bat
- Nyctalus leisleri (Kuhl, 1817) Leisler's bat
- Nyctalus noctula (Schreber, 1774) common noctule
- Pipistrellus kuhlii (Kuhl, 1817) Kuhl's pipistrelle
- Pipistrellus nathusii (Keyserling & Blasius, 1839) Nathusius' pipistrelle
- Pipistrellus pygmaeus (Leach, 1825) soprano pipistrelle

Two species, B. barbastellus and M. dasycneme, are listed under Resolution 6 of the Bern Convention.



Figure 3.4 – Schematic map of survey routes and stationary observations of bats in the KhNPP supervised area

White lines and circles - bat surveys; red contour - 3-km NPP zone, yellow contour - boundaries of the 30-km NPP zone, green contour - boundaries of the Izyaslavsko-Slavutskyi Emerald Network site; yellow polygons - natural reserve fund of Khmelnytskyi region, blue polygons - natural reserve fund of Rivne region

2. *Predators*. Following the surveys, three locations of otters (Lutra lutra) were identified within the KhNPP planned activity territory and 30 km surrounding area, see Table 3.5

Location characteristics	Location
Otter feeding table	A dam between a cooling pond and a relatively small reedbed.
Traces of activity (prints in the sand)	The bank of Dziolova pond near Komarivka village
Otter feeding table	Dam on the Gnylyi Rih River

Table 3.5 – Otter (Lutra lutra) location surveys

3. *Rodents*. Field surveys of beaver (*Castor fiber*) conducted around the KhNPP site indicate that beaver occupation of wetlands is sporadic.

Most of the water bodies are former sand quarries, have a large depth and a large area of open water, and thus are of poor quality for beaver habitat. The Goryn River is also unsuitable for beaver habitat because of its fast flow. Therefore, only two beaver settlements were found in the quarry ponds.

A study of small forest lakes south of the Goryn River channel indicates that they were also inhabited by beavers in the past, as evidenced by very old wood gnawings and several old ruined huts. However, no inhabited locations have been found here.

The reason for the disappearance of the beaver lies in the succession of woody vegetation along the banks of water bodies. Currently, the predominant species here is a

pine tree of a ripening age, which creates a thick canopy of the upper tier, suppressing the growth of hardwoods.

A completely different picture has emerged in the southeastern part of the KhNPP cooling pond. Most of the cooling pond has a large depth and area devoid of near-water vegetation.

In general, the eastern part of the cooling pond coast with numerous canals, streams, and lakes has a very high habitat value for beaver. In particular, 4 beaver huts were recorded on one of the channels (up to 5m wide).

# **3.7.1Emerald Network**

The current scheme of the Emerald Network of Ukraine (https://rm.coe.int/updated-list-of-officially-adopted-emerald-sites-december-2019/168098ef51?fbclid=IwAR3Sfh-F_w0fpHBkCggkU1Xc1bUbo57vMgDhu1Fcgq-gFvM5QaceWsnOlt4) was approved in 2019.

The Emerald Network is focused on the conservation of specific species and habitats in Ukraine, including 264 species and 117 habitats, each of which has prescribed conditions for their conservation.

There are two approved Emerald Network protected territories in the KhNPP supervised area (Updated list..., 2019):

- Derman-Ostroh National Nature Park;

– Male Polissia National Nature Park became a part of the Iziaslavsko-Slavutytskyi Emerald Network, which includes the territories directly adjacent to the KhNPP, including the cooling pond.



Figure 3.5 – The Emerald Network protected territoriers within the KhNPP 30-km supervised area (from the map of the European Emerald Network https://emerald.eea.europa.eu/).

# **3.8** Characterization of the distribution of all negative factors in the area of influence of the planned activity

To analyze the indicators of all negative factors in the area of possible impact of the design facility, the data from the safety analysis reports for KhNPP units were considered and included.

Based on the analysis of external extreme impacts, it was determined that KhNPP buildings and structures may be exposed to the following negative factors:

- tornado;
- earthquake;
- fire (smoke);
- toxic and corrosive emissions into the atmosphere.

In the event of any of these incidents, KhNPP personnel must take adequate measures in accordance with the regulations and instructions of the Ministry of Emergencies of Ukraine.

## 3.8.1 Natural hazards

Natural meteorological phenomena [41] include phenomena that are particularly dangerous in terms of their intensity, area of distribution and duration:

- heavy rains (rainfall of at least 50 mm within 12 hours or less);

- large hail (diameter of at least 20 mm);
- wind with a speed of at least 25 m/s, hurricanes, squalls and tornadoes;
- heavy fog (visibility less than 100 meters);

- severe snowstorms (with a wind speed of at least 15 m/s), snowfall (precipitation of at least 20 mm in 12 hours or less);

- heavy ice (diameter of deposits not less than 20 mm)

# 3.8.1.1 Heavy rain and large hail

During KhNPP operation, heavy rains over the industrial site and events related to large hail did not cause any disruptions in plant operation.

# 3.8.1.2 Tornadoes

According to the zoning of tornado events [31], the KhNPP site is located in a tornado-prone area. According to the catalog of registered tornadoes in the USSR, [31] and the data of the Hydrometeorological Service of Ukraine [42] no tornadoes were registered directly within the KhNPP supervised area.

# 3.8.1.3 Heavy fogs

Heavy fogs are observed during the cold season. The fog classification by its origin is not of fundamental importance for NPPs. Whatever the type of fog, its presence does not contribute to the dispersion of impurities in the surface layer of atmospheric air. No heavy fogs were observed on the territory of Khmelnytskyy region, including in the vicinity of KhNPP.

# 3.8.1.4 Blizzards, snowfalls, ice

During KhNPP operation, severe snowstorms, heavy snowfalls, and ice that occurred in the areas adjacent to the plant did not create any emergencies at the plant.

# 3.8.1.5 Dust storms

The probability of intense natural dust storms in the northern and western regions of Ukraine (where KhNPP is located) is about 5%, i.e. they are possible once in 20 years.

During the plant operation, natural meteorological events that occurred in the areas adjacent to the plant, including severe dust storms did not create any emergencies at KhNPP.

# 3.8.1.6 Earthquake

The results of instrumental seismological surveys show that the KhNPP site is quiet from the seismic point of view.

The seismic score with regard to seismic microzoning for the KhNPP site is:

- design basis earthquake 5 points;
- maximum credible earthquake 6 points;

- seismicity of the site (PGA) according to the deterministic analysis of seismic hazard for KhNPP is 0.1g.

## 3.8.1.7 Floods

Based on the available information on the hydrological water bodies state in the KhNPP 30-km affected area, where the AP1000 units 5& 6 are to be deployed, it can be concluded that this territory is relatively well studied in hydrological terms. Most of the State Emergency Service of Ukraine hydrometeorological stations have more than 50 years of observations. The observation at these stations includes all the water regime elements. The Water Cadastre availability allows using the observation data of the State Emergency Service of Ukraine Hydrometeorological Center as a basis for the regional river water regime features and obtaining calculated hydrological parameters characterizing the water content of the NPP water supply sources.

A flood commission is appointed every year before the spring floods at NPP, and in some cases also before the summer and autumn floods. According to [43], the commission inspects and checks the flood (high water) preparation of all hydraulic structures, their mechanical equipment, lifting devices, and flood passage management and, after the flood has passed, re-inspects the structures.

A flood spillway is located at 14+00 Sta of the barrier dam to allow for reservoir blowdown with water discharge through the bottom outlet. The water is partially discharged during floods of exceptional recurrence, coinciding with periods of full reservoir filling. The flood spillway consists of two spillway shafts, three spillway tunnels with a cross section of  $3.0 \times 3.0$  m, a spillway in the lower reaches and a breakwater in the upper reaches. The flood spillway discharge at 0.01% is 110.0 m3/s.

In case of technological need, the reservoir is purged and water is discharged through the bottom outlet. Water is partially discharged through the flood spillway during floods of exceptional recurrence, coinciding with periods of full reservoir filling.

The maximum design discharge of a rainfall flood in the Hnylyi Rih River with a given probability of exceedance (under natural conditions) is 140 m³/s. The runoff volume is 32.8 million m3. The maximum water discharge from the Hnylyi Rih River with a given probability of exceedance and flood transformation is  $110 \text{ m}^3$ /s. The capacity of the spillway structures is shown in Table 3.6. The dependence of the water discharge Q through the flood spillway on the water level H in the cooling pond is shown in Figure 3.6.

Name of spillway structures	Capacity (m ³ /s) at the water level in the reservoir		
	NWL (203.0 m asl)	MFL (203.7 m asl)	
Flood spillway	does not operate	110.0	
Flood spillway bottom outlet	14.0	-	

Table 3.6 – Spillway	y structures	capacity
----------------------	--------------	----------



Figure 3.6 – Dependence of water flow Q through the flood spillway on the water level H in the cooling pond

Flooding of the areas adjacent to the downstream side of the dam should be monitored regularly. During the inspection of the area adjacent to the dam, flooding is visually detected. At the identified flooded areas, a detailed survey is performed: the spread area and groundwater depth are measured. Following the survey, the need for and methods of drainage of flooded land shall be determined in cooperation with the land user.

#### 3.8.2 Hazards of anthropogenic origin

KhNPP structures and systems that affect safety are designed to withstand extreme external impacts determined by natural site conditions (seismic impact, tornado, extreme ambient temperatures, etc.). Under these impacts, safety conditions are ensured and, accordingly, there is no additional environmental impact of the NPP.

#### 3.8.2.1 Emissions of harmful chemicals into the atmosphere

KhNPP is located on the territory of Khmelnytskyi region, on the border with Rivne region (at a distance of about 6 km), so it is advisable to indicate the current state of atmospheric air for the two regions in this section. The layout of the KhNPP site is shown in Figure 3.7.



Figure 3.7 – The layout schematic of the KhNPP site location

A number of industrial enterprises are located in the KhNPP supervised area that are sources of harmful chemicals emissions into the atmosphere. Pollutants are generated at industrial enterprises as a result of technological processes or fuel combustion.

Despite a significant amount of annual gross emissions of pollutants, they do not have an extremely large impact on the state of the air basin at the KhNPP sanitary protection and supervised areas, since the concentration of these substances in the air, including the KhNPP emissions, does not exceed the permissible limit concentration.

The current state of the atmospheric air is determined on the basis of continuous monitoring conducted by state authorities: Netishyn City Department of the State Institution "Khmelnytskyi Regional Laboratory Center of the Ministry of Health of Ukraine" and Ostroh City Department of the State Institution "Rivne Regional Laboratory Center of the Ministry of Health of Ukraine"; as well as directly by the relevant KhNPP control services. The data on background concentrations are provided in Appendix E.

## 3.8.2.2 Natural and anthropogenic processes

Unfavorable natural and anthropogenic processes include such phenomena as landslides, erosion, suffosion, deflation, karst, rainfall, salinity, and flooding.

Prior to the KhNPP construction, there were the following exogenous geological processes:

territory waterlogging;

– planar erosion (planar washout);

weathering of argillaceous and siltstone rocks in open pits (exfoliation – "peeling" and cracking of the soil massif in some areas).

During the KhNPP units 1&2 construction, these conditions were improved, namely:

- the site was drained, waterlogging was eliminated, marsh deposits were removed and replaced with sand;

– a redeposited chalk was removed and replaced with sand at the main structures area (main buildings of units No. 1-4), which created conditions for the infiltration of precipitation, i.e., waterlogging prevention;

- the territory was planned, asphalted and landscaped, which completely eliminated erosion;

- a set of protective measures was implemented to prevent weathering of siltstone and mudstone rocks in open pits (incomplete excavation to the design mark, installation of protective concrete coatings before the foundations were installed).

A set of measures was taken at the industrial site to reduce the negative impact of exogenous geological processes. As a result of these measures, no EGPs have been identified and no EGPs are expected to appear.

## 3.8.2.3 Accidental radioactive contamination of adjacent territories

At present, the radiological situation at the KhNPP site is mainly determined by radionuclides of natural origin. Short-lived technogenic isotopes were not detected in the NPP supervised area.
#### **DESCRIPTION OF THE ENVIRONMENTAL FACTORS LIKELY TO BE** 4 AFFECTED BY THE PLANNED ACTIVITY AND ITS ALTERNATIVES, **INCLUDING POPULATION HEALTH, FAUNA, FLORA, BIODIVERSITY,** (INCLUDING LAND ACQUISITION), SOIL WATER. LAND AIR. **FACTORS CLIMATIC** (INCLUDING **CLIMATE CHANGE** AND EMISSIONS), **GREENHOUSE** GAS **PROPERTY**, **INCLUDING** ARCHITECTURAL, ARCHAEOLOGICAL AND CULTURAL HERITAGE, LANDSCAPE, SOCIO-ECONOMIC CONDITIONS AND INTERACTIONS **BETWEEN THESE FACTORS**

The planned activity will have an impact on various aspects of the environment. This section is devoted to the description of environmental factors that may be affected as a result of the construction and operation of KhNPP Units 5 and 6. The impact on natural resources, socio-economic aspects, material objects, etc. is considered.

#### 4.1 **Population health**

The level of environmental risk of radiation exposure at Khmelnytskyy NPP during normal operation and possible emergencies outside the sanitary protection zone does not exceed the acceptable level required by the RSSU-97 [22]. Individual risks of the population stochastic effects due to radiation exposure from aerosol emissions of Khmelnytskyy NPP also do not exceed the acceptable level established by the RSSU-97.

The risks of deterministic effects are equal to zero. It means that even in case of an emergency at Khmelnytskyy NPP, when a significant loss of control over radiation emissions may occur, the expected radiation doses will not reach the level that may cause deterministic effects in the population.

It can be concluded that the level of environmental risk in the operation of the additional two Khmelnytskyy NPP units and their impact on human life and population health does not exceed acceptable levels and can be considered insignificant. The above factual data indicate that the safety measures taken at Khmelnytskyy NPP will be sufficient to prevent exceeding the permissible levels of radiation impact on the environment and population health.

#### 4.2 Condition of fauna

Fauna is one of the main environmental factors that can be affected by the implementation of planned activities.

Based on the conducted study, the commissioning of new KhNPP units will not directly damage animal habitats due to the absence of significant changes in the natural landscape. In addition, the commissioning of new units may positively impact migratory birds.

The construction of new KhNPP units does not predict significant negative environmental consequences, such as the destruction of animal populations, and ecosystems or complete elimination of habitats. However, certain changes will affect populations of several animal species. Expanding the non-freezing areas of the cooling pond may increase the number of some insect species, including dragonflies, certain species of beetles, and bedbugs. An increased number of blood-sucking dipterans is also possible due to an increasing number of their food sources and shelter.

Increased noise levels resulting from high traffic may have a restricted impact on insects. On the other hand, ground vibration, which also occurs as a result of high traffic, can be a significant disturbance for some insect species.

Groundwater level changes can have an impact on the entomofauna, primarily through changes in phytocoenoses or plant communities. Each plant species may have its consortium of insect species. When meadows are waterlogged, hydrophilic ones often replace mesophilic insect species. As a result, the meadow ecosystem can be replaced by a marsh one, which leads to a general impoverishment of the insect species range.

The commissioning of Units 5 and 6 and the accident-free operation of KhNPP, including six power units in total, will not have a significant negative impact on the overall species diversity of invertebrates and insects.

Increased anthropogenic impact on landscape areas related to the growing population of Netishyn may affect the number of rare insect species in the urban vicinity. It can be caused by various factors, such as the allocation of meadow areas for vegetable gardens, recreational activities, deforestation, and general environmental pollution. At the same time, it has been confirmed that the commissioning and normal operation of Khmelnytskyy NPP Units 5 and 6 will not hurt the fauna in the NPP observation area.

After the commissioning of new power units, additional impacts on the entomofauna may be possible due to various factors. The following are some of them:

1) Water temperature increasing in a cooling reservoir can affect the aquatic ecosystem as a whole, as well as the species diversity and quantity of aquatic insects. Changes in the temperature regime can affect the diversity and activity of insects living in water. However, it is unlikely that the species diversity of the entomofauna of the reservoir will be impoverished.

2) Increase the number of light sources (lamps that attract insects at night). The most commonly attracted insects are lepidopterans (sphinx moths, Noctuidae (owlet moths), geometer moths, lappet moths, pyralid moths, etc.), beetles (staphylinids - rove beetles, beetles, etc.), dipterans (flies, mosquitoes), hymenopterans, lepidopterans, and shield bugs. As a result, at night, insects may migrate from their habitats (from a distance of several hundred meters to two kilometers) to the NPP site. It can hurt the condition of populations of several insect species, but it will not lead to an impoverishment of the species diversity of Khmelnytskyy NPP's facilities. At the same time, this factor will favorably affect the number of nocturnal predators hunting near artificial light sources (some predatory beetles, arachnids, amphibians, etc).

3) A slight increase in the economic and recreational impact on the ecosystems of KhNPP is expected due to the growth of the number of service personnel and the total population of Netishyn. The area adjacent to the plant is under intensive use. There are recreation areas on the banks of the Goryn River. A large area is allocated for gardens,

vegetable gardens, and pastures. As a result, a valuable meadow entomocomplex has disappeared or been severely degraded over a large area, while at the same time, anthropogenic insect diversity has developed to a large extent. Plenty of synanthropic insects (large species of flies, cockroaches, and scale insects) significantly reduce the overall epidemiological situation in the plant area. To avoid infectious disease outbreaks, it is necessary to constantly monitor compliance with sanitary standards by local sanitary supervision authorities not only in residential areas but also in recreational areas.

### 4.3 Condition of flora

The main impact on the radiation situation in the NPP area is caused by radionuclides of natural origin. This indicates that the contamination levels caused by NPP activities remain at a safe level for nature and people.

The operation of two additional units at KhNPP, includong the above conditions, is not expected to have a significant impact on the structure and dynamics of plant species or change the number of populations of rare and the Red Book plant species.

Under consideration of emergencies, where exposure to flora outside the sanitary protection zone (SPZ) is insignificant, no changes in their structure or other negative changes are expected. However, under significant exposure that may occur in a beyond-design emergency, negative changes are likely to appear in some ecosystems, particularly in pine stands.

Under emergency conditions, the degree of radioactive impact releases on the environment, including forests, depends on various factors, including the release rate and radionuclide structure. It is worth noting that the main dose-forming radionuclide in the beyond-design basis accident is ¹³¹I, which can irradiate woody vegetation through aerial contamination. For plants growing inside the SPZ, doses can be high, which can lead to morphological changes.

In the case of a hypothetical emergency, negative changes mainly for pine plantations are possible, when plants absorb a dose from 6 to 10 Gy. Changes will be less noticeable or absent in hardwood plantations, which are less sensitive to radiation. Mixed stands consisting of pine and hardwoods will be described as intermediate, sometimes showing partial morphological or structural changes.

Assessment of the possible accumulation of radionuclides in valuable agricultural plant species within their ecological ranges is important for ensuring food and ecosystem safety. Many radionuclides released as a result of emergencies may not be significant for plant metabolism, but ¹³⁷Cs and ⁹⁰Sr are easily integrated into biochemical cycles and accumulate in food chains.

#### 4.4 Land resources

Before the construction of KhNPP Units 5 and 6, the project envisages the removal of topsoil formed on top of the fill soils and its storage in dumps near the construction site for further use.

The construction site will be filled with suitable local soils to the appropriate design elevations. The thickness of the fill layer is expected to be from 0.2 m to 1 m.

To reduce the environmental impact, the soil removed during construction will be stored at the KhNPP site.

All earthworks will be carried out with the use of moisturizers to prevent dust generation.

As the construction site has already been prepared for the construction of power units 1-4, no significant impact on the structure of vegetation and soil cover is expected during the construction of new power units.

Heavy metal contamination of soil occurs in three main ways: airborne, waterborne, and through the application of fertilizers and pesticides, as well as agricultural processing.

The most widespread and significant source of contamination is air, which is characterized by the intake of heavy metals in the form of aerosols and gases as a result of emissions from industrial enterprises, internal combustion engines, thermal power plants, CHP plants, and after households use natural fuel (coal) for space heating.

Since there are no large industrial facilities within thirty kilometers of Khmelnytskyy NPP, the main source of heavy metals from the air is roads and the burning of solid fuels for heating residential buildings.

According to the research results, the content of copper, zinc, cadmium, and lead in the soils of the territory adjacent to KhNPP is at the background level. There is a possibility of slight extra anthropogenic lead inputs into the soil of agricultural land adjacent to roads, which will not lead to an excess of average background concentrations of this element and contamination of agricultural products.

In general, pollutant emissions from most pollution sources enter the surface air layer, so they do not enter local and global air flows and are not transported over long distances. Therefore, chemical pollution is local and does not extend beyond the SPZ.

According to the calculations, the KhNPP complex (Units 5 and 6) covers 51.1 hectares. Thus, potential changes in the physical, chemical, and water-physical properties of soils may occur as a result of the impact of already constructed facilities, as well as as a result of increased discharges and emissions from the plant after the launch of power units 5 and 6.

An important impact on soils is man-made contamination of the soil cover with chemical pollutants, including radionuclides. However, being a complex, open, and dynamic system, soils have a certain resistance to anthropogenic load-buffering capacity.

Buffering capacity depends on the physical and chemical properties of soils, their mechanical composition, conditions of economic use, etc. The impact of power units on the soil cover of the region should be assessed taking into account the above-mentioned factors.

Thus, the following conclusions have to be mentioned:

- taking into account insignificant additional environmental contamination with radionuclides during normal operation of the plant (maximum values for Cs-137 - n 10-1 Bq/m2), special agrotechnical measures with changes in the structure of agricultural land

use, agricultural sectors re-profiling, and technological processing changes of products are not appropriate;

- according to the research results, the content of copper, zinc, cadmium, and lead in the soils of the territory adjacent to KhNPP is at the background level (also see parag. 5.3.7). There is a possibility of slight extra anthropogenic lead contamination of the soil of agricultural land adjacent to roads, which will not exceed the TLK (Threshold Limit Value) in agricultural products;

- the radiological situation in the area where the plant is located is currently mainly determined by radionuclides of natural origin, global radioactive contamination, and ChNPP contamination. No short-lived man-made isotopes are detected in the 30 km zone of the plant. The territory's 137Cs contamination is close to global levels (about 3 kBq/m²);

- the surface relief of the area close to the plant and the presence of orographic barriers are taken into account when considering transboundary radionuclide transfer in case of emergencies;

- the impact level will be regional at the maximum design basis emergency and beyond design basis emergency, but neither physical and chemical nor water-physical properties of soils will change; only the chemical elements content will change;

- overall, the analysis of the physical and chemical properties of the region's soils showed that, despite the significant diversity of soil cover, most soils have a significant buffer reserve of resistance to man-made impacts. The construction and operation of KhNPP with six power units will not lead to the exhaustion of this reserve. The landscapes in the vicinity of the plant are a reliable barrier to the expansion of the primary contamination zone through migration.

#### 4.5 Aquatic (water) environment

During the operation of KhNPP Unit 1-2, infiltration of process water caused changes in the groundwater regime within the industrial site. As a result, a rather steady increase in groundwater temperature and salinity is recorded in some areas. This phenomenon is local, does not extend beyond the industrial site, and does not affect water intakes for household water supply.

No additional sources of negative impacts on the hydrosphere are expected to arise in the construction of NPP Units 5-6. Temporary household and storage facilities at the construction site of power units 5 and 6 are provided with water supply and sewerage systems. The existing networks are the source of household water supply for temporary domestic and warehouse facilities. Wastewater is discharged into existing sewerage networks. The generation and discharge of wastewater into the environment is not expected.

NPP impact on surface waters may occur in places of direct contact of NPP process elements and facilities with surface water facilities of common use. These contact points are the water intake and discharge facilities of NPPs.

The operation of NPP Units 5-6 will not cause any significant adverse impact on surface and groundwater, and will not significantly affect the existing groundwater

regime. Thus, the implementation of the design solutions will not hurt surface and groundwater.

The commissioning of NPP Units 5-6 will not result in any significant negative impact on surface and groundwater and will not significantly affect the existing groundwater regime. Thus, the implementation of the design solutions will not hurt surface and groundwater.

The operation of KhNPP, including Units 1-6, in an accident-free mode will not affect the groundwater condition in the 30-kilometer zone. The water quality (chemical and bacteriological composition) of water intakes of the centralized household water supply in the 30-kilometer zone of KhNPP is not related to the NPP operation.

#### 4.6 Geological environment

The geological environment of the KhNPP supervised area is characterized by certain differences within a particular part of the territory. In particular, the geological environment of the land area and the industrial site of KhNPP differ in their structure from the geological environment of the supervised area.

The KhNPP impact on the geological environment within the KhNPP industrial site and land area was almost completely formed during the construction of facilities that are part of the power units 1-4 complex. Groundwater is the most vulnerable element of the geological environment. The commissioning of Units 5 and 6 will not significantly affect the existing groundwater regime; it is possible to get a local increase in water temperature, mineralization, or a local slight increase in the groundwater level in a limited area.

In the monitoring area, the impact of Khmelnytskyy NPP, including the facilities of power units 5 and 6, can only be affected in the event of an emergency, with hypothetical contamination of aquifers.

Before construction works started at the KhNPP site, the following exogenous geological processes were reported:

- slight flooding of the territory;
- weathering of the upper soil layers in the open pits.

The set of measures for the KhNPP Units 5&6 construction includes the following ones to improve site conditions:

- additional drainage;
- protection of soil open areas by arranging lawns of meadow grasses to prevent erosion.

The measures to prevent or limit the possible KhNPP impact on the geological environment of the KhNPP industrial site and land area ensure the stability of the geological situation, so the formation of new and further development of existing exogenous geological processes is not expected.

#### 4.7 Atmospheric air

Atmospheric air may be affected during the construction of new KhNPP units, and depending on the focus of the planned activities, radiation and non-radiation impacts on atmospheric air should be considered.

#### 4.7.1 Radiation impact

Radioactive emissions from the NPP ventilation system are one of the main sources of air contamination.

During the entire period of KhNPP operation, no impact of radioactive emissions and discharges on the radiation situation in the area of its location was detected.

The monitoring results of the KhNPP gas and aerosol emissions into the atmosphere from ventilation pipes of the units 1&2 reactors buildings and the special building in comparison with the established plant emission limits [79] for radionuclide groups are presented below in Table 4.1, according to the data of the Radiation Control Service for 2013-2021.

In accordance with the KhNPP radiation monitoring regulations [51], surface soil contamination is monitored on the territory adjacent to KhNPP.

The observation results show that during the KhNPP operation, the maximum contribution to the specific soil activity is due to the presence of the naturally occurring isotope K-40 and is determined by the natural characteristics of the region. The activity of the most prominent technogenic nuclide Cs-137 is ten times less than K-40.

Table 4.2 shows the measuring results of specific soil cover contamination directly related to the nuclear fuel cycle in 2013-2021, as well as the value of the "zero" background for Cs-137. For each year, the averaged values of sampling activity for measurements are provided, which are combined by the following radius of the KhNPP supervised area:

- 1st radius KhNPP industrial site;
- 2nd radius sanitary protection zone (10 km);
- 3rd radius supervised area (10-20 km);
- 4th radius supervised area (> 20 km).

The measuring results of Co-60 specific activity in environmental objects are not presented, since these values are below the MDA of the equipment. The MDA of Co-60 measurement is 0.099 Bq/kg.

As can be seen in Table 4.2, the activity of the technogenic isotope Cs-137 in all reporting years does not exceed the zero background level.

According to spectrometric measurements [80], the actual content of radionuclides in surface water as well as the activity index of the considered technogenic isotopes Cs-137, Sr-90, M-3 in relation to the permissible concentrations as per DGN 6.6.1-6.5.001-98 [22], are given in Table 4.3 from 2013 to 2021.

These data indicate a negligible impact of KhNPP on water activity and a large margin of activity up to the regulated values.

Voor	Inert	gases	LI	LN	Radio	iodine	Total
Tear	Arelease	% of RL	Arelease	% of RL	Arelease	% of RL	index, %
2013	4,47E+10	9,72E-02	3,35E+04	8,17E-04	4,10E+04	4,10E-03	0,10
2014	5,17E+10	1,12E-01	3,73E+04	9,10E-04	1,08E+05	1,08E-02	0,12
2015	5,14E+10	1,12E-01	2,98E+04	7,28E-04	6,99E+04	6,99E-03	0,12
2016	4,31E+10	9,38E-02	3,42E+04	8,34E-04	5,22E+04	5,22E-03	0,10
2017	3,40E+10	7,39E-02	3,94E+04	9,60E-04	3,01E+04	3,01E-03	0,08
2018	3,33E+10	7,24E-02	3,37E+04	8,21E-04	2,94E+04	2,94E-03	0,08
2019	3,74E+10	8,13E-02	3,53E+04	8,60E-04	3,39E+04	3,39E-03	0,09
2020	3,15E+10	6,85E-02	3,47E+04	8,45E-04	3,31E+04	3,31E-03	0,07
2021	3,39E+10	7,38E-02	3,32E+04	8,11E-04	4,51E+04	4,51E-03	0,08

Table 4.1 – Average daily gas and aerosol release of radioactive substances into the environment for 2013-2021, Bq/day

Radius	Nuclide	2013	2014	2015	2016	2017	2018	2019	2020	2021
	Cs-137	1,51E-01	9,12E-02	1,87E-01	1,52E-01	1,25E+00	1,82E-01	5,76E-02	1,08E-01	1,20E-01
SD7	Cs-134	2,41E-03*	4,44E-03*	3,13E-03*	2,80E-03*	8,99E-03*	3,92E-02	2,19E-03*	2,94E-03*	2,63E-03*
SPZ	Co-60	3,37E-03*	5,63E-03*	4,10E-03*	4,54E-03*	9,71E-03*	2,55E-03*	2,41E-03*	3,21E-03*	2,86E-03*
	Sr-90	1,41E-01	1,80E-01	2,47E-01	1,41E-01	1,51E-02*	2,27E-01	2,49E-01	1,69E-01	1,38E-01
	Cs-137	3,33E-01	2,31E-01	3,58E-01	2,24E-01	1,08E+00	1,61E-01	1,84E-01	2,42E-01	1,83E-01
SD7 10 lare	Cs-134	2,61E-03*	4,86E-03*	3,25E-03*	2,72E-03*	1,51E-01	2,75E-02	1,87E-03*	4,01E-03*	2,15E-03*
SPZ - 10 km	Co-60	3,38E-03*	6,23E-03*	3,97E-03*	2,91E-03*	9,17E-03*	1,83E-03*	1,94E-03*	4,86E-03*	2,30E-03*
	Sr-90	1,80E-01	9,60E-02	3,21E-01	1,23E-01	2,03E-02*	2,42E-01	2,58E-01	2,10E-01	2,00E-01
	Cs-137	3,59E-01	3,42E-01	2,78E-01	3,09E-01	2,52E+00	1,57E-01	1,71E-01	1,78E-01	1,99E-01
10 20 1	Cs-134	3,16E-03*	1,06E-02*	3,97E-03*	3,04E-03*	1,49E-02*	4,94E-02	2,10E-03*	3,86E-03*	2,83E-03*
10 - 20 km	Co-60	4,59E-03*	9,25E-03*	4,96E-03*	3,80E-03*	1,49E-02*	3,34E-03*	2,24E-03*	4,17E-03*	3,16E-03*
	Sr-90	4,12E-01	1,77E-01	2,95E-01	1,18E-01	3,01E-02*	2,96E-01	2,71E-01	2,35E-01	2,65E-01
	Cs-137	2,88E-01	2,09E-01	4,55E-01	3,17E-01	-	1,93E-01	1,77E-01	2,19E-01	2,73E-01
> 20 1	Cs-134	3,80E-03*	8,32E-03*	5,36E-03*	1,30E-03*	-	2,00E-03*	2,46E-03*	6,43E-03*	5,63E-03*
> 20 km	Co-60	5,71E-03*	1,22E-02*	7,33E-03*	1,55E-03*	-	2,00E-03*	2,78E-03*	7,18E-03*	7,17E-03*
	Sr-90	2,13E-01	1,86E-01	2,32E-01	1,62E-01	-	1,68E-01	2,57E-01	2,73E-01	1,94E-01

Table 4.2 –Soil contamination in the KhNPP supervised area for 2013-2021,  $kBq/m^2$ 

C	Della		Year											
Sampling place		onucide	2013	2014	2015	2016	2017	2018	2019	2020	2021			
Cooling pond	C 127	А	7,08E+00	6,70E+00	6,00E+00	9,69E+00	6,19E+00	8,34E+00	1,22E+01	1,35E+01	9,59E+00			
	CS-157	% of PC	7,08E-03	6,70E-03	6,00E-03	9,69E-03	6,19E-03	8,34E-03	1,22E-02	1,35E-02	9,59E-03			
	$C_{2}$ 124	А	4,78E-01	3,87E-01*	3,41E-01*	5,38E-01*	2,28E-01*	3,69E-01*	4,57E-01*	5,77E-01*	5,29E-01*			
	pling place Radio   ing pond Cs-137   Cs-134 Co-60   Sr-90 H-3   foryn river PP) Cs-137   Cs-134 Co-60   Sr-90 H-3   foryn river PP) Cs-137   Cs-134 Co-60   Sr-90 H-3   H-3 Co-60   Sr-90 H-3	% of PC	6,82E-04	5,53E-04	4,86E-04	7,68E-04	3,26E-04	5,26E-04	6,52E-04	8,24E-04	7,55E-04			
	C - (0	А	4,08E-01*	5,09E-01*	4,00E-01*	4,30E-01*	3,41E-01*	2,41E-01*	4,86E-01*	6,09E-01*	5,66E-01*			
	C0-60	% of PC	5,09E-04	6,36E-04	5,00E-04	5,37E-04	4,26E-04	3,01E-04	6,07E-04	7,61E-04	7,07E-04			
	S# 00	А	1,47E+01	1,59E+01	1,33E+01	1,36E+01	1,45E+01	3,61E+01	7,84E+00	7,56E+00	6,91E+00			
	Sr-90	% of PC	1,47E-01	1,59E-01	1,33E-01	1,36E-01	1,45E-01	3,61E-01	7,84E-02	7,56E-02	6,91E-02			
	11.2	А	clide201320147,08E+00 $6,70E+00$ of PC $7,08E+03$ $6,70E+01$ of PC $7,08E-03$ $6,70E+01$ of PC $6,82E-04$ $5,53E-01$ of PC $6,82E-04$ $5,53E-01$ of PC $5,09E-04$ $6,36E-01$ of PC $5,09E-04$ $6,36E-01$ of PC $1,47E+01$ $1,59E+01$ of PC $1,47E-01$ $1,59E-01$ of PC $1,63E-01$ $2,04E-01$ of PC $5,89E+00$ $3,67E+01$ of PC $5,89E+03$ $3,67E+01$ of PC $5,06E-01*$ $2,95E-011$ of PC $6,32E-04$ $3,09E-01$ of PC $1,04E+01$ $1,45E+01$ of PC $1,04E+01$ $1,45E+01$ of PC $1,04E-01$ $1,45E+01$ of PC $1,04E-01$ $1,45E-01$ of PC $1,04E-01$ $1,02E-01$	6,13E+04	6,59E+04	4,96E+04	3,05E+04	3,93E+04	5,24E+04	7,37E+04	6,86E+04			
	п-э	% of PC	1,63E-01	2,04E-01	2,20E-01	1,65E-01	1,02E-01	1,31E-01	1,75E-01	2,46E-01	2,29E-01			
the Goryn river	$C_{2}$ 127	А	5,89E+00	3,67E+00	4,78E+00	3,89E+00	1,19E+01	6,52E+00	1,01E+01	1,09E+01	1,11E+01			
(to NPP)	CS-157	% of PC	5,89E-03	3,67E-03	4,78E-03	3,89E-03	1,19E-02	6,52E-03	1,01E-02	1,09E-02	1,11E-02			
	$C_{2}$ 124	А	3,28E-01*	2,17E-01*	2,89E-01*	2,34E-01*	2,50E-01*	1,78E-01*	6,51E-01*	5,52E-01*	3,88E-01*			
	CS-154	% of PC	4,69E-04	3,09E-04	4,12E-04	3,34E-04	3,57E-04	2,54E-04	9,29E-04	7,88E-04	5,54E-04			
	C - (0	А	5,06E-01*	2,95E-01*	3,89E-01*	3,12E-01*	2,62E-01*	2,00E-01*	7,63E-01*	5,69E-01*	4,07E-01*			
	C0-00	% of PC	6,32E-04	3,68E-04	4,86E-04	3,89E-04	3,27E-04	2,50E-04	9,53E-04	7,11E-04	5,08E-04			
	S= 00	А	1,04E+01	1,45E+01	1,23E+01	1,16E+01	1,47E+01	1,04E+01	6,16E+00	5,43E+00	5,01E+00			
	Sr-90	% of PC	1,04E-01	1,45E-01	1,23E-01	1,16E-01	1,47E-01	1,04E-01	6,16E-02	5,43E-02	5,01E-02			
	11.2	А	2,71E+04	3,06E+04	2,74E+04	1,73E+04	3,57E+04	2,80E+04	8,27E+03	4,75E+03	4,25E+03			
	п-3	% of PC	9,03E-02	1,02E-01	9,12E-02	5,77E-02	1,19E-01	9,33E-02	2,76E-02	1,58E-02	1,42E-02			

Table 4.3 - Average annual specific activity of radionuclides in rivers and reservoirs in the KhNPP area,  $Bq/m^3$ 

the Goryn river (control station)	Cs 127	А	5,00E+00	5,56E+00	4,34E+00	7,67E+00	7,23E+00	8,49E+00	1,15E+01	1,20E+01	9,28E+00
(control station)	CS-157	% of PC	5,00E-03	5,56E-03	4,34E-03	7,67E-03	7,23E-03	8,49E-03	1,15E-02	1,20E-02	9,28E-03
	$C_{2}$ 124	А	2,61E-01*	2,39E-01*	2,95E-01*	3,89E-01*	1,89E-01*	2,06E-01*	5,82E-01*	4,26E-01*	5,38E-01*
	C8-154	% of PC	3,73E-04	3,41E-04	4,21E-04	5,55E-04	2,69E-04	2,94E-04	8,31E-04	6,08E-04	7,68E-04
	$C_{2}$ (0)	А	3,50E-01*	3,39E-01*	3,95E-01*	3,06E-01*	2,34E-01*	2,22E-01*	5,82E-01*	4,69E-01*	5,94E-01*
	C0-00	% of PC	4,38E-04	4,23E-04	4,93E-04	3,82E-04	2,92E-04	2,78E-04	7,27E-04	5,86E-04	7,43E-04
	S= 00	А	1,57E+01	9,15E+00	1,21E+01	1,25E+01	1,23E+01	1,34E+01	6,76E+00	2,56E+00	4,28E+00
	51-90	% of PC	1,57E-01	9,15E-02	1,21E-01	1,25E-01	1,23E-01	1,34E-01	6,76E-02	2,56E-02	4,28E-02
	11.2	А	3,90E+04	5,59E+04	4,34E+04	3,34E+04	2,18E+04	2,52E+04	1,17E+04	9,08E+03	8,30E+03
	п-3	% of PC	1,30E-01	1,86E-01	1,45E-01	1,11E-01	7,27E-02	8,38E-02	3,91E-02	3,03E-02	2,77E-02
«*» - values corr	responding	g to ½ minin	num measured	activity							

The absolute value of the dose rate measured by the monitoring points of the ARSMS during the entire period of operation does not depend on their location relative to KhNPP and is determined by the natural background and emissions of radionuclides of global origin. The dynamics of changes in the radiation background in the monitoring area are caused by the release of technogenic radionuclides as the Chornobyl accident results, global radiation emissions, and radiation background fluctuations.

The commissioning of KhNPP Units 5 and 6 will not lead to significant changes in the radiation situation at the NPP and adjacent monitoring areas. See also paragraphs 1.5.1 and 5.3.7

#### 4.7.2 Non-radiation impact

The start-up boiler house chimney emissions may have a significant impact on the air environment around Khmelnytskyy NPP.

Since the launch of KhNPP Units 5 and 6 does not include the introduction of new technologies accompanied by the release of new hazardous substances, it can be assumed that the qualitative characteristics of emissions will remain the same as before.

Ground-level concentrations of hazardous substances on the border and beyond the boundary of the SPZ with KhNPP sources comply with the limit values for settlements (See also Appendix C).

The commissioning of KhNPP Units 5 and 6 will not lead to an excess of the standard values of indicators of impact on the environmental situation.

### 4.8 Climate effects

The impact of the cooling pond on the area's microclimate is considered to be insignificant. The area of influence of the cooling pond is assessed to the distance of approximately 1000 meters. Quantitative changes in meteorological characteristics in the impact zone of KhNPP cooling systems are considered insignificant. The commissioning of power units 5-6 will not significantly affect the region's climate, see also parag. 4.2. Information on the climatic characteristics of the region is provided in Appendix D.

### 4.9 Architectural, historical, and cultural monuments

The placement of cultural heritage sites outside the site planned for the construction of KhNPP Units 5 and 6 indicates that these properties will not be directly damaged by the construction of new power units and related infrastructure. Thus, it can be confirmed that cultural heritage sites remain protected from the negative impact of construction.

New driveways, soil dumps, and other infrastructure that may appear in the construction zone of the power units have no impact on cultural property, as they are located outside their territory.

#### 4.10 Socio-economic factors

The construction of nuclear power units can have a significant impact on the socioeconomic population's life. Here are some aspects that may be particularly notable:

- increase in the employment rate;
- development of energy supply and affordable electricity;

- improving the living conditions and welfare of the population.

The commissioning of new NPP units is not directly related to the increase in disease and deterioration of population health in the surrounding areas.

The construction of units 5 and 6 will contribute to the provision of new employment opportunities in Netishyn and the KhNPP region, and facilitate drifting of the qualified personnel.

The involvement of university graduates from different Ukrainian regions with the appropriate education and qualifications is a reasonable and effective approach to the construction and commissioning of KhNPP Units 5 and 6. It makes it possible to take advantage of the potential of employees from different regions of the country and ensure a sufficient number of qualified specialists to perform various jobs at the plant.

The involvement of the required number of engineering and technical specialists and workers to perform construction and assembly works at KhNPPs 5 and 6 units of Khmelnytskyy NPP will be decided by construction and assembly organizations on their own.

The development of social infrastructure during the construction of Units 5 and 6 is an important aspect of the project, as it contributes to the improvement of living conditions for residents and employees involved in the project. The construction of social facilities, such as residential buildings, schools, kindergartens, medical facilities, cultural centers, and sports facilities, helps to create comfortable conditions for the life and development of the local community.

The substantial expansion of KhNPP indeed provides prospects for the development of new production facilities in the region. The expected population growth in Netishyn, which is associated with the expansion of the energy infrastructure, creates demand for various services and goods.

During the construction phase, 2,880 temporary workers are planned to be employed in the 1st shift at Units 5 and 6, who will live in Netishyn only during the units' construction.

The staffing for KhNPP Units 5 and 6 is 1003 people (see Table 4.4). Due to simplification of the design of units with AP 1000 because of the reduction in the number of equipment, the operation of such units requires significantly less technical personnel .

Table 4.4 – Comparative table of staffing levels

Position according to Westinghouse Electric	Position by Ukrainian	IAEA	EA PWR	PWR	AP1000	AP1000 Recommend	Adopted AP1000 o	for power uni f Khmelnytsky	its 5,6 yy NPP
Company	standards	1052	Spain	USA	USA	ation.	Power unit No.5	Power unit No.6	Total
	<b>BUSINESS OPERATIONS</b>	S / EXEC	CUTIVE	MANAC	GEMENT	Г ЕРІВНИЦТ	ГВО АЕС		
NPP Manager	CEO (General Director ) of NPP	1	1	1	1	1	1	-	1
Chief Engineer of NPP	Chief Engineer - First Deputy CEO	-	-	-	-	-	1	-	1
		QUAI	<b>JTY AS</b>	SURAN	CE				
QA Manager	The Head of the QA section	-	1	-	1	1	1	-	1
QA Supervisor	Deputy Head of the QA section	19	5	14	3	3	3	3	6
QA Auditor	Quality auditor	-	4	-	10	8	8	8	16
SUBTOTAL QA		19	10	14	14	12	12	11	23
		NU	CLEAR S	SAFETY	7				
Nuclear Safety Manager	Deputy Chief Engineer for Nuclear Safety	1	1	1	1	1	1	-	1
Nuclear Safety Specialist	Lead Nuclear Safety Engineer	5	-	10	-	1	1	1	2
Emergency preparedness manager	Head of the Emergency Management	2	2	1	1	1	1	1	2
Guard	Guard	77	-	135	7	75	75	75	150
Security Manager	Head of the Security Service	1	1	1	1	1	1	-	1
Emergency Preparedness Specialist	Lead Engineer for the Emergency Management and Response	5	1	4	2	2	2	2	4
SUBTOTAL NUCLEAR S	SAFETY	91	5	152	12	81	81	79	160
		E	NGINEE	RING					

Position according to	Position by Ukrainian	IAEA	PWR	PWR	R AP1000 USA	AP1000 Recommand	Adopted for power units 5,6 AP1000 of Khmelnytskyy NPP			
Company	standards	1052	Spain	USA	USA	ation.	Power unit No.5	Power unit No.6	Total	
Engineering Manager	Deputy Chief Engineer for Technology and Engineering	-	1	-	1	1	1	-	1	
Systems Engineering Manager	Head of the Systems Engineering Department	-	1	-	1	1	1	1	2	
System engineers	System engineers	26	14	42	23	25	25	25	50	
Program Engineering Manager	Head of the Program Engineering Department	-	1	-	1	1	1	1	2	
Program Engineers	Software engineer	24	3	16	10	10	10	10	20	
Reactor and Fuel Engineering Manager	Head of Fuel and Reactor Department	-	1	-	1	1	1	1	2	
Reactor Engineers	NPP equipment operation engineer	28	8	4	3	3	3	3	6	
Fuel Handling Supervisor	Deputy Head of Department for Fuel and Reactor Handling	-	1	-	-	1	1	1	2	
Fuel Handling Technician	Fuel handling engineer	-	3	-	-	3	3	3	6	
Engineering analysis manager	Head of the Engineering and Technical Analysis Department	-	1	-	1	1	1	1	2	
Engineer Analysis	Process and analysis engineer	21	4	10	3	2	2	2	4	
Design Engineering Manager	Head of the design and development department	-	1	-	1	1	1	1	2	
Design Engineers	Designer	15	58	28	15	15	10	10	20	
Design / Drafting	Designer	5	-	-	-	_	5	5	10	
Licensing Manager	Head of the licensing department	-	1	1	1	1	1	1	2	
Licensing Engineers	Licensing engineer	9	11	6	6	6	6	6	12	

Position according to	Position by Ukrainian	IAEA	PWR	PWR	AP1000	AP1000 A Recommend	Adopted AP1000 o	l for power uni f Khmelnytsky	its 5,6 yy NPP
Company	standards	1052	Spain	USA	USA	ation.	Power unit No.5	Power unit No.6	Total
Engineers	Head of the Information Technology Department	-	1	1	-	1	1	1	2
IT Technicians	IT Engineer	-	26	4	-	20	20	20	40
Document Control and Records Manager	Head of the Department of Documentation Management and	-	1	1	-	1	1	1	2
Desument control clorin	Accounting	0	4	6		1	1	1	2
Reords /archiving clerk	Records and archiving	-	4	6	-	1	1	1	2
SUBTOTAL ENGINEERI	NG	136	145	125	67	96	96	95	191
		(	<b>PERAT</b>	IONS	1				
Operations Manager	Deputy Chief Operating Engineer	2	1	-	3	1	1	-	1
Shift Manager (licensed)	NPP shift supervisor	-	8	-	5	7	7	7	14
Shift Supervisor (licensed)	Reactor shift supervisor	-	8	-	5	7	7	7	14
Shift Support Supervisor	Shift supervisor of a NPP power unit (stage)	-	-	71	5	7	7	7	14
RO (Reactor operator) / TO (turbine operator) (licensed)	Turbine and reactor control operator	78	16	-	10	14	14	14	28
Work Control (SRO)	Leading engineer of reactor control	-	-	-	5	7	7	7	14
Non-licensed Operator	Operator	-	40	-	30	35	35	35	70
SUBTOTAL OPERATORS	5	80	81	71	68	78	78	77	155
Operations Support Manager	Head of the Operations Support Department	-	1	-	-	1	1	1	2
Operational support Engineers	Operation support engineer	-	12	19	-	7	7	7	14

Position according to	Position by Ukrainian	IAEA	PWR	PWR	AP1000	AP1000	Adopted AP1000 or	for power uni f Khmelnytsky	its 5,6 /y NPP
Company	standards	1052	Spain	USA	USA	ation.	Power unit No.5	Power unit No.6	Total
Lock out-Tag out Engineers	Head of the Department of Repair Planning and Implementation	-	1	-	-	1	1	1	2
Lock out-Tag out technicians	Repair planning engineer		4	I	-	3	3	3	3
SUBTOTAL OD		-	18	19	-	12	12	12	24
Chemistry Manager	Head of the chemical shop	1	1	-	1	1	1	1	2
Chemistry Supervisor	Deputy Head of the chemical shop	5	2	21	5	2	2	2	4
Chemistry Analyst	Laboratory assistant of chemical analysis	6	8	-	12	9	9	9	18
Chemistry Programs Technicians	NPP chemical water treatment worker	8	3	3	-	2	2	2	4
SUBTOTAL CHEMISTRY	Ż.	20	14	24	18	14	14	14	28
Fire Protection Engineer	Head of the Fire Protection Service	-	2	2	1	1	1	-	1
Fire Protection Brigade	Lifeguard	-	35	-	-	35	35	-	35
SUBTOTAL FIRE PROTE	ECTION	-	37	-	-	36	36	-	36
SUBTOTAL OPERATION	1S	100	150	116	87	140	140	103	243
		Μ	AINTEN	ANCE					
Maintenance Manager	Deputy Chief Maintenance Engineer	-	1	-	1	1	1	-	1
Maintenance Superintendents (mechanical / electrical / I&C / Inspections and Test /Preparation)	Heads of Maintenance Departments	-	5	-	4	5	5	5	10
Maintenance Supervisors	Maintenance Specialist	-	22	-	8	10	10	10	20

Position according to	Position by Ukrainian	IAEA	EA PWR	PWR	R AP1000	AP1000 Recommond	Adopted for power units 5,6 AP1000 of Khmelnytskyy NPP		
Company	standards	1052	Spain	USA	USA	ation.	Power unit No.5	Power unit No.6	Total
Mechanical Technicians	Mechanical maintenance Engineer	-	56	-	40	20	20	20	40
Electrical Technicians	Electrical Engineer	-	44	-	20	20	20	20	40
I&C Technician	Engineer for maintaing the control measurement devices and automatic	-	47	-	30	25	25	25	50
Inspection and Test Technicians	Control Computer System Inspection and Test Engineer	-	7	-	-	9	9	9	18
Maintenance Support	Maintenance Locksmith	27	57	30	-	15	15	15	30
SUBTOTAL MAINTENA	NCE	27	239	30	103	105	105	104	209
	I	LEARNI	NG ORG	ANIZA	<b>FION</b>				
Learning Organization Manager	Deputy Chief Engineer for Personnel Training	-	-	-	1	1	1	-	1
Training Manager	The Head of the Personnel Training Center	2	1	-	-	1	1	1	2
Training Supervisors	Senior Personnel Training Instructor	-	5	-	2	3	3	3	6
Licensed Operator Instructors	licensed Operator Instructor	-	5	-	6	10	10	10	20
Non-licensed Operator Instructors	Operator training instructor	-	3	-	-	4	4	4	8
Maintenance Instructors	Maintenance Instructor	-	3	-	2	3	3	3	6
RP Instructors	RP Instructor	40	1	36	2	1	1	1	2
Chemistry Instructors	Chemistry Instructors	-	1	-	2	1	1	1	2
Engineering Instructors	Engineering Instructor	-	2	I	2	2	2	2	4
Simulator coordinator and IT	Simulator Instructor	-	3	-	-	3	3	3	6
Training Technician	Personnel training instructor	-	4	-	-	2	2	2	4

Position according to	Position by Ukrainian	IAEA	PWR	PWR	R AP1000	AP1000	Adopted AP1000 of	for power uni f Khmelnytsky	its 5,6 yy NPP
Company	standards	1052	Spain	USA	USA	Recommend ation.	Power unit No.5	Power unit No.6	Total
SUBTOTAL TRAINING		42	28	36	17	31	31	30	61
Performance Improvement Manager	Head of the HR Department	-	1	-	-	1	1	-	1
Analysis and Improvement Supervisor	Deputy Head of the HR Department	-	1	-	-	1	1	1	2
Analytics Improvement specialist	Human Resources Management Engineer	-	8	-	2	6	6	6	12
HP Supervisor	Head of the Occupational Safety Department	-	1	-	-	1	1	1	2
HP Specialist	Senior Occupational Safety and Health Engineer	-	3	-	-	2	2	2	4
SUBTOTAL PI/HP		$\begin{array}{c c c c c c c c c c c c c c c c c c c $							
SUBTOTAL LEARNING	ORGANIZATION	42	42	36	20	42	42	40	82
		RADIA	<b>FION PR</b>	OTECT	ION				
Radiation Protection (RP) Manager	Head of the Radiation Protection shop	-	1	-	1	1	1	1	2
RP Supervisor (Head of the laboratory, RP daytime/shift supervisors)	Shop Shift Supervisor (head of the laboratory, RP day/shift supervisors)	_	5	-	5	5	5	5	10
RP Technician	RP Engineer	37	-	30	18	7	7	7	14
ALAR Technician	Radiologist Engineer	7	21	5	-	3	3	3	6
Dosimetrist (daytime/shift personnel)	Dosimetrist (daytime/shift personnel)	-	24	-	-	4	4	4	8
Radwaste Technician	Radwaste processing engineer	14	5	16	-	1	1	1	2
SUBTOTAL RADIATION	PROTECTION	58	56	51	24	21	21	21	42
		WOR	K MAN	AGMEN	Т				

Position according to	Position by Ukrainian IA	IAEA	PWR	PWR	AP1000	0 0 Recommend	Adopted for power units 5,6 AP1000 of Khmelnytskyy NPP			
Company	standards	1052	Spain	USA	USA	ation.	Power unit No.5	Power unit No.6	Total	
Work control manager	Head of the Production Support Department	1	1	-	1	1	1	_	1	
Work control Supervisor	Deputy Head of the Production Support Department	-	1	11	2	1	1	1	2	
Week coordinator	Work Control Engineer	-	4	-	-	2	2	2	4	
Work preparation technician	Work Organization Engineer	-	11	-	-	5	5	5	10	
Planner Technician	Planner Engineer	2	4	-	-	2	2	2	4	
Outage Planner Manager	Head of the Outage Planner Department	-	1	-	-	1	1	1	2	
Outage Planner Technician	Maintenance preparation Engineer	2	4	7	-	2	2	2	4	
Planner Technician (design modifications)	Design Modifications Engineer	-	1	-	-	1	1	1	2	
SUBTOTAL WORK MAN	AGEMENT	5	27	18	3	15	15	14	29	
		SU	UPPLY C	CHAIN						
Supply Chain Manager	Head of Logistics Department	-	1	-	-	1	1	1	2	
Materials Purchasing Supervisor	Deputy Head of the Logistics Department	-	1	1	-	1	1	1	2	
Services Purchasing Supervisor	Senior Logistics Engineer	6	1	2	-	1	1	1	2	
Materials Purchasing Tech.	Equipment Purchase Engineer	-	1	1	-	1	1	1	2	
Services Purchasing Tech.	Logistics economist	2	2	2	-	1	1	1	2	
Warehouse Supervisor	Head of the Warehouse Management	-	1	4	-	1	1	1	2	
Warehouse Technician	Warehouseman	10	21	10	-	5	5	5	10	

Position according to Westinghouse Electric	Position by Ukrainian	IAEA	PWR	PWR	AP1000	AP1000 Recommend	Adopted for power units 5,6 AP1000 of Khmelnytskyy NPP			
Company	standards	1052	Spain	USA	USA	ation.	Power unit No.5	Power unit No.6	Total	
SUBTOTAL SUPPLY CHA	AIN	18	28	20	-	11	11	11	22	
TOTAL		497	703	563	331	524	525	478	1003	

The total number of operating personnel at Unit 5, including NPP management and deputy chief engineers, is 525 people, and at Unit 6 - 478 people.

The number of administrative personnel for each unit is assumed to be 15% of the total number of employees at the unit:

The total number of operating personnel at power units 5 and 6 is:

_	for power unit No. 5	$525 \cdot 0,15=79$ people,
_	for power unit No. 6	$478 \cdot 0.15 = 72$ people,

The total number of operating personnel at Units 5 and 6 is as follows:

_	for power unit No. 5	525+79=604 people,
---	----------------------	--------------------

- for power unit No. 5 478+72=550 people.

The total number of personnel at the facility is 604+550=1154 people.

#### 4.11 Technogenic environment

Subject to compliance with design and construction standards, the impact of the new facilities within Units 5 and 6 on the existing utilities, buildings and structures is acceptable.

#### 5 DESCRIPTION AND THE POSSIBLE ENVIRONMENTAL IMPACT ASSESSMENT OF THE PLANNED ACTIVITY, INCLUDING THE MAGNITUDE AND SCOPE OF SUCH IMPACT (AREA AND POPULATION THAT MAY BE AFFECTED), NATURE (IF ANY -TRANSBOUNDARY), INTENSITY AND COMPLEXITY, PROBABILITY, EXPECTED ONSET, DURATION, FREQUENCY AND INEVITABILITY OF THE IMPACT (INCLUDING DIRECT AND ANY INDIRECT, CUMULATIVE, TRANSBOUNDARY, SHORT-TERM, MEDIUM-TERM AND LONG-TERM, PERMANENT AND TEMPORARY, POSITIVE AND NEGATIVE IMPACT)

Investigating the potential environmental impact of a planned activity is critical to ensuring environmental sustainability. For this reason, it is important to analyze the magnitude and scope of the possible impact, including the area of the territory and the number of people who may be affected. It is necessary to take into account the nature of the impact, its intensity, and complexity, probability, expected onset, duration, frequency and inevitability. Various possibilities of impacts are considered, including direct and indirect, spillover, cumulative, transboundary, different durations and types of impacts, and positive and negative aspects.

# 5.1 Preparatory and construction works and the planned activity implementation, including (if necessary) dismantling works after completion of such activity

Description and assessment of the possible environmental impact of the planned activities include impact assessment during preparatory and construction activities, operation of KhNPP, and decommissioning and emergency situations.

#### 5.1.1 Preparatory and construction works

Description and assessment of potential environmental impacts of preparatory and construction activities, namely [5]: location, duration, content, main equipment and technologies, planned temporary structures, transport and utility networks, natural resource requirements and sources of these resources, management of construction and other solid waste, as well as liquid waste generated during their implementation, compliance with air quality standards and maximum permissible noise levels concerning the nearest residential area is presented in paragraphs 1.5.2, 5.3 of this report, respectively.

During units construction, there will be some negative impacts, namely:

- excavation of soil for the arrangement of technological equipment, structures, pipelines, other utility networks, and wells for household needs;

road pavement construction;

- preparation of construction materials (cement, concrete, etc.) and their transportation;

- welding and cutting of metal structures ("assembly works");

– painting works;

– operation of motor vehicles and other construction equipment with internal combustion engines, etc.

Construction and assembly machines and vehicles are the main sources of pollutant emissions due to the combustion of gasoline and diesel fuel in internal combustion engines, as well as the use of mineral oil for the operation of mobile compressors and transformers. To reduce the impact of emissions, all facilities must be subjected to a second full technical inspection following their operating manuals before being put into operation.

To reduce emissions of pollutants from internal combustion engines, the following measures are provided:

- timely inspection and maintenance of equipment;

- regulation of engines following the established standards to ensure complete combustion of fuel, which reduces its consumption and emissions of toxic substances;

- controlling the toxicity of exhaust gases before they are released to the line;
- reducing the time spent driving cars in variable modes;
- reducing the time of engine operation in low-speed and idling modes;
- maximizing the use of electrically driven lifting mechanisms.

To control pollutant release from motor vehicles, as well as to ensure measures to reduce them to the regulatory level, it is planned to organize a control and regulation point for vehicles equipped with gas sampling equipment and gas analyzers.

To reduce the noise level during construction and demolition works at the construction site and ensure the noise regime at the site follows the requirements of regulatory documents, it is necessary to:

- ensure that vehicle engines are silenced while on the construction site;
- turn off loudspeaker communication;
- do not carry out welding work without installing protective screens;

exclude the performance of work accompanied by noise over the permissible norm;

- exclude the operation of equipment that has a noise and vibration level exceeding the permissible standards;

– to soundproof the engines of construction machines, replace protective covers and hoods with multilayer coverage;

- to isolate local sources, use temporary noise screens, noise curtains, and tents (for example, place compressors in a sound-absorbing tent).

Fire safety of NPP units is ensured following the requirements of regulatory acts on fire safety in force in Ukraine:

- Civil Protection Code of Ukraine No. 5403-VI dated 02.10.2012;
- NAPB A.01.001-2014 "Fire Safety Rules in Ukraine";

– DBN B.1.1-7:2016 "Fire Safety of Construction Facilities. General requirements";

- NAPB B.01.014-2007 "Fire Safety Rules for Nuclear Power Plant Operation";
- NAPB B.01.012-2019 "Rules for Fire Protection";

- IB.0.3707.0003 "Facility-wide Instruction on Fire Safety Measures".

Fire safety measures include a set of forces and means, as well as legal, organizational, scientific, technical, economic, and social measures aimed at fighting fires. The measures are aimed at preventing fires, ensuring facility safety, and protecting personnel and equipment in the event of a fire.

Particular attention is focused on fire protection of security systems, as well as systems and elements important for safety. When designing them, we used noncombustible cable products, efficient heat dissipation systems, and means of physical separation and isolation of backup channels from each other to prevent common cause failures in case of fire.

At each operating NPP, in normal operation (as well as at all stages of occurrence and elimination of consequences of a possible nuclear or radiological emergency and in the post-accident period), continuous monitoring of the radiation situation in the premises, on the plant site and in the supervised area is carried out under the Regulations on Radiation Monitoring during Operation of NPP Facilities. In addition to laboratory monitoring methods, each NPP has an automated radiation monitoring system (ARMS) consisting of automated monitoring stations. Monitoring is carried out continuously in an automatic mode, which allows promptly receiving information from the monitoring stations, conducting systematic data analysis, and performing radiation forecasts for all settlements within the 30-kilometer zone around the NPP.

Each operating NPP has an Emergency Plan document that regulates the actions of the administration and personnel in the event of an emergency at the site. The NPP emergency plan is developed based on the requirements of the current legislation of Ukraine in the civil protection field, norms, rules, and standards on nuclear and radiation safety, adopted with due regard to the recommendations of international organizations in the field of nuclear energy use, in particular, the IAEA practical recommendations on the organization of emergency planning.

During the construction period, the operating organization will make additions to the NPP emergency plan action existing records that regulate the procedure for actions of the administration and personnel in the event of an emergency depending on its classification and will be approved by the regulatory authority.

#### 5.1.2 Implementation of planned activities

Operation is an activity aimed at achieving safely the purpose for which the NPP was constructed, including operation at capacity, start-ups, shutdowns, tests, maintenance, repairs, nuclear fuel transfers, in-service inspections, and other related activities. Scheduled activities are carried out in compliance with safety principles and criteria [24].

KhNPP safety is ensured through the consistent implementation of the deep-combat defense strategy based on the use of [24]:

 systems of physical barriers to the spread of ionizing radiation and radioactive substances into the environment; - systems of technical and organizational measures to protect physical barriers and maintain their effectiveness to protect personnel, the population, and the environment.

It is envisaged to control the radiation situation in the buildings, on the KhNPP territory, in the sanitary protection zone and supervised area and to carry out radiation monitoring [24].

Automatic fire detection and extinguishing systems, automated fire warning systems, and smoke protection of buildings (premises) that are not connected to the environment are envisaged [24].

As part of the planned activities, the Company plans to establish a repair and maintenance service. The equipment repair is aimed at solving the following tasks:

- ensuring the trouble-free functioning of systems and equipment within the project's design life and maintaining their design characteristics;

 restoration of the equipment and systems operability in case of their failure (damage) during operation;

– improving the performance of systems and equipment through their reconstruction and modernization.

Repair and maintenance of systems and equipment is carried out to maintain their technical condition and reliability throughout their service life within the efficiency and safety limits established by regulatory documents, and to ensure that the nuclear power plant meets safety requirements throughout its entire service life. Repair and maintenance is carried out in compliance with nuclear and radiation safety requirements and do not threaten the environment and the population.

The planned activities take into account the possible impact on the environment emissions and discharges of pollutants, noise, vibration, light, heat and radiation pollution, radiation and other impact factors, as well as waste management operations. Constructions, systems, and components are designed to reduce radiation and any other impact on the environment, the public, and personnel. During operation and temporary maintenance and repair activities, the generation of solid and liquid radioactive waste and other waste is considered under the current legislation. The environmental impact assessment during the planned activities is presented in paragraph 5.3.

#### 5.1.3 Works during the decommissioning

The general safety objective during NI decommissioning is to protect personnel, the population, future generations of people, and the environment from possible negative impacts of nuclear facility decommissioning activities.

The operating company, based on the regulatory legal acts requirements, determines in the decommissioning strategy specific safety objectives for NI decommissioning and criteria for the final state of the NI site.

Safety of NI decommissioning is ensured through consistent implementation of the defense-in-depth strategy based on the use of a system of physical barriers to prevent the spread of radioactive substances and ionizing radiation and to ensure the protection of

personnel, the population, and the environment from the ionizing radiation effects as well as a system of technical and organizational measures to protect physical barriers and maintain their effectiveness.

The main objective of implementing the defense-in-depth strategy during NI decommissioning is to timely identify and eliminate factors that lead to violations of safe NI decommissioning, emergencies, and prevent their development into accidents, limit and eliminate the consequences of accidents

#### 5.1.3.1 Basic options for power unit decommissioning

To select and justify a decommissioning strategy for a power unit, the following options for its decommissioning may be considered:

immediate dismantling ("Immediate dismantling" option);

- safe preservation of the power unit with further delayed dismantling ("Delayed Dismantling" option).

Both of these scenarios have identical initial and final states, and approximately the same focus of work and activities, but differ in cost characteristics.

*Immediate dismantling* is characterized by a rather short duration (on average, 10-15 years), a large amount of radioactive waste generated, and increased dose exposure of personnel involved in the work (2-3 times higher than for the delayed dismantling option), and a higher potential for impact on the public and the environment. The use of this scenario requires a developed infrastructure and a high level of technological training (development of unique robotic devices, special technologies, etc.) as well as the availability of sufficient cost in the decommissioning fund. The advantage of immediate dismantling is that the site can be vacated in a short time.

Delayed dismantling depends on the following decisions:

- the duration of the period of safe storage (from 30 to 100 years);
- the amount of equipment to be stored;
- the defined limits of protective barriers.

Currently, the option of delayed dismantling is preferable for the decommissioning of KhNPP units 5, 6 for the following reasons:

- there are no technologies for remote dismantling of radiation-contaminated equipment in Ukraine;

- the amount of high-level radioactive waste produced is much less than for the immediate dismantling option;

- there is no confidence that sufficient funds will be accumulated in the decommissioning fund for immediate dismantling at the time of power units' decommissioning;

there is no information on the future need to immediately vacate the power unit site.

Based on the experience of decommissioning PWRs around the world, the best option for a delayed dismantling strategy is to allow the most contaminated part of the reactor facility to remain in operation for 30 years. This period is determined primarily

by the nuclide cobalt-60, which is the main dose-forming nuclide with a half-life of 5.27 years.

#### 5.1.3.2 Nuclear safety in decommissioning

The main source of nuclear hazard at the decommissioning stage is spent nuclear fuel to be removed from power unit.

The design of the AP1000 unit includes a technological complex for nuclear fuel management that ensures the safety of storage, transportation, technological and operational procedures.

5.1.3.3 Ensuring industrial and fire safety and physical protection in decommissioning

During decommissioning, the general industrial, fire, and physical safety of power units 5 and 6 is ensured based on design solutions under applicable rules and regulations.

These types of security should be ensured under the requirements of the relevant regulatory documents in force at the time of implementation of specific stages and performance of a series of works.

Fire safety must also meet the requirements of the defense-in-depth strategy under the requirements of the NPP.

The Decommissioning Project should determine the main approaches to ensuring general industrial and fire safety.

#### 5.1.4 Conclusion

During preparatory and construction works and implementation of the planned activity, including dismantling works after the completed activity, the possible environmental impact is taken into account - emissions and discharges of contaminants, noise, vibration, light, heat and radiation pollution, radiation, and other impact factors, as well as waste management operations. Means of controlling and monitoring the impact of the planned activity, as well as measures to mitigate this impact, will be provided. The decommissioning strategy and approaches will be defined in the Decommissioning Project.

## 5.2 Use of natural resources, including land, soil, water and biodiversity during the planned activity

This section describes and assesses the potential environmental impact of the planned activity arising from the use of natural resources, including land, soil, water resources, and biodiversity. The assessment is important for ensuring sustainable development and maintaining ecological balance.

#### 5.2.1 Soils

The geosystems of the 30-km area belong to 15 geochemical classes, distinguished by typomorphic elements. The main classes are acidic calcium (more than 30% of the territory) and calcium acid-gleyed (more than 10%).

The main type-forming elements are As, Zr, Br, Y, Ca, Si, and sometimes Pb, Mn, and Se. Type-forming elements can be of both natural and technogenic origin.

Of the 15 types, the most abundant is the AsZrBr type. It should be noted that As and Zr are the most common microelements in the 30-kilometer area. They are type-forming elements both in geosystems of broad-leaved landscapes with light gray and gray forest sandy and sandy loamy soils with low humus content (forest terraces), neutral acidity and low content of exchangeable cations (Ca, Mg), and in geosystems with higher humus content and exchangeable cations.

The diversity of soil-forming rocks in these landscapes allows us to conclude that the presence of As, Zr, Br in amounts exceeding the Clarke number is not so much related to the chemical composition of the soil-forming rocks as to agricultural activities and the secondary supply of these elements to the soil after the decomposition of the primordial vegetation that accumulated them. Most geosystems of the H+ - Ca2+ class have long been used in agriculture, and the amount of humus in soils is gradually decreasing and mostly does not exceed 3%, with sporadic 5-7% of humus in leveled areas of elevated forest plains under natural vegetation.

#### 5.2.1.1 Radionuclide contamination of the Khmelnytskyy NPP supervised area

Radionuclides are of technogenic and natural origin. This contamination consists of a superposition of global fallout, fallout from the Chornobyl accident, and fallout caused by aerosol emissions from KhNPP Units 1&2. No other technogenic radionuclides were detected. This confirms the plant operation in normal mode. The level of the supervised area contamination does not create any restrictions on agriculture. Thus, the radiological situation in the vicinity of KhNPP is currently within the regulatory limits. The construction and operation of the new power units will not significantly affect the environment, see also parag. 5.3.7.

### 5.2.1.2 Sources of contamination and assessment of the impact of Units 5 and 6 on soils during normal operation

As a result of the mechanical impact associated with any construction, the soil cover is almost completely destroyed in the allocated area. However, the area of mechanical impact during KhNPP construction is clearly delineated by the agreed allocated boundaries.

Given the fact that radionuclides enter the soil in very small amounts, they cannot affect soil fertility in any way. Their content in soils is much lower than nutrients and they do not pose a threat during plant growing. The radionuclides in soils also do not affect the products quality and yield, see also parag. 5.3.7.2.

From the agrotechnical point of view, correct land use helps to reduce the intensity of radionuclide migration through agrobiocenoses, increase soil fertility and reduce erosion processes. 5.2.1.3 Assessment of possible changes in physical, chemical and water-physical properties of soils including land use characteristics

Following the analysis of the NPP operation impact on soil conditions, radioactive substances have normally no effect on the physicochemical and water-physical properties.

#### 5.2.1.4 Assessment of soil degradation

The main types of soil degradation in the study area are associated with intensive agricultural cultivation, drainage reclamation, and further development of reclaimed land. Their intensive cultivation and use in field crop rotations lead to unjustified losses of soil organic matter.

The degree of degradation is influenced by an increase in ash content (by 20-40%) of the volume weight, the decrease in filtration coefficients and in total moisture content, in some cases by almost 2 times.

Changes in agrochemical parameters during soil degradation are shown by a certain acidification of the soil solution reaction, a decrease in the gross nitrogen content of intensively cultivated areas. There are profound changes in the qualitative composition of organic matter. Humic acids accumulate and fulvic acids decrease. The humus type develops from fulvate-humate to humate-fulvate and humate. According to the modern assessment scale, the degree of soil degradation is defined as weak, medium, high and crisis.

Mineral soils are not subject to degradation processes as intensively as organogenic soils. The research identified that mineral soils are subject to mechanical, biochemical and chemical degradation. If the soils are located within operating drainage systems, the degradation processes usually increase.

Mechanical degradation is shown in two forms: erosion and physical. Water and wind erosion processes are particularly common on soils of light texture. Water erosion increases with the steepness of slopes and plowed areas, which affect the soil washout. Wind erosion degradation in the study area is weak, and water degradation is weak and medium.

Physical soil degradation is caused by soil compaction and decrease in agronomically valuable meso-aggregates. Drained soils, which include sod-podzolic, sod and meadow soils, are characterized by a weak, medium and high degree of degradation.

According to the parameters of soil degradation in terms of total porosity, mineral soils in the KhNPP impact area are characterized by weak, medium, high, and some areas of reclaimed sod-podzolic soils are characterized by a crisis degree of arable horizons degradation. Drainage and intensive agricultural use of soils have led to a deterioration in their structural and aggregate state.

Most of the drained loamy granulometric soils are characterized by a high and crisis level of degradation.

Soil degradation processes associated with the KhNPP construction extend only to the industrial site area. Their presence in the KhNPP 30-kilometer area is practically unrelated to the plant's operation.

### 5.2.1.5 Influence of landscape and geochemical barriers on the pollutant redistribution

Various barriers may be formed based on the complexity of geomorphological, orographic, landscape and geochemical structures, heterogeneity of ravine, gully and river network density, mosaic soil cover, diversity of physical, chemical and water-physical soil properties in geosystems of the KhNPP 30-kilometer impact area. These are phytobarriers, orographic and landscape-geochemical barriers, on which deposition, accumulation and long-term radioactive elements fixation will occur.

The role of these barriers is ambiguous, both in the structure of biogeocenoses and in the time aspect. The soil is a significant biogeochemical barrier in automorphic landscapes, while biota and, above all, microbiota in accumulative landscapes and coniferous communities. The soil barrier functions in relation to radionuclides are the most significant among all components of biogeocenoses. At the same time, the release of radionuclides with gravitational moisture outside the soil profile is no more than one hundredth of a percent per year. Thus, the soil accumulates the main amount of activity: from 80 to 95% (including microbiota) in forest ecosystems and up to 100% in agroecosystems (depending on the period of plant vegetation). However, the barrier functions of soils in relation to different radionuclides are shown in different ways. 137Cs is adsorbed most intensively, 90Sr, 106Ru, Pu are adsorbed much less intensively. It should be emphasized that high sorption capacity of soils for 137Cs is typical for all types of soils and phytocoenoses, including sod-podzolic sandy soils. The exception is peat soils.

The barrier functions of ecosystems and their individual components have a certain temporal dynamics. It is known that a phytocoenotic barrier is clearly visible for a number of macro- and microelements during the growing season. The barrier functions are further transferred to the soil again. For radionuclides, this is shown in a different way. Nevertheless, these aspects have not yet been fully studied and require clarification and confirmation.

A horizontal migration of radionuclides with surface and intra-soil runoff is practically not expressed in forest landscapes. However, a process of radionuclide redistribution between landscapes is evidenced by a constant decrease in contamination density of eluvial landscapes and a corresponding increase in contamination density of accumulative landscapes. From the ecological point of view, the absence of significant inter-landscape radionuclide redistribution indicates localization of releases within the area of primary contamination. This makes it possible to state that the expansion of the primary contamination area and the formation of secondary contamination areas in forest landscapes do not occur.

The border and the forest canopy will serve as short-term barriers to radioactive pollutants. The KhNPP 30-kilometer impact area is well forested (average forest cover is 28%), while the forest cover of the nearer, 10-kilometer area is about 50%. Pine forests

with a share of hornbeam and oak predominate, with hornbeam and oak forests in the northern part. Thus, forests in the supervised area are one of the main barriers that prevent secondary radionuclide redistribution.

As mentioned above, soils are an excellent barrier to further spread (redistribution) of the main radiologically significant radionuclides. There are also quite mobile radionuclides (tritium, chlorine-36), but their role in the formation of dose loads on the population is insignificant. A secondary contamination will occur mainly due to mechanical migration with erosion products and flushing (radionuclides adsorbed on soil particles move with them). The experience of Chornobyl proves that secondary radionuclide redistribution between landscape elements does not have a significant impact on changes in the radiological situation and amounts to less than 1% of the total initial contamination of the landscape.

Given the low mass (substance) radionuclide content even in the NPP accidental releases, the natural mechanisms of soils are sufficient for its firm fixation.

5.2.1.6 Assessment of orographic factors for the primary and secondary pollution fields and orographic barriers

A primary contamination field, under KhNPP normal operation, will occur depending on climatic conditions, primarily a long-term wind regime (recurrence of winds of certain direction and speed), including the roughness of the earth's surface (orographic conditions and height of vegetation cover).

According to the nearest meteorological stations (Shepetivka and Rivne), the area is dominated by winds from the west (20.4% and 24.7%, respectively), southeast (14%) and south (15.7% and 11.4%). Thus, radioactive releases will be mainly dispersed in the east, northwest and north directions.

To the east and north of KhNPP, there is the valley of the Goryn River with elevations ranging from 212 to 230 m and a forested denudation-accumulative flat plain with elevations ranging from 233 to 240 m. In the northwestern direction from the station, there is a denudation-accumulative wavy plain with elevations of 266 m.

The elevation at the KhNPP site reaches 204.6 m, and vent stack is 100 m. Even if we assume that the emissions will be included exclusively into horizontal air mass flows, there are no surfaces on their way that exceed 300-meter hypsometric marks. The height of trees is on average 20 to 25 meters. Therefore, the area does not contain landscape and morphological obstacles to the pollution dispersion in the direction of the prevailing winds.

Orographic and phytobarriers to the airborne pollutant transfer can be formed exclusively in the southern, south-southwest and west-north directions at a distance of 12.5 to 17 km from the station. It is in the southern and south-southwest directions from the plant that the forested elevated structural and denudation plain of the Podolian Upland with elevations up to 320 m is located, and in the west-west-north direction – the forested elevated structural and denudation plain of the elevations up to 315 m.

There will be either orographic barriers or phytobarriers under hypothetical design basis accidental releases with release heights of 0 m and 48 m (206.6 m above sea level – the height of the earth's surface in the industrial site area), if they are absorbed into horizontal air flows, in whatever direction they are dispersed. The height of the accidental release source is determined in terms of conservative estimates for the near area -0 m and applies to both beyond design basis and design basis accidents. This height is not conservative for a transboundary transfer.

Thus, the surface relief of the plant's near area and the barriers should be included when modeling the dispersion of gas and aerosol emissions and the dispersion of emissions in hypothetical emergency situations.

### 5.2.1.7 Characteristics of long-term behavior of radiologically significant radionuclides in soils

This section describes the behavior of the main radiologically significant radionuclides (⁹⁰Sr, ¹³⁷Cs, and actinides) and their long-term migration in soil in case of one-time (accidental) contamination. Chronic normalized NPP radionuclide releases under normal operation do not lead to significant contamination of the environment and its components; the role of short-lived release radionuclides and radioactive inert gases in long-term migration through biological chains and, accordingly, formation of additional exposure of the population when entering the body with food is negligible.

Geochemical radionuclide migration has very low rates even under favorable conditions, and therefore landscape self-cleaning, based on it only, is impossible. Landscape and geochemical barriers are volumetric natural bodies within which the conditions of chemical elements migration change. Mechanical barriers are, first of all, relief features that prevent or facilitate radionuclide redistribution between landscapes and their elements. Biocenoses or their separate layers form biogeochemical barriers. In this case, the stand, shrub layer, grass and moss-lichen covers, forest floor, sod or steppe mat, as well as underground biomass and soil itself perform the role of the barrier. The soil is the main barrier to further radionuclide redistribution between biogeocenose elements.

Soil is a crucial link for incorporating the radionuclides into biogeochemical migration chains in terrestrial ecosystems. The dynamics of radionuclide transfer in the soil-plant link is generally determined by the dynamics of three groups of processes, namely:

- radionuclide mobilization-immobilization in the soil root layer;
- radionuclide removal from the root layer;
- radionuclide physical decay.

The intensity of the first two processes depends primarily on the initial physical and chemical forms of fallout and the dynamics of their transformation in the soil and the soil cover characteristics.

To estimate the removal of radiologically significant radionuclides from the soil root horizons of meadows and arable lands, the values of ecological ( $T_{ecol}$  – without the rate of radionuclide physical decay) and effective ( $T_{ef} = T_{ecol} \cdot T_{1/2}/(T_{ecol} + T_{1/2})$  – with the rate

of radionuclide physical decay) semi-cleaning periods of these horizons are calculated in Table 5.1.

	⁹⁰ Sr				¹³⁷ Cs			
Soil	Meadow (0-5 cm)		Arable land (20 cm)		Meadow (0-5 cm)		Arable land (20 cm)	
	Tecol	Tef	Tecol	Tef	Tecol	Tef	Tecol	Tef
Automorphic mineral light mech composition	11-19	6-8	21-32	12-15	60-150	20-25	26-45	14-18
Automorphic mineral heavy mech composition	15-32	10-15	37-129	16-23	150-400	25-27	60-150	20-25
Hydromorphic organic soil	100-160	22-24	103-233	22-25	11-20	8-12	30-60	15-20
Organogenic dehydrated	-	-	-	-	17-83	11-22	-	-
Hydromorphic minerals	-	_	-	-	45	18	-	_
Low humic sand	2.2-6	2-5	2.2-7.6	2-6	-	-	-	_

Table 5.1 –Semi-cleaning periods of root soil horizons of meadows and arable lands from  $^{90}\text{Sr}$  and  $^{137}\text{Cs}$ 

The intensity of ¹³⁷Cs redistribution in the soil profile is low. In the period from 10 to 15 years after passing the ecological semi-cleaning periods from ¹³⁷Cs of 5-cm natural meadows soil horizons formed on the light mechanical automorphic mineral soils are from 60 to 150 years, on the heavy mechanical automorphic mineral soils from 150 to 400 years, on the hydromorphic organogenic soils from 11 to 20 years, on the hydromorphic mineral soils about 45 years.

The vertical ⁹⁰Sr transfer intensity in mineral soils significantly exceeds cesium radioisotopes transfer intensity. The ecological semi-cleaning periods from ⁹⁰Sr of 5-cm natural meadows soil horizons formed on the light mechanical automorphic mineral soils are from 11 to 19 years, on the heavy mechanical automorphic mineral soils from 15 to 32 years, on the hydromorphic organogenic soils from 100 to 160 years.

The problem should be noted on intensive vertical ⁹⁰Sr transfer in the profile of soil differences typical for Ukrainian Polissia – weakly sodded, slightly humic sands, where the radionuclide migrates at a rate of 2 to 6 cm year-1.

Depending on the physical and chemical properties and moisture content, the values of the ecological semi-cleaning periods of the upper 5-cm horizons of soils, meadows and fallow lands from  241 Am vary from 25±10 to 650±180 years, and from 100±45 to 260±120 years for Pu isotopes.

The values of ecological semi-cleaning periods from ¹³⁷Cs in the agrocenoses soil arable horizons are as follows: for light mechanical automorphic mineral soils – from 26 to 45 years, for heavy mechanical automorphic mineral soils – from 60 to 150 years, for hydromorphic organogenic soils – from 30 to 60 years. The values of ecological semi-cleaning periods from ⁹⁰Sr in the agrocenoses soil arable horizons: for light mechanical

automorphic mineral soils – from 21 to 32 years, for heavy mechanical automorphic mineral soils – from 37 to 130 years, for hydromorphic organogenic soils – from 100 to 230 years.

The values of effective semi-cleaning periods of the upper natural meadow soil horizons and arable soil horizons, which reflect the radionuclide redistribution in the soil profile and the rate of physical radionuclide decay are characterized by lower values – from units to tens of years.

The maximum values of effective semi-cleaning periods from  137 Cs are characterized by the upper natural meadow soil horizons (25 to 27 years) and agricultural land arable soil horizons (20 to 25 years) formed on the heavy mechanical automorphic mineral soils. Hydromorphic organogenic soils (8 to 12 years old) are characterized by the lowest T_{ef} values. In general, according to the  137 Cs migration mobility index, the supervised area soils can be classified as follows: peat-bog > sod-podzolic > gray forest > black soil.

Areas with critical (in terms of ¹³⁷Cs biological mobility) hydromorphic organogenic soils are located in small areas throughout the 30-km area, but they occupy the largest areas in the eastern (semi-hydromorphic), southeastern (semi-hydromorphic and hydromorphic), and southwestern (hydromorphic) sectors.

The minimum values of effective semi-cleaning periods from ⁹⁰Sr were established for light mechanical mineral soils (6 to 8 years and 12 to 15 years, respectively, for the upper natural meadow soil horizons and agricultural land arable soil horizons). An abnormally high intensity of ⁹⁰Sr transfer was noted above in weakly sodded, slightly humic sands, the effective semi-cleaning period from this radionuclide being from 2 to 6 years.

Thus, sandy and sandy loamy sod-podzolic soils are critical in terms of ⁹⁰Sr mobility in them. Such soils occupy more than 20% of the supervised area territory.

Additional levels of soil contamination under plant normal operation are very low in mass terms (concentration units) and therefore the impact of contamination on soil will be negligible. Therefore, special agrotechnical measures aimed at reducing the impact on soil are not appropriate under plant normal operation, see also clause. 5.3.7.2.

#### 5.2.2 Water resources

Water is one of the main resources used under planned activity, and it is important to avoid negative impacts on both surface and groundwater.

#### 5.2.2.1 Surface water

NPP impact on surface waters can be observed in places of direct contact of NPP process elements and structures with surface water bodies of public use. Such contact points are KhNPP water intake facilities.

The service water supply plan for units 5 and 6 was designed as a reverse scheme with cooling towers as radiators. The KhNPP cooling pond in the adopted cooling and process water supply plan for units 5 and 6 is a source of water intake for cooling tower feeding and blowdown and for other process systems needs, and is also used to discharge

blowdown water. The cooling pond water losses are replenished by 100% accumulation of the Hnylyi Rih river flow and makeup by a pumping station from the Goryn River based on a special water use permit.

The additional water pumping station functions to feed the cooling pond to replenish the irrecoverable losses in the service water consumption system and is designed to take in water from the Goryn River at a flow rate of up to 30m ³/s. To meet the total service water demand of Units 5 and 6, it is necessary to withdraw additional water throughout the year, cover the demand in the summer and autumn period by operating the cooling pond, replenish it with runoff from the Goryn River from December to April and with runoff from the Hnylyi Rih River throughout the year, and set a new limit for additional water withdrawal.

The cooling pond chemistry and its quality are determined by such components as the Hnylyi Rih River flow, partial water intake from the Horyn River, and the hydrochemical features of the makeup water (also see Table 5.2). According to O.O. Alekin classification, the Hnylyi Rih River and the Goryn River water belongs to the hydrocarbonate class, calcium group, the second type, according to the predominant quantitative composition of calcium ions and hydrocarbonate ions. Due to changes in the concentrations of total water hardness and bicarbonate ions, as well as their ratio, the cooling pond gradually changes and moves from the calcium to the sodium group.

№	Chemical indicator	Measurement units	Circulation water (cooling pond)	Make-up water of the cooling pond (Gnylyi Rih River)	Makeup water of the cooling pond (Goryn River)
1	Hydrogen index	unit	8,275	7,83	7,91
2	General hardness	mg-eq/dm ³	5,25	6,11	6,20
3	Calcium	mg/dm ³	70,14	97,65	97,92
4	Sodium	mg/dm ³	79,9	7,41	15,27
5	Transparency	cm	49,5	35,91	30,64
6	Alkalinity	mg-eq/dm ³	5,27	5,46	5,48
7	Permanganate oxidizability	mgO/dm ³	8,32	7,67	4,70
8	Chlorides	mg/dm ³	50,76	13,57	24,99
9	Sulfates	mg/dm ³	120,15	44,37	50,04

Table 5.2 - Quality of the KhNPP cooling pond source water and the Gnylyi Rih and Goryn rivers feedwater

Commissioning of the designed AP1000 units 5&6 will not increase the heat load on the cooling pond due to the use of cooling towers as a cooling system and the absence of heat exchange water discharge.

The commissioning of units 3&4 will increase the volume of heated water entering the pond, which will facilitate increasing the temperature, water evaporating and creating conditions for increasing the total mineralization content.
The hydrochemical forecast of the cooling pond water quality during the power units commissioning was made by PLC Lviv ORGRES. The forecast was made using a mathematical modeling program for an average water year (50% availability), for the first and second design years and the year of salt equilibrium. The results of the hydrochemical forecast are presented in Table 5.3.

Season	General H _{gen,} mg-eq/dm ³	Carbonate H _{carb,} mg-eq/dm ³	Noncarb onate H _{noncarb,} mg-eq/dm ³	CalciumC a ²⁺ mg/dm ³	Magnesi um Mg, _{mg/dm} ³	Natrium Na, _{mg/dm} ³	Chloride Cl, mg/dm ³	Sulphate SO4 ⁻² , mg/dm ³	Hydrocar bonateH CO ₃₋ mg/dm ³	lons sum _{mg/dm} ³
				First ca	lculated y	/ear				
Spring	5.07	2.53	2.54	65.30	21.98	14.55	25.70	54.78	154.47	336.80
Summer	6.03	3.20	2.83	79.49	25.11	16.83	29.35	62.27	195.07	408.10
Autumn	6.93	4.10	2.83	95.73	26.13	19.92	30.92	63.61	250.03	483.30
Winter	7.72	4.88	2.94	110.0	27.11	16.51	32.15	64.59	297.92	548.60
		·		Second of	alculated	l year				
Spring	6.50	4.37	2.13	92.40	23.00	13.32	28.14	51.76	266.32	475.10
Summer	6.07	3.68	2.39	78.49	26.24	15.47	32.06	58.92	224.78	436.00
Autumn	6.97	4.56	2.41	94.80	27.19	15.64	33.45	60.47	277.88	509.40
Winter	5.53	5.32	2.44	109.12	28.12	15.30	34.55	61.98	324.29	573.30
		·		The year	of salt ba	lance				
Spring	6.31	4.44	1.87	87.56	23.66	12.53	29.70	49.83	270.69	474.00
Summer	5.90	3.00	2.90	73.85	26.97	14.60	33.78	56.78	231.57	437.50
Autumn	6.80	4.66	2.14	90.44	27.87	14.82	35.07	58.46	284.24	510.90
Winter	7.60	5.42	2.18	104.99	28.76	14.52	36.08	60.08	330.32	574.80

Table 5.3 – The hydrochemical forecast

According to the forecast results, we can note changes in the main ingredient concentrations, such as total water hardness (5.90-7.60 mg-equ/dm³), the amount of calcium ions (73.85-104.99 m/dm³) and hydrocarbonate ions (231.57-330.32 mg/dm³). The total mineralization will be 437.5-574.8 mg/dm³. In terms of total mineralization (including calcium, magnesium, chlorides, sulfates, and bicarbonates), the concentration will increase without exceeding the MPC for fishery water bodies, while the sodium concentration will slightly decrease.

In addition, the commissioning of units 3 and 4 will increase the heated water flow into the cooling pond from 50.0 m^{3/s} for one power unit to 200.0 m^{3/s} for four power units and the existing water cooling technology, and water losses due to additional evaporation from the cooling pond surface will amount to 53.1 million m^{3/}year. Due to the use of cooling towers, the amount of additional evaporation will not change when units 5 and 6 are commissioned.

Based on the permissible cooling water temperature (no more than 33 °C), the cooling pond is designed to remove heat from the 4000 MW NPP equipment (4 power units), including master repair schedule. When units 3 and 4 are commissioned, the heat load on the pond will significantly increase, with production and natural losses increasing significantly.

In 2023, the affiliate SS "Atomprojectengineering" performed the KhNPP cooling pond thermal calculation including the KhNPP AP1000 units 5 and 6 construction [34], which resulted in determining the temperature of the cooled water in the pond during operation of one, two, three and four and six power units with a capacity of 1000 MW each:

- during the operation of two power units in summer, the water temperature in the underwater channel (at the turbine condensers inlet) is 24.4-25.5 °C

- during the operation of three power units in summer, the water temperature is 30.0-30.8 °C, which is less than the standard temperature value of 33.0 °C;

- during the operation of four power units, the water temperature in summer months is  $33.0-33.2^{\circ}$ C and will already reach the critical temperature under technological conditions  $33.0^{\circ}$ C [27];

– during the operation of units 5 and 6 – according to the AP1000 process flow diagram, developed by Westinghouse, the purge water from the tower cooling basin of the CWS system is combined with the purge flow from the SWS system, taken from the pump discharge header and sent to the purge sump of the wastewater collection and disposal system for further discharge into the KhNPP cooling pond. Based on the operating conditions of the cooling and blowdown water removal scheme, it can be concluded that the cooling pond receives water already cooled to ambient temperature. Thus, the operation of power units 5 and 6 will not affect the value of additional evaporation from the reservoir and the temperature regime in the reservoir as a whole.

For the operation of units 3&4, in order not to exceed the permissible cooling water temperature, it is envisaged to construct a 1300.0 m flow control dam.

The dam construction will allow to improve the efficiency of cooling circulating water in the pond and ensure the necessary temperature conditions for the operation of six units even in the most unfavorable "hot" hydrometeorological conditions.

The construction of such a dike also helps to avoid a significant dependence of the pond temperature on the most unfavorable wind conditions with westerly winds of 3-6 m/s.

5.2.2.2 Groundwater

The NPP groundwater monitoring network was installed in 1989. The stationary groundwater monitoring network includes 189 wells, which are equipped for the Upper Proterozoic aquifer. The purpose of hydrogeological monitoring is to control the stability of groundwater regime criteria (level, temperature, chemistry) and assess the impact of anthropogenic factors on groundwater. The monitoring network is almost constantly being reconstructed to repair out-of-service wells. Groundwater conditions are assessed in two aquifers:

- quaternary aquifer (groundwater H1);
- pressure aquifer (Vendian H2).

During the operation of units 1 and 2, the anthropogenic impact of the NPP had almost no effect on the groundwater level, since the observation point and the NPP industrial site are located in close proximity to drainage systems, but it affected their chemical composition and temperature as a result of infiltration of industrial water into the ground due to leaks from water supply systems. The groundwater level and the piezometric level of the pressure aquifer can be traced at almost the same elevations, as there is a hydraulic connection between the horizons.

Seasonal fluctuations in the groundwater level averaged 0.5 m in 2017, 0.56 m in 2016, 0.7 m in 2015, 0.52 m in 2014, 0.72 m in 2013, and 0.73 m in 2012. Groundwater levels are stable and respond only to seasonal climate changes.

The difference in groundwater temperature within the industrial site is 10.0 °C. The background temperature is 9.0-10.0 °C. The increase in groundwater temperature was recorded in the following areas: in the unit 1 main building area along the western wall, along the unit 2 main building perimeter, along the eastern wall of the unit 3 main building (central part), near the unit pumping station No. 1, east of the spray cooling pond, along the special building perimeter.

The pressure aquifer water temperature, and the groundwater temperature, is also higher than the background temperature in some areas. The background temperature is 10 °C. The temperature fluctuations within the industrial site is 10.5 °C, with a maximum of 20.5 °C.

Areas with increased groundwater temperatures and the pressure aquifer water practically coincide, water temperature in both horizons being almost identical.

The groundwater chemistry has not significantly changed in recent years. The groundwater chemical composition in both aquifers within the industrial site is characterized by heterogeneity, i.e. significant fluctuations in mineralization over short distances.

The following areas where increased groundwater mineralization was recorded were identified as follows: in the southwestern corner of the unit 3 main building, along the eastern wall of the unit 1 main building, near the northeastern corner of the special building, in the area between unit pumping stations 1 and 2.

It should be emphasized that, according to the monitoring data, chemical and thermal contamination of groundwater and hydraulically connected groundwater in the upper part of the Upper Proterozoic horizon is localized only within the NPP site; groundwater chemistry background (i.e., unaffected by technogenesis) values and temperature were recorded on the site periphery. At the same time, within the industrial site, there is not a continuous field of technogenically contaminated groundwater, but rather disconnected local areas where groundwater is characterized by increased mineralization and temperature.

Groundwater is widespread in the area. The creation of the KhNPP cooling pond, laying of the outlet and inlet channels, and drainage network led to the transformation of hydrogeological conditions. The drainage network allows maintaining, in general, the level as specified during the design.

It is almost impossible to fully predict quantitative changes in the groundwater chemistry and temperature associated with the anthropogenic impact of NPP facilities, as possible losses of chemically contaminated and hot water are unpredictable. However, no significant changes in groundwater are expected during normal operation. The dynamics of changes in groundwater conditions for 2018-2023 according to the "KhNPP Non-Radiation Factors Impact Assessment Report" for 2022 and 2023 see also [44, 45], are in Table 5.4.

Nome of pollutor to/ohomiools		<b>Content of</b>	pollutants/	chemicals,	mg/dm3	
Name of pollutants/chemicals	2023	2022	2021	2020	2019	2018
	Industrial s	site (H1 hori	izon)	-	-	-
pH, units	6.85	7.53	7.50	7.70	7.39	7.33
total hardness, mg-eq/l	10.54	10.90	13.01	9.38	8.94	8.3
calcium, mg/l	151.8	155.33	182.51	131.9	130.73	_
magnesium, mg/l	36.3	38.30	46.42	33.90	29.42	_
potassium, mg/l	9.03	8.60	9.59	10.07	9.24	—
natrium, mg/l	22.4	20.96	20.26	27.36	25.53	26.33
salt ammonium, mg/l	5.83	5.12	1.48	1.99	1.7	3.04
carbonate, mg-eq/l	0.02	0.03	0.02	0.02	0.02	0.02
bicarbonate, mg-eq/l	4.03	4.15	3.97	3.51	3.37	3.46
chloride, mg/l	27.47	28.56	29.68	25.86	21.31	20.27
nitrite, mg/l	0.08	0.05	0.13	0.05	0.07	0.57
nitrate, mg/l	3.65	2.14	2.34	2.00	2.61	5.96
sulfate, mg/l	483.01	517.44	582.04	388.00	357.03	325.15
permanganate oxide, mgO/l			_	—		_
dry residue, mg/l	869.11	901.71	996.40	692.24	676.98	642.41
	Industrial s	ite (H2 hori	zon)		<u>.</u>	
pH, units	7.17	7.71	7.72	8.00	7.66	7.62
total hardness mg-eq/l	7.22	8.58	9.55	6.97	7.46	6.82
calcium, mg/l	92.91	116.19	130.26	90.02	101.99	—
magnesium, mg/l	31.37	33.83	37.15	30.04	28.95	—
potassium, mg/l	9.84	8.81	8.29	9.20	8.98	—
natrium, mg/l	64.96	63.70	61.91	63.37	66.38	69.56
salt ammonium, mg/l	1.62	1.91	1.67	1.58	1.79	2.38
carbonate, mg-eq/l	0.07	0.05	0.04	0.08	0.06	0.07
bicarbonate, mg-eq/l	4	4.38	4.28	3.48	3.93	3.69
chloride, mg/l	24.9	42.88	39.43	36.97	40.75	37.13
nitrite, mg/l	0.07	0.08	0.12	0.03	0.05	0.06
nitrate mg/l	1.13	0.68	0.63	0.56	0.63	0.49
sulfate, mg/l	255.5	322.17	314.40	308.94	256.84	236.52
permanganate oxide, mgO/l						
dry residue, mg/l	610.37	717.92	724.84	644.78	624.84	583.2

Table 5.4 – Dynamics of changes in groundwater conditions (in terms of pollutants/chemicals)

# 5.2.3 Biodiversity

The KhNPP is located in the forest-steppe region of western Ukraine. The ecological state around the nuclear power plant is in harmony with nature, flora and fauna, and is characterized by a high degree of species diversity.

The Hnylyi Rih River has turned into a large lake with powerful waves. Commercial fish are grown in its depths, many birds' nests and fowl are found on its stretch and riverside. The surrounding forests are natural parks protected by the state. A variety of endemic plants are preserved here.

The cooling pond includes a unique ecosystem, which is peculiar by the fact that the pond's southwestern and eastern near-shore areas are shallow and marshy. These are the excellent places for spawning most fish, of which 23 species have been registered here, including a significant crayfish population. According to the KhNPP engineering design, fishing activities are carried out in full at the pond. In spring, several thousands of black carp fry are released into the KhNPP pond, from which individuals with an average weight of 3-4 kilograms grow by autumn. The material for offspring is obtained in incubators located in the biomelioration section of the hydrotechnical shop.

The territory around the pond is characterized by biodiversity: pine, hornbeam-oak, and birch forests, swamp, and lakes surrounding it contribute to the comfortable living of entire bird colonies. In winter, many waterfowl find shelter here: the warm water and the availability of various foods attract them. Swans, in particular, have been choosing a pond for wintering for ten years.

Within the KhNPP SPA and SA, the radiation parameters do not exceed natural values and are at the level of natural background values measured before the start-up of the nuclear plant; no significant changes are envisaged during the construction and operation of new power units, see also clause 5.3.7.3.

The start-up of units 5 and 6 and the accident-free operation of KhNPP as part of six units will not have a negative impact on the general species diversity of invertebrates and insects, see also subpara. 1.5.2 and 5.3.7.3. An increase in water temperature will lead to a quantitative redistribution in the aquatic entomocomplex, and the share of thermophilic and euribiont species will increase. Only the growth of anthropogenic load on the area landscapes, associated with the growth of the population of the city of Netishyn, can lead to a reduction in the number of some rare species of insects in the city's surroundings. This is due to the action of the following factors: the further diversion of meadow areas for vegetable gardens, recreational load (trampling of grass in places of rest, gathering of mushrooms and berries), cutting of forests, general pollution of the environment. At the same time, it was established that the commissioning and normal operation of the KhNPP units 5 and 6 will not have a negative impact on the animal world in the KhNPP supervised area.

It is worth noting that the new power units of KhNPP are being built on the existing NPP site, and therefore, during the implementation of the planned activity, including the expected transport connections for the implementation of the planned activity, there is no impact on the flora and fauna. In case of deforestation during construction, compensatory measures will be provided. Violations of the fodder base, shelters, nesting places and migration routes of animals are not expected.

#### 5.2.4 Conclusions

Based on the results of the assessment, it was determined that the impact on the environment caused by the use of natural resources in the process of carrying out the planned activity, in particular land, soil, water and biodiversity, is negligible, see also clauses 1.5, 5.3.

Soil degradation processes associated with the construction of KhNPP extend only to the area of the industrial site. Their presence in the 30-kilometer area of KhNPP is practically unrelated to the station's operation. Considering the insignificant mass (material) content of radionuclides, even in the emergency emissions of nuclear power plants, the natural mechanisms of soils are sufficient for their firm fixation.

The construction of power units  $N_{2}$  5 and  $N_{2}$  6 with the AP1000 will not lead to changes in the temperature regime of the water cooling since cooling towers are used as a cooling system. In addition, due to the use of cooling towers as a cooling system and the absence of discharge of heat exchange water, the influence on the chemical composition of the water in the cooling pond is practically excluded. A significant impact on groundwater is not expected.

The commissioning and normal operation of the KhNPP units №5 and №6 will not have a negative impact on the fauna in the KhNPP supervised area.

# 5.3 Pollutant emissions and discharges, noise, vibration, light, heat, contamination, radiation and other impact factors, as well as waste management operations

The environmental impact should be analyzed based on the assessment of emissions and discharges, including the impact of pollutants, noise, vibration, and heat and radiation pollution.

#### **5.3.1** Pollutant exposure

The impact of pollutants on atmospheric air is assessed together with a current atmospheric air condition. The impact assessment is carried out by design method during the construction and operation stages see para. 1.5.2.1. Based on the results, the MAC (maximum acceptable concentration) is not expected to exceed and no environmental impact is expected to occur.

#### 5.3.2 Thermal effects

A nuclear power plant is a source of significant heat releases. Approximately twothirds of thermal generation cannot be used to generate electricity and is discharged into the environment.

The heat is mainly transferred to the environment by convection and water evaporation.

Water evaporation in circulating cooling systems leads to salt accumulation. Processing limits on salt in the cooling water require that cooling systems be purged to maintain the salt regime at an acceptable level.

The service water supply plan for KhNPP units 1, 2 (3, 4) nonessential services (Group B) is recycled with a combined cooling pond type: off-stream from the Goryn River and channel from the Hnylyi Rih River. The KhNPP cooling pond was constructed for the plant process needs and is a body of special water use. The service water supply plan for essential services (Group A) is designed as a recycling system with spray ponds.

The service water supply plan for units 5 and 6 is designed as a reverse water supply with cooling towers as a coolant. The KhNPP cooling pond in the adopted cooling and service water supply plan for units 5 and 6 is a source of water intake for cooling tower basin makeup and blowdown and for the needs of other process systems, and is also used for blowdown water discharge.

The service water system (SWS) and circulating water system (CWS) refer to the designed systems involved in the equipment cooling at the AP1000 units 5 and 6. SWS and CWS are used to supply cooling water to the components of the AP1000 reactor and turbine units and use pond water for blowdown and makeup.

CWS and SWS for units 5 and 6 are cooled using cooling towers. Upon cooling tower basin blowdown, water is taken and transfered to the waste water system blowdown sump for further discharge into the cooling pond. Based on this cooling and blowdown water discharge scheme, it can be concluded that the cooling pond receives an ambient temperature cooled water. Thus, the operation of units 5 and 6 will not affect the additional pond evaporation value.

For the units 1-4 cooling systems, the heat exchanger emissions are transferred to circulating cooling water, which removes heat to the atmosphere using the cooling pond for the turbine building main cooling system and Group B cooling system, as well as spray ponds for Group A cooling system.



The cooling system for Group A at units 1, 2 (3, 4) is a safety system designed to remove excess heat from the reactor to the ultimate heat sink during normal operation and in emergency situations. The system is designed to be independent of other cooling systems. The number of cooling system independent channels corresponds to the number of Emergency Core Cooling System channels.



Table 5.5 shows the power units 1-4 designed heat discharges to the cooling pond.

The Group A designed heat releases of the units 1-4 reactor building under nominal mode to the spray ponds are given in Table 5.6.

Table 5.5 – Units 1-4 design heat discharges to the cooling pond
------------------------------------------------------------------

Power unit number	Heat amount									
	Gcal/year	Gcal/day	Gcal/year*							
1	2095	50250	13618 x 10 ³							
2	4190	100500	27235 x 10 ³							
3	6285	150750	40854 x 10 ³							
4	8380	201000	54472 x 10 ³							

Notice*: Heat amount for the annual installed capacity utilization time of 7200 hours

Table 5.6 – The Group A design heat releases of the units 1-4 reactor building under nominal mode to the spray ponds

Power unit number	er Heat amount (maximum/minimum)								
	Gcal/year	Gcal/year							
1	20/2.5	480/60	175.2·10 ³ /21.9·10 ³						
2	40/5	960/120	350.4·10 ³ /43.8·10 ³						
3	60/7.5	1440/180	525.6·10 ³ /65.7·10 ³						
4	80/10	1920/240	700.8·10 ³ /87.6·10 ³						

The environment is also affected by water that enters the atmosphere through additional evaporation and wind droplets from the cooling pond surface. The quantitive values of such water losses are given in Table 5.7.

Table 5.7 – Design values of additional cooling pond evaporation during units 1-4 operation

Month	01	02	03	04	05	06	07	08	09	10	11	12	Per year
		V	Vater e	vapora	ated du	iring ui	nits ope	eration	, millio	n m ³			

Unit 1	1.05	1.08	1.12	1.14	1.26	1.18	1.08	1.03	1.12	1.10	1.06	1.06	13.28
Units 1, 2	2.10	2.16	2.24	2.28	2.52	2.36	2.16	2.06	2.24	2.20	2.12	2.12	26.56
Units 1-3	3.15	3.24	3.36	3.42	3.78	3.54	3.24	3.09	3.36	3.30	3.18	3.18	39.84
Units 1-4	4.20	4.32	4.48	4.56	5.04	4.72	4.32	4.12	4.48	4.40	4.24	4.24	5.12

The evaporation value was determined based on the unit installed capacity utilization hours, equivalent to 7200 hours per year.

For Group A cooling system, the design evaporation and wind drift values for spray pond and cooling pond are given in Tables 5.8 and 5.9.

Table 5.8 – Total design water losses due to evaporation and wind drift during operation of units 1-4  $\,$ 

Month	01	02	03	04	05	06	07	08	09	10	11	12	Per year
		W	ater los	sses du	e to ev	vaporat	ion an	d wind	drift, r	nln.m³			
Unit 1	1.16	1.15	1.15	1.18	1.22	1.24	1.36	1.28	1.18	1.13	1.22	1.20	14.48
Units 1, 2	2.32	2.30	2.30	2.36	2.44	2.48	2.72	2.56	2.36	2.26	2.44	2.40	28.48
Units 1-3	3.48	3.45	3.45	3.54	3.66	3.72	4.08	3.84	3.54	3.39	3.66	3.60	43.44
Units 1-4	4.64	4.60	4.60	4.72	4.88	4.88	5.44	5.12	4.72	4.52	4.88	4.80	57.92

Table 5.9 - Specific design water losses due to evaporation and wind drift from spray pond and cooling pond

Daramators		Quantity									
Parameters	per hour	per day	per year								
	Water losse	es mln.m ³									
Unit 1	0.0017	0.04	14.48								
Units 1, 2	0.0033	0.08	28.96								
Units 1, 2, 3	0.0051	0.12	43.44								
Units 1, 2, 3, 4	0.0068	0.16	57.92								

For CWS units 5 and 6, a naturally ventilated cooling tower was adopted to receive heated circulating water from the main condensers and support the heat exchangers. The circulating water is cooled by evaporation and returned to the cooling tower basin. One cooling tower is provided for each of units 5 and 6.



The cooling tower has a hyperbolic design and is constructed of concrete with PVC film (also see Figure 5.1). A cooling tower bypass system is provided for freeze

protection. The AP1000 standard main condenser is designed to operate properly over a wide range of circulating water flow rates and inlet temperatures.



Figure 5.1 - Tower cooling tower

The SWS for units 5 and 6 is cooled by a fan cooling tower (also see Figure 5.2). The cooling tower is a straight, counter-current tower with a suction fan and a clog-resistant film filling. The cooling tower is divided into two chambers. Each chamber utilizes a single propeller-type fan located at the top of the chamber to draw air upward through the fill against the downstream.



Figure 5.2 - Fan-type two-section cooling tower

The configuration and size of the cooling tower allow to achieve the required temperature indicators when cooling the heated water.

The cooling tower design is a fiberglass composite made of refractory materials. The tower skeleton and fan deck are made of pultruded fiberglass, and a tower shell is made of corrugated glass-reinforced polyester or equivalent material. The tower filling, located below the water distribution piping, is made of PVC or polypropylene and is designed to avoid clogging.

The cooling tower performance requirements are tied to the ambient wet bulb temperature of 30°C. The requirements for normal operation at capacity apply continuously and must be met even on the hottest day of the year.

The values of heat emissions to the atmosphere from the CWS and SWS cooling towers in accordance with [32, 33] are given in Table 5.10. Estimated values of water losses due to evaporation and wind removal from cooling towers during operation of units 5&6 are given in Table 5.11. Specific estimated values of water losses due to evaporation and wind removal from the CWS tower-type cooling towers are given in Table 5.12.

Table 5.10 – Thermal releases to the atmosphere from the CWS and SWS cooling towers

Cooling tower type	Cooling tower	Heat amount maximum/minimum						
	number, pcs.	Gcal/h	Gcal/day	Gcal/year				
CWS cooling towers	2	3810.0/0	91440.0/0	27.4 x 10 ⁶ /0				
SWS fan cooling	Two double-chamber	174 6/27 74	4100 A/005 76	1.26 x 10 ⁶ /				
towers	cooling towers	1/4.0/57.74	4190.4/905.76	0.74 x 10 ⁶				
Total	Λ	2 004 6/27 74		28.66 x 10 ⁶ /				
TULAI	4	5 964.0/57.74	95 050.4/905.70	0.74 x 10 ⁶				

Table 5.11 – Design water losses due to evaporation and wind drift during units 5 and 6 operation

Month	01	02	03	04	05	06	07	08	09	10	11	12	Per year
<b>Operating hours</b>	744	647	558	540	558	540	558	558	540	558	655	744	7200
	Amou	unt of	water	evapo	rated	during	unit c	perati	on, m	illion n	n ³		
Units 5, 6	5.07	4.41	3.80	3.68	3.80	3.68	3.80	3.80	3.68	3.80	4.46	5.07	49.06

Table 5.12 – Specific design water losses due to evaporation and wind drift from the CWS cooling towers

Doromotoro	Quantity								
Parameters	per hour	per day	per year						
	Water losse	s mln. m ³							
Units 5, 6	0.0068	0.16	48.96						

#### 5.3.3 Wastewater impact

NPP industrial, domestic effluents, stormwater drainage and Netishyn domestic wastewater are not discharged directly into public water bodies. The treated wastewater is discharged to the cooling pond, which is the NPP water body, and thus compensates for water losses in the pond and does not directly affect the degree of surface water pollution. Inputs to eventual AP1000 plant wastewater discharge volumes are in accordance with [46].

The AP1000 units 5 and 6 provide features and provisions to minimize discharges and associated impact on the environment. The AP1000 plant wastewater discharge scheme in accordance with [46] is shown in Figure 5.3.



Figure 5.3 – The AP1000 plant wastewater discharge

The WWS collects the non-radioactive plant equipment wastewater, floor drains, process fluids and residual system flushing effluents before treatment and subsequent discharge to the cooling pond. Effluent is collected in the turbine building sumps and discharged to an oil separator, where the oil is coagulated and separated from the water, and solids settle. The waste water without an oil (e.g. rainwater) can be discharged to a special waste water retention basin (WWRB) for settling and treatment. The WWS oil separator removes oil and settling solids from the plant wastewater before discharge into a cooling pond. The used oil is collected in the separator's internal tank. The effluent in the tank is pumped to the blowdown sump. In the event radioactivity is detected in the discharge from the sumps, the waste water is diverted from the sumps to the WLS for processing and disposal. The WWS parameters are shown in Table 5.13.

Table 5.13 – The WWS parameters per one unit

Fluid type	Flow rate, m ³ /h	Temperature (max), °C	Pressure (max), MPa
Wastewater	56.78	48.9	0.76

The CWS cooling tower basin is fed by the RWS. The cooling tower basin also receives a blowdown flow from the SWS. The CWS blowdown flow is taken from the pump discharge header and directed to the WWS blowdown sump for further discharge to the KhNPP cooling pond. Inputs to eventual blowdown water discharge volume from the CWS and SWS are given in Table 5.14.

Table 5.14 – Eventual blowdown water discharge volume from the CWS and SWS per one unit

Fluid type	Flow rate, m ³ /h	Temperature (max), °C

The CWS cooling tower basin blowdown water	1071.0	3215.0
The SWS fan cooling tower basin blowdown water	14.0	47.0

The wastewater from the WWS and Passive Containment Cooling System (PCS) flows to the Storm Drain System (DRS). Safe leaks and flushing effluent from the tanks, hydrants, and basins in the SWS, RWS, Fire Protection System (FPS), Heating System, and Demineralized Water Transfer and Storage System (DWS) will be diverted to the DRS. The DRS also receives and discharges wastewater from the Gravity and Roof Drain Collection System (RDS).

The Sanitary Drainage System (SDS) collects the sanitary waste from units 5 and 6 restrooms, locker rooms in the turbine building, auxiliary building and annex building and transports it through collectors connected to the building sewage systems to the domestic wastewater treatment plant for a full biological treatment and additional treatment at biological facilities. The capacity of the treatment facilities is 20.0 thousand m³/day, see also Appendix F the actual average load, as informed by the Khmelnytskyy NPP utility company during the operation of units 1 and 2, ranges from 9.0 to 11.0 thousand m³/day, including the wastewater from Netishyn in the amount of approximately 8.5 thousand m³/day. Basen on the preliminary calculations, the power units utility wastewater is approximately 0.38 thousand m³/day. Thus, the capacity of the existing treatment facilities is sufficient to accept and treat domestic wastewater from the designed units 5 and 6. The SDS output parameters are shown in Table 5.15.

Table 5.15 – The SDS output parameters

Fluid type Flow rate, m ³ /h		Temperature (max), C	Pressure (max)	
Sanitary waste water	7.95	21.1	Atmospheric	

Rainwater from the roofs of buildings and the adjacent territory of units 5 and 6 are to be discharged through the DRS into the SWS inlet channel and used in the NPP cycle, which will reduce the need for fresh additional service water. The discharge volume mostly depends on the expected precipitation. Aside from rainfall collection from each building to the ground and into the DRS, the storm drain release point also handles flows from the PCS, which has a normal expected flow of 0.013 m3 /h, and a maximum flow of 114 m3 /h during off normal events.

#### 5.3.4 Noise exposure

The following factors are considered in the noise impact assessment on the environment:

- the impact of additional noise sources due to the commissioning of units 5 and 6;

- the absence of permanent workplaces for service personnel at the industrial site, outside the production buildings and structures, it is reasonable to conduct a noise impact assessment only inside of these buildings and structures.

In case of construction works at units 3&4, noise impact assessment shall include the noise of construction equipment.

The main building No. 5, 6, a number of auxiliary production buildings and structures are put into operation with units 5&6 at the NPP industrial site.

In these buildings and structures, the source of sound impact on the operating personnel is the rotating equipment installed in them (turbine unit, pumping units, ventilation units), as well as the reducing equipment.

Thermal and sound insulation is performed to reduce noise levels. Sound absorption is also provided by the walls and ceilings of the turbine hall.

In the main building, the production process is fully automated, permanent workplaces of the duty personnel are located in special rooms equipped with soundproof enclosing structures. The design sound pressure level in these rooms, provided by the insulating structures, meets the regulatory requirements for control rooms and laboratories, namely, does not exceed the permissible values given in Table 5.16.

Table 5.16 – Permissible sound pressure levels in control rooms, laboratories, computing machinery

Name	Octave bands with geometric mean frequencies, Hz								
	31.5	63	125	250	500	1000	2000	4000	8000
Permissible sound pressure level, dB	93/96*	79/83	70/74	63/68	58/63	55/60	52/57	50/55	49/54
Integrated sound level, dBA					60/65				

Note*: The table shows values for control rooms in the numerator and for laboratories in the denominator

During the operation of the installed equipment, the operating personnel either are absent or are there periodically and for a short time (walkdowns). In this case, it is necessary to protect personnel from noise using soundproof cabins and personal protective equipment (earplugs, earphones).

Soundproof cabins of the KVU type are installed.

Areas with a sound level above 80 dBA are marked with safety signs in accordance with the standards.

To protect service personnel temporarily staying on the territory (duty and repair), it is provided to use earphones to reduce noise in the range of 31.5 - 8000 Hz by 7 - 45 dB.

The entity must ensure monitoring the noise level at workplaces in accordance with the standards at least once a year.

#### 5.3.4.1 Noise protection measures during construction

To reduce the noise level during construction and installation works on the construction site, it is necessary to:

- ensure that vehicle engines are turned off while on the construction site;
- turn off loudspeaker communication;
- do not carry out welding without protective screens;

- exclude the performance of work accompanied by noise exceeding the permissible standards;

- exclude the operation of equipment with a noise and vibration level exceeding the permissible standards;

– to soundproof construction machinery engines, use protective covers and hoods with multilayer coating;

- to isolate local sources, use temporary noise protection screens, noise control curtain, tents (for example, place compressors in a sound-absorbing tent).

Construction and assembly machinery and vehicles are the main sources of exhaust gases and noise, which cannot be avoided during the construction process.

#### 5.3.4.2 Noise limitations under the AP1000 operation

The environmental conditions (lighting, noise, ambient temperature, radiation, air quality, and humidity) in the control room, remote shutdown room, and local control stations are determined using generally accepted industrial and human technology standards.

Noise limit criteria applies only to permanently manned locations (e.g., the main control room (MCR) or local control room) in the standard plant design. As a result, few equipment have noise limitation design features in the Nuclear Island area of the AP1000 plant design.

The AP1000 standard design has a noise criterion (NC) NC-45 for rooms with computer and communication equipment; auxiliary control room, access control areas; chemical laboratories, radioactive waste control room, locker room, and resting rooms. Definitions and explanations of NC-45 are given in the relevant documents [47].

*Main control room and/or remote shutdown room.* The background noise level shall not exceed 65 A-weighted decibels (dB(A)) when the nuclear island HVAC non-radioactive ventilation system (VBS) is in operation. The background noise level shall not exceed 65 dB(A) when the VES associated with the HVAC main control room safety is in operation.

*Other areas.* The noise level in the auxiliary building areas, except for the MCR and remote shutdown room, is 5 dBA lower than the prescribed level.

*Containment.* The Containment noise level should be limited to 70 dBA during refueling to maintain proper operation and communication between operators during refueling breaks.

#### Reducing excessive noise levels

Noise limitation criteria are applied only to areas with permanent personnel (e.g., main control room) in the standard plant design. Thus, not much equipment has noise control design features. Westinghouse uses built-in equipment with silencers, noise-absorbing acoustic panels, and carpeted floors to protect personnel from excessive noise during operation.

# Silencers

Silencers are used on the Steam Generator System (SGS) main steam line power operated relief valves (PORV) and in the Main Control Room Emergency Habitability

System (VES) air amplifier and eductor to protect operators from excessive noise when these components are in use.

*Main steam line power operated relief valve silencers.* The function of the steam generator relief valve silencers is to reduce the audio output of the main steam PORVs to 110 dB(A) at the outlet of the silencer while maintaining a differential pressure less than 0.160 MPa abs (23.2 psia). In addition to limiting, the noise provided from the PORVs operation to the operators, this audio level is the maximum allowed due to operating pressure waves that result in metal fatigue issues that would result in damage to the main steam PORVs. The steam generator relief valve silencers shall be cylindrical in cross section and vertical in orientation, utilizing internal component parts, which include but are not limited to, a pressurized inlet flow diffuser, expansion plenum and acoustic tube module to attenuate the noise generated from the steam from the relief valves.

Air amplifier silencers. The air amplifier silencers are safety-related HVAC silencers and their function is to reduce the noise in the passive air filtration line generated by the operation of the eductor. They are part of the MCR passive air filtration and they act to keep noise levels within the requirement of 65 dB(A) during VES operation. There is one silencer upstream of the eductor and one downstream of the eductor. These silencers are located within the main control room area.

*Eductor Bypass Line Silencer*. This eductor bypass line silencer is used to reduce the noise level of the VES breathable air into the main control room in the event the bypass line is used in the VES. Under full flow conditions the sound level at a location of 1 meter downstream and 1 meter away from the outside of the uninsulated chamber shall not exceed 65 dB(A). This silencer is located within the main control room area.

#### MCR acoustic panels

For additional protection against excessive noise in the MCR, acoustic panels are used.

The panels are 25.4 mm thick, and the material composition consists of fiberglass core, zinc-plated steel eyed screws along top edge, and 100% polyester fabric.

These panels are located within the main control room to provide noise abatement for the operators.

# **Other MCR features**

The MCR has concrete and steel walls and ceiling. To reduce the overall noise-level requirements in the MCR, antistatic carpet should be installed on the floor of the MCR. The carpet electrostatic discharge shall have a limit of 3.5kV.

# **5.3.5** Vibration effect

The AP1000 unit Special Monitoring System (SMS) is a system consisting of the I&C-related systems that provides a special diagnostic and long-term monitoring functions, namely:

- − Digital metal impact monitoring system (DMIMS-DXTM)
- Vibration integrity monitoring system (VIMS)

- Feedwater vibration monitoring system (FWVMS)

#### 5.3.5.1 Digital Metal Impact Monitoring System (DMIMS-DX)

The Digital Metal Impact Monitoring System (DMIMS-DX) provides for the detection of the presence of metallic debris in the RCS when the debris impacts against the internal parts of the RCS.

The DMIMS-DX system is designed to warn operators about the presence of metal debris in the RCS.

#### 5.3.5.2 Vibration Integrity Monitoring System (VIMS)

The VIMS system is a subsystem of the SMS, which consists of the reactor coolant pump vibration monitoring (RCPVM) and control rod drive mechanism (CFVM) systems.

The main purpose of the VIMS is to provide early detection of abnormal vibration levels of the RCPs and CRDM fans to avoid or reduce damage to this equipment. The plant personnel can use this information to focus their efforts on performing corrective actions to eliminate faults in the primary coolant system and minimize the need for overhauls. The VIMS is a continuous monitoring system that provides information to assist in the evaluation of the performance of the RCPs and CRDM fans.

The main function of the RCPVMS is to monitor the vibration of the four reactor coolant pumps. In addition, the RCPVMS monitors the speed of each pump. The RCPVMS is one of the special monitoring systems designed to protect critical components. The CFVMS processes vibration sensor signals from the CRDM four fans and monitors vibration levels.

All monitored data, along with the configured signal settings and signal status, are formatted and displayed on a local display. The graphical user interface allows the user to look at the various monitors to quickly evaluate the monitoring data of each channel.

#### 5.3.5.3 Feedwater Vibration Monitoring System (FWVMS)

The main purpose of the FWVMS is to provide early detection of abnormal vibration levels of the main and booster feedwater pumps, pump motor and reductor to avoid or reduce damage to this equipment. The FWVMS is a continuous monitoring system that also provides pump speed monitoring.

The AP1000 unit has three feedwater pump channels A, B, and C. Each channel consists of a main pump, a booster pump, an electric motor, and a reductor. All the FWVMS sensors are non-contact vibration displacement sensors.

All monitored data, along with the configured signal setpoints and signal status, are formatted and displayed on the local display.

#### 5.3.6 Ultrasound exposure

No ultrasound exposure from operating mechanical equipment was recorded during the Khmelnytskyy NPP units operation.

A single short-term localized ultrasonic exposure is possible during ultrasonic weld control.

The increase in noise background under heavy traffic will not affect insects. At the same time, ground vibration caused by heavy traffic can be considered a significant disturbance factor. Protected species will not be affected by this factor.

# 5.3.7 Radiation effect

The key types of the Khmelnytskyy NPP (KhNPP) radiation effects on the air are gaseous radioactive releases from the NPP ventilation system. The impact of KhNPP releases and discharges of radioactive substances over the entire operating period on the radiation situation within the KhNPP area was not detected against the global fallout, as evidenced by the environmental sample results.

The absolute dose rate value at the monitoring stations during the entire operating period does not depend on their location relative to KhNPP and is caused by the natural background and radionuclides fallout of global origin.

The dynamics of radiation background in the supervised area is caused by the release of technogenic radionuclides as a result of the Chornobyl accident, global radiation fallout and fluctuations in the radiation background. Based on the calculations, it can be concluded that the commissioning of units 5 and 6 will not lead to significant radiation situation changes both on the KhNPP industrial site and in the SPZ and SA.

Within the investigations for the construction of units 3 and 4, the experimental works on soil sampling with gamma-emitting radionuclides measuring were performed to obtain data on the territory contamination. Such studies for units 3 and 4 can also be used for the assessment of the AP1000 units 5&6 construction due to the absence of significant anthropogenic impact.

Long-term observations carried out by the KhNPP external dosimetry laboratory prove that there is no significant impact on the SPZ and SA territories.

Figure 5.4 shows the results of gamma background monitoring in the KhNPP SPZ and SA based on the ARSMS stations data.

As shown by the available measurements, the KhNPP site dose exposure values are fully consistent and do not exceed the zero background values specified in the [48].



Figure 5.4 – Gamma background monitoring in the KhNPP SPZ and SA based on the ARSMS stations data

5.3.7.1 Comparative impact analysis of the planned radioactive substances releases during the AP1000 reactors operation and actual KhNPP VVER units releases

Table 5.17 shows the measuring results of the KhNPP gas and aerosol emissions for the period of 8 years: from 2016 to 2023, as reported in [49].

Table 5.17 – Gas and aerosol emissions	measuring results at	KhNPP for the pe	eriod
of 8 years: from 2016 to 2023, (Bq/year)			

	2016	2017	2018	2019	2020	2021	2022	2023
RIG	1,57E+13	1,24E+13	1,22E+13	1,37E+13	1,15E+13	1,24E+13	1,12E+13	1,05E+13
Radioiodine	1,91E+07	1,10E+07	1,07E+07	1,24E+07	1,21E+07	1,65E+07	1,60E+07	1,31E+07
3H*	5,48E+11	3,19E+11	3,07E+11	3,30E+11	5,26E+11	6,35E+11	4,38E+11	2,10E+12
51 <b>Cr</b>	1,27E+06	1,16E+06	1,40E+06	1,12E+06	1,47E+06	1,28E+06	1,85E+06	1,86E+06
54Mn	1,95E+05	1,57E+05	1,40E+05	2,26E+05	1,84E+05	1,66E+05	2,32E+05	2,07E+05
59Fe	3,16E+05	2,96E+05	2,88E+05	2,43E+05	3,31E+05	2,92E+05	3,94E+05	3,87E+05
58Co	4,78E+05	1,54E+05	1,44E+05	1,60E+05	3,04E+05	1,55E+05	2,62E+05	2,20E+05
60 <b>C</b> O	2,00E+06	1,87E+05	4,23E+05	6,10E+05	1,20E+06	1,70E+05	2,54E+05	2,35E+05
95 <b>Zr</b>	2,79E+05	2,73E+05	2,44E+05	2,07E+05	3,08E+05	2,70E+05	3,55E+05	3,65E+05
95Nb	2,39E+05	4,64E+05	1,72E+05	1,95E+05	2,62E+05	1,70E+05	2,72E+05	2,68E+05
110mAg	1,51E+06	7,15E+06	2,36E+06	1,10E+06	3,98E+05	9,53E+05	4,60E+05	1,19E+06
134 <b>C</b> s	7,67E+05	1,46E+05	1,60E+05	1,43E+05	1,81E+05	1,61E+05	2,23E+05	2,32E+05
137 <b>C</b> s	1,66E+06	1,60E+06	5,22E+05	2,20E+05	2,19E+05	2,47E+05	4,38E+05	4,38E+05
89Sr	1,20E+05	2,15E+04	8,40E+04	6,57E+04	1,96E+04	1,46E+03	1,20E+04	1,30E+04
90 <b>S</b> r	2,66E+05	1,98E+05	2,34E+05	2,46E+05	1,22E+05	5,69E+04	8,58E+04	1,05E+05

According to the information [50], the noble gas releases from the AP1000 unit during normal operation and anticipated operational occurrences are presented in Table 5.18.

		Activity release (Bq/year) of airborne gases (Note)								
Nuclide	Waste Gas System	Containment Building	Auxiliary building	Turbine Building	Condenser Air Removal System	Total release				
Kr-85m	4.60E+11	1.40E+11	1.60E+13	8.50E+08	7.80E+12	2.40E+13				
Kr-85	3.00E+15	1.10E+13	5.20E+13	2.90E+09	2.60E+13	3.10E+15				
Kr-87	negligible	4.40E+10	1.70E+13	2.60E+08	2.20E+12	1.90E+13				
Kr-88	6.70E+09	1.00E+11	1.80E+13	9.60E+08	8.50E+12	2.70E+13				
Xe-131m	1.10E+15	3.10E+13	1.80E+14	9.30E+09	8.10E+13	1.40E+15				
Xe-133m	3.60E+10	6.70E+12	7.40E+13	4.10E+09	3.50E+13	1.10E+14				
Xe-133	2.40E+14	8.90E+13	6.30E+14	3.30E+10	2.90E+14	1.30E+15				
Xe-135m	negligible	6.70E+10	1.30E+14	7.00E+09	5.90E+13	1.90E+14				
Xe-135	negligible	3.10E+12	1.70E+14	2.90E+10	2.60E+14	4.40E+14				
Xe-137	negligible	negligible	3.40E+13	1.80E+09	1.60E+13	4.80E+13				
Xe-138	negligible	2.90E+10	5.90E+13	3.30E+09	2.90E+13	8.90E+13				
Total noble	e gas					6.70E+12				

Table 5.18 – Noble gas releases during normal operation and anticipated operational occurrences

Note: Values less than 3.7E-5 GBq/year are considered to be negligible.

Comparing the values of the total anticipated release of the AP1000 unit with the actual RIG release of the KhNPP units, it can be seen that the anticipated value of the AP1000 unit RIG release is significantly lower.

The anticipated gas and aerosol radionuclide release values for the AP1000 units are given in Table 5.19.

Table 5.19 – Gas and aerosol releases of the AP1000 plant during normal operation and anticipated operational occurrences

	Release (Bq/year) of airborne gases (Note)							
Nuclide	Waste Gas	Containment	Auxiliary	Fuel Handling	Total release			
	System	Building	building	Area	Total release			
Cr-51	negligible	negligible	1.20E+05	6.70E+04	2.30E+05			
Mn-54	negligible	negligible	negligible	1.10E+05	1.60E+05			
Co-57	negligible	negligible	negligible	negligible	negligible			
Co-58	negligible	9.30E+04	7.00E+05	7.80E+06	8.50E+06			
Co-60	negligible	negligible	1.90E+05	3.00E+06	3.20E+06			
Fe-59	negligible	negligible	negligible	negligible	negligible			
Sr-89	negligible	4.80E+04	2.80E+05	7.80E+05	1.10E+06			
Sr-90	negligible	negligible	1.10E+05	3.00E+05	4.40E+05			

	Release (Bq/year) of airborne gases (Note)								
Nuclide	Waste Gas	Containment	Auxiliary	Fuel Handling	Total release				
	System	Building	building	Area					
Zr-95	negligible	negligible	3.70E+05	negligible	3.70E+05				
Nb-95	negligible	negligible	negligible	8.90E+05	9.30E+05				
Ru-103	negligible	negligible	negligible	negligible	negligible				
Ru-106	negligible	negligible	negligible	negligible	negligible				
Sb-125	negligible	negligible	negligible	negligible	negligible				
Cs-134	negligible	negligible	2.00E+05	6.30E+05	8.50E+05				
Cs-136	negligible	negligible	negligible	negligible	negligible				
Cs-137	negligible	negligible	2.70E+05	1.00E+06	1.30E+06				
Ba-140	negligible	negligible	1.50E+05	negligible	1.60E+05				
Ce-141	negligible	negligible	negligible	negligible	negligible				
Total radionuc	ide release				1.70E+07				

Note: Values less than 3.7E-5 GBq/year are considered to be negligible.

In addition, to detail the information on the site radiation state where the AP1000 units are planned to be located, Tables 5.20 - 5.22 below show the measuring data of the ARSMS station and the KhNPP laboratory measurements. The monitoring station (contamination monitoring station-4) is located in the immediate vicinity of the planned AP1000 units location. The sampling frequency is defined in [51]. It should be noted that the radionuclide activity value, which is at the limit of the lower measurement range, is given in terms of I¹/₂ minimum measured activity.

Table 5.20 – Radioactive airborne substances at the contamination monitoring station-4 for 8 years, microBq/m³

Sampling period	Cs-137	Cs-134	Co-60	I-131	Ag-110 m	Sr-90
3 qtr. 2016	1.56E+00	1.95E-02*	2.50E-02*	2.50E-02*	-	1.70E-01
4 qtr. 2016	6.13E-01	1.21E-02*	1.57E-02*	3.06E-02*	-	4.70E-01
1 qtr. 2017	1.24E+00	1.31E-02*	1.88E-02*	3.22E-02*	-	2.20E-01
2 qtr. 2017	8.03E-01	1.30E-02*	1.53E-02*	3.72E-02*	-	6.70E-01
3 qtr. 2017	3.01E+00	7.42E-03*	7.42E-03*	1.64E-02*	-	2.45E-01
4 qtr. 2017	2.03E+00	1.26E-02*	1.39E-02*	1.92E-02*	-	3.19E-01
1 qtr. 2018	1.46E+00	3.79E-03*	5.30E-03*	9.46E-03*	-	1.76E-01
2 qtr. 2018	4.17E+00	3.06E-02*	3.35E-02*	7.72E-02*	-	3.44E-01
3 qtr. 2018	7.69E-01	1.38E-02*	1.81E-02*	4.23E-02*	-	5.15E-01
4 qtr. 2018	1.18E-00	6.26E-03*	1.02E-02*	1.22E-02*	-	4.09E-01
1 qtr. 2019	7.16E-01	6.41E-03*	8.01E-03*	1.47E-02*	-	1.49E-01
2 qtr. 2019	6.38E-01	2.41E-02*	4.36E-02*	5.28E-02*	-	1.58E-01
3 qtr. 2019	3.82E-01	1.36E-02*	1.58E-02*	2.98E-02*	-	3.85E-01
4 qtr. 2019	3.48E-01	8.54E-03*	9.49E-03*	2.15E-02*	-	3.62E-01
1 qtr. 2020	3.87E-01	5.17E-03*	5.72E-03*	1.25E-02*	-	3.12E-02
2 qtr. 2020	1.63E+00	1.27E-02*	1.40E-02*	2.73E-02*	-	9.16E-02
3 qtr. 2020	3.13E-01	8.20E-03*	9.76E-03*	2.16E-02*	-	2.34E-02
4 qtr. 2020	3.49E-01	2.15E-02*	3.19E-02*	4.41E-02*	-	4.89E-02
1 qtr. 2021	6.40E-01	2.01E-02*	2.99E-02*	4.50E-02*	2.18E-02*	5.57E-02

Sampling period	Cs-137	Cs-134	Co-60	I-131	Ag-110 m	Sr-90
2 qtr. 2021	2.03E+00	2.87E-02*	4.17E-02*	8.25E-02*	3.17E-02*	4.93E-02
3 qtr. 2021	3.46E-01	1.30E-02*	6.06E-02*	4.06E-02*	2.84E-02*	5.36E-02
4 qtr. 2021	6.12E-01	4.10E-02*	2.42E-01*	7.51E-02*	5.53E-02*	7.12E-02
1 qtr. 2022	7.23E-01	3.43E-02*	4.02E-02*	7.06E-02*	4.37E-02*	7.55E-02
2 qtr. 2022	6.82E-01	4.86E-02*	5.48E-02*	1.14E-01*	5.98E-02*	1.26E-01
3 qtr. 2022	8.28E-01	1.08E-02*	1.08E-02*	2.59E-02*	9.81E-03*	5.61E-02
4 qtr. 2022	6.19E-01	1.45E-02*	1.83E-02*	3.54E-02*	1.32E-01*	1.40E-01
1 qtr. 2023	6.26E-01	8.52E-03*	1.03E-02*	2.37E-02*	9.43E-03*	8.06E-02
2 qtr. 2023	1.67E+00	7.25E-03*	7.57E-03*	2.33E-02*	7.88E-03*	1.38E-01
3 qtr. 2023	7.51E-01	8.01E-02*	7.92E-02*	1.64E-01*	3.31E-01	4.51E-02
4 qtr. 2023	4.30E-01	2.02E-02*	6.06E-02*	3.82E-02*	1.05E-01	9.95E-02

Note: Values corresponding to ½ minimum measured activity are marked.

Table  $5.21 - \text{The absorbed gamma radiation dose values (mSv) for <math>2016 - 2023$ 

Monitoring station	2016 1 qtr.	2016 2 qtr.	2016 3 qtr.	2016 4 qtr.
	0.21	0.19	0.19	0.16
	2017 1 qtr.	2017 2 qtr.	2017 3 qtr.	2017 4 qtr.
	0.18	0.21	0.20	0.21
	2018 1 qtr.	2018 2 qtr.	2018 3 qtr.	2018 4 qtr.
	0.18	0.19	0.21	0.23
	2019 1 qtr.	2019 2 qtr.	2019 3 qtr.	2019 4 qtr.
	0.23	0.23	0.24	0.19
Site-4	2020 1 qtr.	2020 2 qtr.	2020 3 qtr.	2020 4 qtr.
	0.22	0.24	0.22	0.26
	2021 1 qtr.	2021 2 qtr.	2021 3 qtr.	2021 4 qtr.
	0.21	0.23	0.22	0.25
	2022 1 qtr.	2022 2 qtr.	2022 3 qtr.	2022 4 qtr.
	0.25	0.24	0.23	0.23
	2023 1 qtr.	2023 2 qtr.	2023 3 qtr.	2023 4 qtr.
	0.22	0.23	0.24	0.23

Table 5.22 – Exposure dose rate (EDR) at the contamination monitoring station-4 (CMS-4) for 8 years,  $\mu Sv/h$ 

Year	Month	Minimum EDR on CMS-4, μSv/h	Maximum EDR on CMS-4, μSv/h	Average EDR on CMS-4, μSv/h
2016	1	0,09	0,13	0,1
2016	2	0,09	0,13	0,11
2016	3	0,09	0,14	0,11
2016	4	0,09	0,14	0,11
2016	5	0,09	0,15	0,11
2016	6	0,09	0,14	0,11
2016	7	0,09	0,15	0,11
2016	8	0,09	0,14	0,11
2016	9	0,09	0,13	0,11
2016	10	0,09	0,15	0,11

Year Month		Minimum EDR on	Maximum EDR on	Average EDR on
		CMS-4, μSv/h	CMS-4, μSv/h	CMS-4, μSv/h
2016	11	0,08	0,14	0,1
2016	12	0,09	0,13	0,1
2017	1	0,09	0,12	0,1
2017	2	0,08	0,12	0,1
2017	3	0,09	0,12	0,11
2017	4	0,09	0,13	0,11
2017	5	0,09	0,14	0,11
2017	6	0,09	0,16	0,11
2017	7	0,09	0,13	0,11
2017	8	0,09	0,13	0,11
2017	9	0,09	0,17	0,11
2017	10	0,09	0,16	0,11
2017	11	0,09	0,13	0,11
2017	12	0,09	0,13	0,11
2018	1	0,09	0,14	0,11
2018	2	0,09	0,13	0,1
2018	3	0,09	0,13	0,1
2018	4	0,09	0,13	0,11
2018	5	0,09	0,14	0,11
2018	6	0,1	0,16	0,11
2018	7	0,09	0,16	0,11
2018	8	0,09	0,18	0,11
2018	9	0,09	0,13	0,11
2018	10	0,09	0,18	0,11
2018	11	0,09	0,14	0,11
2018	12	0,09	0,13	0,11
2019	1	0,09	0,12	0,1
2019	2	0,09	0,12	0,1
2019	3	0,09	0,13	0,11
2019	4	0,09	0,15	0,11
2019	5	0,09	0,14	0,11
2019	6	0,09	0,15	0,11
2019	7	0,1	0,13	0,11
2019	8	0,09	0,15	0,11
2019	9	0,09	0,18	0,11
2019	10	0,09	0,13	0,11
2019	11	0,1	0,13	0,11
2019	12	0,09	0,13	0,11
2020	1	0,09	0,16	0,11
2020	2	0,09	0,13	0,11
2020	3	0,09	0,12	0,1
2020	4	0,09	0,14	0,11
2020	5	0,09	0,13	0,11
2020	6	0,09	0,15	0,11
2020	7	0,09	0,15	0,1
2020	8	0,09	0,14	0,11
2020	9	0,09	0,16	0,11

Voor	Month	Minimum EDR on	Maximum EDR on	Average EDR on		
I cal	WIOITUI	CMS-4, μSv/h	CMS-4, μSv/h	CMS-4, μSv/h		
2020	10	0,09	0,18	0,11		
2020	11	0,09	0,14	0,1		
2020	12	0,09	0,15	0,11		
2021	1	0,09	0,13	0,11		
2021	2	0,08	0,12	0,1		
2021	3	0,09	0,12	0,1		
2021	4	0,09	0,12	0,11		
2021	5	0,1	0,13	0,11		
2021	6	0,09	0,13	0,11		
2021	7	0,09	0,17	0,11		
2021	8	0,09	0,14	0,11		
2021	9	0,09	0,18	0,11		
2021	10	0,09	0,12	0,1		
2021	11	0,09	0,15	0,11		
2021	12	0,09	0,15	0,11		
2022	1	0,09	0,13	0,1		
2022	2	0,09	0,12	0,1		
2022	3	0,09	0,13	0,11		
2022	4	0,1	0,14	0,11		
2022	5	0,09	0,13	0,11		
2022	6	0,09	0,14	0,11		
2022	7	0,09	0,14	0,11		
2022	8	0,09	0,16	0,11		
2022	9	0,09	0,16	0,11		
2022	10	0,09	0,12	0,11		
2022	11	0,09	0,13	0,11		
2022	12	0,09	0,15	0,11		
2023	1	0,09	0,13	0,11		
2023	2	0,09	0,13	0,1		
2023	3	0,05	0,17	0,08		
2023	4	0,05	0,15	0,08		
2023	5	0,05	0,12	0,08		
2023	6	0,05	0,15	0,08		
2023	7	0,05	0,14	0,08		
2023	8	0,04	0,19	0,08		
2023	9	0,05	0,11	0,08		
2023	10	0,04	0,17	0,08		
2023	11	0,05	0,13	0,08		
2023	12	0,05	0,14	0,07		

Thus, following the analysis of the statistical release data for the KhNPP existing units with the anticipated data for the AP1000 units operation, it can be concluded that no significant environmental impact related to emissions during the construction and further operation of units using the AP1000 technology is expected.

In addition, the information provided allows using the input data on the radiation impact assessment performed for KhNPP units 3 and 4, which are also pressurized water units.

#### 5.3.7.2 Anticipated radiation effect on land resources

To obtain data on radioactive contamination of the area under study during the units 3 and 4 completion, the experimental works on soil sampling with gamma-emitting radionuclides measuring were performed. In view of the absence of significant anthropogenic impact, as evidenced by long-term observations and measurements throughout the KhNPP operation, the data of such studies can also be used for the impact assessment during the completion and further operation of units 5 and 6 using the AP1000 technology.

Soil sampling was performed to determine the contamination density in the KhNPP area according to a standard methodology using a drive sampler (sampling depth was 20 cm). About 100 soil samples were analyzed in the process.

The results on ¹³⁷Cs in the soil within the KhNPP 30-km area are shown in Figure 5.5.



# Figure 5.5 – Density of ¹³⁷Cs contamination in the KhNPP proximity

This radionuclide is of technogenic origin. The contamination shown on the map consists of a superposition of global fallout, fallout from the Chornobyl accident, and fallout caused by aerosol emissions from KhNPP units 1 and 2. The last contamination source is so negligible that its separation from the total contamination is impossible. No other technogenic radionuclides were detected. This confirms the NPP operation in normal mode.

On that basis, it can be stated that the territory under study was contaminated with  137 Cs as a result of the Chornobyl accident with the activity from 2 to 10 kBq/m². The total contamination is mainly represented by density in the range from 1 to 5 kBq/m². A small spot near the village of Khorovytsia is obviously of Chornobyl origin.

With that level of ¹³⁷Cs contamination, there are no restrictions on agriculture.

The results for  40 K in the soil are shown in Figure 5.6.



Figure 5.6 – Specific activity of  40 K in the soils in the KhNPP proximity (sampling depth – 0.2 m)

This radionuclide is of natural origin and is one of the main natural radionuclides in soils, plants and agricultural production facilities. Potassium radioactivity in soils is primarily determined by its content in the parent rock. The highest concentration is observed in the soil finely dispersed clay fraction. The results showed that the specific activity of the radionuclide in the soils within the studied area ranges from 40 to 700 Bq/kg. The minimum values are typical for light mechanical composition soils (sodpodzolic sandy, sandy loam). The  40 K increases with the burden of mechanical composition.

The results for ²³²Th and its decay products are shown in Figure 5.7. The specific activity of that radionuclide for the territory under consideration varies from 5 to 50 Bq/kg. The wide range of changes in specific activity is explained by the great diversity of soil cover within the investigated area.

Similar to ⁴⁰K, ²³²Th in soil is determined by the parent rock, and increases with the burden of soil mechanical composition.



# Figure 5.7 – Specific activity of 232 Th in soils within the KhNPP proximity (sampling depth – 0.2 m)

Figure 5.8 shows the content of ²²⁶Ra, which is a derivative of ²³⁸U, in the soils of the area under study. This nuclide and its derivative radionuclides (primarily ²²²Rn) are of important radiological importance. The main source of this radionuclide in the biosphere is the earth's crust. Clarke number in soils is about 30 Bq/kg. The content of that radionuclide in the topsoil in the KhNPP adjacent territory varies from 3 to 40 Bq/kg. Therefore, the radiological situation in the KhNPP area is currently determined mainly by radionuclides of natural origin.



Figure 5.8 – Specific activity of  226 Ra in soils within the KhNPP near area (sampling depth – 0.2 m)

As a result of the mechanical impact associated with any construction, the soil cover in the allocated area is almost fully destroyed. However, the area of mechanical impact during the KhNPP construction is clearly delineated by the agreed allocated boundaries.

An important factor affecting the soils is a technogenic contamination of the soil cover with chemical pollutants, including radionuclides. However, being a complex, open and dynamic system, soils have a certain resistance to technogenic load – buffering capacity. Buffering capacity depends on physical and chemical properties of soils, their mechanical composition, conditions of economic use, etc.

The following can be noted based on the abovementioned factors:

- due to the diversity of mechanical composition (from sandy to heavy loam), various degrees of removability, gleying and waterlogging, and a variety of soil-forming rocks, the soil cover of the 30-km area is diverse both in terms of types and area. Gray forest and sod-podzolic soils cover the largest areas in almost all sectors.

- automorphic soils cover the largest areas in the northeastern and northwestern sectors, hydromorphic and semi-hydromorphic soils are found throughout the area, but they cover the largest areas in the eastern (semi-hydromorphic), southeastern (semi-hydromorphic), southeastern (semi-hydromorphic) sectors.

– marsh and meadow-marsh (hydromorphic) soils (approx. 10% of the territory) are critical in terms of the cesium isotopes migration mobility. As for the automorphic soils, the most critical for some radionuclides (e.g., 90Sr) are sod-podzolic sandy soils with the lowest porosity, moisture capacity and maximum hygroscopicity (approx. 20% of the territory).

– considering negligible additional environmental radionuclide contamination during normal plant operation (maximum values for  $Cs^{137}$  - n  $10^{-1}$  Bq/m²), special agrotechnical measures with changes in the agricultural land use structure, re-profiling of agricultural sectors and changes in technological processing of products are not appropriate.

- the surface relief of the area close to the plant and the orographic barriers are included when considering transboundary radionuclide transfer in case of accidents.

- under MDBA and BDBA, the impact will be regional, but neither the physical, chemical nor the water-physical soil properties will change; only the content of chemical elements will change.

- in general, the analysis of physical and chemical properties of the region's soils showed that, despite the significant soil cover diversity, most soils have a significant buffer resistance reserve to technogenic loads. The KhNPP construction and operation with 6 units will not lead to the exhaustion of this reserve. The landscapes of the area close to the plant are a reliable barrier to the expansion of the primary contamination area through migration.

The results of long-term observations over radionuclide in the KhNPP supervised area confirm the soil buffering and the absence of significant radiation effects on the soils.

Table 5.23 below shows the data on soil contamination in the KhNPP supervised area for 2023.

Table 5.23 – Soil contamination in the KhNPP supervised area for 2023, kBq/m²

	SPA		SPA - 10 km		10 - 20 km		>20 km	
Radionuclide	Average	Maximum	Average	Maximum	Average	Maximum	Average	Maximum
	value	value	value	value	value	value	value	value
Cs-137	6,92E-02	1,11E-01	7,20E-02	2,64E-01	8,27E-02	1,69E-01	1,60E-01	-
Cs-134	2,63E-03*	_	4,92E-03*	-	3,04E-03*	-	5,60E-03*	-
Co-60	2,67E-03*	-	3,11E-03*	-	3,22E-03*	-	5,62E-03*	-
I-131	3,09E-03*	-	5,56E-03*	-	3,41E-03*	-	6,47E-03*	-
Sr-90	2,36E-01	-	3,35E-01	3,44E-01	2,94E-01	3,02E-01	1,65E-01	-

**Note*:** values corresponding to ½ minimum measured activity are marked.

According to the procedure [51] soil contamination is determined once a year.

In 2023, as well as for the entire monitoring period, no cases of exceeding the levels of radioactive substances in environmental objects established by regulatory documents were registered. The volumetric activity of the technogenic cesium-137 isotope in the air did not exceed 4.35  $\mu$ Bq/m³, which is approx. 180 thousand times lower than the permissible value of cesium-137 in the air for the population established by the regulations [22] – 0.8 Bq/m³. The volumetric activity of technogenic cesium-137 isotope in water did not exceed 25.3 Bq/m³, which is almost 4 thousand times lower than the permissible value of cesium-137 in drinking water for the population established by the regulations [22] – 100 000 Bq/m³. Soil contamination with cesium-137 in the KhNPP area did not exceed 264 Bq/m².

#### 5.3.7.3 Impact on flora and fauna

The operation of two additional AP1000 units, including the completion of units 3 and 4, will not affect the structure and dynamics of plant species in general, and will not cause changes in the number of rare and Red Book plant species. The radiation situation within the KhNPP area is currently determined mainly by radionuclides of natural origin. The density of ¹³⁷Cs contamination of the territory varies in the range of 3-7 kBq/m2. At this level of ¹³⁷Cs contamination, the flora does not feel any impact and there are no restrictions on nature management.

In the considered emergency situations, the dosed loads on the flora outside the SPA will not lead to changes in its structure and other negative changes. The commissioning of units 5 and 6 and accident-free KhNPP operation as a six-unit plant (including planned units 3 and 4) will not adversely affect the overall species diversity of invertebrates and insects.

#### 5.3.7.4 Technogenic environment

After the commissioning of units 5 and 6, no new technological processes with air emissions of any pollutants other than the existing ones will be introduced at the NPP, i.e. there is no negative impact on the SA facilities.

#### 5.3.7.5 Water bodies impact assessment

No industrial, rain and domestic wastewater from the NPP and domestic effluents from Netishyn are discharged directly into public water bodies. The treated wastewater is discharged into the cooling pond, which is a water body for NPP production use, and thus compensates for water losses in the cooling pond. Given that cooling towers are envisaged for AP1000 reactors, the amount and quality of pollutants entering the cooling pond will be minimal. Moreover, it is envisaged that the parameters of their composition and condition are continuously monitored throughout the entire process media management.

The treated wastewater from the NPP industrial site and Netishyn is not discharged directly into surface water bodies and thus does not have a direct impact on their contamination degree.

When discharging to external water bodies, the radioactive discharge is calculated using the following formula:

$$K_{PHB} = \sum_{i}^{n} \frac{q_i}{Q_{iMC}} \times 100\%$$

where  $q_i$  is the actual total intake of the i-radionuclide into external water bodies at the end of the reported quarter since the beginning of the year, Bq, which, in turn, is calculated by the following formula:

$$q_i = \frac{4}{k} \sum_{j=1}^k c_i^j$$

Where k is the quarter number in which the monitoring is performed (1, 2, 3, 4);  $c^{j}_{i}$  is the actual discharge of the i- radionuclide for the j- quarter;  $Q_{iMS}$  is a discharge limit (permissible discharge) of the i- radionuclide into external water bodies per year, Bq/year.

To assess and compare the discharges from the existing KhNPP units [49] with the planned discharges according to [50], the data on radioactive discharges to the environment (external water bodies) by KhNPP for 2016-2023 (Table 5.24) and expected AP1000 units discharges (Table 5.25) are presented below.

Table 5.24 – Radioactive discharges into the environment (external water bodies) by the KhNPP units for 2016-2023 (values greater than minimum measured activity were used for calculations)

Period,	Radioactive discharges – radionuclide activity, MBq								Discharge
year	¹³⁷ Cs	¹³⁴ Cs	⁶⁰ Co	⁵⁸ Co	⁵⁴ Mn	⁹⁰ Sr	^{110m} Ag	³ H	rate, %
2016	11,01	3,85	-	-	-	2,54	-	2292245	0,16
2017	24,70	15,19	-	-	-	1,27	-	1368952	0,16
2018	11,81	4,27	-	-	-	1,91	-	1979921	0,146
2019	23,70	4,24	-	-	-	1,45	-	2820000	0,214
2020	9,12	1,47	-	-	-	0,72	-	1920000	0,125
2021	8,41	0,87	-	-	-	0,64	0,07	2260000	0,139
2022	6,32	1,33	0,275	-	-	0,68	-	1850000	0,143
2023	7,65	1,21	-	-	-	0,56	-	1040000	0,102

	Activity release (MBq/year) to discharge canal (Note)								
Nuclide	Shim bleed and equip.	Miscellaneous	Turbine building	Total release					
C-14	7 00E-02	neolioihle	negligible	7.00E-02					
Na-24	3 50E+01	2 30E-01	2.80E+00	3 80E+01					
Cl-36	negligible	negligible	negligible	negligible					
Cr-51	4 50E+01	1.30E-01	2.80E-01	4.60E+01					
Mn-54	3.20E+01	7.20E-02	1.40E-01	3.20E+01					
Fe-55	4 80E+02	1.10E+00	2.10E+00	4.90E+02					
Fe-59	4 90E+02	negligible	negligible	5.00E+00					
Co-58	4.10E+02	1.00E+00	2.00E+00	4.10E+02					
Co-60	2.20E+02	5.00E-01	9.40E-01	2.30E+02					
Ni-63	5.30E+02	1.20E+00	2.10E+00	5.40E+02					
Zn-65	1.00E+01	negligible	4.50E-02	1.00E+01					
Nb-94	negligible	negligible	negligible	negligible					
W-187	2.80E+00	negligible	1.70E-01	3.00E+00					
U-234	negligible	negligible	negligible	negligible					
U-235	negligible	negligible	negligible	negligible					
U-238	negligible	negligible	negligible	negligible					
Np-237	negligible	negligible	negligible	negligible					
Pu-238	negligible	negligible	negligible	negligible					
Pu-239	negligible	negligible	negligible	negligible					
Pu-240	negligible	negligible	negligible	negligible					
Pu-241	8.00E-02	negligible	negligible	8.00E-02					
Pu-242	negligible	negligible	negligible	negligible					
Am-241	negligible	negligible	negligible	negligible					
Am-243	negligible	negligible	negligible	negligible					
Cm-242	negligible	negligible	negligible	negligible					
Cm-244	negligible	negligible	negligible	negligible					
As-76	negligible	negligible	negligible	negligible					
Br-82	negligible	negligible	negligible	negligible					
Rb-86	negligible	negligible	negligible	negligible					
Rb-88	3.90E-01	negligible	negligible	3.90E-01					
Sr-89	2.40E+00	negligible	negligible	2.40E+00					
Sr-90	2.50E-01	negligible	negligible	2.50E-01					
Y-91	9.00E-02	negligible	negligible	9.10E-02					
Zr-95	6.80E+00	negligible	negligible	6.90E+00					
Nb-95	6.10E+00	negligible	negligible	6.10E+00					
Mo-99	1.90E+01	1.10E-01	5.30E-01	1.90E+01					
Tc-99m	1.80E+01	1.10E-01	3.80E-01	1.80E+01					
Tc-99	negligible	negligible	negligible	negligible					
Ru-103	1.20E+02	3.10E-01	6.60E-01	1.20E+02					
Ru-106	negligible	negligible	negligible	negligible					
Ag-110m	2.60E+01	5.80E-02	1.10E-01	2.60E+01					

Table 5.25 – Liquid discharge during normal AP1000 plant operation and anticipated operational occurrences

	Activity release (MBq/year) to discharge canal (Note)							
Nuclide	Shim bleed and equip. drains	Miscellaneous waste	Turbine building	Total release				
Sn-117m	negligible	negligible	negligible	negligible				
Sb-122	negligible	negligible	negligible	negligible				
Sb-124	negligible	negligible	negligible	negligible				
Sb-125	negligible	negligible	negligible	negligible				
I-129	negligible	negligible	negligible	negligible				
I-131	1.50E+01	6.30E-02	2.50E-01	1.50E+01				
I-132	1.90E+01	9.10E-02	8.50E-01	2.00E+01				
I-133	2.60E+01	1.70E-01	2.70E+00	2.90E+01				
I-134	5.80E+00	3.90E-02	negligible	5.90E+00				
Cs-134	7.50E+00	negligible	negligible	7.60E+00				
I-135	2.00E+01	1.30E-01	3.20E+00	2.40E+01				
Cs-136	9.20E+00	negligible	8.50E-02	9.30E+00				
Cs-137	2.30E+01	5.00E-02	1.10E-01	2.30E+01				
Ba-140	1.30E+01	4.60E-02	1.10E-01	1.40E+01				
La-140	1.80E+01	6.60E-02	2.00E-01	1.80E+01				
Ce-144	7.90E+01	1.80E-01	3.40E-01	8.00E+01				
Pr-144	7.90E+01	1.80E-01	3.40E-01	8.00E+01				
All others	negligible	negligible	negligible	negligible				
Total	2.40E+03	6.30E+00	2.10E+01	2.50E+03				

Note: values less than 3.7E-5 GBq/year are considered to be negligible.

Comparing the planned AP1000 units releases with the actual KhNPP two units releases, it can be concluded that the impact on water bodies during AP1000 plant operation will be at the level of existing area background values and will not lead to an increase in the radiation background and will not exceed the control levels.

#### 5.3.8 Transboundary transfer

Mathematical modeling of dispersion of gas and aerosol radioactive releases performs assessment of transboundary transfer of radioactive emissions in case of a radiation accident at Khmelnytskyy NPP during normal operation of the plant and emergencies, dose loads assessment on the population using the spatial contamination field.

Radioactive transfer modeling was performed using JRODOS (JRodosServerFebruary2019 version). Diffusion model – DIPCOT.

The meteorological conditions of emissions propagation in the atmosphere play a key role in the field formation of the air radioactive contamination and the underlying surface. Since for this problem the period for the release from Khmelnytskyy NPP to reach the borders, for example, with Poland, is about half a day, the temporal dynamics of meteorological parameters due to both the daily variation of the boundary layer characteristics and weather changes on a synoptic scale plays an important role for such periods. Therefore, the most reasonable approach to selecting meteorological scenarios for radioactive release propagation in the atmosphere is not to construct artificial "ultraconservative" scenarios but to use real data of atmospheric characteristics measurements.

A conservative meteorological forecast based on the results of long-term observations of the meteorological post of the KhNPP branch was used to assess the effects of transboundary transfer:

- wind direction corresponds to the shortest distance to the neighboring countries Poland, Slovakia, Romania, Hungary, and Moldova during the entire emission period;

- wind speed of 1, 2, 3, 5, 10 m/s during the entire emission period (since, based on long-term metrological observations, the average daily wind speed near KhNPP is in the range of 0.8-5 m/s with wind gusts up to 20 m/s);

– atmospheric stability category D.

The height of the reactor vessel pipe of the AP1000 unit, which is planned to be constructed using Westinghouse technology, is approximately 72.500 m from 0.000.

For the calculation, the total emission value corresponding to the maximum design basis accident according to the document [52] and shown in Table 5.26 was taken.

Table 5.26 – Expected total radionuclide composition emission to the environment under the maximum design basis accident (Bq)

Nuclide	0-2 hours	2-8 hours	8-24 hours	24-96 hours	96-720 hour
Xe-133	2.17E+14	1.173E+15	4.33E+15	7.69E+15	1.52E+16
Xe-133m	1.19E+12	9.41E+12	2.44E+13	4.94E+13	2.01E+14
Cs-137	3.03E+12	8.13E+12	5.07E+11	2.59E+09	2.40E+10
I-131	3.34E+13	9.27E+13	9.50E+12	6.24E+12	1.87E+13
Te-131m	7.48E+11	2.32E+12	1.26E+11	2.22E+08	7.40E+07
SR-90	1.37E+11	4.51E+11	2.82E+10	1.48E+08	1.33E+09
Ru-103	2.96E+11	9.71E+11	6.03E+10	2.96E+08	2.15E+09
La-140	2.84E+10	8.94E+10	5.03E+09	1.48E+07	3.70E+06
Ce-141	6.54E+10	2.14E+11	1.33E+10	7.03E+07	4.48E+08
Ba-140	2.76E+12	9.01E+12	5.54E+11	2.96E+09	1.15E+10

Distribution of iodine isotopes by fractions according to the document [52]:

- aerosol form -95%;
- molecular form -4,85%;
- organic form -0,15%.

Google Maps determined the minimum distances to neighboring countries (Figure 5.9). The distance from Khmelnytskyy NPP to the border of neighboring states is shown in Table 5.27.



Figure 5.9 – Minimum distances from Khmelnytskyy NPP to the borders of neighboring countries



Figures 5.10-5.14 show the effective radiation dose distribution map for the neighboring states listed in Table 5.27.


Figure 5.10 – Effective dose map distribution in the direction of Poland



Figure 5.11 – Effective dose distribution map in the direction of Slovakia



Figure 5.12 – Effective dose distribution map in the direction of Hungary



Figure 5.13 – Effective dose distribution map in the direction of Romania



Figure 5.14 – Effective dose distribution map in the direction of Moldova

Section 3.3 of this EIA report provides information on the soil characteristics, the composition of which ensures low mobility and almost complete fixation of heavy metals and radionuclides. Thus, the contamination migration to the neighboring countries is virtually impossible. In addition, when assessing the impact on fauna, according to paragraph 4.3 of this EIA report, it is noted that the releases impact from the six NPP units under normal operating conditions on forestry, mushroom and berry lands, as well as on recreational activities and the quality of forest products is practically absent. Based on the results of the calculations and analysis of the obtained results, it can be concluded that the maximum annual effective dose in the territory of neighboring states at the maximum design-basis accident "from all pathways" is 0.153 mSv, "from all pathways except consumption" is 0.0323 mSv, which is significantly less than the permissible dose limit for the population – 1 mSv under [22]. Thus, considering the peculiarities of the natural and anthropogenic character of the area, the impact on the neighboring countries' territories is practically absent.

It is also worth mentioning that the probability of an accident is extremely low, and the probability of a simultaneous accident at all power units at one site is even more so. Thus, all calculations are usually performed for a single source of a possible accidental release. Moreover, the software tools (RODOS) used to calculate transboundary transfers of releases allow entering data from a specific source.

#### 5.3.9 Calculation of the sanitary protection zone and supervised area

Following the Law of Ukraine on Nuclear Energy Use and Radiation Safety [53], a sanitary protection zone and a supervised area shall be established at the locations of a

nuclear facility or a facility intended for radwaste handling. The size and boundaries of these zones shall be determined in the project by the norms and rules in the field of nuclear energy use and shall be agreed with the state nuclear and radiation safety regulatory authority. Radiation monitoring shall be performed in the sanitary protection zone and supervised area.

## 5.3.9.1 Calculation of the Sanitary Protection Zone

To determine the size of the Sanitary Protection Zone of the AP1000 units, relevant model calculations were performed [54].

Such calculations to determine the size and boundaries of the Khmelnytskyy NPP SPZ were performed following the requirements of the document ISO NNEGC 023:2014[55], which was agreed with the SNRIU and the Ministry of Health of Ukraine. Besides, the impact of operating KhNPP units 1, 2, planned units 3, and 4 with WWER-1000 reactors and planned units 5, and 6 with AP1000 reactors is considered. The size of the sanitary protection zone of KhNPP was determined based on the conditions of not exceeding the established dose criteria for public exposure for:

– normal operation of power units;

- violations of the normal operation of power units with WWER-1000 and AP1000 reactor units;

- design basis accidents at power units with WWER-1000 and AP1000 reactors.

In accordance with [22], the criteria for determining the NPP SPZ size are to follow the non-exceedance the certain radiation levels within and outside the SPZ. The specified exposure levels are set separately for normal operation, as well as for design basis accidents and abnormal operation.

In particular, during normal operation, the dose limit quota for NPP gas and aerosol emissions established by NRBU-97 should not be exceeded, namely, the annual effective dose to the public due to all pathways of dose formation outside the SPZ should not exceed 40  $\mu$ Sv. Non-exceedance of the specified dose shall be ensured for all reference ages specified in NRBU-97 and for all distances equal to or exceeding the distance from the places of possible releases to the SPZ boundaries.

In case of abnormal operation, effective doses to the public outside the SPZ should not exceed 40  $\mu Sv$  for 120 days after such an event.

In the event of design basis accidents, the doses to the public within and outside the SPZ should not exceed the values of unconditional justification levels given in Table D.7.1 of NRBU-97, namely:

a) effective dose:

- children 10 mSv;
- adults 20 mSv;

b) thyroid dose:

– children - 100 mSv;

- adults - 300 mSv;

c) skin dose:

- children - 300 mSv;

- adults - 500 mSv.

Non-exceedance of radiation doses in case of design basis accidents and abnormal operation should be ensured:

- for all reference ages specified in NRBU-97;

- for all distances equal to or exceeding the distance from the places of possible releases to the SPZ boundaries.

According to the results of the calculations, the dimensions of the KhNPP SPZ, which ensure that the dose criteria for public exposure following the requirements of the SOU NNEGC 023:2014 [55], NRBU-97 [22], are not exceeded within and outside its boundary, are shown in Figure 5.15. The calculated boundary of the SPZ is an ellipse with the center in the middle of the segment connecting the locations of the vent pipes of Units 2 and 3. The major axis of the ellipse is on a line

connecting the locations of the vent pipes of power units 1 and 6. The major semi-axis length of the ellipse is 2600 m, and the minor semi-axis one is 2200 m.

The KhNPP SPZ boundaries, defined as a result of the calculation, are almost fully consistent with the existing boundaries (Fig. 5.15), except for its western part. In this area, the calculated SPZ boundary extends beyond the existing one in the western direction by no more than 250 m. At the same time, most of this additional area falls on the water surface of the KhNPP pond. Such a small increase in the calculated SPZ size in the western direction is explained by taking into account the potential impacts of new units 5 and 6. The area of the calculated SPZ is 18.0 km², or about 78% of the existing one.

Before construction activities and after the approval of the construction project for KhNPP Units 5 and 6 with the AP1000 reactor, the size of the SPZ will be finally determined based on the results of the Project's expert review.



Figure 5.15 – The KhNPP SPZ boundaries are defined as a result of calculations based on the requirements of the SOU NNEGC 023:2014 [55] (in red). The existing KhNPP SPZ boundaries are shown in a purple circle with a radius of 2700 m.

#### 5.3.9.2 Calculation of the supervised area

To determine the size of the AP1000 units SA, the corresponding model calculations were performed [54].

Such calculations of the SA were performed based on the requirements of the NP 306.2.173-2011 "Requirements for determining the size and boundaries of the NPP supervised area" [60]. At the same time, the calculation of all functions that are compared with the established criteria for determining the NPP SA size was performed taking into account the potential impact of all sources of radioactive releases at the KhNPP site. This means that modeling of consequences of beyond-design-basis accidents, which determines the KhNPP SA size, was performed for each of the six (operating and planned) KhNPP units, namely:

a) operating units 1 and 2 with VVER-1000/V-320 reactors;

b) units 3 and 4 with VVER-1000/466B reactors planned for design and construction;

c) units 5 No. 6 planned for design and construction with a unit electric capacity of about 1100 MW and a thermal capacity of 3400 MW with AP1000 reactor type manufactured by Westinghouse Electric Company.

The SA size is determined so that in case of beyond-design basis accidents, the frequency of which is equal to or exceeds the values of the indicators established by the "General Safety Provisions for Nuclear Power Plants" [6], the doses to the public on the SA border and beyond do not exceed the criteria for the introduction of urgent countermeasures (lower limits of justification) - evacuation and iodine prophylaxis, given in Table D.7.1 of NRBU 97¹ [22]), namely:

a) effective dose -50 mSv;

b) thyroid dose:

- children - 50 mSv;

- adults - 200 mSv;

c) skin dose – 500 mSv.

Non-exceedance of these doses was demonstrated:

- for all reference ages defined in NRBU-97;

- for all distances equal to or exceeding the distance from the places of possible releases to the SA boundaries.

When calculating the non-exceedance of the specified radiation doses, the types of terrain in the SA and the likely meteorological conditions during the atmospheric transfer of an accidental release are taken into account.

For the considered beyond-design-basis accident scenarios at the VVER-1000 and AP1000 power units, the formation of the following set of circles is required to meet the criteria for not exceeding the effective dose to the public and equivalent skin and thyroid doses:



The combination of circles is shown in Figure 5.16.

[1]⁾ The values of justification limits for doses to the thyroid gland and skin in Table D.7.1 of NRBU-97 are given in terms of absorbed doses (mGy). The numerical values of absorbed doses (mGy) formed by beta and gamma radiation only are equal to the values of equivalent doses (mSv).



Figure 5.16 – The KhNPP SA boundaries are determined by the calculations in accordance with the requirements of NP 306.2.173-2011 (in red). The boundary of the existing KhNPP SA is shown in a purple circle with a 30 km radius

The Figure 5.16 shows the above four circles in blue for each of the VVER-1000 units 1-4. Two circles with a radius of 7.6 km (shown in brown in Fig. 5.16), corresponding to the calculations for units 5-6 with AP1000 reactors, are shifted from each other by 270 m (distance between the units' vent stacks) and are completely absorbed by the other 4 circles with a radius of 13.1 km obtained for units 1-4. If we assume that the center of the supervised area should remain unchanged, then the circle radius drawn around this center and covering all six circles is 13600 m (shown in red in Fig. 5.16). This value can be taken as the size of the supervised area resulting from the calculations.

The specified size of the KhNPP supervised area, based on the document NP 306.2.173-2011, ensures that the established dose criteria are not exceeded for all reference age groups and in the entire range of meteorological conditions for the

dissemination of radioactive releases in the atmosphere and characteristics of the underlying surface.

Prior to the commence of construction activities and after the approval of construction project for the KhNPP units 5 and 6 with the AP1000 reactor, the SA size will be finalized based on the project review results.

## 5.3.10 Waste treatment

In the course of NPP operation, both radioactive waste and non-radioactive waste are inevitably generated, see also Sections 1.5.1, 1.5.2.3.

Radwaste includes such material objects and substances that are not subject to further use, and whose radionuclide activity or radioactive contamination exceeds the levels established by applicable regulations.

NPP operation inevitably generates liquid, solid, and gaseous radioactive waste. The sources of radioactive waste are NPP equipment and systems containing liquid, solid, and gaseous media.

Liquid radioactive waste is generated during the operation and repair of the main and auxiliary equipment of the reactor unit, and plant-wide systems containing radioactive media or contaminated with radioactive substances, see also Section 1.5.1.4.

Solid radioactive waste is spent reactor internals, pump parts, pipelines, fittings, thermal insulation, ventilation system filters, wiping materials, cotton and film overalls, used PPE, various construction waste, etc., see 1.5.1.5.

Gaseous radioactive waste - volatile emissions (aerosols) of the primary coolant occurred during repair work, small leaks, and disorganized coolant leaks, see also 1.5.1.3.

The AP1000 power unit provides a radioactive waste treatment and management system to support the expected waste generation from plant operation and provides the flexibility and space to install mobile systems during emergencies and special waste treatment systems that may be required in the event of such situations. Below is a list of radioactive waste treatment systems for the AP1000:

- Gaseous Radwaste system (WGS);
- Liquid Radwaste system (WLS);
- Radioactive Waste Draine System (WRS);
- Solid Radwaste System (WSS).

The operating organization is currently implementing a project for the construction of the KhNPP Radioactive Waste Processing Complex (RWPC) at the Khmelnytskyy NPP site. The RWPC is intended for processing radwaste already accumulated in KhNPP storage facilities those generated during the current operation of power units and those that will be generated during the decommissioning of KhNPP power units.

The construction of the RWPS at the KhNPP site will ensure

- increasing the level of personnel protection, the public and the environment from negative radiation impact;

- improvement of the radwaste management system at KhNPP, as well as reduction of the probability of radiation accidents during SRW handling;

– preparation of processed SRW for further shipment to specialized enterprises for disposal following the established acceptance criteria;

- significant reduction of the amount of SRW stored at the KhNPP site due to the introduction of new technologies for their processing.

The main RWPC facilities:

- organic fuel combustion facility;
- pressing facility;
- metal decontamination facility;
- activity measurement facility;
- waste sorting and fragmentation facility;
- cementation facility;
- facility for exemption from regulatory control.

The commissioning of the RWPC in 2025 will ensure the availability of free radwaste storage facilities for the lifetime extension and decommissioning of the operating KhNPP units, storage, and processing of SRW from the new KhNPP units 3, 4 and units using the AP1000 technology. The commissioning of the RWPC will generally improve the safety culture in radwaste management at the KhNPP site.

#### 5.3.10.1 Sources of gaseous radwaste generation

The Radwaste Gas System (WGS) is designed to accept, process, and discharge radioactive waste gases off-site in compliance with the threshold limit value for normal plant operation, including power generation, fuel shutdowns, and reloading.

The AP1000 reactor is designed to accept all potentially hydrogen-containing and radioactive gases generated during the plant operation. The largest amount of gaseous waste is generated during plant operation at the end of the fuel cycle. Gaseous radioactive waste includes:

– gas exhausts, non-condensable gases, and volatile aerosols from equipment operating on radioactive media;

air that is removed from workspaces.

The gases accumulated in the primary circuit during operation are vented from it. This results in the formation of a LRW stream. Gaseous emissions can also be generated by venting volatile emissions from the primary coolant resulting from small, organized, unorganized leaks.

Before being released into the atmosphere, gaseous waste is subject to treatment.

#### 5.3.10.2 Sources of liquid radwaste generation

The WLS system is designed to manage, collect, treat, handle, store, and dispose of liquid radwaste generated during normal operation.

The system contains components that accept and store radioactive or potentially radioactive liquid waste.

The system's main equipment is the reactor primary coolant drainage tank (RCDT), containment pit, wastewater settling tanks, waste holding tanks, etc.

The system components store and treat waste during normal operation and expected disruptions of normal operation.

The main sources of liquid radioactive waste generation at the AP1000 facilities are:

– primary circuit coolant;

- decontamination water generated during decontamination activities on NPP equipment

organized and unorganized leaks from equipment containing liquid radioactive media;

- steam generator blowdown water;
- drainage water coming from the Radioactive Waste Drain System (WRS);
- other.

Before liquid waste is discharged and throughout its management, it is mandatory to monitor and control the compliance of radiation parameters of the process medium.

5.3.10.3 Sources of solid radwaste

The main solid radwaste from AP1000 reactors for which the system is designed can be identified as follows:

- spent ion exchange resins;
- spent filter cartridges;

- dry and mixed operational waste generated as a result of the operation of NPP equipment containing radioactive contamination;

– other.

The Solid Radwaste System collects spent ionizers, mixed-action filters, spent filter elements, and dry (solid) and mixed radioactive waste generated during NPP operation in normal operation and in case of expected violations of normal operation. The system is located in the auxiliary and the radioactive waste management buildings.

The mobile systems at the loading area located in the auxiliary building and mobile facilities in the radioactive waste management building are responsible for radioactive waste processing and packaging.

The packaged radwaste is stored in the annex buildings, auxiliary and RAW management buildings before the disposal to an appropriate storage facility.

Solid radwastes are accepted before being processed and packaged by mobile equipment for shipment off-site.

# 5.4 Risks to human health, cultural heritage sites, and the environment, including through the possibility of emergencies

Consideration of risks to human health, cultural heritage, and the environment is one of the key aspects of developing an EIA report.

Environmental risk is the likelihood of an event with adverse consequences for the environment and population health caused by the predicted negative impact of economic and other activities that pose a threat of natural or man-made emergencies.

## 5.4.1 Risks to population health and the environment

Risks to human health and the environment are assessed considering both radiation and non-radiation effects. The risks of non-carcinogenic effects are also accounted for.

The SPZ is organized around any NPP following the Ukrainian radiation safety legislation. In the SPZ, the population is prohibited from living, and restrictions are imposed on production activities that are not related to the nuclear facility and where radiation monitoring is carried out. Thus, neither exports nor agricultural activities are allowed in this area.

In addition, a supervised area is organized around the NPP, which is larger than the Exclusion Zone, and where radiation monitoring is also conducted for various indicators. The scope and frequency of such monitoring is determined by the SNRIU-approved NPP radiation monitoring regulations.

## 5.4.1.1 Assessment of public dose from natural radiation sources

The existing air pollution is well below the permissible concentrations.

Table 5.28 shows the content of radionuclides in surface waters of water bodies in 2021 [35], as well as the permissible concentration of radionuclides in potable water [23].

Radionuclide	Permissible concentrations in potable water	Cooling pond	Horyn River ( to NPP)	Horyn River (control gate)
¹³⁷ Cs	1·10 ⁵	9,59	11,100	9,280
$^{134}Cs$	$7.10^{4}$	0,529	0,388	0,538
⁶⁰ Co	$8.10^{4}$	0,566	0,407	0,594
⁹⁰ Sr	$1.10^{4}$	6,91	5,010	4,280
³ H	$3.10^{7}$	6,86·10 ⁴	$4,25 \cdot 10^3$	8,30·10 ³

Table 5.28 – Radionuclide content in surface waters of water bodies, Bq/m³

The table shows that the concentration of radionuclides in surface water bodies is significantly below the permissible levels.

Based on the data from KhNPP, the content of ¹³⁷Cs and 90Sr in staple foods [56] was assessed in comparison with the permissible values of ¹³⁷Cs and ⁹⁰Sr [57] in these foods, shown in Table 5.29.

Table 5.29 – Average values of  $^{137}\mathrm{Cs}$  and  $^{90}\mathrm{Sr}$  concentration in basic food products, Bq/kg

Sample	Radionuclide con	centration values	Normative indicator of permissible lev		
Sample	⁹⁰ Sr, Bq/kg	¹³⁷ Cs, Bq/kg	⁹⁰ Sr, Bq/kg	¹³⁷ Cs, Bq/kg	
Wheat	0,300 - 0,673	0,059 - 0,777	20	50	

Comple	Radionuclide con	centration values	Normative indicator of permissible levels		
Sample	⁹⁰ Sr, Bq/kg	¹³⁷ Cs, Bq/kg	⁹⁰ Sr, Bq/kg	¹³⁷ Cs, Bq/kg	
Rye	0,451 - 1,260	0,085 - 0,133	20	50	
Milk	0,21	0,83	20	100	
Fish	0,49	0,61	35	150	
Potato	0,059 - 0,851	0,041 - 0,340	20	60	
Cabbage	0,370	0,133	20	60	

Natural radionuclides and their decay products, as well as cosmic radiation make the main contribution to the formation of dose loads on the population of the KhNPP monitoring area. At the same time, artificial radionuclides of global fallout and radionuclides of Chornobyl origin make a significantly smaller contribution to the radiation dose.

Based on the analysis, construction and accident-free commissioning of the new KhNPP units will not cause a significant increase in radionuclides in surface water and basic food products. In emergencies, no significant impact on the population and the environment outside the SPZ is expected, see also. 5.3.7 and 5.3.9.

5.4.1.2 Assessment of risks from radiation exposure to the public and the environment

According to [22], the concept of risk is introduced for both stochastic and deterministic effects.

The individual (r) and collective (R) risk of stochastic effects from exposure is determined accordingly:

$$r = r_E \cdot E, \tag{5.1}$$
$$R = r_E \cdot S_E \tag{5.2}$$

 $E, S_E$  - individual and collective effective doses;

 $r_E$  - risk factor for fatal and non-fatal cancer and serious hereditary effects.

The risk coefficient per unit of individual or collective dose for the population is assumed to be equal to  $_{rE} = 7.3 \cdot 10^{-2} \text{ Sv}^{-1}$ .

When exposed to doses that cause deterministic effects, it is assumed that the risk of severe consequences is equal to the probability of the consequence itself:

$$r = p(E)$$
(5.3)  

$$R = p(S_E) \cdot N$$
(5.4)

p(E),  $p(S_E)$  is the probability of events that generate doses *E* and S_E, respectively; *N* is the number of the population exposed to radiation with equivalent doses E > 0.5 Sv.

#### Risk of deterministic effects

An accident at the AP1000 power unit is considered to be a beyond-design-basis loss of coolant accident (LOCA) with subsequent core meltdown. Although the analysis of the core reaction in the AP1000 reactor during a loss of coolant accident shows that its integrity is preserved, it is assumed that significant core degradation and meltdown occurs to assess the radiological consequences of the accident.

The radionuclide composition and activity of each radionuclide in the release to the environment in a beyond design basis accident was set in accordance with [81] (Table 5.30). The value of the effective release height was conservatively assumed  $_{heff} = 1$  m.

Nuclide	Activity (Bq)
Xe-133	2,86E+16
Xe-133m	2,85E+14
Cs-137	1,17E+13
I-131	1,61E+14
Te-131m	3,19E+12
Sr-90	6,18E+11
Ru-103	1,33E+12
La-140	1,23E+11
Ce-141	2,93E+11
Ba-140	1,23E+13

Table 5.30 – The expected release of activity into the environment in case of an beyond design basis accident at the AP1000 power unit

The ratio of activity in the release between different physico-chemical forms of ¹³¹I was set according to [82] to 95% (aerosol): 4.85% (molecular) : 0.15% (organic).

The values of the integrated volumetric specific activity and surface specific activity of radionuclides were calculated using a standard Gaussian model of the distribution of impurities in the atmosphere (its description is given in [60]). The obtained values were used to estimate the effective and equivalent doses to the public for the period of 14 days after the accident. The effective dose to the public was calculated for each of the 6 reference ages defined in [22], including the following dose formation ways: inhalation of radionuclides, external exposure from the release cloud and external exposure from radionuclides falling on the soil surface. Equivalent doses to the thyroid gland and skin were calculated in a similar manner.



Figure 5.17 – The estimated values of equivalent doses to the thyroid gland (1) and skin (2); 3 - dose criterion of 500 mSv [22].

Figure 5.17 shows the results of calculations of the equivalent dose to the thyroid gland and skin for the critical age group "1 year" and the "worst" meteorological conditions of release propagation (wind speed of 1 m/s, atmospheric stability category F, no precipitation). The value of the roughness parameter of the underlying surface was set to 1 m in the modeling. Non-exceedance of the dose criterion of 0.5 Sv, established in [22], is ensured at distances of 750 m and more for skin exposure, and at distances of 1600 m and more for thyroid exposure. Since both of these values are within the sanitary protection zone, it can be concluded that in this case, the population N exposed to radiation with equivalent doses > 0.5 Sv is zero. Thus, according to formula (5.4), the collective risk of deterministic effects R is absent, and according to formula (5.3), the individual risk of deterministic effects r corresponds to the value of the frequency of the maximum accidental release and is 3.83E-08 events per reactor-year.

#### Risk of stochastic effects

The value of the risk of stochastic effects was calculated for the conditions of normal operation of the AP1000 power units. Air emissions from each of the two KhNPP units 5 and 6 are released through 2 ventilation pipes: the main ventilation pipe and the turbine building ventilation pipe of the plant [5]. The parameters of the 4 emission sources required for model calculations of atmospheric transfer of radionuclides are given in Table 5.31.

Table 5.31- Characteristics of radioactive release sources from the KhNPP Units 5 and 6 under normal operation

Release source	Unit 5 main ventilation pipe	Unit 5 TB ventilation pipe	Unit 6 main ventilation pipe	Unit 6 TB ventilation pipe
Effective height, m	81,6	39,8	81,6	39,8
Nuclear composition	Emission activity (Bq/year)			
H-3	1,8E+12	-	1,8E+12	-
C-14	2,7E+11	-	2,7E+11	-

Release source	Unit 5 main ventilation pipe	Unit 5 TB ventilation pipe	Unit 6 main ventilation pipe	Unit 6 TB ventilation pipe
Ar-41	4,8E+10	-	4,8E+10	-
Kr-85m	1,66E+10	7,80E+09	1,66E+10	7,80E+09
Kr-85	3,06E+12	2,60E+10	3,06E+12	2,60E+10
Kr-87	1,70E+10	2,20E+09	1,70E+10	2,20E+09
Kr-88	1,81E+10	8,50E+09	1,81E+10	8,50E+09
Xe-131m	1,31E+12	8,10E+10	1,31E+12	8,10E+10
Xe-133m	8,07E+10	3,50E+10	8,07E+10	3,50E+10
Xe-133	9,59E+11	2,90E+11	9,59E+11	2,90E+11
Xe-135m	1,30E+11	5,90E+10	1,30E+11	5,90E+10
Xe-135	1,73E+11	2,60E+11	1,73E+11	2,60E+11
Xe-138	5,90E+10	2,90E+10	5,90E+10	2,90E+10
I-131	2,06E+08	3,36E+06	2,06E+08	3,36E+06
I-133	3,45E+08	3,74E+06	3,45E+08	3,74E+06
Cr-51	1,20E+05	6,70E+04	1,20E+05	6,70E+04
Mn-54	-	1,10E+05	-	1,10E+05
Co-58	7,93E+05	7,80E+06	7,93E+05	7,80E+06
Co-60	1,90E+05	3,00E+06	1,90E+05	3,00E+06
Sr-89	3,28E+05	7,80E+05	3,28E+05	7,80E+05
Sr-90	1,10E+05	3,00E+05	1,10E+05	3,00E+05
Zr-95	3,70E+05	-	3,70E+05	-
Cs-134	2,00E+05	6,30E+05	2,00E+05	6,30E+05
Cs-137	2,70E+05	1,00E+06	2,70E+05	1,00E+06

The values of integrated volumetric specific activity and surface specific activity of radionuclides were calculated using a Gaussian model of impurities distribution in the atmosphere for continuous release, in which the dilution factor (or directly the volumetric activity of a radionuclide in the surface air) is calculated as an average for a period of 1 year, including the climatic average values of meteorological parameters repeatability. The repeatability values of meteorological parameters were calculated based on the data of KhNPP meteorological station measurements for 2013-2023 as elements of a three-dimensional matrix characterized by different values 1) wind direction (16 categories), 2) wind speed (8 categories) and 3) atmospheric stability category (6 categories).

The calculated values of radionuclide activity summed with the contributions of all 4 sources (Table 5.31) were used to estimate the effective dose to the population for a period of 1 year, and using formulas (5.1) and (5.2), the values of individual and collective risk of stochastic effects from exposure.

Figure 5.18 shows the values of the individual risk of stochastic effects for the population *r* calculated using formula (5.1) using the results of effective dose calculations for the critical age group "1 year" and average climatic conditions of release propagation with east-northeast wind, at which the calculated values of nuclide activity in air and fallout and the corresponding risks are the highest. The maximum value of individual risk  $r = 3.23 \cdot 10^{-7}$  1/year is achieved at a distance of about 800 m. Thus, for normal operation of two units (Nos. 5 and 6) with AP1000 reactors, the individual risk of stochastic effects

for the population *r* is less than the neglected risk level of  $10^{-6}$  1/year [22] for all distances from the source, reference age groups and meteorological conditions of radioactive release in the atmosphere. Accordingly, it can be assumed that the value of the collective risk of stochastic effects *R* can also be neglected.



Figure 5.18 - Individual risk values for stochastic effects for the population in dependence from distance to the source (1) and level of neglected risk (2).

#### 5.4.1.3 Assessment of the risk of the planned activity to public health

The risk of the planned activity's impact on public health from air pollution is assessed based on the calculations of the risk of non-carcinogenic and carcinogenic effects. Risk assessment is carried out under the following guidelines [17].

Characterization of the risk of non-carcinogenic effects under combined and complex effects of chemical compounds is based on calculating the hazard index (HI). The hazard index for the conditions of simultaneous intake of several substances by the same route (e.g., inhalation or oral) is calculated using the following formula:

$$HI = \sum HQ_i \tag{5.5}$$

 $HQ_i$  – hazard quotients for individual substances, which are determined by:

$$HQ_i = C_i / RfC_i \tag{5.6}$$

 $C_i$  – the estimated average annual concentration of the *i* substance, mg/m³;

 $RfC_i$  – the reference (safe) concentration of the *i* substance, mg/m³.

 $HQ_i = 1$  – the limit value of the accepted risk.

The assessment of non-carcinogenic risk is carried out under the criteria given in Table 5.32.

T 11 F 00 NT	• •	• 1	• . •
Table 5 $37 - No$	n_carcinogenic	rick	criteria
14010 3.32 110	1-caremogenie	TION	cincina

Risk characterization	Hazard quotient (HQ)
The risk of harmful effects is considered as negligibly small.	< 1
The threshold value, which does not require urgent measures, however, cannot be considered sufficiently acceptable.	1
The probability of developing harmful effects increases in proportion to the increase in HQ.	> 1

The hazard index of the planned activity was calculated as the sum of the hazard factors of individual substances contained in the emissions from the diesel generator (including background): nitrogen dioxide, sulfur dioxide, carbon monoxide, hydrocarbons C12-C19, and suspended solids. The results are shown in Table 5.33.

	Nitrogen dioxide	Sulfur dioxide	Carbon monoxide	Hydrocarbon limit C12- C19	Suspended solids non-differentiated by structure
Maximum one-time concentration at the limit of the residential area, $mg/m^3$	0,224	0,064	0,418	0,413	0,064
Estimated average annual concentration, $C_i$ mg/m ³ ,	0,00001	0,000004	0,00002	0,00002	0,000004
Reference (safe) concentration of the $i$ - substance, RfC _i , mg/m ³	0,04	0,08	5,0	1	0,5
Hazard quotient HOi	0.00025	0.00005	0.000004	0.00002	0.000002

Table 5.33 – Calculation of non-carcinogenic effects

HI = 0,00025 + 0,00005 + 0,000004 + 0,00002 + 0,000002 = 0,00033 < 1

According to Table B.1 of Annex B DBN A.2.2-1:2021 [17], the risk of non-carcinogenic effects is estimated to be extremely low.

The risk of developing carcinogenic effects is not determined due to the absence of carcinogenic substances in the emissions from diesel generators.

The social risk assessment of the planned activity is not determined because the emissions from the backup diesel generators do not contain carcinogenic substances.

The results of the risk assessment of the planned activities for the population's health prove that the commissioning of KhNPP Units 5 and 6 will not significantly affect the population's health. Moreover, it should be noted that the planned activity does not involve emissions of carcinogenic substances. Based on the results of the assessment, the impact of the planned activity on the social environment is characterized as environmentally acceptable.

## 5.4.2 Risks to Cultural Heritage Sites

Cultural heritage sites are located beyond the territory planned for the industrial site of KhNPP Units 5 and 6. Therefore, the construction and commissioning of the new power units does not pose any risks to cultural monuments.

5.5 The cumulative impact of other existing facilities, planned activities, and facilities for which a decision on the planned activity has been received, taking into account all existing environmental issues related to areas of special environmental protection significance that may be affected or where natural resources may be used

The KhNPP supervised area covers the territory of four districts of the Khmelnytskyy and Rivne regions. In the Khmelnytskyy region: the Shepetivka and Khmelnytskyy districts, in the Rivne region: the Rivne and Dubno districts.





Figure 5.19 – Location of facilities in the KhNPP 10-km area

There are no large hydraulic facilities upstream and downstream of the Goryn and Viliya rivers in the 30-kilometer zone of the NPP. Reclamation drainage systems in the floodplains of these water sources do not affect the water supply to the NPP plant.

Industrial facilities within 30 kilometers of the NPP are concentrated mainly in the cities of Netishyn, Slavuta, Izyaslav, and Ostroh. These facilities are of local importance with a small production output. The industry of the neighboring towns is represented by the following enterprises:

- Slavuta town – a silicate products plant, a building faience plant, a woodworking plant, food industry enterprises (butter, fruit canning, brewery, etc.), local and consumer goods industry enterprises;

– Izyaslav town – a brick factory, a furniture factory, a musical instrument factory, a butter factory;

– Ostroh town – a brick factory, a drainage pipe factory, a woodworking plant, a furniture factory, a sugar factory, a fruit canning factory, a brewery, and a food processing plant.

The list of major enterprises with their locations is presented in Tables 5.34 and 5.35.

Location area	Distance to NPP (km)	Name of the enterprise			
	Izyaslav territorial community				
Izyaslav town	24-28	1 "Kharchomash" plant			
(southeast)		2 Motor transport company (MTC) - 16840			
		3 Heating network			
		4 Fruit canning plant			
		5 Sales base of bakery products (SBBP)			
		6 Feed mill			
		7 Bread factory			
		8 Butter factory			
		9 Repair and transportation enterprise			
		10 Region Construction Ryagrobud			
		11 Full production and commercial company (FPCC)			
		12 Regional consumer socierty			
		13 State forestry			
		14 State institution№ 3			
		15 Interdistrict base			
		16 MH 324/31			
		17 MH 324/58			
		18 Factory "Octave"			
		19 MTC OSS			
		20 Agricultural chemical plant			
		21 Water utility company			
		22 Izyaslav Waste Processing Complex LLC			
Velyka	20-21	23 Peat briquette plant			
Radohoshch					
village (southwest)					
Pluzhne village	14-15	24 Bread factory			
(south)		25 Brick factory			
Kliubivka village	26-27	26 Sugar factory.			
(south)	~				
	Slavuta te	rritorial community			
Slavuta town (east)	14-18	I OJSC Silikat			
		2 Oil depot			
		3 Budtarfor plant			
		4 Municipal enterprise "Vyrobnychnyk"			
		5 Woodworking plant			
		6 ATP-16845			
		/ District management company			
		8 Combine of public utilities			
		9 Heating network			

Table 5.34 – List of enterprises in the Khmelnytskyy region

II Sewing factory           II ATP-202           II Cagroutural supply company           II Chicory drying plant           II Firm Progress"           II Briam "Progress"           II B Brewery           20 Repair and mechanical plant           II Stafe forestry enterprise           II Stafe forestry entill forestry enterprise	Location area	Distance to NPP (km)	Name of the enterprise
11 ATP-2202           12 Agricultural supply company           13 Bread factory           14 Chicory drying plant           15 Bread products plant           16 RTP           17 Water Utility Company           18 Firm "Progress"           19 Brewery           20 Repair and mechanical plant           21 State forestry enterprise           22 Cloth factory           23 Plant "Estafeta"           24 Furniture shop           25 Paper mill           26 Slavuta Roofing Plant LLC           27 PJSC "Slavuta Glass Factory"           Tsvitokha village           (south-east)           29 RTP           Krupets village           10-12           30 "KIINELNYTSKEBUDENERGO" LTD           31 PJSC "Slavuta malt plant"           Polyan village           5-7           32 Soury of JSC "Volyn"           36 OOO "ENERGOPUDENZAKHIDMONTAZH"           Hannopil village           5-7           35 OOO "ENERGOPUDENZAKHIDMONTAZH"           Hannopil village           5-7           37 OJSC " Agricultural Chemicals"           38 State Enterprise "Gannopol Distillery"           Vanivka village           5-			10 Sewing factory
12 Agricultural supply company           13 Bread factory           14 Chicory drying plant           15 Bread products plant           16 RTP           17 Water Utility Company           18 Firm "Progress"           19 Brewery           20 Repair and mechanical plant           21 State forestry enterprise           22 Cloth factory           23 Plant "Estafeta"           24 Furniture shop           25 Paper mill           26 Slavuta Roofing Plant LLC           27 PJSC "Slavuta Glass Factory"           Tsvitokha village           (south-east)           29 RTP           Krupets village           10-12           30 Paper mill           20 Repir Mill           21 State Enterprise (Agricultural Chemicals)           Kryvyn village           5-7           37 OSC " Agricultural Chemicals"           38 State Enterprise "Ganopol Distillery"           17 Valse           38 Atter Enterprise           38 Atter Enterprise           38 State Enterprise           39 OO STENERGOPIVDENZAKHIDMONTAZH"           17 Als Care Main           38 State Enterprise           39 Atterprise "Ganopol Distillery" <td></td> <td></td> <td>11 ATP-2202</td>			11 ATP-2202
13 Bread factory           14 Chicory drying plant           15 Bread products plant           16 RTP           17 Water Utility Company           18 Firm "Progress"           20 Repair and mechanical plant           21 State forestry entreprise           22 Cloth factory           23 Plant "Estafeta"           24 Furniture shop           25 Paper mill           26 Slavuta Roofing Plant LLC           27 PJSC "Slavuta Glass Factory"           Tsvitokha village           (south-east)           29 RTP           Krupets village           10-12           30 "KHMELNYTSKEBUDENERGO" LTD           31 PJSC "Slavuta malt plant"           Polyan village           (north-east)           25-7           32 Qarry of JSC "Volyn"           34 Agrochemistry (Agricultural Chemicals)           Kryvyn village           5-7           35 Quarry of JSC "Volyn"           36 GOO" "ENERGOPUDENZAKHIDMONTAZH"           Hannopil village           5-7           37 OJSC " Agricultural Chemicals"           Strate Enterprise "Gannopol Distillery"           Ivanivka village           5-7           39 Feed			12 Agricultural supply company
If Chicory drying plant           If S Bread products plant           If RTP           If RTP           If S Firm "Progress"           If B Firm "Progress"           State Enterprise           If B Firm "Progress"           State Enterprise           If B Firm "Progress"           If B Firm "Progress"           State Enterprise "Gannopol Distillery"           If a A for cohemistry (Agricultral Chemicals)           If Prochemistry (Agricultral Chemicals)           If B Firm "Progress"           If B Firm "Progress"           If B Firm "Progress"           If If State Enterprise "Gannopol Distillery"           If a transprid gate			13 Bread factory
I5 Bread products plant           16 RTP           17 Water Utility Company           18 Firm "Progress"           19 Brewery           20 Repair and mechanical plant           21 State forestry enterprise           22 Cloth factory           23 Plant "Estafeta"           24 Furniture shop           25 Paper mill           26 Slavuta Roofing Plant LLC           27 PISC "Slavuta Glass Factory"           Tsvitokha village           (south-east)           29 RTP           Krupets village           10-12           30 "KHMELNYTSKEBUDENERGO" LTD           31 PJSC "Slavuta Glass Factory"           29 RTP           Krupets village           10-12           30 "KHMELNYTSKEBUDENERGO" LTD           31 PJSC "Slavuta malt plant"           Polyan village           67           32 Quarry of JSC "Volyn"           36 OOO "ENERGOPIVDENZAKHDMONTAZH"           Hannopil village           5-7           37 OJSC " Agricultural Chemicals"           38 State Enterprise "Gannopol Distillery"           Yanivka village           5-7           39 Feed mill           Stryhany village			14 Chicory drying plant
16 RTP         17 Water Utility Company           18 Fim "Progress"         19 Brewery           20 Repair and mechanical plant         21 State forestry enterprise           22 Cloth factory         23 Plant "Estafeta"           24 Furniture shop         26 Slavuta Roofing Plant LLC           27 PJSC "Slavuta Glass Factory"         26 Slavuta Glass Factory"           Tsvitokha village         22-23           (south-east)         29 RTP           Krupets village         10-12           30 "KHMELNYTSKEBUDENERGO" LTD           31 PJSC "Slavuta Glass Factory"           73 Sgary of JSC "Slavuta Chemicals           8 The Stafe at the stafe			15 Bread products plant
17 Water Utility Company           18 Firm "Progress"           19 Brewery           20 Repair and mechanical plant           21 State forestry enterprise           22 Cloth factory           23 Plant "Estafeta"           24 Furniture shop           25 Paper mill           26 Slavuta Roofing Plant LLC           27 PJSC "Slavuta Glass Factory"           Tsvitokha village           22-23           29 RTP           Krupets village           10-12           30 "KHMELNYTSKEBUDENERGO" LTD           31 PJSC "Slavuta malt plant"           Polyan village           (north-east)           33 Paper mill           34 Agrochemistry (Agricultural Chemicals)           Kryvyn village           5-7           35 Quarry of JSC "Volyn"           43 Agrochemistry (Agricultural Chemicals)           35 Quarry of JSC "Volyn"           43 Report Plant           Hannopil village           5-7           39 Eeed mill           Stryhany village           5-7           39 Feed mill           Stryhany village           5-7           30 Size Agricultural Chemicals"           34			16 RTP
I8 Firm "Progress"           19 Brewery           20 Repair and mechanical plant           21 State forestry enterprise           22 Cloth factory           23 Plant "Estafeta"           24 Furniture shop           25 Paper mill           26 Slavuta Roofing Plant LLC           27 PJSC "Slavuta Glass Factory"           Tsvitokha village           (south-east)           29 RTP           Krupets village           10-12           30 "KHMELNYTSKEBUDENERGO" LTD           31 PJSC "Slavuta malt plant"           Polyan village           (north-east)           33 Paper mill           34 Agrochemistry (Agricultural Chemicals)           Kryvyn village           5-7           35 Quarry of JSC "Volyn"           36 OOO "ENERGOPIVDENZAKHIDMONTAZH"           Hannopil village           5-7           35 Quarry of JSC "Volyn"           38 State Enterprise "Gannopol Distillery"           Ivanivka village           5-7           39 Feed mill           Stryhany village           5-7           40 Sand quarry           Romaniny village           5-7           41 Brick p			17 Water Utility Company
19 Brewery           20 Repair and mechanical plant           21 State forestry enterprise           22 Cloth factory           23 Plant "Estafeta"           24 Furniture shop           25 Paper mill           26 Slavuta Roofing Plant LLC           27 PJSC "Slavuta Glass Factory"           Tsvitokha village           (south-east)           29 RTP           Krupets village           10-12           30 "KHMELNYTSKEBUDENERGO" LTD           31 PJSC "Slavuta malt plant"           Polyan village           (north-east)           33 Paper mill           34 Agrochemistry (Agricultural Chemicals)           Kryvyn village           5-7           35 Quarry of JSC "Volyn"           36 OOO "ENERGOPIVDENZAKHIDMONTAZH"           Hannopil village           5-7           35 Quarry of JSC "Nolyn"           38 State Enterprise "Gannopol Distillery"           Ivanivka village           5-7           39 Feed mill           Stryhany village           5-7           30 And quarry           Romaniny village           5-7           40 Sand quarry           Romaniny villag			18 Firm "Progress"
20 Repair and mechanical plant           21 State forestry enterprise           22 Cloth factory           23 Plant "Estafeta"           24 Furniture shop           25 Paper mill           26 Slavuta Roofing Plant LLC           27 PJSC "Slavuta Glass Factory"           Tsvitokha village           (south-east)           29 RTP           Krupets village           10-12           30 "KHMELNYTSKEBUDENERGO" LTD           31 PJSC "Slavuta malt plant"           Polyan village           (north-east)           32 Sand quarry           33 Paper mill           34 Agrochemistry (Agricultural Chemicals)           Kryvyn village           5-7           35 Quarry of JSC "Volya"           36 OOO "ENERGOPIVDENZAKHIDMONTAZH"           Hannopil village           5-7           37 OJSC " Agricultural Chemicals"           38 State Enterprise "Gannopol Distillery"           Ivanivka village           5-7           39 Feed mill           Stryhany village           5-7           40 Sand quarry           Romaniny village           5-7           41 Brick plant           N			19 Brewery
21 State forestry enterprise         22 Cloth factory         23 Plant "Estafeta"         24 Furniture shop         25 Paper mill         26 Slavuta Roofing Plant LLC         27 PJSC "Slavuta Glass Factory"         Tsvitokha village         (south-east)         29 RTP         Krupets village         10-12       30 "KHMELNYTSKEBUDENERGO" LTD         31 PJSC "Slavuta malt plant"         Polyan village         (north-east)         32 Sand quary         33 Paper mill         34 Agrochemistry (Agricultural Chemicals)         Kryvyn village         5-7       35 Quarry of JSC "Volyn"         36 OOO "ENERGOPIVDENZAKHIDMONTAZH"         Hannopil village       5-7         5-7       37 OJSC " Agricultural Chemicals"         38 State Enterprise "Gannopol Distillery"         Ivanivka village       5-7         Stryhany village       5-7         5-7       40 Sand quary         Romaniny village       5-7         Stryhany village       5-7         41 Brick plant       42 CISC "Notishinharchprom"         43 Atomenergoremont       43 Atomenergoremont         44 KHNPP Construction Department       45 MU-7			20 Repair and mechanical plant
22 Cloth factory           23 Plant "Estafeta"           24 Furniture shop           25 Paper mill           26 Slavuta Roofing Plant LLC           27 PJSC "Slavuta Glass Factory"           Tsvitokha village           22-23           27 PJSC "Slavuta Glass Factory"           Tsvitokha village           22 P RTP           Krupets village           10-12           30 "KHMELNYTSKEBUDENERGO" LTD           31 PJSC "Slavuta malt plant"           Polyan village           (north-east)           32 Paper mill           33 Paper mill           34 Agrochemistry (Agricultural Chemicals)           Kryvyn village           5-7         35 Quarry of JSC "Volyn"           36 OOO "ENERGOPIVDENZAKHIDMONTAZH"           Hannopil village         5-7           57         37 OJSC " Agricultural Chemicals"           18 State Enterprise "Gannopol Distillery"           Ivanivka village         5-7           40 Sand quarry           Romaniny village         5-7           41 Brick plant           Netishyn town         1-3           42 CJSC "Netishinharchprom"           44 KHNPP Construction Department           44 S MU-7 </td <td></td> <td></td> <td>21 State forestry enterprise</td>			21 State forestry enterprise
23 Plant "Estafeta"           24 Furniture shop           25 Paper mill           26 Slavuta Roofing Plant LLC           27 PJSC "Slavuta Roofing Plant LLC           27 PJSC "Slavuta Roofing Plant LLC           27 PJSC "Slavuta Roofing Plant LLC           28 PRP           Krupets village           10-12         30 "KHMELNYTSKEBUDENERGO" LTD           31 PJSC "Slavuta malt plant"           Polyan village         5-7           10-12         30 "KHMELNYTSKEBUDENERGO" LTD           31 PJSC "Slavuta malt plant"           Polyan village         5-7           10-12         30 "KIMELNYTSKEBUDENERGO" LTD           31 PJSC "Slavuta malt plant"           Polyan village         5-7           10-12         35 Quarry of JSC "Volyn"           38 Gate Enterprise "Gannopol Distillery"           Ivanivka village         5-7           5-7         40 Sand quarry           Romaniny village         5-7           5-7         40 Sand quarry           Romaniny village         5-7           40 Sund quarry           Romaniny village         5-7           40 Sand quarry           Romaniny village         5-7           41 Brick plant      <			22 Cloth factory
24 Furniture shop           25 Paper mill           26 Slavuta Roofing Plant LLC           27 PJSC "Slavuta Glass Factory"           Tsvitokha village           (south-east)           29 RTP           Krupets village           10-12           30 "KHMELNYTSKEBUDENERGO" LTD           31 PJSC "Slavuta malt plant"           Polyan village           (north-east)           32 Sand quarry           (north-east)           34 Agrochemistry (Agricultural Chemicals)           Kryvyn village           5-7           35 Quarry of JSC "Volyn"           36 OOO "ENERGOPIVDENZAKHIDMONTAZH"           Hannopil village           5-7           39 Feed mill           Stryhany village           5-7           39 Feed mill           Stryhany village           5-7           40 Sand quarry           Romaniny village           5-7           41 Brick plant           Netishyn town           1-3           42 CJSC "Netishinharchprom"           43 Atomenergoremont           44 KHNPP Construction Department           45 MU-7           46 MU-13     <			23 Plant "Estafeta"
25 Paper mill           26 Slavuta Roofing Plant LLC           27 PJSC "Slavuta Glass Factory"           Tsvitokha village           (south-east)           Krupets village           10-12           30 "KHMELNYTSKEBUDENERGO" LTD           31 PJSC "Slavuta malt plant"           Polyan village           (north-east)           32 Sand quarry           (north-east)           34 Agrochemistry (Agricultural Chemicals)           Kryvyn village           5-7           35 Quarry of JSC "Volyn"           36 OOO "ENERGOPIVDENZAKHIDMONTAZH"           Hannopil village           5-7           37 OJSC " Agricultural Chemicals"           38 State Enterprise "Gannopol Distillery"           Ivanivka village           5-7           39 Feed mill           Stryhany village           5-7           41 Brick plant           Netishyn town           1-3           42 CJSC "Netishinharchprom"           43 Atomenergoremont           44 KHNPP Construction Department           45 MU-7           46 MU-13           47 Khmelnytskyy ATP-16827           50 Production and Trading Company "Dolphin"     <			24 Furniture shop
26 Slavuta Roofing Plant LLC           27 PJSC "Slavuta Glass Factory"           Tsvitokha village         22-23           (south-east)         29 RTP           Krupets village         10-12         30 "KHMELNYTSKEBUDENERGO" LTD           31 PJSC "Slavuta malt plant"         31 PJSC "Slavuta malt plant"           Polyan village         5-7         32 Sand quarry           (north-east)         33 Paper mill         34 Agrochemistry (Agricultural Chemicals)           Kryvyn village         5-7         35 Quarry of JSC "Volyn"           Kryvyn village         5-7         37 OJSC " Agricultural Chemicals"           Mannopil village         5-7         39 Feed mill           Stryhany village         5-7         40 Sand quarry           Romaniny village         5-7         40 Sand quarry           Romaniny village         5-7         41 Brick plant           Netishyn town         1-3         42 CJSC "Netishinharchprom"           44 KHNPP Construction Department         45 MU-7           46 MU-13         49 Khmelnytskyy Heat and Power Installation           48 SU-3         49 Khmelnytskyy ATP-16827           50 Production and Trading Company "Dolphin"         51 Atommontazhservice LLC           52 LLC "Construction Industry KhNPP Construction Department"         5			25 Paper mill
27 PJSC "Slavua Glass Factory"         Tsvitokha village (south-east)       22-23         Yrupets village       10-12         30 "KHMELNYTSKEBUDENERGO" LTD 31 PJSC "Slavuta malt plant"         Polyan village (north-east)       5-7         32 Sand quarry (north-east)       32 Sand quarry 33 Paper mill 34 Agrochemistry (Agricultural Chemicals)         Kryvyn village       5-7         35 Quarry of JSC "Volyn"         36 OOO "ENERGOPIVDENZAKHIDMONTAZH"         Hannopil village       5-7         37 OJSC " Agricultural Chemicals"         38 State Enterprise "Gannopol Distillery"         Ivanivka village       5-7         Stryhany village       5-7         40 Sand quarry         Romaniny village       5-7         50 CV "Netishinharchprom"         43 Atomenergoremont         44 KHNPP Construction Department         45 MU-7         46 MU-13         47 Khmelnytskyy Heat and Power Installation         48 SU-3         49 Khmelnytskyy ATP-16827         50 Production and Trading Company "Dolphin"         51 Atommontazhservice LLC         52 LLC "Construction Industry KhNPP Construction Department"         53 PE "Eximbud"         54 Research and Production Commercial Enterprise Alfa LTD LLC </td <td></td> <td></td> <td>26 Slavuta Roofing Plant LLC</td>			26 Slavuta Roofing Plant LLC
Tsvitokha village (south-east)       22-23       29 RTP         Krupets village       10-12       30 "KHMELNYTSKEBUDENERGO" LTD 31 PJSC "Slavuta malt plant"         Polyan village (north-east)       5-7       32 Sand quarry 33 Paper mill 34 Agrochemistry (Agricultural Chemicals)         Kryvyn village       5-7       35 Quarry of JSC "Volyn"         Hannopil village       5-7       37 OJSC " Agricultural Chemicals)         Kryvyn village       5-7       39 Feed mill         State Enterprise "Gannopol Distillery"       38 State Enterprise "Gannopol Distillery"         Ivanivka village       5-7       40 Sand quarry         Romaniny village       5-7       40 Sand quarry         Romaniny village       5-7       40 Sand quarry         Netishyn town       1-3       42 CJSC "Netishinharchprom"         43 Atomenergoremont       44 KHNPP Construction Department         44 S MU-7       46 MU-13         49 Khmelnytskyy Heat and Power Installation       48 SU-3         49 Khmelnytskyy ATP-16827       50 Production and Trading Company "Dolphin"         51 Atommontazhservice LLC       52 LLC "Construction Industry KhNPP Construction Department"         53 PE "Eximbud"       54 Research and Production Commercial Enterprise Alfa LTD LLC         55 LLC "Energy Complex KhNPP Construction Department "       55 LLC "E			27 PJSC "Slavuta Glass Factory"
(south-east)       29 RTP         Krupets village       10-12       30 "KHMELNYTSKEBUDENERGO" LTD         31 PJSC "Slavuta malt plant"       33 Paper mill         Yender       33 Paper mill         34 Agrochemistry (Agricultural Chemicals)         Kryvyn village       5-7         35 Quarry of JSC "Volyn"         36 OOO "ENERGOPIVDENZAKHIDMONTAZH"         Hannopil village       5-7         57       39 Feed mill         Stryhany village       5-7         57       39 Feed mill         Stryhany village       5-7         6000 "ENERGOPIVDENZAKHIDMONTAZH"         Hannopil village       5-7         57       39 Feed mill         Stryhany village       5-7         40 Sand quarry         Romaniny village       5-7         41 Brick plant         Netishyn town       1-3         42 CJSC "Netishinharchprom"         43 Atomenergoremont         44 KHNPP Construction Department         45 MU-7         46 MU-13         47 Khmelnytskyy Heat and Power Installation         48 SU-3         49 Khmelnytskyy ATP-16827         50 Production and Trading Company "Dolphin"         51 Atommontazhservice L	Tsvitokha village	22-23	
Krupets village       10-12       30 "KHMELNYTSKEBUDENERGO" LTD         90lyan village (north-east)       31 PJSC "Slavuta malt plant"         33 Paper mill       34 Agrochemistry (Agricultural Chemicals)         34 Agrochemistry (Agricultural Chemicals)       35 Quarry of JSC "Volyn"         36 OOO "ENERGOPIVDENZAKHIDMONTAZH"         Hannopil village       5-7         37 OJSC " Agricultural Chemicals"         38 State Enterprise "Gannopol Distillery"         Ivanivka village       5-7         57       39 Feed mill         Stryhany village       5-7         40 Sand quarry         Romaniny village       5-7         41 Brick plant         Netishyn town       1-3         42 CJSC "Netishinharchprom"         43 Atomenergoremont         44 KHNPP Construction Department         45 MU-7         46 MU-13         47 Khmelnytskyy Heat and Power Installation         48 SU-3         49 Khmelnytskyy ATP-16827         50 Production and Trading Company "Dolphin"         51 Atommontazhservice LLC         52 LLC "Construction Industry KhNPP Construction         53 PE "Eximbud"         54 Research and Production Commercial Enterprise         Alfa LTD LLC       55 LLC "Energy Comp	(south-east)		29 RTP
Arrow and a second s	Krupets village	10-12	30 "KHMELNYTSKEBUDENERGO" LTD
Polyan village (north-east)       5-7       32 Sand quarry         33 Paper mill       34 Agrochemistry (Agricultural Chemicals)         Kryvyn village       5-7       35 Quarry of JSC "Volyn"         Hannopil village       5-7       37 OJSC " Agricultural Chemicals"         With a village       5-7       37 OJSC " Agricultural Chemicals"         Stryhany village       5-7       39 Feed mill         Stryhany village       5-7       40 Sand quarry         Romaniny village       5-7       41 Brick plant         Netishyn town       1-3       42 CJSC "Netishinharchprom"         43 Atomenergoremont       44 KHNPP Construction Department         45 MU-7       46 MU-13         47 Khmelnytskyy Heat and Power Installation       48 SU-3         49 Khmelnytskyy ATP-16827       50 Production and Trading Company "Dolphin"         51 Atommontazhservice LLC       52 LLC "Construction Industry KhNPP Construction Department"         53 PE "Eximbud"       54 Research and Production Commercial Enterprise Alfa LTD LLC         51 LC "Energy Complex KhNPP Construction Department "       51 LC "Energy Complex KhNPP	in apois (mago	10 12	31 PISC "Slavuta malt plant"
(north-east)       33 Paper mill         33 Paper mill       34 Agrochemistry (Agricultural Chemicals)         Kryvyn village       5-7         35 Quarry of JSC "Volyn"         36 OOO "ENERGOPIVDENZAKHIDMONTAZH"         Hannopil village       5-7         37 OJSC " Agricultural Chemicals"         38 State Enterprise "Gannopol Distillery"         Ivanivka village       5-7         39 Feed mill         Stryhany village         5-7       40 Sand quarry         Romaniny village       5-7         41 Brick plant         Netishyn town         1-3       42 CJSC "Netishinharchprom"         43 Atomenergoremont         44 KHNPP Construction Department         45 MU-7         46 MU-13         47 Khmelnytskyy Heat and Power Installation         48 SU-3         49 Khmelnytskyy ATP-16827         50 Production and Trading Company "Dolphin"         51 Atommontazhservice LLC         52 LLC "Construction Industry KhNPP Construction         Department"         53 PE "Eximbud"         54 Research and Production Commercial Enterprise         Alfa LTD LLC         55 LLC "Energy Complex KhNPP Construction         Department" </td <td>Polyan village</td> <td>5-7</td> <td>32 Sand quarry</td>	Polyan village	5-7	32 Sand quarry
34 Agrochemistry (Agricultural Chemicals)         Kryvyn village       5-7         35 Quarry of JSC "Volyn"         36 OOO "ENERGOPIVDENZAKHIDMONTAZH"         Hannopil village       5-7         37 OJSC " Agricultural Chemicals"         38 State Enterprise "Gannopol Distillery"         Ivanivka village       5-7         39 Feed mill         Stryhany village       5-7         40 Sand quarry         Romaniny village         5-7       40 Sand quarry         Romaniny village         5-7       41 Brick plant         Netishyn town       1-3         42 CJSC "Netishinharchprom"         43 Atomenergoremont         44 KHNPP Construction Department         45 MU-7         46 MU-13         47 Khmelnytskyy Heat and Power Installation         48 SU-3         49 Khmelnytskyy ATP-16827         50 Production and Trading Company "Dolphin"         51 Atommontazhservice LLC         52 LLC "Construction Industry KhNPP Construction         Department"         53 PE "Eximbud"         54 Research and Production Commercial Enterprise         Alfa LTD LLC         55 LLC "Energy Complex KhNPP Construction         Department "	(north-east)		33 Paper mill
Kryvyn village       5-7       35 Quarry of JSC "Volyn"         Hannopil village       5-7       37 OJSC " Agricultural Chemicals"         Wanivka village       5-7       39 Feed mill         Stryhany village       5-7       40 Sand quarry         Romaniny village       5-7       40 Sand quarry         Romaniny village       5-7       40 Sand quarry         Romaniny village       5-7       41 Brick plant         Netishyn town       1-3       42 CJSC "Netishinharchprom"         43 Atomenergoremont       44 KHNPP Construction Department         45 MU-7       46 MU-13         47 Khmelnytskyy Heat and Power Installation       48 SU-3         49 Khmelnytskyy ATP-16827       50 Production and Trading Company "Dolphin"         51 Atommontazhservice LLC       52 LLC "Construction Industry KhNPP Construction Department"         53 PE "Eximbud"       54 Research and Production Commercial Enterprise Alfa LTD LLC         55 LLC "Energy Complex KhNPP Construction Department "       55 LLC "Energy Complex KhNPP Construction Department "	(		34 Agrochemistry (Agricultural Chemicals)
All y fundage     Description       36 OOO "ENERGOPIVDENZAKHIDMONTAZH"       Hannopil village       5-7       37 OJSC " Agricultural Chemicals"       38 State Enterprise "Gannopol Distillery"       Ivanivka village       5-7       39 Feed mill       Stryhany village       5-7       40 Sand quarry       Romaniny village       5-7       41 Brick plant       Netishyn town       1-3       42 CJSC "Netishinharchprom"       43 Atomenergoremont       44 KHNPP Construction Department       45 MU-7       46 MU-13       47 Khmelnytskyy Heat and Power Installation       48 SU-3       49 Khmelnytskyy ATP-16827       50 Production and Trading Company "Dolphin"       51 Atommontazhservice LLC       52 LLC "Construction Industry KhNPP Construction       Department"       53 PE "Eximbud"       54 Research and Production Commercial Enterprise       Alfa LTD LLC       55 LLC "Energy Complex KhNPP Construction       Department "	Krvvvn village	5-7	35 Quarry of JSC "Volyn"
Hannopil village       5-7       37 OJSC "Agricultural Chemicals"         Ivanivka village       5-7       39 Feed mill         Stryhany village       5-7       40 Sand quarry         Romaniny village       5-7       41 Brick plant         Netishyn town       1-3       42 CJSC "Netishinharchprom"         43 Atomenergoremont       43 Atomenergoremont         44 KHNPP Construction Department       45 MU-7         45 MU-7       46 MU-13         47 Khmelnytskyy Heat and Power Installation         48 SU-3       49 Khmelnytskyy ATP-16827         50 Production and Trading Company "Dolphin"         51 Atommontazhservice LLC       52 LLC "Construction Industry KhNPP Construction Department"         53 PE "Eximbud"       54 Research and Production Commercial Enterprise Alfa LTD LLC         55 LLC "Energy Complex KhNPP Construction Department "       55 LLC "Energy Complex KhNPP Construction Department "	in j i jii i inage		36 OOO "ENERGOPIVDENZAKHIDMONTAZH"
38 State Enterprise "Gannopol Distillery"       Ivanivka village     5-7       Stryhany village     5-7       Romaniny village     5-7       Netishyn town     1-3       42 CJSC "Netishinharchprom"       43 Atomenergoremont       44 KHNPP Construction Department       45 MU-7       46 MU-13       47 Khmelnytskyy Heat and Power Installation       48 SU-3       49 Khmelnytskyy ATP-16827       50 Production and Trading Company "Dolphin"       51 Atommontazhservice LLC       52 LLC "Construction Industry KhNPP Construction       Department"       53 PE "Eximbud"       54 Research and Production Commercial Enterprise       Alfa LTD LLC       55 LLC "Energy Complex KhNPP Construction       Department "	Hannopil village	5-7	37 OISC " Agricultural Chemicals"
Ivanivka village       5-7       39 Feed mill         Stryhany village       5-7       40 Sand quarry         Romaniny village       5-7       41 Brick plant         Netishyn town       1-3       42 CJSC "Netishinharchprom"         43 Atomenergoremont       43 Atomenergoremont         44 KHNPP Construction Department       45 MU-7         46 MU-13       47 Khmelnytskyy Heat and Power Installation         48 SU-3       49 Khmelnytskyy ATP-16827         50 Production and Trading Company "Dolphin"       51 Atommontaphservice LLC         52 LLC "Construction Industry KhNPP Construction       Department"         53 PE "Eximbud"       54 Research and Production Commercial Enterprise Alfa LTD LLC         55 LLC "Energy Complex KhNPP Construction       Department "			38 State Enterprise "Gannopol Distillery"
Stryhany village       5-7       40 Sand quarry         Romaniny village       5-7       41 Brick plant         Netishyn town       1-3       42 CJSC "Netishinharchprom"         43 Atomenergoremont       43 Atomenergoremont         44 KHNPP Construction Department       45 MU-7         46 MU-13       47 Khmelnytskyy Heat and Power Installation         48 SU-3       49 Khmelnytskyy ATP-16827         50 Production and Trading Company "Dolphin"       51 Atommontazhservice LLC         52 LLC "Construction Industry KhNPP Construction Department"       53 PE "Eximbud"         54 Research and Production Commercial Enterprise Alfa LTD LLC       55 LLC "Energy Complex KhNPP Construction Department "	Ivaniyka village	5-7	39 Feed mill
Romaniny village       5.7       41 Brick plant         Netishyn town       1-3       42 CJSC "Netishinharchprom"         43 Atomenergoremont       43 Atomenergoremont         44 KHNPP Construction Department         45 MU-7         46 MU-13         47 Khmelnytskyy Heat and Power Installation         48 SU-3         49 Khmelnytskyy ATP-16827         50 Production and Trading Company "Dolphin"         51 Atommontazhservice LLC         52 LLC "Construction Industry KhNPP Construction         Department"         53 PE "Eximbud"         54 Research and Production Commercial Enterprise         Alfa LTD LLC         55 LLC "Energy Complex KhNPP Construction         Department "	Stryhany village	5-7	40 Sand quarry
Netishyn town       1-3       42 CJSC "Netishinharchprom"         43 Atomenergoremont       43 Atomenergoremont         44 KHNPP Construction Department         45 MU-7         46 MU-13         47 Khmelnytskyy Heat and Power Installation         48 SU-3         49 Khmelnytskyy ATP-16827         50 Production and Trading Company "Dolphin"         51 Atommontazhservice LLC         52 LLC "Construction Industry KhNPP Construction         Department"         53 PE "Eximbud"         54 Research and Production Commercial Enterprise         Alfa LTD LLC         55 LLC "Energy Complex KhNPP Construction         Department "	Romaniny village	5-7	41 Brick plant
12 Cost Contraction Induction         43 Atomenergoremont         44 KHNPP Construction Department         45 MU-7         46 MU-13         47 Khmelnytskyy Heat and Power Installation         48 SU-3         49 Khmelnytskyy ATP-16827         50 Production and Trading Company "Dolphin"         51 Atommontazhservice LLC         52 LLC "Construction Industry KhNPP Construction         Department"         53 PE "Eximbud"         54 Research and Production Commercial Enterprise         Alfa LTD LLC         55 LLC "Energy Complex KhNPP Construction         Department "	Netishyn town	1-3	42 CISC "Netishinharchnrom"
44 KHNPP Construction Department         45 MU-7         46 MU-13         47 Khmelnytskyy Heat and Power Installation         48 SU-3         49 Khmelnytskyy ATP-16827         50 Production and Trading Company "Dolphin"         51 Atommontazhservice LLC         52 LLC "Construction Industry KhNPP Construction         Department"         53 PE "Eximbud"         54 Research and Production Commercial Enterprise         Alfa LTD LLC         55 LLC "Energy Complex KhNPP Construction         Department "	i totishiyii town	1.5	43 Atomenergoremont
45 MU-7         46 MU-13         47 Khmelnytskyy Heat and Power Installation         48 SU-3         49 Khmelnytskyy ATP-16827         50 Production and Trading Company "Dolphin"         51 Atommontazhservice LLC         52 LLC "Construction Industry KhNPP Construction         Department"         53 PE "Eximbud"         54 Research and Production Commercial Enterprise         Alfa LTD LLC         55 LLC "Energy Complex KhNPP Construction         Department "			44 KHNPP Construction Department
45 kter         46 MU-13         47 Khmelnytskyy Heat and Power Installation         48 SU-3         49 Khmelnytskyy ATP-16827         50 Production and Trading Company "Dolphin"         51 Atommontazhservice LLC         52 LLC "Construction Industry KhNPP Construction         Department"         53 PE "Eximbud"         54 Research and Production Commercial Enterprise         Alfa LTD LLC         55 LLC "Energy Complex KhNPP Construction         Department "			45 MU-7
47 Khmelnytskyy Heat and Power Installation         48 SU-3         49 Khmelnytskyy ATP-16827         50 Production and Trading Company "Dolphin"         51 Atommontazhservice LLC         52 LLC "Construction Industry KhNPP Construction         Department"         53 PE "Eximbud"         54 Research and Production Commercial Enterprise         Alfa LTD LLC         55 LLC "Energy Complex KhNPP Construction         Department "			46 MU-13
48 SU-3         49 Khmelnytskyy ATP-16827         50 Production and Trading Company "Dolphin"         51 Atommontazhservice LLC         52 LLC "Construction Industry KhNPP Construction         Department"         53 PE "Eximbud"         54 Research and Production Commercial Enterprise         Alfa LTD LLC         55 LLC "Energy Complex KhNPP Construction         Department "			47 Khmelnytskyy Heat and Power Installation
49 Khmelnytskyy ATP-16827 50 Production and Trading Company "Dolphin" 51 Atommontazhservice LLC 52 LLC "Construction Industry KhNPP Construction Department" 53 PE "Eximbud" 54 Research and Production Commercial Enterprise Alfa LTD LLC 55 LLC "Energy Complex KhNPP Construction Department "			48 SU-3
50 Production and Trading Company "Dolphin"         51 Atommontazhservice LLC         52 LLC "Construction Industry KhNPP Construction         Department"         53 PE "Eximbud"         54 Research and Production Commercial Enterprise         Alfa LTD LLC         55 LLC "Energy Complex KhNPP Construction         Department "			49 Khmelnytskyy ATP-16827
50 Froduction and Frading Company Dorphil         51 Atommontazhservice LLC         52 LLC "Construction Industry KhNPP Construction         Department"         53 PE "Eximbud"         54 Research and Production Commercial Enterprise         Alfa LTD LLC         55 LLC "Energy Complex KhNPP Construction         Department "			50 Production and Trading Company "Dolphin"
51 Atominointaziser vice LLC         52 LLC "Construction Industry KhNPP Construction         Department"         53 PE "Eximbud"         54 Research and Production Commercial Enterprise         Alfa LTD LLC         55 LLC "Energy Complex KhNPP Construction         Department "			51 Atommontazhservice LLC
52 EEC Construction Industry Kinver Construction         Department"         53 PE "Eximbud"         54 Research and Production Commercial Enterprise         Alfa LTD LLC         55 LLC "Energy Complex KhNPP Construction         Department "			52 LLC "Construction Industry KhNPP Construction
53 PE "Eximbud" 54 Research and Production Commercial Enterprise Alfa LTD LLC 55 LLC "Energy Complex KhNPP Construction Department "			Department"
54 Research and Production Commercial Enterprise Alfa LTD LLC 55 LLC "Energy Complex KhNPP Construction Department "			53 PF "Eximbud"
Alfa LTD LLC 55 LLC "Energy Complex KhNPP Construction Department "			54 Research and Production Commercial Enterprise
55 LLC "Energy Complex KhNPP Construction Department "			Alfa LTD LLC
Department "			55 LLC "Energy Complex KhNPP Construction
			Department "

Location area	Distance to NPP (km)	Name of the enterprise				
		56 Simex LLC				
		57 Ukrainian-Polish joint venture "Jurex-Ukraine"				
		58 Collective Trade and Production Optical Enterprise				
		"RITA"				
		59 LLC "Production Enterprise				
		"Turboelektromontazh"				
		60 PE Company "Energogaz"				
		61 Ukrainian-Polish joint venture in the form of a				
		limited liability company "V.V.T."				

Table 5.35 – List of enterprises in the Rivne region

Location area	Distance to NPP (km)	Name of the enterprise				
Ostroh territorial community						
Ostroh town	Ostroh town 8-10 1 Ostroh Brick Plant					
		2 Ostroh Mineral Water Plant LLC				
		3 OJSC Ostroh Furniture Factory				
		4 Vilia factory				
		5 Dairy plant				
		6 Sugar factory				
		7 Agriculture and construction company				
	8 Fruit canning plant					
		9 Ostroh Sugar Plant LLC				
		10 Rivne Feed Mill LLC				
		11 PJSC "Ostroh Grain Receiving Enterprise"				
	4-6	12 CJSC "Agrotech"				
	15-17	13 Private Production and Commercial Enterprise				
		"Rembudmontazhservice"				
9-11 14 PJSC "Mohyla Plant of Building Materials"						
	19-20	15 PE "Biogaztekhmontazh"				
	Hoshcha territorial community					
Hoshcha town	30-32	16 Promin LLC				
(north)		17 Food products plant				

The table shows that the closest to the NPP site enterprises are located as follows:

37 enterprises within a radius of 3-10 km. Of these, 20 are located in Netishyn and
 7 in Ostroh;

- three economic facilities are in the village of Polyany; two are in the village of Kryvyn and the village of Hannopil; one is in the village of Stryhany, the village of Romany, and the village of Velbivno.

Under normal operation and in emergencies, dose loads on flora and fauna outside the SPZ will not lead to changes in their structure or other negative changes. The commissioning of Units 5 and 6, and the accident-free operation of KhNPP with six units (including planned Units 3 and 4) will not adversely affect the overall species diversity of invertebrates and insects (see also 5.2). Following the commissioning of Units 5 and 6, no new technological processes are expected to be created that will result in the emission of any harmful substances into the atmosphere that are different from the existing ones, i.e. the launch of Units 5 and 6 will not have any negative impact on the SA facilities.

It is worth mentioning that Units 5 and 6 will not have any technological connections with Units 1-4, except for the technological wastewater systems, see also paragraph 5.3.3.

The performed dispersion measurements, taking into account the background air pollution and the influence of other air pollutants, showed no exceedances of the maximum permissible concentration standards, see also 1.5, 5.3, 5.4. The cumulative impact of the planned activity and neighboring polluting industrial enterprises is acceptable.

## 5.6 Impact of the planned activity on the climate, including the nature and scale of greenhouse gas emissions, and the sensitivity to climate change

The impact of KhNPP on the microclimate is determined by the performance of the cooling systems during the operation of six units, which include heat discharges, hydrothermal conditions of the cooling pond, cooling towers, and the nature of meteorological processes in the surface layer of the atmosphere.

Cooling systems can affect the microclimate of the air space directly above reservoirs and the environment in their boundaries.

The area of KhNPP impact will be local, namely: the impact of the cooling pond within 1.0 km, and the existing spray ponds -0.3 km.

#### 5.7 Technology and Substances Used

The environmental impacts caused by the technology and substances used are assessed considering the technology of hazardous chemicals and nuclear fuel management.

## 5.7.1 Handling chemicals and solutions

Chemicals that may impact the environment should be assessed using the Hazardous Substances Identification Methodology [58].





It is worth noting that the types, amounts, and quantities of chemicals consumed and the detailed list of chemicals for closed or open cooling systems depend on the sitespecific WCR program, site water quality and treatment requirements, and the choice of site-specific equipment. Therefore, a detailed list will be compiled after the design and selection of site-specific equipment is completed.

Within the EIA report, the following hazardous substances are expected to be used in the technological process of Units 5 and 6:

- boric acid
- diesel fuel
- hydrogen
- lithium hydroxide;
- hydrazine
- zinc acetate;
- nitrogen.

**Boric acid**  $(H_3BO_3)$  is added to the reactor coolant to control reactivity. The Boric Acid Solution Tank (BAST) supplies borated water to the Reactor Coolant System (RCS) for replenishment after restarting operations to maintain the required boron content in the Reactor Coolant System (RCS) during power operation. The Chemical and Volume Control System (CVS) supplies boron-rich feed water to the Core Maintenance Tanks (CMTs), accumulators, the Inner Reactor Water Storage Tank (IRWST), and the spent fuel pool. The Boric Acid Solution Tank (BAST) is located in a separate bunded area with a concrete floor and walls of sufficient volume to prevent acid spills into the environment. Emissions from the boric acid solution preparation tank and chemical mixing tank will be discharged to the sump of the annex/auxiliary building (9.5  $m^3$ ). Discharge from the sump is carried out to the Waste Water system (WWS). During the Reactor Coolant System (RCS) loop draining and depressurization operations, the CVS receives water from the RCS. This water is transferred from the CVS to the liquid radioactive waste management system for treatment and discharge. There is a possibility of reactor coolant leakage through the reactor vessel flange O-rings. This wastewater is treated and discharged through the Liquid Radwaste System (WLS). There is a potential for reactor coolant to leak through the pressure compensator relief valves and the automatic pressure relief system (ADS) valves (saddle leakage). These leaks accumulate in collectors in the lower part of the distribution manifolds of the ADS valves. This wastewater is quantified, treated, and discharged through the Liquid Radwaste System (WLS). Thus, any potential emissions occur either through the Waste Water system (WWS) or through the liquid discharge point from the Liquid Radwaste System (WLS).

*Diesel fuel* is used for backup diesel generators, auxiliary diesel generators, and a diesel fire pump. The main components in the Standby Diesel Fuel Oil System (DOS) for diesel engines are located in separate, heated, fan-ventilated rooms installed on racks, and modular enclosures of transfer platforms on the site. The fuel transfer platform has a perimeter edge to hold spilled oil. All the underground fuel oil pipelines of the Standby Diesel Fuel Oil System (DOS) are protected and encased in a corrosion-resistant piping system to protect from oil leaks. In addition, the protective pipe has a monitoring and alarm system to detect leaks. Each fuel oil storage tank of the Standby Diesel Fuel Oil System (DOS) is located in a separate area with a reinforced concrete floor and walls, surrounded by a dam of sufficient volume to contain the entire contents of the tank in the case of an accidental leak or failure to prevent contamination of the underground aquifer. The purpose of the tank design is to localize the leak. Therefore, the only leakage point is the Waste Water System (WWS), as the tanks are discharged to the WWS. The WWS ensures that the oil is separated from the effluent from the sumps before it is discharged from the plant outfall.

*Hydrogen gas*  $(H_2)$  is added to the reactor coolant under pressure and maintained in solution to provide reducing conditions at normal operating temperatures to minimize general corrosion of primary circuit surfaces and fuel element vessel surfaces and reduce the risk of stress corrosion cracking (SCC) of sensitized stainless steels and nickel-based alloys. Hydrogen is added to maintain a reducing chemical environment and to create thermodynamically suitable oxide films on the surfaces of the Reactor Coolant System (RCS) that are similar to those that will be stable during operation. The hydrogen supply

pipeline serving the Chemical and Volume Control System (CVS), as well as the pipelines of the Liquid Radwaste System (WLS) and the Gaseous Radwaste System (WGS), are postulated systems in which hydrogen is present during operation and in which the potential for leakage in the surrounding area is taken into account. Leakage points from other systems, including the Primary Sampling System (PSS) and auxiliary systems, were also investigated. The areas around the potential hydrogen leakage points have electrical classification zones defined under NFPA 70 (National Electrical Code). Gas release points are controlled as a part of the PU AP 1000's heating, ventilation, and air conditioning systems (HVAC) and do not generate any hazardous substances.

*Lithium-7 hydroxide (⁷LiOH)* is added to the Reactor Coolant System (RCS) to create an alkaline atmosphere with the presence of boric acid at operating temperatures. 7LiOH is added as a concentrated solution from the chemical mixing tank in the CVS and if necessary mixed with demineralized water to be added regularly to the RCS. ⁷LiOH is used to maintain an alkaline pH to minimize corrosion and the rate of corrosion product formation. The company will determine the amount of 7LiOH stored at the site in a warehouse.

*Hydrazine* ( $N_2H_4$ ) is added to the reactor coolant when the reactor is restarted after a fuel reloading shutdown to remove dissolved oxygen after mechanical oxygen removal methods are completed. Hydrazine reacts with iron oxide and copper oxide to form protective oxide (passive) films. Hydrazine is added through the chemical mixing tank by the CVS. Hydrazine does not pose a threat to nuclear safety due to its infrequent use. In addition, employees involved in the work required to maintain safety functions do not need to be in areas where hydrazine may leak or be released (e.g., in the CVS pump station in the radioactive part of the annex building). Both the lithium hydroxide and a small portion of the hydrazine stockpile are stored in an annex building, which is a reinforced concrete structure of seismic resistance category I. It shares a common pedestal deck with the containment building. The annex building provides a sufficient ventilation area from the main steam and feedwater pipeline shut-off valve compartments in the event of a postulated leak in the main steam or feedwater pipeline.

**Zinc acetate** is created and stored in a zinc injection equipment set that is part of the CVS to inject zinc acetate into the RCS. The addition of soluble zinc acetate to the reactor coolant causes zinc to incorporate into oxide films on the wetted surfaces of reactor coolant components, steam generator tubes, and RCS piping. It reduces the ongoing corrosion of austenitic stainless steel and nickel-based alloys. Zinc acetate is stored in a stainless steel zinc additive tank. The turbine compartment sump serves as a secondary containment for the zinc addition tank.

*Nitrogen* has no toxic effects under atmospheric pressure. The nitrogen injection equipment is located outdoors. The rest of the pipelines located indoors are of all-welded construction with valves without seals. The set of equipment for hydrogen injection is located outdoors. Other pipelines located indoors are of all-welded construction with valves without seals.

The storage places, hazard class, mass of hazardous substances, and class of the arresters are shown in Table 5.37.

## Table 5.37 – Identification of hazardous substances

Production item	Production item Name of hazardous substance/weight, t			Class				
Power unit No. 5 within the protected zone								
Annex building	Zinc acetate solution / 1.32	H3 RANDOM TOXICITY						
	Hydrazine solution / 0.10	Hydrazine 2-naphthylamine		-				
	Lithium hydroxide solution / 0.02	H3 RANDOM TOXICITY						
Building of the turbine room and	Zinc acetate solution / 1.32	Hydrazine 2-naphthylamine		2				
the turbine room	Hydrazine solution / 1.00	H3 RANDOM TOXICITY		3				
Diesel generator building Diesel fuel № 2D / 13.34		Oil products and alternative fuels		-				
Diesel fuel storage tank (DOS-MT- 01A) and diesel fuel pumping unit (DOS-MS-01A)	Diesel fuel № 2D / 313,28	Oil products and alternative fuels		3				
Diesel fuel storage tank (DOS-MT- 01B) and diesel fuel pumping unit (DOS-MS-01B) Diesel fuel № 2D/ 313,28		Oil products and alternative fuels		3				
Boric acid tank Boric acid solution / 306.13		E1 Category 1		1				
	Power unit No. 6 wit	hin the protected zone						
Auxiliary building	Zinc acetate solution / 1.32	H3 RANDOM TOXICITY						
	Hydrazine solution / 0.10	Hydrazine 2-naphthylamine		_				
	Lithium hydroxide solution / 0.02	H3 RANDOM TOXICITY						
Building of the turbine room and	Zinc acetate solution / 1.32	Hydrazine 2-naphthylamine		2				
the turbine room	Hydrazine solution / 1.00	H3 RANDOM TOXICITY		3				
Diesel generator building	Diesel fuel № 2D / 13,34	Oil products and alternative fuels		-				

Production item	Name of hazardous substance/weight, t	Class or individual name of the hazardous substance		Class			
Diesel fuel storage tank (DOS-MT- 01A) and diesel fuel pumping unit (DOS-MS-01A)	Diesel fuel № 2D / 313,28	Oil products and alternative fuels		3			
Diesel fuel storage tank (DOS-MT- 01B) and diesel fuel pumping unit (DOS-MS-01B)	Diesel fuel № 2D / 313,28	Oil products and alternative fuels		3			
Boric acid tank	Boric acid solution / 306.13	E1 Category 1		1			
	Permanent items wit	hin the protected zone					
Nitrogen-oxygen station	Liquid nitrogen / 10.50 Liquid hydrogen / 0.40	E1 Category 1 Hydrogen		3			
Permanent NPP facilities outside the protected zone							
Fire Protection System (FPS) diesel tank	Diesel fuel № 2D / 0,79	Oil products and alternative fuels		-			

According to the materials provided by Westinghouse Electric Company, certain production units were identified as high-risk objects:

• The turbine building and the first bay of the turbine building of unit 5 were identified as Class 3 high-risk objects;

• The diesel fuel storage tank (DOS-MT-01A) and diesel fuel pumping unit (DOS-MS-01A) of unit 5 were identified as high-risk objects of Class 3;

• The diesel fuel storage tank (DOS-MT-01B) and diesel fuel pumping unit (DOS-MS-01B) of unit 5 are identified as high-risk objects of Class 3;

• The boric acid tank of unit 5 is identified as a high-risk object of Class 1;

• The turbine building and the first bay of the turbine building of unit 6 is identified as Class 3 high-risk objects;

• The diesel fuel storage tank (DOS-MT-01A) and diesel fuel pumping unit (DOS-MS-01A) of unit 6 are identified as a high-risk object of Class 3;

• The diesel fuel storage tank (DOS-MT-01B) and diesel fuel pumping unit (DOS-MS-01B) of unit 6 are identified as a high-risk object of Class 3;

- The boric acid tank of unit 6 is identified as a high-risk object of Class 1;
- The nitrogen-oxygen station is identified as a high-risk object of Class 3.

## 5.7.2 Nuclear Fuel Handling

The following stages of nuclear fuel management are envisaged within the planned activities

- storage of new nuclear fuel;
- use of nuclear fuel in nuclear reactors;
- transportation of new and spent nuclear fuel, radioactive materials, and waste;
- storage of spent nuclear fuel, radioactive materials, and waste.

The Fuel Handling and Refueling System (FHS) ensure safe nuclear fuel management. The main function of the Fuel Handling and Refueling System (FHS) is to move and store new and irradiated fuel within the power unit. The FHS system maintains fuel assemblies and active components in a subcritical state during handling and storage under both normal and emergency conditions. In addition, the FHS system is designed to ensure that equipment of seismic resistance category I function after a safe shutdown of the power unit during an MRS. Subcriticality during fuel transfer and storage is ensured by the geometrically safe configuration of the equipment involved in the transfer.



Currently, the design scheme for spent fuel handling based on Holtec International technology is being implemented for Ukrainian NPPs.

The storage site area is designed with the possibility of expansion considering the new generating and/or replacement construction facilities.

## 6 A DESCRIPTION OF THE FORECASTING METHODS USED TO ASSESS ENVIRONMENTAL IMPACTS AND THE ASSUMPTIONS UNDERLYING SUCH FORECASTING, AS WELL AS THE USED ENVIRONMENTAL DATA

To assess the dynamics of environmental indicators, impacts were assessed throughout the entire range of activities related to the work performed at the KhNPP industrial site.

The environmental impact assessment considered the aspects of the activity that may affect the state of resources and components of the natural environment, as well as the consequences associated with the impacts.

To determine the quantitative parameters of possible impacts, calculation methods, and expert assessments based on the results of engineering surveys at the project site were used.

The assessment of radiation factors takes into account the radioactive decay of radionuclides of existing environmental contamination, as well as additional potential contamination as a result of the planned activity.

In cases where there is no reliable initial data required to assess the environmental impact of the planned activity, a conservative approach is used. In particular, the calculations were performed under approved industry standards and regulatory requirements.

In addition, appropriate software tools approved by the SNRIU were used to calculate the possible movement of radioactive contamination in the event of hypothetical maximum design basis accidents.

All available materials characterizing the current state of environmental components in the KhNPP area, monitoring data, results of engineering and other studies of previous years, cartographic materials, and other information were used as sources of information for the EIA.

Forecasting the dynamics of environmental indicators was performed using the EOL 2000 software, an automated system for calculating the dispersion of harmful substances emissions, which is included in the list of programs recommended by the Ministry of Environment of Ukraine. The calculation module of the system is implemented under OND-86 [61], the Methodology for Calculating the Concentrations of Harmful Substances in the air contained in the enterprises' emissions.

The specific information sources used in the preparation of the EIA materials are mentioned in the text as references and are provided in the section "List of References".
### 7 A DESCRIPTION OF THE MEASURES ENVISAGED TO PREVENT, PRECLUDE, AVOID, REDUCE, ELIMINATE A SIGNIFICANT ADVERSE ENVIRONMENTAL IMPACT, INCLUDING COMPENSATORY MEASURES (IF POSSIBLE)

The AP1000 plant has been designed with environmental concerns as a priority. The safety of the public and the power plant workers, as well as the environmental impact has been addressed as follows:

• operational releases and discharges were minimized by design features;

• strict targets for personnel radiation exposure were set and satisfied. The targets are aimed at ensuring maximum employee safety and are implemented in accordance with ALARA principles. The targets are implemented through modern design solutions, high-tech monitoring systems, training programs and strict adherence to international safety standards;

• total radwaste volumes were minimized compared to reactors of previous generations by implementing technological and engineering solutions that increase the efficiency of nuclear fuel use, reduce the amount of waste generated during reactor operation, and optimize waste treatment and storage systems;

• improved safety performance by the introduction of passive safety systems, automation of technological processes and optimization of the number of operating personnel;

• other hazardous waste volumes (non-radioactive) were minimized.

The AP1000 plant safety approach has been specifically designed to maximize the plant robustness against catastrophic events resulting in extensive loss of infrastructure and a common-cause, complete loss of electrical power – both onsite and offsite. Specifically, the AP1000 plant is unique in that the plant response to a complete station blackout is considered as a design basis event and is embedded in the licensing basis.

This fundamental approach is at the basis of the AP1000 robustness against extreme external events.

# 7.1 Description of technical solutions to prevent accidents and localize releases of hazardous substances, ensure fire and explosion safety

The AP1000 is designed to prevent abnormal operation by utilizing its own safety margin and physical features (e.g., larger water volumes and reserves, negative power and temperature coefficients of reactivity).

To reduce the probability of deviations from normal operation that could lead to emergency situations, non-safety-related active systems are provided. These highly reliable active systems are the first level of protection against more likely events, as they are automatically activated and prevent unnecessary operation of passive systems important for safety.

To mitigate design basis accidents, the AP1000 plant is designed with passive safety features as a second level of protection. The passive safety systems do not require operator actions or AC power. These systems use only natural forces, such as gravity, natural circulation, and compressed gas to make the systems work. Simple valves actuate passive safety systems automatically. The valves are energized and held in the closed position under normal operation. An emergency signal or loss of power is required to trigger these valves in an emergency.

Fire safety of KhNPP Units 5 and 6 is ensured by fire prevention and fire protection subsystems, including organizational and technical measures. The specificity of KhNPP fire safety is that the fire safety systems of the facility are targeted and should primarily ensure radiation and nuclear safety of the facility.

The designed fire safety system for KhNPP Units 5 and 6 performs fire prevention and fire protection functions.

Preventing the formation of a flammable environment is ensured by a combination of the following measures:

– maximizing the use of non-combustible and not easily combustible substances and materials;

– limiting the mass of combustible substances and materials that are compactly located, placing them in the safest way;

- isolation of flammable environment (allocation of fire compartments and sections);

- installation of fire hazardous equipment in isolated rooms (fire compartments and sections);

– maximum isolation and automation of the technological process with spent nuclear fuel;

– use of devices to protect production equipment with flammable substances from damage and accidents;

– installation of cutoff, shutdown and other devices (including on ventilation system air ducts).

Preventing the formation of ignition sources in a flammable environment is achieved by a combination of the following measures:

- use of electrical equipment that meets the conditions of operation in explosive and fire hazardous areas in accordance with the Electrical Installation Rules;

- use of fast-acting means of protective disconnection of possible sources of ignition;

- compliance with electrostatic safety requirements;

– lightning protection of buildings and structures;

– eliminating conditions for thermal, chemical and/or microbiological spontaneous combustion of flammable substances.

Fire protection is achieved by a combination of the following measures:

• use of fire extinguishing means and appropriate types of fire fighting equipment;

• application of fire alarm systems and automatic fire extinguishing systems;

• use of non-combustible building structures and materials with standardized fire hazard indicators;

- use of space-planning solutions and devices that limit the spread of fire, such as:
  - division of the building into fire compartments and sections;
  - installation of fire barriers;
  - refractory elements in the equipment;
  - emergency shutdown of equipment, etc.;

• ensuring timely warning and evacuation of personnel through space-planning solutions and technical means;

• use of collective and individual protection means for people (personnel, fire units) against fire hazards and radiation exposure;

• use of technical means of smoke protection.

### 7.2 Justification for the acceptability of the design solution package

Westinghouse's AP1000 technology is the only Generation III+ reactor technology licensed by the U.S. Nuclear Regulatory Commission and in several European and Asian countries. The advantage of AP1000 reactors is a very high installed capacity utilization rate of 93% and an extended fuel cycle of 18 months. The AP1000 plant is a two-loop pressurized water reactor (PWR) that uses a simplified, innovative and effective approach to safety. The AP1000 design offers clear advantages, including high safety, economic competitiveness, and improved and more efficient operation.

The AP1000 reactor design is distinguished by its passive safety systems, record short fuel overload time, and, therefore, high efficiency. In addition, such a powerful reactor occupies a small area.

The safety of NPPs with the AP1000 reactor is ensured by a system of technological and organizational means:

- use of the internal self-protection properties of the reactor unit;
- application of the defence-in-depth concept;

- use of safety systems designed using a single failure principle, diversity, redundancy and physical separation;

- use of approved engineering and technical practices;

- compliance with nuclear and radiation safety regulations, as well as compliance with the requirements set forth in the AP1000 basic design;

- maintaining and improving the safety culture;
- use of the quality management system at all stages of the nuclear facility life cycle;
  - ensuring appropriate training and qualification of personnel;
  - use of operational experience;

The AP1000 plant safety is ensured by the consistent implementation of a defence in depth strategy based on the following:

- preventing deviations from normal operation;

- detecting deviations from normal operation and providing means to prevent such deviations that may lead to emergency conditions;

- providing engineering safety measures to control and mitigate emergency conditions. In addition, the prevention and mitigation of severe accidents is considered through the development and use of probabilistic risk assessment.

The AP1000 design combines proven design concepts and operational experience from existing PWR plants. It is applied to a defined set of functional requirements in the simplest and most efficient way possible.

Compliance with environmental requirements shall be ensured in accordance with the established restrictions: environmental, sanitary and epidemiological, urban planning and fire protection. The limits are set according to a number of criteria:

- internal and external exposure of personnel and the public;
- maximum permissible emissions and discharges to the environment;
- the amount of chemical emissions of harmful substances into the atmosphere;
- concentration of harmful substances in the surface layer of the atmosphere;
- volumes of special water use and discharges;
- area of the sanitary protection area and supervised area.

During the siting, design, construction, operation, lifetime extension and decommissioning of power units, the requirements of legislative acts and applicable environmental regulations are met, i.e., environmental safety (the state of environmental protection against disruption of its ecological balance) shall be ensured.

The rational use of natural resources, compliance with the established requirements for environmental pollution, and compliance with existing agreements will be ensured by implementing the comprehensive measures envisaged by the design of units 5 and 6, which cover all aspects of environmental compliance:

• measures to conserve resources (conservation and rational use of land, water, energy, and fuel resources, reuse of resources);

- protective measures:
  - architectural, construction and planning solutions;
  - measures to reduce radiation impact on the environment;
  - measures to reduce non-radiation impact on the environment;
- restoration measures:
  - technical and biological reclamation;
  - normalization of the state of certain environmental components;
- compensation measures:
  - compensation for inherent damage during normal operation of the power unit;

- assessment of the necessary costs to compensate for damage to the public and the environment in the event of an accident;

• security measures (monitoring systems):

- radiation monitoring system at the NPP site and adjacent territory;

- surface and groundwater monitoring systems;

– monitoring biocenoses and populations of surface water (bioindicators of thermal impact, eutrophication and radionuclide contamination);

- system for monitoring geological processes and soil conditions;

- system for monitoring the condition of the foundations of buildings and structures;

– warning system in the NPP supervised area.

Liquid or solid radioactive waste management and storage is carried out in accordance with the requirements of the General Safety Provisions for Radioactive Waste Management Prior to Disposal.

To detect possible leaks of liquid radioactive media on the NPP territory, the observation wells equipped with water sampling tools for monitoring are provided.

The NPP is a source of environmental pollution, the radioactive impact of which is manifested only in the event of beyond design basis accidents.

## 7.3 **Protection of the environment from ionizing radiation**

During plant operation, the following measures and devices ensure environmental protection against ionizing radiation:

- localization barriers in accordance with the defence in depth principle;
- creation of closed loops with radioactive media;
- arrangement of primary pressurized systems within the containment;
- creation of intermediate cooling water circuits;
- division of production rooms into restricted and normally occupied areas;
- division of ventilation systems of restricted and normally occupied areas;
- creation of an organized collection and treatment of radioactive leaks;
- creation of an organized collection of liquid and solid radioactive waste;
- storage and processing of waste in a special building;

– maintaining radiation and climatic conditions in production rooms by means of ventilation systems;

– accident localization system.

Protective measures against the release of radioactive substances into the atmosphere, surface and groundwater, soil and food chains are ensured as follows:

• a special containment system that prevents the spread of radioactive substances into the environment;

• an air purification and removal system that provides for the following basic measures:

- the exhaust air containing radioactive isotopes is purified using aerosol and iodine filters;

- separate measurements are performed to determine the presence of inert gases, aerosols and iodine;

- the ventilation system maintains the pressure in the containment area below atmospheric pressure, which ensures that in the event of any leakage from the primary circuit, air from the containment area passes through the filters with constant monitoring of its activity;

- cleaning of process blowdowns is carried out on filters where most of the radioactive isotopes of xenon and krypton decay;

- organized air emission from the restricted area premises of the nuclear island and the radioactive waste management building (carried out through ventilation pipes that ensure the necessary dispersion of emitted radionuclides in the atmospheric air to concentrations below permissible levels);

• establishment of sanitary and protection area;

• emission control system in each ventilation pipe, which is carried out continuously using a centralized system, as well as individual devices;

• control of the concentration of radioactive iodine aerosols before and after filters of ventilation systems.

The Containment Air Filtration System (VFS) [62] provides the following functions to limit the exposure of plant personnel and the general public to airborne radioactivity:

• the VFS purges the containment atmosphere of airborne contamination during normal operation to an acceptable radiation level;

• the VFS purges the containment atmosphere to reduce potential airborne contamination during cold shutdown for personnel protection;

• the VFS treats the containment exhaust air to maintain normal gaseous radiological releases at safe levels when combined with gaseous discharges from other sources;

• the VFS monitors the radioactive effluents discharged to the environment through the plant vent;

• the plant vent minimizes the re-entry of effluents into the HVAC fresh air intakes.

The Gaseous Radwaste System (WGS) receives, processes, and discharges the radioactive waste gases received within acceptable off-site release limits during normal modes of plant operation including power generation, shutdown and refueling [63]. The WGS is a once-through ambient temperature activated carbon delay system.

The WGS is designed to eliminate sources of hydrogen ignition. Discharge to the VFS is downstream of the Radiologically Controlled Area Ventilation System (VAS) exhaust fans to provide additional protection against hydrogen ignition.

The Radwaste Building HVAC System function is to maintain suitable environmental conditions for equipment and personnel working in the area. The specific functions of the system are:

– provide conditioned air to all work areas to maintain acceptable temperatures for equipment and personnel working in the areas;

- ensure that air movement is from clean to potentially contaminated areas to minimize the spread of airborne contaminants;

- collect the vented discharges from all potentially contaminated mobile equipment;

- radiation monitoring by the Radiation Monitoring System (RMS) of all potentially contaminated exhaust air prior to release to the plant vent;

– maintain the Radwaste Building at a negative pressure with respect to outdoors to prevent unmonitored releases from the Radwaste Building;

- provide once-through ventilation of the Radwaste Building with no recirculation of potentially contaminated air.

To protect groundwater from contamination, units 5 and 6 have a liquid radwaste system (WLS) designed to monitor, collect, process, treat, store and dispose the liquid radwastes generated during normal operation.

To prevent a water-borne activity discharge, the containment isolation valves, piping and electrical penetrations, hatches, and the containment are installed with the following safety functions:

• automatic isolation of pipelines crossing the containment is necessary to ensure the containment integrity;

• leakage of the containment, penetrations and isolation valves is less than the design leakage at the design containment pressure, which meets the requirements for leakage tests [64].

The WLS performs the following safety-related functions:

– automatic isolation of WLS lines that penetrate the Containment Building;

- overpressure protection of containment penetrations.

The WLS provides these non-safety related functions:

- to receive and process Reactor Coolant System (RCS) effluent;

to receive and process radioactive equipment drains, including those from the
WLS:

WLD;

- to receive and process radioactive floor drains, including those from the WLS;

- to receive and accumulate radioactive hand-wash and shower-drain wastes from the Radioactive Waste Drain System (WRS);

to receive and accumulate radioactive chemical wastes from the WLS;

- to receive and process water from the spent fuel handling subsystem and the Solid Radwaste System;

- to receive and process water from relief-valve discharges into the containment, including those from the Normal Residual Heat Removal System (RNS), Chemical and Volume Control System (CVS), and Passive Core Cooling System (PXS);

- to transfer any of the chemically contaminated influents to the WLS to a mobile equipment processing facility in the Radwaste Building. After treatment by mobile equipment, the WLS may receive the treated water for additional processing or discharge;

to maintain a separation between incompatible waste streams.

Radiation protection of the operating personnel and contamination aversion of the environment in the SRW system is ensured by:

- special equipment for SRW handling (containers, vehicles, etc.);
- mechanical means for transfering operations;
- radiation monitoring and decontamination equipment.

All SRW transport and processing operations are performed with radiation monitoring to ensure radiation safety of the plant personnel.

All shields designed to protect against the release of radionuclides into the environment are monitored by the NPP's RM system.

## 7.4 Measures to reduce non-radiation environmental impact

A nuclear power plant with AP1000 provides features and capabilities to minimize emissions and related environmental impacts. NPP emissions can be divided into two categories: liquid and gaseous emissions.

All chemical emissions into the atmosphere are periodic and do not lead to a violation of the regulatory state of the atmosphere surface layers and ensure the lowest possible level of impact. Thus, the design does not provide for special measures to reduce air pollutant emissions.

The following measures will be taken to prevent or reduce the flow of pollutants into the aquatic environment, as well as to protect surface waters from depletion during the construction and operation of the facility:

• the NPP has a reverse process water supply system designed to minimize the environmental impact on surface water bodies of public use;

• it is a designed equipment cooling system of power units 5 and 6 with cooling towers at the NPP. It allows to reduce the volume of water taken from the reservoir and use one only for feeding the Service Water System (SWS), Circulating Water System (CWS) and cooling tower blowdown;

• all NPP wastewater after treatment is directed to the cooling pond and is not discharged directly into public water bodies;

• the NPP cooling pond is a regulating reservoir that allows water intake from the Goryn River only during high-water periods and, accordingly, does not affect the water management situation in the Goryn River basin;

• all NPP reclaimed water entering the cooling pond are subject to hydrochemical control;

• in case of exceeding the level of mineralization and concentration of biogenic elements in the cooling pond, it shall be purged within the limits of their maximum permissible concentrations (MPC) established by sanitary standards;

• to comply with the temperature conditions in the control point stipulated by the sanitary standards by diluting the blowdown water, which is controlled by appropriate water temperature measurements;

• to monitor the hydrobiological state of the KhNPP technical and ecological system, to study the conditions of existence of aquatic communities, their structure and quantitative indicators in the background water bodies (Hnilyi Rih River, Horyn River);

• the laboratory of the KhNPP subdivision monitors the chemical composition and water temperature at sampling points in the cooling pond, inlet and outlet channels;

• compliance with the requirements of the current legislation on the protection and rational use of surface water.

In addition, the following measures will be taken to prevent or reduce the flow of pollutants into the water environment:

• use of water for drinking purposes should be carried out only upon confirmation of compliance of water quality with the requirements [20];

• keeping regular records of water withdrawn in the logs of the established form, its quality and water level depth in the wells;

• maintaining water meters for water accounting, crane devices for water sampling for laboratory analysis, compliance with the requirements of the sanitary mode of the groundwater protection areas [65];

• monitoring of groundwater regime (level, temperature, chemical composition);

• compliance with the requirements of the current legislation on the protection and rational use of groundwater.

The noise impact from sources located on the industrial site on facilities outside the NPP sanitary protection area is within normal limits.

Chemicals used in the AP1000 plant are stored on the nuclear island, Turbine Island, and in the yard.

The chemicals are used for a variety of purposes and are applied to the primary side, secondary side, passive cooling systems, closed cooling water systems, and other AP1000 applications [59].

Many of these chemicals are stored in Chemical Storage Area in the Turbine Building. To prevent any leakages from the Turbine Building to the environment, the area of potential releases are diked with sufficient volume to contain the leaked volume of the chemical. For chemicals stored in the Service Water System (SWS) Chemical Treatment Building, there are individual prefabricated cubicles with individual containments to separate chemicals and their vapors to avoid chemical interactions [59].

The Chemical and Volume Control System (CVS) provides the means for adding chemicals to the RCS.

The boric acid storage tank (BAST) supplies borated water to the RCS. The BAST is contained in a separate concrete floored and walled diked area with sufficient volume to prevent spilling the acid to the environment. The Annex/Auxiliary Building Sump will manage releases from the Boric Acid Batching Tank and Chemical Mixing Tank (9.5 m³ of volume). Discharge of the sump is to the Waste Water System (WWS) [59].

Therefore, any potential discharges are either through the WWS or the WLS liquid discharge release point.

Diesel Oil Fuel is used for standby diesel generators, ancillary diesel generators, and the diesel driven fire protection pump. Diesel fuel is stored on the AP1000 plant site.

The Standby Diesel Fuel Oil System (DOS) design conforms to the requirements [59]. The major components in the DOS for the diesel engines are enclosed in separate, heated, fan ventilated, skid mounted, modularized transfer skids enclosures in the yard. The fuel transfer skid has a lip around the perimeter to retain diesel fuel spills. The skid has four floor drains to catch fuel oil spills. The diesel fuel from the drains is routed to the WWS for its separation and processing [59].

All buried DOS diesel fuel transfer piping is protected and enclosed within a corrosion resistant containment spill protection piping system. The protection pipe has a leak detection monitoring and alarm system [59, 66].

Each DOS Diesel Fuel Storage Tank is contained in a separate reinforced concrete floored and walled diked area with sufficient volume to contain the entire tank contents of an accidental leak or failure in order to prevent contamination of the underground aquifer. Each diesel fuel tank area includes a sump area for water collection and drainage to the WWS. The dike is designed to be capable of capturing the maximum probable precipitation rainfall for testing of the rainwater contained in the dike before discharge [59].

For each tank, the design objective is to contain the leak. Therefore, the only release point is via the WWS as the sumps are discharged to the WWS. The WWS provides separation of diesel fuel from the sump discharges before release from the plant outfall [59].

The equipment and floor drainage systems collect liquid wastes from equipment and floor drains during normal operation, startup, shutdown, and refueling [67]. The liquid wastes are then transferred to appropriate processing and disposal systems.

Equipment and floor drainage is segregated according to the type of waste. Liquid wastes are classified and segregated for collection as follows:

- radioactive liquid waste
- nonradioactive liquid waste
- chemical and detergent liquid waste
- oily liquid waste

#### Nonradioactive and Potentially Radioactive Waste Drains

The waste water system collects nonradioactive waste from floor and equipment drains in auxiliary, annex, turbine, and diesel generator building sumps or tanks. Selected nonradioactive liquid wastes and tanks are monitored for radioactivity to determine the liquid wastes have not been inadvertently contaminated. If contaminated, the wastes are diverted to the liquid radwaste system for processing and ultimate disposal.

#### **Chemical Waste Drains**

The radioactive waste drain system collects chemical wastes from the auxiliary building chemical laboratory and decontamination solution drains from the annex building and directs these wastes to the chemical waste tank of the liquid radwaste system.

#### **Detergent Waste Drains**

The laundry and respirator cleaning functions that generate detergent wastes are performed offsite. Detergent wastes from hot sinks and showers are routed to the chemical waste tank.

#### **Oily Waste Drains**

The waste water system collects nonradioactive, oily, liquid waste in drain tanks and sumps. Drain tank and sump liquid wastes are pumped through an oil separator prior to further processing. The oil is collected in a tank for disposal.

Sampling for oil in the waste holdup tank of the liquid radwaste system is provided to detect oil contamination before the ion exchanger resins are damaged. Oily water is pumped from the tank through an oil adsorbing bag filter before further processing. The spent bag filters are transferred to drums and stored in the radwaste building.

# 7.5 Comprehensive measures to reduce negative impact on the air environment

The project provides for the following measures to prevent negative impact on the air environment and reduce pollutant emissions into the atmosphere:

- preventive measures - planning, standardization, regulation, design of construction and reconstruction of factories and other facilities that impact or may impact the air quality, establishment of a sanitary protection zone, sanitary-hygienic expertise, governmental accounting, monitoring, etc.;

– implementation of measures to prevent and reduce air pollution by transport emissions and other mobile vehicles and facilities and their physical factors impact. They include development and implementation of a set of measures to reduce emissions, neutralize harmful substances and physical impact during the design, manufacture, commissioning and repair of transport and other vehicles and facilities; conversion of transport and other vehicles and facilities into less toxic fuels; improving the maintenance of transportation routes; improving fuel transportation and storage technologies;

– implementation of legal measures – regulating the implementation of preventive measures against its contamination by industrial, household and other releases. It is determined that to warehouse, place, store or transport the industrial and household release that is a source of air contamination with pollutants and substances with an unpleasant smell or other harmful effects, is allowed only with a special authorization in the territories designated by local executive authorities and local self-government bodies, in compliance with environmental safety standards and subsequent utilization or disposal. It is prohibited to incinerate such wastes on the enterprise territory, except when it is

carried out using special facilities under the requirements established by the legislation on atmospheric air protection. The owners or the authorized bodies are obliged to ensure the processing, utilization and timely release transportation that pollutes the air to enterprises that use it as raw materials or to specially allocated places or facilities;

- implementation of security measures – preventing, reducing and achieving safe levels of industrial and other noise. They should be ensured by creating and implementing low-noise machines and mechanisms; improving the transport design and other mobile vehicles, facilities and their operating conditions, as well as maintaining railways, roads, pavement, etc. in proper condition;

– implementation of preventive measures – regulating the conditions for designing, building and reconstructing an enterprise. They are carried out with obligatory compliance with environmental safety standards, government sanitary requirements and rules at the facilities planned for this purpose, as well as taking into account the accumulation and pollution transformation in the atmosphere, its transboundary transfer, and the specifics of climatic conditions;

– approval of development projects for construction and reconstruction of enterprises and other facilities, that impact or may impact the atmospheric air is carried out by a specially authorized executive body for urban planning and architecture, under the conclusions of environmental, sanitary and other executive authorities or local self-government bodies in accordance with their powers as defined by law. The construction and commissioning of new and reconstructed enterprises and other facilities is prohibited if it does not meet the requirements for air protection prescribed by law;

- use of equipment with high environmental characteristics and systems for emissions purification from harmful components, application of the control measurement devices and automatic system for reliable protection of operating equipment, control over the density of all connections.

During the equipment operation, a set of organizational and technical measures is envisaged to reduce pollutant emissions into the air:

- compliance of technological regulations and fire safety requirements;
- maintaining complete technical readiness of equipment and pipeline integrity;
- maintaining equipment in proper operating condition;
- the systematic valves tightness monitoring, fittings and pipeline connections;

- compliance of the prescribed standards of permissible limit release for contaminants into the air (performed by specialized services using the instruments equipped in the company's laboratory);

- regulating pollutant emission into the atmosphere during periods of adverse meteorological conditions (AMC).

#### 7.6 Remediation measures

### 7.6.1 Technical and biological reclamation

The land of the State Forestry Fund and marginal agricultural land were allocated for the construction of the KhNPP. Land was alienated for the KhNPP construction. The reclamation activities were completed by the time unit 1 was commissioned.

Soil degradation processes associated with the construction of NPP units are common only in the industrial site area. Their presence in the KhNPP SA is practically not related to the plant operation.

Given the absence of additional environmental contamination with radionuclides during normal operation of the plant, special agrotechnical measures with changes in the structure of agricultural land use, re-profiling of agricultural sectors and changes in technological processing of products are considered inappropriate.

The need for special measures may be caused only by exceeding the established levels of radionuclide contamination of agricultural products, which may exceed the dose limit established for the public. According to the assessment, the additional contribution to contamination of various agricultural products with long-lived radionuclides due to gas and aerosol emissions is lower than the existing contamination, which, in turn, is significantly lower than the established permissible levels.

In the considered emergency situations, contamination of certain territories is possible at levels that will determine the agricultural products produced on these lands as not meeting the established permissible levels. In this case, countermeasures may be applied, the decision to take which is based on an analysis of the actual situation. The procedure for applying countermeasures is regulated by DGN 6.6.1.-6.5.001-98. Countermeasures are used not to reduce the impact on the soil, but to reduce the dose load on the population.

It is advisable to use mushrooms, as well as pine, blueberries, mosses and lichens (for each of the tiers) as bioindicators of contamination, for which there is a sufficient database and the corresponding dependencies have been established.

#### 7.6.2 Normalization of vegetation, fauna and nature reserves

Anthropogenic impacts vary in scale and impact under the current KhNPP operating mode. The main factors that change and reduce the resilience of plant communities are the following: plowing, felling, grazing, haymaking, recreation, land reclamation, construction, quarrying (removal of soil and geological rocks), artificial plantations that do not meet environmental conditions, fires, etc.

A comprehensive assessment of the impact of all these factors shows significant changes in vegetation cover and destruction of ecosystems.

The commissioning of new power units, further development of infrastructure, and an increase in the number of service personnel will lead to a certain increase in the impact on natural complexes and vegetation, but it is possible to limit it by creating new protected areas, monitoring and control.

The KhNPP SA contains 47 nature reserve fund objects with different protection degrees, the area of which is more than 3000 hectares. This is slightly more than 1% of

the area territory, which is four times lower than the national indicator. Protected objects belong to four categories: nature reserves (of national and local importance), natural monuments (of national and local importance), protected tracts, and park and garden art monuments. Of the 47 nature reserve fund objects, 7 are of national importance, and the remaining 40 are of local importance.

Forest management should adhere to the optimal ratio of natural and transformed community types, which will stabilize the situation in the context of the normal operation of new power units.

According to the Decree of the President of Ukraine, a national nature park "Male Polissia" with an area of 8762.7 hectares is being created in Slavuta and Iziaslav districts of Khmelnytskyy region.

The boundaries of the national park are conditionally laid out by river valleys and the KhNPP cooling pond. In the north - the Goryn River and the cooling pond; in the east - the Goryn River; in the northwest - the Vilia River; in the south - the tributaries of the Goryn River and the Vilia River. Most of the southern and southeastern part of the nearer supervised area will be included in this National Park. The creation of the park will help preserve the unique natural resources of the region.

There is every reason to believe that the regional structure of natural ecosystems will be preserved.

## 7.7 Compensation measures

The legislation of Ukraine provides for economic measures to stimulate activities aimed at reducing the environmental impact, namely:

• limits on the use of natural resources and pollutant emission standards;

• setting environmental tax rates for environmental pollution and the use of natural resources;

• compensation in accordance with the established procedure for damages caused to the environment as a result of violation of environmental legislation.

The above payments, which are regularly made by KhNPP during the operational period, go to the state and local budgets and can be used to finance measures related to environmental protection and damage compensation, depending on the budget allocation at the state or regional level.

Compensation measures in the social sphere include the following aspects:

• increase in electricity generation for the needs of the population, utilities and industrial enterprises;

• new jobs for construction and installation personnel and engineering and technical personnel, additional jobs for the population both during the construction of the NPP and during operation;

• resettlement of additional construction and installation personnel and then operational personnel near the NPP under construction will stimulate the need to expand the scope of social services that also cover the local population.

Fish farming of herbivorous fish has been organized in the cooling pond area to prevent its siltation.

The additional levels of contamination of the soil under normal operation of the plant are tens of thousands of times lower than the existing background contamination. Given the fact that radionuclides enter the soil in ultra-micro quantities, they cannot affect soil fertility in any way. Their content in soils is much lower than the content of nutrients and they cannot compete with them when it comes to plant growth. The radionuclides in soils also do not affect the quality of the products and yields. Therefore, under normal operation of the plant, it is not advisable to take special agrotechnical measures aimed at reducing the impact on the soil. Thus, no damage will be caused by contamination of land and agricultural products under normal operating conditions of KhNPP Units 5 and 6.

Atmospheric air and land resources will not experience anthropogenic load from the operation of KhNPP Units 5 and 6.

#### 8 DESCRIPTION OF THE EXPECTED SIGNIFICANT ADVERSE ENVIRONMENTAL ACTIVITY IMPACT CAUSED BY THE PROJECT'S VULNERABILITY TO DISASTER RISKS, MEASURES TO PREVENT OR MITIGATE THE DISASTER IMPACT ON THE ENVIRONMENT AND EMERGENCY RESPONSE MEASURES

Following the IAEA recommendations set out in the document "Basic Safety Principles for Nuclear Power Plants" INSAG 12, the goals and obligations of Energoatom to ensure and maintain the safety of nuclear facilities are defined in the Safety Policy Statement of Energoatom, which is available on the official website of the company.

Protecting people and the environment from radiation exposure during the construction, commissioning, operation, and decommissioning of nuclear facilities is the highest priority of Energoatom's activities.

Energoatom is entirely responsible for the nuclear and radiation safety of nuclear facilities at all life cycle stages and gives unconditional priority to safety over other goals.

The main objectives of Energoatom are to maintain and improve the design safety level of nuclear facilities based on the obligations under the Convention on Nuclear Safety, the results of scientific and technical research and development, operational experience, as well as the implementation of a new effective program for the protection of nuclear facilities from hazardous impacts, including ones caused by military aggression.

Energoatom ensures continuous monitoring and analysis of the safety of nuclear facilities.

Energoatom maintains constant communication with the public, informing them about the safety status and activities to improve safety.

#### 8.1 AP1000 design

The AP1000 plant has been designed with environmental requirements as a priority. The public safety and power plant employees, as well as the environmental impact, are addressed in this way:

• Minimized Operational emissions and discharges through design features.

• Set and met strict targets for personnel radiation exposure. These targets are aimed at ensuring maximum employee safety and are implemented in accordance with the ALARA principles. The targets are implemented through modern design solutions, high-tech monitoring systems, training programs and strict adherence to international safety standards.

• The reduced total amount of radioactive waste compared to previous generations of reactors.

• Other minimized hazardous waste (non-radioactive).

AP1000's approach to safety has been specifically designed to maximize the nuclear power plant's resistance to disaster events that result in major infrastructure damage and total loss of power from a common cause, both on and off-site. In particular, the AP1000 plant is unique in that a complete power outage is considered a design event and is built into the design. The fundamental approach that underpins the AP1000's resistance to extreme external events (such as Fukushima): the AP1000 was designed with passive safety systems to eliminate reliance on additional support systems, thereby creating a safer design and more independent:

1. In the case of a power outage, critical Structures, Systems, and Components (SSCs) automatically achieve a fail-safe configuration without the need for operator action or AC/DC power.

2. The AP1000's passive approach to safety reduces the importance of the AC power supply and cooling by providing long recovery periods from events, that would result in a prolonged plant outage and/or prolonged loss of connection to the final heat sink.

3. The structures, systems, and components (SSCs) critical to bringing the reactor to a safe shutdown state are protected within a steel containment and surrounded by a robust reinforced concrete composite shield. The CS building is designed following the latest US NRC, EUR, and WENRA regulations to withstand commercial aircraft impact.

The facilities of KhNPP Units 5 and 6 are based on the design configuration of the reference NPP. The reference NPP is the Vogtle NPP with two AP1000 units 3 &4 located in Burke County near Waynesboro, Georgia in the southeastern United States. The design life of the AP1000 power unit is 60 years.

The American AP1000 technology is the only Generation III+ reactor technology that has received a license for construction and operation from the U.S. Nuclear Regulatory Commission (U.S. NRC).

Today, NPPs with AP1000 reactor units are safely and reliably operated in China and the United States (6 units in total):

– Sanmen NPP Unit 1 in China began commercial operation in September 2018. It became the first power unit with AP1000, which was built and commissioned using this technology for the first time.

– Haiyang-1 and Sanmen-2 were commissioned in November 2018. The Haiyang-2 power unit began commercial operation in January 2019.

- In the United States, two Vogtle NPP units were commissioned - No. 3 (in July 2023) and No. 4 (in April 2024).

The launch, commissioning, and operation of AP1000 units demonstrated the effectiveness of the inherent design features of passive safety systems. Moreover, tests of AP1000 power units proved the availability of sufficient reliability reserves and high power generation performance.

The AP1000 design has been developed under a set of US codes and standards, the vast majority of which are recognized as international best practices and are recognized

in EU countries. Therefore, to ensure the implementation of the AP1000 project, the operating organization is already coordinating with the regulatory authority the regulatory framework that will be used for design, fabrication, installation, normative regulation plan (NRP), and licensing. This task consists of two fields:

- acceptance of construction and general industrial standards of the United States in the form of State Standards, ISO, and Departmental Building Codes (DBC);

- resolving the issue of applying NUREG, ANS, ANSI, and ASME norms,) rules and standards and using them in licensing and nuclear and radiation safety reviews.

Licensing of certain stages of the nuclear facility life cycle (construction and commissioning, operation, and decommissioning) will be carried out following the national legislation.

## 8.2 Probabilistic Safety Assessment

The probabilistic safety analysis (PSA) for AP1000 reactor units was performed in accordance with ASME/ANSI standards approved by the NRC. The PSA models are recognized as meeting the NRC PSA safety objectives and relevant regulatory guidance. PSA models focus on the magnitude and outcomes of internal initial events, internal flood, internal fire, seismic hazards, and other external hazards. The PSA also describes the defenth in depth model during shutdown and the process of maintaining the proper condition.

Probabilistic safety analysis for power units is a comprehensive analysis that includes full-scale Level 1 and Level 2 "at power" PSA.

Level 1 analysis includes:

- assessment of internal initial events;
- analysis of the event tree and success criteria;
- analysis of plant systems using fault tree models;
- analysis of common cause failures and human reliability;
- data analysis;
- quantitative calculation of the CDF.

Level 2 analysis includes:

- assessment of severe accident phenomena and fission product source conditions;
- modeling of the containment event tree and associated success criteria;
- analysis of hydrogen combustion and mixing;
- quantitative calculation of the CDF.

The PSA results indicate that the AP1000 design meets the higher expectations and goals for the next generation of passive pressurized water reactors (PWRs).

The NF severe damage frequency in the core and maximum accidental release frequency (MARF) for internal events during power operation (excluding seismic events,

fires, and floods) are 3,94E-07 one-reactor events per year and 3,83E-08 one-reactor events per year (according to NP 306.2.245-2024, such criteria should not exceed 1E-6 per reactor per year and 1E-7 per power unit per year). These frequencies are at least two orders of magnitude lower than for typically operated pressurized water reactors. The risk reduction is attributed to many plant design features, with the dominant reduction coming from highly reliable and redundant passive safety systems that address both reactor operation and shutdown risks. These passive systems are much less dependent on operator actions and auxiliary systems than the factory systems at currently operating plants.

Conservative, limit estimates of fires and floods show that the reactor core damage risk due to these events is small compared to the reactor core damage risk due to events related to power line accidents and outages.

A summary of the PSA's conclusions on the AP1000 construction:

• The AP1000 design benefits from the high level of redundancy and diversity of passive safety systems. Passive systems have proven highly reliable; the design is simple, requiring a limited number of components to function.

• The AP1000 is less reliant on non-safety-related systems than modern plants or advanced evolutionary light water reactors.

• Non-safety support systems (AC power, cooling water for components, process water, and air for instrumentation) play a limited role in the plant's risk profile because passive safety systems do not require cooling water or AC power.

• The AP1000 is less reliant on human actions than modern plants or advanced light water reactors in the process of evolution.

• The core damage frequency and the limited accidental release frequency are low, despite the conservative assumptions made when defining the success criteria for passive systems.

• Individual systems failures or components are not too important due to the redundancy and diversity of safety-related systems.

• Passive safety systems are available in all shutdown modes. Scheduled maintenance of passive functions is only performed during shutdown modes when the function is not important for risk. In addition, scheduled maintenance of the non-safety-related advanced protection functions used during shutdown is performed during power-up.

• The passive cooling construction of the AP1000's containment vessel is very reliable. Air cooling is of great importance and can prevent containment destruction, although the construction has other defense lines to cool the containment, such as fan coolers and passive water cooling of the vessel.

• The potential for containment bypass is reduced due to fewer pinholes that could lead to fission product leakage.

• The lower part of the reactor vessel has no through-holes, which eliminates through-hole failure as a potential cause of reactor vessel failure. Preventing the movement of molten core fragments into the containment eliminates several severe accidents, such as fuel-coolant interaction outside the reactor vessel and core-concrete interaction, which could threaten the integrity of the containment. Therefore, AP1000, by

preventing the movement of core debris into the containment, significantly reduces the containment failure probability.

• The potential for fires and flooding to spread to safety-related equipment is significantly reduced by the AP1000 design.

The PSA results for internal events that initiate the emergency at the power plant demonstrate that the AP1000 containment construction is reliable in preventing releases following a severe accident and that the risk to the public from severe accidents for the AP1000 is very low. The frequency of the limited accidental release (containment failure rate) at AP1000 can be divided into two types of failures:

1) the defective containment, in which the containment integrity is either compromised due to the initiating event or has never been achieved since the beginning of the accident;

2) containment failure caused by high-energy phenomena of a severe accident. The sum of these failures is the total frequency of the limited accident release.

The quantitative assessment results of the event tree in the containment structure for the frequency of the limited accidental release are important:

• the total frequency of the limited accidental release for the AP1000 is 1.95E-08 events per year. This is approximately 8 percent of the reactor containment damage frequency for internal initial power-up events. The ability of the containment to prevent releases (i.e., containment effectiveness) is 92 percent.

• the Level 3 analysis shows that the risk to the public is small and within acceptable limits.

The PRA's results in accordance with the AP1000 Design Control Document for Units 5 and 6, namely the nuclear fuel severe damage frequency in the core and the frequency of accidental releases, are presented in Table 8.1. The results of the PSA models demonstrate that the ones and results meet the safety objectives of the NRC PRA. In addition, for comparison, Table 8.1 shows the assessment of the fulfillment of the safety criteria for KhNPP Units 1 and 2.

Table 8.1 – Results of probabilistic risk assessment of AP1000 and WWER-1000 for internal events

	Core damage frequency per reactor per year	Limited accidental release frequency per reactor per year
Power unit No. 5 (AP 1000)	3,94E-07	3,83E-08
Power unit No. 6 (AP 1000)	3,94E-07	3,83E-08
Power unit No. 1 (WWER -1000)	4,08E-06	1,62E-06
Power unit No. 2 (WWER -1000)	4,12E-06	9,10E-07

Based on the summary results of the Probabilistic Risk Assessment of the AP1000 unit, it can be concluded that the core damage frequency and the frequency of the limited

accidental release of the AP1000 unit meet the safety criteria for NPP units by the Basic Safety Principles for Nuclear Power Plants.

# **8.3** Maximum design basis accident. Beyond design basis accident (design extension conditions)

Power unit condition categories are defined in accordance with ANSI N18.2, Nuclear Safety Criteria for the Design of Stationary PWR Plants, which distinguish between four different reactor plant conditions depending on their frequency of occurrence and the potential impact of their consequences on the public.

There are four categories:

- State I Normal operation and transients
- State II Medium frequency failures
- State III Infrequent failures
- State IV Limit failures

The main principle underlying the ratio of design requirements to each condition is that the most frequent conditions should have the lowest potential risks associated with the release of radioactive substances, and those extreme situations that pose the greatest risks to the public should be the least likely to occur.

The maximum design basis and beyond design basis accidents are characterized by events of state IV.

State IV events are faults that are not expected but are postulated because their consequences include the potential to release significant amounts of radioactive material. These faults represent design limit cases. State IV faults shall not cause the release of fission products to the environment, resulting in doses to personnel and the public exceeding regulatory values. A single Condition IV event should not cause an indirect loss of essential functions of systems required to eliminate the malfunction, including emergency core cooling systems and containment. The following faults are classified in this category:

- steam pipeline failure (major);
- feedwater system pipeline break;
- GCS shaft jamming (blocked rotor);
- the HPF shaft breakage;
- a range of accidents with the release of the SCS body;
- steam generator tube rupture;

• LOCA'resulting of some alleged ruptures of the first circuit pipelines (large rupture);

• fuel handling accidents.

Based on the results of the analysis of relevant accidents in the category – the State IV events, it was determined that the most serious radiological consequences arising from

a loss of coolant accident due to a series of suspected ruptures in the primary pipelines (Loss of Coolant Accident - LOCA).

A LOCA is the result of a pipe rupture at the pressure limit of the reactor coolant system. A large pipe rupture is defined as a rupture with a total cross-sectional area equal to or greater than 1.0 ft2 (0.3 m2). This event is considered a State IV (Limit Failure) event because it is not expected to occur during the plant life cycle, but is postulated as a conservative design basis.

Even though the analysis of the AP1000 core during a loss-of-coolant accident (LOCA) shows that the integrity of the core is preserved, it is assumed that major core degradation and melting occurred to assess the radiological consequences of the accident. A core melt accident with a large loss of coolant (LOCA) is the most severe accident scenario considered in the framework of beyond design basis accidents.

The accident conditions model is based on NUREG-1465 [68] and Regulatory Guide 1.183 [69]. The releases from a core melt accident with a large loss of coolant (LOCA) result in design basis accidents.

A design basis accident is an accident for which the design defines initial events and final states and provides for safety systems that ensure, taking into account the principle of a single failure of the safety system or one additional personnel error, limiting its consequences to the established limits.

Maximum design basis accident is a design basis accident with the most severe radiation consequences.

Design extension conditions are conditions caused by initial events not considered as part of a design basis accident, in particular, the expected probability of occurrence of which is lower than that taken into account for design basis accidents, or the course (development) of which is accompanied by additional failures of safety systems or human errors compared to design basis accidents. Design extension conditions are divided into two categories: category A, which includes design extension conditions without severe nuclear fuel damage, and category B, which includes accidents with severe damage to nuclear fuel (severe accidents).

Radiation accidents of this class result in the release of radioactive substances into the environment and contamination of the natural environment, including agricultural land. The degree of contamination of the territory adjacent to the plant depends primarily on the amount of radionuclides released into the environment and the nature of their dispersion in the atmosphere. Unfavorable weather conditions are those under which there is minimal dispersion of radioactive substances in the atmosphere. Under such conditions, contamination of a small area with very large density gradients occurs.

The sources of radioactive substances entering agroecosystems in case of accidents are primarily determined by the radionuclide composition of the release, the structure of agricultural land near the plant, and the season.

In the case of an accident, the following radioactive mixture of fission products is released into the environment: radioactive noble gases (RNG), radioisotopes of iodine,

¹³⁷Cs, ⁹⁰Sr. RBGs do not participate in migration through agrobiocenoses due to their inertness. Short-lived radioisotopes of iodine play the main role in the contamination of agricultural products at the early stage of the accident (several months). It is at the early stage of the accident that the peculiarities of the behavior of accidentally released radionuclides in agrobiocenoses are manifested. At further stages of the accident, long-lived radionuclides 137Cs and 90Sr will play a leading role in the product contamination.

The maximum contamination of agricultural products may occur in the case of an accident during the growing season. Most of the radionuclides falling from the atmosphere during the passage of a cloud are retained on the surface of plants, while the rest fall directly on the soil cover. During harvesting or grazing on contaminated pastures, radionuclides (a certain amount of those deposited on the soil surface) are transferred to agricultural products. The pasture-animal-milk chain is a particularly important way for iodine radioisotopes to migrate at the early accident stage. If the radioactive fallout occurs during the grazing period, iodine radionuclides are quickly included in trophic migration chains when animals consume pasture vegetation contaminated by aerial means. The first weeks after the accident are particularly acute in terms of radiation. The duration of the period of predominantly foliar contamination is determined by the first growing season after the accident.

During the next growing season after the accident, the main mechanism of radionuclides migration through agricultural ways is the root uptake of radionuclides from the soil by plants. In the case of such a design basis accident, these radionuclides will be long-lived 137Cs and 90Sr, and the duration of this period may cover tens of years. Root uptake of radionuclides into plants is significantly less than aerial contamination of plants due to the sorption of radioactive substances by soil.

All major types of agricultural land are represented in the land use structure around KhNPP. Therefore, the main migration path of radioactive fallout in agrobiocenoses in the event of hypothetical accidents will be determined by their initial dispersion in the atmosphere (wind direction, atmospheric condition, precipitation).

The critical source of radionuclides in agricultural products during probable accidents will be meadows and pastures located in the floodplain of the Goryn River. Therefore, the critical pathway for radionuclide migration both in the early phase of the accident and in the subsequent phases will be the chain "pastures-animals-livestock products-human".

The value of the total release corresponding to the maximum design basis accident in compliance with the document [70] is given in Table 8.2 (Bq).

Nuclide	0-2 hours	2-8 hours	8-24 hours	24-96 hours	96-720 hours
Xe-133	2.17E+14	1.173E+15	4.33E+15	7.69E+15	1.52E+16
Xe-133m	1.19E+12	9.41E+12	2.44E+13	4.94E+13	2.01E+14
Cs-137	3.03E+12	8.13E+12	5.07E+11	2.59E+09	2.40E+10

Table 8.2 - Expected total release of radionuclide composition to the environment under the design extension conditions at the AP1000 power unit (Bq) [70].

3.34E+13	9.27E+13	9.50E+12	6.24E+12	1.87E+13
7.48E+11	2.32E+12	1.26E+11	2.22E+08	7.40E+07
1.37E+11	4.51E+11	2.82E+10	1.48E+08	1.33E+09
2.96E+11	9.71E+11	6.03E+10	2.96E+08	2.15E+09
2.84E+10	8.94E+10	5.03E+09	1.48E+07	3.70E+06
6.54E+10	2.14E+11	1.33E+10	7.03E+07	4.48E+08
2.76E+12	9.01E+12	5.54E+11	2.96E+09	1.15E+10
	3.34E+13 7.48E+11 1.37E+11 2.96E+11 2.84E+10 6.54E+10 2.76E+12	3.34E+139.27E+137.48E+112.32E+121.37E+114.51E+112.96E+119.71E+112.84E+108.94E+106.54E+102.14E+112.76E+129.01E+12	3.34E+139.27E+139.50E+127.48E+112.32E+121.26E+111.37E+114.51E+112.82E+102.96E+119.71E+116.03E+102.84E+108.94E+105.03E+096.54E+102.14E+111.33E+102.76E+129.01E+125.54E+11	3.34E+139.27E+139.50E+126.24E+127.48E+112.32E+121.26E+112.22E+081.37E+114.51E+112.82E+101.48E+082.96E+119.71E+116.03E+102.96E+082.84E+108.94E+105.03E+091.48E+076.54E+102.14E+111.33E+107.03E+072.76E+129.01E+125.54E+112.96E+09

#### 8.4 Dose load assessment of an accident at the AP1000 unit

The dose load assessment in emergency situations and emergencies is a key task since the entire modern methodology for ensuring the radiation safety of nuclear power facilities is based on the standardization of population exposure individual doses.

#### 8.4.1 Design basis accident at the KhNPP Units 5-6 with AP1000 reactors

In case of design basis accidents, in accordance with subpara. 7.1-7.2 [55], public exposure doses on the SPZ boundary and beyond should not exceed the values of unconditional justification levels given in Table D.7.1 of NRBU-97, namely:

a) effective dose:

- children 10 mSv;
- adults 20 mSv;
- b) thyroid dose:
- children 100 mSv;
- adults 300 mSv;
- c) skin dose:
- children 300 mSv;
- adults 500 mSv.

Non-exceedance of radiation doses in case of design basis accidents and abnormal operation should be ensured:

- for all reference ages specified in NRBU-97;

- for all distances equal to or exceeding the distance from the places of possible releases to the SPZ boundaries.

The modeling of release propagation from KhNPP AP1000 Units 5-6 was performed for the design basis LOCA scenario (release activity according to Table 8.2). The calculations of contamination parameters and radiation doses were performed for stationary meteorological conditions at points on the plume axis with a step of 50 m up to a maximum distance of 3000 m from the source.

8.4.1.1 Effective dose

The results of calculations of effective dose values for all six reference groups of the population for the atmospheric stability category D (neutral conditions (equilibrium stratification) according to the Paskvill-Gifford classification) in the range of distances 500 - 1500 m are shown in Fig. 8.1. The wind speed was set equal to  $1 \text{ M/s}^{-1}$ , the intensity of rainfall was 10.7 mm h⁻¹. According to the results, the critical age group for such an accident is the reference group "10 years".



## Figure 8.1 - Dependence of the calculated effective dose value for different reference ages on the distance for the atmospheric stability category D

The horizontal lines indicate dose rates of 10 mSv for children and 20 mSv for adults in design basis accidents

The results of calculations of the effective exposure dose of the reference group "10 years" for all stability categories are shown in Fig. 8.2 For the stability category F (moderately stable atmosphere, according to the Paskvill-Gifford classification), the dose limit for children is not exceeded at a distance of 1750 m from the release source.



Figure 8.2 – Dependence of the calculated value of the effective dose of the reference age "10 years" on the distance for different categories of atmospheric stability *The horizontal line indicates the dose rate of 10 mSv for children in design basis accidents* 

The set of conditions for release propagation in the event of a design basis accident at the AP1000 unit, under which the maximum effective dose values are achieved, is the wind speed of  $1 \text{ m/s}^{-1}$ , the atmospheric stability category F (moderately stable atmosphere) and the value of the underlying surface roughness parameter z0 = 0.01 m (type of surface microrelief - mowed and low grass up to 15 cm). The critical population group is the age category "10 years".

Thus, in the case of the considered design basis LOCA at the AP1000 unit, the criterion of not exceeding the dose limit is ensured under all considered meteorological conditions and for all age groups at distances  $\geq$  1750 m from the release source.

Table 8.3 shows the results of calculations of the effective dose for individual nuclides and dose pathways for the age category "10 years", atmospheric stability category F (moderately stable atmosphere) and roughness parameter 0.01 m at a distance of 1750 m from the source, which ensures that the dose criterion for children of 10 mSv is not exceeded. According to the SOU NNEGC [55], the exposure pathway through consumption of contaminated products was not considered for accidental releases. The calculated value of the effective dose *Eeff* = 9.3 mSv consists of the contributions of the "air" component *Eair* (inhalation pathway and external exposure from activity in the air) and external exposure from fallout on the earth's surface *Esrf*, the relative contribution of which to the effective dose is 69 % and 31 %, respectively.

Table 8.3 - Contribution of individual nuclides to the calculated dose values due to inhalation of radionuclides and external exposure from the cloud *Eair* for the age category "10 years" in the case of a design basis accident at the AP1000 power unit, external exposure from fallout on the soil surface *Esrf* and total effective dose *Eeff* at a distance of 1750 m from the source

Nuclide	Emission, Bq	<i>Eair</i> , mSv	<i>Esrf</i> , mSv	<i>Eeff</i> , mSv

I-131 (aer)	3.17E+13	3.81E+00	2.22E+00	6.03E+00
I-131 (mol.)	1.62E+12	2.59E-03	4.31E-03	6.90E-03
I-131 (org.)	5.01E+10	3.71E-02	6.31E-04	3.77E-02
Xe-133m	1.19E+12	1.76E-04	0.00E+00	1.76E-04
Xe-133	2.17E+14	3.51E-02	0.00E+00	3.51E-02
Sr-90	1.37E+11	3.70E-01	2.97E-03	3.73E-01
Ru-103	2.96E+11	2.85E-02	2.56E-02	5.41E-02
Cs-137	3.03E+12	2.04E+00	3.28E-01	2.37E+00
Ba-140	2.76E+12	6.77E-02	9.69E-02	1.65E-01
La-140	2.84E+10	5.01E-04	1.18E-02	1.23E-02
Ce-141	6.54E+10	6.93E-03	8.84E-04	7.81E-03
Te-131m	7.48E+11	4.97E-03	1.95E-01	2.00E-01
Sum		6.41	2.89	9.30

#### 8.4.1.2 Equivalent skin dose

Equivalent skin doses to the public were calculated in the event of a design basis LOCA at the AP1000 unit. Fig. 8.3 shows the dependence of the equivalent skin dose on the distance to the source for different categories of atmospheric stability at a wind speed of  $1 \text{ m/s}^{-1}$  and a roughness parameter value of 0.01 m. Under the given conditions of atmospheric distribution of the release, the calculated values of the equivalent skin dose are the same for all reference groups of the population.





The horizontal lines indicate dose rates of 300 mSv for children and 500 mSv for adults in case of accidental releases

As follows from the calculation results, in the event of a design basis LOCA at the AP1000 unit, the non-exceedance of skin dose rate exposure for all considered meteorological conditions is ensured at distances of  $\geq$  350 m from the release source for children and  $\geq$  250 m from the release source for adults.

8.4.1.3 Equivalent thyroid dose

The equivalent thyroid doses for the public in the event of a maximum design basis accident at the AP1000 unit were calculated. Fig. 8.4 shows the dependence of the equivalent thyroid dose on the distance to the source for different categories of atmospheric stability at a wind speed of  $1 \text{ M/s}^{-1}$  and a roughness parameter value of 1 m, and Fig. 8.5 shows the dose-distance dependence for different reference ages and atmospheric stability category F (moderately stable atmosphere). The critical population group is the reference age of "1 year".



Figure 8.4 – Dependence of the calculated equivalent thyroid dose value on the distance for different categories of atmospheric stability and the reference age category "1 year"



Figure 8.5 – Dependence of the calculated equivalent thyroid dose value on the distance for different reference ages and atmospheric stability category F

Thus, in the event of a design basis LOCA at the AP1000 unit, the non-exceedance of thyroid dose rate for all considered meteorological conditions is ensured at a distances of  $\geq$  1700 m from the release source for children and  $\geq$  850 m from the release source for adults.

The model calculations for the release scenarios for a design basis LOCA at the AP1000 unit resulted in the following:

1. In the case of the considered design-basis LOCA at the AP1000 unit, the non-exceedance limit criterion for the effective dose is ensured under all considered meteorological conditions and for all age groups at a distances of  $\geq$  1750 m from the release source.

2. In the event of a design basis accident with loss of coolant at a power unit with an AP1000 reactor, the dose limit for skin exposure for all considered meteorological conditions is ensured at distances  $\geq$  350 m from the release source for children and  $\geq$  250 m from the release source for adults.

3. In the event of a design basis LOCA at the AP1000 unit, the non-exceedance of thyroid dose rate for all considered meteorological conditions is ensured at the distances of  $\geq$  1700 m from the release source for children and  $\geq$  850 m from the release source for adults.

4. The SPZ boundary for KhNPP AP1000 Units 5-6 (1700 m) practically does not exceed the existing KhNPP SPZ (2700 m radius), except for its western part. In this area, the SPZ boundary of the AP1000 units 5 and 6 exceeds the existing one in the western direction by no more than 250 meters. Most of this additional area falls on the water surface of the KhNPP cooling pond.

5. The KhNPP SPZ sizes ensure that the established dose criteria are not exceeded for all reference age groups and in the entire range of meteorological conditions for the propagation of radioactive releases in the atmosphere and characteristics of the underlying surface.

## 8.4.2 Beyond Design Basis Accident at the KhNPP AP1000 Units 5-6

Calculations of contamination characteristics and radiation doses were performed for stationary meteorological conditions at points on the plume axis with a step of 100 m up to a maximum distance of 30 km from the source.

## 8.4.2.1 Effective dose

Effective doses to the public were calculated for different age groups. In accordance with the requirements of [60], all the results presented below were obtained by modeling the spread of radionuclides in the atmosphere with a value of the wet removal constant  $\Lambda = 2 \text{ h}^{-1}$ , which corresponds to the intensity of rainfall  $J = 21.4 \text{ mm h}^{-1}$ .

Based on the results of model calculations, it was found that in the event of a *beyond* design basis accident (design extension conditions) at the AP1000 unit, the nonexceedance of dose criterion for the effective dose to the public for beyond design basis accidents (50 mSv) is provided under all considered meteorological conditions and for all age groups at the distances of  $\geq$  3500 m from the release source.

The set of conditions for release propagation at the BDBA (design extension conditions) under which the maximum effective dose values are achieved, is wind speed

1  $\text{M/s}^{-1}$  at a height of 10 m, atmospheric stability category F, rainfall intensity 21.4 mm h⁻¹, and the value of the roughness parameter of the underlying surface  $z_0 = 1$  m. The critical population group is the age category "1 year".

Table 8.4 shows the results of calculations of the effective radiation dose for two age categories "1 year" and "Adults" in case of an BDBA at one of the KhNPP AP1000 Units 5-6 at a distance of 3500 m from the release source. The main way of forming the effective radiation dose is external human exposure from fallout on the soil surface (68% for children of 1 year and 76% for adults).

Table 8.4 – Contribution of individual nuclides to the calculated dose values due to inhalation of radionuclides and external exposure from the cloud *Eair*, external exposure from deposition on the soil surface *Esrf* and total effective dose *Eeff* for two age categories "1 year" and "Adults" in case of a BDBA (design extension conditions) at AP1000 unit

Nuclida	Emission	1 year				Adults	
Inuclide	, Bq	<i>Eair</i> , mSv	<i>Esrf</i> , mSv	<i>Eair</i> , mSv	<i>Esrf</i> , mSv	<i>Eair</i> , mSv	<i>Esrf</i> , mSv
I-131 (aer.)	1,53E+14	1,10E+01	2,65E+01	3,75E+01	5,06E+00	2,65E+01	3,16E+01
I-131 (mol.)	7,79E+12	1,48E-01	6,71E-01	8,18E-01	7,60E-02	6,71E-01	7,47E-01
I-131 (org.)	2,41E+11	1,31E-01	7,16E-03	1,38E-01	6,79E-02	7,16E-03	7,50E-02
Xe-133m	2,85E+14	2,66E-02	0,00E+00	2,66E-02	2,66E-02	0,00E+00	2,66E-02
Xe-133	2,86E+16	2,92E+00	0,00E+00	2,92E+00	2,92E+00	0,00E+00	2,92E+00
Sr-90	6,18E+11	2,62E-01	3,31E-02	2,95E-01	4,46E-01	3,31E-02	4,79E-01
Ru-103	1,33E+12	1,41E-02	2,84E-01	2,99E-01	1,79E-02	2,84E-01	3,02E-01
Cs-137	1,17E+13	1,24E+00	3,13E+00	4,37E+00	2,03E+00	3,13E+00	5,16E+00
Ce-141	2,93E+11	3,66E-03	9,79E-03	1,35E-02	4,93E-03	9,79E-03	1,47E-02
La-140	1,23E+11	1,05E-03	1,26E-01	1,27E-01	8,14E-04	1,26E-01	1,27E-01
Ba-140	1,23E+13	2,83E-01	1,07E+00	1,35E+00	3,19E-01	1,07E+00	1,39E+00
Te-131m	3,19E+12	2,85E-02	2,04E+00	2,07E+00	1,50E-02	2,04E+00	2,06E+00
Sum		16,1	33,9	49,9	11,0	33,9	44,9

#### 8.4.2.2 Equivalent skin dose

Equivalent skin doses were calculated for the public in the event of an BDBA (design extension conditions) at the AP1000 unit. Fig. 8.6 shows the dependence of equivalent skin dose on the distance to the source for different categories of atmospheric stability at wind speed 1 m/s⁻¹ and roughness parameter value 1 m. Under the given conditions of atmospheric propagation of the release, the calculated values of the equivalent skin dose are the same for all reference groups of the population.



Figure 8.6 – Dependence of the calculated equivalent skin dose value on the distance for different categories of atmospheric stability

The horizontal line indicates the dose criterion of 500 mSv for all reference age categories in case of accidental releases

According to the calculations, in the event of a BDBA at the AP1000 unit, the dose criterion for skin exposure for all considered meteorological conditions and types of bedding surface is not exceeded at the distances of  $\geq$  1200 m from the release source for all reference age categories.

#### 8.4.2.3 Equivalent thyroid dose

Calculations of equivalent thyroid doses were performed for the population in case of a BDBA at the AP1000 unit. Fig. 8.7 shows the dependence of equivalent thyroid dose on the distance to the source for different categories of atmospheric stability at wind speed  $1 \text{ M/s}^{-1}$  and roughness parameter value of 1 m, and Fig. 8.8 shows the dependence of dose on distance for different reference age groups and atmospheric stability category F. The critical population group is the reference age "1 year", and for all age groups the maximum dose values are obtained for stability category F.



Figure 8.7 – Dependence of the calculated equivalent thyroid dose value on the distance for different stability categories and reference age categories "1 year" (a) and "Adults" (b)

The horizontal lines show the values of the dose criteria of 50 mSv for the "1 year" category and 200 mSv for the "Adults" category



Figure 8.8 – Dependence of the calculated equivalent thyroid dose value on the distance for different reference ages and atmospheric stability category F.

Thus, in the event of a BDBA (design extension conditions) at the AP1000 unit, the dose criteria for thyroid exposure are not exceeded under all considered meteorological conditions and for all reference age categories at the distances of  $\geq$  7600 m from the release source.

As a result of model calculations for the considered release scenario in case of a BDBA at the AP1000 units, the following results were obtained:

1. The critical population group is the age group "1 year". Non-exceedance of the criteria for the introduction of urgent countermeasures (lower limits of justification) for the *effective dose* under all considered meteorological conditions, all types of bedding surface and for all reference age groups is ensured at the distances of  $\geq 3500$  m from the release source.

2. Non-exceedance of the dose criterion for *skin exposure* for all considered meteorological conditions and types of underlying surface is ensured at the distances of  $\geq 1200$  m from the release source for all reference age groups.

3. Non-exceedance of the dose criteria for *thyroid exposure* is ensured for all considered meteorological conditions and for all reference age categories at the distances of  $\geq$  7600 m from the release source.

4. The main way of forming the effective dose is external human exposure from fallout on the soil surface (68% for children of 1 year and 76% for adults). The largest contribution to the effective dose for the critical age group "1 year" is  $^{131}I - 77\%$ .

5. The defined SA boundary for the AP1000 units fully fits into the current KhNPP SA with a radius of 30 km.

## 8.5 The Ensuring Safety Principles of the NPP

The design of the AP1000 unit was developed in accordance with the requirements of the U.S. legislation 10 CFR 20 "Standards for Protection Against Radiation" [71], internationally recognized standards (ASTME, ASME, etc.) and in accordance with the requirements of international institutions (IAEA, EUR and WENRA). One of the features of the AP1000 plant is the use of passive safety systems, which makes the power unit less dependent on external power supply in case of emergencies.

The AP1000 design is a combination of proven design concepts. In addition, is the result of the experience of operating existing PWR/WWR plants, applied to a defined set of functional requirements most simply and efficiently possible.

The principles of ensuring nuclear and radiation safety of Westinghouse Electric Company LLC AP1000 power units are based on:

• requirements established by the US nuclear safety codes, norms, and standards for the specifics of the power units under development and do not contradict the relevant nuclear safety rules and regulations in force in Ukraine;

• modern safety philosophy and principles developed by the global nuclear community and enshrined in the IAEA safety standards and ICRP publications;

• application of well-known deterministic methods, supplemented by probabilistic methods using appropriate numerical targets and analysis for detailed safety assessment;

- using proven components and technologies in the design of the AP1000;
- utilization of innovative safety features and capabilities in the AP1000 design;
- use of modular construction technology;

• simplification of the AP1000 design in comparison with WWER facilities operated in Ukraine;

- minimizing the impact of internal and external hazards and human errors;
- use of an advanced digital interface and human-machine interface.

The general safety objective is to ensure the personnel protection, the population, and the environment from unacceptable radiation exposure during the commissioning, operation, and decommissioning of a nuclear facility.

The general objective is achieved by safety management at all stages of the NPP nuclear facility life cycle and in all operational conditions through the implementation of radiological and technical safety objectives.

The operating organization continuously implements and keeps up-to-date the regulatory documents/standards of the enterprise that establish general provisions according to which the organizational culture is formed, maintained, and developed to ensure safety to protect personnel, the public, and the environment from unacceptable radiation exposure during the activities implementation in the field of nuclear energy use. These regulatory documents comply with the current regulatory requirements, take into account the IAEA recommendations, and experience in implementing a safety culture in the operating organization.

The above documents are mandatory for all operating organization personnel and should be included in the relevant regulatory documents, job descriptions, other company

production documents, briefing programs, training materials for company personnel, etc. Understanding of and compliance with the requirements of these documents is verified during self-assessments, safety culture audits, knowledge testing, and personnel certification.

By the law, the licensee (operating organization) must ensure compliance with the terms and conditions of the activity, which are defined in the license conditions or specified directly in the license when it is issued. The license conditions are usually developed by the licensing authority and approved by the Cabinet of Ministers of Ukraine. The current legislation provides for administrative liability for violation of the conditions for conducting the licensed type of activity and sets out the grounds for the licensing authority to refuse to issue a license, suspend it, or revoke it.

In its activities, the licensee should follow the basic principles of the state policy in the field of nuclear energy use and radiation protection, which are:

• priority of protecting people and the environment from ionizing radiation;

• ensuring the minimum level of radioactive waste generation during the use of nuclear energy;

• prohibition of any activity in the field of nuclear energy use that results in a reasonably foreseeable greater negative impact on future generations than that allowed for the current generation;

- ensuring safety in the use of nuclear energy;
- openness and accessibility of information related to the use of nuclear energy;
- providing compensation for damage caused by radiation exposure;

• creation of a legal and financial mechanism for socio-economic compensation of the risk for the population living in the observation zones;

• creation of special social infrastructure in the observation zones;

• ensuring measures for the socio-economic interest of local executive authorities and local self-government bodies on the territory of which nuclear facilities and facilities intended for radioactive waste management are located;

• establishing liability for violation of the legal safety regime in the field of nuclear energy use;

• separation of state administration functions in the field of nuclear energy use and state regulation of nuclear and radiation safety;

• separation of functions of state administration in the field of nuclear energy use and direct economic activity in the field of nuclear energy use;

• distribution of duties, rights and responsibilities among all subjects of legal relations in the field of nuclear energy use;

• regulation, licensing and supervision in the field of nuclear energy use;

• creation of a legal and financial mechanism of liability of the operating organization to citizens and business entities for nuclear damage caused;

• establishing a legal and financial mechanism for the licensee's liability to citizens and business entities for damage caused in the event of a radiation accident;

• disposal and long-term storage of radioactive waste at the expense of waste producers;

• participation of citizens and their associations in the formation of state policy in the field of nuclear energy use;

• prohibition of any activity related to ionizing radiation, if the benefits of such activity are less than the possible harm caused by it;

• compliance with the limits of exposure doses to personnel and the public established by the rules and regulations on nuclear and radiation safety;

• establishing the lowest indicators of individual doses, number of exposed persons, and probability of exposure from any specific radiation source under the rules and regulations on radiation safety, taking into account economic and social conditions of the state;

• implementation of international treaties, development of international cooperation in the field of peaceful uses of nuclear energy, and strengthening of the international regime of safety and radiation protection of the population;

• separation of state administration in the field of nuclear energy use and radioactive waste disposal.

The radiological safety objective is to ensure that the limits of radiation exposure to personnel, the public, and 10 CFR 20 "Standards for Protection Against Radiation" [71], as well as those established by the RSSU [22], limits of radiation exposure to personnel, the population and the environment during normal operation, violations of normal operation and design accidents, as well as limiting radiation exposure in the event of a severe accident. At the same time, it is necessary to ensure that the specified radiation exposure is at the lowest possible level, taking into account economic and social factors, and the achieved level of science and technology.

According to the established standards, the following categories of persons exposed to radiation (personnel and the population) are established:

• Category A (personnel) - persons who permanently or temporarily work directly with sources of ionizing radiation.

• Category B (personnel) - persons who are not directly involved in work with sources of ionizing radiation, but due to the location of workplaces in the premises and industrial sites of facilities with radiation and nuclear technologies, they may receive additional exposure

• Category C - the entire population.

Thus, each category of persons has its dose limits and permissible levels, which are given in Table 8.5.

Dece limite	Person exposure category				
Dose limits	Α	В	С		
Annual effective dose	20*	2	1		
The annual equivalent dose in:					
The lens of the eye	150	15	15		
Skin	500	50	50		

Table 8.5 – Dose limits (mSv per year)

Hands and fe	et		500	50	-
••• • • <b>*</b>	~	 -	1	 	

**Note** *: average for any consecutive 5 years, but not more than 50 mSv in a single year.

Following the radiological safety objective, it is necessary to:

• ensure that, in normal operation, radiation exposure at the plant and doses resulting from any release of radioactive material from the facility are kept as low as reasonably achievable and below the established limits;

• ensure that for all accidents considered in the design of the plant, the radiological consequences, if any, are negligible;

• reduce the likelihood of off-site events and their associated radiological consequences to avoid the need for extensive countermeasures and to enable the authorities to simplify off-site emergency planning.

The design of the AP1000 plant includes features that keep radiation exposure to personnel at a reasonably achievable level (ALARA principle) at all stages of the nuclear facility life cycle.

The main principles of radiation safety in accordance with the ALARA, which are provided for in the design of the AP1000 plant, are as follows:

• structures, systems and components are designed to be reliable and maintainable, thereby effectively reducing maintenance requirements for radioactive components;

• structures, systems and components are designed in a way to reduce radiation exposure, which allows for operation, maintenance and inspection of equipment within the minimum design radiation field;

• structures, systems and components are designed to ensure the optimal layout and access conditions to reduce the time for repair and dismantling, thereby effectively reducing the time spent in the radiation exposure zone during operation, maintenance and inspections;

• structures, systems and components are designed in such a way as to ensure remote and semi-remote operation, maintenance and inspections, thereby effectively reducing the time spent in the radiation exposure zone.

For radiation protection, the entire territory of the NPP with an AP1000 unit, where personnel may be present, is classified following the defined radiation zones by the requirements and definitions [71] and is based primarily on the expected personnel access. All areas of the plant are classified into radiation-safe areas and areas with limited access (strict regime areas) subject to radiological control. The designation of radiation zones and corresponding radiation levels of NPPs with AP1000 units according to [71] are given in Table 8.6.

Restricted zones (strict regime zones) following [71] are, in turn, divided into radiation hazardous zones, zones with high radiation levels, zones of radioactive air pollution, zones of surface radioactive pollution and zones of radioactive materials. During the operation of the plant, the requirements for mandatory designation of zones will be established by the radiation safety personnel based on the actual radiation levels
as specified in paragraph 1902 [71]. The requirements for access to each piece of plant equipment, instrumentation, or control devices used at the plant will be considered when establishing the design radiation level in this zone.



The safety of NPPs with the AP1000 reactor is ensured by a system of technological and organizational means through:

- use of the internal self-protection properties of the reactor unit;
- application of the principle of deeply shielded protection;

• use of safety systems designed using the principle of a single failure, diversity, redundancy, and physical separation;

• use of approved engineering and technical practices;

• Compliance with nuclear and radiation safety rules and regulations, as well as compliance with the requirements outlined in the AP1000 basic design;

• maintaining and improving the safety culture;

• use of the quality management system at all stages of the nuclear facility life cycle;

- ensuring appropriate training and qualification of personnel;
- taking into account operational experience, in particular, AP 600;

• availability of necessary operational documentation.

The principles of ensuring NPP safety are divided into fundamental and general organizational and technical principles.

The fundamental principles include:

- ensuring a safety culture;
- responsibility of the OO (Operating Organization);
- state regulation of safety;
- implementation of the defense-in-depth strategy.
- General organizational and technical principles include:
- application of proven engineering and technical practices;
- quality management;
- self-assessment of NPP safety;
- security analysis;
- departmental supervision;
- independent inspections;
- consideration of the human factor;
- ensuring radiation safety;
- consideration of operational experience;
- scientific and technical support.

The safety of the AP1000 installation is ensured by the consistent implementation of a translaty defended protection strategy based on

• prevention of deviations from normal operation;

• detection of deviations from normal operation and provision of means to prevent such deviations that may lead to emergency conditions;

• provision of engineering safety features to control and mitigate emergency conditions. In addition, the prevention and mitigation of severe accidents is considered through the development and use of probabilistic risk assessment and supporting analysis results.

The AP1000's ability to prevent abnormal operation is based on its safety margin and physical characteristics (e.g., larger water volumes and reserves, negative power and temperature coefficients).

# 8.6 Measures to protect the population and the environment

Measures to protect the population and the environment include the general procedure for determining emergency and urgent protective countermeasures for the population, informing and providing recommendations on the protection of the population to executive authorities, the state nuclear and radiation safety regulatory authority, and other interested organizations.

The population radiation protection in the context of an emergency is based on a system of radiation protection measures (countermeasures), which usually interfere with

normal human activity, as well as with normal social, economic and cultural functioning of the territories.

Intervention is a type of human activity that is always aimed at reducing and preventing uncontrolled and unpredictable exposure or the probability of exposure in situations of emergency exposure.

### Countermeasure types

All protective countermeasures used in the event of a radiation accident are divided into direct and indirect ones.

Direct countermeasures include countermeasures, the implementation of which leads to the prevention or reduction of individual and/or collective doses of the population's accidental exposure

Indirect countermeasures include all types of countermeasures that do not prevent individual and collective doses to the public but reduce (compensate) the amount of health damage associated with this accidental exposure.

Indirect countermeasures, in particular, include those aimed at improving the life quality of the population exposed to accidental exposure: introduction of socio-economic and medical benefits and monetary compensation, improvement of nutrition, etc.

The emergency plan developed for each NPP provides the scope of monitoring of radiation indicators of environmental objects in the exclusion (observation) zone, as well as information on the location of sampling sites for environmental objects, sites for monitoring dose rate, and integrated dose on the ground in the NPP observation area, locations of stationary radiation monitoring facilities (air filter of facilities, integrated dosimeters, etc.).

Protection of the public and the environment in the event of an accident at KhNPP is ensured by implementing the following key measures of the KhNPP's Emergency Preparedness and Response Systems.

• enhanced monitoring of radiation indicators of environmental objects and public exposure in the Zone

• forecasting of radiation doses to the population in the Zone;

• informing central and local executive authorities and local self-government bodies about the results of monitoring and radiation dose forecasting;

• provision of recommendations to central and local executive authorities and local self-government bodies on countermeasures to protect the population.

The monitoring of radiation indicators performed to protect the population and the environment includes:

• continuous monitoring of gas and aerosol emissions through vent pipes using stationary measuring instruments;

• radiation control over the release of radioactive substances into the environment, the state of the radiation situation in the SPZ and OZ and radioactive contamination of environmental objects;

• the scale assessment and the significance of accidental releases and discharges to the environment, and forecasts of public exposure doses.

The scale assessment and the significance of accidental releases and discharges to the environment are performed under the methodology [72].

Based on the results of the forecast of public exposure doses, protective countermeasures for the public are determined based on the levels of intervention and action levels established in [22].

The decision to take protective countermeasures is made not only by taking into account the current state of the radiation situation but is primarily based on the forecast of its development in connection with expected accidental releases and discharges, as well as using hydrometeorological forecasts.

The Radiation Situation Monitoring Group (RSMG) is responsible for forecasting public exposure doses and developing recommendations for public protection. Before the emergency groups and brigades (EG&B) are assembled, the shift personnel of the Radiation Safety Shop (RSS) develop recommendations for public protection for transmission to local executive authorities.

The results of assessments and forecasts and recommendations on protective countermeasures for the public are immediately transmitted to the headquarters of the emergency manager (HEMO). Before the meeting of the headquarters of the EMO, the information is reported to the NPP shift supervisor (NPP SS).

Analysis and clarification or verification of monitoring results, forecasts, and protective measures recommended for the population is carried out by the HEOS Headquarters on behalf of the HEOS. Recommendations on public protection for transfer to external organizations are approved by the HEOS (NPP SS (Shift Supervisor of the NPP) before the HEOS takes over its responsibilities).

The results of monitoring of radiation indicators, forecasts of radiation situations, public exposure doses, and recommended protective countermeasures for the public are immediately communicated to the Executive Committee of the Netishyn City Council and the Mayor of Netishyn, Energoatom Commission on Emergency Situations and other interested organizations.

During the accident, assessments and forecasts of radiation situation in the supervised area and public exposure doses are periodically repeated or performed immediately in case of significant changes in the hydrometeorological situation in the accident area.

8.7 Emergency preparedness and emergency response in the event of a radiation emergency

According to the IAEA Safety Glossary, emergency preparedness is the ability to take measures that effectively mitigate the consequences of an emergency for human health and safety, quality of life, property, or the environment.

Emergency response is defined by the IAEA as the implementation of measures aimed at mitigating the consequences of an emergency for human health and safety, quality of life, property, and the environment. It can also provide a basis for restoring normal social and economic activities.

Facilities that use radioactive materials (RM) and other ionizing radiation sources (IRS) in their practical activities are classified according to the category of potential radiation hazard to the public and the environment in the design (normal) mode and the event of a radiation emergency. Five categories of such facilities are defined. Emergency planning in case of a radiation emergency is carried out by these categories:

Category I – facilities (nuclear power plants) for which hazardous events on the site, including events with a low probability of occurrence, may lead to severe deterministic medical off-site effects.

Category II – facilities (some types of research reactors or enterprises for the production of sealed radiation sources, etc.) for which hazardous accidents at the industrial site may lead to an increase in off-site doses to the public, which justifies urgent countermeasures.

Category III – facilities (industrial radiation facilities, enterprises storing spent nuclear fuel or collecting and disposing of low-level waste, etc.) for which hazardous accidents at the site may lead to increased radiation doses or contamination, which justifies urgent protective actions at the site.

Category IV – activities that may lead to a radiation emergency that justifies the application of urgent countermeasures in an unexpected location. These activities include transportation and other authorized activities related to mobile objects (such as industrial radiographic radiation sources, nuclear-powered satellites or radiothermal generators), as well as unauthorized activities (such as activities with illegally obtained radiation sources). Hazard category IV represents the minimum hazard level applicable to the entire country territory;

Category V – activities that do not involve the use of radiation sources under normal conditions, but there is a significant probability of radioactive contamination of agricultural products to levels that require an immediate ban on the use of products as a result of an accident at facilities of radiation hazard categories I or II, including such facilities in other countries.

Radiation hazard categories I and II include power and research nuclear facilities, where radiation emergencies may result in deterministic and stochastic effects on personnel and the public, which requires planning of preventive and urgent protective actions such facilities off-site.

NPP safety is achieved through the consistent implementation of certain special measures at five levels of the defense-in-depth strategy, which is a set of consistent

physical barriers to the spread of radioactive substances and ionizing radiation together with technical means and organizational measures aimed at preventing deviations from normal operating conditions, preventing emergencies and limiting their consequences.

Emergency preparedness and response is the last of the five levels of defense-indepth strategy implementation which provides for:

• the establishment of a sanitary protection zone and supervised area around the NPP;

• availability of emergency plans, emergency response plans, the effectiveness and implementation readiness which are periodically checked during emergency training and activities;

construction of radiation shelters and Incident and Emergency Centers.

The NPP Incident and Emergency Center and the Emergency Preparedness and Response System of the Operating Organization (EPR) to emergencies and accidents at Ukrainian NPPs ensure implementation of measures of this level at NPPs, as the most radiation-hazardous facilities.

According to [73], the NPP Incident and Emergency Center is an infrastructure facility in the operating organization's emergency preparedness and response system that includes a set of specially equipped premises, equipment, information, and communication systems and is intended to manage the response to nuclear and radiation emergencies and other accidents at the NPP site and within its sanitary protection zone and to coordinate interaction with external organizations to ensure radiation protection of personnel

The Emergency Preparedness and Response System of the Operating Organization (EPR) is an organizational structure of an operating organization and operates following the IAEA recommendations, requirements of the nuclear legislation of Ukraine, norms, rules, and standards on nuclear and radiation safety, as well as legislation in the civil protection field.

The operating organization has a basic document that establishes the principles of organization of the operating organization's EPR, defines its goals, objectives, composition, operation procedure, distribution of powers and responsibilities between structural units and officials of the operating organization regarding emergency planning, preparedness, and response, interaction with third-party bodies, enterprises, and organizations. This document is developed and approved by the operating organization and coordinated with the Ministry of Energy, Ministry of Health, SNRIU, and State Emergency Service of Ukraine.

The operating organization's EPR performs its functions as part of the Unified State Civil Protection System (USCPS), which ensures state policy implementation in the civil protection field in Ukraine.

Interaction of central executive authorities, functional and territorial subsystems, and the operating organization within the preparedness and emergency response system for radiation and radiation-nuclear emergencies is carried out through the USCPS.

The scheme of interaction between the functional system of nuclear and radiation safety within the USCPS and the international obligations of Ukraine is shown in Fig. 8.9.

The authorities of the territorial subsystem of the USCPS (local executive authorities and local self-administration bodies) shall ensure, respectively

- notification of the population about protective measures;
- informing the population about the radiation situation;
- making decisions on iodine prophylaxis and the population evacuation;

– implementation of other measures within the competence and of the bodies' authorities under nuclear legislation.

In the event of a nuclear or radiation emergency at operating NPPs in Ukraine, central and local authorities should, by applicable law, make decisions and act based on relevant departmental and territorial plans, considering recommendations of the NPP and the operating organization, as well as other authorized bodies.



Figure 8.9 – Scheme of interaction of the functional system of nuclear and radiation safety

According to the requirements of the current legislative and regulatory acts in the civil protection field, all operating NPPs in Ukraine have identified high-risk facilities and approved lists of facilities to be equipped with automated systems for early detection of emergency hazards and notification of personnel in case of emergency. The operating organization has implemented automated systems for early warning of emergencies and notification of personnel in case of emergencies and notification of personnel in case of emergencies at each separate NPP unit and established control over their implementation.

## 8.7.1 Radiation monitoring

At any operating NPP, in normal operation (as well as at all stages of occurrence and elimination of consequences of a possible nuclear or radiological emergency and in the post-accident period), continuous monitoring of gas and aerosol emissions through vent pipes and liquid discharges, radiation situation in the premises, on the plant site and in the observation area is carried out (provided) following the Regulations on Radiation Monitoring during Operation of NPP Facilities. It ensures the accident detection of signs based on the readings of measuring instruments, alarm devices, relay protection devices, and automatics that have been triggered. In most cases, it is possible to detect signs of a radiation accident and declare the appropriate accident class even before radioactive substances leave the NPP site.

In addition to laboratory monitoring methods, each NPP operates an automated radiation monitoring system (ARMS) consisting of automated monitoring stations. Supervision is carried out continuously in an automatic mode, which allows receiving information from the control posts, conducting systematic data analysis, and performing a forecast of the radiation situation for all settlements in the 30-kilometer zone around the NPP.

To provide the public with the most accurate information and protect the population on time, the Ministry of Environmental Protection and Natural Resources together with the data owners: Energoatom, the Ministry of Health of Ukraine, the State Emergency Service of Ukraine, the Ukrainian Hydrometeorological Center, local authorities and business entities, have collected all relevant data on a single map of radiation background in Ukraine [74]. It shows measurements from monitoring points located throughout the country. Specialists, even in the most hazardous settlements, collect the data daily automatically or manually. Today, there are 514 measurement points throughout the country.

Summary information on the radiation situation at industrial sites and in the supervised areas of NPP branches is published on the official website.

#### 8.7.2 International commitments on early notification of a nuclear accident

In the event of a potential or actual emergency on the territory of Ukraine, according to the Procedure for Classification of Emergencies by their Levels [75], the State Emergency Response Plan [76] is put into effect.

Based on the IAEA recommendations and requirements, the following classes of accidents are established: emergency preparedness - industrial accident - site accident - municipal accident. They correspond to hazard warning - emergency at the facility - emergency on the site - general emergency.

In particular, in the event of a utility accident (the largest one) at an NPP, they are immediately implemented:

- emergency plan of the site, i.e. NPP;
- emergency plan of the operating organization;

- response plans of local and regional territorial subsystems of the Unified State Civil Protection System, whose territory belongs to the NPP supervised area;

response plans of relevant functional subsystems;

– radiation accident response plan.

The plan is intended for:

- prompt response of governmental authorities and civil protection forces, prevention of loss of life, material losses reduction, and organization of priority life support for the victims;

- organizing management, interaction and informing of the governing authorities and civil protection forces involved in disaster relief;

- determining the sequence and scope of organizational and practical measures to respond to an emergency and eliminate its consequences, deadlines, responsible executors, and resources required for this purpose.

The plan is implemented by:

– informing and notifying about the risk or occurrence of an emergency;

- transferring governing bodies and civil protection forces to the high alert mode and emergency mode;

- actions of governing bodies and civil protection forces in the high alert mode and emergency mode;

– management during the elimination of the consequences of an emergency;

- involvement of civil protection forces and conducting emergency rescue and other urgent works;

– interaction between governing authorities and civil protection forces;

- organization of the main types of support during rescue and other emergency operations and elimination of the consequences of an emergency.

Since 1986, Ukraine has ratified the IAEA Convention on Early Notification of a Nuclear Accident and the Convention on Assistance in the Case of a Nuclear Accident or Radiological Emergency by a decree of the Presidium of the Verkhovna Rada.

The SNRIU is authorized to act as a single national point of contact according to the IAEA Convention on Early Notification of Nuclear Accidents, the Convention on Assistance in the Case of a Nuclear Accident or Radiological Emergency, and the Convention on the Physical Protection of Nuclear Material, and Nuclear Facilities; to exchange internationally operational information on nuclear events within the International Nuclear Event Scale (INES).

The Convention on Early Notification of a Nuclear Accident applies in the event of any accident involving the operation of a nuclear facility, the activities of a State Party, or persons or entities under its jurisdiction or control, which results in a release of radioactive substances or may result in a release of radioactive substances and which has resulted in or may result in an international transboundary release that, in terms of radiation safety, could be important to other States. According to the Convention, a State Party in whose territory a nuclear accident has occurred shall immediately notify the IAEA, as well as (directly or through the IAEA) those States that have been or may be exposed to radiation, indicating the nature of the accident, the time when it occurred, and its exact location. Each State Party shall notify the IAEA and other States Parties of its competent authorities responsible for accident response, notification, and information relating to the accident.

The SNRIU provides the Ukrainian Hydrometeorological Center (UkrHMC) of the State Emergency Service with information on the source and (if available) composition of the radioactive release for modeling atmospheric transport of radioactive substances based on numerical weather forecasts using the Real-time On-line DecisiOn Support system (RODOS) and ensuring information exchange within the World Meteorological Organization. The UkrHMC provides SNRIU with available meteorological information for assessing and forecasting the development of emergencies, including risks of transboundary impact. At the state level, the SNRIU notifies the IAEA and competent authorities of other states, as well as the media through the SNRIU Information and Incident and Emergency Center.

# 9 IDENTIFICATION OF ALL DIFFICULTIES (TECHNICAL DEFICIENCIES, LACK OF SUFFICIENT TECHNICAL MEANS OR KNOWLEDGE) IDENTIFIED IN THE PREPARATION PROCESS OF THE ENVIRONMENTAL IMPACT ASSESSMENT REPORT

There are no difficulties in the preparation process of the environmental impact assessment report for the construction of KhNPP Units 5 and 6. The available technical means and knowledge are sufficient.

### 10 ALL COMMENTS AND SUGGESTIONS RECEIVED BY THE AUTHORIZED CENTRAL BODY AFTER THE PUBLISHED NOTIFICATION OF THE PLANNED ACTIVITY, AND A TABLE WITH INFORMATION ON FULL CONSIDERATION, PARTIAL ONE OR JUSTIFICATION FOR COMMENTS REJECTION AND SUGGESTIONS RECEIVED DURING THE PUBLIC DISCUSSION

All comments and suggestions received by the authorized central body after the publication of the Notification of the planned activity and information on consideration and incorporation are presented in Table 10.1, see also Appendix A.

The conditions of the Ministry of Environmental Protection and Natural Resources for the scope of studies and level of information details to be included in the EIA Report and information on consideration are presented in Table 10.2, see also Appendix B.

All comments and suggestions received from the Polish authorities on the scope of environmental impact assessment documentation and information on consideration and incorporation are presented in Table 10.3, see also Appendix K.

Table 10.1 - All comments and suggestions received by the authorized central body after the publication of the Notification of planned activities and responses to them

No.	Content of the comment or suggestion	Information on review and consideration	
	The public organization letter "Ecoclub" "Ecodia" "Khmelnytskyy Energy Cluster" dated 12.04.2024 №73		
1	Analysis of the reactor unit selection, including:	Considered	
	- justification and selection procedure for the Westinghouse Electric Company's AP 1000 reactor;	Information is provided in Section 2 of the EIA Report.	
	- justification of the possibility of selecting an alternative reactor unit, including the environmental impact determination of the alternative unit operation;		
	- information on consideration of other technical alternatives to generate the appropriate amount of electricity, including the option of refusing to construct KhNPP Units 5 and 6 and using renewable energy sources to generate the same amount of electricity.		
2	<ul> <li>Site selection for KhNPP 5 and 6 units: justification and procedure for site selection for KhNPP 5 and 6 units siting;</li> <li>justification of the possibility of selecting an alternative location (site) for KhNPP 5 and 6 units;</li> </ul>	Considered Information is provided in Section 2 of the EIA Report.	
	- list and results of studies conducted to determine the suitability of the selected territory (site) for the construction of KhNPP Units 5 and 6.		
3	Design characteristics, including:	Considered	
	-information on the development stage or design stage at which the object of the planned activity is, if available - project documentation adopted at this stage;	Information is provided in paragraph 5.1 of the EIA Report.	
	- information on the pre-project stage (pre-investment) works;		
	- dates of the completion of all design stages.		

No.	Content of the comment or suggestion	Information on review and consideration
4	Description of a permissible alternative, its feasibility and effectiveness, its advantages and disadvantages in terms of technical and economic indicators (including, if available, a feasibility study), environmental safety, and consumption of natural resources per unit of output.	Not considered The EIA Report discusses the territorial alternative in Section 2 of the EIA Report.
5	Detail the planned activity place and location of the main objects of the material and technical base (including buildings and structures, engineering and transport networks, and other graphic materials as part of design and planning decisions, which show the buildings' layouts and structures) of the planned activity on the following materials:	Partially considered The information is partially provided in paragraphs 1.1 and 1.4.1 of the EIA Report.
	-on the general plan of the territory and/or detailed plan of the territory;	
	-on orthophotomaps (satellite images) of high resolution;	
	-on the scheme (plan).	
	We propose to display the following information on the mapping materials:	
	-consumed land resources;	
	-30-kilometer zone of KhNPP;	
	-the industrial site boundaries;	
	-location and parameters of buildings, roads, and other facilities;	
	-location of sources of pollutant emissions;	
	-location of the nearby residential development to the object of the planned activity.	
	We also emphasize that the cartographic materials attached to the Report in electronic format should maintain high resolution and quality.	
6	Description of preparatory and construction works, their duration; content, fixed assets, and technologies of works related to the removal of green or other plants, engineering preparation, and land protection, change of relief, drainage of surface wastewater and groundwater; planned temporary structures, transport, and engineering networks.	Considered The information is provided in paragraph 1.3 of the EIA Report.

No.	Content of the comment or suggestion	Information on review and consideration
7	Description of the planned activity, namely:	Partially considered
	-production processes that are the content of the planned activity; in addition, it is recommended that the process description or equipment include the projected production capacity, maximum consumption of raw materials or other materials per hour, and other parameters of production capacity;	Information on production processes is provided in paragraph 1.4 of the EIA Report.
	-annual demand for raw materials, fuel and fuels, other materials, and water (in appropriate units);	Information on the annual demand for raw materials and water for cooling the units is provided in paragraphs 1.4.3 and 1.4.4 of the EIA
	-annual water demand for unit cooling in the following combinations: a) KhNPP Units 5, 6 b)	Report.
	KhNPP Units 1, 2 and 5, 6 c) KhNPP Units 1, 2, 3, 4, 5, 6	Information on the unit's lifetime is provided in
	-the planned activity lifetime ;	paragraph 1.4.2 of the EIA Report.
	-characteristics of the nuclear fuel cycle with its description;	Information on the nuclear fuel cycle
	-compliance of the planned KhNPP Units 5, and 6 with the operational and safety requirements defined by the International Atomic Energy Agency and the Western European Nuclear	5.7.2 (nuclear fuel use) of the EIA Report.
	Regulatory Association (WENRA);	Information on the compliance of the planned
	-organizational, technical and technological solutions for environmental protection and rational use of natural resources adopted by the planned activity and for which costs are provided, as well as restoration and componention measures (if applicable)	requirements is provided in paragraph 8.6 of the EIA Report.
	wen as restoration and compensation measures (in applicable).	Information on organizational, technical, and
		technological solutions is partially provided in Section 7 of the EIA Report.
8	Characteristics of the technological equipment to be constructed, reconstructed, re-equipped or	Considered
	dismantled (replaced) in connection with the planned activity, which is: - pollutant generation source;	Information on the sources of pollutants is provided in paragraph 1.5 of the EIA Report.
	- the source of the significant impact of physical factors (noise, infra- or ultrasound that spreads beyond the production buildings and the industrial site, significant electromagnetic radiation	Information on sources of significant impact of physical factors is provided in paragraph 5.3 of the EIA Report.

No.	Content of the comment or suggestion	Information on review and consideration
	following state sanitary norms and rules for the protection of the population from the effects of electromagnetic radiation, ionizing radiation, etc.;)	Information on pollutants is provided in paragraph 1.5.
	-accumulators of pollutants, solid and liquid waste;	Information on belonging to the objects subject to
	-belongs to the objects subject to EIA according to the Law.	EIA according to the Law is provided in Section 1 of the EIA Report.
9	The assessment (characterization) of the impact on environmental components is proposed to be	Considered
	considered according to the following options.	Information is provided in paragraph 5.5 of the EIA
	-Considering the impact of the operation of KhNPP Units 1 and 2 and planned KhNPP Units 5 and 6;	Report
	-considering the impact of the operation of KhNPP Units 1-4 (KhNPP Units 3, 4 following the EIA Conclusion, registration number of the case 2018111232231) and planned KhNPP Units 5,	
10		<u> </u>
10	Description of the waste, including its name or code by the state classifications, the annual volume of waste (and, if available, the specific generation rate adopted for the object of the	Considered
	planned activity), one of disposal at own waste sites, the volume of recycling and transfer to other organizations for recycling or placement (disposal).	Information is provided in paragraph 1.5.2.3 of the EIA Report.
11	Characterization of water use and water disposal in terms of the type of these systems, volumes	Considered
	per year, the quality composition of wastewater and other wastewater in terms of natural physical, chemical, and chemical indicators of water and the content of pollutants, and the equipment of wastewater treatment facilities. Water use and water intake should be	Information is provided in paragraphs 1.4.3, 1.4.4, 5.2.2, and 5.3.3 of the EIA Report.
	characterized in terms of the type of water supply system, type of source(s), name and type of water body - the supply source, water quality category, water intake volume per year by water supply source, and uneven consumption by month.	
12	Provide in the EIA report the calculation of the water balance for the Goryn River, considering	Partially considered
	the operation of 6 KhNPP units (No. 1-6) and 4 KhNPP units (No. 1-2, 5-6), with due regard to their lifetime and the trend of changes in the Goryn River water level due to climate change, and accounting for existing and projected economic activities within the Goryn River basin.	The information is partially provided in paragraph 1.4.5.1 of the EIA Report.

No.	Content of the comment or suggestion	Information on review and consideration
		More detailed information will be provided at the project stage.
13	Provide hydrothermal calculations of the cooling water reservoir for the period of operation of 6 KhNPP units (No. 1-6) and 4 KhNPP units (No. 1-2, 5-6), considering their lifetime.	Considered The information is provided in paragraph 5.3.2 of the EIA Report.
14	Characterization of land plots for areas allocated or to be allocated (withdrawn) for the planned activity, land plots (recommended list - by cadastral numbers), category and purpose for the planned and existing states, restrictions on the use of land for the existing and planned state following land management documentation, measures for engineering preparation and protection of land, terms, scope and technologies for reclamation of disturbed land, management of fertile soil layer on the land to be disturbed.	Considered The information is provided in paragraph 1.4.1 of the EIA Report.
15	Characterization of changes in the micro- or macroclimatic conditions of the area due to the planned activity, for example, in temperature, air humidity, freezing of soil or water, fog, or the probability of artificial fog formation over a residential area, etc. Additionally, assess the vulnerability of the planned activity as a whole or its processes or facilities to the adverse effects of climate change, such as an increase in average temperatures, unevenness of river flow, a drop in their water content, etc.	Considered Information is provided in paragraphs 5.6, and 5.3.2 of the EIA Report.
16	The impact characteristics of Units 5 and 6 on flora and fauna. The cumulative impact assessment of flora and fauna from the simultaneous operation of six KhNPP units, including:	Considered
	-identification of bioindicators for environmental assessments;	Information is provided in paragraph 7.5.1 of the EIA Report.
	-identification of zoo-indicators for environmental assessments;	Information is provided in paragraph 7.5.1 of the EIA Report.
	-assessment of changes in forest composition; plant and animal populations;	Information is provided in paragraph 5.2 of the EIA Report.
	-assessment of possible destruction of animal populations, destruction of ecosystems, partial or complete elimination of habitats;	Information is provided in paragraph 5.2 of the EIA Report.

No.	Content of the comment or suggestion	Information on review and consideration
	-assessment of changes in the composition of aquatic plant and animal populations.	Information is provided in paragraph 5.2 of the EIA Report.
17	Emission characteristics: a list of stationary sources with an indication of their type, including stationary sites where mobile sources of emissions (construction and installation equipment, trucks, mobile devices, and installations, except for hand tools) are regularly (daily) operated.	Considered Information is provided in paragraph 1.5 of the EIA Report
18	The chemical emissions characteristics: a list of chemical emissions to the atmosphere and the surrounding water environment (sources and quantitative characteristics of discharges).	Considered Information is provided in paragraph 1.5.2 of the EIA Report.
19	Characteristics of the impact on the population, including the rate of cancer in the population (considering the sex and age) in the nearest settlements in comparison with the average rate in Ukraine and calculations: -risk of non-carcinogenic effects; -risk of carcinogenic effects; -the social risk of the planned activity.	Considered Information is provided in paragraph 5.4 of the EIA Report.
20	If there are territories or objects of the NRF, Ecological Network on the territory of the planned activity, within the sanitary protection zone, or on the territory of natural environment objects directly affected by the planned activity, assess the impact of the planned activity on natural complexes and objects, in particular flora and fauna, their communities and habitats (including protected ones), and provide measures to minimize such impact by compensatory measures.	Considered Information is provided in paragraphs 5.2 and 7.6 of the EIA Report.
21	Description ("characteristics") of spent nuclear fuel and radioactive waste management, including the quantity, storage site, and storage period. Also, provide alternative options for spent nuclear fuel and radioactive waste handling from the planned one.	Considered Information is provided in paragraphs 5.7.2.4, 5.3.10, and 1.5.1 of the EIA Report.
22	Description (characterization) of the radioactive materials transportation, including nuclear fuel, spent nuclear fuel, and radwaste (RW), including:	Partially considered Information is partially provided in paragraph 5.7.2 of the EIA Report.

No.	Content of the comment or suggestion	Information on review and consideration
	-assessment of radiation impact during transportation of all types of radioactive materials to and from Khmelnytskyy NPP;	Detailed information will be at the "project" stage.
	-determination of the degree of environmental risk during transportation of all types of radioactive materials to and from Khmelnytskyy NPP under normal transportation conditions;	
	-assessment of consequences for the population and the environment during transportation of all types of radioactive materials to and from Khmelnytskyy NPP under normal transportation conditions;	
	-assessment of possible accident risks during transportation of all types of radioactive materials to and from Khmelnytskyy NPP with breakdown by types of transport used (rail, road, etc.)	
	-assessment of environmental and economic damage in case of an accident during transportation and costs for its compensation under all scenarios;	
	-assessment of environmental and economic costs of overcoming the consequences of transportation accidents under all scenarios.	
23	Characterize the environmental impact during emergencies:	Partially considered
	-to consider accident scenarios at one of KhNPP Units 5-6 and at two units at once, including those caused by the impact of military operations;	Accident scenarios are considered in the Preliminary Safety Analysis Report (PSAR).
	-to consider scenarios of different types of weapons hitting the vital life support points, destruction of the turbine hall or cooling system of KhNPP Units 5 and 6;	
	-to consider the scenario of an accident with the crash of a large passenger aircraft similar to a Boeing 777/Boeing 737 with full fuel tanks on the nodal points of KhNPP Units 5 and 6;	
	-maximum design and beyond design basis accidents (with their description);	Information is provided in Section 8 of the EIA Report.
	-meteorological situations that were considered in such assessments, the extent of impact on environmental components;	The information is provided in paragraphs 5.3.8, 5.3.9 of the EIA Report.
	-impact area in case of emergencies;	Information is provided in paragraphs. 5.3.8, 5.3.9 of the EIA Report.

No.	Content of the comment or suggestion	Information on review and consideration	
	-list of pollutants and their potential release mass in case of emergency;	Information is provided in paragraphs 1.5, 5.3.8, and 5.3.9 of the EIA Report.	
	-number of people potentially affected in the impact area in case of emergencies;	Information is provided in Section 2 of the EIA Report.	
	-potential economic losses from emergencies;	Information on economic losses from accidents and	
	-measures to prevent, minimize, and eliminate such consequences.	the measures envisaged are partially presented in Section 8, and detailed information will be provided in the Preliminary Safety Analysis Report	
		(PSAR).	
24	Description of the socio-economic impact and forecasting of the difference between the price of	Considered	
	electricity generated per unit of fuel and the market price of a unit of fuel and the storage price and further disposal of a spent nuclear fuel unit at present and every five years during the power units' lifetime.	Information is provided in paragraph 4.10 (socio- economic factors) of the EIA Report.	
25	Provide information related to decommissioning, liquidation (dismantling) of the facility,	Partially considered	
	cessation of planned activities, focusing on dismantling, removal of waste and hazardous chemicals, and land reclamation.	Information is partially provided in paragraphs 5.1 and 7.5.1 of the EIA Report.	
26	The EIA report should include:	Not considered	
	- Preliminary Safety Analysis Report (PSAR);	The PSAR and Ageing Management Program will	
	- Ageing Management Program;	be implemented at the design stage.	
	- A report on the inspection of the condition of the existing structures of the cooling pond dam.	Information on the need to construct a dam is provided in paragraphs 1.4 and 1.5.1.2.	
The	The Ministry of Environmental Protection and Natural Resources letter with comments and suggestions from the public on the planned activities dated 17 04 2024 No. 21/21-03/1681-24:		
	Letter of the Executive Committee of Netishyn City Council dated 04.04.2024 No. 01/01-18-1272/2024		
1	We suggest the Notification should be revised in Section 4 Social and Economic Impact of the	Partially considered	
	planned activity as follows:	The project will comply with all legal and	
	"The socio-economic justification for the planned activities is to strengthen the energy	organizational frameworks for environmental	

No.	Content of the comment or suggestion	Information on review and consideration
	independence of the state and provide the population and industry with electricity and heat. The	impact assessment following the requirements of
	project implementation is of national importance and involves a large number of local	the Law of Ukraine "On Environmental Impact
	contractors, which will create new jobs, and increase production capacity utilization and	Assessment" dated 23.05.2017 No. 2059-VIII.
	economic development of the country.	It should be noted that in connection with the entry
2	In the process of activities implementation, it is planned to:	into force on 29.12.2023 of the Law of Ukraine
	1) spend at least 10 percent of the cost of the power unit construction project on the social	dated 13.07.2023 No. 3227-IX "On Amendments
	development of the region.	to Certain Laws of Ukraine on Improving and
	$(2) = \frac{1}{2} + \frac{1}{2} $	Digitalizing the Environmental Impact Assessment
	2) ensure full implementation of Article 12 of the Law of Ukraine "On the Use of Nuclear Energy and Padiation Safety" in particular:	Procedure", the environmental impact assessment
	Energy and Radiation Safety, in particular.	procedure will now be implemented through a new
	-preserving benefits for payment for consumed electricity for the population permanently	Unified EIA Register on the Unified Environmental
	residing in the 30-kilometer zone of nuclear power plants;	Platform of the Ministry of Environmental
	-allocating to a special fund of the local budget socio-economic compensation for the risk of the	Protection and Natural Resources of Ukraine
	population living in the observation zone:	(Ministry of Ecology) "EcoSystem". The
		considering the peculiarities of martial law
	-creation and maintenance of special social infrastructure facilities, including protective	considering the peculiarities of martial law.
	structures designed to shelter and protect people, equipment, and property from radiation	The planned activities notification under the project
	exposure in the event of a radiation accident.	"Construction of power units No. 5, 6 with the
	33) provision of the specialized medical institution status located on the community territory	AP1000 reactor unit at the Khmelnytskyy NPP
	where the nuclear facility is, with appropriate financial support;	site" has been published on the EcoSystem
	A) personal income tay payment by business antities performing work on the completion of new	platform in the Unified EIA Register and is not
	a) power units at the Khmelnytskyy NPP site to the community budget where the nuclear facility	subject to change.
	is "	Within 12 working days from the date of notice
		publication, the public could submit comments and
		suggestions to the Ministry of Ecology on the
		planned activities, the research scope, and the level
		of detail of information to be included in the EIA
		keport.

No.	Content of the comment or suggestion	Information on review and consideration
		Subsequently, the Announcement of the public discussion and the EIA Report will be posted on the EcoSystem platform in the EIA Register.
		The duration of the public discussion is at least 25 working days dated the official publication of the Announcement and the EIA Report.
		At the public discussion stage of the EIA report, the public is allowed to participate in public hearings and provide any comments and suggestions on the environmental impact assessment report and the planned activities.
	Letter from Ulashanivka village council dated 04.04.2024 N	No. 06-59/87
1	In Section 4 of the notification, the last sentence should be revised to read as follows: "In the activity implementation planned to spend at least 10 percent (up to 10 percent in the published version) of the construction project cost on the social development of the region", since the development and economic growth of the caustic region, provision of business guarantees and its insurance should be ensured first and foremost.	Partially considered The project will comply with all legal and organizational frameworks for environmental impact assessment following the requirements of the Law of Ukraine "On Environmental Impact
2	It is worth noting that, in addition to the positive effects of the construction of such a facility, it is also necessary to specify its negative consequences for the population of the 30-kilometer risk zone, the threat to the health of this category of people, the threat to the environment and ecology, and the increased risks for investors, enterprises, and production. Section 4 of the notification should be expanded with paragraph 2 as follows: Decisions to build new power units will also have the following negative consequences:	Assessment" No. 2059-VIII dated 23.05.2017. It is important to note that due to the entry into force on 29.12.2023 of the Law of Ukraine No. 3227-IX dated 13.07.2023 "On Amendments to Certain Laws of Ukraine on Improving and Digitalizing the Environmental Impact Assessment Procedure", the environmental impact assessment procedure will now be implemented through a new
	NPPs have repeatedly faced the problem of a lack of water for reactor cooling. Fresh water is used to cool them. Most of the regions of Ukraine have insufficient water resources: 0.14-0.72 thousand cubic meters per person per year. In Europe, the average figure per person is 4.56 thousand mSv/year. In 2016 and 2017, Khmelnytskyy NPP additionally pumped water from the	Unified EIA Register on the Unified Environmental Platform of the Ministry of Environmental Protection and Natural Resources of Ukraine (the Ministry of Ecology), EcoSystem. The posted

No.	Content of the comment or suggestion	Information on review and consideration
	Goryn River to cool two power units. Despite this, KhNPP plans to build two additional units. This will affect the river's ecosystem;	documentation will be open, considering the peculiarities of martial law.
	- a threat to the public health of the 30-kilometer risk zone of Khmelnytskyy NPP. Increasing number of people suffering from cancer, cardiovascular diseases, etc.;	The notification on the planned activities under the project "Construction of power units No. 5, 6 with
	- climate change. According to the 2021 Analysis of the Impact of Climate Change on Water Resources. Ukraine', most river basins will experience a decrease in average annual runoff this century. The average annual temperature in Ukraine is projected to increase by 0.8-1.10 C. The redistribution of precipitation throughout the year will also increase: there will be more	the AP1000 reactor unit at the Khmelnytskyy NPP site" has been published on the EcoSystem platform in the Unified EIA Register and is not subject to change.
	precipitation in the cold season, and less in the warm season when the need for cooling of nuclear power plants increases;	Within 12 working days from the date of notice publication, the public could submit comments and
	-safety factors - construction safety of the AP 1000 technology, Westinghouse Electric Company claims that due to simplified systems, AP 1000 reactors are cheaper to build, operate, and maintain (due to lower piping safety, fewer cables and pumps, and a smaller overall building footprint). However, according to the U.S. Nuclear Regulatory Commission's report on	suggestions to the Ministry of Ecology on the planned activities, the research scope, and the level of detail of information to be included in the EIA Report.
	the safety assessment of the AP 1000 reactor, the reactor shield would not resist a direct aircraft hit, which is the most important in martial law;	Subsequently, the Announcement of the public discussion and the EIA Report will be posted on the EcoSystem platform in the EIA Register.
	- economic factors include the risks of cost overspending and delayed construction under martial law.	The duration of the public discussion is at least 25
	Please be aware that before the new nuclear reactor construction it is necessary to study the sufficiency of water resources and the impact on other facilities.	working days dated the official publication of the Announcement and the EIA Report.
	It is also essential to consider climate change forecasts in each region where construction is planned (increased emissions and average annual temperature, atypical weather changes during seasons such as droughts or floods, changes in air and water quality, etc.)	At the public discussion stage of the EIA report, the public is allowed to participate in public hearings and provide any comments and suggestions to the EIA report and the planned activities.
	Considering the above, we urge you to follow the above suggestions and comments, ensure compliance with the legal guarantees of the population of the risk zone, and prevent neglect of their rights and interests, as for several years in a row there have been trends of non-compliance	Information on the sufficiency of water resources is provided in paragraph 1.4.4 of the EIA Report.
	with the provisions of Articles 12-1, 12-2 of the Law of Ukraine "On the Use of Nuclear Energy and Radiation Safety", suspension of its scope from 2019 to 2024 by the Laws of	Information on threats to public health is provided in paragraph 5.4 of the EIA Report.

No.	Content of the comment or suggestion	Information on review and consideration
	Ukraine, which contradicts the requirements of Article 22 of the Constitution of Ukraine; non- return of electricity benefits for consumers, etc. In addition, the location of an object of national importance in the region should provide guarantees for business and its insurance, and attract investors for the territorial community development.	Paragraph 5.6 of the EIA Report provides information on climate change forecasts. Information on safety factors is provided in Sections 7 and 8 of the EIA Report.
	Letter of Pluzhne village council dated 04.04.2024 №03-3	3/365/2024
1	In Section 4 of the notification, the last sentence should be revised to read as follows: "In the activity implementation it is planned to spend at least 10 percent (up to 10 percent in the published version) of the construction project cost on the social development of the region", since the development and economic growth of the caustic region, provision of business guarantees and its insurance should be ensured first and foremost.	Partially considered The project will comply with all legal and organizational frameworks for environmental impact assessment following the requirements of the Law of Ukraine "On Environmental Impact
2	It is worth noting that besides the positive consequences of the construction of such a facility, it is also necessary to specify its negative consequences for the population of the 30-kilometer risk zone, the threat to the health of this category of people, the threat to the environment and ecology, and the increased risks for investors, enterprises, and production, Section 4 of the notification should be supplemented with paragraph 2 as follows: The decision to build new power units will also have the following negative consequences: - a threat to the environment is a potential water resource shortage, as Ukrainian NPPs have repeatedly faced the problem of a lack of water for reactor cooling. Fresh water is used to cool them. Most of the regions of Ukraine have insufficient water resources: 0.14-0.72 thousand cubic meters per person per year. In Europe, the average per capita figure is 4.56 thousand mSv/year. In 2016 and 2017, Khmelnytskyy NPP additionally pumped water from the Goryn River to cool two power units. Despite this, KhNPP plans to build two additional units. This will affect the river's ecosystem; - a threat to the population health in the 30-kilometer risk zone of the Khmelnytskyy NPP. Increase in the number of people suffering from cancer, cardiovascular diseases, etc.; - climate change. According to the 2021 Analysis of the Impact of Climate Change on Water	Assessment" dated 23.05.2017 No. 2059-VIII. It should be noted that in connection with the entry into force on 29.12.2023 of the Law of Ukraine dated 13.07.2023 No. 3227-IX "On Amendments to Certain Laws of Ukraine on Improving and Digitalizing the Environmental Impact Assessment Procedure", the environmental impact assessment procedure will now be implemented through a new Unified EIA Register on the Unified Environmental Platform of the Ministry of Environmental Protection and Natural Resources of Ukraine (the Ministry of Ecology) "EcoSystem". The documentation to be posted there will be open, considering the peculiarities of martial law. The notification on the planned activities under the project "Construction of Power Units 5, 6 with the AP 1000 Reactor Unit at the Khmelnytskyy NPP Site" was published on the EcoSystem platform in

No.	Content of the comment or suggestion	Information on review and consideration
	century. The average annual temperature in Ukraine is projected to increase by 0.8-1.10 C. The redistribution of precipitation throughout the year will also increase: there will be more precipitation in the cold season, and less in the warm season when the need for cooling of nuclear power plants increases;	change. Within 12 working days from the date of publication of the notice, the public had the right to provide the Ministry of Environment with
	- safety factors - construction safety of the AP 1000 technology, Westinghouse Electric Company claims that due to simplified systems, AP 1000 reactors are cheaper to build, operate, and maintain (due to lower piping safety, fewer cables and pumps, and a smaller overall building footprint). However, according to the U.S. Nuclear Regulatory Commission's report on the sofety exception of the AP 1000 reactors the meeter shield would not reside a direct size of the terms of terms of the terms of the terms of terms of the terms of the terms of the terms of the terms of te	comments and suggestions on the planned activities, the scope of studies, and the level of detail of information to be included in the EIA Report.
	hit, which is the most important in martial law; - economic factors, including the risks of cost overspending and delayed construction under	Subsequently, the Announcement on the start of public discussion and the EIA Report will be posted on the EcoSystem platform in the EIA
	martial law.	Register.
	Please be aware that before the construction of new nuclear reactors, it is necessary to study the sufficiency of water resources and the impact of other facilities.	The public discussion duration is at least 25 working days dated the official publication of the Announcement and the EIA Report.
	It is also essential to consider climate change forecasts in each region where construction is planned (increased emissions and average annual temperature, atypical weather changes during seasons such as droughts or floods, changes in air and water quality, etc.)	At the stage of public discussion of the EIA report, the public is allowed to participate in public
	Given the above, we urge you to take into account the above proposals and comments, ensure compliance with the legal guarantees of the population of the risk zone, and prevent neglect of their rights and interests, as for several years in a row there have been trends of non-compliance	hearings and provide any comments and suggestions to the environmental impact assessment report and the planned activities.
	with the provisions of Articles 12-1, 12-2 of the Law of Ukraine "On the Use of Nuclear Energy and Radiation Safety", suspension of its scope from 2019 to 2024 by the Laws of	Information on water resource sufficiency is provided in paragraph 1.4.4 of the EIA Report.
	Ukraine, which contradicts the requirements of Article 22 of the Constitution of Ukraine; non- return of electricity benefits for consumers, etc. In addition, the location of an object of national importance in the region should provide guarantees for business and its insurance, and attract	Information on threats to public health is provided in paragraph 5.4 of the EIA Report.
	investors for the territorial community development. Besides, according to a preliminary public opinion poll, the public is extremely concerned and against the 5-6 power unit construction at the Khmelnytskyy NPP site due to the systematic failure to comply with the above legal	Paragraph 5.6 of the EIA Report provides information on climate change forecasts.
	requirements and disregard for their legitimate rights and interests.	Information on safety factors is provided in

No.	Content of the comment or suggestion	Information on review and consideration
		Sections 7 and 8 of the EIA Report.
	Letter of the Iziaslav City Council of Shepetivka District, Khmelnytskyy Region	, dated 04.04.2024 No. 1013/03-14
1	In Section 4 of the notification, the last sentence should be revised to read as follows: "In the course of carrying out the activity, it is planned to spend at least 10 percent (up to 10 percent in the published version) of the construction project cost on the social development of the region", since the development and economic growth of the caustic region, provision of business guarantees and its insurance should be ensured first and foremost.	Partially considered The project will comply with all legal and organizational frameworks for environmental impact assessment following the requirements of the Law of Ukraine "On Environmental Impact
2	It is worth noting that besides the positive consequences of the construction of such a facility, it is also necessary to specify its negative consequences for the population of the 30-kilometer risk zone, the threat to the health of this category of people, the threat to the environment and ecology, and the increased risks for investors, enterprises, and production.	Assessment" dated 23.05.2017 No. 2059-VIII. It should be noted that in connection with the entry into force on 29.12.2023 of the Law of Ukraine dated 13.07.2023 No. 3227-IX "On Amendments to Certain Laws of Ukraine on Improving and
	Section 4 of the notification should be supplemented with paragraph 2 as follows: The decision to build new power units will also have the following negative consequences: - a threat to the environment is a potential water resource shortage, as Ukrainian NPPs have repeatedly faced the problem of a lack of water for reactor cooling. Fresh water is used to cool them. Most of the regions of Ukraine have insufficient water resources: 0.14-0.72 thousand cubic meters per person per year. In Europe, the average per capita figure is 4.56 thousand mSv/year. In 2016 and 2017, Khmelnytskyy NPP additionally pumped water from the Goryn River to cool two power units. Despite this, KhNPP plans to build two additional units. This will affect the river's ecosystem;	to Certain Laws of Okraine on Improving and Digitalizing the Environmental Impact Assessment Procedure", the environmental impact assessment procedure will now be implemented through a new Unified EIA Register on the Unified Environmental Platform of the Ministry of Environmental Protection and Natural Resources of Ukraine (the Ministry of Ecology) "EcoSystem". The documentation to be posted there will be open, considering the peculiarities of martial law.
	<ul> <li>- a threat to the population health in the 30-kilometer risk zone of the Khmelnytskyy NPP. Increase in the number of people suffering from cancer, cardiovascular diseases, etc.</li> <li>- climate change. According to the 2021 Analysis of the Impact of Climate Change on Water Resources. Ukraine', most river basins will experience a decrease in average annual runoff this century. The average annual temperature in Ukraine is projected to increase by 0.8-1.10 C. The redistribution of precipitation throughout the year will also increase: there will be more precipitation in the cold season, and less in the warm season when the need for cooling of nuclear power plants increases;</li> </ul>	The notification on the planned activities under the project "Construction of Power Units 5, 6 with the AP 1000 Reactor Unit at the Khmelnytskyy NPP Site" was published on the EcoSystem platform in the Unified EIA Register and is not subject to change. Within 12 working days from the date of publication of the notice, the public had the right to

No.	Content of the comment or suggestion	Information on review and consideration
	- safety factors - construction safety of the AP 1000 technology, Westinghouse Electric Company claims that due to simplified systems, AP 1000 reactors are cheaper to build, operate, and maintain (due to lower piping safety, fewer cables and pumps, and a smaller overall building footprint). However, according to the U.S. Nuclear Regulatory Commission's report on the safety assessment of the AP 1000 reactor, the reactor shield would not resist a direct aircraft bit which is the most important in mortial laws	provide the Ministry of Environment with comments and suggestions on the planned activities, the scope of studies, and the level of detail of information to be included in the EIA Report.
	<ul> <li>economic factors, including the risks of cost overspending and delayed construction under martial law.</li> <li>Please be aware that before the construction of new nuclear reactors, it is necessary to study the</li> </ul>	Subsequently, the Announcement on the start of public discussion and the EIA Report will be posted on the EcoSystem platform in the EIA Register.
	sufficiency of water resources and the impact of other facilities.	The public discussion duration is at least 25 working days dated the official publication of the
	planned (increased emissions and average annual temperature, atypical weather changes during seasons such as droughts or floods, changes in air and water quality, etc.).	Announcement and the EIA Report. At the stage of public discussion of the EIA report, the public is allowed to participate in public
3	Given the above, we urge you to take into account the above proposals and comments, ensure compliance with the legal guarantees of the population of the risk zone, and prevent neglect of their rights and interests, as for several years in a row there have been trends of non-compliance with the provisions of Articles 12-1, 12-2 of the Law of Ukraine "On the Use of Nuclear	hearings and provide any comments and suggestions to the environmental impact assessment report and the planned activities.
	Energy and Radiation Safety", suspension of its scope from 2019 to 2024 by the Laws of Ukraine, which contradicts the requirements of Article 22 of the Constitution of Ukraine; non-	Information on water resource sufficiency is provided in paragraph 1.4.4 of the EIA Report.
	return of electricity benefits for consumers, etc. In addition, the location of an object of national importance in the region should provide guarantees for business and its insurance, and attract investors for the territorial community development. Besides, according to a preliminary public	Information on threats to public health is provided in paragraph 5.4 of the EIA Report.
	opinion poll, the public is extremely concerned and against the 5-6 power unit construction at the Khmelnytskyy NPP site due to the systematic failure to comply with the above legal requirements and disregard for their legitimate rights and interests	Paragraph 5.6 of the EIA Report provides information on climate change forecasts.
	requirements and disregard for their regitimate rights and interests.	Information on safety factors is provided in Sections 7 and 8 of the EIA Report.
	Letter of Shepetivka City Council dated 05.04.2024 No. 0	3-24/1319
	During the clarification of the coordinates of the KhNPP's 30-kilometer zone was determined that Shepetivka was not included in the list of 125 settlements in this zone, which is located	The size and boundaries of the observation zone (hereinafter referred to as the OZ) for

No.	Content of the comment or suggestion	Information on review and consideration
	27.5 km from the plant, and four other settlements of the former Shepetivka and Slavuta districts, including Plesnat village, which is currently part of the Shepetivka city territorial community. Based on the results of topographic surveys carried out with consideration of changes in the boundaries of Shepetivka, as defined by the Verkhovna Rada Resolution No. 64/98-VR of February 3, 1998, almost 30% of the town's territory is located in the KhNPP's 30-kilometer zone.	Khmelnytskyy NPP operating units were determined by the NPP technical design. The sufficiency of the size and boundaries of the observation zone for the operating Khmelnytskyy NPP units was substantiated and confirmed in the final safety analysis report for Unit 2, which passed
	In 2007, OJSC "Kyiv Research and Design Institute 'ENERGOPROEKT' together with the Ukrainian Research Institute of Environmental Problems (UkrRIEP) of the Ministry of Environmental Protection of Ukraine, Kharkiv, at the request of the Executive Committee of the Shepetivka City Council, developed a 'Scientific and Economic Justification for the Necessity of Including Shepetivka in the 30-kilometer observation zone of the Khmelnytskyy NPP'. The conclusions of this study state:	the state nuclear and radiation safety assessment. The "Feasibility Study for the Construction of Khmelnytskyy NPP Units 3 and 4", which passed a comprehensive state review and was approved, also proved the sufficiency of the existing facilities for the operation of Khmelnytskyy NPP consisting of four VVER-1000 power units.
	<ul> <li>- about 30% of the city's territory is located within the 30-kilometer zone of the KhNPP;</li> <li>- about one-third of the city's housing stock, the main recreation area, children's health camp, new drinking water intake "Lisova Haliavyna", water supply from Kamiankivske drinking water intake, as well as several enterprises, organizations, and institutions are located within the 30-km zone;</li> </ul>	For the design of new Khmelnytskyy NPP units 5 and 6 with the AP1000 reactor unit, the dimensions of the SS are set following the requirements of paragraphs 4.13-4.16 of NP 306.2.144-2008 "Safety Requirements for NPP Site Selection" and paragraphs 9.1 and 9.4 of the "Basic Sanitary Rules
	- Section 4.1 of the justification proves that the radiation impact of the KhNPP on the entire territory of Shepetivka and all its residents is the same, and therefore it is meaningless to divide the city into two parts (by the boundary of the 30-kilometer zone) based on the radiation intensity of the KhNPP;	for Ensuring Radiation Safety of Ukraine" (OSPU- 2005). The OZ is established based on calculations specified in NP 306.2.173-2011 "Requirements for Determining the Size and Boundaries of the
	- about a third of the city's population (almost 14 thousand residents) already lives in areas less than 30 km from the Khmelnytskyy NPP. There are also prospective residential areas in this zone with a population of more than 7 thousand people;	Nuclear Power Plant Surveillance Zone", considering the cumulative impact from all power units.
	- in the vicinity of Shepetivka, the natural factors analysis proves that the radiation impact on Shepetivka residents is no less than on residents of the nearest settlements within the 30- kilometer zone (for instance, Plishchyn village, which borders the city, is included in the Khmelnytskyy NPP's 30-km zone list of settlements around and is currently part of the	Article 45 of the Law of Ukraine "On the Use of Nuclear Energy and Radiation Safety" prescribes that the size and boundaries of the observation zone are determined in the design of a nuclear power plant following the rules and regulations in the field

No.	Content of the comment or suggestion	Information on review and consideration
	settlements of the Shepetivka city territorial community); - there is a strong communication link between the city districts located inside and outside the 30-km zone, and between the city and non-urban areas within the 30-km zone. It proves that the	of nuclear energy use and are also agreed with the state nuclear and radiation safety regulatory authority.
	factors related to KhNPP operation have a joint effect on the entire population of Shepetivka and, therefore, the same impact on the city population and other settlements within the 30-km zone.	The legal standard is also contained in clause 6.8.3 of NP 306.2.141-2008 "General Safety Provisions for Nuclear Power Plants", which defines that the
	The State Sanitary and Epidemiological expertise of the above-mentioned scientific and economic justification was conducted in 2007 by the Marzeev Institute of Hygiene and Medical Ecology of the Academy of Medical Sciences of Ukraine. According to the conclusions of the	size of the sanitary protection zone and observation zone is justified by the design of the nuclear power plant.
	State Sanitary and Epidemiological Expertise of the Ministry of Health of Ukraine (№05.03.02- 07/47840 of 25.09.2007), it was found possible to include Shepetivka in the 30-kilometer zone of the Khmelnytskyy NPP. It should be noted that this conclusion is valid without any term limit.	Decision-making on the siting, design, and construction of nuclear facilities and radioactive waste handling facilities of national importance is within the exclusive competence of the Verkhovna
	Paragraph 2.1 of the Requirements for Determining the Size and Boundaries of the Nuclear Power Plant Surveillance Zone, approved by the joint order of the State Nuclear Regulatory Inspectorate of Ukraine and the Ministry of Health of Ukraine No. 153/766 dated 07.11.2011 (as amended by the Order of the State Nuclear Regulatory Inspectorate of Ukraine No. 206/765 of 23.11.2015) provides that in case the outer boundary of the observation zone (hereinafter referred to as the OZ) passes through a settlement, the OZ boundary is adjusted so that the entire settlement within its administrative boundary is included in the NPP OZ.	Rada of Ukraine following Article 17 of the Law of Ukraine "On Nuclear Energy Use and Radiation Safety" and Article 2 of the Law of Ukraine "On the Procedure for Decision-Making on Siting, Design, and Construction of Nuclear Facilities and Radioactive Waste Handling Facilities of National Importance".
	Considering the mentioned above, we urge you to consider and resolve the issue of including the city of Shepetivka and the village of Plesna, which is part of the Shepetivka urban territorial community (hromada), in the list of administrative-territorial units that are part of the 30-kilometer observation zone of the Khmelnytskyy NPP.	Therefore, revision and approval of the list of settlements to be included in the Exclusion Zone is possible during the design development of new power units of Khmelnytskyy NPP by the procedure established by law, after the Verkhovna Rada of Ukraine decides on siting, design, and construction of the nuclear facilities.
	Letter of Slavuta City Council dated 05.04.2024 No. 04	-29/519
1	In Section 4 of the notification, the last sentence should be revised to read as follows: "In the	Partially considered
	course of carrying out the activity, it is planned to spend at least 10 percent (up to 10 percent in	

No.	Content of the comment or suggestion	Information on review and consideration
	the published version) of the construction project cost on the social development of the region",	The project will comply with all legal and
	since the development and economic growth of the caustic region, provision of business	organizational frameworks for environmental
	guarantees and its insurance should be ensured first and foremost.	impact assessment following the requirements of
2	It is worth noting that besides the positive consequences of the construction of such a facility it	the Law of Ukraine "On Environmental Impact
2	is also necessary to specify its negative consequences for the population of the 30-kilometer	Assessment" dated 23.05.2017 No. 2059-VIII.
	risk zone, the threat to the health of this category of people, the threat to the environment and	It should be noted that in connection with the entry
	ecology, and the increased risks for investors, enterprises, and production.	into force on 29.12.2023 of the Law of Ukraine
	Section 4 of the notification should be supplemented with paragraph 2 as follows:	dated 13.07.2023 No. 3227-IX "On Amendments to Certain Laws of Ukraine on Improving and
	The decision to build new power units will also have the following negative consequences:	Digitalizing the Environmental Impact Assessment Procedure", the environmental impact assessment
	- a threat to the environment is a potential water resource shortage, as Ukrainian NPPs have repeatedly faced the problem of a lack of water for reactor cooling. Fresh water is used to cool them. Most of the regions of Ukraine have insufficient water resources: 0.14-0.72 thousand cubic meters per person per year. In Europe, the average per capita figure is 4.56 thousand mSv/year. In 2016 and 2017, Khmelnytskyy NPP additionally pumped water from the Goryn River to cool two power units. Despite this, KhNPP plans to build two additional units. This will affect the river's ecosystem;	procedure will now be implemented through a new Unified EIA Register on the Unified Environmental Platform of the Ministry of Environmental Protection and Natural Resources of Ukraine (the Ministry of Ecology) "EcoSystem". The documentation to be posted there will be open, considering the peculiarities of martial law.
	- a threat to the population health in the 30-kilometer risk zone of the Khmelnytskyy NPP. Increase in the number of people suffering from cancer, cardiovascular diseases, etc.	The notification on the planned activities under the project "Construction of Power Units 5, 6 with the
	- climate change. According to the 2021 Analysis of the Impact of Climate Change on Water Resources. Ukraine', most river basins will experience a decrease in average annual runoff this century. The average annual temperature in Ukraine is projected to increase by 0.8-1.10 C. The redistribution of precipitation throughout the year will also increase: there will be more	AP1000 Reactor Unit at the Khmelnytskyy NPP Site" was published on the EcoSystem platform in the Unified EIA Register and is not subject to change.
	precipitation in the cold season, and less in the warm season when the need for cooling of nuclear power plants increases;	Within 12 working days from the date of publication of the notice, the public had the right to
	- safety factors - construction safety of the AP 1000 technology, Westinghouse Electric Company claims that due to simplified systems, AP 1000 reactors are cheaper to build, operate, and maintain (due to lower piping safety, fewer cables and pumps, and a smaller overall building footprint). However, according to the U.S. Nuclear Regulatory Commission's report on	provide the Ministry of Environment with comments and suggestions on the planned activities, the scope of studies, and the level of detail of information to be included in the EIA

No.	Content of the comment or suggestion	Information on review and consideration
	the safety assessment of the AP 1000 reactor, the reactor shield would not resist a direct aircraft hit, which is the most important in martial law;	Report.
	- economic factors, including the risks of cost overspending and delayed construction under martial law.	public discussion and the EIA Report will be posted on the EcoSystem platform in the EIA
	Please be aware that before the construction of new nuclear reactors, it is necessary to study the sufficiency of water resources and the impact of other facilities.	The public discussion duration is at least 25
	It is also essential to consider climate change forecasts in each region where construction is planned (increased emissions and average annual temperature, atypical weather changes during	working days dated the official publication of the Announcement and the EIA Report.
	seasons such as droughts or floods, changes in air and water quality, etc.).	At the stage of public discussion of the EIA report, the public is allowed to participate in public
3	Given the above, we urge you to take into account the above proposals and comments, ensure compliance with the legal guarantees of the population of the risk zone, and prevent neglect of their rights and interests, as for several years in a row there have been trends of non-compliance with the provisions of Articles 12-1, 12-2 of the Law of Ukraine "On the Use of Nuclear	hearings and provide any comments and suggestions to the environmental impact assessment report and the planned activities.
E U	Energy and Radiation Safety", suspension of its scope from 2019 to 2024 by the Laws of Ukraine, which contradicts the requirements of Article 22 of the Constitution of Ukraine; non- return of electricity benefits for consumers, etc. In addition, the location of an object of national importance in the region should provide guarantees for business and its insurance, and attract investors for the territorial community development. Besides, according to a preliminary public opinion poll, the public is extremely concerned and against the 5-6 power unit construction at the Khmelnytskyy NPP site due to the systematic failure to comply with the above legal requirements and disregard for their legitimate rights and interests.	Information on water resource sufficiency is provided in paragraph 1.4.4 of the EIA Report.
		Information on threats to public health is provided in paragraph 5.4 of the EIA Report.
		Paragraph 5.6 of the EIA Report provides information on climate change forecasts.
		Information on safety factors is provided in Sections 7 and 8 of the EIA Report.

Table 10.2 – Requirements of the Ministry of Environment for the scope of studies and level of detail of information to be included in the EIA Report

No.	Requirements of the Ministry of Environment	Information about the consideration
1	Include the technical characteristics information of the planned activity in the EIA Report, in particular:	
	-type of equipment and its technical characteristics to be used in the implementation of the planned activity;	Information is provided in paragraph 1.4.2 of the EIA Report
	-technical documentation, information, and data confirming the compliance assessment of the proposed equipment with the regulatory documentation for its manufacture (passports, etc.);	Information is provided in paragraph 1.4.2 of the EIA Report
	-data on hazardous factors (substances, materials) to be used at the facility of the planned activity;	Information is provided in paragraph 1.5 of the EIA Report
	-information on the available material and technical base and its technical characteristics necessary for the planned activity.	Information is provided in paragraphs 1.3.3 and 1.3.4 of the EIA Report
2	Provide the calculations for assessing the seismic condition of the territory (location) of the planned activity in the EIA Report. If necessary, identify restrictions in the seismic zone.	Information is considered and provided in paragraph 1.3.7 of the EIA Report
3	Provide detailed information on the assessment of the projected impact on the geological environment and tectonics, as well as materials on the studies of karst phenomena at the planned activity in the EIA Report.	Information considered and provided in paragraph 1.4.1 of the EIA Report
	Describe the engineering and geological zonation of the territory according to the risk of landslide and collapse processes, as well as the features of their development.	
	Provide a description of slope stability and expected changes, indicating the type of possible landslide and collapse processes, their location, size, and the magnitude and speed of movement of soil masses (considering indirect consequences caused by landslide and collapse processes (deformation of existing buildings and structures, flooding of valleys during the formation of landslide dams, high waves during rapid displacement of earth masses into the water area, etc.)	
4	Detail the geological structure characteristics, composition and properties of soils within the area of possible impact during construction and operation of the proposed activity, as well as	The information is considered and provided in paragraphs 5.2.1 and 5.3.7.2 of the EIA Report

No.	Requirements of the Ministry of Environment	Information about the consideration
	information on the composition and properties of soils within the compressible stratum at the base of the structure in the EIA Report.	
	Provide information on the categories and soil quality, analysis of the planned activity impacts on soils, considering the occurrence of hazardous geological processes and phenomena and other factors that negatively affect the soil condition.	
	Reflect the measures to prevent or reduce the development of hazardous geological processes and phenomena, in particular, soil erosion in the EIA Report.	
5	Detail the description of the technological process of the planned activity, indicating all factors of impact on the water environment and technical solutions aimed at eliminating or reducing harmful impacts (emissions, releases, leaks into water bodies) in the EIA Report, including measures to prevent or reduce pollutants from entering the water environment, disruption of the hydrodynamic regime, depletion of surface and groundwater resources, water conditions deterioration, degradation of aquatic organisms and possible changes in the water balance of the territory.	Information is provided in paragraphs 1.5.2.2, 5.2.2, 5.3.7.5, and 7.4 of the EIA Report.
	The EIA Report should detail the treatment technology of all types of wastewater, provide information on the results of laboratory monitoring of water sources and water bodies in the affected area, and provide measures for the future control of their condition.	Information is provided in paragraph 5.3.3 of the EIA Report.
	Describe the peculiarities of the hydrological regime of the territory of the planned activity with a description of the water balance (site, catchment, and water body) and forecasting and modeling of extreme floods, development of channel processes, bank erosion, etc.	Information is provided in paragraphs 1.4.4, 3.3, 3.8.1.7 of the EIA Report.
	Include a detailed description of nuclear reactor cooling systems in the EIA Report.	Information is provided in paragraph 1.4.2.2 of the EIA Report.
	Investigate possible thermal contamination of water bodies within the observation zone (30 km).	Information is provided in paragraph 5.3.2 of the EIA Report.
	Provide data and analysis on the impact of the planned activity on the hydrological regime of the Goryn River and groundwater level.	Information is provided in paragraph 1.4.4 of the EIA Report.

No.	Requirements of the Ministry of Environment	Information about the consideration
6	In the EIA Report, in assessing the impact on atmospheric air, calculate the predicted air pollution by modeling the dispersion of pollutant emissions into the atmosphere, including consideration of the preliminary analysis of the existing anthropogenic background and the prospects for its change, and indicate measures to prevent or minimize such pollution.	Information is provided in paragraphs 1.5, 7.4, Appendix C of the EIA Report.
	Provide information on official characteristics certificates on the climatic conditions of the planned activity area (average annual wind speeds in the eight wind rose, etc.) and on the background content of pollutants in the air of the area where the facility is located for the current year.	Information is provided in Appendix D and Appendix E of the EIA Report.
7	Provide information on the current state of the treatment facilities at the enterprise, as well as assess their operation, considering the prospective additional load from the planned activities implementation. Also, include in the EIA Report data with supporting materials on limestone sludge accumulation sites, volumes of limestone sludge release, available sludge SFs with their parametric characteristics, and provide for ways to manage limestone sludge in the long term (if such sludge is available and operations with it are performed).	Information is provided in. Paragraphs 1.4.3.1, 5.3.3 of the EIA Report.
	To detail the relevant section of the EIA Report with information on waste handling (qualitative and quantitative characteristics of waste generated during the implementation of the planned activity, in particular, radwaste at all stages of the nuclear facility life cycle, including site selection, design, construction, and decommissioning).	Information is provided in paragraphs 1.3.7, 1.5.2.3, 5.3.10 of the EIA Report.
	Classify waste under the requirements of the Law of Ukraine "On Waste Management".	Information is provided in paragraph 1.5.2.3 of the EIA Report.
8	The EIA Report includes information on the relation of the planned activity area to the areas and objects of the nature reserve fund (including the forest reserve of local importance "Pralis"), areas reserved for further conservation, areas and objects of the ecological network, cultural heritage sites, sanitary protection and security zones, water intakes, water bodies with indication of their status (including the width of the coastal protection zone, water protection zone, etc.) with the appropriate display of this information on a topographic basis (with the definition of scale).	Information is provided in paragraphs 3.7, and 5.2.3 of the EIA Report.

No.	Requirements of the Ministry of Environment	Information about the consideration
	Include calculations and analysis of the likely impact on these areas and objects, including compensation and environmental protection measures in the EIA Report.	Information is provided in paragraph 7.6 and Appendix C of the EIA Report.
9	Particular attention should be paid to the assessment of the likely impact on flora and fauna (habitats, migration routes, breeding conditions, impacts) in the implementation of the planned activity, including the proposed transport links for the implementation of the planned activity, indicating compensatory and environmental protection measures.	Information is considered and reflected in paragraph 5.2.3 of the EIA Report.
10	Update the EIA Report with information on the company's mandatory policy aimed at continuous monitoring and analysis of NPP safety, including the planned units 5 and 6 at the existing Khmelnytskyy NPP site.	Information is considered and provided in Section 8 of the EIA Report
11	Reflect the envisaged principles of NPP safety (ensuring safety culture; responsibility of the operating organization, state safety regulation; implementation of the defense-in-depth strategy; application of proven engineering and technical practices, quality management; NPP safety self-assessment; safety analysis; departmental supervision; independent inspections; consideration of the human factor; ensuring radiation safety; consideration of operational experience; scientific and technical support, etc.) to specify the mentioned principles and requirements arising from them following the norms, rules and standards on nuclear and radiation safety in the EIA Report.	Information is considered and provided in paragraph 8.6 of the EIA Report.
12	Include in the EIA Report the CCSUP Schedule (Comprehensive (Consolidated) Safety Upgrade Program for NPP Units) based on schedules of preventive maintenance and operations with fresh and spent nuclear fuel (in case of confidentiality, apply part eight of Article 4 of the Law of Ukraine "On Environmental Impact Assessment").	On 24.04.2024, Energoatom sent a letter No. 01- 9098/46 to the Ministry of Environment with a justification for the exclusion of this item from the conditions list.
	Provide data with forecasts of changes in the state of the environment under normal conditions and in case of design basis accidents at NPPs, considering the implementation of the CCSUP.	The purpose of the Comprehensive (Consolidated) Safety Upgrade Program for NPPs is to increase the operational safety level and reliability of Russian-designed WWER power units due to increased requirements for nuclear and radiation safety in connection with the issuance of new or updating of existing regulatory documents on nuclear and radiation safety of Ukraine and considering the

No.	Requirements of the Ministry of Environment	Information about the consideration
		consequences of global events at NPPs, such as the accident at Fukushima Daiichi NPP, which revealed a significant safety deficit of NPPs.
		The design of power units 5 and 6 at Khmelnytskyy NPP will be based on the AP1000 Generation III+ design certified by the US regulatory authority, which is based on fundamental improvements in safety and the ability to respond to extreme events that led to the Fukushima Daiichi accident.
13	Model emergencies with defined zones of possible contamination (including the affected countries), and provide a list of measures to localize and eliminate emergencies.	Information is considered and provided in paragraphs 5.3.8, 5.3.9 and Section 8 of the EIA Report.
14	Include in the EIA Report an activities analysis considering the socio-economic burden of the region, existing ancillary facilities of the NPP and new planned auxiliary facilities.	Information is considered and provided in paragraph 4.10 of the EIA Report.
15	Investigate and include information on socio-economic risks from the planned activities in the EIA Report:	Information is considered and provided in paragraph 5.4 of the EIA Report
	-analysis of possible social conflicts related to the implementation of the planned activity;	
	-display the dynamics of the population level around the facility;	
	-in terms of calculating the economics of environmental management, justify further facility operation	
16	Detail the assessment of likely impacts on human health, in particular, indicating the levels of electromagnetic and radiation pollution.	Information is considered and provided in paragraph 5.4 of the EIA Report.
17	Provide a references list indicating the sources used for descriptions and assessments in the development of the environmental impact assessment report (including the available list of certificates, letters received from competent organizations and services, etc.)	The references list is provided in the EIA Report in the section "References".
Table 10.3 – All comments and suggestions received from the Polish authorities on the scope of environmental impact assessment documentation and information on consideration and incorporation

№	Comments	Information on consideration
	Regional Director for Environme	ental Protection in Lublin
1	Presentation of planned technical, technological, and organizational solutions to counteract the possibility of a design and beyond design accident, aimed at preventing exceeding the permissible level of pollution and environmental impact	Information is provided in paragraph 7.1 of the EIA Report
2	Presentation of information on the severe off-design accident or terrorist situation impact on the health and life of the Polish population	KhNPP is a nuclear and categorized facility Measures for a specific period are envisaged following SBC B.1.1- 5:2007 "Composition, content, development procedure, coordination and approval of the section ENGINEERING AND TECHNICAL MEASURES OF CIVIL PROTECTION (CIVIL DEFENSE) in urban planning documentation for a specific period (for official use (FOU)". Nuclear material, radioactive waste and other ionizing radiation sources that are a potential source of environmental hazard at KhNPP is a subject to physical protection under the Law of Ukraine "On Physical Protection of Nuclear Facilities, Nuclear Materials, Radioactive Waste and Other Sources of Ionizing Radiation". Sections on physical protection are developed based on the design risk and are restricted materials. Also, see paras. 5.3.8, 8.5 of the EIA Report.
3	Summary and conclusions of the submitted analyzes in a form accessible to the public regarding its participation in the transboundary procedure	The project will comply with all legal and organizational frameworks for environmental impact assessment according to the requirements of the Law of Ukraine "On Environmental Impact Assessment" dated 23.05.2017 No. 2059-VIII.

№	Comments	Information on consideration	
		The transboundary environmental impact assessment of the construction of power units 5 and 6 at the Khmelnytskyy NPP site will be carried out following the requirements of the Law of Ukraine "On Environmental Impact Assessment" and the Convention on Environmental Impact Assessment in a Transboundary Context.	
4	Presentation of procedures for informing the Polish party on the current radiological situation and in case of emergencies.	Information is provided in c 8.7 of the EIA Report.	
5	Presentation of information on radwaste and spent nuclear fuel handling.	Information is provided in paras. 1.5.1, 5.3.10, 5.4.1 and 5.7.2.4 of the EIA Report.	
6	Presentation of cumulative impacts considering the possibility of combining impacts	The information is provided in paras. 5.3.8, 5.5 of the EIA Report.	
	The Regional Director of Environmental Protection in Rzeszów identified the following issues to be considered during the documentation		
	preparati	on	
1	• Presentation of the transboundary radioactive contamination assessment with assumptions:	Information is provided in paras 5.3.6, 5.3.8 of the EIA Report	
	• with consideration of the cumulative impact of the planned power units 5 and 6 and other ones operated at a certain time at a certain site;		
	• based on current meteorological conditions for the affected countries;		
	• under normal operating conditions and in the event of an accident at a nuclear power plant.		
2	Presentation of possible emergency and accident scenarios.	Information is provided in Section 7 of the EIA Report	
3	Presentation of the planned preventive measures to be applied at nuclear facilities to avoid accidents and releases of hazardous substances and pollutants into the environment.	The information is provided in Section 8 and paragraph 7.1 of the EIA Report	
4	Presentation of the planned radiological monitoring system	Information is provided in Section 11 of the EIA Report	

N⁰	Comments	Information on consideration	
5	Presentation of the planned technology for storage and handling of spent nuclear fuel	Information is provided in paragraph 5.7.2.4 of the EIA Report	
	The Polish Geological Institute - National Research Institute has ident	tified the following issues to be presented in the documentation	
1	Expected impact of the planned project on the hydrodynamic system of the active groundwater exchange zone	Information is provided in paragraph 5.2.2.2 of the EIA Report	
2	Expected impact of the planned project on the temperature and chemical composition of the active groundwater exchange zone	Information is provided in paragraph 5.2.2.2 of the EIA Report	
3	Expected impact of the planned activity on chemical denudation on soil stability under all power plant facilities	Information is provided in paragraph 5.2.1 of the EIA Report	
4	Proposed method and location for reprocessing and handling spent nuclear fuel and radwaste	The information is provided in paras. 5.3.10, 5.4.1, 5.7.2.4 of the EIA Report	
	The National Nuclear Energy Agency has identified the following issues to be addressed in the prepared documentation		
1	Impact of construction of Units 5 and 6 on the safe operation of Units 1 and 2 and unfinished ones (Units 3,4) including impact on emergency plan measures	Information is provided in paras 5.1.1 and 5.6 of the EIA Report	
2	Interaction with other Khmelnytskyy NPP units after commissioning of Units 5 and 6, including common systems for Units 5 and 6 and Units 1, 2, 3, 4.	Information is provided in paragraph 5.6 of the EIA Report	
3	Organization of radiation monitoring of the environment in the area around Khmelnytskyy Nuclear Power Plant during the construction of power units 5 and 6 and their commissioning.	Information is provided in Section 11 of the EIA Report	
4	Describe how KhNPP Units 5 and 6 will be similar to the AP1000 reference unit and the differences between them. Describe how KhNPP Units 5 and 6 will be similar to the AP1000 reference unit and the differences between them. What will be the impact of Ukrainian nuclear safety and radiological protection requirements on the design changes.	Information is provided in paragraph 8.1 of the EIA Report	
5	Changes in the national power grid and power plant system in connection with the operation of Units 5 and 6. The method of thermal energy utilization	Regarding changes in the national power system: free connections at switchgear are envisaged to enable the connection of prospective consumers. Additional grid construction and construction of	

N⁰	Comments	Information on consideration
	of power units No. 5 and 6, as well as the impact on the existing heat supply	additional powerlines $330 \div 750 \text{ kV} (\Pi \Pi 330 \div 750 \text{ kB})$ will be
	system of the power plant.	implemented according to specific design decisions.
		Changes to the main power output scheme of the power units are
		<u>envisaged</u> - construction and installation of a new 330 kV switchgear:
		<ul> <li>installation of a 750/330 kV autotransformer at the existing open</li> </ul>
		switchgear 750 kV;
		- construction of 330 kV transmission lines.
		existing lines, it is possible to postpone the connection of the 750
		kV Rzeszów–Khmelnytskyy powerline.
		The final decisions will be made at the Design (Project) stage.
		impact on the existing heat supply system of the power plant will be
		made at the Design (Project) stage.
6	The impact of the current situation in Ukraine on the construction and commissioning of power Units 5 and 6, as well as on the AP1000 reactor	The design of the new power units is based on all decisions on Vogtle Unit 4.
	design itself with the reference design.	To provide additional protection against terrorist actions of the aggressor country, the option of placing the switchgear underground will be considered at the design stage.
7	Spent Nuclear Fuel and Radioactive Waste Management from Units 5 and 6 and Impact of Increased Spent Nuclear Fuel and Radioactive Waste on the National Spent Nuclear Fuel and Radioactive Waste Management Plan	Information is provided in paras. 5.7.2.4, 5.3.10 and 1.5.1 of the EIA Report.
8	Types and activity of radionuclides released to the environment, their physical and chemical form, methods, ways, and rates of release to the environment, as well as mechanisms of transfer of radionuclides in the environment, including mechanisms of dispersion and accumulation and their seasonal variability under normal conditions.	Information is provided in paras. 5.3.7, 5.3.8 and 1.5.1 of the EIA Report.

N⁰	Comments	Information on consideration
9	Existing levels of radionuclide concentrations in the environment and their variability, as well as the presence of physical or chemical contaminants that may affect radionuclide transfer.	Information is provided in paragraph 5.3.7 of the EIA Report.
10	Natural and anthropogenic environmental features that will affect the transfer of radionuclides (e.g. geological, hydrological, and meteorological conditions, vegetation, or presence of water bodies, with special attention to features, that may affect the territory of the Polish Republic).	Considering the peculiarities of the natural and anthropogenic area character, the impact on the territory of the Polish Republic is practically absent. Information is provided in paragraph 5.3.8 of the EIA Report.
11	Detailed assumptions, including meteorological data and data on releases of radioactive substances into the environment in the event of a severe accident, as well as the results of forecasts of the radiological situation, including a detailed assessment of the potential impact on public health and the environment in the territory of the Polish Republic	To assess the radiological situation forecasts under the worst-case scenario - the beyond design-basis accident, the transboundary impact on the states, including the Polish Republic, was calculated. Such information is provided in Section 5.3.8 of the EIA. More details of such calculations will be provided in the safety analysis report.
12	Structure of agriculture, exports of agricultural products from the territories affected by the nuclear facility, and export trends of agricultural products.	Information is provided in paragraph .5.4.1 of the EIA Report.
13	Identified critical groups	Information is provided in paragraph 1.5.1 of the EIA Report.
14	The procedure for informing the Polish authorities in the event of an accident at the Khmelnytskyy NPP with the expected forecast of the radiological situation in the Republic of Poland, strategies for communication with residents of transboundary areas, including in the border areas of the Polish Republic, to ensure transparency and access to information.	Information is provided in paragraph 8.7 of the EIA Report.
15	Conclusions on the increase in cumulative environmental impact of the entire NPP after commissioning of the planned units	Information is provided in paragraph 5.5 of the EIA Report.
16	Complex assessment of the impact of a severe accident on all nuclear facilities located at the selected site	The accident consequence assessment, which was performed for the reference AP1000 project and passed the relevant approval procedures with the US regulatory authorities, showed the safety of the implemented project. The EIA Report uses the initial data from this analysis and will be implemented in the KhNPP AP1000 design. In addition, Section 5.3.8

№	Comments	Information on consideration
		of the EIA provides the radiological situation estimates under the worst- case scenario - a design-basis accident at AP1000; and Section 8 provides all the necessary information on the accidents analysis and their consequences. In turn, detailed assessments of the severe accident impact will be performed according to regulatory requirements and will be presented in the safety analysis report (SAR).
	In addition, in the next stages of transboundary consultations, the Polish party is interested in obtaining accurate external emergency plans, as well as access to the results of measurements within the framework of radiation monitoring of the environment in the area around Khmelnytskyy NPP during the construction of Units 5, 6 and their commissioning.	Online data on gamma background values, both during the operation and construction of NPP power units, are displayed on the website www.xaec.org.ua. However, to prevent the site from being disabled due to intensified hacker attacks during the Russian invasion of Ukraine, the site was restricted. Also, the current radiation background at all sites of operating Ukrainian NPPs can be found at the following link: https://energoatom.com.ua/ua/ The population of the 30-kilometer observation zone of the KhNPP branch is informed through social networks, media platforms, regional media, and information portals.
		Ukraine continues to implement the Strategy for the Integrated Automated Radiation Monitoring System (RM) according to the Resolution of the Cabinet of Ministers of Ukraine No. 323-r dated 29.04.2022. To improve the radiation safety level of the population of Ukraine and neighboring countries, increase the reliability and functionality of the radiation monitoring system throughout Ukraine, and integrate the Ukrainian RM system into the European and international RM network, the project "Support for the establishment of an integrated automated radiation monitoring system covering the entire territory of Ukraine" (IARMS) is currently being implemented.

### 11 A SUMMARY OF THE MONITORING AND CONTROL PROGRAMS FOR ENVIRONMENTAL IMPACTS DURING THE IMPLEMENTATION OF THE PLANNED ACTIVITY, AS WELL AS (IF NECESSARY) POST-PROJECT MONITORING PLANS

Ukraine has established a system of state environmental monitoring to ensure the collecting, processing, storage, and analysis of information on the environmental condition, forecasting its changes, and developing scientifically based recommendations for making effective management decisions.

The executive body that implements the state policy in the field of environmental protection, other specially authorized state bodies, as well as enterprises, institutions, and organizations whose activities lead to or may lead to environmental deterioration, shall monitor the state of the environment and the pollution level.

The monitoring system is based on the following principles:

- comprehensiveness / integrated (all components of the environment that may be affected by NPPs are covered and all types of pollution are taken into account)

- closed loop (there is a feedback between the NPP and the environment);
- systemic unity of NPP and environment;
- ensuring the radiation safety of people and the environment.

The monitoring system includes:

- a system for monitoring the radiation situation at the NPP site and the adjacent territory, including elements of the man-made environment;

- Atmospheric air control system;
- Surface and groundwater monitoring system;

- system for monitoring biological objects of surface waters (monitoring biocenoses and populations);

- system for monitoring the condition of silt and bottom sediments;
- system for monitoring geological processes and soil conditions;

 $-\,$  a system for monitoring the condition of the foundations of buildings and structures;

warning system.

The radiation impact of NPPs on the environment, including the man-made environment, is assessed using radiation monitoring equipment that controls both the sources of radionuclides entering the environment (liquid discharges, gas and aerosol emissions, etc.) and the radiation situation at the NPP industrial site and the adjacent territory.

The radiation monitoring system is a set of technical means and organizational measures that perform the following main tasks:

 radiation monitoring of the state protective barriers conditions on the way of radioactive substances and ionizing radiation spreading; – technological systems radiation monitoring;

- the environment radiation monitoring;

– control of organized emissions and discharges;

leak detection in technological equipment;

- radiation dosimetric monitoring of internal and external personnel exposure;

- regulation and control of public exposure based on annual effective calculations and equivalent doses to critical groups;

– preparation of accounting and reporting documents on the radiation situation at NPPs, in the SPZ and AF, and personnel exposure.

The radiation monitoring system is installed permanently and operates in conjunction with regular and ad hoc radiation monitoring programs to help meet relevant regulatory requirements.

The AP1000 radiation monitoring system is functionally divided into two subsystems:

- radiological monitoring and sampling of process media, emissions, and discharges;

- radiation monitoring of the adjacent area.

The radiation monitoring system consists of distributed measuring instruments that include one or more detectors and a special radiation processor. Each one receives averages, stores the measured data, and transmits alarms and data to the plant control system (safety and security monitoring system for safety-related monitors) for monitoring (if necessary), display, and recording. These alarms are categorized into the following alarm levels: low (failure), warning, and high. Selected channels have alarms based on the rate of increase. It is also possible to store measurements.

To solve the above tasks, it is envisaged:

– remote (permanent or periodic) monitoring;

- control via permanently installed local means and portable devices;

- control via sampling methods of controlled environments with subsequent processing and measurement.

Organizationally, the radiation monitoring system includes the following subsystems:

- radiation technological monitoring;
- radiation dosimetric monitoring;
- individual dosimetric monitoring;
- radiation monitoring of the environment.

A special subdivision of Khmelnytskyy NPP performs permanent laboratory monitoring of air quality on the boundary of the SPZ. Air monitoring at Khmelnytskyy NPP includes five observation points located both in the area of the main industrial site (HVAC and greenhouse area) and settlements located at a distance of 4-8 km (Netishyn - sports complex and mill area, Komarivka village). All the obtained data are processed and, based on their results, a consolidated "Report on the Assessment of the Impact of

Non-Radiation Factors on the Environment" is prepared annually, which provides conclusions on the environmental impact of the NPP and formulates directions for improving environmental protection activities.

The impact of industrial and other economic facilities in Netishyn, as well as Khmelnytskyy NPP itself, on the Goryn River, as a source of water supply for the number of man-made objects was minimized by implementing the following measures:

– observations at the NPP industrial site, the territory of Netishyn, and floodplains of the Viliya and Horyn rivers;

- the sewage Disposal System (SDS) collects sanitary waste from the turbine building, annex building, and one of Units 5 and 6. It transports it to the existing household wastewater treatment plant, where it undergoes full biological treatment and post-treatment at the biological facilities and is then drained to the NPP cooling pond. The capacity of the existing treatment facilities is sufficient to accept and treat household wastewater from the designed power units. Thus, the treated wastewater from the industrial site of power units 5 and 6 is not discharged into the Goryn River and does not affect the pollution level in the river;

- the Waste Water System (WWS) collects non-radioactive wastewater from power unit equipment, building floor drains, process fluids, and system flushing residues before treatment. The wastewater is collected in turbine building sumps and sent to an oil separator, where the oil is coagulated and separated from the water and then transported off-site. If the wastewater does not contain oil, it is sent to a special wastewater retention tank (WWRB) for settling and treatment. Wastewater from the tank is pumped to a purge sump. In case of detection of radioactivity in the drainages from the settling tanks, the wastewater is sent from the settling tanks to the WLS for treatment and disposal. The Waste Water System (WWS) and the Passive Containment Cooling System (PCS) discharge to the Storm Drain System (DRS);

- the Storm Drain System (DRS) receives flows from the Gravity and Roof Drainage Collection system (RDS), treated drains from the Waste Water System (WWS) and liquid from the passive cooling system of the Passive Containment Cooling System (PCS) and discharges them from the power units. Wastewater from the Storm Drainage System (DRS) is supposed to be discharged into the inlet channel for use in the NPP cycle, which will reduce the need for fresh additional service water. A radiation detector is located on the common drainage pipeline of the turbine building pit pumps to enable timely shutdown of the wastewater flow and prevent its drainage into the environment.

- the purge flow of the cooling tower basins of the Circulating Water Supply (CWS) and Service Water (SWS) Systems is taken from the pump drainage manifold and directed to the purge sump of the wastewater collection and disposal system with subsequent drainage into the KhNPP cooling pond.

Liquid and solid radioactive waste is treated and stored under the requirements of the laws and regulations in force in Ukraine. The equipment layout, technological operations, and placement of liquid radioactive waste and solid radioactive waste in closed rooms prevent the leakage and emission of radioactive waste into the environment.

Soil monitoring is carried out at Khmelnytskyy NPP under the Regulations for Physical and Chemical Control of Soils, Sludge and Bottom Sediments at KhNPP, in the Sanitary Protection Zone and Observation Zone No. 0.LO.6210.RG-17 and the Schedule for Sampling and Chemical Analysis of Soils, Sludge and Bottom Sediments  $N_0.\Pi O.6210.P\Gamma$ -17 and «Schedule of sampling and chemical analysis of soils, silts and bottom sediments».



By regulatory requirements, the SPZ accommodates NPP auxiliary and maintenance facilities and carries out radiation monitoring.

During the startup of power units, the only potential source of chemical impact on the environment may be the cooling pond that receives treated industrial and household wastewater from the NPP site that does not contain radioactive contaminants, treated industrial and household wastewater (which is directed to the general household sewerage network), and stormwater drains from the NPP site.

The sanitary supervision authorities monitor water quality at the blowdown water drainage point, water quality in the Goryn River before the blowdown water is released, and water quality in the Goryn River at a distance of 500 meters downstream from the drainage point. In case of exceeding the threshold limit value (TLK) for pollutants, the bottom water outlet in the last hydraulic gate valve is closed and the blowdown is stopped.

Radioactive contamination of the territory during the maximum design basis accident and the beyond design basis accident will not change the surface soil layer's physical, chemical, or water-physical properties.

Under normal operating conditions of Khmelnytskyy NPP consisting of six power units and implementation of the project environmental protection measures, no negative impact on the environment will be ensured.

# 12 SUMMARY OF NON-TECHNICAL INFORMATION INTENDED FOR A WIDE AUDIENCE

The mission of the Energy Strategy of Ukraine until 2050 [1] is to create conditions for the sustainable development of the national economy by ensuring access to reliable, sustainable and modern energy sources.

Some of the main measures to achieve strategic goals in the electricity generation are as follows:

• make decisions on the construction of new generating capacities to replace those that will be decommissioned after 2025;

• make decisions and program to replace the NPP capacities that will be decommissioned after 2030;

• select the reactor technologies for the construction of new nuclear power units to replace the NPP capacities that will be decommissioned after 2030.

On August 31, 2021, in Washington, D.C., Petro Kotin, Energoatom President, and Patrick Fragman, President & CEO of Westinghouse Electric Company (WEC), in the presence of President of Ukraine Volodymyr Zelenskyy, signed a Memorandum of Understanding that provides for the construction of new nuclear power units using the AP1000 technology in Ukraine. The pilot project is the construction of two new AP1000 power units at the KhNPP site.

On June 02, 2022, at the KhNPP site, the Declaration on the Start of Energoatom and WEC Joint Project Implementation for the construction of AP1000 power units at the KhNPP site was signed.

The project envisages the construction of two power units with a relevant group of auxiliary buildings and structures within the KhNPP site. The land alienation for KhNPP construction was performed in accordance with engineering design including the future construction of power units. By the time Unit 1 was commissioned, the activities related to land alienation, reclamation and compensation for land occupation were fully completed.

The construction of power units 5&6 ensures:

- energy supply development and electricity availability;
- general improvement of living conditions and welfare of the population;
- employment generation in Netishyn and KhNPP region;
- qualified labor influx;
- social infrastructure development.

The AP1000 has been designed with environmental requirements as a priority. The public and NPP employees safety, as well as the environmental impact, are addressed as follows:

• Design features minimize operational releases.

- Strict targets for staff exposure.
- Reduction in total amount of radioactive wastes.
- Minimized amount of other hazardous waste (non-radioactive).

• AP1000 technology approach to plant safety was specifically designed to maximize the plant's resistance to catastrophic events that result in major infrastructure damage and total loss of power from a common cause, both on and off site.

Radiation protection of the operating personnel and exclusion of environmental contamination in the SRW system is ensured by:

- special equipment for SRW handling (containers, vehicles, etc.);
- mechanical means for handling operations;
- radiation monitoring and decontamination equipment.

All SRW transport and processing operations are performed with radiation monitoring to ensure radiation safety of the plant personnel.

All shields against the radionuclide release into the environment are controlled by the NPP radiation monitoring system.

The main source of air pollution with harmful chemical impurities is the startup boiler house, which, after the startup of power units 5 & 6, can be used only in some emergencies related to the simultaneous shutdown of all power units. Other sources of chemical emissions into the atmosphere are periodic and do not lead to a violation of the regulatory state of the surface layers.

The following measures will be taken to prevent or reduce the flow of pollutants into the aquatic environment, as well as to protect surface waters from depletion during the construction and operation of the facility:

• the NPP has a reverse process water supply system designed to minimize the environmental impact on surface water bodies of general use;

• all NPP wastewater after treatment is directed to the cooling pond and is not discharged directly into public water bodies;

• the NPP cooling pond is a regulating reservoir that allows water intake from the Goryn River only during high-water periods and, accordingly, does not affect the water management situation in the Goryn River basin;

• all NPP effluents entering the cooling pond are subject to hydrochemical control;

• in case of exceeding the level of mineralization and concentration of biogenic elements in the cooling pond, the cooling pond should be purged within the limits of reaching their MPC established by sanitary standards;

• compliance with the temperature conditions in the control points stipulated by the sanitary standards by diluting the blowdown water, controlled by appropriate water temperature measurements;

• compliance with the current legislation requirements for protection and rational use of surface water.

A noise impact from sources located on the industrial site on objects outside the NPP sanitary protection area is within normal limits.

There are no sources of electromagnetic radiation at KhNPP that exceed the limit values established by regulatory documents.

The legislation of Ukraine provides for economic measures to stimulate activities aimed at reducing the environmental impact, namely:

- limits on the use of natural resources and pollutant emissions;
- charge for the use of natural resources and pollutant emissions;

• compensation in accordance with the established procedure for damages caused by violation of applicable law.

The above payments, which are regularly made by KhNPP during the operation, are used to implement measures at the regional level to compensate for environmental damage caused by business activities.

No significant adverse environmental impact is expected from construction activities and implementation of the planned activities. The impact on the environment and population is assessed as acceptable.

# REFERENCES

- 1. "Energy Strategy up to 2050"
- 2. Order of the Cabinet of Ministers of Ukraine "On Approval of the Energy Strategy of Ukraine until 2050" of April 21, 2023, No. 373-p
- 3. "Memorandum of Understanding between the State Enterprise National Nuclear Energy Generating Company Energoatom and Westinghouse Electric Company dated 31.08.2021
- 4. Law of Ukraine "On Environmental Impact Assessment" of 23.05.2017 No. 2059-VIII.
- Order of the Ministry of Environmental Protection and Natural Resources of Ukraine "On Approval of the General Methodological Recommendations on the Content and Procedure for Preparing Environmental Impact Assessment Reports" of March 15, 2021 No. 193
- 6. Decree of the President of Ukraine "On the Establishment of the Male Polissia National Nature Park" of August 02, 2013 No. 420/2013
- 7. Resolution of the Cabinet of Ministers of Ukraine "Some Issues of Preparatory and Construction Works" of April 13, 2011, No. 466
- Resolution of the Cabinet of Ministers of Ukraine "Some Issues of Permitting and Registration Procedures in Construction under Martial Law" of June 24, 2022, No. 722
- 9. DSTU B A.3.1-22: 2013 Determination of the duration for facility construction
- 10. SNiP 1.04.03-85* Norms of construction duration and delay in the construction of enterprises, buildings and structures
- 11. DBN A.3.1-5-2016 Organization of construction production
- 12. DSTU-N B V.2.1-28: 2013 Guidelines for earthworks and arrangement of bases and construction of foundations
- 13. DSTU-N B B.2.1-32: 2014 Guidelines for the design of excavation pits for foundations and buried structures
- 14. DSTU-N B V.2.6-203: 2015 Guidelines for the performance of works in the manufacture and installation of building structures
- 15. DSTU B V.2.6-200: 2014 Metal building structures. Requirements for installation
- 16. WESTINGHOUSE UNP_EGA_000004 Temporary Facilities and Support Requirement Information
- 17. DBN A.2.2-1:2021 Composition and content of environmental impact assessment materials (EIA)
- 18. NP 306.2.144-2008 General Safety Provisions for Nuclear Power Plants

- 19. DBN B.1.1-12:2014 Construction in seismic areas of Ukraine
- 20. DSanPiN 2.2.4-171-10 Hygienic requirements for drinking water intended for human consumption
- 21. SOU NNEGC 173:2020. Ensuring technical safety. Technical requirements for the structure and safe operation of process pipelines and equipment
- 22. DGN 6.6.1.-6.5.001-98 Radiation safety standards of Ukraine. NRBU-97
- 23. DSP 6.177-2005-09-02. State Sanitary Rules "Basic Sanitary Rules for Ensuring Radiation Safety of Ukraine" OSPU-2005.
- 24. NP 306.2.141-2008. General Safety Provisions for Nuclear Power Plants. OPBU-2008
- 25. Constitution of Ukraine of 28.06.1996 No. 254k/96-VR
- 26. Law of Ukraine "On Environmental Protection" of 25.06.1991 No. 1264-XII
- 27.RD 210.006-90 Rules for technological design of nuclear power plants with VVER reactors
- 28. Inv. no. 44973. 43-814.203.004.OE.13.07.05. Feasibility studies. Volume 7.5. Basic technological solutions. Hydraulic engineering part. SE "NNEGC 'ENERGOATOM' SS 'ATMPROJECTENEERING', 2017.
- 29. DSN 3.3.6.037-99 Sanitary standards of industrial noise, ultrasound and infrasound
- 30. DSTU 3.3.6.039-99 "State sanitary norms of industrial general and local vibration"
- 31. RD 95.10444-91 Recommendations for determining the design characteristics of tornadoes when placing NPP
- 32. Westinghouse AP1000 Design Control Document Rev. 19 Tier 2 Chapter 9 Auxiliary Systems Section 9.2 Water Systems.
- 33. Westinghouse AP1000 Design Control Document Rev. 19 Tier 2 Chapter 10 -Steam and Power Conversion System - Section 10.4 Other Features of Steam and Power Conversion Systems
- 34. 46.3.7.KhNPP/23-02.P. Thermal calculation of the KhmNPP cooling pond including the construction of the AP1000 power units 5 and 6, SE NNEGC "ENERGOATOM" SS "ATOMPRJECTENGINEERING", 2023.
- 35. 24.2.42.OB.02 Khmelnytskyi NPP. Unit No. 2. Final Safety Analysis Report. Chapter 2. Characterization of the NPP area and site
- 36. Report on the Strategic Environmental Assessment of the Regional Waste Management Plan in Khmelnytskyi Oblast until 2030 (Draft), Kyiv, 2020
- 37. DSTU-N B V.1.1-27: 2010 Construction climatology.

- 38. Stock data of meteorological observations for 1945-2007 at the Shepetivka weather station: temperature and humidity, precipitation, wind, snow cover, fog, atmospheric phenomena, natural meteorological phenomena: Sectoral State Archive of the Central Geophysical Observatory of the Hydrometeorological Service of Ukraine, 2008.
- 39.Technical report on engineering and geological surveys at the facility: "Construction of power units No. 5, 6 with reactor unit AP 1000 at the Khmelnytski NPP site. The first stage of construction". Geobest Engineering Center LLC, 2024.
- 40. 24.1.42.OB.02.01 Khmelnytskyi NPP. Unit No. 1. Safety analysis report. Chapter 3. Section 1. Characterization of the site area and NPP location area.
- 41. Hygienic regulations for approximate safe levels of exposure to chemical and biological substances in the atmospheric air of populated areas, approved by the Order of the Ministry of Health of Ukraine No. 52 dated January 14, 2020.
- 42. Catalog of tornadoes that occurred in Ukraine and Belarus according to the Ukrainian Hydrometeorological Center. Letter from the Hydrometeorological Center of Ukraine, December 2001.
- 43. GKD 34.20.507-2003 "Technical Operation of Power Plants and Networks. Rules".
- 44. Report on the Environmental Impact Assessment of Non-Radiation Factors at the affiliate SS Khmelnytskyi NPP, JSC NNEGC Energoatom for 2023. JSC NNEGC Energoatom, affiliate SS Khmelnytskyi NPP, 2023.
- 45. Report on the Environmental Impact Assessment of the Non-Radiation Factors of the SS KhNPP, SE NNEGC Energoatom for 2022. SE NNEGC Energoatom, SS KhNPP, 2022.
- 46. WESTINGHOUSE URP-GW-G0R-011 Input data for Khmelnytskyi Feasibility Study. Deliverabke 21.3 (Liquid and gas release points), 2022.
- 47. ANSI S12.2-2019, "American National Standard Criteria for Indoor Noise Evaluation"
- 48. Completed report "Study of the zero background in the area of construction of the Khmelnytskyi Nuclear Power Plant" /topic code 07.01.003.286.82x/. State registration number 01.83.0048665.
- 49. "Report on the state of radiation safety at KhNPP for 2023" No. 174 of 26.02.2024
- 50. WESTINGHOUSE URP-GW-G0R-031 Input data for Khmelnytskyi Feasibility Study. Deliverable 21.1 (Radiological Releases), 2022.
- 51. Radiation Control Regulations of Khmelnytskyi NPP 0.RB.2509. IE-22.
- 52. "AP1000® Plant Design Overview"
- 53. Law of Ukraine "On the Use of Nuclear Energy and Radiation Safety" dated 08.02.1995 No. 39/95-BP (rev. 01.01.2024)

- 54. Report on the research work "Pre-design work on the definition of the sanitary protection zone and supervised area for the facility "Construction of power units 5, 6 with the reactor unit AP1000 at the Khmelnytskyi NPP site" Stage 1. Calculation and determination of the size of the sanitary protection zone of KhNPP Units 5, 6 with the AP1000 reactor unit under design
- 55. SOU NNEGC 023:2014. Radiation safety. Procedure for establishing the NPP sanitary protection zone
- 56. 43-251.203.006.OB.06. Khmelnytskyi NPP, Unit 2. Report on periodic safety review. Safety factor No. 14. Volume 6, Chapter 6: Environmental Impact.
- 57. GN 6.6.1.1-130-2006. State hygienic standards. Permissible levels of 137Cs and 90Sr radionuclides in food and drinking water
- 58. Resolution of the Cabinet of Ministers of Ukraine "Some issues of identification of high-risk objects" of September 13, 2022, No. 1030
- 59. WESTINGHOUSE URP-GW-G0R-049. Revision 0 AP1000[™] Plant Input Data for Khmelnytskyi Feasibility Study Deliverable 21.2 (Hazardous Materials and Release Pathways)
- 60. NP 306.2.173-2011. Requirements for determining the size and boundaries of the supervised area of a nuclear power plant
- 61. OND-86. "Goskomhydromet. Methodology for calculating concentrations in the atmospheric air of harmful substances contained in the emissions of enterprises". Л.: Gidrometizdat, 1987.
- 62. WESTINGHOUSE URP-GW-G0R-029, AP1000 Input data for Khmelnytskyi Feasibility Study. Deliverable 17.1 (HVAC Systems).
- 63. WESTINGHOUSE URP-GW-G0R-018, AP1000 Input data for Khmelnytskyi Feasibility Study. Deliverables 23.1 (Radioactive Waste Management Systems).
- 64. 10 CFR 50, Appendix J Appendix J to Part 50-Primary Reactor Containment Leakage Testing for Water-Cooled Power Reactors
- 65. DBN B.2.5-74:2013 Water supply. External networks and structures. Basic design provisions. With Amendment No. 1
- 66. 40 CFR 112 "Prevention of oil pollution of the environment"
- 67. WESTINGHOUSE ML071580933, AP1000 Design Control Document, Auxiliary Systems
- 68. NUREG-1465, Accident Source Terms for Light-Water Nuclear Power Plants
- 69. Regulatory guide 1.183. Alternative radiological source terms for evaluating design basis accidents at nuclear power reactors
- 70. WESTINGHOUSE SBD_EGA_00004 EA Questionnaire Response
- 71. 10 CFR 20 "Standards for Protection Against Radiation" ("NRC Regulation Title 10, Part 20, Standards for Protection Against Radiation")

- 72. SOU NNEGC 110:2015 Radiation Safety. Methodology for assessing the scale and significance of accidental releases and discharges from nuclear power plants to the environment. General requirements
- 73. NP 306.2.02/3.077-2003 Requirements for internal and external NPP crisis centers
- 74. SaveEcoBot https://www.saveecobot.com/radiation-maps
- 75. Resolution of the Cabinet of Ministers of Ukraine of 24.03.2004 No. 368 "On Approval of the Procedure for Classification of Emergency Situations by Their Levels"
- 76. NP 306.5.01/3.083-2004 Response Plan for Radiation Accidents.
- 77. Report «The impact assessment of the planned activity on construction of new units at KhNPP on populations of rare plant species of international, national and regional protection status, rare plant communities, habitats of the Emerald Network and nature reserve sites». Natural History Museum at the National Academy of Sciences of Ukraine, 2024.
- 78. Report «The impact assessment of the planned activity on construction of new units at KhNPP on the species of animals listed in Resolution 6 (1998) of the Convention on the Conservation of Wild Flora and Fauna and Natural Habitats (Bern Convention)». Schmalhausen Institute of Zoology at the National Academy of Sciences of Ukraine, 2024.
- 79. 0.RB.0139.NR-22. Permissible gas and aerosol emission and water discharge of radioactive substances of Khmelnytskyi NPP (Radiation and Hygiene Regulations of the first group).
- 80. Reports on the radiation situation in the vicinity of Khmelnytskyi NPP for 2013-2021. The KhNPP External Radiation Monitoring Laboratory.
- 81. EA Questionnaire Response with AP1000 Overview and Plant Description. Attachment 4. Supporting Tables with detailed information for the EA Questionnaire, 2022.
- 82. AP1000 Safety Analysis Report Input Document Tier 2 Chapter 15, Revision A. Westinghouse Electric Company LLC, 2023.
- 83. 46.3.7.KhNPP/23-01.P Assessment of the total cooling water demand of KhNPP including the construction of power units 5&6 with the AP1000 reactor unit, ATOMPROJECTENGINEERING, SE NNEGC ENERGOATOM, 2024.
- 84. Construction and forecasting of the water balance of the water management site M5.1.4.45 of the Goryn River from the source to the border of Khmelnytskyi and Rivne regions". Report on research work. UDK: 005; 519.7; 303.732, state registration number 0124U004377/Vinnytsia National Technical University. Kyiv, 2024.

# **RESPONSIBLE EXECUTORS LIST**

Name, surname	Job title	Signature
Oleksandr Kosinskyi	Head of the Design and Technical Department	toel
Mariia Dubovenko	Mariia Dubovenko Head of the Project Implementation Division	
Nataliia Niedielkova	Head of Sector for Design Documentation Issue	
Olha Blokhina Senior Engineer		St.
Vitalii Boiko	Senior Engineer	Het /
Dmytro Cherkas	Senior Engineer	diff
Yaroslav Rekukha	Senior Engineer	AD
Mykhailo Bortnik	Lead Engineer	Totia
Anzhelika Nefedova	Lead Engineer	Ale
Olena Rossovska	Lead Engineer	Alber -
Daria Khokhriakova	Lead Engineer	lan
Natalia Dyma	Engineer of the I category	Dato
Oleksii Sharhin	Engineer of the I category	the

### **APPENDIX A**

### Comments and proposals from the public on the planned activities, scope of studies



МІНІСТЕРСТВО ЗАХИСТУ ДОВКІЛЛЯ ТА ПРИРОДНИХ РЕСУРСІВ УКРАЇНИ (МІНДОВКІЛЛЯ)

> Департамент екологічної оцінки вул. Митрополита Василя Липківського, 35, м. Київ, 03035, 206-31-40, E-mail: info@mepr.gov.ua

> > Ha №____

АКЦІОНЕРНЕ ТОВАРИСТВО «НАЦІОНАЛЬНА АТОМНА ЕНЕРГОГЕНЕРУЮЧА КОМПАНІЯ «ЕНЕРГОАТОМ» 01032, місто Київ, вул. Назарівська, будинок 3

Департамент екологічної оцінки Міністерства захисту довкілля та природних ресурсів України повідомляє, що:

відповідно до Повідомлення про плановану діяльність АТ «НАЕК «ЕНЕРГОАТОМ», яка підлягає оцінці впливу на довкілля (реєстраційний номер справи 5333 у Єдиному реєстрі з оцінки впливу на довкілля), щодо спорудження енергоблоків № 5 і № 6 на майданчику Хмельницької АЕС із застосуванням технічних характеристик реакторної установки типу АР1000 компанії Westinghouse Electric Company з метою подальшої експлуатації та вироблення електроенергії, розпочато процедуру оцінки впливу на довкілля у відповідності до законодавства;

з дня офіційного оприлюднення зазначеного Повідомлення про плановану діяльність до Міністерства захисту довкілля та природних ресурсів України надходили зауваження і пропозиції від громадськості, що додаються.

Звертаємо увагу, що засобами Єдиного реєстру з оцінки впливу на довкілля були подані зауваження та пропозиції до планованої діяльності, обсягу досліджень та рівня деталізації інформації, що підлягає включенню до звіту з оцінки впливу на довкілля ГО «Екоклуб» від 12,04,2024 № 73.

Додатки: на _ арк. в 1 прим.

Директор Департаменту

Інна Теличко 206 31 40



Марина ШИМКУС



Мівдовкіпта №21/21-03/1681-24 від 17.04.2024 КБІГ: Півскус М. О. 17.04.2024 11:59 588:ЕD98:ГР900307B04000000E8PC3400E6DBA600 Седтивукат лійсений з 03.08.2022 00:00 до 02.08.2024 23:59

### Виконавчий комітет Нетішинської міської ради

вул. Шевченка, 1, м. Нетішин, Хмельницька область, 30100, тел.(03842) 9-05-90, 9-00-94 E-mail: <u>miskrada@netishynrada.gov.ua</u> Код ЄДРПОУ 05399231

.2024 №

/2024 на №

від

Міністерство захисту довкілля природних ресурсів України

Про зауваження та пропозиції до повідомлення про планову діяльність, яка підлягає оцінці впливу на довкілля (реєстраційний номер 5333)

Міністерством захисту довкілля та природних ресурсів України оприлюднено 20 березня 2024 року у газеті «Голос України» за № 55 (8334) ПОВІДОМЛЕННЯ про планову діяльність, яка підлягає оцінці впливу довкілля (надалі-Повідомлення). Планова діяльність полягає у спорудженні енергоблоків № 5 та № 6 на майданчику Хмельницької АЕС із застосуванням технічних характеристик реакторної установки АР 1000.

Відповідно до Закону України «Про оцінку впливу на довкілля» пропонуємо у пункті 4. Соціально-економічний вплив планової діяльності Повідомлення викласти у такій редакції:

«Соціально-економічним обґрунтуванням діяльності від провадження планованої діяльності є посилення енергетичної незалежності держави, забезпечення населення та промисловості електричною та тепловою енергією. Реалізація проекту має загальнонаціональне значення, передбачає залучення великої кількості місцевих підрядних організацій, що забезпечить створення нових робочих місць, завантаження виробничих потужностей та економічний розвиток країни.

У процесі здійснення діяльності планується:

1) витратити на соціальний розвиток регіону не менше 10 відсотків від вартості проекту будівництва енергоблоків.

2) забезпечити повне виконання статті 12 Закону України «Про використання ядерної енергії та радіаційну безпеку», зокрема:

-збереження пільг з оплати за спожиту електричну енергію для населення, яке постійно проживає в 30-кілометровій зоні атомних електростанцій;

-спрямування до спеціального фонду місцевого бюджету соціальноекономічної компенсацію ризику населення, яке проживає на території зони спостереження;



-створення та підтримання у справному стані об'єктів спеціальної соціальної інфраструктури, у тому числі захисних споруд, призначених для укриття і захисту населення, техніки та майна від дії радіаційного опромінення у разі радіаційної аварії.

3) надання статусу спеціалізованого медичного закладу, який розташований на території населеного пункту де розташований (розташовується) ядерний об'єкт, з відповідним фінансовим забезпеченням;

4) сплата податку на доходи фізичних осіб суб'єктами господарювання, які виконують роботи щодо добудови нових енергоблоків на майданчику Хмельницької атомної електростанції, до бюджету громади де розташований (розташовується) ядерний об'єкт.»

Впевнені, що Нетішинська міська територіальна громада при умові позитивного впливу на довкілля, отримає усі визначені законами України соціально-економічні компенсації від будівництва енергоблоків.

Міський голова

Олександр СУПРУНЮК

Валентина Кравчук, 9-15-24

#### Виконавчий комітет Нетішинської міської ради

вул. Шевченка, 1, м. Нетішин, Хмельницька область, 30100, тел.(03842) 9-05-90, 9-00-94 E-mail: <u>miskrada@netishynrada.gov.ua</u> Код ЄДРПОУ 05399231

04.04.2024 Nº 01/01-18-18-26 12024 на №

від

Міністерство захисту довкілля та природних ресурсів України

Про зауваження та пропозиції до повідомлення про планову діяльність, яка підлягає оцінці впливу на довкілля (ресстраційний номер 5333)

Міністерством захисту довкілля та природних ресурсів України оприлюднено 20 березня 2024 року у газеті «Голос України» за № 55 (8334) ПОВІДОМЛЕННЯ про планову діяльність, яка підлягає оцінці впливу довкілля (надалі-Повідомлення). Планова діяльність полягає у спорудженні

№ 5 та № 6 на майданчику Хмельницької АЕС із застосуванням технічних характеристик реакторної установки АР 1000.

Відповідно до Закону України «Про оцінку впливу на довкілля» пропонуємо у пункті 4. Соціально-економічний вплив планової діяльності Повідомлення викласти у такій редакції:

«Соціально-економічним обґрунтуванням діяльності від провадження планованої діяльності є посилення енергетичної незалежності держави, забезпечення населення та промисловості електричною та тепловою енергією. Реалізація проекту має загальнонаціональне значення, передбачає залучення великої кількості місцевих підрядних організацій, що забезпечить створення нових робочих місць, завантаження виробничих потужностей та економічний розвиток країни.

У процесі здійснення діяльності планується:

1) витратити на соціальний розвиток регіону не менше 10 відсотків від вартості проекту будівництва енергоблоків.

2) забезпечити повне виконання статті 12 Закону України «Про використання ядерної енергії та радіаційну безпеку», зокрема:

-збереження пільг з оплати за спожиту електричну енергію для населення, яке постійно проживає в 30-кілометровій зоні атомних електростанцій;

-спрямування до спеціального фонду місцевого бюджету соціальноекономічної компенсацію ризику населення, яке проживає на території зони спостереження;



-створення та підтримання у справному стані об'єктів спеціальної соціальної інфраструктури, у тому числі захисних споруд, призначених для укриття і захисту населення, техніки та майна від дії радіаційного опромінення у разі радіаційної аварії.

 надання статусу спеціалізованого медичного закладу, який розташований на території населеного пункту де розташований (розташовується) ядерний об'єкт, з відповідним фінансовим забезпеченням;

4) сплата податку на доходи фізичних осіб суб'єктами господарювання, які виконують роботи щодо добудови нових енергоблоків на майданчику Хмельницької атомної електростанції, до бюджету громади де розташований (розташовується) ядерний об'єкт.»

Впевнені, що Нетішинська міська територіальна громада при умові позитивного впливу на довкілля, отримає усі визначені законами України соціально-економічні компенсації від будівництва енергоблоків.

Міський голова

Олександр СУПРУНЮК

Валентина Кравчук, 9-15-24



### УКРАЇНА УЛАШАНІВСЬКА СІЛЬСЬКА РАДА ШЕПЕТІВСЬКОГО РАЙОНУ ХМЕЛЬНИЦЬКОЇ ОБЛАСТІ

вул.Перемоги, 4, с.Улашанівка, Шепетівського району, Хмельницької області, 30070, тел.0989740679 Е-mail:04405064@mail.gov.ua Код €ДРПОУ 04405064

No 16 59/12 nig OH DA Lich від Ha No

Міністерство захисту довкілля та природних ресурсів України

У газеті «Голос України» №55 (8334) 20.03.2024 року у рубриці оголошення оприлюднено Повідомлення про планову діяльність, яка підлягає оцінці впливу на довкіяля (реєстраційний номер 5333 у Єднному реєстрі з оцінки впливу на довкіяля). Планова діяльність полягає у спорудженні енергоблоків №5 і №6 на майданчику Хмельницької АЕС із застосуванням технічних характеристик реакторної установки АР1000.

Чинним законодавством України передбачено, що громадськості надається можливість надавати будь-які зауваження і пропозиції до планової діяльності, а також взяти участь у громадських слуханнях, що проводяться у режимі відеоконференції у період воєнного стану.

Тому, ознайомившися зі змістом оприлюдненого повідомлення, надаємо наступні пропозиції та зауваження:

 У пункті 4 повідомлення останис речення необхідно викласти у такій редакції: «У процесі здійснення діяльності плануються витрати на сопіальний розвиток регіону у розмірі <u>не менше 10 відсотків (у оприлюдненій редакції до 10 відсотків)</u> від варгості проєкту будівництва», адже насамперед повинен забезпечуватися розвиток та економічне зростання такого регіону, надання гарантій бізнесу та його страхування.

2) Варто зазначити, що окрім позитивних наслідків спорудження такого об'єкту необхідно також прописати і його негативні наслідки для населення 30-кілометрової зони ризику, загроза для здоров'я цієї категорії осіб, загроза для довкілля та екології, зростання ризиків для інвесторів, підприємств та виробництва.

Пункт 4 повідомлення доречно доповнити абзацом 2 такого змісту:

Рішення про будівництво нових енергоблоків також матимуть наступні негативні наслідки:

- загроза для довкілля, що полягає у потенційній нестачі водних ресурсів, адже Українські АЕС неодпоразово зіштовхувалися із проблемою нестачі води для охолодження реакторів. Для їхнього охолодження використовують прісну воду. Більша частина регіонів України мало забезпечена водними ресурсами: 0,14-0,72 тис. м3/рік на одну людину. В Європі усереднений показник на одну людину складає 4,56 тис. м3/рік. На Хмельницькій АЕС у 2016 та 2017 роках додатково підкачували воду з річки Горинь для охолодження двох енергоблоків. Попри це, на ХАЕС планують збудувати ще два додаткових блоки. Це вплине на екосистему річки;



загроза для здоров'я населення 30-кілометрової зони ризнку Хмельницької АЕС.
 Зростання кількості людей на ракові захворювання, серцево-судиниі захворювання тощо;

- зміна клімату. Згідно із проведеним у 2021 році Аналізом впливу кліматичних змін на водні ресурси України, у більшості річкових басейнів вілбудеться зниження серелнього річного стоку у цьому столітті. Необхідно враховувати можливу нестачу води у випадку проєктування та будівництва нових АЕС. Середня річна температури в Україні прогнозовано зросте на 0,8-1,1° С. Також посилиться перерозподіл опадів протягом року: їх стане більше у холодний період, у теплий період, коли зростає потреба в охолодженні АЕС, стане менше;

- безпекові фактори — безпечність будівництва за технологією АР1000. Компанія Westinghouse Electric Company заявляє, що завдяки спрощенню систем, будівництво, експлуатація та обслуговування реакторів АР 1000 є дешевшим (завдяки нижчій безпеці трубопроводів, меншій кількості кабелів та насосів, а також меншій загальній площі будівлі). Проте згідно зі звітом Комісії з ядерного регулювання США про оцінку безпеки атомного реактора за технологією АР 1000, захисний щит реактора не витримає прямого влучання літака, що є особливо важливим в умовах воснного стану;

 економічні фактори, що полягають у ризиках перевитрат та затягнутого будівництва в умовах воснного стану;

Звертаємо увагу, що перед будівництвом нових атомних реакторів потрібно провести дослідження достатності водних ресурсів та вплив на інші об'єкти. Також обов'язково врахувати прогнози зміни клімату в кожному з регіонів, де заплановано будівництво (зростання обсягів викидів CO2 та середньорічної температури, нетипові зміни погоди упродовж сезонів як-то посухи чи повені, зміна якості повітря та води тощо).

З огляду на вказане, закликаємо врахувати вищезазначені пропозиції та зауваження, забезпечити дотримання законних гарантій населення зони ризику, недопущення нехтуванням їхніми правами та інтересами, адже вже котрий рік постіль спостерігається тенденції щодо невиконання положень ст.ст. 12-1, 12-2 Закону України «Про використання ядерної енергії та радіаційної безпеки», призупинення Законами України сфери її дії з 2019 по 2024 роки, що суперечить вимогам ст. 22 Конституції України; неповернення пільг на електроенергію для споживачів тощо. Також розміщення на території регіону об'єкту загальнодержавного значення повинно забезпечувати гарантії для бізнесу та його страхування, залучення інвесторів для розвитку територіальних громад.

Також більш розгорнуті зауваження, пропозиції та вимоги будуть наданні під час обговорення безпосередньо звіту оцінки впливу на довкілля із врахуванням думки громадськості.

Сільський голова 🥂

AR I

Микола ГИЗИМЧУК



#### ПЛУЖНЕНСЬКА СІЛЬСЬКА РАДА

вул. Бортника, 7, с.Плужне, Шепетівський район, Хмельницька область, 30320 E-mail: plugnerada@ukr.net. Код ЄДРПОУ 04406745

04.04.2024 No anger Ha No

від

Міністерство захисту довкілля та природних ресурсів України

У газеті «Голос України» №55 (8334) 20.03.2024 року у рубриці оголошення оприлюднено Повідомлення про планову діяльність, яка підлягає оцінці впливу на довкілля (реєстраційний номер 5333 у Єдиному ресстрі з оцінки впливу на довкілля). Планова діяльність полягає у спорудженні енергоблоків №5 і №6 на майданчику Хмельницької АЕС із застосуванням технічних характеристик реакторної установки АР1000.

Чинним законодавством України передбачено, що громадськості надається можливість надавати будь-які зауваження і пропозиції до планової діяльності, а також взяти участь у громадських слуханнях. що проводяться у режимі відеоконференції у період воєнного стану.

Тому, ознайомившися зі змістом оприлюдненого повідомлення, надаємо наступні пропозиції та зауваження:

1) У пункті 4 повідомлення останнє речення необхідно викласти у такій редакції: «У процесі здійснення діяльності плануються витрати на соціальний розвиток регіону у розмірі <u>не менше 10 відсотків (у оприлюдненій редакції до 10 відсотків)</u> від вартості проскту будівництва», адже насамперед новинен забезпечуватися розвиток та економічне зростання такого регіону, надання гаралтій бізпесу та його страхування.

2) Варто зазначити, що окрім позитивних наслідків спорудження такого об'єкту необхідно також прописати і його пегативні наслідки для населення 30-кілометрової зони ризику, загроза для здоров'я цісї категорії осіб, загроза для довкілля та екології, зростання ризиків для інвесторів, підприємств та виробництва.

Пункт 4 повідомлення доречно доповнити абзацом 2 такого змісту: Рішення про будівництво нових енергоблоків також матимуть наступні негативні наслідки:

- загроза для довкілля, що полягає у потенційній нестачі водних ресурсів, адже Українські АЕС неодноразово зіштовхувалися із проблемою нестачі води для охолодження реакторів. Для їхнього охолодження використовують прісну воду. Більша частина регіонів України мало забезпечена водними ресурсами: 0,14-0,72 тис. м3/рік на одну людину. В Європі усереднений показник на одну людину складає 4,56 тис. м3/рік. На Хмельницькій АЕС у 2016 та 2017 роках додатково підкачували воду з річки Горинь для охолодження двох енергоблоків. Попри це, на ХАЕС планують збудувати ще два додаткових блоки. Це вплине на екосистему річки;

- загроза для здоров'я населення 30-кілометрової зони ризику Хмельницької АЕС. Зростання кількості людей на ракові захворювання, серцево-судинні захворювання тощо: - зміна клімату. Згідно із проведеним у 2021 році Аналізом впливу кліматичних змін на водні ресурси України, у більшості річкових басейнів відбудеться зниження середнього річного стоку у цьому столітті. Необхідно враховувати можливу иестачу води у випадку проєктування та будівництва нових АЕС. Середня річна температури в Україні прогнозовано зросте на 0,8-1,1° С. Також посилиться перерозноділ онадів протягом року:

UB

Міндовкілля №14634/6/24 від 04.04 2024 їх стане більше у холодний період, у теплий період, коли зростає потреба в охолодженні AEC, стане менше;

- безпекові фактори — безпечність будівництва за технологією AP1000. Компанія Westinghouse Electric Company заявляє, що завдяки спрощенню систем, будівництво, експлуатація та обслуговування реакторів AP 1000 є дешевшим (завдяки нижчій безпеці трубопроводів, меншій кількості кабелів та насосів, а також меншій загальній площі будівлі). Проте згідно зі звітом Комісії з ядерного регулювання США про оцінку безпеки атомного реактора за технологією AP 1000, захисний щит реактора не витримає прямого влучання літака, що є особливо важливим в умовах воєнного стану;

 економічні фактори, що полягають у ризиках перевитрат та затягнутого будівництва в умовах воєнного стану;

Звертаємо увагу, що перед будівництвом нових атомних реакторів потрібно провести дослідження достатності водних ресурсів та вплив на інші об'єкти. Також обов'язково врахувати прогнози зміни клімату в кожному з регіонів, де заплановано будівництво (зростання обсягів викидів CO2 та середньорічної температури, нетипові зміни погоди упродовж сезонів як-то посухи чи повені, зміна якості повітря та води тощо).

З огляду на вказане, закликаємо врахувати вищезазначені пропозиції та зауваження, забезпечити дотримання законних гарантій населення зони ризику, недопушення нехтуванням їхніми правами та інтересами, алже вже котрий рік постіль спостерігається тенденції щодо невиконання положень ст.ст. 12-1, 12-2 Закону України «Про використання ядерної енергії та радіаційної безпеки», призупинення Законами України сфери її дії з 2019 по 2024 роки, що суперечить вимогам ст. 22 Конституції України; неповернення пільг на електроенергію для споживачів тощо. Також розміщення на території регіону об'єкту загальнодержавного значення повинно забезпечувати гарантії для бізнесу та його страхувания, залучення інвесторів для розвитку гериторіальних громад.

Крім того, за попереднім опитуванням громадськості, громадськість вкрай занепокоєна та виступає проти будівництва енергоблоків №5 та №6 на майданчику Хмельницької атомної електростанції через систематичне невиконання зазначених вище вимог законодавства, нехтування їхніми законними правами та інтересами.

Також більш розгорнуті зауваження, пропозиції та вимоги будуть наданні під час обговорення безпосередньо звіту оцінки впливу на довкілля із врахуванням думки громадськості.

Сільський голова, Член Асоціації міст України 🔿 Віталій МАРТИНЮК

Ilodanna CEMEHRIPHA 0562312425



### ІЗЯСЛАВСЬКА МІСЬКА РАДА ШЕПЕТІВСЬКОГО РАЙОНУ ХМЕЛЬНИЦЬКОЇ ОБЛАСТІ

вул. Незалежності, 43, м. Ізяслав, Хмельницька область, 30300, тел./факс: (03852) 4-24-96, E-mail:iz.mr@ ukr.net Код СДРПОУ 04060720

<u>04.04 dOd.4</u> № 1613/03-14 На № _____ від

> Міністерство захисту довкілля та природних ресурсів України

У газеті «Голос України» №55 (8334) 20.03.2024 року у рубриці оголошення оприлюднено Повідомлені я про планову діяльність, яка підлягає оцінці впливу на довкілля (реєстраційний номер 5333 у Єдиному реєстрі з оцінки впливу на довкілля). Планова діялья ість полягає у спорудженні енергоблоків №5 і №6 па майданчику Хмельницької АЕС із застосуванням технічних характеристик реакторної установки АР1000.

Чинним законодавством України передбачено, що громадськості надається можливість надавати будь які зауваження і пропозиції до планової діяльності, а також взяти участь у громадських слуханнях, що проводяться у режимі відеоконференції у період воєнного стану.

Тому, ознайомившися зі змістом оприлодненого повідомлення, надаємо наступні пропозиції та зауваження:

1) У пункті 4 повідомлення останнє речення необхідно викласти у такій редакції: «У процесі здійснення дія льності плануються витрати на соціальний розвиток регіону у розмірі не менше 10 відсотків (у оприлюдненій редакції до 10 відсотків) від вартості проєкту будівництва», адже насамперед повинен забезпечуватися розвиток та економічне зростання такого регіону, надання гарантій бізнесу та його страхування.

2) Варто зазначити, що окрім позитивних наслідків спорудження такого об'єкту необхідно також прописати і його негативні наслідки для населення 30кілометрової зони ризику, загроза для здоров'я цієї категорії осіб, загроза для довкілля та екології, фостання ризиків для інвесторів, підприємств та виробництва.

Пункт 4 повідомлення доречно доповнити абзацом 2 такого змісту: Рішення про будівництво нових снергоблоків також матимуть наступні негативні наслідки:

 загроза для довкілля, що полягає у потенційній нестачі водних ресурсів, адже Українські АЕС неоднораз ово зіштовхувалися із проблемою нестачі води для охолодження реакторів. Для їхнього охолодження використовують прісну воду.
 Більша частина регіонів України мало забезпечена водними ресурсами: 0,14-0,72



тис. м3/рік на одну людину. В Європі усереднений показник на одну людину складає 4,56 тис. м3/рік. На Хмельницькій АЕС у 2016 та 2017 роках додатково підкачували воду з річки Горинь для охолодження двох енергоблоків. Попри це, на ХАЕС планують збудувати ще два додаткових блоки. Це вплине на екосистему річки;

- загроза для здоров'я населення 30-кілометрової зони ризику Хмельницької АЕС. Зростання кількості людей на ракові захворювання, серцево-судинні захворювання тощо;

- зміна клімату. Згідно із проведеним у 2021 році Аналізом впливу кліматичних змін на водні ресурси України, у більшості річкових басейнів відбудеться знижения середнього річного стоку у цьому столітті. Необхідно вр вувати можливу нестачу води у випадку просктування та будівництва но AEC. Середня річна температури в Україні прогнозовано зросте на 0,8-1,1° С. Також ься перерозподіл опадів протягом року: їх стане більше у холодний noc період, у теплий період, коли зростає потреба в охолодженні АЕС, стане менше: - безпекові ф ри — безпечність будівництва за технологією AP1000. Компанія Westinghouse Electric C any заявляє, що завдяки спрощенню систем. будівництво, експлуатація та обслуговування реакторів АР 1000 є дешевшим (завляки нижчій безпеці трубопроводів, меншій кількості кабелів та насосів, а також меншій загальній площі будівлі). Проте згідно зі звітом Комісії з ядерного регулювання США про оцінку безпеки атомного реактора за технологією АР 1000, захисний щит реактора не витримає прямого влучання літака, що є особливо в вим в умовах военного стану;

- економічні фактори, що по у ризиках перевитрат та затягнутого будівництва в умовах воєнного стану;

Звертаємо увагу, що перед будівництвом нових атомних реакторів потрібно провести дослідження достатності водних ресурсів та вплив на інші об'єкти. Також обов'язково врахувати прогн зміни клімату в кожному з регіонів, де заплановано будівництво (зростання обсягів викидів СО2 та середньорічної температури, нетипові зміни погоди упродовж сезонів як-то посухи чи повені, зміна якості повітря та води тощо).

З огляду на вказане, закликаємо врахувати вищезазначені пропозиції та зауваження, забезпечити дотримання законних гарантій населення зони ризику, недопущення исхтуванням їхніми правами та інтересами, адже вже котрий рік постіль спостерігається тенденції щодо невиконання положень ст.ст. 12-1, 12-2 стання ядерної енергії та радіаційної безпеки», Закону України «Про ви призупинення Законами України сфери її дії з 2019 по 2024 роки, що суперечить вимогам ст. 22 Конституції України; неповернення пільг на електроенергію для споживачів TOILIO. Також розміщення на території регіону об'єкту ьнодержавного значення повинно забезпечувати гарантії для бізнесу та його 3 страхування, залучення інвесторів для розвитку територіальних громад.

Крім того, за попереднім опитуванням громадськості, громадськість вкрай занепокоєна та виступає проти будівництва енергоблоків №5 та №6 на майданчику Хмельницької атомної електростанції через систематичне невиконання зазначених вище вимог законодавства, нехтування їхніми

законними правами та інтересами.

Також більш розгорнуті зауваження, пропозиції та вимоги будуть наданні під час обговорення безпосередньо звіту оцінки впливу на довкілля із врахуванням думки громадськості.

Сергій ШЛЕГЕЛЬ Міський голова //



#### ШЕПЕТІВСЬКА МІСЬКА РАДА ВИКОНАВЧИЙ КОМІТЕТ

вул. Соборності, 4, м. Шепетівка, Хмельницька обл., 30405, тел. 4-08-89, факс (03840) 4-01-71 E-mail: 04060789@mail.gov.ua Код ЄДРПОУ 04060789 05.04.2024 No 03-241 1319

> Міністерство захисту довкілля та природних ресурсів України

Про надання пропозицій (реєстраційний номер справи 5333 в Ресстрі ОВД)

20.03.2024 року в газеті «Голос України» №55 (8334) опубліковано Повідомлення про планову діяльність, яка підлягає оцінці впливу на довкілля (реєстраційний номер 5333 в Єдиному реєстрі з оцінки впливу на довкілля) АТ «НАЕК «ЕНЕРГОАТОМ». Планова діяльність полягає у спорудженні енергоблоків №5 і №6 на майданчику Хмельницької АЕС із застосуванням технічних характеристик реакторної установки типу АР1000 компанії Westinghouse Electric Company з метою подальшої експлуатації та вироблення електроенергії.

Чинним законодавством України передбачено, що громадськість має право подавати будь-які зауваження чи пропозиції, які, на її думку, стосуються планованої діяльності, без необхідності їх обґрунтування, приймати участь у громадських слуханнях.

Тому, ознайомившись зі змістом оприлюдненого повідомлення, пропонуємо врахувати наступне.

В 1991 році Київським науково-дослідним і проектним інститутом «Атоменергопроект», на підставі техніко-економічного розрахунку, був складений перелік населених пунктів, що входять до 30-кілометрової зони навколо Хмельницької атомної електростанції (далі -ХАЕС). При уточненні координат межі 30-кілометрової зони навколо ХАЕС було визначено, що до переліку з 125 населених пунктів, що входять до даної зони не включено місто Шепетівка, яке знаходиться на відстані 27,5 км. від станції та ще чотири населених пункти колишніх Шепетівського та Славутського районів, в т.ч. село Плесна, яке на даний час входить до складу Шепетівської міської територіальної громади. За результатами топогеодезичних робіт, виконаних з врахуванням змін меж м.Шепетівка, визначених Постановою Верховної Ради від 03.02.1998 року №64/98-ВР, майже 30% території міста знаходиться в 30-кілометровій зоні ХАЕС.

В 2007 році ВАТ «Київський науково – дослідний та проектно-конструкторський інститут «ЕНЕРГОПРОЕКТ» спільно з Українським науково – дослідним інститутом екологічних проблем (УкрНДІЕП) Міністерства охорони навколишнього природного середовища України, м. Харків, на замовлення виконавчого комітету Шепетівської міської ради, розробили «Науково – економічного обґрунтування необхідності включення м.Шепетівки до 30-кілометрової зони спостереження Хмельницької АЕС». У висновках даного обґрунтування зазначено:

- близько 30% території міста знаходиться в 30-кілометровій зоні ХАЕС;

- у межах 30-кілометрової зони розташована приблизно третина житлового фонду міста, основна зона рекреації, дитячий оздоровчий табір, новий питний водозабір «Лісова галявина», водогін від Кам'янківського питного водозабору, а також ряд підприємств, організацій, установ;

- у розділі 4.1 обгрунтування доведено, що радіаційний вплив ХАЕС на усю територію м.Шепетівка та на всіх її мешканців є однаковим і тому ділити місто на дві частини (межею 30-кілометрової зони) за ознаками інтенсивності радіаційного впливу від ХАЕС не має ніякого сенсу;

UB



- близько третини населення міста ( майже 14 тис. мешканців ) вже зараз має житло у районах, які знаходяться на відстані менше 30 км. від Хмельницької АЕС. В цій зоні знаходяться і перспективні житлові масиви, населення яких становитиме більше 7 тис. чоловік;

- аналіз природних факторів в районі Шепетівки доводить, що радіаційний вплив на мешканців Шепетівки є не меншим, ніж для мешканців найближчих населених пунктів, включених до 30-кілометрової зони ( до прикладу село Пліщин, яке фактично межує з містом, включене до Переліку населених пунктів, що входять до 30-кілометрової зони навколо Хмельницької АЕС і на даний час входить до складу населених пунктів Шепетівської міської територіальної громади);

- існує тісний комунікаційний зв'язок між районами міста, розташованими всередині і зовні території 30-кілометрової зони, а також між містом і позаміськими територіями в 30кілометровій зоні. Все це доводить наявність спільної дії факторів, пов'язаних з експлуатацією ХАЕС, на все населення м.Шепетівки і, відповідно, однаковість впливів на населення міста і інших населених пунктів 30-кілометрової зони.

Державну санітарно - епідеміологічну експертизу вищезазначеного науково – економічного обґрунтування провів у 2007 році Інститут гігієни та медичної екології ім. О.М. Марзєєва Академії медичних наук України. За висновками державної санітарно – епідеміологічної експертизи Міністерства охорони здоров'я України (№05.03.02-07/47840 від 25.09.2007р.) визнано за можливе включення м. Шепетівки до 30 – кілометрової зони Хмельницької АЕС. Слід зазначити що даний висновок дійсний без обмеження терміну дії.

Пунктом 2.1 Вимог щодо визначення розмірів і меж зони спостереження атомної електричної станції, затверджених спільним наказом Державної інспекції ядерного регулювання України, Міністерства охорони здоров'я України №153/766 від 07.11.2011р. (в редакції наказу Державної інспекції ядерного регулювання України №206/765 від 23.11.2015р.) передбачено, що у разі проходження зовнішньої межі зони спостереження (далі – 3С) по населеному пункту, межа 3С коригується таким чином, щоб весь населений пункт, у межах його адміністративного кордону, входив до 3С АЕС.

Враховуючи вищезазначене закликаємо розглянути та вирішити питання <u>включення</u> <u>міста Шепетівки та села Плесна, яке входить до складу Шепетівської МТГ до Переліку</u> <u>адміністративно-територіальних одиниць</u>, що входять до 30-кілометрової зони спостереження Хмельницької AEC.



Bully

Віталій БУЗИЛЬ



# СЛАВУТСЬКА МІСЬКА РАДА ХМЕЛЬНИЦЬКОЇ ОБЛАСТІ ВИКОНАВЧИЙ КОМІТЕТ

вул. Соборності, 7, м. Славута, Хмельницька обл., 30000 тел. 7-11-68, 7-11-66, факс 7-12-33 E-mail: <u>mail@slavuta-mvk.gov.ua</u>, веб-сайт: www.slavuta-mvk.gov.ua код ЄДРПОУ 23563639

U. 04. 2024 № 04-29 519 Ha No

Міністерство захисту довкілля та природних ресурсів України

від

У газеті «Голос України» №55 (8334) 20.03.2024 року у рубриці оголошення оприлюднено Повідомлення про планову діяльність, яка підлягає оцінці впливу на довкілля (реєстраційний номер 5333 у Єдиному реєстрі з оцінки впливу на довкілля). Планова діяльність полягає у спорудженні енергоблоків №5 і №6 на майданчику Хмельницької АЕС із застосуванням технічних характеристик реакторної установки АР1000.

Чинним законодавством України передбачено, що громадськості надається можливість надавати будь-які зауваження і пропозиції до планової діяльності, а також взяти участь у громадських слуханнях, що проводяться у режимі відеоконференції у період воєнного стану.

Тому, ознайомившися зі змістом оприлюдненого повідомлення, надаємо наступні пропозиції та зауваження:

1) У пункті 4 повідомлення останнє речення необхідно викласти у такій редакції: «У процесі здійснення діяльності плануються витрати на соціальний розвиток регіону у розмірі <u>не менше ГІ відсотків (у оприлюдненій редакції до 10 відсотків)</u> від вартості проєкту будівництва», адже насамперед повинен забезпечуватися розвиток та економічне зростання такого регіону, надання гарантій бізнесу та його страхування.

2) Варто зазначити, що окрім позитивних наслідків спорудження такого об'єкту необхідно також прописати і його негативні наслідки для населення 30-кілометрової зони ризику, загроза для здоров'я цієї категорії осіб, загро за для довкілля та екології, зростання ризиків для інвесторів, підприємств та виробництва.

Пункт 4 повідомлення доречно доповнити абзацом 2 такого змісту:

Рішення про будівництво нових енергоблоків також матимуть наступні негативні наслідки:

- загроза для довкілля, що полягає у потенційній нестачі водних ресурсів, адже Українські АЕС неодноразово зіштовкувалися із проблемою нестачі води для охолодження реакторів. Для їхнього охолодження використовують прісн у воду. Більша частина регіонів України мало забезпечена водними ресурсами: 0,14-0,72 тис. м3/рік на одну людину. В Європі усереднений показник на одну людину складає 4,56 тис. м3/рік. На Хмельницькій АЕС у 2016 та 2017 роках додатково підкачували воду з річки Горинь для охолодження двох енергоблоків. Попри це, на ХАЕС планують збудувати ще два додаткових блоки. Це вплине на екосистему річки;

загроза для здоров'я населення 30-кілометрової зони ризику Хмельницької АЕС.
 Зростання кількості людей на ракові захворювання, серцево-судинні захворювання тощо;

UB Міндовкілля №15368/6/24 від 09.04.2024 - зміна клімату. Згідно із проведеним у 2021 році Аналізом впливу кліматичних змін на водні ресурси України, у більшості річкових басейнів відбудеться зниження середнього річного стоку у цьому столітті. Необхідно враховувати можливу нестачу води у випадку проєктування та будівництва нових АЕС. Середня річна температури в Україні прогнозовано зросте на 0,8-1,1° С. Також посилиться перерозподіл опадів протягом року: їх стане більше у холодний період, у теплий період, коли зростає потреба в охолодженні АЕС, стане менше;

- безпекові фактори — безпечність будівництва за технологією АР1000. Компанія Westinghouse Electric Company заявляє, що завдяки спрощенню систем, будівництво, експлуатація та обслуговування реакторів АР 1000 є дешевшим (завдяки нижчій безпеці трубопроводів, меншій кількості кабелів та насосів, а також меншій загальній площі будівлі). Проте згідно зі звітом Комісії з ядерного регулювання США про оцінку безпеки атомного реактора за технологією АР 1000, захисний щит реактора не витримає прямого влучання літака, що є особливо важливим в умовах воєнного стану;

 економічні фактори, що полягають у ризиках перевитрат та затягнутого будівництва в умовах воєнного стану;

Звертаємо увагу, що перед будівництвом нових атомних реакторів потрібно провести дослідження достатності водних ресурсів та вплив на інші об'єкти. Також обов'язково врахувати прогнози зміни клімату в кожному з регіонів, де заплановано будівництво (зростання обсягів викидів CO2 та середньорічної температури, нетипові зміни погоди упродовж сезонів як-то посухи чи повені, зміна якості повітря та води тощо).

З огляду на вказане, закликаємо врахувати вищезазначені пропозиції та зауваження, забезпечити дотримання законних гарантій населення зони ризику, недопущення нехтуванням їхніми правами та інтересами, адже вже котрий рік постіль спостерігається тенденції щодо невиконання положень ст.ст. 12-1, 12-2 Закону України «Про використання ядерної енергії та радіаційної безпеки», призупинення Законами України сфери її дії з 2019 по 2024 роки, що суперечить вимогам ст. 22 Конституції України; неповернення пільг на електроенергію для споживачів тощо. Також розміщення на території регіону об'єкту загальн одержавного значення повинно забезпечувати гарантії для бізнесу та його страхування, залучення інвесторів для розвитку територіальних громад.

Крім того, за попереднім опитуванням громадськості, громадськість вкрай занепокоєна та виступає проти будівництва енергоблоків №5 та №6 на майданчику Хмельницької атомної електростанції через систематичне невиконання зазначених вище вимог законодавства, нехтування їхніми законними правами та інтересами.

Також більш розгорнуті зауваження, пропозиції та вимоги будуть наданні під час обговорення безпосередньо звіту оцінки впливу на довкілля із врахуванням думки громадськості.

#### Міський голова,

Голова Секції з питань захисту прав та інтересів монофункціональних міст і їх населення Асоціації міст України

Василь СИДОР

Анна ТАЛАШОК (03842)7-11-68

От:	Управління економічного розвитку Шепетівський BK <shepeconom@ukr.net></shepeconom@ukr.net>	
Отправлено:	16 квітня 2024 р. 14:08	
Кому:	ОВД	
Тема:	Про надання пропозицій (реєстраційний номер справи 5333 в реєстрі ОВД)	
Вложения:	Лист.pdf	

Виконавчий комітет Шепетівської міської ради повторно надсилає пропозиції по реєстраційній справі про оцінку впливу на довкілля планової діяльності 5333 в Єдиному реєстрі з оцінки впливу на довкілля.


#### ШЕПЕТІВСЬКА МІСЬКА РАДА

ВИКОНАВЧИЙ КОМІТЕТ

вул. Соборності, 4, м. Шепетівка, Хмельницька обл., 30405, тел. 4-08-89, факс (03840) 4-01-71 E-mail: <u>04060789@mail.gov.ua</u> Код ЄДРПОУ 04060789

№_____

Міністерство захисту довкілля та природних ресурсів України

Про надання пропозицій (реєстраційний номер справи 5333 в Реєстрі ОВД)

20.03.2024 року в газеті «Голос України» №55 (8334) опубліковано Повідомлення про планову діяльність, яка підлягає оцінці впливу на довкілля (реєстраційний номер 5333 в Єдиному реєстрі з оцінки впливу на довкілля) АТ «НАЕК «ЕНЕРГОАТОМ». Планова діяльність полягає у спорудженні енергоблоків №5 і №6 на майданчику Хмельницької АЕС із застосуванням технічних характеристик реакторної установки типу AP1000 компанії Westinghouse Electric Company з метою подальшої експлуатації та вироблення електроенергії.

Чинним законодавством України передбачено, що громадськість має право подавати будь-які зауваження чи пропозиції, які, на її думку, стосуються планованої діяльності, без необхідності їх обгрунтування, приймати участь у громадських слуханнях.

Тому, ознайомившись зі змістом оприлюдненого повідомлення, пропонуємо врахувати наступне.

В 1991 році Київським науково-дослідним і проектним інститутом «Атоменергопроект», на підставі техніко-економічного розрахунку, був складений перелік населених пунктів, що входять до 30-кілометрової зони навколо Хмельницької атомної електростанції (далі – ХАЕС). При уточненні координат межі 30-кілометрової зони навколо ХАЕС було визначено, що до переліку з 125 населених пунктів, що входять до даної зони не включено місто Шепетівка, яке знаходиться на відстані 27,5 км. від станції та ще чотири населених пункти колишніх Шепетівського та Славутського районів, в т.ч. село Плесна, яке на даний час входить до складу Шепетівської міської територіальної громади. За результатами топогеодезичних робіт, виконаних з врахуванням змін меж м.Шепетівка, визначених Постановою Верховної Ради від 03.02.1998 року №64/98-ВР, майже 30% території міста знаходиться в 30-кілометровій зоні ХАЕС.

В 2007 році ВАТ «Київський науково – дослідний та проектно-конструкторський інститут «ЕНЕРГОПРОЕКТ» спільно з Українським науково – дослідним інститутом екологічних проблем (УкрНДІЕП) Міністерства охорони навколишнього природного середовища України, м. Харків, на замовлення виконавчого комітету Шепетівської міської ради, розробили «Науково – економічного обґрунтування необхідності включення м.Шепетівки до 30-кілометрової зони спостереження Хмельницької АЕС». У висновках даного обґрунтування зазначено:

близько 30% території міста знаходиться в 30-кілометровій зоні ХАЕС;

 у межах 30-кілометрової зони розташована приблизно третина житлового фонду міста, основна зона рекреації, дитячий оздоровчий табір, новий питний водозабір «Лісова галявина», водогін від Кам'янківського питного водозабору, а також ряд підприємств, організацій, установ;



ВИКОНАВЧИЙ КОМІТЕТ ШЕПЕТІВСЬКОЇ МІСЬКОЇ РАДИ 03-24/1319 від 05.04.2024 Бузиль Віталій Володимирович 05.04.2024 10:43

Міський голова

- у розділі 4.1 обгрунтування доведено, що радіаційний вплив ХАЕС на усю територію м.Шепетівка та на всіх її мешканців є однаковим і тому ділити місто на дві частини (межею 30-кілометрової зони) за ознаками інтенсивності радіаційного впливу від ХАЕС не має ніякого сенсу;

- близько третини населення міста ( майже 14 тис. мешканців ) вже зараз має житло у районах, які знаходяться на відстані менше 30 км. від Хмельницької АЕС. В цій зоні знаходяться і перспективні житлові масиви, населення яких становитиме більше 7 тис. чоловік;

- аналіз природних факторів в районі Шепетівки доводить, що радіаційний вплив на мешканців Шепетівки є не меншим, ніж для мешканців найближчих населених пунктів, включених до 30-кілометрової зони ( до прикладу село Пліщин, яке фактично межує з містом, включене до Переліку населених пунктів, що входять до 30-кілометрової зони навколо Хмельницької АЕС і на даний час входить до складу населених пунктів Шепетівської міської територіальної громади);

- існує тісний комунікаційний зв'язок між районами міста, розташованими всередині і зовні території 30-кілометрової зони, а також між містом і позаміськими територіями в 30-кілометровій зоні. Все це доводить наявність спільної дії факторів, пов'язаних з експлуатацією ХАЕС, на все населення м.Шепетівки і, відповідно, однаковість впливів на населення міста і інших населених пунктів 30-кілометрової зони.

Державну санітарно - епідеміологічну експертизу вищезазначеного науково – економічного обґрунтування провів у 2007 році Інститут гігієни та медичної екології ім. О.М. Марзєєва Академії медичних наук України. За висновками державної санітарно – епідеміологічної експертизи Міністерства охорони здоров'я України (№05.03.02-07/47840 від 25.09.2007р.) визнано за можливе включення м. Шепетівки до 30 – кілометрової зони Хмельницької АЕС. Слід зазначити що даний висновок дійсний без обмеження терміну дії.

Пунктом 2.1 Вимог щодо визначення розмірів і меж зони спостереження атомної електричної станції, затверджених спільним наказом Державної інспекції ядерного регулювання України, Міністерства охорони здоров'я України №153/766 від 07.11.2011р. (в редакції наказу Державної інспекції ядерного регулювання України №206/765 від 23.11.2015р.) передбачено, що у разі проходження зовнішньої межі зони спостереження (далі – ЗС) по населеному пункту, межа ЗС коригується таким чином, щоб весь населений пункт, у межах його адміністративного кордону, входив до ЗС АЕС.

Враховуючи вищезазначене закликаємо розглянути та вирішити питання включення

спостереження Хмельницької АЕС.

Міський голова

Віталій БУЗИЛЬ

Олександр ГРИНЬ (03840) 4-03-93





Міністерство захисту довкілля та природних ресурсів України 03035, м. Київ, вул. Митрополита Василя Липківського,35 Відділу оцінки впливу на довкілля Info@mepr.gov.ua, OVD@mepr.gov.ua

Вих. №73 від 12.04.2024 р.

#### Зауваження і пропозиції до планованої діяльності, обсягу досліджень та рівня деталізації інформації для Звіту з ОВД, реєстраційний № 5333

Згідно з пунктом 7 статті 5 Закону України «Про оцінку впливу на довкілля» громадськість може надати зауваження і пропозиції до планованої діяльності, обсягу досліджень та рівня деталізації інформації, що підлягає включенню до звіту з оцінки впливу на довкілля.

Враховуючи вищенаведене, надаємо зауваження і пропозиції до планованої діяльності, обсягу досліджень та рівня деталізації інформації для Звіту з ОВД планованої діяльності АТ «НАЦІОНАЛЬНА АТОМНА ЕНЕРГОГЕНЕРУЮЧА КОМПАНІЯ «ЕНЕРГОАТОМ», щодо спорудженні енергоблоків № 5 і № 6 на майданчику Хмельницької АЕС із застосуванням технічних характеристик реакторної установки типу AP1000 компанії Westinghouse Electric Company з метою подальшої експлуатації та вироблення електроенергії (ресстраційний номер справи в Єдиному реєстрі з ОВД: 5333).

Пропонусмо включити до Звіту з ОВД наступну інформацію:

Аналіз вибору реакторної установки, що включає: 1.

 Обгрунтування та процедуру вибору реакторної установки AP1000 компанії Westinghouse Electric Company;

 обгрунтування можливості вибору альтернативної реакторної установки включно з визначенням впливу на навколишнє середовище від роботи альтернативної установки:

 інформацію про розгляд інших технічних альтернатив для отримання відповідної кількості електричної енергії, включно із варіантом відмови від будівництва блоків № 5 та





6 ХАЕС та варіантом використання відновлювальних джерел енергії для отримання такої ж кількості електричної енергії.

Вибір майданчика для розташування енергоблоків ХАЕС 5,6:

 обгрунтування та процедуру вибору майданчика для розташування енергоблоків XAEC 5.6

 обтрунтування можливості вибору альтернативного місця розташування (майданчика) для розміщення енергоблоків ХАЕС 5,6;

- перелік та результати досліджень, які були проведені для визначення придатності обраної території (майданчика) під будівництво 5-го та 6-го енергоблоків ХАЕС.

3. Характеристику проектування, що включає:

 інформацію щодо стадії розроблення або стадії проєктування, на якій перебуває об'єкт планована діяльність, за наявності - прийняту на даному етапі проєктну документацію;

 інформацію, щодо стапу, на якому знаходяться передпроєктні (передінвестиційні) роботи;

дати завершення всіх стадій проєктування.

4. Характеристику альтернативи в аспекті її допустимості, доцільності та ефективності, розглянути її переваги і недоліки за техніко-економічними показниками (додаючи, за наявності, техніко-скономічний розрахунок), екологічною безпекою, обсягами споживання природних ресурсів на одиницю продукції.

 Деталізувати місце провадження планованої діяльності та розташування основних об'єктів матеріально-технічної бази (що включає будівлі і споруди, інженерні і транспортні мережи, інші графічні матеріали у складі просктно-планувальних рішень, на яких відображені схеми розташування будівель і споруд) планованої діяльності на наступних матеріалах:

на генеральному плані території та/або детальному плані території;

на ортофотопланах (супутникові знімки) високої роздільної здатності;

схемі (плані).

На картографічних матеріалах пропонуємо відобразити наступну інформацію:

- споживані земельні ресурси;
- 30-кілометрову зону ХАЕС;
- межі промислового майданчику;
- розташування та параметри будівель, доріг та інших об'єктів;
- розташування джерел викидів забруднюючих речовин;

 розташування найближчої житлової забудови по відношенню до об'єкту планованої ліяльності.



Також зауважуємо на тому, щоб картографічні матеріали, при їх додаванні до Звіту в електронному форматі, зберігали високу роздільну здатність і якість.

6. Опис проведення підготовчих і будівельних робіт, тривалість їх виконання; зміст, основних засобів і технологій робіт, пов'язаних з видаленням зелених або інших насаджень, з інженерною підготовкою і захистом земель, зміною рельєфу, відведенням поверхневих стічних вод і грунтових вод; запланованих тимчасових споруд, транспортних та інженерних мереж.

7. Опис планованої діяльність, а саме:

 виробничих процесів, що є змістом планованої діяльності, крім того у характеристиці процесу або устаткування рекомендується зазначати прогнозовану виробничу продуктивність, максимальну внтрату сировини або інших матеріалів за годину, інші параметри виробничої продуктивності;

 річної потреби у сировині, паливі і пальному, інших матеріалах, у воді (у відповідних одиницях виміру);

 - річної потреби води для охолодження блоків в комбінаціях: а) енергоблоків № 5, 6 ХАЕС б) енергоблоків № 1,2 та 5, 6 ХАЕС в) енергоблоків № 1,2, 3, 4, 5, 6 ХАЕС

термін експлуатації об'єкта планованої діяльності;

характеристику ядерно-паливного циклу з його описом;

 відповідність планованих енергоблоків № 5, 6 ХАЕС експлуатаційним вимогам та вимогам безпеки, визначеними Міжнародним агентством з атомної енергії та Західносвропейської асоціації ядерного нагляду WENRA;

 організаційно-технічних і технологічних рішень з охорони навколишнього природного середовища і раціонального використання природних ресурсів, що прийняті планованою діяльністю і на які передбачені витрати, а також відновлювальних та компенсаційних заходів (якщо передбачено).

 Характеристику технологічного устаткування, а саме таку, що буде збудовано, реконструйовано, переоснащено чи демонтовано (замінено) у зв'язку з планованою діяльністю, і при цьому є:

джерелом утворення забруднюючих речовин;

 джерелом значного впливу фізичних факторів (шуму, інфра- або ультразвуку, які поширюються за межі виробничих корпусів і проммайданчика, значного електромагнітного випромінювання згідно з державними санітарними нормами і правилами захисту населення від впливу електромагнітних випромінювань, іонізуючого випромінювання, );

с накопичувачами забруднюючих речовин, твердих і рідких відходів;

належить до об'єктів, що згідно з Законом підлягають ОВД.



 Оцінку (характеристику) впливу на компоненти довкілля пропонуємо розглядати за наступними варіантами:

 – з врахуванням впливу від експлуатації енергоблоків №1-2 та планованих енергоблоків № 5, 6 ХАЕС;

-з врахуванням впливу від експлуатації енергоблоків № 1-4 ХАЕС (енергоблоки № 3, 4 ХАЕС відповідно до Висновку з ОВД, реєстраційний номер справи 2018111232231) та планованих енергоблоків № 5, 6 ХАЕС.

10. Характеристику відходів із зазначенням їх найменування або коду згідно з державними класифікаторами, річний обсяг відходів (а також, за наявності, питомий показник утворення, прийнятий для об'єкта планованої діяльності), обсяг розміщення на власних місцях видаления відходів, обсяг утилізації, обсяг передачі іншим організаціям на утилізацію або розміщення (захоронення).

11. Характеристику водокористування та водовідведення на предмет типу цих систем, обсягів за рік, якісного складу стічних та інших зворотних вод за природними фізико-хімічними і хімічними показниками води і за вмістом забруднюючих речовин, облаштування засобами спорудах для очищення стічних вод. Водокористування і забір води характеризувати на предмет типу системи водопостачання, типу джерела (джерел), найменування і типу водного об'єкта – джерела постачання, категорії якості води, обсягу забору води за рік у розрізі джерел водопостачання і нерівномірності споживання за місяцями.

12. Надати у звіті з ОВД розрахунок водогосподарського балансу по річці Горинь з урахуванням експлуатації 6-х енергоблоків ХАЕС (№1-6) та 4-х енергоблоків ХАЕС (№1-2, 5-6) з врахуванням терміну їх експлуатації та тенденції зміни водності річки Горинь внаслідок зміни клімату і врахуванням наявної та прогнозною господарською діяльності в межах басейну річки Горинь.

 Надати гідротермічні розрахунки водосховища охолоджувача на період експлуатації 6-х енергоблоків ХАЕС (№1-6) та 4-х енергоблоків ХАЕС (№1-2, 5-6) з урахуванням терміну їх експлуатації.

14. Характеристику земельних ділянок на предмет: площ відведених або таких, що будуть відведені (вилучені) під плановану діяльність, земельних ділянок (рекомендований перелік – згідно з кадастровими номерами), категорії і цільового призначення на планований та існуючий стани, обмежень у використанні земель на існуючий і планований стан згідно із землевпорядною документацією, заходів інженерної підготовки і захисту земель, строків, обсягів і технологій рекультивації порушених земель, управління родючим шаром грунту на землях, що будуть порушені.



15. Характеристику змін у мікро- або мезокліматі місцевості у зв'язку з планованою діяльністю, наприклад, у температурному режимі, вологості повітря, замерзанні грунту чи води, настанні туманів або ймовірності штучного туманоутворення над житловою зоною та ін. Додатково оцінити вразливість планованої діяльності в цілому або її окремих технологічних процесів чи об'єктів до несприятливих наслідків зміни клімату, таких як зростання середніх температур, збільшення нерівномірності водного стоку рік і падіння їх водності тощо.

16. Характеристику впливу енергоблоків 5 та 6 на рослинний і тваринний світ. Оцінку кумулятивного впливу на рослинний і тварриний світ від роботи одночасно шести енергоблоків ХАЕС, що включас:

- визначення біоіндикаторов для екологічних оцінок;
- визначення зооіндикаторів для екологічних оцінок;
- оцінку зміни складу лісів; популяцій рослин і тварин;

 оцінку можливої деструкції популяцій тварин, руйнування екосистем, часткова чи повна ліквідація арсалів;

оцінку зміни складу популяцій водних рослин і тварин

17. Характеристику викидів, а саме: перелік стаціонарних джерел із зазначенням їх типу, у тому числі стаціонарних майданчиків, на яких регулярно (щодня) експлуатуються пересувні джерела викидів (будівельно-монтажна спецтехніка, вантажний автотранспорт, мобільні пристрої та установки, за винятком ручних інструментів).

 Характеристику хімічних викидів, а саме перелік хімічних викидів до атмосфери та навколишкнього водного середовища (джерела та кількісну характеристику скидних вод).

 Характеристику впливу на населення, що включає показник ракових захворювань населення (з врахуванням статево-вікової піраміди) в найближчих населених пунктах в порівняні з середнім показником по Україні і розрахунки:

- ризику розвитку неканцерогенних ефектів;
- ризику канцерогенних ефектів;
- соціального ризику планованої діяльності.

20. В разі наявності територій чи об'єктів ПЗФ, Екомережі на території планованої діяльності, в межах санітарно-захисної зони або на території об'єктів природного середовища які безпосередньо зазнають впливу планованої діяльності оцінити вплив планованої діяльності на природні комплекси та об'єкти, зокрема флору та фауну, їх угрупування та оселища (зокрема, що охороняються) з наданням заходів мінімізації такого впливу компенсаційних заходів.



21. Опис (характеристику) поводження з відпрацьованим ядерним паливом та радіоактивними відходами його кількість, місце зберігання, термін зберігання. Також надати альтернативні варіанти поводження з відпрацьованим ядерним паливом та радіаційними відходами від планованого.

 Опис (характеристику) перевезення радіоактивних матеріалів, включно із ядерним пливом, відпрацьованим ядерним паливом та радіоактивними відходами (РАВ), що включає:

 оцінку радіаційного впливу при транспортуванні всіх видів радіоактивних матеріалів до та з Хмельницької АЕС;

 визначення ступеня екологічного ризику під час транспортування всіх видів радіоактивних матеріалів до та з Хмельницької АЕС при нормальних умовах транспортування;

 оцінку наслідків для населення і навколишнього середовища під час транспортування всіх видів радіоактивних матеріалів до та з Хмельницької АЕС при нормальних умовах транспортування;

 оцінку можливих ризиків у випадку аварій під час транспортування всіх видів радіоактивних матеріалів до та з Хмельницької АЕС із розподілом за видами транспорту, який використовується (залізничний, автомобільний тощо);

 оцінку еколого-економічних збитків у випадку аварії при перевезенні і витрат на його компенсацію за всіма сценаріями;

 оцінку еколого-економічних витрат на подолання наслідків аварій при перевезенні за всіма сценаріям.

23. Характеристику впливу на довкілля під час аварійних ситуацій:

 розглянути сценарії аварій на одному з енергоблоків № 5-6 ХАЕС та одразу на двох, в тому числі спричинених впливом бойових дій;

 розглянути сценарії результатів влучання різних видів зброї у вузлові точки забезпечення життєдіяльності, руйнування машинного залу чи системи охолодження енергоблоків ХАЕС № 5 та 6;

 розглянути сценарій аварії із падінням великого пасажирського літака, аналогічного Boeing 777/Boeing 737, із повними баками пального на вузлові точки енергоблоків ХАЕС № 5 та 6;

максимальній проектній та запроекній аварійних ситуаціях (з їх описом);

 метеорологічних ситуацій які враховувалися при таких оцінках, масштабів впливу на компоненти довкілля;

зона впливу при аварійних ситуаціях;



перелік забруднюючих речовин та потенційна їх маса викиду при аварійних ситуаціях;

 кількість населення що потенційно перебуває в зоні ураження під час аварійних ситуацій;

потенційні економічні втрати від аварійних ситуацій;

передбачені заходи попередження, мінімізації та ліквідації таких наслідків.

24. Характеристику соціально-економічного впливу, та прогнозування різницю ціни виробленої електроенергії з одиниці пального з ринковою ціною одиниці пального та ціною зберігання та подальшої утилізації одиниці відпрацьованого ядерного пального станом на зараз та кожні п'ять років протягом всього терміну експлуатації енергоблоків.

25. Надати інформацію, що стосується виведення з експлуатації, ліквідації (демонтаж) об'єкта, припиненням планованої діяльності, акцентуючи увагу на роботах з демонтажу, видаленні відходів і небезпечних хімічних речовин, рекультивації земель.

26. До звіту з ОВД додати:

попередній звіт з аналізу безпеки (ПЗАБ);

- програму «управління старінням»;

 звіт з обстеження стану існуючих будівель і конструкцій дамби водосховища-охолоджувача.

З повагою

Заступник директора ГО «Центр екологічних ініціатив «Екодія»,

Пасюк О. Г.

Hora

Виконавчий директор ГО «Екоклуб», Мартинюк А. М.

Голова правління ГО «Хмельницький енергетичний кластер», Кушнір С.С.



## APPENDIX B Conditions for the scope of studies and level of information detail



Паперова колія електронного документа

#### МІНІСТЕРСТВО ЗАХИСТУ ДОВКІЛЛЯ ТА ПРИРОДНИХ РЕСУРСІВ УКРАЇНИ (МІНДОВКІЛЛЯ)

Департамент екологічної оцінки вул. Митрополита Василя Липківського, 35, м. Київ, 03035, 206-31-40, E-mail: info@mepr.gov.ua

Ha Nr _____

АКЦІОНЕРНЕ ТОВАРИСТВО «НАЦІОНАЛЬНА АТОМНА ЕНЕРГОГЕНЕРУЮЧА КОМПАНІЯ «ЕНЕРГОАТОМ» 01032, місто Київ, вул. Назарівська, будинок 3

Про умови визначення обсягу досліджень та рівня дегалізації інформації

Департамент екологічної оцінки Міністерства захисту довкілля та природних ресурсів України відповідно до вимог частин восьмої та дев'ятої статті 5 Закону України «Про оцінку впливу на довкілля» надає умови щодо обсягу досліджень та рівня деталізації інформації, що підлягає включенню до Звіту з оцінки впливу на довкілля АТ «НАЕК «ЕНЕРГОАТОМ» щодо планованої діяльності зі спорудження енергоблоків № 5 і № 6 на майданчику Хмельницької АЕС із застосуванням технічних характеристик реакторної установки типу АР1000 компанії Westinghouse Electric Company з метою подальшої експлуатації та вироблення електроенергії (реєстраційний номер справи в Єдиному реєстрі з оцінки впливу на довкілля – 5333).

Водночас повідомляємо, що у разі надходження умов щодо обсягу досліджень та рівня деталізації інформації, що підлягає включенню до Звіту з оцінки впливу на довкілля від зачеплених сторін, готових взяти участь у транскордонних консультаціях, відповідно до вимог Конвенції про оцінку впливу на навколишнє середовище у транскордонному контексті (Конвенція Еспо) їх буде надано додатково.

Додаток: на 5 арк. в 1 прим.

Директор Департаменту

Інна Теличко 206 31 40



Mingomaliaat Na21/21-03/1685-24 aig 17.04.2024 KEIE Illumaye M. O. 17.04.2024 15:22 S8E2D9E7P900307B04000000E8PC3400E6DBA600 Coprospisor aillemeil v 03.08.2022 00:90 ao 02.08.2024 23:59

Марина ШИМКУС

#### Умови визначення обсягу досліджень, рівень деталізації інформації, що підлягає включенню до Звіту з оцінки впливу на довкілля

«Спорудження енергоблоків № 5 і № 6 на майданчику Хмельницької АЕС із застосуванням технічних характеристик реакторної установки типу АР1000 компанії Westinghouse Electric Company з метою подальшої експлуатації та вироблення електроенергії»

(назва планованої діяльності)

#### реєстраційний номер справи в Єдиному реєстрі з оцінки впливу на довкілля – 5333

На виконання статті 5 Закону України «Про оцінку впливу на довкілля» Міндовкілля розглянуло повідомлення про плановану діяльність, яка підлягає оцінці впливу на довкілля АТ «НАЕК «ЕНЕРГОАТОМ» щодо планованої діяльності зі спорудження енергоблоків № 5 і № 6 на майданчику Хмельницької АЕС із застосуванням технічних характеристик реакторної установки типу АР1000 компанії Westinghouse Electric Company з метою подальшої експлуатації та вироблення електроенергії та надає умови визначення обсягу досліджень і рівня деталізації інформації, що підлягає включенню до Звіту з оцінки впливу на довкілля.

Цей документ видається відповідно до законодавства України, зокрема, Закону України «Про оцінку впливу на довкілля» та Закону України «Про охорону навколишнього природного середовища».

Вимоги до структури та змісту Звіту з оцінки впливу на довкілля, визначені статтею 6 Закону України «Про оцінку впливу на довкілля» та додатком 3 постанови Кабінету Міністрів України від 13.12.2017 № 1026, є обов'язковими.

Звіт з оцінки впливу на довкілля має містити інформацію від заінтересованих органів щодо можливості реалізації планованої діяльності з огляду на вплив на здоров'я людей, водні та земельні ресурси, біорізноманіття, інші фактори довкілля (тобто до Звіту з оцінки впливу на довкілля підлягає включенню інформація про дозвільні документи щодо планованої діяльності за наявності).

#### Деталізувати у Звіті з оцінки впливу на довкілля (далі – Звіт з ОВД) наступне:

 Включити до Звіту з ОВД інформацію про технічні характеристики планованої діяльності, зокрема:

- тип обладнання та його технічні характеристики, що використовуватимуться при реалізації планованої діяльності;

 технічну документацію, інформацію та відомості про підтвердження оцінки відповідності передбаченого обладнання нормативній документації на виготовлення (паспорти тощо);

- дані щодо небезпечних чинників (речовин, матеріалів), що передбачаються до використання на об'єкті планованої діяльності;

- відомості стосовно наявної матеріально-технічної бази та її технічні характеристики, необхідної для провадження планованої діяльності.

 Навести у Звіті з ОВД розрахунки оцінки сейсмічного стану території (місця розташування) планованої діяльності. За необхідності визначити обмеження в умовах сейсмічної зони.

 Відобразити у Звіті з ОВД детальну інформацію про оцінку прогнозованого впливу на геологічне середовище та тектоніку, а також матеріали щодо проведених досліджень карстових явищ на території планованої діяльності.

Описати інженерно-геологічне районування території за небезпекою виникнення зсувних і обвальних процесів, а також за особливостями їх розвитку. Навести характеристику стійкості схилів і очікуваних її змін із зазначенням типу можливих зсувних і обвальних процесів, їх місцезнаходження, розмірів, а також величин і швидкості переміщення ґрунтових мас (із урахуванням непрямих наслідків, викликаних зсувними і обвальними процесами (деформації існуючих будівель і споруд, затоплення долин при утворенні обвально-зсувних загат, виникнення високої хвилі при швидкому зміщенні земляних мас у акваторію тощо).

4. Деталізувати у Звіті з ОВД характеристику геологічної будови, складу та властивостей ґрунтів в межах території можливого впливу при будівництві та експлуатації об'єкта планованої діяльності, а також інформацію щодо складу властивостей ґрунтів в межах стискуваної товщі в основі споруди.

Надати відомості щодо категорій та якості ґрунтів, аналізу впливів планованої діяльності на ґрунти з урахуванням виникнення небезпечних інженерно-геологічних процесів і явищ та інших чинників, які негативно впливають на стан ґрунтів.

Відобразити у Звіті з ОВД заходи щодо запобігання або зменшення розвитку небезпечних геологічних процесів і явищ, зокрема, ерозії ґрунтів.

5. Деталізувати у Звіті з ОВД опис технологічного процесу планованої діяльності із зазначенням усіх чинників впливу на водне середовище і технічних рішень, спрямованих на усунення чи зменшення шкідливих впливів (викидів, скидів, витоків у водні об'єкти), у тому числі, заходи щодо запобігання або зменшення надходжень у водне середовище забруднюючих речовин, порушення гідродинамічного режиму, виснаження поверхневих і підземних водних ресурсів, погіршення стану вод, деградації угруповань водних організмів та ймовірні зміни водного балансу території.

У Звіті з ОВД деталізувати технологію очистки всіх видів стічних вод, надати відомості щодо результатів лабораторного контролю за станом джерел водопостачання та водойм, що знаходяться у зоні впливу, із наданням заходів щодо перспективи контролю їх стану.

Описати особливості гідрологічного режиму території об'єкту планованої діяльності із наведенням характеристики водного балансу (майданчика, водозбору та водойми) та прогнозуванням і моделюванням екстремальних паводків, розвитку руслових процесів, розмивів берегів тощо.

Включити до Звіту з ОВД детальний опис систем охолодження ядерних реакторів.

Дослідити ймовірне теплове забруднення водних об'єктів, що знаходяться в зоні спостереження (30 км).

Надати дані та аналіз щодо впливу планованої діяльності на гідрологічний режим р. Горинь та рівень ґрунтових вод.

6. У Звіті з ОВД при оцінці впливу на атмосферне повітря здійснити розрахунок прогнозованого забруднення атмосферного повітря методом моделювання розсіювання викидів забруднюючих речових в атмосферу, у т. ч. з урахуванням попередньо проведеного аналізу щодо існуючого антропогенного фону та перспективи його зміни, а також зазначити заходи щодо запобігання або мінімізації такого забруднення.

Надати відомості щодо офіційних довідок-характеристик про кліматичні умови району розміщення об'єкту планованої діяльності (середньорічні швидкості вітру по румбах восьми румбової рози вітрів тощо) і про фоновий вміст забруднюючих речовин в атмосферному повітрі району розміщення об'єкту на поточний рік.

7. Надати інформацію щодо сучасного стану очисних споруд на підприємстві, а також оцінити їх експлуатацію з урахуванням перспективного додаткового навантаження від реалізації планованої діяльності. Також включити до Звіту з ОВД дані з обґрунтовуючими матеріалами щодо місць накопичення вапнякового шламу, обсяги скидання вапнякового шламу, наявні шламонакопичувачі із зазначенням їх параметральних характеристик, передбачити шляхи поводження з вапняковими шламами на довгострокову перспективу (у разі наявності такого шламу та здійснення операцій з ним).

Деталізувати відповідний розділ Звіту з ОВД інформацією про управління відходами (якісні та кількісні характеристики відходів, що утворюються під час реалізації планованої діяльності, зокрема, радіоактивних відходів на усіх етапах життєвого циклу ядерної установки, включаючи вибір майданчику, проектування, будівництво та зняття з експлуатації).

Класифікувати відходи з урахуванням вимог Закону України «Про управління відходами». 8. У Звіті з ОВД зазначити відношення території планованої діяльності до територій та об'єктів природно-заповідного фонду (у тому числі до лісового заказника місцевого значення «Праліс»), територій, зарезервованих до наступного заповідання, територій та об'єктів екомережі, об'єктів культурної спадщини, санітарно-захисних та охоронних зон, водозаборів, водних об'єктів із зазначенням їх статусу (у тому числі ширини прибережної захисної смуги, водоохоронної зони тощо) з відповідним відображенням даної інформації на топографічній основі (з визначенням масштабу).

Включити до Звіту з ОВД розрахунки та аналіз ймовірного впливу на зазначені території та об'єкти із зазначенням компенсаційних та природоохоронних заходів.

9. Особливу увагу приділити оцінці ймовірного впливу на флору і фауну (середовища існування, шляхи міграції, умови розмноження, наслідки впливу) при реалізації планованої діяльності, у т. ч. з урахуванням передбачуваних транспортних зв'язків для реалізації планованої діяльності із зазначенням компенсаційних та природоохоронних заходів.

10. Доповнити Звіт з ОВД інформацією щодо обов'язкової політики підприємства, що спрямована на постійний контроль і аналіз стану безпеки АЕС, в тому числі планованих енергоблоків № 5 і № 6 на існуючому майданчику Хмельницької АЕС.

Відобразити у Звіті з ОВД передбачені принципи забезпечення 11. безпеки АЕС (забезпечення культури безпеки; відповідальність експлуатуючої реалізація стратегії безпеки; державне регулювання організації; глибокоешелонованого захисту; застосування апробованої інженерно-технічної практики; управління якістю; самооцінка безпеки АЕС; аналіз безпеки; відомчий нагляд; незалежні перевірки; урахування людського фактору; забезпечення радіаційної безпеки; урахування досвіду експлуатації; науковотехнічна підтримка тощо) і конкретизувати вищезазначені принципи та вимоги, що випливають з них, у відповідності до норм, правил і стандартів з ядерної та радіаційної безпеки.

12. Долучити до Звіту з ОВД План-графік КЗПБ (Комплексна (зведена) програма підвищення безпеки енергоблоків атомних електростанцій) базований на графіках планово-профілактичних ремонтів та операцій із свіжим та відпрацьованим ядерним паливом (у випадку конфіденційності застосовувати частину восьму статті 4 Закону України «Про оцінку впливу на довкілля»).

Навести дані у Звіті з ОВД з прогнозами змін стану довкілля за нормальних умов та при проєктних аваріях на АЕС з урахуванням реалізації КЗПБ.

13. Здійснити моделювання аварійних ситуацій із визначеними зонами можливого забруднення (у т. ч. на зачеплені країни), передбачити перелік заходів щодо локалізації та ліквідації аварійних ситуацій.

14. Включити до Звіту з ОВД аналіз діяльності з урахуванням соціально-економічного навантаження регіону, наявних допоміжних об'єктів АЕС та нових запланованих допоміжних об'єктів.

15. Дослідити та включити до Звіту з ОВД інформацію про соціальноекономічні ризики від планованої діяльності:

- аналіз можливих соціальних конфліктів, пов'язаних із реалізацією планованої діяльності;

- відобразити динаміку рівня чисельності населення навколо об'єкту;

- в частині розрахунку економіки природокористування надати обгрунтування виправданості подальшої експлуатації об'єкту.

16. Деталізувати оцінку ймовірних впливів на здоров'я людини, зокрема, із зазначенням рівнів електромагнітного та радіаційного забруднення.

17. Надати список посилань із зазначенням джерел, що використовуються для описів та оцінок, при розробці звіту з оцінки впливу на довкілля (у т. ч. наявний перелік довідок, листів, отриманих від компетентних організацій та служб тощо).

KININ TA

ДЛЯ ЮВІДОК

MH

Директор Департаменту

Марина ШИМКУС

#### **APPENDIX C**

### Calculations of air pollutant dispersion during construction and operation

# C.1 Calculation of pollutant dispersion from construction equipment during the KhNPP units 5 and 6 construction

Copyright(C) ТОВ «Coфr фонд» м. Кайв Тел. (044)599 35 57 E-Mail info@stimd.kiev.ua

AT «HAEK «Enepsoamosa», Jimenvin Ne133565957

## EOЛ 2000[h] (Windows версія)

Автоматизована система розрахунку розсіювання викидів шкідливих речовин

Загальний звіт про результати розрахунку розсіювання

"Період будівництва"

Розрахунковий модуль системи реахілує методику ОПД-86 Програма рекомендована для використання Міністерством охорони навколицикого природного середовища України(2464/19/4-10 от 15.03.2006)

				Завдання	на розраху	BOK.			
Будуі	Швядкіс Швядкіс вати розр	Найма Кодин пр Кодин Шимдв видкість в видкість в бількі Фіхос Кількі Ча вра ахункову Висога	нування мії ом. майдан прин суман груп суман ість вітру (1 істик U се сбору напр. ві ть найб. вк. сть майб. вк. сть майб. вк. сть майб. вк. судзови ві розрахунку	ета ників м/с) U сер. зн.) р. надфакел вітру ру задн. яц. яц. мц. (м)	ьної) рисметва	Herimu 1 301 32 31 0.5 2 0.5 1 10 - 3 10 Tax Hi/Hi 2	# 8 330 337 ; 7 1.5	2754	
			п	араметри р	опрахунков	вих майдая	สายหลัด		
Ne n/n	Коорд. Х	Коорд. Ү	Довжния	Ширина	Кут. пов. розр. майд. відн. вісі ОХ осн. сист. коорд		Крок по сітці вісь ОХ	Крок по сітці вісь ОҮ	Особа. вимоги
1	0.0	0.0	7000.0	7000.0	0.	0	250.0	250.0	0

	-2-		
12.06.2024	"AT «HAEK «Eneproarom»"	ЕОЛ 2000 [h] v4.0,	Jlinemis №133565957

Код міста	Найменувания міста	Сер. температура самого теплого місяця (град С)	Сер. температура самого холодного місяця (град С)	Гранячна швидкість вітру (м/с)	Регіональний коефіцієнт стратифікації	Кут між північним напрям. та віссю ОХ осн. сист. коорд. (град)	Площа міста (кв. км)
100	Heriunn	23.0	-5.7	7.0	200	90	70

Широта (град.,хв.,сек.)	Широта (пиш. чи пдиг.)	Довгота (град.,хв.,сек.)	Довгота (зд. чи сд.)	Ймовірпість повтору вітру(Пя)	Ймовірність повтору вітру(ПиСх)	Ймовірність повтору вітру(Сх)	Имовірпість повтору вітру(ПдСх)	Ймовірність повтору вітру(Пд)
50rpag.19/48*	104	26rpag.38*24*	сд	10.8	5	9.4	13.9	16.6

Имовірність	Ймовірність	Ймовірність
повтору	повтору	повтору
вітру(ПдЗх)	вітру(3х)	вітру(ПпЗх)
11.7	20.8	11.8

Код пр. майд.	Найменування промислового майданчика	Код речовин (групя сумації)	Найменування речовнини (Коди речовни, що входять у групу сумації ).	Потужність викиду (г/с)	Потужність викиду (г/рік)
а. Г	Хмельаннцька ХАЕС	Гр. сум. № 31 Код р-ни 301 Код р-ни 328 Код р-ни 330 Код р-ни 337 Код р-ни 2754	301 330 Азоту діоксид Сажа Ангідрид сірчистий Вуглеводні граничні є 12-є 19(розчинник РПК-26611 і ін.)	5.2000 1.9520 0.4160 0.3200 2.9760 0.4800	0.0000 0.0000 0.0000 0.0000 0.0000 0.0000

Код речовани	Найменування резовини	ГДК (мг/м.куб)
301	Азоту діокенд	0.20000000

# - 3 -12.06.2024 "АТ «НАЕК «Епергоатом»" - ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Фонові концентрації, які вміщують впески діючих джерел (Частки ГДК) (частки ГДК) (Вихідні рівні забруднення) для речовини : Азоту діоксид. Варіант завдання фону : а.

Коорд. Х поста спостережения	Коорд. Ү воста спостережения	U<2 м/с (штиль)	IIInnenciers. Birpy 2-U-U* Elu	Швилкість вітру 2 <u<u* ПнС</u<u* 	Швидкість вітру 2-U-U* С	Швидисть вirpy 2 <u<u* ПлС</u<u* 	Швядкість вітру 2-U-U* Пд	Швидисість вітру 2 <u<u* Пд3</u<u* 	IIInsentions sirpy 2-U-U* 3	Швидкість вітру 2 <u<u* Пв3</u<u* 
0.00	0.00	0.04	0.04	0.04	0.04	0.04	0,04	0.04	0.04	0.04

Фонові концентрації без урахування внесків діючих джерел (Частки ГДК) (частки ГДК) (Власне фон - верхис число, вклад - нижис) для речовним : Азоту діоксид, Варіант завдання фону : а.

Коорд. Х поста спостережения	Коорд. У поста спостережения	U<2 м/с (штиль)	Швидисість niтру 2⊲U <u* Пи</u* 	Швидкість nitpv 2 <u<u* ПпС</u<u* 	Швидкість пітру 2 <u<u* С</u<u* 	Швидкість niтpv 2 <u-u* ПлС</u-u* 	Швидкість пітру 2 <u<u* Пл</u<u* 	Швилкість пітру 2 <u<u* Пл3</u<u* 	Швидкість вітру 2 <u<u* 3</u<u* 	Швндю́сть вітру 2≪U≪U* ПпЗ
0:00	0:00	0.0400	0.0400	0.0400	0.0400	0.0400	0.0400	0.0400	0.0400	0.0400

	- 4 -	2	
12.06.2024	"АТ «НАЕК «Енергоатом»"	ЕОЛ 2000 [h] v4.0,	Ліцензія №133565957

Перелік джерел, у викидах яких с Азоту діоксид

Кол джерела Технологічні параметри	11001	11002	11003	11004	11005	11006	11007	11008
Викид г/с	0.0610	0.0610	0.0610	0.0610	0.0610	0.0610	0.0610	0.0610
Клас небезпечн.	1	1	1	1	1	1	1	1
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мс/м. куб	13.4208	13.4208	13.4208	13.4208	13.4208	13.4208	13.4208	13.4208
ХМ (м)	28.62	28.62	28.62	28.62	28.62	28.62	28.62	28.62
UM[h=2.00m] (m/c)	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50
Х Ү Коорд. точеч. початок лін-го, центр симетр. пл-го (м)	-600.00 100.00	-610.00 105.00	-620.00 103.00	-630.00 110.00	-640.00 90.00	-650.00 102.00	-660.00 108.00	-670.00 85.00
Х. Ү Коорд. кінця лін-то, дов. і ширина пл-го(м)	5.00 5.00	5,00 5,00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00
Коеф-т рель'ефу	1.0000	1.0000	1.0000	1,0000	1.0000	1.0000	1.0000	1.0000
Витрата ПГПС(м. куб/с)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
Шв-ть вихіду ПГПС: м/с	0	0	0	0	0	0	0	0
Діаметр (м)	*	*		8	÷	×	- 8	-
Висота (м)	5.0000	5.0000	5.0000	5.0000	5.0000	5.0000	5.0000	5.0000
Температура (С)	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000
Коеф-т впоряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Викид т/р	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0,0000

Код джерела Технологічні параметри	11009	11010	11011	11012	11013	11014	11015	11016
Виюнд г/с	0.0610	0.0610	0.0610	0.0610	0.0610	0.0610	0.0610	0.0610
Клас небезпечн.	1	1	1	1	1	1	1	1
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/M[h=2.00м] мс/м. куб	13.4208	13.4208	13.4208	13.4208	13.4208	13.4208	13.4208	13.4208
ХМ (м)	28.62	28.62	28.62	28.62	28.62	28.62	28.62	28.62
UM[h=2.00m] (m/c)	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50
Х Ү Коорд. гочеч. почеток лін-го, центр симетр. пл-го (м)	-680.00 90.00	-690.00 100.00	-700.00 98.00	-710.00 95.00	-720.00 110.00	-730.00 90.00	-740.00 85.00	-750.00 115.00
Х У Коорд. кінця лін-го, дов. і ширяца пл-го(м)	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00
Коеф-т рель'сфу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ШПС(м. куб'с)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
Шв-ть вихіду ПГПС: м/с	0	0	0	0	0	0	0	0
Діаметр (м)	÷	•	÷	÷		6		
Висота (м)	5:0000	5:0000	5,0000	5:0000	5.0000	5.0000	5.0000	5.0000
Температура (С)	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000
Коеф-т вноряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1,0000	1.0000
Викид т/р	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000

-5-12.06.2024 "AT «НАЕК «Еперговтом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Код джерела Технологічні параметри	11017	11018	11019	11020	11021	11022	11023	11024
Вионд п/с	0.0610	0.0610	0.0610	0.0610	0.0610	0.0610	0.0610	0.0610
Клас небезпечн.	1	1	1	1	1	1	1	1
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/M[h=2.00м] мс/м. куб	13.4208	13.4208	13.4208	13.4208	13.4208	13.4208	13.4208	13,4208
XM (st)	28.62	28.62	28.62	28.62	28.62	28.62	28.62	28.62
UM[h=2.00m] (m/c)	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50
Х Ү Коорд. гочеч, почяток лін-го, центр саметр, пл-го (м)	-760.00 200.00	-770.00 220.00	-780.00 230.00	-790.00 250.00	-800.00 220.00	-810.00 250.00	-820.00 220.00	-830.00 250.00
Х Ү Коорд. кінця лін-го, дов. і ширица пл-го(м)	5.00 5.00							
Коеф-т рель' сфу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ПГПС(м. куб'с)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
Шв-ть вихіду ШПС: м/с	0	0	0	0	0	0	0	0
Діаметр (м)	÷	-	÷	-	•	-		
Висота (м)	5.0000	5:0000	5.0000	5.0000	5.0000	5.0000	5.0000	\$,0000
Температура (С)	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000
Коеф-т виоряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Викид т/р	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000

- 6 -12.06.2024 "AT «НАЕК «Еперговтом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Код джерела Технологічні параметри	11025	11026	11027	11028	11029	11030	11031	11032
Виюнд г/с	0.0610	0.0610	0.0610	0.0610	0.0610	0.0610	0.0610	0.0610
Клас небезпечн.	1	1	1	1	1	1	1	1
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/M[h=2.00м] мс/м. куб	13.4208	13.4208	13.4208	13.4208	13.4208	13.4208	13.4208	13,4208
ХМ (м)	28.62	28.62	28,62	28.62	28.62	28.62	28.62	28.62
UM[h=2.00st] (st/c)	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50
Х	-840.00 200.00	-850.00 150.00	-860.00 205.00	-870.00 250.00	-880.00 260.00	-890.00 220.00	-900.00 260.00	-910.00 270.00
Х У Коорд. кінця лін-го, дов. і пиряца пл-го(м)	5.00 5.00							
Коеф-т рель'ефу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ПГПС(м. куб'с)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
Шв-ть вихіду ШПІС: м/с	0	0	0	0	0	0	0	0
Діаметр (м)	÷	-	÷		•	-		8
Висота (м)	5:0000	5:0000	5.0000	5.0000	5.0000	5.0000	5.0000	5.0000
Температура (C)	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000
Коеф-т вноряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Викид т/р	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000

	- / -	·8	
12.06.2024	"АТ «НАЕК «Епергоатом»"	EOJI 2000 [h] v4.0,	Jlinemia №133565957

Розрахункові концентрації речовнин: Алоту діоксид в розрахункових точках та номера джерел, що надають найбільший внесок на высоті 2.00 м

№ розр. точки	Концентр. у точці частки ГДК	Коорд. розр. точки Х	Коорд. розр. точки Ү	Напрамок вітру	Шандиість вітру	Розмір внеску Q0	уе джерела N0	Розмір вноску Q1	№ джерела N1	Розмір вноску Q2	Nê Josepela N2
п	0.1846	0,0	2700.0	73	0.75	0.0047	11020	0.0047	11019	0.0047	11022
22	0.1730	\$50.0	2700.0	63	0.75	0.0043	11020	0.0043	11019	0.0043	11018

	- 8 -	8	
12.06.2024	"AT «HAEK «Eneproarom»"	ЕОЛ 2000 [h] v4.0,	Ліцензія №133565957

Точки найбільших концентрацій речовини Азоту діоксид		
На розрахун. площадці № 1 та помера джерел, що надають найбільший впесок на высот	2.00 x	à

Концентрації у точці частки ГДК	Коорд. розр. точван Х	Коорд. розр. точки Ү	Напрям. вітру	Швадкість вітру	Розмір внеску Q0	№ джерела №0	Розмір внеску Q1	№ джереля N1	Розмір внеску Q2	№ джерела N2
4.7106	-1000.0	250.0	171	0.50	0.4500	11030	0.4020	11031	0.3484	11028
4.3664	→500.0	0.0	327	0.75	0.3041	11005	0.2882	11003	0.2796	11001
4.2077	-750.0	0.0	262	0.50	0.6653	11015	0.5998	11014	0.4412	11013
3.2523	-750.0	250.0	97	0.50	0.6213	11017	0.3494	11016	0.3263	11013
2.3380	-500.0	250.0	29	0.50	0.1402	11007	0.1329	11004	0.1275	11006
2.0490	-1000.0	500.0	126	0.75	0.1108	11029	0.1077	11024	0.1057	11028
1.9415	-750.0	500.0	79	0.75	0.1505	11020	0.1453	11022	0.1289	11024
1.8352	-1000.0	0,0	224	0,50	0.1766	11026	0.1157	11025	0.1095	11027
1.5287	-1250.0	500.0	146	7.00	0.0797	11032	0.0769	11031	0.0723	11029
1.4924	-250.0	0.0	342	7.00	0.0839	11001	0.0832	11002	0.0786	11004



Азоту діокенд Карта-ехема Н=2.00 м

#### - 10 -12.06.2024 "АТ «НАЕК «Еперговтом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Сод речовини 328	Найменувания речовним	ГДК (мг/м.куб)
328	Сажа	0.15000000

Фонові концентрації, які вміщують внески дізочих джерел (Частки ГДК) (частки ГДК) (Вихідні рівні забруднення) для речовник : Сажа. Варіант завдання фону : а.

Коорд. Х поста спостережения	Коорд. У поста спостереженно	U<2 м/с (штиль)	IIInnqueiers Birpy 2 <u<u* Hu</u<u* 	IIInnuncieus sirpy 2 <u<u* HuC</u<u* 	Швидкість вітру 2 <u=u* С</u=u* 	Шинджість вітру 2 <u<u* ПдС</u<u* 	Швяджість вітру 2 <u<u* Пд</u<u* 	III видкість вітру 2-U-U* ПаЗ	Шимдкість вітру 2 <u<u* 3</u<u* 	Швидкість вітру 2 <u<u* ПяЗ</u<u* 
0.00	0.00	0.40	0.40	0.40	0.40	0.40	0.40	0.40	0.40	0.40

Фонові концентрації без урахування внесків діючих джерел (Частки ГДК) (частки ГДК) (Власне фон - верхис число, вклад - нижис) для речовини : Сяжа. Варіант завдання фону : а.

Коорд. Х поста спостережения	Коорд. Ү поста спостережения	U<2 м/с (апасаь)	Шпяджість вітру 2 <u<u* Пп</u<u* 	IIIвидијењ вiтру 2 <u<u* ПиС</u<u* 	Шиюлкість вітру 2 <u<u* С</u<u* 	Швилкість нітру 2 <u<u* ПлС</u<u* 	Шюядкість вітру 2≤U <u* Пд</u* 	Шпилкість вітру 2 <u<u* ПдЗ</u<u* 	Швядкість вітру 2 <u<u* 3</u<u* 	Швида́сть вітру 2⊲U⊲U* Пн3
0.00	0.00	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000

	- 11	-	
12.06.2024	"АТ «НАЕК «Епергоатом»"	ЕОЛ 2000 [h] v4.0,	Ліцензія №133565957
	Перелік ажеред у	PRICEDURAL REPORT	

Сажа

Кол джерели Технологічні параметри	11001	11002	11003	11004	11005	11006	11007	11008
Викид г/с	0.0130	0.0130	0.0130	0.0130	0.0130	0.0130	0.0130	0.0130
Клас небезлечи.	1	1	1	1	1	1	1	1
СМfh=2.00м1 (частки ГДК) СМfh=2.00м1 мг/м. куб СМ:М[h=2.00м] мс/м. куб	3.8136	3.8136	3.8136	3.8136	3.8136	3,8136	3.8136	3.8136
ХМ (м)	28.62	28.62	28.62	28.62	28.62	28.62	28.62	28.62
UM[h=2.00m] (m/c)	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50
Х Ү Коорд. точеч. початок ліп-го, центр симетр. пл-го (м)	-600.00 100.00	-610.00 105.00	-620.00 103.00	-630.00 110.00	-640.00 90.00	-650.00 102.00	-660.00 108.00	-670.00 85.00
Х Ү Коорд. кінця лін-то, дов. і ширина пл-то(м)	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5,00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00
Коеф-т рель'ефу	1.0000	1.0000	1.0000	1,0000	1.0000	1.0000	1.0000	1.0000
Витрата ПГПС(м. куб'с)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
Шв-ть вихіду ПППС: м/с	0	0	0	0	0	0	0	0
Діаметр (м)	*	*	×.	8	÷	2	8	-
Висота (м)	5.0000	5.0000	5.0000	5.0000	5.0000	5.0000	5.0000	5.0000
Температура (С)	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000
Коеф-т впоряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Викид т/р	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000

Код джерела Технологічні параметри	11009	11010	11011	11012	11013	11014	11015	11016
Виюнд г/с	0.0130	0.0130	0.0130	0.0130	0.0130	0.0130	0.0130	0.0130
Клас небезпечн.	1	1	1	1	1	1	1	1
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/M[h=2.00м] мс/м. куб	3.8136	3.8136	3,8136	3.8136	3.8136	3.8136	3.8136	3,8136
ХМ (м)	28.62	28.62	28,62	28.62	28.62	28.62	28.62	28.62
UM[h=2.00m] (m/c)	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50
Х Ү Коорд. гочеч. почеток лін-го, центр симетр. пл-го (м)	-680.00 90.00	-690.00 100.00	-700.00 98.00	-710.00 95.00	-720.00 110.00	-730.00 90.00	-740.00 85.00	-750.00 115.00
Х У Коорд. кінця лін-го, дов. і ширяща пл-го(м)	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00
Коеф-т рель'сфу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ПГПС(м. куб'с)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
Шв-ть вихіду ШПІС: м/с	0	0	0	0	0	0	0	0
Діаметр (м)	÷	-	÷		-	-		
Висота (м)	5.0000	5:0000	5.0000	5.0000	5.0000	5.0000	5.0000	\$,0000
Температура (С)	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000
Коеф-т вноряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Вакид т/р	0.0000	0.0000	0.0000	0.0000	0.0000	0.000	0.0000	0.0000

- 12 -12.06.2024 "АТ «НАЕК «Епергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Код джерела Технологічні параметри	11017	11018	11019	11020	11021	11022	11023	11024
Вижня г/с	0.0130	0.0130	0.0130	0.0130	0.0130	0.0130	0.0130	0.0130
Клас небезпечн.	1	1	1	1	1	1	1	1
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мт/м. куб СМ/M[h=2.00м] мс/м. куб	3.8136	3.8136	3,8136	3.8136	3.8136	3.8136	3.8136	3.8136
XM (st)	28.62	28.62	28,62	28.62	28.62	28.62	28.62	28.62
UM[h=2.00m] (m/c)	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50
Х Ү Коорд. гочеч. почяток лін-го, центр саметр. пл-го (м)	-760.00 200.00	-770.00 220.00	-780.00 230.00	-790.00 250.00	-800.00 220.00	-810.00 250.00	-820.00 220.00	-830.00 250.00
Х Ү Коорд. кінця лін-го, дов. і ширица пл-го(м)	5.00 5.00							
Коеф-т рель' сфу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Витраза ПГПС(м. куб'с)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
Шв-ть вяхіду ШПС: м/с	0	0	0	0	0	0	0	0
Діаметр (м)	÷		÷		-		-	- 20
Висота (м)	5:0000	5:0000	5.0000	5.0000	5.0000	5.0000	5.0000	5.0000
Температура (С)	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000
Коеф-т вноряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Викид т/р	0.0000	0.0000	0.0000	0.0000	0.0000	0.000	0.0000	0.0000

- 13 -12.06.2024 "АТ «НАЕК «Епергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Код джерела Технологічні параметри	11025	11026	11027	11028	11029	11030	11031	11032
Виюнд г/с	0.0130	0.0130	0.0130	0.0130	0.0130	0.0130	0.0130	0.0130
Клас небезпечн.	1	1	1	1	1	1	1	1
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/M[h=2.00м] мс/м. куб	3.8136	3.8136	3,8136	3.8136	3.8136	3.8136	3.8136	3.8136
ХМ (м)	28.62	28.62	28.62	28.62	28.62	28.62	28.62	28.62
UM[h=2.00m] (m/c)	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50
Х	-840.00 200.00	-850.00 150.00	-860.00 205.00	-870.00 250.00	-880.00 260.00	-890.00 220.00	-900.00 260.00	-910.00 270.00
Х У Коорд. кінця лін-го, дов. і ширяца пл-го(м)	5.00 5.00							
Коеф-т рель'офу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ПГПС(м. куб'с)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
Шв-ть вихіду ПГПС: м/с	0	0	0	0	0	0	0	0
Діаметр (м)	÷	-	÷	÷		-		÷.
Висота (м)	5:0000	5:0000	5.0000	5.0000	5.0000	5.0000	5.0000	\$,0000
Температура (C)	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000
Коеф-т вноряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Викид т/р	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000

- 14 -12.06.2024 "АТ «НАЕК «Епергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Розрахункова концентрації речовнин: Сажа в розрахунковах точках та помера джерел, що надають найбільший внесок на высоті 2.00 м

№ розр. точки	Концентр. у точці частки ГДК	Коорд. розр. точки Х	Коорд. розр. точки Ү	Напрамок вітру	Шандиість вітру	Розмір внеску Q0	уе джерела N0	Розмір вноску Q1	ме джерела N1	Розмір вноску Q2	Nê Josepela N2
ш	0.4411	0,0	2700.0	73	0.75	0.0013	11020	0.0013	11019	0.0013	11022
22	0.4378	\$50.0	2700.0	63	0.75	0.0012	11020	0.0012	11019	0.0012	11018

	- 15	-	
12.06.2024	"АТ «НАЕК «Епергоатом»"	ЕОЛ 2000 [h] v4.0,	Ліцензія №133565957
	111111111111111111111111111111111111111	100	

Точки найбільних концентрацій речовани Сажа На розрахун. площадці № 1 та помера джерел, що надають найбільний внесок на высоті 2.00 м

Концентрації у точці частки ГДК	Коорд, розр, точван Х	Коорд. розр. точки Ү	Напрям. вітру	Швадкість вітру	Розмір внеску Q0	№ джерела N0	Розмір внеску Q1	№ джерели N1	Розмір внеску Q2	№ джерсла N2
1.7272	-1000.0	250.0	171	0.50	0.1279	11030	0.1142	11031	0.0990	11028
1.6294	-500.0	0.0	327	0.75	0.0864	11005	0.0819	11003	0.0795	11001
1.5843	-750.0	0.0	262	0.50	0.1890	11015	0.1704	11014	0.1254	11013
1.3128	-750.0	250.0	97	0.50	0.1765	11017	0.0993	11016	0.0927	11013
1.0530	-500.0	250.0	29	0.50	0.0398	11007	0.0378	11004	0.0362	11006
0.9709	-1000.0	500.0	126	0.75	0.0315	11029	0.0306	11024	0.0300	11028
0.9403	-750.0	500.0	79	0.75	0.0428	11020	0.0413	11022	0.0366	11024
0.9101	-1000.0	0.0	224	0,50	0.0502	11026	0.0329	11025	0.0311	11027
0.8230	-1250.0	500.0	146	7.00	0.0226	11032	0.0219	11031	0.0205	11029
0.8127	-250.0	0.0	342	7.00	0.0238	11001	0.0236	11002	0.0223	11004





#### - 17 -12.06.2024 "AT «НАЕК «Епергоптом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Код речовини	Найменувания речовнии	ГДК (мг/м.куб)
330	Ангідрид сірчистий	0.50000000

Фонові концентрації, які вміщують внески діючих джерел (Частки ГДК) (частки ГДК) (Вихідні рівні забруднення) для речовини : Ангідрид сірчистий. Варіант завдання фону : а.

Коорд. Х поста спостережения	Коорд. У поста спостереженно	U<2 м/с (штиль)	IIImagaicens Birpy 2 <u<u* Hu</u<u* 	IIInunkiens sirpy 2 <u<u* HuC</u<u* 	Швидкість вітру 2 <u<u* С</u<u* 	Шинджість вітру 2 <u<u* ПдС</u<u* 	Швялкість вітру 2 <u<u* Пд</u<u* 	III saakien, sirpy 2=U=U* IIa3	Шимдкість вітру 2 <u<u* 3</u<u* 	Швидкість вітру 2 <u<u* ПяЗ</u<u* 
0.00	0.00	0.04	0.04	0.04	0.04	0.04	0.04	0.04	0.04	0.04

Фонові концентрації без урахування внесків діючих джерел (Частки ГДК) (частки ГДК) (Власне фон - верхис число, вклад - нижие) для речовини : Ангідрид сірчистий. Варіант завдання фону : а.

Коорд. Х поста спостережения	Коорд. У поста спостережения	U<2 м/с (шталь)	Шпядкість вітру 2 <u<u* Пя</u<u* 	IIIвндкість вітру 2 <u<u* ПиС</u<u* 	Шпидисість вітру 2 <u<u* С</u<u* 	Швилкість вітру 2≪U <u* ПдС</u* 	Швидкість вітру 2≤U <u* Пд</u* 	Шпилкість вітру 2 <u<u* ПдЗ</u<u* 	Швидкість вітру 2 <u<u* 3</u<u* 	Швндийсть вітру 2⊲U⊲U* Пн3
0.00	0.00	0.0400	0.0400	0.0400	0.0400	0.0400	0.0400	0.0400	0.0400	0.0400

Кол джерели Технологічні параметри	11001	11002	11003	11004	11005	11006	11007	11008
Винид г/с	0.0100	0.0100	0.0100	0.0100	0.0100	0.0100	0.0100	0.0100
Клас небезпечи.	1	1	1	1	1	1	1	1
СМfh=2.00м] (частки ГДК) СМfh=2.00м] мг/м. куб СМ:М[h=2.00м] мс/м. куб	0.8801	0.8801	0.8801	0.8801	0.8801	0.8801	0.8801	0.8801
ХМ (м)	28.62	28.62	28.62	28.62	28.62	28.62	28.62	28.62
UM[h=2.00 _M ] (m/c)	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50
Х Ү Коорд. точеч. початок лін-го, центр симетр. пл-го (м)	-600.00 100.00	-610.00 105.00	-620.00 103.00	-630.00 110.00	-640.00 90.00	-650.00 102.00	-660.00 108.00	-670.00 85.00
Х. Ү Коорд. кінця лін-то, дов. і ширипа п.1-то(м)	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5,00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00
Коеф-т рель'ефу	1.0000	1.0000	1.0000	1,0000	1.0000	1.0000	1.0000	1.0000
Витрата ПГПС(м. куб/с)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
Шв-ть вихіду ППІС: м/с	0	0	0	0	0	0	0	0
Діаметр (м)	*	*	×.	8	÷	2	8	
Висота (м)	5.0000	5.0000	5,0000	5.0000	5.0000	5.0000	5.0000	5.0000
Температура (C)	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000
Коеф-т впоряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Викид т/р	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000

# 12.05.2024 "AT «НАЕК «Епергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Перелік джерел, у викидах яких є Ангідрид сірчистий

Код джерела Технологічні параметри	11009	11010	11011	11012	11013	11014	11015	11016
Виюнд г/с	0.0100	0.0100	0.0100	0.0100	0.0100	0.0100	0.0100	0.0100
Клас небезпечн.	1	1	1	1	1	1	1	1
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/M[h=2.00м] мс/м. куб	0.8801	0.8801	0.8801	0.8801	0.8801	0.8801	0.8801	0.8801
ХМ (м)	28.62	28.62	28,62	28.62	28.62	28.62	28.62	28.62
UM[h=2.00m] (m/c)	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50
Х Ү Коорд. точеч, початок лін-го, центр саметр, цл-го (м)	-680.00 90.00	-690.00 100.00	-700.00 98.00	-710.00 95.00	-720.00 110.00	-730.00 90.00	-740.00 85.00	-750.00 115.00
Х У Коорд. кінця лін-го, дов. і ширяца пл-го(м)	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00
Коеф-т рель'сфу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ПГПС(м. куб'с)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
Шв-ть вихіду ШПІС: м/с	0	0	0	0	0	0	0	0
Діаметр (м)	÷	-	÷		•	-		
Висота (м)	5.0000	5:0000	5.0000	5.0000	5.0000	5.0000	5.0000	\$,0000
Температура (С)	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000
Коеф-т вноряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Вакид т/р	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000

- 19 -12.06.2024 "АТ «НАЕК «Епергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Код джерела Технологічні параметри	11017	11018	11019	11020	11021	11022	11023	11024
Вижня г/с	0.0100	0.0100	0.0100	0.0100	0.0100	0.0100	0.0100	0.0100
Клас небезпечн.	1	1	1	1	1	1	1	1
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мт/м. куб СМ/M[h=2.00м] мс/м. куб	0.8801	0.8801	0.8801	0.8801	0.8801	0.8801	0.8801	0.8801
XM (st)	28.62	28.62	28,62	28.62	28.62	28.62	28.62	28.62
UM[h=2.00m] (m/c)	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50
Х Ү Коорд. гочеч. почяток лін-го, центр саметр. пл-го (м)	-760.00 200.00	-770.00 220.00	-780.00 230.00	-790.00 250.00	-800.00 220.00	-810.00 250.00	-820.00 220.00	-830.00 250.00
Х Ү Коорд. кінця лін-го, дов. і ширица пл-го(м)	5.00 5.00							
Коеф-т рель' сфу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ПГПС(м. куб'с)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
Шв-ть вяхіду ШПС: м/с	0	0	0	0	0	0	0	0
Діаметр (м)	÷		÷		-		-	- 8
Висота (м)	5:0000	5:0000	5.0000	5.0000	5.0000	5.0000	5.0000	5.0000
Температура (С)	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000
Коеф-т вноряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Викид т/р	0.0000	0.0000	0.0000	0.0000	0.0000	0.000	0.0000	0.0000

-20 -12.06.2024 "АТ «НАЕК «Епергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Код джерела Технологічні параметри	11025	11026	11027	11028	11029	11030	11031	11032
Виюнд г/с	0.0100	0.0100	0.0100	0.0100	0.0100	0.0100	0.0100	0.0100
Клас небезпечн.	1	1	1	1	1	1	1	1
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/M[h=2.00м] мс/м. куб	0.8801	0.8801	0.8801	0,8801	0.8801	0.8801	0.8801	0.8801
ХМ (м)	28.62	28.62	28,62	28.62	28.62	28.62	28.62	28.62
UM[h=2.00m] (m/c)	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50
Х	-840.00 200.00	-850.00 150.00	-860.00 205.00	-870.00 250.00	-880.00 260.00	-890.00 220.00	-900.00 260.00	-910.00 270.00
Х У Коорд. кінця лін-го, дов. і ширица пл-го(м)	5.00 5.00							
Коеф-т рель'ефу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ШПІС(м. куб'с)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
Шв-ть вихіду ПГПС: м/с	0	0	0	0	0	0	0	0
Діаметр (м)	÷	-	÷		•	-		÷.
Висота (м)	5:0000	5:0000	5.0000	5.0000	5.0000	5.0000	5.0000	\$,0000
Температура (С)	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000
Коеф-т виоряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1,0000	1.0000
Викид т/р	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000

	-21-								
12.06.2024	"AT «HAEK «Eneprostom»"	EOJI 2000 [h] v4.0.	Jlinemia №133565957						

Розрахункові концентрації речовини: Ангідрид сірчнетий в розрахункових точках та номера джерел, що надають найбільший внесок на высоті 2.00 м

№ розр. точки	Концентр. у точці частки ГДК	Коорд. розр. точки Х	Коорд. розр. точки Ү	Напрамок вітру	Швюлсість вітру	Розмір внеску Q0	№ джерела N0	Розмір вноску Q1	ме джерела N1	Розмір вноску Q2	Nê Josepela N2
п	0.0495	0,0	2700.0	73	0.75	0.0003	11020	0.0003	11019	0.0003	11022
22	0.0487	\$50.0	2700.0	63	0.75	0.0003	11020	0.0003	11019	0.0003	11018
	- 22										
------------	-------------------------	--------------------	---------------------								
12.06.2024	"АТ «НАЕК «Еперговтом»"	ЕОЛ 2000 [h] v4.0,	Ліцензія №133565957								

Точки найбільших концентрацій речовнии Ангідрид сірчистий На розрахун. площадці № 1 та помера джерел, що надають найбільший внесок на высоті 2.00 м

Концентрації у точці частки ГДК	Коорд, розр, точкан Х	Коорд. розр. точки Ү	Напрям. вітру	Швидкість вітру	Розмір внеску Q0	№ джерела №0	Розмір внеску Q1	№ джереля N1	Розмір внеску Q2	№ джерела N2
0.3463	-1000.0	250.0	171	0.50	0.0295	11030	0.0264	11031	0.0228	11028
0.3237	-500.0	0.0	327	0.75	0.0199	11005	0.0189	11003	0.0183	11001
0.3133	-750.0	0.0	262	0.50	0,0436	11015	0.0393	11014	0.0289	11013
0.2506	-750.0	250.0	97	0.50	0.0407	11017	0.0229	11016	0.0214	11013
0.1907	-500.0	250.0	29	0.50	0.0092	11007	0.0087	11004	0.0084	11006
0.1717	-1000.0	500.0	126	0.75	0.0073	11029	0.0071	11024	0.0069	11028
0.1647	-750.0	500.0	79	0.75	0.0099	11020	0:0095	11022	0.0085	11024
0.1577	-1000.0	0,0	224	0,50	0.0116	11026	0.0076	11025	0.0072	11027
0.1376	-1250.0	500.0	146	7.00	0.0052	11032	0.0050	11031	0.0047	11029
0.1352	-250.0	0.0	342	7.00	0.0055	11001	0.0055	11002	0.0052	11004



Ангідрид сірчистий Карта-схема Н=2.00 м

### - 24 -12.06.2024 "АТ «НАЕК «Еперговтом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Код речовини	Найменувания речовним	ГДК (мг/м.куб)
337	Вуглещо оксид	5.00000000

Фонові концентрації, які вміщують внески діючих джерел (Частки ГДК) (частки ГДК) (Вихідні рівні забруднення) для речовини : Вуглецю оксяд, Варіант завдання фону : а.

Коорд. Х поста спостережения	Коорд. У поста спостережения	U<2 м/с (штиль)	IIInnqueiers Birpy 2 <u<u* Hu</u<u* 	IIInnuncieus sirpy 2 <u<u* HuC</u<u* 	Швидкість вітру 2 <u=u* С</u=u* 	Шинджість вітру 2 <u<u* ПдС</u<u* 	Швяджість вітру 2 <u<u* Пд</u<u* 	III видкість вітру 2-U-U* ПаЗ	Шимдкість вітру 2 <u<u* 3</u<u* 	Швидкість вітру 2 <u<u* Пв3</u<u* 
0.00	0.00	0.08	0.08	0.08	0.08	0.08	0.08	0.08	0.08	0.08

Фонові концентрації без урахування внесків діючих джерел (Частки ГДК) (частки ГДК) (Власне фон - верхис число, вклад - нижис) для речовини : Вуглецко оксад. Варіант завдання фону : а.

Коорд. Х поста спостережения	Коорд. Ү поста спостереження	U<2 м/с (вписть)	Шпилкість вітру 2 <u<u* Пи</u<u* 	IIIвндкість вітру 2 <u<u* ПиС</u<u* 	Шиндисість вітру 2 <u<u* С</u<u* 	Швилкість вітру 2≪U <u* ПдС</u* 	Швидкість вітру 2≤U <u* Пд</u* 	Шпилкість вітру 2 <u<u* ПдЗ</u<u* 	Швядкість вітру 2 <u<u* 3</u<u* 	Швндийсть вітру 2⊲U⊲U* Пн3
0.00	0.00	0.0800	0.0800	0.0800	0.0800	0.0800	0.0800	0.0800	0.0800	0.0800

	-25	2	
12.06.2024	"AT «НАЕК «Еперговтом»"	ЕОЛ 2000 [h] v4.0,	Ліцензія №133565957
	CONTRACTOR AND A CONTRACTOR		

Перелік джерел, у викидах яких є Вуглецю оксид

Код джерели Технологічні параметри	11001	11002	11003	11004	11005	11006	11007	11008
Вижд г/с	0.0930	0.0930	0.0930	0.0930	0.0930	0.0930	0.0930	0.0930
Клас небезлечи.	1	1	1	1	1	1	1	1
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ:М[h=2.00м] мс/м. куб	0.8184	0.8184	0.8184	0.8184	0.8184	0.8184	0.8184	0.8184
ХМ (м)	28.62	28.62	28.62	28.62	28.62	28.62	28.62	28.62
UM[h=2.00m] (m/c)	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50
Х Ү Коорд. точеч. початок лин-го, центр симетр. пл-го (м)	-600.00 100.00	-610.00 105.00	-620.00 103.00	-630.00 110.00	-640.00 90.00	-650.00 102.00	-660.00 108.00	-670.00 85.00
Х Ү Коорд. кінця лін-то, дов. і ширина пл-го(м)	5.00 5.00	5,00 5,00	5.00 5.00	5.00 5,00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00
Коеф-т рель'ефу	1.0000	1.0000	1.0000	1,0000	1.0000	1.0000	1.0000	1.0000
Витрата ПГПС(м. куб/с)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
Шв-ть вихіду ПППС: м/с	0	0	0	0	0	0	0	0
Діаметр (м)	*	*	×.	÷	÷	×	8	
Висота (м)	5.0000	5.0000	5.0000	5.0000	5.0000	5.0000	5.0000	5.0000
Температура (C)	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000
Коеф-т впоряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Викид т/р	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0,0000

Код джерела Технологічні параметри	11009	11010	11011	11012	11013	11014	11015	11016
Вижня г/с	0.0930	0.0930	0.0930	0.0930	0.0930	0.0930	0.0930	0.0930
Клас небезпечн.	1	1	1	1	1	1	1	1
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/M[h=2.00м] мг/м. куб	0.8184	0.8184	0.8184	0,8184	0.8184	0.8184	0.8184	0.8184
XM (st)	28.62	28.62	28.62	28.62	28.62	28.62	28.62	28.62
UM[h=2.00m] (m/c)	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50
Х Ү Коорд. гочеч. почяток лін-го, центр саметр. пл-го (м)	-680.00 90.00	-690.00 100.00	-700.00 98.00	-710.00 95.00	-720.00 110.00	-730.00 90.00	-740.00 85.00	-750.00 115.00
Х Ү Коорд. кінця лін-го, дов. і ширица пл-го(м)	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00
Коеф-т рель' сфу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ПГПС(м. куб/с)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
Шв-ть вяхіду ШПС: м/с	0	0	0	0	0	0	0	0
Діаметр (м)	÷	-	÷		•	-		
Висота (м)	5:0000	5:0000	5.0000	5.0000	5.0000	5.0000	5.0000	\$,0000
Температура (С)	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000
Коеф-т впоряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1,0000	1.0000
Викид т/р	0.0000	0.0000	0.0000	0.0000	0.0000	0.000	0.0000	0.0000

-26 -12.06.2024 "АТ «НАЕК «Епергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

					<u> </u>			
Код джерела Технологічні параметри	11017	11018	11019	11020	11021	11022	11023	11024
Виюна п/с	0.0930	0.0930	0.0930	0.0930	0.0930	0.0930	0.0930	0.0930
Клас небезпечн.	1	1	1	1	1	1	1	1
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/M[h=2.00м] мг/м. куб	0.8184	0.8184	0.8184	0.8184	0.8184	0.8184	0.8184	0.8184
XM (st)	28.62	28.62	28,62	28.62	28.62	28.62	28.62	28.62
UM[h=2.00m] (m/c)	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50
Х	-760.00 200.00	-770.00 220.00	-780.00 230.00	-790.00 250.00	-800.00 220.00	-810.00 250.00	-820.00 220.00	-830.00 250.00
Х У Коорд. кінця лін-го, дов. і ширица пл-го(м)	5.00 5.00							
Коеф-т рель'сфу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ШПС(м. куб'с)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
Шв-ть вяхіду ПГПС: м/с	0	0	0	0	0	0	0	0
Діаметр (м)	÷	-	÷		•	-		
Висота (м)	5:0000	5:0000	5.0000	5.0000	5.0000	5.0000	5.0000	5.0000
Температура (С)	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.000
Коеф-т виоряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1,0000	1.0000
Вакад т/р	0.0000	0.0000	0.0000	0.0000	0.0000	0.000	0.0000	0.0000

-27 -12.06.2024 "АТ «НАЕК «Епергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Код джерела Технологічні параметри	11025	11026	11027	11028	11029	11030	11031	11032
Виюна г/с	0.0930	0.0930	0.0930	0.0930	0.0930	0.0930	0.0930	0.0930
Клас небезпечн.	1	1	1	1	1	1	1	1
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/M[h=2.00м] мс/м. куб	0.8184	0.8184	0.8184	0.8184	0.8184	0.8184	0.8184	0.8184
ХМ (м)	28.62	28.62	28,62	28.62	28.62	28.62	28.62	28.62
UM[h=2.00x1] (M/c)	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50
Х Ү Коорд. гочеч, початок лін-го, центр симетр. цл-го (м)	-840.00 200.00	-850.00 150.00	-860.00 205.00	-870.00 250.00	-880.00 260.00	-890.00 220.00	-900.00 260.00	-910.00 270.00
Х У Коорд. кінця лін-го, дов. і ширяца пл-го(м)	5.00 5.00							
Коеф-т рель'офу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ПГПС(м. куб'с)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
Шв-ть вихіду ПГПС: м/с	0	0	0	0	0	0	0	0
Діаметр (м)	÷	-	÷			-	2	
Висота (м)	5:0000	5:0000	5.0000	5.0000	5.0000	5.0000	5.0000	\$,0000
Температура (C)	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000
Коеф-т впоряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Викид т/р	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000

- 28 -12.06.2024 "АТ «НАЕК «Епергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Розрахункові копцептрації речовини: Вуглецю оксид в розрахункових точках та помера джерел, що надають найбільший внесок на высоті 2.00 м

№ розр. точки	Концентр. у точці частки ГДК	Коорд. розр. точки Х	Коорд. розр. точки Ү	Напрамок вітру	Шандиість вітру	Розмір внеску Q0	уе джерела N0	Розмір вноску Q1	ме джерела N1	Розмір вноску Q2	Ne Josepejia N2
ш	0.0888	0,0	2700.0	73	0.75	0.0003	11020	0.0003	11019	0.0003	11022
22	0.0881	\$50.0	2700.0	63	0.75	0.0003	11020	0.0003	11019	0.0003	11018

	- 29	2	
12.06.2024	"АТ «НАЕК «Епергоатом»"	ЕОЛ 2000 [h] v4.0,	Ліцецзія №133565957

Точки найбільших концентрацій речовний Вуглецю окенд На розрахун. площадці № 1 та помера джерел, що надають найбільший внесок на высоті 2.00 м

Концентрації у точці частки ГДК	Коорд. розр. точкая Х	Коорд. розр. точки Ү	Напрям. вітру	Швидкість вітру	Розмір внеску Q0	№ джерела №0	Розмір внеску Q1	№ джереля N1	Розмір внеску Q2	№ джерела N2
0.3648	-1000.0	250.0	171	0.50	0.0274	11030	0.0245	11031	0.0212	11028
0.3438	-500.0	0.0	327	0.75	0.0185	11005	0.0176	11003	0.0171	11001
0.3342	-750.0	0.0	262	0.50	0,0406	11015	0.0366	11014	0.0269	11013
0.2759	-750.0	250.0	97	0.50	0.0379	11017	0.0213	11016	0.0199	11013
0.2201	-500.0	250.0	29	0.50	0.0085	11007	0.0081	11004	0.0078	11006
0.2025	-1000.0	500.0	126	0.75	0.0068	11029	0.0066	11024	0.0064	11028
0.1960	-750.0	500.0	79	0.75	0.0092	11020	0:0089	11022	0.0079	11024
0.1895	-1000.0	0,0	224	0,50	0.0108	11026	0.0071	11025	0.0067	11027
0.1708	-1250.0	500.0	146	7.00	0.0049	11032	0.0047	11031	0.0044	11029
0.1686	-250.0	0.0	342	7.00	0.0051	11001	0.0051	11002	0.0048	11004



Вуглецю окенд Карта-схема Н=2.00 м

### - 31 -12.06.2024 "AT «НАЕК «Еперговтом»" ЕОЛ 2000 [h] v4.0, Ліцецзія №133565957

Код речовини	Найменувания речовини	ГДК (мг/м.куб)
2754	Вуглеводні граничні с12-с19(розчинник РПК-2661	t i 1.00000000

Фонові концентрації, які вміщують внески діючих джерел (Частки ГДК) (частки ГДК) (Вихідні рівні забруднення) для речовнии : Вуглеводні граничні с12-с19(розчилник РПК-26611 і ін.). Варіант завдання фону : а.

Коорд. Х поста спостережения	Коорд. У поста спостережения	U<2 м/с (штиль)	Швидкість вітру 2⊲U <u* Пи</u* 	Illmunkiens sinpy 2 <u=u* HnC</u=u* 	Швидкість вітру 2 <u<u* С</u<u* 	Швиджість вітру 2⊲U <u* ПдС</u* 	Швялкість вітру 2 <u<u* Пд</u<u* 	Швядкість вітру 2 <u<u* ПлЗ</u<u* 	Шимджість вітру 2 <u<u* 3</u<u* 	Швидкість вітру 2 <u<u* ПяЗ</u<u* 
0.00	0.00	0.40	0.40	0.40	0.40	0.40	0.40	0.40	0.40	0.40

Фонові концентрації без урахування внесків діючих джерел (Часткя ГДК) (частки ГДК) (Власне фон - верхис число, вклад - нижие) для речовини : Вуглеводні граничні с12-с19(розчинник РПК-26611 і ін.). Варіянт завлання фону : а.

Коорд. Х поста спостережения	Коорд. У поста спостережения	U<2 м/с (вписть)	Шпядкість вітру 2 <u<u* Пя</u<u* 	Швидкість вітру 2 <u<u* ПиС</u<u* 	Швидийсть вітру 2 <u<u* С</u<u* 	Швилкість вітру 2≪U <u* ПдС</u* 	Швидкість вітру 2≤U <u* Пд</u* 	Шпилкість вітру 2 <u<u* ПлЗ</u<u* 	Швидкість вітру 2 <u<u* 3</u<u* 	Швидийсть вітру 2 <u<u* Пн3</u<u* 
0.00	0.00	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000

11005 0.0150 1 0.6600	11006 0.0150 1	0.0150	11008
0.0150	0.0150	0.0150	0.0100
0.6600	1		0.0150
0.6600	6	1	1
-	0.6600	0.6600	0.6600
28.62	28.62	28.62	28.62
0.50	0.50	0.50	0.50
) -640.00 90.00	-650.00 102.00	-660.00 108.00	-670.00 85.00
5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00
1.0000	1.0000	1.0000	1.0000
0.0000	0.0000	0.0000	0.0000
0	0	0	0
	2	8	-
5.0000	5.0000	5.0000	5.0000
0 23.0000	23.0000	23.0000	23.000
Vienoc v	10000-000		1.0000
1.0000	1.0000	1.0000	1.0000
	0.50 640.00 90.00 5.00 5.00 0 1.0000 0 0.0000 0 1 5.0000 0 23.0000	0.50 0.50   -640.00 -650.00   90.00 -650.00   5.00 5.00   5.00 5.00   0 1.0000   0 0.0000   0 0.0000   0 0.0000   0 0.0000   0 2.0000   0 23.0000	0.50 0.50 0.50   -640.00 90.00 -650.00 102.00 -660.00 108.00   5.00 5.00 5.00 5.00 5.00 5.00   1.0000 1.0000 1.0000   0 0.0000 0.0000 0.0000   0 0.0000 0.0000 0.0000   0 5.0000 5.0000 5.0000   0 0.0000 0.0000 0.0000   0 5.0000 5.0000 5.0000   0 23.0000 23.0000 1.0000

# - 32 -12.06.2024 "АТ «НАЕК «Епергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

## Перелік джерел, у викидах яких є Вуглеводні граничні с12-с19(розчинник РПК-26611 і ін.)

Код джерела Технологічні параметри	11009	11010	11011	11012	11013	11014	11015	11016
Виюнд г/с	0.0150	0.0150	0.0150	0.0150	0.0150	0.0150	0.0150	0.0150
Клас небезпечн.	1	1	1	1	1	1	1	1
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/M[h=2.00м] мс/м. куб	0.6600	0.6600	0.6600	0,6600	0.6600	0.6600	0.6600	0.6600
ХМ (м)	28.62	28.62	28,62	28.62	28.62	28.62	28.62	28.62
UM[h=2.00m] (m/c)	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50
Х Y Коорд. точеч, почеток лін-го, центр саметр, цл-го (м)	-680.00 90.00	-690.00 100.00	-700.00 98.00	-710.00 95.00	-720.00 110.00	-730.00 90.00	-740.00 85.00	-750.00 115.00
Х У Коорд. кінця лін-го, дов. і ширяца пл-го(м)	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00
Коеф-т рель'сфу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ПГПС(м. куб'с)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
Шв-ть вихіду ШПІС: м/с	0	0	0	0	0	0	0	0
Діаметр (м)	÷		÷		-		-	2
Висота (м)	5:0000	5:0000	5.0000	5.0000	5.0000	5.0000	5.0000	\$,0000
Температура (С)	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000
Коеф-т впоряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Вакад т/р	0.0000	0.0000	0.0000	0.0000	0.0000	0.000	0.0000	0.0000

- 33 -12.06.2024 "АТ «НАЕК «Епергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Код джерела Технологічні параметри	11017	11018	11019	11020	11021	11022	11023	11024
Викид г/с	0.0150	0.0150	0.0150	0.0150	0.0150	0.0150	0.0150	0.0150
Клас небезпечн.	1	1	1	1	1	1	1	1
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/M[h=2.00м] мс/м. куб	0.6600	0.6600 - -	0.6600	0,6600	0.6600	0.6600	0.6600	0.6600
ХМ (м)	28.62	28.62	28,62	28.62	28.62	28.62	28.62	28.62
UM[h=2.00m] (m/c)	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50
Х Ү Коорд. точеч, початок лін-го, центр саметр, пл-го (м)	-760.00 200.00	-770.00 220.00	-780.00 230.00	-790.00 250.00	-800.00 220.00	-810.00 250.00	-820.00 220.00	-830.00 250.00
Х У Коорд. кінця лін-го, дов. і ширяща пл-го(м)	5.00 5.00							
Коеф-т рель'сфу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ПГПС(м. куб'с)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
Шв-ть вяхіду ШПС: м/с	0	0	0	0	0	0	0	0
Діаметр (м)	÷		÷			-		8
Висота (м)	5:0000	5:0000	5.0000	5.0000	5.0000	5.0000	5.0000	\$,0000
Температура (С)	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000
Коеф-т виоряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Вихид т/р	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000

- 34 -12.06.2024 "АТ «НАЕК «Епергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Код джерела Технологічні параметри	11025	11026	11027	11028	11029	11030	11031	11032
Викид г/с	0.0150	0.0150	0.0150	0.0150	0.0150	0.0150	0.0150	0.0150
Клас небезпечн.	1	1	1	1	1	1	1	1
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/M[h=2.00м] мс/м. куб	0.6600	0.6600	0.6600	0.6600	0.6600	0.6600	0.6600	0.6600
ХМ (м)	28.62	28.62	28,62	28.62	28.62	28.62	28.62	28.62
UM[h=2.00m] (m/c)	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50
Х Ү Коорд. гочеч. початок лін-го, центр симетр. пл-го (м)	-840.00 200.00	-850.00 150.00	-860.00 205.00	-870.00 250.00	-880.00 260.00	-890.00 220.00	-900.00 260.00	-910.00 270.00
Х Ү Коорд. кінця дін-го, дов. і пиряна пл-го(м)	5.00 5.00							
Коеф-т рель'офу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ПГПС(м. куб/с)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
Шв-ть вихіду ШПС: м/с	0	0	0	0	0	0	0	0
Діаметр (м)	÷	-	÷	÷		-		
Висота (м)	5.0000	5.0000	5.0000	5.0000	5.0000	5.0000	5.0000	5.0000
Температура (C)	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000
Коеф-т вноряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Викид т/р	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000

- 35 -12.06.2024 "АТ «НАЕК «Епергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Розрахункові концентрації резовини: Вуглеводні граничні с12-с19(розчинник РПК-26611 і ін.) в розрахункових точках та номера джерел, що надають найбільший внесок на высоті 2.00 м

№ розр. точки	Концентр. у точці частки ГДК	Коорд. розр. точки Х	Коорд. розр. точки Ү	Напрамок вітру	Шандиість вітру	Розмір внеску Q0	уе джерела N0	Розмір вноску Q1	№ джерела N1	Розмір вноску Q2	Nê Jokepeja N2
п	0.4071	0,0	2700.0	73	0.75	0.0002	11020	0.0002	11019	0.0002	11022
22	0.4065	\$50.0	2700.0	63	0.75	0.0002	11020	0.0002	11019	0.0002	11018

	- 36	2	
12.06.2024	"AT «HAEK «Eneprostom»"	ЕОЛ 2000 [h] v4.0,	Jlinemia №133565957

Точки найбільних концентрацій речовани Вуглеводні граничні с12-с19(розчинник РПК-26611 і іп.) На розрахун. площадці № 1 та помера джерел, що надають найбільший внесок на высоті 2.00 м

Концентрації у точці частки ГДК	Коорд. розр. точкан Х	Коорд. розр. точки Ү	Напрям. вітру	Швидкість вітру	Розмір внеску Q0	№ джерела №0	Розмір внеску Q1	№ джереля N1	Розмір внеску Q2	№ джерела N2
0.6297	-1000.0	250.0	171	0.50	0.0221	11030	0.0198	11031	0.0171	11028
0.6128	-500.0	0.0	327	0.75	0.0150	11005	0.0142	11003	0.0138	11001
0.6050	-750.0	0.0	262	0.50	0.0327	11015	0.0295	11014	0.0217	11013
0.5580	-750.0	250.0	97	0.50	0.0306	11017	0.0172	11016	0.0160	11013
0.5130	-500.0	250.0	29	0.50	0.0069	11007	0.0065	11004	0.0063	11006
0.4988	-1000.0	500.0	126	0.75	0.0054	11029	0.0053	11024	0.0052	11028
0.4935	-750.0	500.0	79	0.75	0.0074	11020	0:0071	11022	0.0063	11024
0.4883	-1000.0	0.0	224	0,50	0.0087	11026	0.0057	11025	0.0054	11027
0.4732	-1250.0	500.0	146	7.00	0.0039	11032	0.0038	11031	0.0036	11029
0.4714	-250.0	0.0	342	7.00	0.0041	11001	0.0041	11002	0.0039	11004

3500.00 Bertain 3250.00 3000.00 141 0.41 2750.00 2500.00 2250.00 2000.00 1750.00 1500.00 1250.00 1000.00 750.00 \$00.00 250.00 Europerius, http 0.00 CONTRACTOR DEPENDENCIES AND INCOMES -250.00 -500.00 -750.00 -1000.00 1250.00 -1500.00 -1759.00 -2000.00 -2250.00 -2500.00 -2750.00 E -3000.00 -3150.00 3500.00 -3000.00 -2503.00 -2000.00 -1500.00 -1000.00 -500.00 0.00 500.00 1000.00 1500.00 2000.00 2500.00 3000.00 3500.00 Нормативна санітарно-захисна зона

Вуглеводні граничні е12-е19(розчинник РНК-26611 і ін.) Карта-ехема H=2.00 м

### - 38 -12.06.2024 "AT «НАЕК «Еперговтом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Код гр. сум.	Код речования	Найменувания речовнии	ГДК (мг/м.куб)
31	301	Азоту діокена	0.20000000
	330	Ангідряд сірчистий	0.50000000

Фонові концентрації, які вміщують внески діючих джерел (Частки ГДК) (частки ГДК) (Вихідні рівні забруднення) для групи сумації № 31. Варіант завдання фону : а.

Коорд. Х поста спостереження	Коорд. У поста спостереження	U<2 м/е (штиль)	Шинджість и́тру 2 <u<u* Пи</u<u* 	Швидкість nirpv 2 <u<u* ПнС</u<u* 	Швидијеть nirpy 2 <u<u* С</u<u* 	Швидисість nirpy 2 <u<u* ПаС</u<u* 	Швидкість вітру 2 <u<u* Па</u<u* 	Швидкість літру 2-U-U* ПдЗ	Швядкість вітру 2 <u<u* 3</u<u* 	Швядкість nirpv 2 <u<u* Пи3</u<u* 
0.00	0.00	0.08	0.08	0.08	0.08	0.08	0.08	0.08	0.08	0.08

Фонові концентрації без урахування внесків діючих джерел (Частки ГДК) (частки ГДК) (Власне фон - верхне число, вклад - нижне) для групи сумації № 31. Варіант завдання фону : а.

Коорд. Х поста спостереження	Коорд. У поста спостереження	U<2 м/с (штиль)	Швидкість вітру 2 <u<u* Пя</u<u* 	Швидкість вітру 2 <u<u* ПнС</u<u* 	Швидкість ^{Вітру} 2 <u<u* С</u<u* 	Швидкість ^{Birpy} 2 <u<u* ПдС</u<u* 	Швндкість вітру 2 <u<u* Па</u<u* 	Швидкість вітру 2 <u<u* ПаЗ</u<u* 	Швидкість вітру 2 <u<u* 3</u<u* 	Швядкість вітру 2 <u<u* Пв3</u<u* 
0.00	0.00	0.0800	0.0800	0.0800	0.0800	0.0800	0.0800	0.0800	0.0800	0.0800

	- 39	-	
12.06.2024	"АТ «НАЕК «Епергоатом»"	ЕОЛ 2000 [h] v4.0,	Ліцензія №133565957

Перелік джерел, у викидах яких с Група сумації № 31

Кол джерела Технологічні	***11001	***11002	***11003	***11004	***11005	***11006
параметри						
Викид г/с	0.162500009	0.162500009	0.162500009	0.162500009	0.162500009	0.162500009
Клас небезпечн.	1	1	1	1	1	1
СМІһ=2.00м] (частан ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мс/м. куб	14.3009	14.3009	14.3009	14,3009	14.3009	14.3009
ХМ (м)	28.62	28.62	28.62	28,62	28.62	28.62
UM[h=2.00m] (m/c)	0.50	0.50	0.50	0.50	0.50	0.50
Х Ү Коорд. гочеч. початок лін-го, центр симетр. пл-го (м)	-600.00 100.00	-610.00 105.00	-610.00 -620.00 105.00 103.00		-640.00 90.00	-650,00 102.00
Х Ү Коорд, кінця лін-то, дов. і ширина пл-го(м)	5.00 5.00	5.00 5.00	5.00 5.00	\$.00 5.00	5.00 5.00	5.00 5.00
Коеф-т рель'ефу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ПГПС(м. куб/с)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
Шв-ть вихіду ПППС: м/с	0	0	0	0	0	0
Діаметр (м)			(A)	38	÷	*
Висота (м)	5.0000	5.0000	5.0000	5.0000	5.0000	5.0000
Температура (C)	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000
Коеф-т впоряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Викид т/р	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000

Код джерела Технологічні параметри	***11007	***11008	***11009	***11010	***11011	***11012
Виюня п/е	0.162500009	0.162500009	0.162500009	0.162500009	0.162500009	0.162500009
Клас небезпечн.	1	1	1	1	1	1
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мг/м. куб	14,3009	14.3009	14.3009	14.3009	14.3009	14.3009
ХМ (м)	28.62	28.62	28.62	28.62	28.62	28.62
UM[h=2.00st] (ss/c)	0.50	0.50	0.50	0.50	0.50	0.50
Х Ү Коорд. точеч. початок лін-го, центр симетр. пл-го (м)	-660.00 108.00	-670.00 85.00	-680.00 90.00	-690.00 100.00	-700.00 98.00	-710.00 95.00
Х Ү Коорд, кінця лін-го, дов. і ширяца пл-го(м)	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00
Коеф-т рель'ефу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ШПС(м. куб/с)	0.0000	0.0000	0.0000	0.0000	0.0000	0.000
Шв-ть вихіду ШПС: м/с	0	0	0	0	0	0
Діаметр (м)		14	-			
Висота (м)	5.0000	5.0000	5.0000	5.0000	5,0000	5.0000
Температура (С)	23.0000	23,0000	23,0000	23.0000	23.0000	23.0000
Коеф-т вноряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Вяжид т/р	0.0000	0.0000	0.0000	0:0000	0.0000	0.0000

## - 40 -12.06.2024 "АТ «НАЕК «Еперговтом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Код джерела	***11013	***11014	***11015	***11016	***11017	***11018
Технологічні параметри						
Виюд п/с	0.162500009	0.162500009	0.162500009	0.162500009	0.162500009	0.162500009
Клас небезпечн.	1	1	1	1	1	1
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мг/м. куб	14,3009	14.3009	14.3009	14.3009	14.3009 - -	14.3009
ХМ (м)	28.62	28.62	28.62	28.62	28.62	28.62
UM[h=2.00st] (ss/c)	0.50	0.50	0.50	0.50	0.50	0.50
Х Ү Коорд. точеч. початок лін-го, центр симетр. пл-го (м)	-720.00 110.00	-730.00 90.00	-740.00 85.00	-750.00 115.00	-760.00 200.00	-770.00 220.00
Х Ү Коорд, кінця лін-го, дов. і ширяца пл-го(м)	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00
Коеф-т рель'ефу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ШПС(м. куб/с)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
Шв-ть вихіду ШПІС: м/с	0	0	0	0	0	0
Діяметр (м)		4	-	-		
Висота (м)	5.0000	5.0000	5.0000	5.0000	5.0000	5.0000
Температура (С)	23.0000	23,0000	23,0000	23.0000	23.0000	23.0000
Коеф-т впоряд. осід.	1.0000	1.0000	1,0000	1.0000	1.0000	1.0000
Вакад т/р	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000

# -41-12.06.2024 "АТ «НАЕК «Еперговгом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Код джерела Технологічні цараметри	***11019	***11020	***11021	***11022	***11023	***11024
Виюнд г/с	0.162500009	0.162500009	0.162500009	0.162500009	0.162500009	0.162500009
Клас небезпечн.	1	1	1	1	1	1
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мг/м. куб	14.3009	14.3009	14.3009	14.3009	14.3009	14.3009
ХМ (м)	28.62	28.62	28.62	28.62	28.62	28.62
UM[h=2.00st] (ss/c)	0.50	0.50	0.50	0.50	0.50	0.50
Х Ү Коорд. гочеч. початок лін-го, центр симетр. пл-го (м)	-780.00 230.00	-790.00 250.00	-800.00 220.00	-810.00 250.00	-820.00 220.00	-830.00 250.00
Х Ү Коорд, кінця лін-го, дов. і ширяца пл-го(м)	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00
Коеф-т рель'ефу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ШПС(м. куб/с)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
Шв-ть вихіду ШПС: м/с	0	0	0	0	0	0
Діаметр (м)			•			-
Висота (м)	5.0000	5.0000	5.0000	5:0000	5,0000	5.0000
Температура (С)	23.0000	23.0000	23,0000	23.0000	23.0000	23,0000
Коеф-т вноряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Вяжид т/р	0.0000	0.0000	0.0000	0:0000	0.0000	0.0000

# - 42 -12.06.2024 "AT «НАЕК «Еперговгом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Код джерела Технологічні параметри	***11025	***11026	***11027	***11028	***11029	***11030
Вяюнд г/с	0.162500009	0.162500009	0.162500009	0.162500009	0.162500009	0.162500009
Клас небезпечн.	1	1	1	1	1	1
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мг/м. куб	14.3009	14.3009	14.3009	14.3009	14.3009	14.3009
ХМ (м)	28.62	28.62	28.62	28.62	28.62	28.62
UM[h=2.00st] (ss/c)	0.50	0.50	0.50	0.50	0.50	0.50
Х Ү Коорд. гочеч. початок лін-го, центр симетр. пл-го (м)	-840.00 200.00	-850.00 150.00	-860.00 205.00	-870.00 250.00	-880.00 260.00	-890.00 220.00
Х Ү Коорд, кінця лін-го, дов. і ширяца пл-го(м)	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00	5.00 5.00
Коеф-т рель'ефу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ШПІС(м. куб'є)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
Шв-ть вихіду ШПС: м/с	0	0	0	0	0	0
Діаметр (м)	÷		•			
Висота (м)	5.0000	5.0000	5.0000	5.0000	5,0000	5.0000
Температура (С)	23.0000	23.0000	23,0000	23.0000	23.0000	23.0000
Коеф-т вноряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Вакад т/р	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000

-43 -12.06.2024 "АТ «НАЕК «Еперговгом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Код джерела Технологічні параметри	***11031	***11032		
Викид г/с	0.162500009	0.162500009		
Клас небезпечн.	1	1		
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мс/м. куб	14,3009	14.3009		
ХМ (м)	28.62	28.62		
UM[h=2.00m] (m/c)	0.50	0.50		
Х Ү Коорд. гочеч. почяток лін-го, центр саметр. пл-го (м)	-900.00 260.00	-910.00 270.00		
Х Ү Коорд. кінця ліп-го, дов. і ширяна пл-го(м)	5.00 5.00	5.00 5.00		
Коеф-т рель' сфу	1.0000	1.0000		
Витрата ШПС(м. куб'с)	0.0000	0.0000		
Шв-ть вихіду ПГПС: м/с	0	0		
Діаметр (м)	÷	1		
Висота (м)	5.0000	5.0000		
Температура (C)	23.0000	23.0000		
Коеф-т впоряд. осід.	1.0000	1.0000		
Викид т/р	0.0000	0.0000		

- 44 -									
12.06.2024	"AT «HAEK «Eneprostom»"	ЕОЛ 2000 [h] v4.0,	Ліцензія №133565957						

Розрахункові концентранії групи сумації № 31 в розрахункових точках та помера джерел, що надають найбільший внесок на высоті 2.00 м

№ розр. точки	Концентр. у точці частки ГДК	Коорд. розр. точки Х	Коорд. розр. точки Ү	Напрамок вітру	Шандність вітру	Розмір внеску Q0	уе джерела N0	Розмір вноску Q1	№ джерела N1	Розмір вноску Q2	Ne Juscepena N2
п	0.2341	0,0	2700.0	73	0.75	0.0050	11020	0.0050	11019	0.0050	11022
22	0.2217	550.0	2700.0	63	0.75	0.0046	11020	0.0046	11019	0.0045	11018

1	Точки найбільших концентрацій трупи сумації № 31 На розрахуп. площадці № 1 та помера джерел, що надають найбільший внесок на высоті 2.00 м											
Концентрації у точці частки ГДК	Коорд. розр. гочван Х	Коорд. розр. точки Ү	Напрям. вітру	Швадкість вітру	Розмір внеску Q0	№ джерела N0	Розмір внеску Q1	№ джереля N1	Розмір внеску Q2	№ джерела N2		
5.0569	-1000.0	250.0	171	0.50	0.4795	11030	0.4283	11031	0.3712	11028		
4.6901	-500.0	0.0	327	0.75	0.3241	11005	0.3071	11003	0.2979	11001		
4.5210	-750.0	0.0	262	0.50	0.7089	11015	0.6391	11014	0.4702	11013		
3.5030	-750.0	250.0	97	0.50	0,6620	11017	0.3724	11016	0.3477	11013		
2.5287	-500.0	250.0	29	0.50	0.1494	11007	0.1416	11004	0.1358	11006		
2.2208	-1000.0	500.0	126	0.75	0.1181	11029	0.1147	11024	0.1126	11028		

0.1604

0.1882

0.0849

0.0894

11020

11026

11032

11001

0.1548

0.1232

0.0820

0.0886

11022

11025

11031

11002

0.1374

0.1166

0.0771

0.0837

11024

11027

11029

11004

2.1062

1.9929

1.6664

1.6276

-750.0

-1000.0

-1250.0

-250.0

500.0

0.0

500.0

0,0

79

224

146

342

0.75

0.50

7.00

7.00

		5-
12.06.2024	"AT «HAEK «Eneproatom»"	¹ ЕОЛ 2000 [h] v4.0, Ліцензія №133565957
	Contraction of the second s	



Група сумації № 31 Карта-схема Н=2.00 м

# C.2 Calculation of pollutant dispersion during the KhNPP units 5 and 6 operation

Copyright(C) ТОВ «Софт фонд» м. Киля

Теп. (044)599 35 57 E-Mail mfo@sfind kiev ua

АТ «НАЕК «Емергоатом», Лицензія №133565957

# EOЛ 2000[h] (Windows версія)

Автоматизована система розрахунку розсіювання викидів шкідливих речовин

Загальний звіт про результати розрахунку розсіювання

"Період експлуатації"

Розрахунковий модуль системи реалізус методику ОНД-86 Програма рекомендована для використання Міністерством охорони навхолишнього природного середовища України(2464/19/4-10 от 13.03.2006)

				Завдання	на розраху	нок.			
Буду	Шы Шансткіе вати розр	Найме Коди пр Коли Швисл Швисл Миже Коли Коли Коли Коли Коли Коли Коли Коли	пувания мі ом. майдан пречовни груп суман кість вітру ( іпру (част. часткя U часткя U часткя Чапр. в. папр. віт ть найб. вк сть маяке ко козалявій фе СЗЗ/зону в розрахунку	та чисків м/с) U сер. зв.) р. надфакел вітру ру тадн. энп. энп. энп. (м) підп. (м)	њної) рисмства	Heriumu 1 1309-37 1317-38 10102-4 1333-86 630-08-4 (2735) 31 35 2 7 0.5 1 - 10 - 3 10 Tax Hi/Hi 2	n -1 (123) 1313-1 -0 (146) 7440-4 4-0 (301) 7631 -4 (328) 7446-6 0 (337) (342) (2754) (290) 11002 1.5	(3-9 (143) (7-3 (203) (9-5 (323) (9-5 (330) 7681-49-4 (343) (10265)	) (344)
			п	араметри р	озрахунко	вііх майдая	PROCE		
№ π/п	Коорд. Х	Коорд. Ү	Довжина	Шарана	Кут. пов. розр. майд. відн. вісі ОХ осн. сист. коорд.		Крок по сіці вісь ОХ	Крок по сітці вісь ОҮ	Особл. вимоги
1	0.0	0.0	7000.0	7000.0	0	.0	250.0	250.0	0

	- 2 -	and an exception of	
19/09/2024	"АТ «НАЕК «Евергоатом»*	ЕОЛ 2000 [h] v4.0,	Ліценція №133565957

Код міста	Найменування міста	Сер. температура самого теплого місяця (град С)	Сер. температура самого холодного місяця (град С)	Гранична швидкість вітру (м'с)	Регіональний коефіцієнт стратифікації	Кут між північним напрям. та віссю ОХ осн. сист. коорд. (град)	Плоша міста (кв. км)
<u>(</u> 1	Нетішин	23.0	-5.7	7.0	200	90	70

Широта (град.,хв.,сек.)	Шнрота (пиш. чи пдш.)	Довгота (град.,хв.,сек.)	Довгота (зд. чи сд.)	Ймовірність повтору вітру(Пв)	Ймовірність повтору вітру(ПнСх)	Ймовірність повтору вітру(Сх)	Ймовірність повтору вітру(ПдСх)	Ймовірність повтору вітру(Пд)
50rpag.19'48''	IIH	26град.38'24"	сд	10.8	5	9.4	13.9	16.6

Ймовірність	Ймовірність	Ймовірність
повтору	повтору	повтору
вітру(ПдЗх)	вітру(3x)	вітру(ПнЗх)
11.7	20.8	11.8

# 

Код пр. майд.	Найменування промислового майданчика	Код речовин (групи сумації)	Найменування речовини (Коди речовин, що входять у групу сумації ).	Потужність викиду (г/с)	Потужність вихиду (т/рік)
1	Хмельницька АЕС	Tp. evat. № 31 Tp. evat. № 35 Tp. evat. № 35 Tp. evat. № 1100 1309-37-1 (123) 1313-13-9 (143) 1317-38-0 (146) 7440-47-3 (203) 10102-44-0 (301) 7631-86-9 (323) 1333-86-4 (328) 7446-09-5 (330) 630-08-0 (337) (342) 7681-49-4 (343) (344) (2735) (2754) (2902) (10265)	301 330 310 342 342 344 Запіза окенд**(у перерахунку на запізо) Марганець та його сполуки (у перерахунку на двоокнє Міді окенд(у перерахунку на мідь) Хром шестивалентинй (в перерахунку на приокнє хрому) Азоту діоксид Кремнію діокенд аморфний (Аероенд-175) Сажа Ангідрид сірчистий Вуглещо окенд Фтористі сполуки газоподібні (фтористий Фтористі сполуки добре розчинні неоргавічні (фторид і Фтористі сполуки добре розчинні неоргавічні (фторид і Фтористі сполуки добре розчинні неорганічні (фторид і Маєдо мінеральне нафтове (веретенне, машине, Вуглеводні гранячий с12-с19(розчинник РПК-26611 і ін.) Зважені речовани, недиференційовані за складом	68.7572 16.0572 0.0256 0.0272 0.0044 0.0048 0.0005 21.1040 0.0026 0.0020 15.9972 2.7043 0.0024 0.0024 0.0026 0.0024 0.0026 0.0026 0.0024 0.0026 0.0026 0.0024 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0026 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.0027 0.	10.1151 3.2786 0.0444 0.0460 0.0005 0.0003 2.7746 0.0018 0.0000 3.1786 0.2753 0.0040 0.0076 0.0044 9.8380 0.1582 0.3065 0.0112

Код речовини	Найменувания речовнии	ГДК (мг/м.куб)
1309-37-1 (123)	Заліза окенд**(у перерахунку на залізо)	0.4000000

Фонові концентрації, які вміщують внескя діючих джерел (Часткя ГДК) (часткя ГДК) (Вихідні рівні забруднення) для речовини : Заліза оксид**(у перерахунку на залізо). Варіант завдання фону : а.

Коорд. Х поста спостереження	Коорд. Ү поста спостереження	U<2 м/с (штиль)	Швядкість вітру 2 <u<u* Пя</u<u* 	IIIвидиість вітру 2 <u<u* ПнС</u<u* 	Швядкість пітру 2 <u<u* С</u<u* 	Швилкість віпуv 2 <u<u* ПлС</u<u* 	Швидкість вітру 2 <u<u* Пд</u<u* 	Швидкість вітру 2 <u<u* Пд3</u<u* 	Швицкість вітру 2 <u<u* 3</u<u* 	Швилкість вітру 2 <u<u* Пв3</u<u* 
0.00	0.00	0.04	0.04	0.04	0.04	0.04	0.04	0.04	0.04	0.04

Фонові концентрації без урахування внесків діючих джерел (Частки ГДК) (частки ГДК) (Власне фон - верхне число, вклад - нижне) для речовник : Заліза оксид**(у перерахувку на залізо). Варіант завдання фону : а.

Коорд, Х поста спостережения	Коорд. Ү поста спостережения	U<2 м/с (штиль)	Швилкість вітру 2 <u<u* Пп</u<u* 	Шпилкість вітру 2 <u<u* ПиС</u<u* 	IIInnakien. Birpy 2 <u<u* C</u<u* 	Шпадкість вітру 2 <u<u* ПдС</u<u* 	Швидкість вітру 2 <u<u* Пд</u<u* 	Швидкість вітру 2 <u<u* Пл3</u<u* 	Illinuxien sirpv 2 <u<u* 3</u<u* 	Швядкість вітру 2 <u<u* Пи3</u<u* 
0.00	0.00	0.0400	0.0400	0.0400	0.0400	0.0400	0.0400	0.0400	0.0400	0.0400

#### -4 -19/09/2024 "АТ «НАЕК «Енергоятом»" ЕОЛ 2000 [b] v4.0, Ліцензія №133565957

Перелік джерел, v викидах яких є Заліза оксид**(у перерахунку на залізо)

Код джерела Технологічні параметри	10201	11046		
Винад г/с	0.0136	0.0136		
Клас небезпечн.	5	5		
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мс/м. куб	0.0532	0.0532		
ХМ (м)	86.53	86.53		
UM[h=2.00м] (м/с)	0.54	0.54		
Х Ү Коорд. точеч. початок лін-го, центр симетр. лл-го (м)	475.00 -120.00	-320.00 400.00		
Х Ү Коорд. кінця лія-го, дов. і цациона плго(м)	0.00 0.00	0.00 0.00		
Коеф-т рель`офу	1.0000	1.0000		
Внтрата ПГПС(м. куб/с)	2.7400	2.7400		
Шв-ть вихіду ПГПС: м/е	9.6908	9.6908		
Діаметр (м)	0.6000	0.6000		
Висота (м)	14.0000	14.0000		
Температура (С)	23.0000	23.0000		
Коеф-т впоряд. осід.	1.0000	1.0000		
Викид т/р	0.0230	0.0230		

Розрахункові концентрації речовини: Заліза оксид^{**}(у перерахунку на залізо) в розрахункових точвах та номера джерел, що надають найбільший внесок на высоті 2.00 м

№ розр. точки	Конпентр. v точші частки ГДК	Коорд. розр. точкн Х	Коорд, розр, точки Ү	Напрямок вітру	Швидкість вітру	Розмір внеску Q0	№ джерела N0	Розмір внеску Q1	№ джерела N1	Розмір внеску Q2	№ джерела N2
11	0.0403	0.0	2700.0	82	7.00	0.0003	11046	0.0000	10201		
22	0.0403	550.0	2700.0	70	7.00	0.0003	11046	0.0000	10201		

# 19/09/2024 "АТ «НАЕК «Епергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Точки найбільших концентрацій речовини Заліза оксид**(v перерахунку на залізо) На розрахун. площадці № 1 та номера джерел, що надають найбільший внесок на высоті 2.00 м

Концентрації у точці частки ГДК	Коорд, розр. точки Х	Коорд. розр. точки Ү	Напрям. вітру	Швядкість вітру	Розмір внеску Q0	№ джерела N0	Розмір внеску Q1	№ джереля N1	Розмір внеску Q2	№ джерела N2
0.0504	-250.0	500.0	52	0.5399	0.0104	11046				
0.0503	500.0	0.0	82	0.5399	0.0103	10201				
0.0500	500.0	-250.0	277	0.5399	0.0100	10201	0.0000	11046		
0.0488	-250.0	250.0	291	0.5399	0.0088	11046				
0.0482	-500.0	500.0	148	0.8099	0.0076	11046	0,0006	10201		
0.0470	-500.0	250.0	219	0.8099	0.0070	11046				
0.0465	250.0	0.0	151	0.8099	0.0065	10201				
0.0462	250.0	-250.0	214	0.8099	0.0062	10201				
0.0457	750.0	-250.0	330	0.8099	0.0052	10201	0.0005	11046		
0.0452	750.0	0.0	28	0.8099	0.0052	10201				



Заліза оксид**(у перерахунку на залізо) Карта-схема H=2.00 м

#### .7-19/09/2024 "АТ «НАЕК «Евергоатом»" ЕОЛ 2000 [h] v4.0, Ліцетія №133565957

Код речовнии	Найменувания речовнин	ГДК (мг/м.куб)
1313-13-9 (143)	Марганець та його сполуки (у перерахунку на	0.01000000

Фонові концентрації, які вміщують внески діючих джерел (Частки ГДК) (частки ГДК) (Вихідні рівні забруднення) для речовини : Марганець та його сполуки (у перерахунку на двоокис марганцю). Варіант завдання фону : а.

Коорд. Х поста спостереження	Коорд. Ү поста спостереження	U<2 м/с (штиль)	Швядкість вітру 2 <u<u* Пя</u<u* 	IIIвилкість вітру 2 <u<u* ПнС</u<u* 	Швядкість вітру 2 <u<u* С</u<u* 	Шпадкість віпуv 2 <u<u* ПдС</u<u* 	Швядкість вітру 2 <u<u* Пд</u<u* 	Швядкість вітру 2 <u<u* Пд3</u<u* 	Швидкість вітру 2 <u<u* 3</u<u* 	Швяджість вітру 2 <u<u* Пн3</u<u* 
0.00	0.00	0.40	0.40	0.40	0.40	0.40	0.40	0.40	0.40	0.40

Фонові концентрації без урахування внесків діючих джерел (Частки ГДК) (частки ГДК) (Власне фон - верхне число, вклад - нижне) для речовния : Марганець та його сполуки (у перерахунку на двоокиє марганшо). Варіант завдання фону : а.

Коорд. Х поста спостереження	Коорд. Ү поста спостереження	U<2 м/с (штиль)	Швидкість вітру 2 <u<u* Пп</u<u* 	Шпилкість вітру 2 <u<u* ПиС</u<u* 	Швидкість вітру 2 <u<u* C</u<u* 	IIImmedicers simpy 2 <u<u* HaC</u<u* 	Швилкість вітру 2 <u<u* Пд</u<u* 	Швилкість вітру 2 <u<u* Пд3</u<u* 	Шпилкість sirpv 2 <u<u* 3</u<u* 	Швидкість вітру 2 <u<u* Пи3</u<u* 
0.00	0.00	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000

# -8 -19/09/2024 "АТ «НАЕК «Енергоатом»" ЕОЛ 2000 [b] v4.0, Ліцензія №133565957

Переліх джерел, <br/> v викидах яких  $\epsilon$ Марганець та його сполуки (у перерахунку на дво<br/>окис марганцю)

Код джерела Технологічні параметри	10201	11046
Винац г/с	0.0022	0.0022
Клас небезпечн.	5	5
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мс/м. куб	0,3444	0.3444
ХМ (м)	86.53	86.53
UM[h=2.00м] (м/с)	0.54	0.54
Х Ү Коорд. точеч. початок лін-го, центр симетр. пл-го (м)	475.00 -120.00	-320.00 400.00
Х Ү Коорд. кінця лін-го, дов. і ширина пл-го(м)	0.00 0.00	0.00 0.00
Коеф-т рель' офу	1.0000	1.0000
Витрата ПППС(м. куб/с)	2.7400	2.7400
Шв-ть вихіду ПГПС: м/е	9.6908	9.6908
Діаметр (м)	0.6000	0.6000
Висота (м)	14.0000	14.0000
Температура (С)	23.0000	23.0000
Коеф-т впоряд. осід.	1.0000	1.0000
Викид т/р	0.0023	0.0023

Розрахункові концентрації речовани: Марганець та його сполуки (у перерахунку на двоокие марганцю) в розрахункових точках та номера джерел, що надають найбільший внесок на высоті 2.00 м

№ розр. точки	Концентр. v точці частки ГДК	Коорд. розр. точкн Х	Коорд, розр. точки Ү	Напрямок вітру	Швядвість вітру	Розмір внеску Q0	№ джерела N0	Розмір внеску Q1	Nb джерела N1	Розмір внеску Q2	№ джерела N2
11	0.4020	0.0	2700.0	82	7.00	0.0020	11046	0.0000	10201		
22	0.4019	550.0	2700.0	70	7.00	0.0019	11046	0.0000	10201		

### -9-19/09/2024 "АТ «НАЕК «Еверговтом»" ЕОЛ 2000 [h] v4.0, Ліцевзія №133565957

Точки найбільших кондентрацій речовання Марганець та його сполуки (у перерахунку на двоокис марганцво) На розрахун. площадці № 1 та номера джерел, що надають найбільший внесок на высоті 2.00 м

Концентрації у точці частки ГДК	Коорд. розр. точки Х	Коорд. розр. точки Ү	Напрям. вітру	Швядкість вітру	Розмір внеску Q0	№ джереца N0	Розмір внеску Q1	№ джерела N1	Розмір внеску Q2	№ джерела N2
0.4675	-250.0	500.0	52	0.5399	0.0675	11046				
0.4669	500.0	0.0	82	0.5399	0.0669	10201				
0.4646	500.0	-250.0	277	0.5399	0.0646	10201	0.0001	11046		
0.4571	-250.0	250.0	291	0.5399	0.0571	11046				
0.4531	-500.0	500.0	148	0.8099	0.0490	11046	0.0041	10201		
0.4452	-500.0	250.0	219	0.8099	0.0452	11046				
0.4419	250.0	0.0	151	0.8099	0.0419	10201				
0.4399	250.0	-250.0	214	0.8099	0.0399	10201				
0.4369	750.0	-250.0	330	0.8099	0.0333	10201	0.0036	11046		
0.4339	750.0	0.0	28	0.8099	0.0339	10201				

Мартанець та його сполуки (у перерахунку на двоокие марганцю) Карта-схема H=2.00 м



### -11 -19/09/2024 "АТ «НАЕК «Евергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Код резовини	Найменування речовнин	ГДК (мг/м.куб)
1317-38-0 (146)	Міді оксид(у перерахунку на мідь)	0.02000000

## Опис фона для речовини : Міді оксид(у перерахунку на мідь)

Опис фону відсутній чи не враховувалось у розрахунках.

Перелік джерел, у викидах яких  $\epsilon$ Міді оксид<br/>(у перерахунку на мідь)

Код джерела Технологічні параметри	10201	11046		
Викид г/е	0.0024	0.0024		
Клас небезпечн.	5	5		
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мс/м. куб	0.1879	0.1879		
ХМ (м)	86.53	86.53		
UM[h=2.00st] (ss/c)	0.54	0.54		
Х Ү Коорд. точеч. початок лін-го, пентр симетр. пл-го (м)	475.00 -120.00	-320.00 400.00		
Х Ү Коорд, кінця лін-го, дов. і цпарина пл-го(м)	0.00 0.00	0.00 0.00		
Коеф-т рель' ефу	1.0000	1.0000		
Ватрата ШПС(м. куб/с)	2.7400	2.7400		
Шв-ть вихіду ПГПС: м/с	9.6908	9.6908		
Діаметр (м)	0.6000	0.6000		
Висота (м)	14.0000	14.0000		
Температура (С)	23.0000	23.0000		
Коеф-т впоряд. осід.	1.0000	1.0000		
Викид т/р	0.00025	0.00025		
### - 12 -19/09/2024 — "AT «НАЕК «Евергоатом»" — ЕОЛ 2000 [h] v4.0, Ліценія №133565957

Розрахункові концентрації речовини: Міді окенд(у перерахунку на мідь) в розрахункових точках та номера джерел, що надають найбільший внесок на высоті 2.00 м

№ розр. точкн	Концентр. у точці частки ГДК	Коорд. розр. точки Х	Коорд. розр. точки Ү	Напрямок вітру	Швильість вітру	Розмір внеску Q0	№ джерела N0	Розмір внеску Q1	№ джерела N1	Розмір внеску Q2	№ джерела N2
11	0.0011	0.0	2700.0	82	7.00	0.0011	11046	0.0000	10201		
22	0.0010	550.0	2700.0	70	7.00	0.0010	11046	0.0000	10201		

Точки найбільших концентрацій речовини Міді оксид(у перерахунку на мідь) На розрахун. плошадці № 1 та номера джерел, що надають найбільший внесок на высоті 2.00 м

Концентрації у точці частки ГДК	Коорд. розр. точки Х	Коорд, розр. точки Ү	Натрям. вітру	Швидкість вітру	Розмір внеску Q0	№ джерена N0	Розмір внеску Q1	№ джерела N1	Розмір внеску Q2	№ джерела N2
0.0368	-250.0	500,0	52	0.5399	0.0368	11046				
0.0365	500.0	0.0	82	0.5399	0.0365	10201				
0.0352	500.0	-250.0	277	0.5399	0.0352	10201	0.0000	11046		
0.0311	-250.0	250.0	291	0.5399	0.0311	11046				
0.0290	-500.0	500.0	148	0.8099	0.0267	11046	0.0023	10201		
0.0246	-500.0	250.0	219	0.8099	0.0246	11046				
0.0228	250.0	0.0	151	0.8099	0.0228	10201				
0.0217	250.0	-250.0	214	0.8099	0.0217	10201				
0.0201	750.0	-250.0	330	0.8099	0.0182	10201	0.0019	11046		
0.0185	750.0	0.0	28	0.8099	0.0185	10201				

-13 -19/09/2024 "АТ «НАЕК «Евергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957



Міді окенд(у перерахунку на мідь) Карта-ехема Н=2.00 м

## - 14 -19/09/2024 "АТ «НАЕК «Енергоатом»" ЕОЛ 2000 [b] v4.0, Ліцензія №133565957

Код речовнии	Найменування речовнии	ГДК (мг/м.куб)
7440-47-3 (203)	Хром шестивалентний (в перерахунку на триокис	0.00150000

Опис фона для речовани : Хром шестивалентний (в перерахунку на тряокис хрому)

Опне фону відсутній чи не враховувалось у розрахунках.

Перелік джерел, у викидах яких  $\epsilon$  Хром шестивалентний (в перерахунку на триокис хрому)

Код джерела Технологічні параметри	10201	11046
Викид r/c	0.00025	0.00025
Клас небезпечи.	5	5
СМ[b=2.00м] (частки ГДК) СМ[b=2.00м] мг/м. куб СМ/М[b=2.00м] мс/м. куб	0.2609	0.2609
ХМ (м)	86.53	86.53
UM[h=2.00st] (st/c)	0.54	0.54
Х У Коорд. точеч. початок лін-го, центр снолетр. пл-го (м)	475.00 -120.00	-320.00 400.00
Х Ү Коорд, кінця лія-го, дов. і ширина пл-го(м)	0.00 0.00	0.00 0.00
Коеф-т рель' сфу	1.0000	1.0000
Витрата ШПС(м. куб/с)	2.7400	2.7400
Шв-ть вахіду ПГПС: м/с	9.6908	9.6908
Діаметр (м)	0.6000	0.6000
Висота (м)	14.0000	14.0000
Температура (С)	23.0000	23.0000
Коеф-т впоряд. осід.	1.0000	1.0000
Викид т/р	0.00017	0.00017

### - 15 -19/09/2024 "АТ «НАЕК «Еверговтом»" ЕОЛ 2000 [h] v4.0, Ліпевзія №133565957

Розрахункові концентрації речовник: Хром шестивалентний (в перерахунку на трнокиє хрому) в розрахункових точках та номера джерел, що надають найбільший внесок на высоті 2.00 м

№ розр. точкн	Концентр. у точці частки ГДК	Коорд. розр. точки Х	Коорд. розр. точки Ү	Напрямок вітру	Швилкість вітру	Розмір внеску Q0	№ джерела N0	Розмір внеску Q1	№ джерела N1	Розмір внеску Q2	Ne джерела N2
11	0.0016	0.0	2700.0	82	7.00	0.0015	11046	0.0000	10201		
22	0.0014	550.0	2700.0	70	7.00	0.0014	11046	0.0000	10201		

Точки найбільших концентрацій речовини Хром шестивалентний (в перерахунку на триокис хрому) На розрахун. плошадці № 1 та номера джерел, що надають найбільший внесок на высоті 2.00 м

Концентрації у точці частки ГДК	Коорд. розр. точки Х	Коорд, розр. точки Ү	Натрям. вітру	Швидкість вітру	Розмір внеску Q0	№ джерена N0	Розмір внеску Q1	№ джерела N1	Розмір внеску Q2	№ джерепа N2
0.0511	-250.0	500,0	52	0.5399	0.0511	11046				
0.0507	500.0	0.0	82	0.5399	0.0507	10201				
0.0490	500.0	-250.0	277	0.5399	0.0489	10201	0.0000	11046		
0.0432	-250.0	250.0	291	0.5399	0.0432	11046				
0.0402	-500.0	500.0	148	0.8099	0.0371	11046	0.0031	10201		
0.0342	-500.0	250.0	219	0.8099	0.0342	11046				
0.0317	250.0	0.0	151	0.8099	0.0317	10201				
0.0302	250.0	-250.0	214	0.8099	0.0302	10201				
0.0279	750.0	-250.0	330	0.8099	0.0253	10201	0.0027	11046		
0.0257	750.0	0.0	28	0.8099	0.0257	10201				



Хром шестивалентний (в перерахунку на трнокие хрому) Карта-ехема H=2.00 м

### -17 -19/09/2024 "АТ «НАЕК «Евергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Код речовини	Найменування речовнии	ГДК (мг/м.куб)
10102-44-0 (301)	Азоту діоксид	0.20000000

Фонові концентрації, які вміщують внески діючих джерел (Частки ГДК) (частки ГДК) (Вихідні рівні забруднення) для речовани : Азоту діоксид. Варіант завдання фону : а.

Коорд. Х поста спостереження	Коорд. Ү поста спостереження	U<2 м/с (штиль)	Швядкість вітру 2 <u<u* Пя</u<u* 	IIIвидкість вітру 2 <u<u* ПнС</u<u* 	IIIвидкість вітру 2 <u<u* C</u<u* 	Шпадкість віпуv 2 <u<u* ПдС</u<u* 	Швядкість вітру 2 <u<u* Пд</u<u* 	Швядкість вітру 2 <u<u* Пд3</u<u* 	Швидкість вітру 2 <u<u* 3</u<u* 	Швяджість вітру 2 <u<u* ПиЗ</u<u* 
0.00	0.00	0.04	0.04	0.04	0.04	0.04	0.04	0.04	0.04	0.04

Фонові концентрації без урахувания внесків діючих джерел (Частки ГДК) (частки ГДК) (Власне фон - верхиє число, вклад - нижнє) для речовнии : Азоту діоксид. Варіант завдання фону : а.

Коорд. Х поста спостереження	Коорд, Ү поста спостереження	U<2 м/с (штиль)	Швядкість вітру 2 <u<u* Пп</u<u* 	Шпилкість вітру 2 <u<u* ПнС</u<u* 	Швилкість вітру 2 <u<u* C</u<u* 	IIInnancierts Birpy 2 <u<u* IIaC</u<u* 	Швилкість вітру 2 <u<u* Пд</u<u* 	Швилкість вітру 2 <u<u* Пд3</u<u* 	Шпилкість вітру 2 <u<u* З</u<u* 	Швядкість вітру 2 <u<u* Пв3</u<u* 
0.00	0.00	0.0400	0.0400	0.0400	0.0400	0.0400	0.0400	0.0400	0.0400	0.0400

	- 18	- 	
19/09/2024	"AT «HAEK «Евергоатом»"	ЕОЛ 2000 [h] v4.0,	Ліценція №133565957

Перелік джерел, у викидах яких є Азоту діяксид

Код джерела Технологічні параметри	10101	10201	11046	11049	11050
Винанд г/с	5.3400	0.0020	0.0020	0.0100	15.7500
Клас небезпечн.	5	5	5	5	5
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мс/м. куб	0.4317	0.0157	0.0157	11.8036	48.1338
ХМ (м)	546.37	86.53	86.53	12.48	201.84
UM[h=2.00м] (м/с)	1.71	0.54	0.54	0.50	5,60
Х Ү Коорд. точеч. початок лін-го, центр симетр. пл-го (м)	330.00 -245.00	475.00 -120.00	-320.00 400.00	-420.00 550.00	-500.00 365.00
Х Ү Коорд. кінця лів-го, дов. і шарнява пл-го(м)	0.00 0.00	0.00 0.00	0.00 0.00	-410.00 695.00	0.00 0.00
Коеф-т рель' ефу	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ПГПС(м. куб/с)	11:4000	2.7400	2.7400	0.2500	44.7000
Шв-ть вахіду ПГПС: м/е	1.4175	9.6908	9.6908	32.0000	3.5571
Діаметр (м)	3.2000	0.6000	0.6000	0.0001	4.0000
Висота (м)	60.0000	14.0000	14.0000	5.0000	8.5000
Температура (С)	118.0000	23.0000	23.0000	250.0000	55.0000
Коеф-т впоряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000
Викид т/р	0.0480	0.0023	0.0023	0.0000	2.7220

Розрахункові концентрації речовини: Азоту діокенд в розрахункових точках та номера джерел, що надають найбільший внесок на высоті 2.00 м

№ розр. точки	Концентр. v точції частки ГДК	Коорд, розр, точки Х	Коорд, розр. точки Ү	Напрямок вітру	Швидвість вітру	Розмір внеску Q0	№ джерела N0	Розмір внеску Q1	Nb лжерела N1	Розмір внеску Q2	№ жкерела N2
11	0.7351	0.0	2700.0	78	2.00	0.6888	11050	0.0056	10101	0.0006	11049
22	0.6574	550.0	2700.0	66	2.00	0.6126	11050	0.0039	10101	0.0009	11049

	- 19	line manager and	
19/09/2024	"АТ «НАЕК «Евергоатом»"	ЕОЛ 2000 [h] v4.0,	Ліцензія №133565957

Точки вайбільших концентрацій речовний Азоту діоксид На розрахун. площадці № 1 та номера джерел, що надяють найбільший внесок на высоті 2.00 м

Концентрації у точці частки ГДК	Коорд. розр. точки Х	Коорд. розр. точки Ү	Напрям. вітру	Швядкість вітру	Розмір внеску Q0	№ джерела N0	Розмір внеску Q1	№ джерепа N1	Розмір внеску Q2	№ джерела N2
15.0694	-500.0	250.0	270	7.00	15.0292	11050	0.0002	11049		
13:5189	-500.0	500.0	90	7.00	13.4789	11050				
8.4434	-750.0	250.0	205	6.8635	8.4032	11050	0.0003	11046	0.0000	11049
8.4433	-250.0	250.0	335	6.8635	8.4033	11050	0.0000	11049	0.0000	11046
8.3881	-750.0	500.0	152	6.8635	8.3074	11050	0.0405	10101	0.0002	10201
8.3473	-250.0	500.0	28	6.8635	8.3073	11050	0.0000	11046	0.0000	11049
7.4590	-500.0	0.0	270	6.8635	7.4179	11050	0.0011	11049	0.0000	11046
7.2377	-500.0	750.0	90	6.8635	7.1977	11050	0.0000	11046	0.0000	10101
6.6197	-750.0	0.0	236	6.8635	6.5793	11050	0.0004	11049	0.0000	11046
6.6193	-250.0	0.0	304	6.8635	6.5793	11050	0.0000	11049	0.0000	11046

-20 -19/09/2024 "АТ «НАЕК «Евергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957



Азоту діокенд Карта-ехема H=2.00 м

### -21 -19/09/2024 "АТ «НАЕК «Енергоатом»" ЕОЛ 2000 [h] v4.0, Ліценія №133565957

Код речовнии	Найменування речовнии	ГДК (мг/м.куб)
7631-86-9 (323)	Кремийо діоксид аморфний (Аеросил-175)	0.02000000

Фонові концентрації, які вміщують внески діючих джерел (Частки ГДК) (частки ГДК) (Вихідні рівні забруднення) для речовини : Кремнію діоксид аморфинй (Аеросил-175). Варіант завдання фону : а.

Коорд. Х поста спостереження	Коорд. Ү поста спостереження	U<2 м/с (штиль)	Швядкість вітру 2⊲U <u* Пв</u* 	Шпилисть вітру 2 <u<u* ПнС</u<u* 	Швидкість вітру 2 <u<u* С</u<u* 	Шпядкість вітру 2 <u<u* ПдС</u<u* 	Швядкість вітру 2 <u<u* Пд</u<u* 	Швядкість вітру 2 <u<u* Пд3</u<u* 	Швидкість вітру 2 <u<u* 3</u<u* 	Швяджість вітру 2 <u<u* Пв3</u<u* 
0.00	0.00	0.40	0.40	0.40	0.40	0.40	0.40	0.40	0.40	0.40

Фонові концентрації без урахування внесків діючих джерел (Частки ГДК) (частки ГДК) (Власне фон - верхне число, вклад - нижне) для речовнин : Кремпію діоксид аморфний (Аеросил-175). Варіант завдання фону : а.

Коорд. Х поста спостереження	Коорд. У поста спостереження	U<2 м/с (штиль)	Швядкість вітру 2 <u<u* Пв</u<u* 	Шпилкість вітру 2 <u<u* ПнС</u<u* 	Швилкість вітру 2 <u<u* C</u<u* 	Шиндкість вітру 2 <u<u* ПдС</u<u* 	Швядкість вітру 2 <u<u* Пд</u<u* 	Шпилкість вітру 2 <u<u* Пл3</u<u* 	Шлилкість вігру 2 <u<u* 3</u<u* 	Швялкість вітру 2 <u<u* Пв3</u<u* 
0.00	0.00	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000

# - 22 -19/09/2024 "AT «НАЕК «Енергоатом»" ЕОЛ 2000 [h] v4.0, Ліценція №133565957

Перелік джерел, у викидах яких є Кремнію діоксид аморфний (Аеросил-175)

Код джерела Технологічні параметри	10201	11046
Викид г/е	0.0013	0.0013
Клас небезпечн.	5	5
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мс/м. куб	0.1018	0.1018
ХМ (м)	86,53	86.53
UM[h=2.00м] (м/с)	0.54	0.54
Х Ү Коорд. точеч. початок пін-го, центр симетр. пл-го (м)	475.00 -120.00	-320.00 400.00
Х Ү Коорд. кінця лін-го, дов. і ширина пл-го(м)	0.00 0.00	0.00 0.00
Коеф-т рель' офу	1.0000	1.0000
Вятрата ПГПС(м. куб/с)	2.7400	2.7400
Шв-ть вихіду ПГПС: м/е	9.6908	9.6908
Діаметр (м)	0.6000	0.6000
Висота (м)	14.0000	14.0000
Температура (С)	23.0000	23.0000
Коеф-т впоряд. осід.	1.0000	1.0000
Викид т/р	0.00089	0.00089

Розрахункові концентрації речовини: Кремнію діоксид аморфний (Аеросил-175) в розрахункових точвах та номера джерел, що надають найбільший внесок на высоті 2.00 м

№ розр. точки	Конпентр. v точші частки ГДК	Коорд. розр. точкн Х	Коорд, розр. точки Ү	Напрямок вітру	Швядвість вітру	Розмір внеску Q0	№ джерела N0	Розмір внеску Q1	№ джерела N1	Розмір внеску Q2	№ джерела N2
11	0.4006	0.0	2700.0	82	7.00	0.0006	11046	0.0000	10201		
22	0.4006	550.0	2700.0	70	7.00	0.0005	11046	0.0000	10201		

### - 23 -19/09/2024 "AT «НАЕК «Енергоятом»" ЕОЛ 2000 [h] ∨4.0, Ліцензія №133565957

Точки найбільших концентрацій речовнни Кремнію діокенд аморфний (Аеросил-175) На розрахув. площадці № 1 та номера джерел, що надають найбільший внесок на высоті 2.00 м

Концентрації у точці частки ГДК	Коорд, розр. точки Х	Коорд. розр. точкя Ү	Напрям. вітру	Швидкість вітру	Розмір внеску Q0	№ джереца N0	Розмір внеску Q1	№ джереля N1	Розмір внеску Q2	№ джерела №2
0.4199	-250.0	500.0	52	0.5399	0.0199	11046				
0.4198	500.0	0.0	82	0.5399	0.0198	10201				
0.4191	500.0	-250.0	277	0.5399	0.0191	10201	0.0000	11046		
0.4169	-250.0	250.0	291	0.5399	0.0169	11046				
0.4157	-500.0	500.0	148	0.8099	0.0145	11046	0.0012	10201		
0.4133	-500.0	250.0	219	0.8099	0.0133	11046				
0.4124	250.0	0.0	151	0.8099	0.0124	10201				
0.4118	250.0	-250.0	214	0.8099	0.0118	10201				
0.4109	750.0	-250.0	330	0.8099	0.0098	10201	0.0011	11046		
0.4100	750.0	0.0	28	0.8099	0.0100	10201				

- 24 - 19/09/2024 "АТ «НАЕК «Евергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957



Кремийо діокенд аморфинй (Аероенл-175) Карта-ехема H=2.00 м

### - 25 -19/09/2024 "АТ «НАЕК «Евергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Код речовини	Найменування речовнии	ГДК (мг/м.куб)
1333-86-4 (328)	Сажа	0.15000000

### Фонові концентрації, які вміщують внески діючих джерел (Частки ГДК) (частки ГДК) (Вихідні рівні забруднення) для речовини : Сажа. Варіант завдання фону : а.

Коорд. Х поста спостереження	Коорд. Ү поста спостереження	U<2 м/с (штиль)	Швядкість вітру 2 <u<u* Пя</u<u* 	Шинджість вітру 2 <u<u* ПнС</u<u* 	Швидкість вітру 2 <u<u* С</u<u* 	Швядкість віру 2 <u<u* ПдС</u<u* 	Швядкість вітру 2 <u<u* Пд</u<u* 	Швидкість вітру 2 <u<u* Пд3</u<u* 	Швиджість вітру 2 <u<u* 3</u<u* 	Швялкість вітру 2 <u<u* ПиЗ</u<u* 
0.00	0.00	0.40	0.40	0.40	0.40	0.40	0.40	0.40	0.40	0.40

Фонові концентрації без урахування внесків діючих джерел (Частки ГДК) (частки ГДК) (Власие фон - верхиє число, вклад - нижиє) для речовшин : Сажа. Варіант завдання фону : а.

Коорд, Х поста спостереження	Коорд, Ү поста спостереження	U<2 м/с (штиль)	Швядвість вітру 2 <u<u* Пи</u<u* 	Шпидкість вітру 2 <u<u* ПнС</u<u* 	Швидкість вітру 2 <u<u* С</u<u* 	Швилисість вітру 2 <u<u* ПдС</u<u* 	Швялкість вітру 2 <u<u* Пд</u<u* 	Швилкість вітру 2 <u<u* Пд3</u<u* 	Швнакість вітру 2 <u<u* 3</u<u* 	Швядкість вітру 2 <u<u* Пи3</u<u* 
0.00	0.00	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000

# - 26 -19/09/2024 — "AT «НАЕК «Еверговтом»" ЕОЛ 2000 [h] v4.0, Ліцевзія №133565957

Перелік джерел, у викидах яких є Сажа

Код джерела Технологічні параметри	11049
Вилад г/с	0.0020
Клас небезпечн.	5
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мс/м. куб	3.1476
ХМ (м)	12.48
UM[h=2.00м] (м/с)	0,50
Х Ү Кооод. точеч. початок лін-го, центр свыетр. пл-го (м)	-420.00 550.00
Х Ү Коорд. кінця лін-го, дов. і шприна пл-го(м)	-410.00 695.00
Коеф-т рель`офу	1.0000
Витрата ПГПС(м. куб/с)	0.2500
Шв-ть вахіду ПГПС: м/с	32.0000
Діаметр (м)	0.0001
Висота (м)	5.0000
Температура (C)	250.0000
Коеф-т впоряд. осід.	1.0000
Викид т/р	0.0000

	Розрахунков: концентрації речовини: Сажа											
в розрахункових точках	та номера джерел,	, що надають найбільши	й внесок на высоті 2.00 м									

№ розр. точки	Конпентр. v точщі часткн ГДК	Коорд. розр. точкн Х	Коорд, розр. точки Ү	Напрямок вітру	Швидкість вітру	Розмір внеску Q0	№ джерела N0	Розмір внеску Q1	№ джерела N1	Розмір внеску Q2	№ джерела N2
11	0.4003	0.0	2700.0	79	0.75	0.0003	11049				
22	0.4003	550.0	2700.0	65	0.75	0.0003	11049				

## - 27 -19/09/2024 "AT «НАЕК «Евергоатом»" ЕОЛ 2000 [b] v4.0, Ліценція №133565957

Точки найбільших концентрацій речовнии Сажа На розрахун. площадці № 1 та номера джерел, що надають найбільший внесок на высоті 2.00 м

Концентрації у точці частки ГДК	Коорд. розр. точки Х	Коорд. розр. точки Ү	Напрям. вітру	Швядкість вітру	Розмір внеску Q0	№ джереца N0	Розмір внеску Q1	№ джерела N1	Розмір внеску Q2	№ джерела N2
0.4150	-500.0	500.0	225	0.50	0.0150	11049				
0.4134	-500.0	750.0	134	0.50	0.0134	11049				
0.4077	-250.0	750.0	38	0.50	0.0077	11049				
0.4076	-250.0	500.0	323	0,50	0.0076	11049				
0.4036	-750.0	500.0	200	0.75	0.0036	11049				
0.4036	-750.0	750.0	159	0.75	0.0036	11049				
0.4035	-500.0	250,0	257	0.75	0.0035	11049				
0.4034	-500.0	1000.0	103	0.50	0.0034	11049				
0.4031	-250.0	250.0	294	0.50	0.0031	11049				
0.4031	-250.0	1000.0	66	0.50	0.0031	11049				

- 28 - 19/09/2024 "АТ «НАЕК «Евергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957



Сажа Карта-ехема H=2.00 м

### - 29 -19/09/2024 "АТ «НАЕК «Евергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Код речовини	Найменування речовнии	ГДК (мг/м.куб)
7446-09-5 (330)	Ангілрид сірчистий	0.50000000

Фонові концентрації, які вміщують внески діючих джерел (Частки ГДК) (частки ГДК) (Вихідні рівні забруднення) для речовини : Ангідрид сірчистий. Варіант завдання фону : а.

Коорд. Х поста спостереження	Коорд. Ү поста спостереження	U<2 м/с (штиль)	Швядкість вітру 2 <u<u* Пя</u<u* 	IIIвидкість вітру 2 <u<u* ПнС</u<u* 	IIIвидкість вітру 2 <u<u* C</u<u* 	Шпадкість віпуv 2 <u<u* ПдС</u<u* 	Швядкість вітру 2 <u<u* Пд</u<u* 	Швядкість вітру 2 <u<u* Пд3</u<u* 	Швидкість вітру 2 <u<u* 3</u<u* 	Швяджість вітру 2 <u<u* ПиЗ</u<u* 
0.00	0.00	0.04	0.04	0.04	0.04	0.04	0.04	0.04	0.04	0.04

Фонові концентрації без урахувания внесків діючих джерел (Частки ГДК) (частки ГДК) (Власне фон - верхиє число, вклад - нижиє) для речовнин : Ангідрид сірчнетий. Варіант завдания фону : а.

Коорд. Х поста спостереження	Коорд. У поста спостереження	U<2 м/с (штиль)	Швядкість вітру 2 <u<u* Пп</u<u* 	Шпидкість вітру 2 <u<u* ПиС</u<u* 	Швидкість вітру 2 <u<u* C</u<u* 	IIImnacierts Bitpy 2 <u<u* HaC</u<u* 	Швялкість вітру 2 <u<u* Пд</u<u* 	Швилкість вітру 2 <u<u* Пл3</u<u* 	Шпнлкість вігру 2 <u<u* 3</u<u* 	Швилкість вітру 2 <u<u* Пи3</u<u* 
0.00	0.00	0.0400	0.0400	0.0400	0.0400	0.0400	0.0400	0.0400	0.0400	0.0400

# - 30 -19/09/2024 "AT «НАЕК «Енергоятом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Перелік джерел, у викидах яких є Ангідрид сірчистий

Код джерела Технологічні параметри	10101	11049	11050
Винид г/с	12.7472	0.0000016	3.2500
Клас небезпечн.	5	5	5
СМ[h=2.00м] (частин ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мс/м. куб	0.4122	0.0008	3.9729
ХМ (м)	546.37	12.48	201,84
UM[h=2.00м] (м/с)	1.71	0.50	5.60
Х Ү Коорд. точеч. початок лін-го, центр симетр. пл-го (м)	330.00 -245.00	-420.00 550.00	-500.00 365.00
Х Ү Коорд. кінця лів-го, дов. і шарина пл-го(м)	0.00 0.00	-410.00 695.00	0.00 0.00
Коеф-т рель' ефу	1.0000	1.0000	1.0000
Витрата ПГПС(м. куб/с)	11.4000	0.2500	44.7000
Шв-ть вихіду ПГПС: м/с	1.4175	32.0000	3.5571
Діаметр (м)	3.2000	0.0001	4.0000
Висота (м)	60:0000	5.0000	8.5000
Температура (С)	118.0000	250.0000	55.0000
Коеф-т впоряд. осід.	1.0000	1.0000	1.0000
Викид т/р	2.6166	0.0000	0.5620

Розрахункові концентрації речовини: Ангідрид сірчнетий в розрахункових точках та номера джерел, що надають найбільший внесок на высоті 2.00 м

№ розр. точки	Концентр. v точції частки ГДК	Коорд, розр. точкн Х	Коорд, розр. точки Ү	Напрямок вітру	Швидвість вітру	Розмір внеску Q0	№ джерела N0	Розмір внеску Q1	Nb джерела N1	Розмір внеску Q2	№ джерела N2
11	0.1026	0.0	2700.0	86	2.00	0.0371	11050	0.0255	10101	0.0000	11049
22	0.0942	550.0	2700.0	66	2.00	0.0506	11050	0.0036	10101	0.0000	11049

## - 31 -19/09/2024 "AT «НАЕК «Еверговтом»" ЕОЛ 2000 [h] v4.0, Ліцевція №133565957

Точки найбільших концентрацій речовний Ангідрид сірчистий На розрахун. площадці № 1 та номера джерел, що надають найбільший внесок на высоті 2.00 м

Концентрації у точці частки ГДК	Коорд. розр. точки Х	Коорд. розр. точки Ү	Напрям. вітру	Швядкість вітру	Розмір внеску Q0	№ джерела N0	Розмір внеску Q1	№ джереля N1	Розмір внеску Q2	№ джерела N2
1.2683	-500.0	250.0	269	7.00	1.2283	11050	0.0000	11049		
1.0928	-500.0	500.0	88	7.00	1.0528	11050				
0.7268	-750.0	250.0	205	7.00	0.6868	11050	0.0000	11049		
0.7215	-250.0	500.0	28	7.00	0.6815	11050	0.0000	11049		
0.6331	-500.0	750.0	90	7,00	0.5931	11050	0.0000	10101	0.0000	11049
0.6054	-750.0	500.0	155	5.2362	0.5472	11050	0.0182	10101	0.0000	11049
0.5669	-250.0	250,0	340	7.00	0.5269	11050	0.0000	11049		
0.5615	-750.0	0.0	237	7.00	0.5215	11050	0.0000	11049		
0.5445	-500.0	0.0	274	7.00	0.5045	11050	0.0000	11049		
0.5264	-250.0	0.0	307	7.00	0.4864	11050	0.0000	11049		

- 32 - 19/09/2024 "АТ «НАЕК «Евергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957



Ангідрид сірчистий Карта-схема H=2.00 м

## - 33 - 19/09/2024 "АТ «НАЕК «Евергоатом»" ЕОЛ 2000 [h] v4.0, Ліпетія №133565957

Код речовнии	Найменування речовнии	ГДК (мг/м.куб)
630-08-0 (337)	Вуглено оксид	5.00000000

Фонові концентрації, які вміщують внески діючих джерел (Частки ГДК) (частки ГДК) (Вихідні рівні забруднення) для речовини : Вуглецю оксид. Варіант завдання фону : а.

Коорд. Х поста спостереження	Коорд. Ү поста спостереження	U<2 м/с (штиль)	Швядкість вітру 2 <u<u* Пя</u<u* 	IIIвидкість вітру 2 <u<u* ПнС</u<u* 	IIInnakiers Bitpy 2 <u<u* C</u<u* 	Шпадкість віпуv 2 <u<u* ПдС</u<u* 	Швядкість вітру 2 <u<u* Пд</u<u* 	Швядкість вітру 2 <u<u* Пд3</u<u* 	Шиндкість вітру 2 <u<u* 3</u<u* 	Швяджість вітру 2 <u<u* Пи3</u<u* 
0.00	0.00	0.08	0.08	0.08	0.08	0.08	0.08	0.08	0.08	0.08

Фонові концентрації без урахування внесків діючих джерел (Частки ГДК) (частки ГДК) (Власне фон - верхиє число, вклад - нижнє) для речовник : Вутлецю оксид. Варіант завдання фону : а.

Коорд. Х поста спостереження	Коорд, Ү поста спостереження	U<2 м/с (штиль)	Швядкість вітру 2 <u<u* Пп</u<u* 	Шпилкість вітру 2 <u<u* ПнС</u<u* 	Швилкість вітру 2 <u<u* C</u<u* 	IIImmediens simpy 2 <u<u* IImC</u<u* 	Швилкість вітру 2 <u<u* Пд</u<u* 	Швилкість вітру 2 <u<u* Пд3</u<u* 	Швилкість вітру 2 <u<u* 3</u<u* 	Швядкість вітру 2 <u<u* Пв3</u<u* 
0.00	0.00	0.0800	0.0800	0.0800	0.0800	0.0800	0.0800	0.0800	0.0800	0.0800

	- 34	Incorrente ac	
19/09/2024	"AT «HAEK «Eneproarom»"	ЕОЛ 2000 [h] v4.0,	Ліцетія №133565957

Перелік джерел, у викидах яких є Вуглецю оксяд

Код джерела Технологічні параметри	10101	10201	11046	11049	11050
Винанд г/е	1.3781	0.0051	0.0051	0.0160	1.3000
Клас небезпечн.	5	5	5	5	5
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мс/м. куб	0.0045	0.0016	0.0016	0.7554	0.1589
ХМ (м)	546.37	86.53	86.53	12.48	201.84
UM[h=2.00м] (м/с)	1.71	0.54	0.54	0.50	5.60
Х Ү Коорд. точеч. початок лін-го, центр симетр. пл-го (м)	330.00 -245.00	475.00 -120.00	-320.00 400.00	-420.00 550.00	-500.00 365.00
Х Ү Коорд. кінпя лів-го, дов. і шприна пл-го(м)	0.00 0.00	0.00 0.00	0.00 0.00	-410.00 695.00	0.00 0.00
Коеф-т рель' өфу	1.0000	1.0000	1.0000	1.0000	1.0000
Внтрата ПГПС(м. куб/с)	11.4000	2.7400	2.7400	0.2500	44.7000
Шв-ть вихіду ШГПС: м/е	1.4175	9.6908	9.6908	32.0000	3.5571
Діаметр (м)	3.2000	0.6000	0.6000	0.0001	4.0000
Висота (м)	60:0000	14.0000	14.0000	5.0000	8.5000
Температура (C)	118.0000	23.0000	23.0000	250.0000	55.0000
Коеф-т впоряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000
Викид т/р	0.0291	0.0106	0.0106	0.0000	0.2250

Розрахункові концентрації речовини: Вуглецю оксид в розрахункових точках та номера джерел, що надають найбільший внесок на высоті 2.00 м

№ розр. точки	Концентр. у точці частки ГДК	Коорд, розр, точкн Х	Коорд. розр. точки Ү	Напоямок вітру	Швидвість вітру	Розмір внеску Q0	№ джерела N0	Розмір внеску Q1	Nb джерела N1	Розмір внеску Q2	N⊵ ,∞kepena N2
11	0.0830	0.0	2700.0	80	1.3856	0.0028	11050	0.0001	10101	0.0000	11049
22	0.0827	550.0	2700.0	69	1.3856	0.0025	11050	0.0001	10101	0.0001	11049

	- 35	e The second second second	
19/09/2024	"АТ «НАЕК «Евергоатом»"	ЕОЛ 2000 [h] v4.0,	Ліцетія №133565957

Точки найбільших концентрацій речовини Вутлецю оксид На розракун. площадці № 1 та номера джерел, що надають найбільший внесок на высоті 2.00 м

Концентрації у точці частки ГДК	Коорд. розр. точки Х	Коорд. розр. точки Ү	Напрям. вітру	Швядкість вітру	Розмір внеску Q0	№ джерела N0	Розмір внеску Q1	№ джереля N1	Розмір внеску Q2	№ джерела N2
0.1232	-500.0	250.0	273	7.00	0.0432	11050	0.0000	11049		
0.1105	-500.0	500.0	85	7.00	0.0305	11050				
0.1076	-250.0	250.0	335	7.00	0.0276	11050	0.0000	11049	0.0000	11046
0.1066	-250.0	500.0	27	7.00	0.0266	11050	0.0000	11046	0.0000	11049
0.1041	-500.0	0.0	269	7,00	0.0240	11050	0.0001	11049	0.0000	11046
0.1029	-750.0	250.0	201	7.00	0.0228	11050	0.0000	11046	0.0000	11049
0.1002	-750.0	500,0	156	7.00	0.0200	11050	0.0001	10101	0.0000	10201
0.1000	-250.0	750.0	59	7.00	0.0200	11050	0.0000	11046	0.0000	11049
0.1000	-750.0	750.0	125	7.00	0.0199	11050	0.0001	10101	0.0000	11046
0.0989	-500.0	750.0	94	7.00	0.0189	11050	0.0000	11046	0.0000	11049

- 36 - 19/09/2024 "АТ «НАЕК «Еверговтом»" ЕОЛ 2000 [h] v4.0, Ліцетія №133565957



Вуглецю оксил Карта-схема H=2.00 м

### - 37 - 19/09/2024 "АТ «НАЕК «Енергоатом»" ЕОЛ 2000 [h] v4.0, Ліценія №133565957

Код речовини	Найменування речовнин	ГДК (мг/м.куб)
(342)	Фтористі сполуки газоподібні (фтористий	0.02000000

Фонові концентрації, які вміщують внески діючих джерел (Частки ГДК) (частки ГДК) (Вихілні рівні забруднення) для речовини : Фтористі сполуки газоподібні (фтористий водень,4-фтористий кремній) у перерахунку на фтор. Варіант завдання фону : а.

Коорд. Х поста спостереження	Коорд. Ү поста спостереження	U<2 м/с (штиль)	Швядкість вітру 2 <u<u* Пв</u<u* 	Шинджість вітру 2 <u<u* ПнС</u<u* 	Швядкість вітру 2 <u<u* С</u<u* 	Швядкість вітру 2 <u<u* ПдС</u<u* 	Швядкість вітру 2 <u<u* Пд</u<u* 	Шандкість вітру 2 <u<u* Пд3</u<u* 	IIInnukiens Birpy 2 <u<u* 3</u<u* 	Швядкість вітру 2 <u<u* Пн3</u<u* 
0.00	0.00	0.40	0.40	0.40	0.40	0.40	0.40	0.40	0.40	0.40

Фонові концентрації без урахування внесків діючих джерел (Частки ГДК) (частки ГДК) (Власне фон - верхнє число, вклад - нижнє) для речовини : Фтористі сполуки газоподібні (фтористий водень,4-фтористий кремній) у перерахунку на фтор. Варіант завдання фону : а.

Коорд. Х поста спостережения	Коорд. Ү поста спостереження	U<2 м/с (птиль)	Швядвість вітру 2 <u<u* Пп</u<u* 	Шпилкість вітру 2 <u<u* ПнС</u<u* 	Швидкість вітру 2 <u<u* С</u<u* 	Швилисість вітру 2 <u<u* ПдС</u<u* 	Швялкість вітру 2 <u<u* Пд</u<u* 	Швилкість вітру 2 <u<u* Пд3</u<u* 	Шанлкість вітру 2 <u<u* З</u<u* 	Швядкість вітру 2 <u<u* Пи3</u<u* 
0.00	0.00	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000

## - 38 -19/09/2024 "АТ «НАЕК «Енергоатом»" ЕОЛ 2000 [b] v4.0, Ліцензія №133565957

Перелік джерел, у викидах яких є Фтористі сполуки газоподібні (фтористий водень,4-фтористий кремній) у перерахунку на фтор

Код джерела Технологічні параметри	10201	11046		
Винанд г/с	0.0012	0.0012		
Клас небезпечн.	5	5		
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мс/м. куб	0.0939	0.0939		
ХМ (м)	86.53	86.53		
UM[b=2.00 _M ] (м/c)	0.54	0.54		
Х Ү Коорд. точеч. початок лін-го, пентр симетр. пл-го (м)	475.00 -120.00	-320.00 400.00		
Х Ү Коорд. кінця лін-го, дов. і ширина пл-го(м)	0.00 0.00	0.00 0.00		
Коеф-т рель`ефу	1.0000	1.0000		
Внтрата ПГПС(м. куб/с)	2.7400	2.7400		
Шв-ть вихіду ПГПС: м/с	9.6908	9.6908		
Діаметр (м)	0.6000	0.6000		
Висота (м)	14.0000	14.0000		
Температура (С)	23.0000	23.0000		
Коеф-т впоряд. осід.	1.0000	1.0000		
Викид т/р	0.0020	0.0020		

Розрахункові концентрації речовання: Фторасті сполуки газоподібні (фторастий водень, 4-фторастий кремній) у перерахунку на фтор в розрахункових точках та номера джерел, що надають найбільший внесок на высоті 2.00 м

№ розр. точки	Концентр. у точці частки ГДК	Коорд. розр. точки Х	Коорд. розр. точки Ү	Напрямок вітру	Швидність вітру	Розмір внеску Q0	№ джерела N0	Розмір внеску Q1	№ джерела N1	Розмір внеску Q2	№ джерела N2
11	0.4006	0.0	2700.0	82	7.00	0.0006	11046	0.0000	10201		
22	0.4005	550.0	2700.0	70	7.00	0.0005	11046	0.0000	10201		

### - 39 -19/09/2024 "АТ «НАЕК «Еверговтом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Точки найбільших концентрацій речовини Фтористі сполуки тазоподібні (фтористий водень.4-фтористий кремній) у перерахунку на фтор На розрахун. площадці № 1 та номера джерел, що надають найбільший внесок на высоті 2.00 м

Концентрації у точці частки ГДК	Коорд. розр. точки Х	Коорд. розр. точки Ү	Напрям. вітру	Швядкість вітру	Розмір внеску Q0	№ джереца N0	Розмір внеску Q1	№ джерела N1	Розмір внеску Q2	№ джерела N2
0.4184	-250.0	500.0	52	0.5399	0.0184	11046				
0.4183	500.0	0.0	82	0.5399	0.0183	10201				
0.4176	500.0	-250.0	277	0.5399	0.0176	10201	0.0000	11046		
0.4156	-250.0	250.0	291	0.5399	0.0156	11046				
0.4145	-500.0	500.0	148	0.8099	0.0134	11046	0.0011	10201		
0.4123	-500.0	250.0	219	0.8099	0.0123	11046				
0.4114	250.0	0.0	151	0.8099	0.0114	10201				
0.4109	250.0	-250.0	214	0.8099	0.0109	10201				
0.4101	750.0	-250.0	330	0.8099	0.0091	10201	0.0010	11046		
0.4093	750.0	0.0	28	0.8099	0.0093	10201				





### -41 -19/09/2024 "АТ «НАЕК «Енергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Код речовнин	Найменування речовнин	ГДК (мг/м.куб)
7681-49-4 (343)	Фтористі сполуки добре розчиний неорганічні	0.03000000

Фонові концентрації, які вміщують внески діючих джерел (Частки ГДК) (частки ГДК) (Вихідні рівні забруднення), для речовини : Фтористі сполуки добре розчинні неорганічні (фторид і гексафторсилікат натрію). Варіант завдання фону : а.

Коорд. Х поста спостереження	Коорд. Ү поста спостереження	U<2 м/с (штиль)	Швядкість вітру 2 <u<u* Пя</u<u* 	Швидкість вітру 2 <u<u* ПнС</u<u* 	IIIвидкість вітру 2 <u<u* C</u<u* 	Швядкість віру 2 <u<u* ПдС</u<u* 	Швядкість вітру 2 <u<u* Пд</u<u* 	IIIвидкість вітру 2 <u<u* Пд3</u<u* 	IIInnakiens Birpy 2 <u<u* 3</u<u* 	Швяджість вітру 2 <u<u* Пи3</u<u* 
0.00	0.00	0.40	0.40	0.40	0.40	0.40	0.40	0.40	0.40	0.40

Фонові концентрації без урахування внесків діючих джерел (Частки ГДК) (частки ГДК) (Власне фон - верхне число, вклад - вняжне) для речовнин : Фтористі сполуки добре розчиниі неорганічні (фторид і гексафторсилікат натрію). Варіант завдання фону : а.

Коорд. Х поста спостереження	Коорд. У поста спостереження	U<2 м/с (штиль)	Швидкість вітру 2 <u<u* Пп</u<u* 	Шпиджість вітру 2 <u<u* ПиС</u<u* 	Швидкість вітру 2 <u<u* C</u<u* 	IIImmedicers simpy 2 <u<u* HaC</u<u* 	Швялкість вітру 2 <u<u* Пд</u<u* 	Швилкість вітру 2 <u<u* Пл3</u<u* 	Шпилкість sirpv 2 <u<u* 3</u<u* 	Швилкість вітру 2 <u<u* Пи3</u<u* 
0.00	0.00	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000

# - 42 -19/09/2024 "АТ «НАЕК «Енергоатом»" ЕОЛ 2000 [b] v4.0, Ліцензія №133565957

Перелік джерел, <br/>у викидах яких  $\epsilon$ Фтористі сполуки добре розчинні неорганічні (ф<br/>торид і гексафторсилію<br/>ят натрію)

Код джерела Технологічні параметри	10201	11046
Винанд т/е	0.0013	0.0013
Клас небезпечн.	5	5
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мс/м. куб	0.0678	0.0678
ХМ (м)	86.53	86.53
UM[h=2.00м] (м/с)	0.54	0.54
Х Ү Коорд. точеч. початок лін-го, центр симетр. шл-го (м)	475.00 -120.00	-320.00 400.00
Х Ү Коорд. кінця лін-го, дов. і шприна пл-го(м)	0.00 0.00	0.00 0.00
Коеф-т рель' офу	1.0000	1.0000
Внтрата ПГПС(м. куб/с)	2.7400	2.7400
Шв-ть вихіду ПГПС: м/е	9.6908	9.6908
Діаметр (м)	0.6000	0.6000
Висота (м)	14.0000	14.0000
Температура (С)	23.0000	23.0000
Коеф-т впоряд. осід.	1.0000	1.0000
Викид т/р	0.0038	0.0038

Розрахункові концентрації речовини: Фтористі сполуки добре розчинні неорганічні (фторид і гексафторсилікат натрію) в розрахункових точках та вомера джерел, що надають найбільший внесок на высоті 2.00 м

№ розр. точки	Концентр. v точці частки ГДК	Коорд. розр. точкн Х	Коорд, розр. точки Ү	Напрямок вітру	Швядвість вітру	Розмір внеску Q0	№ джерела N0	Розмір внеску Q1	Nb джерела N1	Розмір внеску Q2	№ джерела N2
11	0.4004	0.0	2700.0	82	7.00	0.0004	11046	0.0000	10201		
22	0.4004	550.0	2700.0	70	7.00	0.0004	11046	0.0000	10201		

### -43 -19/09/2024 "АТ «НАЕК «Еверговтом»" ЕОЛ 2000 [h] v4.0, Ліцевзія №133565957

Точки найбільших концентрацій речовини Фторнеті сполуки добре розчиниі неорганічні (фторид і гексафторенлікат натрію) На розрахун. площадці № 1 та номера джерел, що надають найбільший внесок на высоті 2.00 м

Концентрації у точці частки ГДК	Коорд. розр. точки Х	Коорд. розр. точки Ү	Напрям. вітру	Швидкість вітру	Розмір внеску Q0	№ джерена N0	Розмір внеску Q1	№ джереля N1	Розмір внеску Q2	№ джерела N2
0.4133	-250.0	500.0	52	0.5399	0.0133	11046				
0.4132	500.0	0.0	82	0.5399	0.0132	10201				
0.4127	500.0	-250.0	277	0.5399	0.0127	10201	0.0000	11046		
0.4112	-250.0	250.0	291	0.5399	0.0112	11046				
0.4105	-500.0	500.0	148	0.8099	0.0096	11046	0.0008	10201		
0.4089	-500.0	250.0	219	0.8099	0.0089	11046				
0.4083	250.0	0.0	151	0.8099	0.0083	10201				
0.4079	250.0	-250.0	214	0.8099	0.0079	10201				
0.4073	750.0	-250.0	330	0.8099	0.0066	10201	0.0007	11046		
0.4067	750.0	0.0	28	0.8099	0.0067	10201				

Фтористі сполуки добре розчинні неорганічні (фторид і гексафторенлікат натрію) Карта-схема H=2.00 м



### -45 -19/09/2024 "АТ «НАЕК «Евергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Код речовнии	Найменування речовнин	ГДК (мг/м.куб)
(344)	Фтористі сполуки погано розчинні неорганічні	0.20000000

Фонові концентрації, які вміщують внески діючих джерел (Частки ГДК) (частки ГДК) (Вихідні рівні забруднення) для речовини : Фтористі сполуки погано розчинні неорганічні (фторид алюмінію, гексафторалюмінат натрію). Варіант завдання фону : а.

Коорд. Х поста спостереження	Коорд. Ү поста спостереження	U<2 м/с (штиль)	Швядкість вітру 2 <u<u* Пя</u<u* 	Швидкість вітру 2 <u<u* ПнС</u<u* 	IIIвядкість вітру 2 <u<u* C</u<u* 	Швядкість вітру 2 <u<u* ПдС</u<u* 	Швядкість вітру 2 <u<u* Пд</u<u* 	Швядкість вітру 2 <u<u* Пд3</u<u* 	IIIBRETRICTS Bitpy 2 <u<u* 3</u<u* 	Швядкість вітру 2 <u<u* Пн3</u<u* 
0.00	0.00	0.40	0.40	0.40	0.40	0.40	0.40	0.40	0.40	0.40

Фонові концентрації без урахування внесків діючих джерел (Частки ГДК) (частки ГДК) (Власне фон - верхне число, вклад - нижне) для речовнин : Фторнсті сполуки погано розчинні неорганічні (фторид апомінію, гексафторадюмінат натрію). Варіант завдання фону : а.

Коорд. Х поста спостереження	Коорд. Ү поста спостереження	U<2 м/с (птиль)	Швидвість вітру 2 <u<u* Пи</u<u* 	Шпилкість вітру 2 <u<u* ПнС</u<u* 	Швидкість вітру 2 <u<u* С</u<u* 	Швилисість вітру 2 <u<u* ПдС</u<u* 	Швялкість вітру 2 <u<u* Пд</u<u* 	Швилкість вітру 2 <u<u* Пд3</u<u* 	Шанлкість вітру 2 <u<u* З</u<u* 	Швядкість вітру 2 <u<u* Пи3</u<u* 
0.00	0.00	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000

### - 46 -"AT «НАЕК «Енергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957 19/09/2024

Перелік джерел, <br/>у викидах яких є Фтористі сполуки погано розчиниї неорганічні (фторид алюмінію, гексафторалюмінат натрію)

Код джерела Технологічні параметри	10201	11046
Винанд т/с	0.0008	0.0008
Клас небезпечн.	5	5
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мс/м. куб	0.0063	0.0063
ХМ (м)	86.53	86.53
UM[h=2.00 _M ] (м/c)	0.54	0.54
Х Ү Коорд. точеч. початок лін-го, центр сяметр. пл-го (м)	475.00 -120.00	-320.00 400.00
Х Ү Коорд. кінця лін-го, дов. і ципряна пл-го(м)	0.00 0.00	0.00 0.00
Коеф-т рель`ефу	1.0000	1.0000
Внтрата ПГПС(м. куб/с)	2.7400	2.7400
Шв-ть вихіду ШПС: м/с	9.6908	9.6908
Діаметр (м)	0.6000	0.6000
Висота (м)	14.0000	14.0000
Температура (С)	23.0000	23.0000
Коеф-т впоряд. осід.	1.0000	1.0000
Викид т/р	0.0022	0.0022

Розрахункові концентрації речовини: Фторасті сполуки погано розчинні неорганічні (фторид алюмінію, гексафторалюмінат натрію) в розрахункових точках та номера джерел, що надають найбільший внесок на высоті 2.00 м

№ розр. точки	Концентр. у точщі частки ГДК	Коорд. розр. точкн Х	Коорд, розр. точки Ү	Напрямок вітру	Швядвість вітру	Розмір внеску Q0	№ джерела N0	Розмір внеску Q1	№ джерела N1	Розмір внеску Q2	№ джерела N2
11	0.4000	0.0	2700.0	82	7.00	0.0000	11046	0.0000	10201		
22	0.4000	550.0	2700.0	70	7.00	0.0000	11046	0.0000	10201		

### - 47 -19/09/2024 "АТ «НАЕК «Евергоатом»" ЕОЛ 2000 [h] v4.0, Ліцевлія №133565957

Точки найбільших концентрацій речовнин Фтористі сполуки погано розчинні неорганічні (фторид алюмінію, гексафторалюмінат натрію) На розракун. площадці № 1 та номера джерел, що надають найбільший внесок на высоті 2.00 м

Концентрації у точці частки ГДК	Коорд. розр. точки Х	Коорд. розр. точки Ү	Напрям. вітру	Швядкість вітру	Розмір внеску Q0	№ джереца N0	Розмір внеску Q1	№ джерела N1	Розмір внеску Q2	№ джерела N2
0.4012	-250.0	500.0	52	0.5399	0.0012	11046				
0.4012	500.0	0.0	82	0.5399	0.0012	10201				
0.4012	500.0	-250.0	277	0.5399	0.0012	10201	0.0000	11046		
0.4010	-250.0	250.0	291	0.5399	0.0010	11046				
0.4010	-500.0	500.0	148	0.8099	0.0009	11046	0.0001	10201		
0.4008	-500.0	250.0	219	0.8099	0.0008	11046				
0.4008	250.0	0.0	151	0.8099	0.0008	10201				
0.4007	250.0	-250.0	214	0.8099	0.0007	10201				
0.4007	750.0	-250.0	330	0.8099	0.0006	10201	0.0001	11046		
0.4006	750.0	0.0	28	0.8099	0.0006	10201				
#### Фтористі сполуки погано розчиний неорганічні (фторид алюмінію, гексафторалюмінат натрію) Карта-схема H=2.00 м



## -49 -19/09/2024 "АТ «НАЕК «Еверговтом»" ЕОЛ 2000 [b] v4.0, Ліцевзія №133565957

Код резовнии	Найменувания речовини	ГДК (мг/м.куб)
(2735)	Масло мінеральне нафтове (веретенне, машинне,	0.05000000

Опис фона для речовини : Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)

Опне фону відсутній чи не враховувалось у розрахунках.

Перелік джерел, у викидах яких є											
Масло мінеральне нафтове (веретенне, машянне, циліндрове	i in.)										

Код джерела Технологічні параметри	10305	10306	10307	10308	10309	10310	10311	10312	10313
Викид г/с	0.00195	0.00195	0.0042	0.0049	0.00029	0.00029	0.0042	0.00073	0.00195
Клас небезпечн.	5	5	5	5	5	5	5	5	5
СМ[h=2.00м] (часткн ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мс/м. куб	0.0018	0.0018	0.0038	0.0045	0.0003	0.0003	0.0038	0.0007	0.0018
ХМ (м)	286.19	286.19	286.19	286.19	286.19	286.19	286.19	286.19	286.19
UM[h=2.00st] (st/c)	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50
Х Ү Коорд. точеч. початок лін-го, центр симетр. пл-го (м)	195.00 50.00	200.00 55.00	180.00 5.00	180.00 -5.00	170.00 -40.00	170.00 -30.00	170.00 -20.00	210.00 0.00	40.00 95.00
Х Ү Коорд, кінця лія-го, дов. і шарана пл-го(м)	0.00 0.00	0.00 0.00	0.00 0.00	0.00 0.00	0.00 0.00	0.00 0.00	0.00 0.00	0.00 0.00	0.00 0.00
Коеф-т рель' өфу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Ватрата ПГПС(м. куб/с)	0.7500	0.7500	1.6100	0.1900	0.1100	0.1100	0.1100	0.2800	0.7500
Шв-ть вихіду ПГПС: м/с	78.9198	78.9198	91.1074	37.7993	19.3850	19.3850	6.2247	15.8448	78.9198
Діаметр (м)	0.1100	0.1100	0.1500	0.0800	0.0850	0.0850	0.1500	0.1500	0.1100
Висота (м)	50.0000	50.0000	50.0000	50.0000	50.0000	50.0000	50.0000	50.0000	50.0000
Температура (C)	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000
Коеф-т впоряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Викид т/р	0.0615	0.0615	0.1325	0.0155	0.0091	0.0091	0.1325	0.0230	0.0615

Код джерела Технологічні параметри	10314	10315	10316	10317	10318	10319	11009	11010	11011
Викид г/с	0.00195	0.0042	0.00029	0.00029	0.00029	0.0054	0.0068	0.0068	0.0068
Клас небезпечн.	3	5	5	5	3	:5	2 <b>3</b> 5	5	5
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мг/м. куб	0.0018	0.0038	0.0003	0.0003	0.0003	0.0049	0.0069	0.0069	0.0069 :
ХМ (м)	286.19	286.19	286.19	286.19	286.19	286.19	274.74	274.74	274.74
UM[h=2.00m] (m/e)	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50
Х Ү Коорд. точеч. початок лін-го, центр симетр. пл-го (м)	45.00 105.00	30.00 55.00	15.00 15.00	20.00 20.00	20.00 30.00	55.00 55.00	-400.00 315.00	-395.00 310.00	-390.00 305.00
Х Ү Коорд. кінця лін-го, дов. і шперана пл-го(м)	0.00 0.00	0.00 0.00	0.00 0.00	0.00 0.00	0.00 0.00	0.00 0.00	0.00 0.00	0.00 0.00	0.00 0.00
Коеф-т рель' ефу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ПГПС(м. куб/с)	0.7500	1.6100	0.1100	0.1100	0.1100	2.0800	2.6400	2.6400	2.6400
Шв-ть вихіду ПГПС: м/с	78.9198	91.1074	19.3850	19.3850	6.2247	117.7039	0.5836	0.5836	0.5836
Діаметр (м)	0.1100	0.1500	0.0850	0.0850	0.1500	0.1500	2,4000	2.4000	2.4000
Висота (м)	50.0000	50.0000	50.0000	50.0000	50.0000	50,0000	48.0000	48.0000	48.0000
Температура (C)	23.0000	23.0000	23.0000	23.0000	23,0000	23.0000	23:0000	23.0000	23.0000
Коеф-т впоряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Вакад т/р	0.0615	0.1325	0.0091	0.0091	0.0091	0.1703	0.2140	0.2140	0.2140

- 50 -19/09/2024 "АТ «НАЕК «Еверговтом»" ЕОЛ 2000 [b] v4.0, Ліцевнія №133565957

Код джерела Технологічні параметри	11012	11013	11014	11015	11016	11017	11018	11019	11020
Викид г/с	0.0068	0.0068	0.0068	0.0068	0.0068	0.0068	0.0068	0.0068	0.0068
Клас небезпечн.	5	3	5	5	5	5	5	5	5
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мс/м. куб	0.0069	0.0069	0.0069	0.0069	0.0069	0.0069	0.0069	0.0069	0.0069
ХМ (м)	274.74	274.74	274.74	274.74	274.74	274.74	274.74	274.74	274.74
UM[h=2.00m] (m/e)	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50
Х Ү Коорд. точеч, початок лін-го, центр саметр, пл-го (м)	-385.00 300.00	-380.00 295.00	-405.00 320.00	-410.00 315.00	-415.00 310.00	-420.00 305.00	-425.00 300.00	-430.00 295.00	-435.0 290.00
Х Ү Коорд. кінця лін-го, дов. і шпряна пл-го(м)	0.00 0.00	0.00 0.00							
Коеф-т рель'ефу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ПГПС(м. куб/с)	2.6400	2.6400	2.6400	2,6400	2.6400	2.6400	2.6400	2.6400	2.6400
Шв-ть вихіду ШГПС: м/с	0.5836	0.5836	0.5836	0.5836	0.5836	0.5836	0.5836	0.5836	0.5830
Діаметр (м)	2.4000	2.4000	2.4000	2.4000	2.4000	2.4000	2.4000	2.4000	2.4000
Висота (м)	48.0000	48.0000	48.0000	48.0000	48.0000	48.0000	48.0000	48.0000	48.000
Температура (C)	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.000
Коеф-т впоряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.000
Викид т/р	0.2140	0.2140	0.2140	0.2140	0.2140	0.2140	0.2140	0.2140	0.2140

- 51 -19/09/2024 "АТ «НАЕК «Еверговтом»" ЕОЛ 2000 [b] v4.0, Ліцевнія №133565957

Код джерела Технологічні параметри	11021	11022	11023	11024	11025	11026	11027	11028	11029
Виюнд г/с	0.0068	0.0068	0.0068	0.0068	0.0068	0.0068	0.0068	0.0068	0.0068
Клас небезпечн.	5	3	5	5	5	5	5	5	5
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/M[h=2.00м] мг/м. куб	0.0069	0.0069	0.0069	0,0069 :	0.0069	0.0069	0.0069	0.0069	0.0069
ХМ (м)	274.74	274.74	274.74	274.74	274.74	274.74	274.74	274.74	274,74
UM[h=2.00m] (m/c)	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50
Х Ү Коорд. точеч. початок лін-го, дентр симетр. пл-го (м)	-440.00 285.00	-445.00 280.00	-410.00 325.00	-415.00 320.00	-420.00 315.00	-425.00 310.00	-430.00 305.00	-561.00 390.00	-556.00 385.00
Х Ү Коорд. кінця лія-го, дов. і шпряна пл-го(м)	0.00 0.00								
Коеф-т рель'ефу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ПГПС(м. куб/с)	2.6400	2.6400	2.6400	2,6400	2.6400	2.6400	2.6400	2.6400	2.6400
Шв-ть вихіду ШГПС: м/с	0.5836	0.5836	0.5836	0.5836	0.5836	0.5836	0.5836	0.5836	0.5836
Діаметр (м)	2.4000	2.4000	2.4000	2.4000	2.4000	2.4000	2.4000	2.4000	2.4000
Висота (м)	48.0000	48.0000	48.0000	48.0000	48.0000	48.0000	48.0000	48.0000	48.000
Температура (C)	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.000
Коеф-т впоряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Викид т/р	0.2140	0.2140	0.2140	0.2140	0.2140	0.2140	0.2140	0.2140	0.2140

- 52 -19/09/2024 "AT «НАЕК «Еверговтом»" ЕОЛ 2000 [h] v4.0, Ліцевзія №133565957

Код джерела Технологічні параметри	11030	11031	11032	11033	11034	11035	11036	11037	11038
Викид r/c	0.0068	0.0068	0.0068	0.0068	0.0068	0.0068	0.0068	0.0068	0.0068
Клас небезпечн.	5	3	5	5	5	5	5	5	5
СМ[h=2.00м] (частин ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мс/м. куб	0.0069	0.0069	0.0069	0,0069	0.0069	0.0069	0.0069	0.0069	0.0069
ХМ (м)	274.74	274.74	274.74	274.74	274.74	274.74	274.74	274.74	274.74
UM[h=2.00m] (m/c)	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50
Х Ү Коорд. точеч. початок лін-го, центр симетр. пл-го (м)	-551.00 380.00	-546.00 375.00	-541.00 370.00	-566.00 395.00	-571.00 390.00	-576.00 385.00	-581.00 380.00	-586.00 375.00	-591.00 370.00
Х Ү Коорд. кінця лін-го, дов. і шарана пл-го(м)	0.00 0.00								
Коеф-т рель'ефу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ПГПС(м. куб/с)	2.6400	2.6400	2.6400	2.6400	2.6400	2.6400	2.6400	2.6400	2.6400
Шв-ть вихіду ПГПС: м/с	0.5836	0.5836	0.5836	0.5836	0.5836	0.5836	0.5836	0.5836	0.5830
Діаметр (м)	2.4000	2.4000	2.4000	2.4000	2.4000	2.4000	2.4000	2.4000	2.4000
Висотя (м)	48.0000	48.0000	48.0000	48.0000	48.0000	48.0000	48.0000	48.0000	48.000
Температура (С)	23.0000	23.0000	23.0000	23.0000	23,0000	23.0000	23.0000	23.0000	23.000
Коеф-т впоряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Викид т/р	0.2140	0.2140	0.2140	0.2140	0.2140	0.2140	0.2140	0.2140	0.2140

- 53 -19/09/2024 "АТ «НАЕК «Еверговтом»" ЕОЛ 2000 [b] v4.0, Ліцевнія №133565957

Код джерела Технологічні параметри	11039	11040	11041	11042	11043	11044	11045	13003	13004
Викид г/с	0.0068	0.0068	0.0068	0.0068	0.0068	0.0068	0.0068	0.0020	0.0042
Клас небезпечн.	5	3	5	5	5	5	5	5	5
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мс/м. куб	0.0069	0.0069	0.0069	0.0069	0.0069	0.0069	0.0069	0.0018	0.0038
ХМ (м)	274.74	274.74	274.74	274.74	274.74	274.74	274.74	286.19	286.19
UM[h=2.00m] (m/e)	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50
Х Ү Коорд. точеч, початок лін-го, центр симетр, пл-го (м)	-596.00 365.00	-601.00 360.00	-606.00 355.00	-571.00 400.00	-576.00 395.00	-581.00 390.00	-586.00 385.00	-140.00 153.00	-155.00 107.00
Х Ү Коорд. кінця лін-го, дов. і шарана пл-го(м)	0.00 0.00								
Коеф-т рель' ефу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ПГПС(м. куб/с)	2.6400	2.6400	2.6400	2.6400	2.6400	2.6400	2.6400	0.7500	1.6100
Шв-ть вихіду ПГПС: м/с	0.5836	0.5836	0.5836	0.5836	0.5836	0.5836	0.5836	78.9198	91.1074
Діаметр (м)	2.4000	2.4000	2.4000	2.4000	2.4000	2.4000	2.4000	0.1100	0.1500
Висота (м)	48.0000	48.0000	48.0000	48.0000	48.0000	48.0000	48.0000	50.0000	50.0000
Температура (С)	23.0000	23.0000	23.0000	23.0000	23,0000	23.0000	23.0000	23.0000	23.0000
Коеф-т впоряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Вакад т/р	0.2140	0.2140	0.2140	0.2140	0.2140	0.2140	0.2140	0.0631	0.1325

- 54 -19/09/2024 "АТ «НАЕК «Еверговтом»" ЕОЛ 2000 [b] v4.0, Ліцевнія №133565957

Код джерела Технологічні параметри	13005	13006	13007	13008	13009	13010	13011	13012	13013
Викид г/с	0.0003	0.0003	0.0003	0.0054	0.0020	0.0020	0.0042	0.0003	0.0003
Клас небезпечн.	5	5	5	್ರ	5	5	5	5	5
СМ[b=2.00м] (частки ГДК) СМ[b=2.00м] мг/м, куб СМ/М[b=2.00м] мг/м, куб	0.0003	0.0003	0.0003	0.0049	0.0018	0.0018	0.0038	0.0003	0.0003
ХМ (м)	286.19	286.19	286.19	286.19	286.19	286.19	286.19	286.19	286.19
UM[h=2.00m] (m/c)	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50	0.50
Х Ү Коорд. точеч. початок лів-го, центр симетр. пл-го (м)	-172.00 68.00	-166.00 74.00	-163.00 82.00	-134.00 107.00	-270.00 217.00	-265.00 222.00	-280.00 176.00	-297.00 137.00	-291.00 143.00
Х Ү Коорд. вінця лія-го, дов. і ппарина пл-го(м)	0.00 0.00	0.00 0.00	0.00 0.00	0.00 0.00	0.00 0.00	0.00 0.00	0.00 0.00	0.00 0.00	0,00 0.00
Коеф-т рель' ефу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ПГПС(м. куб/с)	0.1100	0.1100	0.1100	2.0800	0.7500	0.7500	1.6100	0.1100	0.1100
Шв-ть вихіду ПГПС: м/е	19.3850	19.3850	6.2247	117.7039	78.9198	78.9198	91.1074	19.3850	19.3850
Діаметр (м)	0.0850	0.0850	0.1500	0.1500	0.1100	0.1100	0.1500	0.0850	0.0850
Висотя (м)	50.0000	50.0000	50.0000	50.0000	50.0000	50.0000	50.0000	50.0000	50.0000
Температура (С)	23.0000	23.0000	23.0000	23.0000	23.0000	23:0000	23.0000	23.0000	23.0000
Коеф-т впоряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Викид т/р	0.0095	0.0095	0.0095	0.1703	0.0631	0.0631	0.1325	0.0095	0.0095

- 55 -19/09/2024 "АТ «НАЕК «Еверговтом»" ЕОЛ 2000 [h] v4.0, Ліцевзія №133565957

	- 56	i • · · · · · · · · · · · · · · · · · · ·	
19/09/2024	"АТ «НАЕК «Енергоатом»"	ЕОЛ 2000 [h] v4.0,	Ліпетія №133565957

Код джерела Технологічні параметри	13014	13015	13016
Викид г/с	0.0003	0.0054	0.0054
Клас небезпечн.	3	5	5
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мс/м. куб	0.0003	0.0049	0.0049
ХМ (м)	286.19	286.19	286.19
UM[h=2.00m] (m/c)	0.50	0,50	0.50
Х У Коорд. точеч. початок лін-го, центр саметр. пл-го (м)	-288.00 151.00	-259.00 176.00	-256.00 182.00
Х Ү Коорд. кінця лін-го, дов. і шарина пл-го(м)	0.00 0.00	0,00 0.00	0.00 0.00
Коеф-т рель' ефу	1.0000	1.0000	1.0000
Витрата ПГПС(м. куб/с)	0.1100	2.0800	2.0800
Шв-ть вихіду ПГПС: м/с	6.2247	117.7039	117.7039
Діаметр (м)	0.1500	0.1500	0.1500
Висота (м)	50.0000	50.0000	50.0000
Температура (С)	23.0000	23.0000	23.0000
Коеф-т впоряд. осід.	1.0000	1.0000	1.0000
Викид т/р	0.0095	0.1703	0.1703
errors and a Table 1			

Розрахункові концентрації речовини: Масло мінеральне нафтове (веретенне, мацянне, циліндрове і ін.) в розрахункових точках та номера джерел, що надають найбільший внесок на высоті 2.00 м

№ розр. точки	Концентр. v точш частки ГДК	Коорд. розр. точки Х	Коорд. розр. точная Ү	Напрямок вітру	Швидкість вітру	Розмір впеску Q0	№ джерела N0	Розмір внеску Q1	№ джерела N1	Розмір впеску Q2	№ джерела N2
n.	0.0151	0.0	2700.0	78	0.75	0.0004	11042	0.0004	11033	0.0004	11028
22	0.0134	550.0	2700.0	66	2.00	0.0003	11023	0.0003	11024	0.0003	11032

# - 57 - 19/09/2024 "AT «НАЕК «Евергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Точки найбільших конпентрацій речовнии Масло мінеральне нафтове (веретенне, машинне, циліндове і ін.) На розрахун. площадці № 1 та номера джерел, що надають найбільший внесок на высоті 2.00 м

Концентрації у точці частки ГДК	Коорд. розр. точки Х	Коорд. розр. точки Ү	Напрям. вітру	Швидкість вітру	Розмір внеску Q0	№ джереца N0	Розмір внеску Q1	№ джерела N1	Розмір внеску Q2	№ джерела N2
0.1147	-750.0	500.0	148	0.50	0.0030	11030	0.0030	11029	0.0030	11031
0.1003	-250.0	250.0	340	0.50	0.0029	11026	0.0029	11025	0.0029	11027
0.0921	-1000.0	500.0	162	0.50	0.0025	11040	0.0025	11039	0.0025	11041
0.0911	-750.0	750.0	123	0,50	0.0025	11042	0.0025	11033	0.0025	11028
0.0889	-250.0	0.0	305	0,50	0.0027	11022	0.0027	11021	0.0027	11020
0.0872	0.0	250.0	350	0.50	0.0026	11013	0.0026	11012	0.0026	11011
0.0869	0.0	0.0	325	0.50	0.0023	11013	0.0023	11012	0.0023	11011
0.0864	-1000.0	250.0	190	0.50	0.0025	11041	0.0024	11040	0.0024	11039
0.0811	-500.0	750.0	92	0.50	0.0024	11032	0.0023	11031	0.0023	11022
0.0803	-750.0	250.0	198	0.50	0.0026	11023	0.0026	11024	0.0026	11025

Маело мінеральне нафтове (веретенне, машинне, циліндрове і ів.) Карта-ехема H=2.00 м



### - 59 -19/09/2024 "АТ «НАЕК «Евергоатом»" ЕОЛ 2000 [h] v4.0, Ліцешія №133565957

Код речовини	Найменування речовнин	ГДК (мг/м.куб)
(2754)	Вуглеводні граничні с12-с19(розчанник РПК-2661	i 1.00000000

Фонові концентрації, які вміщують внески діючих джерел (Частки ГДК) (частки ГДК) (Вихідні рівні забруднення) для речовини : Вуглеводні граничні c12-c19(розчинник РПК-26611 і ін.). Варіант завдання фону : а.

Коорд. Х поста спостереження	Коорд. Ү поста спостереження	U<2 м/с (штиль)	Швядкість вітру 2⊲U <u* Пв</u* 	IIIвилкість вітру 2 <u<u* ПнС</u<u* 	Швядкість вітру 2 <u<u* С</u<u* 	Швядкість вітру 2 <u<u* ПдС</u<u* 	Швядкість вітру 2 <u<u* Пд</u<u* 	IIIвидкість вітру 2 <u<u* Пд3</u<u* 	Швидкість вітру 2 <u<u* 3</u<u* 	Швяджість вітру 2 <u<u* Пв3</u<u* 
0.00	0.00	0.40	0.40	0.40	0.40	0.40	0.40	0.40	0.40	0.40

Фонові концентрації без урахування внесків діючих джерел (Частки ГДК) (частки ГДК) (Власне фон - верхис число, вклад - нижиє) для речовнин : Вуглеводні граничні e12-e19(розчинних РПК-26611 і іп.). Варіант завдання фону : а.

Коорд. Х поста спостереження	Коорд. У поста спостереження	U<2 м/с (штиль)	Швядкість вітру 2 <u<u* Пв</u<u* 	Шпилкість вітру 2 <u<u* ПнС</u<u* 	Швилкість вітру 2 <u<u* C</u<u* 	Шиндкість вітру 2 <u<u* ПдС</u<u* 	Швядкість вітру 2 <u<u* Пд</u<u* 	Шпилкість вітру 2 <u<u* Пл3</u<u* 	Шлилкість вігру 2 <u<u* 3</u<u* 	Швялкість вітру 2 <u<u* Пв3</u<u* 
0.00	0.00	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000	0.4000

Код джереля Технологічні	10131	10132	10133	10134	11001	11002	11003	11004	11005
Buana r/e	0.00002	0.00002	0.00002	0.00002	0.00002	0.00002	0.00002	0.00002	0.000008
Клас небезпечн.	5	3	5	5	5	5	5	5	5
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/M[h=2.00м] мс/м. куб	0.0005	0.0005	0.0005	0.0005	0.0001	0.0001	0.0001	0.0001	0.0004
ХМ (м)	34.34	34.34	34.34	34.34	68.68	68.68	68.68	68.68	28.62
UM[h=2.00м] (м/с)	0.50	0.50	0.50	0.50	0.50	0.50	0,50	0.50	0,50
Х Ү Коорд. точеч. початок лін-го, центр симетр. пл-го (м)	535.00 -325.00	550.00 -325.00	545.00 -340.00	532.00 -340.00	-500.00 160.00	-550.00 180.00	-678.00 200.00	-728.00 220.00	-550.00 165.00
Х Ү Коорд. кінця лів-го, дов. і шарная пл-го(м)	0.00 0.00								
Коеф-т рель' өфу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ПГПС(м. куб/с)	0.2500	0.2500	0.2500	0.2500	0.2500	0.2500	0.2500	0.2500	0.2900
Шв-ть вихіду Ш'ПС: м/е	2.2044	2.2044	2.2044	2.2044	2.2044	2.2044	2.2044	2.2044	1.4770
Діаметр (м)	0.3800	0.3800	0.3800	0.3800	0.3800	0.3800	0.3800	0.3800	0.5000
Висота (м)	6.0000	6.0000	6.0000	6.0000	12.0000	12.0000	12.0000	12.0000	5.0000
Температура (С)	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000	23.0000
Коеф-т впоряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Викид т/р	0.0045	0.0045	0.0045	0.0045	0.00063	0.00063	0.00063	0.00063	0.000000

### - 60 -19/09/2024 "AT «НАЕК «Енергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Перелік джерел. у викидах яких є Вуглеводні граничні c12-c19(розчинник РПК-26611 і ін.)

Код джерела	11006	11007	11008	11049	11050	13001	13002
Технологічні параметри							
Ванаяд г/с	0.000008	0.000008	0.000008	0.000002	0.7900	0.00002	0.00000
Клас небезпечн.	5	:5	5	3	5	5	ંક
CM[h=2.00м] (частки ГДК) CM[h=2.00м] мг/м. куб CM/M[h=2.00м] мг/м. куб	0.0004	0.0004	0.0004	0.0005	0.4829	0.0001 :	0.0004
ХМ (м)	28.62	28.62	28.62	12.48	201,84	68.68	28.62
UM[h=2.00м] (м/е)	0.50	0.50	0.50	0.50	5.60	0.50	0.50
Х Ү Коорд. точеч. початок лін-го, центр саметр. пл-го (м)	-600.00 185.00	-683.00 205.00	-733.00 225.00	-420.00 550.00	-500.00 365.00	-270.00 -50.00	-280.00 -65.00
Х Ү Коорд. кінця лін-го, дов. і шарана пл-го(м)	0.00 0.00	0.00 0.00	0.00 0.00	-410.00 695.00	0.00 0.00	0.00 0.00	0.00 0.00
Коеф-т рель'ефу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ПГПС(м. куб/с)	0.2900	0.2900	0.2900	0.2500	44.7000	0.2500	0.2900
Шв-ть вихіду ПГПС: м/с	1.4770	1.4770	1.4770	32.0000	3.5571	2.2044	1.4770
Діаметр (м)	0.5000	0.5000	0.5000	0.0001	4.0000	0.3800	0.5000
Висота (м)	5.0000	5.0000	5.0000	5.0000	8.5000	12.0000	5.0000
Температура (C)	23.0000	23.0000	23.0000	250.0000	55.0000	23.0000	23.0000
Коеф-т впоряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Викид т/р	0.0000007	0.0000007	0.0000007	0.0000	0.1370	0.00063	0.00000

- 61 -19/09/2024 "АТ «НАЕК «Еверговтом»" ЕОЛ 2000 [h] v4.0, Ліцевзія №133565957

Розрахункові концентрації речовини: Вуглеводні граничні с12-с19(розчинник РПК-26611 і ін.) в розрахункових гочках та номера джерел, що надають найбільший внесок на высоті 2.00 м

№ розр. точки	Концентр. v точий частки ГДК	Коорд. розр. точки Х	Коорд. розр. точная Ү	Напрямок вітру	Швидвість вітру	Розмір впеску Q0	№ джерела N0	Розмір внеску Q1	№ джерела N1	Розмір внеску Q2	№ джерела N2
т	0.4069	0.0	2700.0	78	2.00	0.0069	11050	0.0000	11002	0.0000	11001
22	0.4061	550.0	2700.0	66	2.00	0.0061	11050	0.0000	11002	0.0000	11001

# - 62 - 19/09/2024 "AT «НАЕК «Евергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Точки найбільших концентрацій речовини Вуглеводні граничні с12-с19(розчинник РПК-26611 і ін.) На розрахун. площадці № 1 та номера джерел, що надають найбільший внесок на высогі 2.00 м

Концентрації у точці частки ГДК	Коорд, розр. точки Х	Коорд. розр. точки Ү	Напрям. вітру	Швядкість вітру	Розмір внеску Q0	№ джерела N0	Розмір внеску Q1	№ джереля N1	Розмір внеску Q2	№ джерела N2
0.5508	-500.0	250.0	270	7.00	0.1508	11050	0.0000	11049		
0.5352	-500.0	500.0	390	37.00	0.1352	11050	0.0000	11001	0.0000	11005
0.4844	-750.0	250.0	205	5.5576	0.0844	11050	0.0000	11049	0.0000	11002
0.4844	-250.0	250.0	335	5,5576	0.0844	11050	0.0000	11008	0.0000	11004
0.4833	-250.0	500.0	28	5,5576	0.0833	11050	0.0000	11008	0.0000	11004
0.4833	-750.0	500.0	152	5.5576	0.0833	11050	0.0000	10132	0.0000	10131
0.4743	-500.0	0.0	270	7.00	0.0743	11050	0.0000	11001	0.0000	11049
0.4721	-500.0	750.0	90	7.00	0.0721	11050	0.0000	11001	0.0000	11005
0.4660	-750.0	0.0	236	7.00	0.0660	11050	0.0000	11006	0.0000	11002
0.4660	-250.0	0.0	304	7.00	0.0660	11050	0.0000	11001	0.0000	11002



Вуглеводні граничні e12-e19(розчинник РПК-26611 і ін.) Карта-схема H=2.00 м

### - 64 -19/09/2024 "АТ «НАЕК «Енергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Код речовнии	Найменування речовнин	ГДК (мг/м.куб)
(2902)	Зважені речовини, недиференційовані за складом	0.50000000

Фонові концентрації, які вміщують внески діючих джерел (Частки ГДК) (частки ГДК) (Вихідні рівні забруднення) для речовини : Зважені речовини, недиференційовані за складом. Варіант завдання фону : а.

Коорд. Х поста спостереження	Коорд. Ү поста спостережения	U<2 м/с (штиль)	Швядкість вітру 2 <u<u* Пв</u<u* 	IIIвилкість вітру 2 <u<u* ПнС</u<u* 	Швидкість вітру 2 <u<u* С</u<u* 	Швядкість вітру 2 <u<u* ПдС</u<u* 	Швядкість вітру 2 <u<u* Пд</u<u* 	Швядкість вітру 2 <u<u* Пд3</u<u* 	Швидкість вітру 2 <u<u* 3</u<u* 	Швюдкість вітру 2 <u<u* Пв3</u<u* 
0.00	0.00	0.10	0.10	0.10	0.10	0.10	0.10	0.10	0.10	0.10

Фонові концентрації без урахування внесків діючих джерел (Частки ГДК) (частки ГДК) (Власне фон - верхне число, вклад - нижнє) для речовнин : Зважен речовнин, недиференційовані за складом. Варіант завдання фону : а.

Коорд. Х поста спостереження	Коорд. У поста спостереження	U<2 м/с (штиль)	Швядкість вітру 2 <u<u* Пп</u<u* 	Шпилкість вітру 2 <u<u* ПиС</u<u* 	Швидкість вітру 2 <u<u* C</u<u* 	IIImmedicers simpy 2 <u<u* HaC</u<u* 	Швилкість вітру 2 <u<u* Пд</u<u* 	Швилкість вітру 2 <u<u* Пл3</u<u* 	Шпилкість вігру 2 <u<u* 3</u<u* 	Швядкість вітру 2 <u<u* Пв3</u<u* 
0.00	0.00	0.1000	0.1000	0.1000	0.1000	0.1000	0.1000	0.1000	0.1000	0.1000

		Second and the	0.000000000	el a ser a ser a la	8 0.00 MAG	
Код джерела Технологічні параметри	10101	10202	10203	11047	11048	11050
Вананд г/с	0.3731	0.0075	0.0075	0.0075	0.0072	1.0500
Клас небезпечн.	5	5	5	5	5	5
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мг/м. куб	0.0121	0.0528	0.0528	0.0528	0.0507	1.2836
ХМ (м)	546.37	62.96	62.96	62.96	62.96	201.84
UM[h=2.00м] (м/с)	1.71	0.50	0.50	0.50	0.50	5,60
Х Ү Коорд. точеч. початок лін-го, центр симетр. пл-го (м)	330.00 -245.00	485.00 -80.00	480.00 -90.00	-330.00 390.00	-295.00 385.00	-500.00 365.00
Х Ү Коорд. кінця лін-го, дов. і шарняка пл-го(м)	0.00 0.00	0.00 0.00	0.00 0.00	0.00 0.00	0.00 0.00	0.00 0.00
Коеф-т рель' ефу	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Внтрата ПГПС(м. куб/с)	11:4000	1.3400	1.3400	1.3400	1.3400	44.7000
Шв-ть вихіду ПГПС: м/е	1.4175	6.8246	6.8246	6.8246	6.8246	3.5571
Діаметр (м)	3.2000	0.5000	0.5000	0.5000	0.5000	4.0000
Висота (м)	60:0000	11.0000	11.0000	11.0000	11.0000	8.5000
Температура (C)	118.0000	23.0000	23.0000	23.0000	23.0000	55.0000
Коеф-т впоряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
Викид т/р	0.0735	0.0130	0.0130	0.0130	0.0130	0.1810

### - 65 -19/09/2024 "AT «НАЕК «Евергоатом»" ЕОЛ 2000 [h] v4.0, Ліцевнія №133565957

Перелік джерел. v викидах яких є Зважені речовини, недиференційовані за складом

Розрахункові концентрації речовини: Зважені речовини, недиференційовані за складом в розрахункових точвах та номера джерел, що надають найбільший внесок на высоті 2.00 м

№ розр. точки	Концентр. у точції частки ГДК	Коорд. розр. точкн Х	Коорд. розр. точки Ү	Напрямок вітру	Швидвість вітру	Розмір внеску Q0	№ джерела N0	Розмір внеску Q1	Nb джерела N1	Розмір внеску Q2	№ ,жерепа N2
11	0.1187	0.0	2700.0	78	2.00	0.0184	11050	0.0002	10101	0.0001	11047
22	0.1166	550.0	2700.0	66	2.00	0.0163	11050	0.0001	10101	0.0001	11047

### - 66 -19/09/2024 "АТ «НАЕК «Еверговтом»" ЕОЛ 2000 [b] v4.0, Ліпевнія №133565957

Точки найбільших концентрацій речовини Зважені речовини, недиференційовані за складом На розракун. площадці № 1 та номера джерел, що надають найбільший внесок на высоті 2.00 м

Концентрації у точці частки ГДК	Коорд. розр. точки Х	Коорд. розр. точки Ү	Напрям. вітру	Швядкість вітру	Розмір внеску Q0	№ джерена N0	Розмір внеску Q1	№ джереля N1	Розмір внеску Q2	№ джерела N2
0.5007	-500.0	250.0	270	7.00	0.4007	11050				
0.4594	-500.0	500.0	390	37.00	0.3594	11050				
0.3243	-750.0	250.0	205	7.00	0.2234	11050	0.0006	11047	0.0003	11048
0.3234	-250.0	250.0	335	7.00	0.2234	11050	0.0000	11047		
0.3228	-750.0	500.0	152	7,00	0.2209	11050	0.0011	10101	0.0003	10203
0.3209	-250.0	500.0	28	7.00	0.2209	11050	0.0000	11047	0.0000	11048
0.2975	-500.0	0.0	270	7.00	0.1975	11050	0.0000	11047	0.0000	11048
0.2917	-500.0	750.0	90	7.00	0.1917	11050	0.0000	11047	0.0000	11048
0.2755	-750.0	0.0	236	7.00	0.1754	11050	0.0001	11047	0.0000	11048
0.2754	-250.0	0.0	304	7.00	0.1754	11050	0.0000	11047	0.0000	11048

Зважені речовнни, педиференційовані за складом Карта-ехема H=2.00 м



### - 68 -19/09/2024 "АТ «НАЕК «Енергоатом»" ЕОЛ 2000 [b] v4.0, Ліцензія №133565957

Код речовнин	Найменування речовнин	ГДК (мг/м.вуб)
(10265)	Емульсол (склад:вода-97.6%,нітрит натрію-0,2% та .	0.05000000

Опис фона для речовини : Емульсол (склад:вода-97.6%, нітрит натрію-0.2% та інш.)

Опне фону відсутній чи не враховувалось у розрахунках.

. Перелік джерел, у викидах яких <br/>  $\epsilon$  Емульсол (склад:вода-97.6% дітрит натрію-0.2% та інш.)

			1	
Код джерела Технологічні параметри	10202	10203	11047	11048
Викид г/е	0.0007	0.0007	0.0007	0.0006
Клас небезпечн.	5	5	5	5
СМ[h=2.00м] (часткн ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мс/м. куб	0.0493	0.0493	0.0493	0.0423
ХМ (м)	62.96	62.96	62.96	62.96
UM[h=2.00m] (m/c)	0.50	0.50	0.50	0.50
Х Ү Коорд. точеч. початок лін-го, центр симетр. пл-го (м)	485.00 -80.00	480.00 -90.00	-330.00 390.00	+295.00 385.00
Х Ү Коорд. кінця лія-го, дов. і шаряна пл-го(м)	0.00 0.00	0.00 0.00	0.00 0.00	0.00 0.00
Коеф-т рель' өфу	1.0000	1.0000	1.0000	1.0000
Витрата ШПС(м. куб/с)	1.3400	1.3400	1.3400	1.3400
Шв-ть вихіду ПГПС: м/с	6.8246	6.8246	6.8246	6.8246
Діаметр (м)	0.5000	0.5000	0.5000	0.5000
Висота (м)	11.0000	11.0000	11.0000	11.0000
Температура (C)	23.0000	23.0000	23.0000	23.0000
Коеф-т впоряд. осід.	1.0000	1.0000	1.0000	1.0000
Викид т/р	0.0028	0.0028	0.0028	0.0028

### - 69 -19/09/2024 "АТ «НАЕК «Евергоатом»" ЕОЛ 2000 [h] v4.0, Ліцевзія №133565957

Розрахункові концентрації речовнин: Емудьсол (склад:вода-97.6% нітрит вятрію-0.2% та інш.) в розрахункових точках та номера джерел, що надають найбільший внесок на высоті 2.00 м

№ розр. точки	Коншентр. у точці частки ГДК	Коорд. розр. точки Х	Коорд. розр. точки Ү	Напрямок вітру	Швялкість вітру	Розмір внеску Q0	№ джерела N0	Розмір внеску Q1	№ джерела N1	Розмір внеску Q2	№ джерела N2
11	0.0003	0.0	2700.0	94	0.75	0.0001	11047	0.0001	10202	0.0001	10203
22	0.0003	550.0	2700.0	82	0.75	0.0001	10202	0.0001	10203	0.0001	11047

Точки найбільших концентрацій речовини Емульсол (склад:вода-97.6%, нітрит натрію-0.2% та інш.) На розрахун. плошадці № 1 та номера джерел, що надають найбільший внесок на высоті 2.00 м

Концентрації у точпі частки ГДК	Коорд. розр. точки Х	Коорд, розр. точки Ү	Напрям. вітру	Швидкість вітру	Розмір внеску Q0	№ джерена N0	Розмір внеску Q1	№ джерела N1	Розмір внеску Q2	№ джерела N2
0.0164	500.0	0.0	83	0.50	0.0085	10202	0.0079	10203		
0.0113	-250.0	500.0	56	0.50	0.0065	11047	0.0048	11048		
0.0112	500.0	-250.0	274	0.75	0.0056	10203	0.0055	10202		
0.0104	-250.0	250.0	290	0.50	0.0054	11048	0.0050	11047		
0.0087	-500.0	500.0	149	0.75	0.0047	11047	0.0034	11048	0.0003	10202
0.0076	250.0	0.0	159	0.75	0.0038	10203	0.0038	10202		
0.0075	-500.0	250.0	217	0.75	0.0043	11047	0.0032	11048		
0.0064	750.0	0.0	20	0.75	0.0032	10202	0.0032	10203	0.0000	11047
0.0063	250.0	-250.0	213	0.75	0.0032	10203	0.0031	10202		
0.0060	750.0	-250.0	329	0.75	0.0028	10203	0.0028	10202	0.0003	11047

3500.00 3250.00 Some an 3000.00 2750.00 2500.00 2258.00 2000.00 1750.00 Tenus COURT IN 1500.00 1260.00 1000.00 758.00 0 500.00 250.00 11 1 748B 0.00 mine 7820 -250.00 -600.00 -750.00 1000.00 -1250.00 -1500.00 1750.00 -2000.00 -2250.00 -2600.00 -2750.00 3000.00 3258.00 3500.00 1000.00 1500.00 2000.00 2500.00 3000.00 3500.00 -3009.00 -2500.00 -2000.00 -1500.00 -1000.00 -500.00 0.00 500.00

Нормативна санітарно-захисна зопа

Емульсол (склад вода-97.6%, нітрит натрію-0.2% та інш.) Карта-схема H=2.00 м

### -71 -19/09/2024 "АТ «НАЕК «Евергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Код гр. сум.	Код речовнин	Найменування речовани	ГДК (мг/м.куб)
31	301	Азоту діокенд	0.20000000
	330	Ангідрид сірчистий	0.50000000

Фонові концентрації, які вміщують внескя діючих джерел (Часткя ГДК) (часткя ГДК) (Вихідні рівні забруднення) для групн сумації № 31. Варіант завдання фону : а.

Коорд. Х поста спостережения	Коорд. У поста спостережения	U<2 м/с (штиль)	Швидкість вітру 2 <u<u* Пи</u<u* 	Швидкість вітру 2 <u<u* ПиС</u<u* 	Швидкість пітру 2 <u<u* С</u<u* 	Швилкість віпуv 2 <u<u* ПлС</u<u* 	Швндкість вітру 2 <u<u* Пд</u<u* 	Швялкість вітру 2 <u<u* Пд3</u<u* 	Швндкість вітру 2 <u<u* 3</u<u* 	Швядкість вітру 2 <u<u* Пп3</u<u* 
0.00	0.00	0.08	0.08	0.08	0.08	0.08	0.08	0.08	0.08	0.08

Фонові копцентрації без урахування внесків діючих джерел (Частки ГДК) (частки ГДК) (Власне фон - верхис число, вклад - нижиє) для групи сумації № 31. Варіант завдання фону : а.

Коорд. Х поста спостереження	Коорд. У поста спостереження	U<2 м/с (штиль)	Швядкість вітру 2 <u<u* Пя</u<u* 	Швиджість вітру 2 <u<u* ПнС</u<u* 	Швидкість вітру 2 <u<u* C</u<u* 	Швидкість віпуv 2 <u<u* ПдС</u<u* 	Швидкість вітру 2 <u<u* Пд</u<u* 	Швидкість вітру 2 <u<u* Пд3</u<u* 	Швидкість вітру 2 <u<u* 3</u<u* 	Швидкість вітру 2 <u<u* Пн3</u<u* 
0.00	0.00	0.0800	0.0800	0.0800	0.0800	0.0800	0.0800	0.0800	0.0800	0.0800

		Перелік д Гру	керел, у виз па сумації )	андах якнх с № 31	-
Код джерела Технологічні параметри	***10101	10201	11046	***11049	***11050
Викид г/е	26.097200394	0.0020	0.0020	0.0250016	42.6250
Слас небезпечн.	5	5	5	5	5

# -72 -19/09/2024 "АТ «НАЕК «Евергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

	Викид г/е	26.097200394	0.0020	0.0020	0.0250016	42.6250
	Клас небезпечн.	5	5	5	5	5
	СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мс/м. куб	0.8440	0.0157	0.0157	11.8043 -	52.1067
ľ	XM (M)	546.37	86.53	86.53	12.48	201.84
	UM[b=2.00м] (м/с)	1.71	0.54	0.54	0.50	5.60
	Х Ү Кооод. точеч. початок лін-го, центр симетр. пл-го (м)	330.00 -245.00	475.00 -120.00	-320.00 400.00	-420.00 550.00	-500.00 365.00
	Х Ү Коорд. кінця лін-го, дов. і шприма пл-го(м)	0.00 0.00	0.00 0.00	0.00 0.00	-410.00 695.00	0.00 0.00
	Коеф-т рель`офу	1.0000	1.0000	1.0000	1.0000	1.0000
	Витрата ПГПС(м. куб/с)	11.4000	2.7400	2.7400	0.2500	44.7000
	Шв-ть вихіду ШГПС: м/е	1.4175	9.6908	9.6908	32.0000	3.5571
	Діаметр (м)	3.2000	0.6000	0.6000	0.0001	4.0000
	Висота (м)	60.0000	14.0000	14.0000	5.0000	8.5000
	Температура (С)	118.0000	23.0000	23.0000	250.0000	55.0000
Í	Коеф-т впоряд. осіл.	1.0000	1.0000	1.0000	1.0000	1.0000
	Вакид т/р	2.736599922	0.0023	0.0023	0.0000	7.366999626
ļ						

Розрахункові концентрації групи сукації № 31 в розрахункових точках та номера джерел, що надають найбільший внесок на высоті 2.00 м

№ розр. точки	Концентр. у точції частки ГДК	Коорд. розр. точкн Х	Коорд, розр. точки Ү	Напоямок вітру	Швидијсть вітру	Розмір внеску Q0	№ джерела N0	Розмір внеску Q1	ур джерела N1	Розмір внеску Q2	№ джерела N2
11	0.8007	0.0	2700.0	81	2.00	0.6966	11050	0.0235	10101	0.0006	11049
22	0.7169	550.0	2700.0	69	2.00	0.6184	11050	0.0176	10101	0.0008	11049

		<ul> <li>Monoconconstantes</li> </ul>	
19/09/2024	"AT «НАЕК «Енергоатом»"	ЕОЛ 2000 [h] v4.0,	Ліцензія №133565957
	· · · · · · · · · · · · · · · · · · ·		2011 A.M. 1

Точки найбільших концентрацій групи сумації № 31 На розрахун. площадці № 1 та номера джерел, що надають найбільший внесок на высоті 2.00 м

Концентрації у точці частки ГДК	Коорд. розр. точки Х	Коорд. розр. точки Ү	Напрям. вітру	Швидкість вітру	Розмір внеску Q0	№ джерела N0	Розмір внеску Q1	№ джерета N1	Розмір внеску Q2	№ джерела N2
16.0213	-500.0	250.0	269	7.00	15.9410	11050	0.0003	11049		
14.5719	-500.0	500.0	89	7.00	14.4919	11050				
8.7874	-750.0	250.0	203	6.9305	8.7071	11050	0.0003	11046	0,0000	11049
8.1278	-750.0	500.0	149	6.9305	7.9280	11050	0.1 196	10101	0.0001	10201
7.2002	-750.0	0.0	235	6.9305	7.1198	11050	0.0004	11049	0.0000	11046
7.1996	-250.0	500.0	32	6.9305	7.1196	11050	0.0000	11046	0.0000	11049
7.1917	-250.0	0.0	304	6.9305	7.1117	11050	0.0000	11049	0.0000	11046
6.8803	-250.0	750.0	56	6.9305	6.8003	11050	0.0000	11046	0.0000	11049
6.5347	-250.0	250.0	331	6.9305	6.4547	11050	0.0000	11049	0.0000	11046
6.4296	0.0	250.0	347	7.00	6.3495	11050	0.0000	11046	0.0000	11049

-74 -19/09/2024 "АТ «НАЕК «Евергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957



Група суманії № 31 Карта-схема H=2.00 м

### -75 -19/09/2024 "АТ «НАЕК «Евергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Код гр. сум.	Код речовнин	Найменування речовини	ГДК (мг/м.куб)
35	330 342	Ангідряд сірчнетий Фтористі сполуки газоподібні (фтористий	0.50000000 0.02000000

Фонові концентрації, які вміщують внескя діючих джерел (Часткя ГДК) (часткя ГДК) (Вихідні рівні забруднення) для групн сумації № 35. Варіант завдання фону : а.

Коорд. Х поста спостережения	Коорд, У поста спостережения	U<2 м/с (штяль)	Швядкість вітру 2 <u<u* Пя</u<u* 	Швидкість вітру 2 <u<u* ПиС</u<u* 	Швидкість niтру 2 <u<u* С</u<u* 	Швилкість віпуv 2 <u<u* ПлС</u<u* 	Швидкість вітру 2 <u<u* Пд</u<u* 	Швялкість вітру 2 <u<u* Пд3</u<u* 	Швндкість вітру 2 <u<u* 3</u<u* 	Швядкість вітру 2 <u<u* Пп3</u<u* 
0.00	0.00	0.44	0.44	0.44	0.44	0.44	0.44	0.44	0.44	0.44

Фонові копцентрації без урахування внесків діючих джерел (Частки ГДК) (частки ГДК) (Власне фон - верхис число, вклад - нижиє) для групи сумації № 35. Варіант завдання фону : а.

Коорд. Х поста спостереження	Коорд. У поста спостереження	U<2 м/с (штиль)	Швядкість вітру 2 <u<u* Пя</u<u* 	Швиджість вітру 2 <u<u* ПнС</u<u* 	Швидкість вітру 2 <u<u* C</u<u* 	Швидкість віпуv 2 <u<u* ПдС</u<u* 	Швидкість вітру 2 <u<u* Пд</u<u* 	Швидкість вітру 2 <u<u* Пл3</u<u* 	Швидкість вітру 2 <u<u* 3</u<u* 	Швидкість вітру 2 <u<u* Пн3</u<u* 
0.00	0.00	0.4400	0.4400	0.4400	0.4400	0.4400	0.4400	0.4400	0.4400	0.4400

	- 76	- 76 -						
19/09/2024	"АТ «НАЕК «Евергоатом»"	ЕОЛ 2000 [h] v4.0,	Ліценція №133565957					

Перелік джерел, у викидах яких є Група сумації № 35

Код джерела Технологічні параметри	10101	10201	11046	11049	11050
Винанд г/с	12.747200012	0.0012	0.0012	0.0000016	3.2500
Клас небезпечн.	5	5	5	5	5
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мс/м. куб	0.4122	0.0939	0.0939	0.0008	3,9729
ХМ (м)	546.37	86.53	86.53	12.48	201.84
UM[h=2.00м] (м/с)	1.71	0.54	0.54	0.50	5.60
Х Ү Коорд. точеч. початок лін-го, центр симетр. пл-го (м)	330.00 -245.00	475.00 -120.00	-320.00 400.00	-420.00 550.00	-500.00 365.00
Х Ү Коорд. кінця лів-го, дов. і шарняка пл-го(м)	0.00 0.00	0.00 0.00	0.00 0.00	-410.00 695.00	0.00 0.00
Коеф-т рель' ефу	1.0000	1.0000	1.0000	1.0000	1.0000
Витрата ПГПС(м. куб/с)	11.4000	2.7400	2.7400	0.2500	44.7000
Шв-ть вихіду Ш'ПС: м/е	1.4175	9.6908	9.6908	32.0000	3.5571
Діаметр (м)	3.2000	0.6000	0.6000	0.0001	4.0000
Висота (м)	60.0000	14.0000	14.0000	5.0000	8.5000
Температура (С)	118.0000	23.0000	23.0000	250.0000	55.0000
Коеф-т впоряд. осід.	1.0000	1.0000	1.0000	1.0000	1.0000
Викид т/р	2.616600037	0.0020	0.0020	0.0000	0.561999977

Розрахункові концентрації групи сукації № 35 в розрахункових точках та номера джерел, що надають найбільший внесок на высоті 2.00 м

№ розр. точки	Концентр. у точції частки ГДК	Коорд. розр. точкн Х	Коорд. розр. точки Ү	Напоямок вітру	Швидвість вітру	Розмір внеску Q0	№ джерела N0	Розмір внеску Q1	Nb джерела N1	Розмір внеску Q2	N⊵ ,∞kepena N2
11	0.5030	0.0	2700.0	86	2.00	0.0371	11050	0.0255	10101	0.0003	11046
22	0.4945	550.0	2700.0	66	2.00	0.0506	11050	0.0036	10101	0.0003	11046

	- 77		
19/09/2024	"AT «НАЕК «Евергоатом»"	ЕОЛ 2000 [h] v4.0,	Jinemia №133565957

Точки найбільших концентрацій групи сумації № 35 На розрахун. площадці № 1 та номера джерел, що надають найбільший внесок на высоті 2.00 м

Концентрації у точці частки ГДК	Коорд. розр. точки Х	Коорд. розр. точки Ү	Напрям. вітру	Швядкість вітру	Розмір внеску Q0	№ джереца N0	Розмір внеску Q1	№ джерела N1	Розмір внеску Q2	№ джерела N2
1.6683	-500.0	250.0	269	7.00	1.2283	11050	0.0000	11049		
1.4928	-500.0	500.0	88	7.00	1.0528	11050				
1.1282	-750.0	250.0	205	7.00	0.6868	11050	0.0014	11046	0.0000	10201
1.1215	-250.0	500.0	28	7.00	0.6815	11050	0.0000	11046	0.0000	11049
1,0331	-500.0	750.0	90	7,00	0.5931	11050	0.0000	11046	0.0000	10101
1.0066	-750.0	500.0	155	7.00	0.5506	11050	0.0146	10101	0.0011	10201
0.9669	-250.0	250.0	340	7.00	0.5269	11050	0.0000	11046	0.0000	11049
0.9616	-750.0	0.0	237	7.00	0.5215	11050	0.0001	11046	0.0000	11049
0.9445	-500.0	0.0	274	7.00	0.5045	11050	0.0000	11046	0.0000	11049
0.9264	-250.0	0.0	307	7.00	0.4864	11050	0.0000	11046	0.0000	11049

-78 -19/09/2024 "АТ «НАЕК «Евергоатом»" ЕОЛ 2000 [b] v4.0, Ліцензія №133565957



Група суманії № 35 Карта-схема H=2.00 м

### -79 -19/09/2024 "АТ «НАЕК «Еверговтом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957

Код гр. сум.	Код речовини	Найменування резовани	ГДК (мг/м.куб)
11002	342	Фтористі сполуки газоподібні (фтористий	0.02000000
	344	Фтористі сполуки погано розчинні неорганічні	0.2000000

Фонові концентрації, які вміщують внески діючих джерел (Частки ГДК) (частки ГДК) (Вихідні рівні забруднения) для трупи сумації № 11002. Варіант завдяння фону : в.

Коорд. Х поста спостережения	Коорд. У поста спостережения	U<2 м/с (штиль)	Швядвість вітру 2 <u<u* Пп</u<u* 	Швидкість вітру 2 <u<u* ПиС</u<u* 	Швидкість niтру 2 <u<u* С</u<u* 	Швилкість віпуv 2 <u<u* ПаС</u<u* 	Швидкість вітру 2 <u<u* Пд</u<u* 	Швадкість вітру 2 <u<u* Пд3</u<u* 	Швндкість вітру 2 <u<u* 3</u<u* 	Швядкість вітру 2 <u<u* Пи3</u<u* 
0.00	0.00	0.80	0.80	0.80	0.80	0.80	0.80	0.80	0.80	0.80

Фонові копцентрації без урахування внесків діючих джерел (Частки ГДК) (частки ГДК) (Власне фон - верхис число, вклад - нижиє) для групи сумації № 11002. Варіант завдання фону : а.

Коорд. Х поста спостереження	Коорд. У поста спостереження	U<2 м/с (штиль)	Швядкість вітру 2 <u<u* Пя</u<u* 	Швиджість вітру 2 <u<u* ПнС</u<u* 	Швидкість вітру 2 <u<u* C</u<u* 	Швидкість віпуv 2 <u<u* ПдС</u<u* 	Швидкість вітру 2 <u<u* Пд</u<u* 	Швидкість вітру 2 <u<u* Пл3</u<u* 	Швидкість вітру 2 <u<u* 3</u<u* 	Швидкість вітру 2 <u<u* Пн3</u<u* 
0.00	0.00	0.8000	0.8000	0.8000	0.8000	0.8000	0.8000	0.8000	0.8000	0.8000

# - 80 -19/09/2024 "АТ «НАЕК «Евергоатом»" ЕОЛ 2000 [h] v4.0, Ліцевзія №133565957

Перелік джерел, у викидах яких є Група сумації № 11002

Код джерела Технологічні параметри	***10201	***11046		
Викад г/е	0.0128	0.0128		
Клас небезпечн.	5	5		
СМ[h=2.00м] (частки ГДК) СМ[h=2.00м] мг/м. куб СМ/М[h=2.00м] мс/м. куб	0.1002	0.1002		
ХМ (м)	86.53	86.53		
UM[b=2.00м] (м/с)	0.54	0,54 -320,00 400,00 0,00 0,00 1,0000 2,7400 9,6908 0,6000 14,0000		
Х Ү Коорд. точеч. початок лін-го, центр симетр. лл-го (м)	475.00 -120.00			
Х Ү Коорд. кінця лін-го, дов. і шприна пл-го(м)	0.00 0.00			
Коеф-т рель' офу	1,0000			
Внтратя ПГПС(м. куб/с)	2.7400			
Шв-ть вихіду ШГПС: м/е	9.6908			
Діаметр (м)	0.6000			
Висота (м)	14.0000			
Температура (С)	23.0000	23.0000		
Коеф-т впоряд. осід.	1.0000	1.0000		
Викид т/р	0.022200001 0.0222000			

Розрахункові концентрації групи сумації № 11002 в розрахункових точвах та номера джерел, що надають найбільший внесок на высоті 2.00 м

№ розр. точки	Конпентр. у точщі частки ГДК	Коорд. розр. точкн Х	Коорд. розр. точки Ү	Напоямок вітру	Швидкість вітру	Розмір внеску Q0	№ джерела N0	Розмір внеску Q1	Ме джерела N1	Розмір внеску Q2	№ джерела N2
11	0.8007	0.0	2700.0	82	7.00	0.0007	11046	0.0000	10201		
22	0.8007	550.0	2700.0	70	7.00	0.0007	11046	0.0000	10201		

	- 81		
19/09/2024	"AT «НАЕК «Евергоатом»"	ЕОЛ 2000 [h] v4.0,	Ліцензія №133565957

Точки вайбільших концентрацій групп сумації № 11002 На розрахун. площадці № 1 та номера джерел, що надають найбільший внесок на высоті 2.00 м

Конпентрації у точці частки ГДК	Коорд. розр. точки Х	Коорд. розр. точки Ү	Напрям. вітру	Швидкість вітру	Розмір внеску Q0	№ джерела N0	Розмір внеску Q1	№ джерела N1	Розмір внеску Q2	№ джерела N2
0.8245	-250.0	500.0	52	0.5399	0.0245	11046				
0.8243	500.0	0.0	82	0.5399	0.0243	10201				
0.8235	500.0	-250.0	277	0.5399	0.0235	10201	0.0000	11046		
0.8207	-250.0	250.0	291	0.5399	0.0207	11046				
0.8193	-500.0	500.0	148	0.8099	0.0178	11046	0.0015	10201		
0.8164	-500.0	250.0	219	0.8099	0.0164	11046				
0.8152	250.0	0.0	151	0.8099	0.0152	10201				
0.8145	250.0	-250.0	214	0.8099	0.0145	10201				
0.8134	750.0	-250.0	330	0.8099	0.0121	10201	0.0013	11046		
0.8123	750.0	0.0	28	0.8099	0.0123	10201				

- 82 - 19/09/2024 "АТ «НАЕК «Евергоатом»" ЕОЛ 2000 [h] v4.0, Ліцензія №133565957



Група сумації № 11002 Карта-схема Н=2.00 м

### **APPENDIX D Reference on climatic characteristics**

	Хмелы	ницький	обласни	ий центр	з гід	ромете	орологі	i	
$\boxtimes$	29000, м Хма вул.Грушев кімн.204	сльницький ського,87,	Ť	T.: 79-45-88 T.: 65-64-92 T/\$\overline{this}: 76-27-56	5		e-mail: p	gdhmel@n	neteo.gov.u
<u>№01-49</u>	1 <i>86</i> від <i>о</i>	18. 03.	<u>2018p</u>	Дирен ТОВ Непр	ктору Наук яхініі	ово-вн і Н.Ф.	робнича	а фірма «	KJIIFOC
2				*					
Н аероло промис що р	адаємо гічної слового і озташов	клімат станції майданч аний :	ичну х Шепе ика ВП « за адре	арактери тівка, д Хмельни есою: Х	стику дані цька мельн	за якої АЕС» Д іицька	даними репрез ДП НАЕ облас	спосте ентативн К «Енерг ть, м.Н	режень і для оатом», етішин,
вул.Ен Коефіт	ергетикі пент. які	в, 20 : ай залеж	ить від с	тратифік	ації ат	мосфе	ри — 200		
Коєфі	цієнт рел	ьєфу міс	цевості	- 1.0	•	1	•		
Повтор	юваніст	ь (%) на	прямку в	ітру та ш	тилю	/роза в	ітрів/:		
TIH ]	ПнС С	ПдС	Пд Пд	3 3 I	ІнЗ	Штил	6		
10.8	5.0 9.4	13.9	16.6 11.	7 20.8	11.8	13.1		4.000	
Середи	ня швиді	сість віт	у за рік	- 3.1 M/CE	K	a avoi i	илалас 4	5% - 7 M/	Per
Серели	acts Brij	TeMpena	тура пов	15  hepess	⁰ тепл	A NKOI V	лладае.	770 - 7 MI/	JOK
Макси	мальна т	темпера	ура пові	тря за ли	тень -	найтег	іліший м	иісяць :	
- cepez	цня – 23.	0 ⁰ тепла	JF	-1				and a free state of the second	
- cepez	цня з абс	олютних	$x - 29.6^{\circ}$	тепла		÷			
- абсол	пютна –	35.9 [°] тег	ла				• •	•	
Мінім	альна те	мперату	ра повітр	ля за січен	IP — На	айхоло,	дніший	місяць:	
- cepez	цня — 8.7	морозу	10 10	MODORN					
- afcor	цня з аос пютна –	33.6 ⁰ мо	no3V	Moposy					
Серед Серед Серед	ня темпе ня темпе ня темпе	ратура 1 ратура 1 пература	ювітря н ювітря н за шіс	айбільш з айбільш з сть самиз	колод геплоі х теп	ного м го міся лих м	ісяця (січ ця (липе ісяців ]	чень): - 5 нь): - 17 року (кв	.7 ⁰ моро: .7 ⁰ тепла ітень-
вересе	ень) — 14	.3° тепла		on ury you	10 JULI	v wingt	ip nory	WORTHUL	_
берез	$H_{\rm H} = 0$		v	самих хол	юдни	AMICH	цв року	MOBICIB	
Серед	ня місяч	на і річн	, а кількіс	ть опадів	B MM	-			
I	II III	IУ [′] У	УІ	УП У	III I	X X	XI	XII Pik	
42	37 34	55 69	0 100	108 7	6	59 39	44	51 714	HO 3
Серед Серед	ня кільк ня кільк	ість опа, ість опа	дів за хо. дів за те	лодний пе плий пері	еріод ( од (ІУ	( XI-III (-X) – 5	) - 208 м 606 м́м.	OPNEL	налог
Conforded Name		Довід Нача	ка дійсн альник	а тільки д	ля ТС И К	B HBQ	Þ «Лігос ювський	« JIIFC	)C» Min Billione
вик.Маз	ур тел.76-27	-56 0						1800	100
## APPENDIX E Information on background pollutant concentration



## УКРАЇНА

## ХМЕЛЬНИЦЬКА ОБЛАСНА ДЕРЖАВНА АДМІНІСТРАЦІЯ ХМЕЛЬНИЦЬКА ОБЛАСНА ВІЙСЬКОВА АДМІНІСТРАЦІЯ ДЕПАРТАМЕНТ ПРИРОДНИХ РЕСУРСІВ ТА ЕКОЛОГІЇ

вул. Свободи, 70 м. Хмельницький, 29000

тел.: 61-85-08, E-mail: 42814282@mail.gov.ua Код ЄДРПОУ: 42814282

На № 62-10/2-2432/9362 від 02.06.2023 року

## ВЕЛИЧИНИ ФОНОВИХ КОНЦЕНТРАЦІЙ ЗАБРУДНЮВАЛЬНИХ РЕЧОВИН (визначені розрахунковим методом)

Департамент природних ресурсів та екології Хмельницької обласної державної адміністрації

(назва організації, яка видає величний фонових концентрацій)

## м. Нетішин, Шенетівський район, Хмельницька область (населений пункт, назва)

(населении пункт, на

Організація, що запитує величини фонових концентрацій ДП «Національна атомна енергогеперуюча компанія «Енергоатом» в особі відокремленого підрозділу Хмельницька АЕС

(пазва)

Підприємство, для якого встановлюються величини фонових концентрацій ДП «Національна атомна енергогенеруюча компанія «Епергоатом» в особі відокремленого

## підрозділу Хмельницька АЕС (проводить реконструкцію)

(назва, зазначити:діюче, проводить реконструкцію, пове будівництво)

Перелік забруднювальних речовин, для яких установлюються величини фонових концентрацій, а також речовин, які мають властивість сумації шкідливого впливу: бутан, пропан, метил меркаптан, азоту діоксид, водень хлористий за молекулою HCI, вуглецю оксид, сірчистий ангідрид, метан, пил недифенційований за складом, бенз(а)пірен, суспендовані тверді частинки (сажа), вуглеводні граничні C12-C19, бензин, аерозоль лакофарбових матеріалів, уайт-спірит, залізо, марганець, фториди добре розчинні неорганічнія, фториди погано розчинні неорганічні, фтористі газоподіблі сполуки, пил неорганічний з вмістом діоксину кремнію, ксилол, туолол, кремнію діоксид аморфний, титану діоксид, бутилацетат, ацетон.

Величини фонових концентрацій визначено з урахуванням вкладу підприємства, для якого вони запитуються: так

За результатами розрахунків (для населених пунктів до 50 тис. чол.) установлюються такі величини фонових концентрацій забрудновальних речовин:

ВП «Хмель»ицька АЕС» ДП «НАЕК «Енергоатом»		
ДАТА	Вхідний №	
31.07.2023	14143	



**JOKYMEHT CED** 

Департаменту природних ресурсів та екології Хмельницької ОДА 06-101-1820/23 від 12.06.2023

Підписувач <mark>КЛІПАЦЬКА ІРИНА АНАТОЛІЇВНА</mark> Сертифікат <u>248197DDFAB977E504000000EB67DC0071D21B04</u> Дійсний з <u>01.05.2023 8:38:36</u> по <u>30.04.2024 23:59:59</u>

Вуглецю оксид	0,4	0,4	0,4	0,4	0,4	0,4	0,4	0,4
 Діоксид азоту	0,008	0,008	0,008	0,008	0,008	0,008	0,008	0,008
Речовини у вигляді твердих суспендованих частинок	0,05	0,05	0,05	0,05	0,05	0,05	0,05	0,05
Водень хлористий (соляна кислота) за молекулою HCl	0,08	0,08	0,08	0,08	0,08	0,08	0,08	0,08
Сажа	0,06	0,06	0,06	0,06	0,06	0,06	0,06	0,06
Ксилол	0,08	0,08	0,08	0,08	0,08	0,08	0,08	0,08
Толуол	0,24	0,24	0,24	0,24	0,24	0,24	0,24	0,24
Пропан	26	26	26	26	26	26	26	26
 Бутан	80	80	80	80	80	80	80	80
Уайт-спірит	0,4	0,4	0,4	0,4	0,4	0,4	0,4	0,4
метилмеркапта н	0,00004	0,0000 4	0,0000 4	0,000 04	0,00004	0,0000 4	0,0000 4	0,00004
Метан	20	20	20	20	20	20	20	20
Ангідрид сірчистий	0,02	0,02	0,02	0,02	0,02	0,02	0,02	0,02
Бензин (нафтовий, малосірчастий у перерахунку на вуглець)	2,0	2,0	2,0	2,0	2,0	2,0	2,0	2,0
Аерозоль лакофарбуваль них матеріалів	0,04	0,04	0,04	0,04	0,04	0,04	0,04	0,04
Метан	20	20	20	20	20	20	20	20
Толуол	0,24	0,24	0,24	0,24	0,24	0,24	0,24	0,24
Бенз(а)пірен	0,04мкг на 100 м ³	0,04мк г на 100 м ³	0,04мк г на 100 м ³	0,04м кг на 100 м ³	0,04мкг на 100 м ³	0,04мк г на 100 м ³	0,04мк г на 100 м ³	0,04мкг на 100 м ³
Кремнію діоксид аморфний (Аеросил-175)	0,008	0,008	0,008	0,008	0,008	0,008	0,008	0,008
Етантіол (етилмеркапта н)	1,2*10- 6	1,2*10- 6	1,2*10 -6	1,2*1 0-6	1,2*10-6	1,2*10- 6	1,2*10- 6	1,2*10- 6

Вуглеводні насичені С12- С19	0,4	0,4	0,4	0,4	0,4	0,4	0,4	0,4
Залізо	0,016	0,016	0,016	0,016	0,016	0,016	0,016	0,016
Марганець і його сполуки (у пере- рахунку на діоксид марганцю)	0,004	0,004	0,004	0,004	0,004	0,004	0,004	0,004
Фториди добре розчинні неорганічні	0,012	0,012	0,012	0,012	0,012	0,012	0,012	0,012
Фториди погано розчинні неорганічні	0,08	0,08	0,08	0,08	0,08	0,08	0,08	0,08
Фтористі газоподібні сполуки (фтористий водень, чотирифто- ристий кремній) (у перерахунку на фтор)	0,008	0,008	0,008	0,008	0,008	0,008	0,008	0,008
Пил неорганічний, з вмістом діоксиду кремнію в %: - 70-20 (шамот, цемент і ін.)	0,12	0,12	0,12	0,12	0,12	0,12	0,12	0,12

Величини фонових концентрацій розраховані відповідно до "Порядку визначення величин фонових концентрацій забруднювальних речовин в атмосферному повітрі" затвердженого наказом Міністерства екології та природних ресурсів України №286 від 30 липня 2001 року та зареєстрованого в міністерстві юстиції України 15 серпня 2001 року за №700/5891.

Термін дії величин фонових концентрацій три роки.

Виконувач обов'язків директора Департаменту

Ірина КЛІПАЦЬКА

Олена СНІГУРСЬКА, 656959

# APPENDIX F Information on treatment facilities



вул. Енергетиків 20, м. Нетішин, Хмельницька обл., Україна 30100 ЄДРПОУ 21313677 Телефон: +38 (03842) 90262. Факс: +38 (03842) 90260 Комутатор: +38 (03842) 40400 E-mail: <u>office(</u>

## Філія «Відокремлений підрозділ «ХМЕЛЬНИЦЬКА АЕС»

Affiliate - "Separate Subdivision "Khmelnytskyi NPP"

uska οδπ., 20 Enerhetykiv St., Netishyn, Khmelnytska Oblast, Ukralne, 30100 3842) 90260 Office: +38 (03842) 90262. Fax: +38 (03842) 90260 E-mail: office@khnpp.atom.gov.ua Switchboard: +38 (03842) 40400

Т. в. о. генерального директора

філії «ВП «Атомпроектінжиніринг»

Тетяні АМОСОВІЙ

м. Київ

office_api@direkcy.atom.gov.ua

Про надання інформації по очисним спорудам ГПС

Шановна Тетяно Володимирівно!

Для розробки розподілу «Скидання стічних вод» ОВД енергоблоків №5, 6 надаємо таку інформацію:

- проєктна потужність 20 000 м³ за добу;
- фактична потужність 8 000 м³ за добу;
- фактичні обсяги господарсько-побутових стоків, що скидалися за 2023 рік:
- від м. Нетішин 2403763 м³;
- від енергоблоків № 1, 2 222556 м³.

Використання існуючих очисних споруд господарсько-побутових стоків для потреб енергоблоків №5, 6 залежить від обсягів скидання стоків з цих споруд і розширення м. Нетішин.

З повагою

Заступник головного інженера з загальностанційних об'єктів

Андрій ДАНИЛЮК

КГ Вячеслав Жураковський 6 27 78



Документ АСКВД філії «ВП ХАЕС». Підписаний КЕП. Підписувач: Данилюк Андрій Васильович Сертифікат 382367105294АР9704000000АD540600710ADB01 Заборовено розповсюджения без дозволу філі «ВП ХАЕС»

№86-04-1983/8868 від 24.04.2024



# APPENDIX G Certificate of measurement capabilities approval



# APPENDIX H Site maps

H.1 Layout of buildings and structures



# H.2 Plant layout



УМОВНЕ ПОЗНАЧЕННЯ



існуючий промлайданник ХАЕС

- проммайданчик енергоблоків NR5 та NR6

# H.3 Location of emission sources





# H.4 Plant location with the indicated sanitary protection area

# H.5 Scheme of the KhNPP 30-km area



# H.6 Location scheme of pollutant sources



## **APPENDIX J** Information on the dam





Відокремлений підрозділ «ХМЕЛЬНИЦЬКА АЕС»

NAECOKINPR

National Nuclear Energy Generating Company «Energoatom» SE «Khmelnytskyi NPP»

вул. Енергетнків 20, м. Нетішин, Хмальницька обл., Україна 30100 єДРПОУ 21313677 Тепефон: +38 (03642) 90262. Факс: +38 (03642) 90260 Комутатор: +38 (03842) 40400 Е-mail: <u>office@k</u>

цька обл., 20 Enerhetykiv St., Netishyn, Khmeinytska Oblast, 17 Ukraine, 30100 13842) 90260 Office: +38 (03842) 90262. Fax: +38 (03842) 90260 E-mail: <u>office@khnpp.atom.gov.ua</u> Switchboard: +38 (03842) 40400

> Т. в. о. генерального директора ВП «Атомпроектінжиніринг» Тетяні АМОСОВІЙ вул. Сім'ї Правових, 6 м. Київ, 01032

Про надання інформації

Шановна Тетяно Володимирівно!

У відповідь на Ваш лист від 07.09.2023 №02-46/4141 повідомляємо, що при розробці ТЕО блоків №3, 4 були передбачені ремонтні роботи з відновлення фільтра дренажного каналу і низового укосу земляної греблі з урахуванням зрізанням кущів. Робота з відновлення похилого дренажу буде включена в обсяги ремонтних робіт на наступні періоди, щодо бетонного кріплення укосу земляної греблі деякі ділянки з ремонту були виконані в 2023 році. По низовому укосу періодично здійснюється вирубка кущів та дерев, що не призводить до руйнування відкосу земляної греблі.

Також на Ваше прохання надаемо дані результатів рівня води в п'єзометрах за 2021-2022 роки та матеріали роботи «Уточнення допустимих експлуатаційних значень положення кривої депресії у земляній греблі Хмельницької АЕС УДК 626.862.3:532.001», що була виконана «Національним університетом водного господарства та природокористування (НУВГП)», м. Рівне у 2014 році.

Додаток: Таблиці ресстрації вимірювання грунтових вод у свердловина-п'єзометрах за 2021-2022 роки та положення кривої депресії у земляній греблі Хмельницької АЕС УДК 626.862.3:532.001», що була виконана «Національним університетом водного господарства та природокористування (НУВГП)», м. Рівне у 2014 році на 95 аркушах в 1 примірнику.

#### 3 повагою

В. о. заступн	ика	генерального	о директора з	капітального		Ацарій СА	воченко
будівництва	*	начальник	управління	капітального		distant of	10.00
будівництва				EXT. He	43	.92	110

K-the opposituality

ГТЦ Віктор Турицький 6 20 21



0910 2023



Документ СЕД АСКВД ВП ХАЕС, підписаннії КЕП, Підписувач: Савоченко Андрій Олександрович Сертифікат 382367105294АР9704000000С0Е16А00DCA44801 Таборового ринокозденной без шикогу IID ХАЕС

228-05-3529/16515 BUL 04.10.2023

62621

481

# APPENDIX K Comments and proposals on the planned activity, scope of research from the affected countries



## МІНІСТЕРСТВО ЗАХИСТУ ДОВКІЛЛЯ ТА ПРИРОДНИХ РЕСУРСІВ УКРАЇНИ

(MIHJOBKLIJIS)

Департамент екологічної оцінки вул. Митрополита Васпля Липківського, 35, м. Київ, 03035, 206-31-40, E-mail: info@mepr.gov.ua

Ha Ne _____

АКЦІОНЕРНЕ ТОВАРИСТВО «НАЦІОНАЛЬНА АТОМНА ЕНЕРГОГЕНЕРУЮЧА КОМПАНІЯ «ЕНЕРГОАТОМ» 01032, місто Київ, вул. Назарівська, будинок 3

На додаток до листа Департаменту екологічної оцінки Міністерства захисту довкілля та природних ресурсів України від 17.04.2024 № 21/21-03/1685-24 (реєстраційний номер справи 5333 у Єдиному реєстрі з оцінки впливу на довкілля), надаємо зауваження і пропозиції до планованої діяльності, обсягу досліджень та рівня деталізації інформації, що підлятає включенню до звіту з оцінки впливу на довкілля АТ «НАЕК «ЕНЕРГОАТОМ» щодо спорудження енергоблоків № 5 і № 6 на майданчику Хмельницької АЕС із застосуванням технічних характеристик реакторної установки типу АР1000 компанії Westinghouse Electric Company з метою подальшої експлуатації та вироблення електроенергії, отримані під час транскордонних консультацій від Австрійської Республіки, Словацької Республіки, Республіки Молдова, Республіки Польща та Румунської сторони, відповідно до вимог Конвенції про оцінку впливу на навколишнє середовище у транскордонному контексті (Конвенція Еспо).

Додатки: на 29 арк. в 1 прим.

Директор Департаменту

Марина ШИМКУС

Інна Теличко 206 31 40



Милловкітия №21/21-03/5346-24 від 05.11.2024 №21/21-03/5346-24 від 05.11.2024 КПП: Шновус М. О. 05.11.2024 16:01 5 97А.092883581:C00304000000E8FC340007B0D600 Серпяфікат дійсний з 16.07.2024 17:52 до 16.07.2026 17:52 Federal Ministry Republic of Austria Climate Action, Environment, Energy, Mobility, Innovation and Technology

Ministry of Environmental Protection and Natural Resources Ms Olena Kramarenko Deputy Minister 35 Mytropolyta V. Lypkivskogko Street 03035 Kyiv Ukraine bmk.gv.at

BMK - Department V/11 (Plant-related Environmental Protection, Environmental Assessment & Air Pollution Control) v11@bmk.gv.at

Ursula Platzer-Schneider Official in Charge

<u>ursula.platzer@bmk.gv.at</u> +43 1 71162 612115 Address: Stubenbastel 5, 1010 Vienna

E-mail replies should be sent to the above email address, quoting the reference number

Reference number: 2024-0.469.188

Vienna, 25.06.2024

## Espoo, NPP Khmelnitsky 5 and 6, AT participation

Dear Deputy Minister Kramarenko,

Thank you for the notification according to Art. 3 Espoo Convention regarding the construction of two new modern Western AP-1000 units at the Khmelnitsky nuclear power plant site. The notification arrived at the Federal Ministry for Climate Action, Environment, Energy, Mobility, Innovation and Technology on 18 June 2024.

Austria will participate in the EIA according to Art. 3 para 3 Espoo Convention for these activities since the construction and operation of the project may have significant adverse effects on Austria, especially in case of severe accidents.

Therefore, Austria kindly asks the Ukraine to send the environmental impact assessment documentation according to Art. 4 Espoo Convention in German or English.

Thank you for your co-operation.

Yours sincerely,

On behalf of the Federal Minister

Dr. Ursula Platzer-Schneider

UB Мидовкілля №26269/7/24 віл 28.06.2024 арк. I

V OSTA	Hinweis	Dieses Dokument wurde amtssigniert.
UBL TE ARE	Datum	2024-06-25T10:49:15+02:00
BUNDESMINISTERIUM FÜR RUMASCHUTZ, UMWELT, ENERGIE, MOBILITÄT, INNOVATION UND TECHNOLOGIE AMTSSIGNATUR	Seriennummer	1871969199
	Aussteller-Zertifikat	CN=a-sign-corporate-05,OU=a-sign-corporate-05,O=A-Trust Ges. f. Sicherheitssysteme im elektr. Datenverkehr GmbH,C=AT
	Prüfinformation	Informationen zur Prüfung des elektronischen Siegels bzw. der elektronischen Signatur finden Sie unter: https://www.signaturpruefung.gv.at/



Section of Environmental Impact Assessment Department of Environmental Impact Assessment II.

> Ministry of environmental protection and natural resources of Ukraine 35 Mytropolyta Vasylya Lypkivskogo Str. 030 35 Kyiv Ukraine

Your letter/from day 6139/22-200-83510 18. 06. 2024 Our number 13078/2024-11.1.2/kp 50282/2024 Contact information Ing. K. Preinerova 02/5956 2196 Bratislava 18. July 2024

## Subject

Assessment of the Impact of Proposed Activities – National Nuclear Energy Generating Company – Statement on participation in the Environmental Impact Assessment process

On June 19, 2024, the Ministry of the Environment of the Slovak Republic, Section of Environmental Impact Assessment, Department of Environmental Impact Assessment - currently the Ministry of the Environment of the Slovak Republic, Section of Environmental Impact Assessment, Department of Environmental Impact Assessment II. (hereinafter referred to as "MoE SR") received letter No. 6139/22-200-83510 dated June 18, 2024, from the Ukrainian Embassy, including a copy of a letter from the Ministry of Environmental Protection and Natural Resources of Ukraine. This letter informed us, as an affected party under the provisions of the Convention on Environmental Impact Assessment in a Transboundary Context (Espoo Convention) and Article 9 paragraphs 1 and 2 of Directive 2011/92/EU of the European Parliament and Council of December 13, 2011, on the assessment of the effects of certain public and private projects on the environment (codified version), about the planned implementation of the proposed activity "Assessment of the Impact of Proposed Activities – National Nuclear Energy Generating Company for Fuel Production" (hereinafter referred to as "proposed activity").

According to the report on the proposed activity, which is subject to Environmental Impact Assessment, the proposed activity involves the construction of the 5th and 6th units of the Khmelnytskyi NPP with installed AP1000 III+ generation pressurized water reactors (PWR) with passive safety systems and a capacity of approximately 1100 MWe by Westinghouse Electric Company for the purpose of electricity production. The projected lifespan of the reactors is 60 years. The operator of both planned units of the Khmelnytskyi NPP is to be the National Nuclear Energy Generating Company (Energoatom). The location of the proposed activity is in the Khmelnytskyi region, the city of Netishyn, approximately 320 km from the border of the Slovak Republic with Ukraine. The area of both units' site is 51.1 hectares with a construction area of 158.2 thousand m². The alternative location No. 1 is in the Rivne region, the city of Varash, Rivne NPP site, approximately 350 km from the border of the Slovak Republic with Ukraine. Alternative location No. 2 is in the Mykolaiv region, the city of Yuzhnoukrainsk, Yuzhnoukrainsk NPP site, approximately 650 km from the border of the Slovak Republic with Ukraine. The report states that the implementation of the project has nationwide significance and involves a large number of local suppliers, thereby creating new job opportunities, increasing production capacities, and promoting economic development in Ukraine. The report indicates that no urbanistic restrictions are necessary for the implementation of the project at the Khmelnytskyi NPP site. The project is to implement a defense-in-depth system that will maintain important safety system functions and ensure nuclear safety and radiation protection. Both planned units of the Khmelnytskyi NPP are to be constructed in accordance with the environmental legislation effective in Ukraine. The report further provides an overview of key technological solutions and protective measures to minimize adverse environmental impacts ensured by the planned project (e.g., measures for the conservation of natural resources; architectural, structural, and planning solutions; measures to minimize the radiation impact on the environment; measures to minimize non-radiation environmental impacts; environmental monitoring systems – radiation situation monitoring system in the NPP site, supervised and controlled areas; air monitoring systems; surface and groundwater monitoring systems; surface water monitoring system; building foundation and structure monitoring system in the NPP site). These solutions and measures are intended to ensure environmental protection during construction.

Neighboring states whose environment may potentially be adversely affected by the transboundary impact of the proposed activity are Poland, Slovakia, Romania, Hungary, Moldova, and Austria. Of all types of transboundary impacts, only ionizing radiation is considered to be significant transboundary impact according to the report. Under normal construction and operation conditions of both planned units of the Khmelnytskyi NPP, it is expected that the radiation impact on the population and the environment of neighboring states will be negligible compared to existing background impacts. Assessments of the transboundary transfer consequences of ionizing radiation for maximum project accidents and extended project condition accidents are to be carried out using radionuclide atmospheric transfer models.

In accordance with § 51(a) of Act No. 24/2006 Coll. on Environmental Impact Assessment and on Amendments and Supplements to Certain Acts, as amended (hereinafter referred to as "the EIA Act"), taking into account the provisions of Act No. 541/2004 Coll. on the Peaceful Use of Nuclear Energy (Atomic Act) and on Amendments and Supplements to Certain Acts, as amended, the MoE SR requested an opinion/expression from the competent authority in matters of state supervision over the nuclear safety of nuclear installations, which is the Nuclear Regulatory Authority of the Slovak Republic (hereinafter referred to as "NRA SR"). The NRA SR assessed the message about the proposed activity as the competent authority, ensuring the assessment of nuclear energy utilization program intentions and the quality of selected nuclear equipment and instruments, or the Slovak Republic's obligations arising from international treaties regarding the nuclear safety of nuclear installations and the handling of nuclear materials.

Based on the NRA SR's statement dated June 9, 2024, and after summarizing all information, the MoE SR draws attention to the fact that it is currently not possible to qualitatively evaluate the potential negative impacts of the proposed activity on the environment in the Slovak Republic territory due to the lack of relevant information from this perspective in the message about proposed activity. The MoE SR welcomes the mention in the report of future analyses of radiation impacts from maximum project accidents and extended project condition accidents using radionuclide atmospheric transfer modeling. However, the MoE SR is unclear on what basis the Ukrainian side assumes that under normal construction and operation conditions of the planned project, the radiation impact on the population and the environment of neighboring states will be negligible compared to existing background impacts. Accordingly, the MoE SR notes that the report does not specify the nuclear safety standards (safety standards, guidelines, etc.) that will guarantee the elimination of potential radiation impacts of the proposed activity on the environment in neighboring states' territories. The MoE SR expects the documentation from the originator side/report on the assessment of the proposed activity's impact on the environment to include a

detailed description of the methodology for analyzing the proposed activity's impacts on the environment in surrounding states' territories, including the selection method for accident scenarios under extended project conditions and the radiation impacts of the proposed activity at the nearest points of the Slovak Republic territory from all three considered locations. The analysis results of the proposed activity's environmental impacts on neighboring states' territories, especially within the scenarios of maximum project accidents and extended project condition accidents, will be necessary to assess their relevance and significance concerning the environment in the Slovak Republic territory.

In accordance with § 51(e) of the EIA Act and Article 3(3) of the Espoo Convention, following your letter dated June 19, 2024, the MoE SR expresses its interest in participating in the transboundary environmental impact assessment process of the proposed activity.

At the same time, the MoE SR wishes to highlight the high level of previous cooperation with Ukraine and expresses hope in maintaining this effort for environmental protection.

Ing. Katarína Jankovičová general director of section

## M I N I S T E R U L MEDIULUI AL REPUBLICII MOLDOVA



## M I N I S T R Y OF ENVIRONMENT OF THE REPUBLIC OF MOLDOVA

MD 2004, mun.Chişinău, bd Ştefan cel Mare și Sfânt,162, tel. 022 20 45 87, e-mail: cancelaria@mediu.gov.md

______Nr. <u>04-07/1697</u> La nr. <u>25/5-21/4370-24</u> din <u>02.04.2024</u>_____

Г

Mr. Ruslan Strilets Minister Ministry of Environmental Protection and Natural Resourses of Ukraine

٦

Dear Mr. Strilets,

The Ministry of Environment of the Republic of Moldova expresses its high consideration to the Ministry of Environmental Protection and Natural Resources of Ukraine and has the honour to inform you of the following.

On 04.04.2024 the Ministry of Environment received by e-mail the notification of the Ministry of Environmental Protection and Natural Resources of Ukraine on the commencement of the environmental impact assessment procedure for the project "Construction of power units No. 5 and No. 6 at the site of the Khmelnytsky NPP using the technical characteristics of the Westinghouse Electric Company AP 1000 reactor for further operation and power generation" (location Netishyn, Khmelnytskyi region, Ukraine), in accordance with Article 3 of the Convention on Environmental Impact Assessment in a Transboundary Context (Espoo Convention).

On 09.04.2024 the notification in English and Ukrainian for the project "Construction of power units No. 5 and No. 6 at the site of the Khmelnytsky NPP using the technical characteristics of the Westinghouse Electric Company AP 1000 reactor for further operation and power generation" (location Netishyn, Khmelnytskyi region, Ukraine), was sent by email to the Environmental Agency (competent authority) with the competence of coordination of environmental impact assessment procedures, including in transboundary context.

In this context, according to the provisions of art. 3 para. (8) of the Espoo Convention and art. 16 of the Law no. 86/2014 on environmental impact assessment, the Environment Agency sent the Notification of the mentioned project for consultation, examination and submission of comments/proposals to the relevant identified national authorities. At the same time, the Notification of the mentioned project was published for public consultation on the official website at the following link: <u>https://am.gov.md/ro/node/1039</u>.

As a result of the examination of the information presented in the Notification for the project "Construction of power units No. 5 and No. 6 at the site of the Khmelnytsky NPP using the technical characteristics of the Westinghouse Electric Company AP 1000 reactor for further operation and power generation" (location Netishyn, Khmelnytskyi region, Ukraine), the opinions of the national authorities, it is found that the project envisages on the existing industrial site of the Khmelnytsky NPP the construction of reactors No. 5 and No. 6 with a thermal capacity of 3400 MW for the generation of electricity for the purpose of strengthening the energy independence of the State by supplying the population and industry with electricity and heat. Is provided modernizing the Soviet-type reactors using the technical characteristics of the Westinghouse Electric Company AP 1000 two-loop pressurized water reactor with 60 years lifetime.

The distance measured in a straight line from the boundary of the Khmelnytsky NPP site to the nearest point of the border is about 200 km from the Republic of Moldova. The Republic of Moldova may be negatively affected by the effects of a transboundary nuclear accident (radiation).

According to the Notification, under normal condition of construction and operation, the radiation impact on the population and environment of neighboring countries is predicting to be negligible compared to the existing background impacts. Assessment of the consequences of transboundary radioactivity transfer, including the analysis of the potential radiological impact on the health of the population will be performed in the environmental assessment report.

In this context, we inform that the Republic of Moldova will participate in the transboundary environmental impact assessment procedure for the project *"Construction of power units No. 5 and No. 6 at the site of the Khmelnytsky NPP using the technical characteristics of the Westinghouse Electric Company AP 1000 reactor for further operation and power generation"* (location Netishyn, Khmelnytskyi region, Ukraine).

We take this opportunity to express our high regards and readiness for fruitful cooperation in the future.

Minister

Digitally signed by Lazarencu Sergiu Date: 2024.06.27 11:12:18 EEST Reason: MoldSign Signature Location: Moldova



Sergiu LAZARENCU

Mariana Petreanu +37322204599 mariana.petreanu@mediu.gov.md



DEPARTAMENT OCEN ODDZIAŁYWANIA NA ŚRODOWISKO

### DOOS-TSOOS.442.12.2024.JA8

Warsaw, 10 May 2024

Ms. Olena Anatoliivna Hrytsak Deputy Director of the Ecological Assessment Department Head of the Environmental Impact Assessment Section Ministry of Environmental Protection and Natural Resources of Ukraine

## email: OVD@mepr.gov.ua

Subject matter: position of the Polish authorities concerning the scope of the environmental impact assessment documentation for the project entitled "Construction of power units no. 5 and 6 at the Khmelnytskyi Nuclear Power Plant using the technical parameters of the Westinghouse Electric Company AP1000 reactor for further operation and electricity generation".

#### Dear Madam,

With reference to the letter of 26 April 2024, I hereby present the positions of the Polish authorities concerning the scope of the environmental impact assessment documentation for the project entitled "Construction of power units no. 5 and 6 at the Khmelnytskyl Nuclear Power Plant using the technical parameters of the Westinghouse Electric Company AP1000 reactor for further operation and electricity generation" (hereinafter: Khmelnytskyi NPP units no. 5 and 6).

Below, I provide a summary of all the observations, and I attach the positions in question to a letter requesting their consideration when preparing the environmental impact assessment documentation for Khmelnytskyi NPP units no. 5 and 6 pursuant to Article 4 of the Espoo Convention¹.

¹ The Convention on Environmental Impact Assessment in a Transboundary Context, signed in Espoo on 25 February 1991.

Aleje Jerozolimskie 136, 02-305 Warszawa, tel.: 22 310-67-00, kancelaria@gdos.gov.pl, www.gov.pl/ados

The Regional Director for Environmental Protection in Lublin identified the following matters as necessary to be considered in the preparation of the documentation.

- Presentation of the planned technical, technological, and organisational solutions to counteract the possibility of a design and beyond design-basis accident, aimed at preventing the exceedance of the acceptable contamination level and the ecological impact;
- Presentation of information on the impact of a major beyond design-basis accident or a terrorist situation on the health and life of the Polish public;
- A summary and conclusion of the submitted analyses in a manner accessible to the public, in relation to its participation in the transboundary procedure;
- Presentation of procedures for informing the Polish Party about the current radiological situation and in case of emergencies;
- Presentation of information on the management of radioactive waste and spent nuclear fuel;
- Presentation of cumulative impacts, i.e. taking into account the possibilities of impact overlaps.

The Regional Director for Environmental Protection in Rzeszów identified the following matters as necessary to be considered in the preparation of the documentation.

- Presentation of an assessment of the transboundary radioactive contamination, with assumptions, and:
  - taking into account the cumulative impacts of the planned units no. 5 and 6 and the other units in operation at a particular time at the designated site;
  - based on the current meteorological conditions for the exposed countries;
  - under normal operating conditions and in case of an accident within the nuclear power plant area.
- Presentation of possible emergency and accident scenarios;
- Presentation of the planned safeguards to be applied at the nuclear installations to prevent accidents and discharge of hazardous substances and pollutants into the environment;
- 4. Presentation of the planned radiological monitoring system;
- 5. Presentation of the planned technology for the storage and management of spent nuclear fuel.

The Polish Geological Institute – National Research Institute identified the following matters as necessary to be presented in the documentation.

 The anticipated impact of the planned project on the hydrodynamic system of the active groundwater exchange zone;

The anticipated impact of the planned project on the temperature and chemical composition of the active groundwater exchange zone;

The anticipated impact of the planned project on chemical denudation in the context of the stability of the ground under all power plant installations;

 Anticipated method and location of spent nuclear fuel and radioactive waste treatment and management.

The National Atomic Energy Agency identified the following matters as necessary to be considered in the prepared documentation.

- The impact of the construction of nuclear power units no. 5 and 6 on the safe operation of units no. 1 and 2 and uncompleted units no. 3 and 4, including the impact on the emergency plan measures;
- Interoperability with other power units of the Khmelnytskyi Nuclear Power Plant after the commissioning of units no. 5 and 6, including common systems for units no. 5 and 6 and units no. 1, 2, 3, and 4;
- Organisation of the environmental radiation monitoring in the area surrounding the Khmelnytskyi Nuclear Power Plant during the construction of units no. 5 and 6 and during their operation;
- A description of the extent to which units no. 5 and 6 of the Khmelnytskyi NPP will be similar to the reference AP1000 unit and the differences between them. What will be the impact of the Ukrainian nuclear safety and radiological protection requirements on design changes;
- 5. Modifications in the national power and power plant system in connection with the operation of units no. 5 and 6. The manner of using heat from units no. 5 and 6, as well as the impact on the current heating system of the power plant;
- The impact of the current situation in Ukraine on the construction and operation of units no. 5 and 6 and the AP1000 reactor design itself in relation to the reference design;
- The management of spent nuclear fuel and radioactive waste from units no. 5 and 6 and the impact of the increase in spent nuclear fuel and radioactive waste volumes on the national spent nuclear fuel and radioactive waste management plan;
- 8. The types and activity of radionuclides discharged into the environment, their physical and chemical form, the methods and routes and rates of discharge into the environment, as well as the mechanisms of radionuclide transfer in the environment, including dispersion and accumulation mechanisms and their seasonal variability in normal conditions;

- The existing radionuclide concentration levels in the environment and their variability, and the presence of physical or chemical contaminants that may affect radionuclide transfer;
- Natural and anthropogenic environmental features that will affect the transfer of radionuclides (e.g. geological, hydrological, and meteorological conditions, vegetation or presence of reservoirs – with particular attention to features that may affect the territory of the Republic of Poland);
- 11. Detailed assumptions, including meteorological data and data on radioactive substance releases into the environment in case of a major accident, as well as the results of the radiological situation development forecasts, including a detailed assessment of potential impact on public health and the environment in the territory of the Republic of Poland;
- Structure of agriculture, exports of agricultural goods from the areas exposed to the impacts of the nuclear installation, and destinations of the agricultural exports;
- 13. Identified critical groups;
- 14. The manner in which the Polish authorities are to be informed in case of an accident at the Khmelnytskyi Nuclear Power Plant with the expected forecast of radiological conditions in the Republic of Poland, the strategies for communication with the residents of cross-border areas, including those in the border areas of the Republic of Poland, in order to ensure transparency, access to information;
- Conclusions on the increase in the cumulative environmental impact of the entire nuclear power plant after commissioning of the planned units;
- A combined impact assessment of a severe accident at all nuclear installations located at the selected site.

Furthermore, at the next stages of transboundary consultations, the Polish Party has an interest in obtaining accurate external emergency plans, as well as in access to the results of measurements as part of the environmental radiation monitoring in the area surrounding the Khmelnytskyi Nuclear Power Plant during the construction of units no. 5 and 6 and during their operation.

Yours sincerely,

Dorota Toryfter-Szumańska Deputy Director Environmental Impact Assessment Department / ~ signed digitally/

## Attachments (in Polish):

- 1. Letter from the President of the National Atomic Energy Agency of 26 April 2024.
- 2. Letter from the Regional Director for Environmental Protection in Rzeszów of 26 April 2024.
- Letter from the Regional Director for Environmental Protection in Lublin of 23 April 2024.
  Letter from the Director of the Polish Geological Institute National Research Institute of 7 May 2024.

#### CC:

- President of the National Atomic Energy Agency
  Director of the Polish Geological Institute National Research Institute 2.
- Regional Director for Environmental Protection in Lublin 3.
- 4. Regional Director for Environmental Protection in Rzeszów



DEPARTAMENT OCEN ODDZIALYWANIA NA ŚRODOWISKO

DOOS-TSOOS.442.12.2024.JA8

Warszawa, 10 maja 2024 r.

Pani Olena Anatolilvna Hrytsak Zastępca Dyrektora Departamentu Oceny Ekologicznej Kierownik Sekcji Oceny Oddziaływania na Środowlsko Ministerstwo Ochrony Środowiska i Zasobów Naturalnych Ukralny

email: OVD@mepr.gov.ua

Dotyczy: stanowiska polskich organów w sprawie zakresu dokumentacji oceny oddziaływania na środowisko dla przedsięwzięcia pn. "Budowa bloków energetycznych nr 5 i 6 na terenie Elektrowni Jądrowej Chmielnicki z wykorzystaniem parametrów technicznych reaktora Westinghouse Electric Company AP1000 do dalszej eksploatacji i wytwarzania energii elektrycznej*.

## Szanowna Pani Dyrektor,

w nawiązaniu do pisma z 26 kwietnia 2024 r. przekazuję stanowiska polskich organów dotyczące zakresu dokumentacji oceny oddziaływania na środowisko dla planowanego przedsięwzięcia pn. "Budowa bloków energetycznych nr 5 i 6 na terenie Elektrowni Jądrowej Chmielnicki z wykorzystaniem parametrów technicznych reaktora Westinghouse Electric Company AP1000 do dalszej eksploatacji i wytwarzania energii elektrycznej" (dalej: Chmielnicki EJ bloki 5 i 6).

Poniżej przedstawiam zestawienie wszystkich uwag, a przekazane stanowiska załączam do pisma z prośbą o ich uwzględnienia przy opracowywaniu dokumentacji oceny oddziaływania na środowisko dla Chmielnicki EJ bloki 5 i 6 zgodnie z art. 4 Konwencji z Espoo¹.

¹ Konwencja o ocenach oddziaływania na środowisko w kontekście transgranicznym, sporządzona w Espoo dnia 25 lutego 1991 r.

Aleje Jerozolimskie 136, 02-305 Warszawa, tol.: 22 310-67-00, kancelaria@gdas.gov.pl, www.gov.pl/gdos

Regionalny Dyrektor Ochrony Środowiska w Lublinie wskazał następujące kwestie za konieczne do uwzględnienia w przygotowywanej dokumentacji.

- Przedstawienie zaplanowanych rozwiązaniach technicznych, technologicznych i organizacyjnych mających na celu przeciwdziałanie możliwości wystąpienia awarii projektowej i pozaprojektowej, mających na celu niedopuszczenie do przekroczenia dopuszczalnego poziomu skażeń i zapobieganie ekologicznym następstwom;
- Przedstawienie informacji na temat wpływu poważnej awarii pozadprojektowej lub sytuacji o charakterze terrorystycznym na zdrowie i życie społeczeństwa polskiego;
- Podsumowane i skonkludowane w sposób przystępny dla społeczeństwa przedłożonych analiz, w związku z jego udziałem w procedurze transgranicznej;
- Przedstawienie procedur informowania Strony polskiej o aktualnej sytuacji radiologicznej oraz w przypadkach wystąpienia sytuacji awaryjnych;
- Przedstawienie informacji o sposobie postępowania z odpadami promieniotwórczymi i wypalonym paliwem jądrowym;
- Przedstawienie oddziaływań w ujęciu skumulowanym, tj. przy uwzględnieniu możliwości ich nakładania się.

Regionalny Dyrektor ochrony Środowiska w Rzeszowie wskazał następujące kwestie za konieczne do uwzględnienia w przygotowywanej dokumentacji.

- Przedstawienie oceny rozprzestrzenienia się transgranicznego zanieczyszczeń, wraz ze wskazaniem przyjętych założeń oraz:
  - przy uwzględnieniu oddziaływań skumulowanych planowanych bloków 5 i 6 oraz pozostałych eksploatowanych w danym czasie w wyznaczonej lokalizacji;
  - w oparciu o aktualne warunki meteorologiczne dla narażonych krajów;
  - w warunkach normalnej pracy oraz w przypadku wystąpienia awarii w obszarze elektrowni jądrowej.
- 2. Przedstawienie możliwych rozpatrywanych scenariuszy sytuacji awaryjnych i awarii;
- Przedstawienie planowanych do zastosowania zabezpieczeń obiektów jądrowych przed wystąpieniem awarii i wyciekiem niebezpiecznych substancji i zanieczyszczeń do środowiska;
- 4. Przedstawienie planowanego do zastosowania system monitoringu radiologicznego;
- Przedstawienie planowanej do zastosowania technologii magazynowania i postępowania z wypalonym paliwem jądrowym.

Państwowy Instytut Geologiczny- Państwowy Instytut Badawczy wskazał następujące kwestie za konieczne do przedstawienia w przygotowywanej dokumentacji.

1. Prognozowane oddziaływanie planowanej inwestycji na układ hydrodynamiczny strefy aktywnej wymiany wód podziemnych;

 Prognozowane oddziaływanie planowanej inwestycji na temperaturę i skład chemiczny strefy aktywnej wymiany wód podziemnych;

3. Prognozowane oddziaływanie planowanej inwestycji na denudację chemiczną w kontekście stabilności podłoża pod wszystkimi obiektami elektrowni;

Przewidywany sposób i miejsce zagospodarowania zużytego paliwa jądrowego i postępowania z odpadami radioaktywnymi.

Państwowa Agencja Atomistyki wskazała następujące kwestie za konieczne do uwzględnienia w przygotowywanej dokumentacji.

- Wpływ budowy bloków jądrowych nr 5 i 6 na bezpieczną eksploatację bloków nr 1 i 2 oraz nieukończone bloki nr 3 i 4, w tym wpływ na działania podejmowane w ramach planu postępowania awaryjnego;
- Współpraca z innymi blokami jądrowymi Elektrowni Jądrowej Chmielnicki po oddaniu bloków nr 5 i 6 do eksploatacji, w tym systemów wspólnych dla bloków 5 i 6 oraz bloków 1, 2, 3 i 4;
- Organizacja środowiskowego monitoringu radiacyjnego wokół Elektrowni Jądrowej Chmielnicki podczas budowy bloków nr 5 i 6 oraz w trakcie ich eksploatacji;
- Opis w jakim stopniu bloki nr 5 i 6 Elektrowni Jądrowej Chmielnicki będą podobne do referencyjnego bloku AP1000 i z czego wynikać będą różnice. Jaki wpływ na zmiany projektowe będą miały ukraińskie wymagania dotyczące bezpieczeństwa jądrowego i ochrony radiologicznej;
- Modyfikacje w systemie elektroenergetycznym kraju i elektrowni w związku z eksploatacją bloków nr 5 i 6. Sposób wykorzystania ciepła z bloków nr 5 i 6, a także wpływ na obecny system ciepłowniczy elektrowni;
- Wpływ obecnej sytuacji na Ukrainie na budowę i eksploatację bloków 5 i 6 oraz sam projekt reaktora AP1000 w stosunku do referencyjnego projektu;
- Postępowanie z wypalonym paliwem jądrowym i odpadami promieniotwórczymi z bloków nr 5 i 6 oraz wpływ wzrostu ilości wypalonego paliwa jądrowego i odpadów promieniotwórczych na krajowy plan postępowania z wypalonym paliwem jądrowym i odpadami promieniotwórczymi;
- Rodzaje i aktywność odprowadzanych do środowiska radionuklidów, ich forma fizyczna i chemiczna, metody i drogi oraz tempo odprowadzania do środowiska, jak również mechanizmy transferu radionuklidów w środowisku, w tym mechanizmy dyspersji i kumulacji oraz ich sezonowa zmienność w warunkach normalnych;

- Istniejące poziomy stężeń radionuklidów w środowisku i ich zmienność oraz obecność zanieczyszczeń fizycznych lub chemicznych, które mogą wpływać na transfer radionuklidów;
- Naturalne i antropogeniczne cechy środowiska, które będą miały wpływ na transfer radionuklidów (np. warunki geologiczne, hydrologiczne i meteorologiczne, roślinność lub obecność zbiorników ze szczególnym uwzględnieniem tych cech, które mogą mieć wpływ na terytorium Rzeczpospolitej Polskiej);
- Szczegółowe założenia, w tym dane meteorologiczne oraz dane dotyczące uwolnień substancji promieniotwórczych do środowiska w przypadku wystąpienia poważnej awarii, a także wyniki prognoz rozwoju sytuacji radiacyjnej, w tym szczegółowa ocena potencjalnego wpływu na zdrowie publiczne i środowisko naturalne na obszarze Rzeczpospolitej Polskiej;
- Struktura rolnictwa, eksportu artykułów rolnych z terenów objętych wpływem obiektu jądrowego oraz kierunki eksportu artykułów rolnych;
- 13. Zidentyfikowane grupy krytyczne;
- 14. Sposób informowania władz polskich w przypadku awarii Elektrowni Jądrowej Chmielnicki z podaniem oczekiwanych prognoz warunków radiologicznych na terenie Rzeczpospolitej Polskiej, strategie komunikacji z mieszkańcami obszarów transgranicznych, w tym z obszarów przygranicznych Rzeczpospolitej Polskiej, w celu zapewnienia transparentności, dostępu do informacji;
- Wnioski na temat wzrostu skumulowanego oddziaływania całej elektrowni jądrowej na środowisko po oddaniu do eksploatacji planowanych bloków;
- Ocena łączna skutków ciężkiej awarii we wszystkich obiektach jądrowych zlokalizowanych w wybranej lokalizacji.

Ponadto, w dalszych etapach konsultacji transgranicznych Strona polska jest zainteresowana możliwością uzyskania dokładnych zewnętrznych planów postępowania awaryjnego, jak również możliwością dostępu do wyników pomiarów środowiskowego monitoringu radiacyjnego wokół Elektrowni Jądrowej Chmielnicki podczas budowy bloków nr 5 i 6 oraz w trakcie ich eksploatacji.

Z poważaniem,

DOROTA TORYFTER-SZUMAŃSKA Zastępca Dyrektora Departament Ocen Oddziaływania na Środowisko / – podpisany cyfrowo/

## Załączniki (w j. polskim):

- 1. Pismo Prezesa Państwowej Agencji Atomistyki z 26 kwietnia 2024 r.
- Pismo Regionalnego Dyrektora ochrony Środowiska w Rzeszowie z 26 kwietnia 2024 r.
  Pismo Regionalnego Dyrektora Ochrony Środowiska w Lublinie z 23 kwietnia 2024 r.
- 4. Pismo Dyrektora Państwowego Instytutu Geologicznego Państwowego Instytutu Badawczego z 7 maja 2024 r.

## Do wiadomości:

- 1. Prezes Paristwowej Agencji Atomistyki
- 2. Dyrektor Państwowego Instytutu Geologicznego- Państwowego Instytutu Badawczego
- Regionalny Dyrektor Ochrony Środowiska w Lublinie
  Regionalny Dyrektor ochrony Środowiska w Rzeszowie



Andrzej Głowacki

DBJ-WLOP.416.2.2024

Warszawa, 24-04-2024r.

Pani Dorota Toryfter-Szumańska Zastępca Dyrektora Departamentu Ocen Oddziaływania na Środowisko Generalna Dyrekcja Ochrony Środowiska Al. Jerozolimskie 136 00-305 Warszawa

Dotyczy: opinii w zakresie transgranicznego oddziaływania na środowisko przedsięwzięcia pn. "Budowa bloków energetycznych nr 5 l 6 na terenie Elektrowni Jądrowej Chmielnicki z wykorzystaniem parametrów technicznych reaktora Westinghouse Electric Company AP1000 do dalszej eksploatacji i wytwarzania energii elektrycznej".

#### Szanowna Pani Dyrektor,

W nawiązaniu do pisma z dnia 4 kwietnia 2024 r. dotyczącego wszczęcia postępowania w sprawie oceny oddziaływania na środowisko w kontekście transgranicznym dla przedsięwzięcia polegającego na budowie bloków energetycznych nr 5 i 6 na terenie Elektrowni Jądrowej Chmielnicki z wykorzystaniem parametrów technicznych reaktora Westinghouse Electric Company AP1000 do dalszej eksploatacji i wytwarzania energii elektrycznej, uprzejmie informuję, że Państwowa Agencja Atomistyki dokonała analizy dostarczonego przez Stronę ukraińską dokumentu i niniejszym przedstawia swoje stanowisko w sprawie.

Elektrownia Jądrowa Chmielnicki znajduje się w odległości około 190 km od granic Rzeczpospolitej Polskiej. W przypadku wystąpienia awarii skutkującej znaczącymi uwolnieniami substancji promieniotwórczych do środowiska, może wystąpić zagrożenie radiologiczne dla ludności Rzeczpospolitej Polskiej, a tym samym może istnieć konieczność podjęcia odpowiednich działań interwencyjnych na terenie Rzeczpospolitej Polskiej. Ponadto sama Strona ukraińska stwierdziła, że planowane przedsięwzięcie należy do pierwszej i drugiej kategorii działalności i oblektów, które zgodnie z prawem ukraińskim mogą mieć znaczący wpływ na środowisko, również na środowisko krajów sąsiednich, m.in. Rzeczpospolitej Polskiej.

Państwowa Agencja Atomistyki 04. Nowy Świat 6/12, 00-400 Warszawa www.paa.gov.pl тец. 22 556 28 00 FAX. 22 621 37 86 Е-мак., kancelaria@paa.gov.pl W związku z przytoczonymi wyżej argumentami oceniono, że zasadne jest przystąpienie Rzeczpospolitej Polskiej do oceny oddziaływania na środowisko w kontekście transgranicznym dla przedmiotowej inwestycji.

Zgodnie z przekazaną informacją zakres raportu oddziaływania na środowisko wskazany jest w dokumencie "Law of Ukraine on Environmental Impact Assessment", jednakże dotyczy ogólnych tematów i nie porusza zagadnień specyficznych dla bezpieczeństwa obiektów jądrowych. W związku z powyższym zasadne jest zwrócenie się z prośbą do Strony ukraińskiej o uwzględnienie następujących zagadnień w raporcie oddziaływania na środowisko:

- Wpływ budowy bloków jądrowych nr 5 i 6 na bezpieczną eksploatację bloków nr 1 i 2 oraz nieukończone bloki nr 3 i 4, w tym wpływ na działania podejmowane w ramach planu postępowania awaryjnego.
- Współpraca z innymi blokami jądrowymi Elektrowni Jądrowej Chmielnicki po oddaniu bloków nr 5 i 6 do eksploatacji, w tym systemów wspólnych dla bloków 5 i 6 oraz bloków 1, 2, 3 i 4.
- Organizacja środowiskowego monitoringu radiacyjnego wokół Elektrowni Jądrowej Chmielnicki podczas budowy bloków nr 5 i 6 oraz w trakcie ich eksploatacji.
- 4. Opis w jakim stopniu bloki nr 5 i 6 Elektrowni Jądrowej Chmielnicki będą podobne do referencyjnego bloku AP1000 i z czego wynikać będą różnice. Jaki wpływ na zmiany projektowe będą miały ukraińskie wymagania dotyczące bezpieczeństwa jądrowego i ochrony radiologicznej.
- Modyfikacje w systemie elektroenergetycznym kraju i elektrowni w związku z eksploatacją bloków nr 5 i 6. Sposób wykorzystania ciepła z bloków nr 5 i 6, a także wpływ na obecny system ciepłowniczy elektrowni.
- Wpływ obecnej stuacji na Ukrainie na budowę i eksploatację bloków 5 i 6 oraz sam projekt reaktora AP1000 w stosunku do referencyjnego projektu.
- Postępowanie z wypalonym paliwem jądrowym i odpadami promieniotwórczymi z bloków nr 5 i 6 oraz wpływ wzrostu ilości wypalonego paliwa jądrowego i odpadów promieniotwórczych na krajowy plan postępowania z wypalonym paliwem jądrowym i odpadami promieniotwórczymi.
- Rodzaje i aktywność odprowadzanych do środowiska radionuklidów, ich forma fizyczna i chemiczna, metody i drogi oraz tempo odprowadzania do środowiska, jak również mechanizmy transferu radionuklidów w środowisku, w tym mechanizmy dyspersji i kumulacji oraz ich sezonowa zmienność w warunkach normalnych.
- Istniejące poziomy stężeń radionuklidów w środowisku i ich zmienność oraz obecność zanieczyszczeń fizycznych lub chemicznych, które mogą wpływać na transfer radionuklidów.
- Naturalne i antropogeniczne cechy środowiska, które będą miały wpływ na transfer radionuklidów (np. warunki geologiczne, hydrologiczne i meteorologiczne, roślinność lub obecność zbiorników ze szczególnym

Strona 2 z 4

uwzględnieniem tych cech, które mogą mieć wpływ na terytorium Rzeczpospolitej Polskiej).

- 11. Szczegółowe założenia, w tym dane meteorologiczne oraz dane dotyczące uwolnień substancji promieniotwórczych do środowiska w przypadku wystąpienia poważnej awarii, a także wyniki prognoz rozwoju sytuacji radiacyjnej, w tym szczegółowa ocena potencjalnego wpływu na zdrowie publiczne i środowisko naturalne na obszarze Rzeczpospolitej Polskiej.
- Struktura rolnictwa, eksportu artykułów rolnych z terenów objętych wpływem obiektu jądrowego oraz kierunki eksportu artykułów rolnych.
- 13. Zidentyfikowane grupy krytyczne.
- 14. Sposób informowania władz polskich w przypadku awarii Elektrowni Jądrowej Chmielnicki z podaniem oczekiwanych prognoz warunków radiologicznych na terenie Rzeczpospolitej Polskiej, strategie komunikacji z mieszkańcami obszarów transgranicznych, w tym z obszarów przygranicznych Rzeczpospolitej Polskiej, w celu zapewnienia transparentności, dostępu do informacji.
- Wnioski na temat wzrostu skumulowanego oddziaływania całej elektrowni jądrowej na środowisko po oddaniu do eksploatacji planowanych bloków.
- Ocena łączna skutków ciężkiej awarii we wszystkich obiektach jądrowych zlokalizowanych w wybranej lokalizacji.

Ponadto w dalszych etapach konsultacji transgranicznych Strona polska jest zainteresowana możliwością uzyskania dokładnych zewnętrznych planów postępowania awaryjnego, jak również możliwością dostępu do wyników pomiarów środowiskowego monitoringu radiacyjnego wokół Elektrowni Jądrowej Chmielnicki podczas budowy bloków nr 5 i 6 oraz w trakcie ich eksploatacji.

> Z poważaniem, Andrzej Głowacki Prezes

Państwowej Agencji Atomistyki / -- podpisany cyfrowo/

Otrzymuje: Adresat Egzemplarz dla: PAA-DBJ

Strona 3 z 4



REGIONALNA DYREKCJA OCHRONY ŚRODOWISKA W RZESZOWIE

#### WOO\$.442.2.2024.AW.8

## Rzeszów, dnia 26 kwietnia 2024 r.

#### Generalny Dyrektor Ochrony Środowiska

Nawiązując do pisma Generalnego Dyrektora Ochrony Środowiska z dnia 09 kwietnia 2024 r. znak: DOOŚ-TSOOŚ.442.12.2024.JA, którym powiadomiono o wszczęciu przez Ukrainę postępowania w sprawie oceny oddziaływania na środowisko w kontekście transgranicznym dla przedsięwzięcia pn.: "Budowa bloków energetycznych nr 5 i 6 na terenie Elektrowni Jądrowej Chmielnicki z wykorzystaniem parametrów technicznych reaktora Westinghouse Electric Company AP1000 do dalszej eksploatacji i wytwarzania energii elektrycznej" na podstawie art. 119 ust, 2 ustawy z dnia 3 października 2008 r. o udostępnianiu informacji o środowisku i jego ochronie, udziale społeczeństwa w ochronie środowiska oraz o ocenach oddziaływania na środowisko (Dz. U. z 2023 r. poz. 1094 ze zm.) oraz pisma Regionalnego Dyrektora Ochrony Środowiska w Rzeszowie z dnia 24 kwietnia 2024 r. znak: WOOŚ.442.2.2024.AW.5, w którym wyrażono stanowisko o zasadności udziału Rzeczypospolitej Polskiej na prawach Strony narażonej w postępowaniu dotyczącym transgranicznego oddziaływania na środowisko przedmiotowego przedsięwzięcia na środowisko.

Zdaniem tut. Dyrekcji, w ww. Raporcie należy:

- Przedstawić ocenę rozprzestrzenienia się transgranicznego zanieczyszczeń, wraz ze wskazaniem podstaw do jej wykonania. Powinna ona zawierać analizę oddziaływań skumulowanych planowanych bloków 5 i 6 oraz pozostałych eksploatowanych w danym czasie w wyznaczonej lokalizacji.
- Ww. ocena powinna być przeprowadzona w oparciu o aktualne warunki meteorologiczne dla narażonych krajów.
- Ocena powinna być przeprowadzona dla warunków normalnej pracy oraz w przypadku wystąpienia awarii obszarze elektrowni jądrowej.
- 4. Przedstawić możliwe rozpatrywane scenariusze awaril.
- Przedstawić planowane do stosowania zabezpieczenia oblektów jądrowych przed wystąpieniem awarii i wycieku niebezpiecznych substancji i zanieczyszczeń.
- Przedstawić planowany do stosowania system monitoringu radiologicznoekologicznego.
- Przedstawić planowaną do stosowania technologię magazynowania i postępowania z wypalonym paliwem jądrowym.

Jednocześnie informuję, iż obwieszczeniem z dnia 26 kwietnia 2024 r. znak: WOOŚ.442.2.2024.AW.6, zapewniono zgodnie z art. 119 ust. 1 ustawy z dnia 3 października 2008 r. o udostępnianiu informacji o środowisku i jego ochronie, udziałe społeczeństwa w ochronie środowiska oraz o ocenach oddziaływania na środowisko, na okres 20 dni, udział społeczeństwa.

Regionalny Dyrektor Ochrony Środowiska w Rzeszowie

(-) Wojciech Wdowik (podpisano bezpiecznym podpisem elektronicznym)



dniany wyrsigania EMAS - zarzydziony arzydeni sfekrywne, aszczednie i prokładowiskowo

Otrzynoują: 1. Adresat doręczenie elektroniczne ePUAP

Do wiadomości: 1. WOOŚ, aa

WOOŚ.442.2.2024 AW.8

Strona 2 z 2
Regionalna Dyrekcja Ochrony Środowiska w Lublinie 20-144 Lublin, ul. Bazylianówka 46

WOO\$.442.2.2024.KK.1

Lublin, dnia 23 kwietnia 2024r.

#### Generalny Dyrektor Ochrony Środowiska

W odpowiedzi na pismo Generalnego Dyrektora Ochrony Środowiska z dnia 9 kwietnia 2024r., znak: DOOŚ-TSOOŚ.442.12.202024.JA2 przekazujące powiadomienie Ministerstwa Ochrony Środowiska i Zasobów Naturalnych Ukrainy dotyczące wszczęcia postępowania w sprawie oceny oddziaływania na środowisko w kontekście transgranicznym dla przedsięwzięcia pn. "Budowa bloków energetycznych nr 5 i 6 na terenie Elektrowni Jądrowej Chmielnicki z wykorzystaniem parametrów technicznych reaktora Westinghouse Electric Company AP1000 do dalszej eksploatacji i wytwarzania energii elektrycznej" wraz z dokumentacją w języku polskim i anglelskim, działając na podstawie art. 119 ustawy z dnia 3 października 2008 r. o udostępnianiu informacji o środowisku i jego ochronie, udziale społeczeństwa w ochronie środowiska oraz ocenach oddziaływania na środowisko (t.j. Dz. U. z 2023 r., poz. 1094 ze zm.) Regionalny Dyrektor Ochrony Środowiska w Lublinie przedstawia stanowisko w przedmiotowej sprawie.

Przedmiotem powiadomienia jest rozbudowa istniejącej i funkcjonującej Elektrowni Jądrowej Chmielnicki o dwa nowe bloki energetyczne. Planowana do rozbudowy Elektrownia położona jest na Wyżynie Wołyńskiej w powiecie Słavuta w pobliżu miejscowości Netishyn, w odlegiości ok. 185 km od granicy województwa lubelskiego. Z uwagi na przewidywaną technologię oraz odległość od granicy Państwa Polskiego nie przewiduje się, by przy normalnej pracy bloków energetycznych wystąpiły oddziaływania mogące mieć wpływ na ludzi i środowisko w kontekście transgranicznym. Przeniesienie radioaktywnych nuklidów na teren Rzeczypospolitej Polskiej może nastąpić w przypadku wystąpienia sytuacji noszącej znamiona awarii projektowej bądź ponadprojektowej oraz sprzyjających warunków atmosferycznych, w tym wiatrów wiejących w kierunku zachodnim. Stąd w opinii Regionalnego Dyrektora Ochrony Środowiska w Lublinie <u>istnieje konieczność przystąpienia do procedury</u> transgranicznej dla planowanego przedsięwzjęcia.

W Raporcie opracowanym na potrzeby transgranicznej oceny oddziaływania na środowisko powinny znaleźć się informacje o:

- zaplanowanych rozwiązaniach technicznych, technologicznych i organizacyjnych mających na celu przeciwdziałanie możliwości wystąpienia awarii projektowej i ponadprojektowej, mających na celu niedopuszczenie do przekroczenia dopuszczalnego poziomu skażeń i zapobieganie ekologicznym następstwom;
- wpływie poważnej awarii pozaprojektowej lub sytuacji o charakterze terrorystycznym na zdrowie i życie społeczeństwa polskiego;

Z uwagi na udział społeczeństwa w procedurze transgranicznej wszelkie analizy powinny zostać podsumowane i skonkludowane w sposób przystępny;

informowaniu strony polskiej o aktualnej sytuacji radiologicznej oraz w przypadku wystąpienia sytuacji awaryjnych,

sposobie postępowania z odpadami promieniotwórczymi i wypalonym paliwem jądrowym.
Zagadnienia powinny zostać opracowane w ujęciu skumulowanym, tj. przy uwzględnieniu możliwości nakładania się oddziaływań.

Zgodnie z art. 119 ust. 1 ustawy z dnia 3 października 2008 r. o udostępnianiu informacji o środowisku i jego ochronie, udziale społeczeństwa w ochronie środowiska oraz ocenach oddziaływania na środowisko (t.j. Dz. U. z 2023r., poz. 1094 ze zm.) Regionalny Dyrektor Ochrony Środowiska w Lublinie Obwieszczeniem znak: WOOŚ.442.2.2024.KK.2 z dnia 23 kwietnia 2024 r. zapewnił społeczeństwu możliwość udziału w postępowaniu zgodnie z zapisami działu III ww. ustawy i wyłożył do wglądu dokumenty przesłane przez Stronę Ukraińską w języku polskim i języku angielskim. Jednocześnie Zawiadomieniem znak: WOOŚ.442.2.2024.KK.3 z dnia 23 kwietnia 2024 r. Regionalny Dyrektor Ochrony Środowiska w Lublinie poinformował wszystkich wójtów, burmistrzów i prezydentów miast województwa lubelskiego o toczącym się postępowaniu z udziałem społeczeństwa.

Wszystkie uwagi i wnioski, które zostaną przedłożone podczas trwającej procedury udziału społeczeństwa zostaną niezwłocznie po jej zakończeniu przekazane Generalnemu Dyrektorowi Ochrony Środowiska.

Regionalny Dyrektor Ochrony Środowiska w Lublinie Beata Sielewicz /Podpisano cyfrowo/

#### Załączniki:

- Obwieszczenie Regionalnego Dyrektora Ochrony Środowiska w Lublinie: WOOŚ.442.2.2024.KK.2 z dnia 23 kwietnia 2024 r.
- Zawiadomienie Regionalnego Dyrektora Ochrony Środowiska w Lublinie: WOOŚ.442.2.2024.KK.3 z dnia 23 kwietnia 2024 r.

Otrzymują:

1. Generalna Dyrekcja Ochrony Środowiska - doręczenie elektroniczne e-puap



Państwowy Instytut Geologiczny Państwowy Instytut Badawczy państwowa służba geologiczna

Warszawa, 07.05.2024

#### BSG.070.19.2024

Szanowny Pan dr Piotr Otawski Generalny Dyrektor Ochrony Środowiska

**Dotyczy:** udziału w transgranicznej ocenie oddziaływania na środowisko dla przedsięwzięcia pn. "Budowa bloków energetycznych nr 5 i 6 na terenie Elektrowni Jądrowej Chmielnicki z wykorzystaniem parametrów technicznych reaktora Westinghouse Electric Company AP1000 do dalszej eksploatacji i wytwarzania energii elektrycznej"

#### Szanowny Panie Dyrektorze,

Państwowy Instytut Geologiczny – Państwowy Instytut Badawczy, podobnie jak w przypadku oceny oddziaływania na środowisko dla Chmielnickiej Elektrowni Jądrowej blok 3 i 4, również w przypadku budowy bloków energetycznych nr 5 i 6 na terenie ww. elektrowni z wykorzystaniem parametrów technicznych reaktora Westinghouse Electric Company AP1000 do dalszej eksploatacji i wytwarzania energii elektrycznej widzi zasadność przystąpienia Strony polskiej do procedury transgranicznej dla planowanego przedsięwzięcia.

Naszym zdaniem podczas przygotowywania raportu oddziaływania na środowisko Strona ukralńska powinna uwzględnić następujące kwestie:

- Prognozowane oddziaływanie planowanej inwestycji na układ hydrodynamiczny strefy aktywnej wymiany wód podziemnych,
- Prognozowane oddziaływanie planowanej inwestycji na temperaturę i skład chemiczny strefy aktywnej wymiany wód podziemnych,

pgi.gov.pl

ul. Rekowiecke 4, 00-975 Warszawa tel. (+48) 22 45 92 000, biuro@pgl.gov.pl

Sąd Rejonowy dle m. st. Werszewy w Warszewie XIII Wydzieł Gospodarczy KRS, Nr 0000122099 NIP 52500040-40

- Prognozowane oddziaływanie planowanej inwestycji na denudację chemiczną w kontekście stabilności podłoża pod wszystkimi obiektami ww. elektrowni,
- Przewidywany sposób i miejsce zagospodarowania zużytego paliwa jądrowego i postępowania z odpadami radioaktywnymi.

#### Z poważaniem,

prof. dr hab. Krzysztof Szamałek

Dyrektor Państwowy Instytut Geologiczny - Państwowy Instytut Badawczy /podpisany cyfrowo/

#### pgi.gov.pl

ul. Rekowiecka 4, 00.975 Warszowa tel. (+48) 22 45 92 000, biuro@pgi.gov.pl

Sąd Rejonowy dla m. sl. Warszawy w Warszawie XIII Wydział Gospodarczy KRS, Nr 0000122099 NIP 525-000-80-40



Ministerul Mediului, Apelor și Pădurilor

### MINISTER'S CABINET

No. DGEICPSC/22629/ HF/16 .07.2024

### To: MINISTRY OF ENVIRONMENTAL PROTECTION AND NATURAL RESOURCES OF UKRAINE

In att: Mr. Ruslan Strilets, Minister

Dear Minister Strilets,

The Ministry of Environment, Waters and Forests of Romania sends cordial greetings to the Ministry of Environmental Protection and Natural Resources of Ukraine and particularly appreciates the bilateral cooperation in the field of environmental protection.

I hereby acknowledge your letter no. 25/5-21/4341-24/02.04.2024 regarding the notification for the proposed project «*Construction of power units No. 5 and No. 6 at the site of the Khmelnytsky NPP using technical characteristics of the Westinghouse Electric Company AP1000 reactor for further operation and electricity generation*», developer JSC National Nuclear Energy Generating Company Energoatom, received by electronic means on the 12th of June 2024. A copy of the Note Verbale from 13th June 2024 accompanying the letter was also received on the 14th of June 2024 by electronic means.

After careful consideration of the information provided in the notification, I would like to inform you that, in accordance to the provisions of the Espoo Convention, we wish to take part in the environmental impact assessment procedure in a transboundary context for the project <u>«Construction of power units No. 5 and No. 6 at the site of the Khmelnytsky NPP using technical characteristics of the Westinghouse Electric Company AP1000 reactor for further operation and electricity generation</u>» and I have the pleasure to forward several comments and proposals from our side related to the contents of the submitted notification.

The Khmelnytsky Nuclear Power Plant (KhNPP) is located in the central part of the Western Ukraine, on the border of three oblasts/counties, namely: Khmelnitsky, Rivne and Ternopil in the vicinity of Netishin city, Khmelnytsky region. The site of the Khmelnytsky nuclear power plant is located to the north of the Romanian national territory, the distance from the northern border of Romania being approximately 227 km and approximately 655 km from Bucharest. Construction of the power plant started in 1981 in Soviet times, being designed as a power plant

with four VVER-1000 type nuclear reactors. At this time, two VVER-1000 reactors are operational, Units 1 and 2 each generating 1000 MW (net) of electricity. The other two VVER-1000 reactors, Units 3 and 4 are under construction. The proposed project will increase the power generated at the nuclear facility with the power of two AP1000 reactors.

The nuclear power plant Khmelnytsky is taken into consideration in the Romanian National Nuclear Emergency Response Plan. The potential direct radiological impact in the event of a severe accident is identified in the north and east of Romania, the areas that include the territory of Romania from the 300 km circle around the plant.

The notification contains very general information on the project and a sum of declarations, part of them predicting the negligible impact on public and environment of the project. Below are presented several aspects that were not included in the notification and we believe that the EIA report should detail.

In the absence of actual technical specifications data from the nuclear facility (confinement, steam management structure, active systems of restraint/reduction of radioactivity losses from the control volumes of the energy block, etc.) as well as a source term confirmed by the nuclear facility operator, the hypothetical scenarios developed by our experts are purely informative, serving the purpose of rapid identification (screening) of potential hazards to Romania's national territory.

We are interested in receiving detailed information on radiological impact studies and assessment of transboundary radiological consequences, as well as source term calculation data for design basis accidents and severe accidents, as soon as they become available. At this stage of the consultations, in the absence of results on the radiological impact assessment, including the consequences of a severe accident, we cannot make any requirements regarding the content of the safety analysis.

The EIA should include information that describes alternative 0 (no project), and, since the notification mentions that no other technological alternatives have been considered, the arguments for the chosen technology. The notification mentions that territorial alternatives (Rivne NPP and South Ukraine NPP) have been taken into account. Please provide arguments for choosing the Khmelnytsky site.

The notification includes a statement regarding the minimization of the negative impact on the environment, through key technological solutions and through protective measures that guarantee environmental protection during construction. The EIA report should include information on the key technological solutions and protective measures, during operation of the NPP (as well as information on aspects of decommissioning). We believe that the EIA report should include such information for all phases of the project.

As mentioned in the notification, the environmental and other restrictions on the planned activities are established pursuant to the legislation of Ukraine in compliance with the maximum permissible levels of anthropogenic impact on the environment, sanitary standards, radiation regulations, etc. The EIA report should include information on the maximum permissible levels required by Ukrainian legislation, the results of analyses performed for the project and their conformity with national legislation, but also in accordance with safety and environmental

standards, reference levels, etc. used at international level. It should be presented if the national nuclear regulatory framework was or will be updated to cover the technology proposed for the two new units.

The EIA report should include information on the accident analysis that would be performed for all type of scenarios required by national but also international safety standards and their results clearly supporting the declaration/prediction of negligible impact on public and environment should be presented.

Aspects related to the management of radioactive waste and spent nuclear fuel resulting from the operation of the nuclear power plant and from the decommissioning process are of interest.

The EIA report should include information on aspects of impact on climate change and biodiversity as well as detailing the potential significant negative effects on the environment, determined by the project's vulnerability to climate change and to all risks (major accidents, war, natural disasters) as well as cumulative impact taking into account the effect of the proposed project and that of other similar existing or planned projects that are located in the area of the proposed project.

Please accept, Mr. Minister, the expression of my high consideration and esteem.

Miroea FECHET Minister of environment, waters and forests

### **APPENDIX L**

# Calculations of the pollutant releases rate from the designed sources of units 5 and 6 with referenced methods

# Розрахунки потужності викидів ЗР від проєктованих джерел енергоблоків № 5 та № 6 з посиланням на використані методики

Під час експлуатації певне забруднення атмосфери викидами шкідливостей буде від резервуарів з дизпаливом, будівель: турбінного відділення, дизельгенератора, ремонтно-механічної майстерні, внутрішнього проїзду автомобілей (під час заїзду та виїзду з парковки).

# Резервуар зберігання дизельного палива в ємності об'ємом 1000 м³ (джерела викидів №№ 1001-1004 на Генплані).

Загальна кількість дизельного палива, що буде прийматися в ємність – 160 т/рік або 193 м³/рік (щільність дизельного палива приймається 830 кг/м³) виходячи із кількості, що необхідна для роботи резервних дизельних електростанцій.

Резервуар наземний, обладнується дихальним клапаном НДКМ-200, що призначений для регулювання тиску парів нафтопродуктів в резервуарі.

Розрахунок викидів забруднюючих речовин виконуються згідно з «Сборник методик по расчету количества загрязняющих веществ в выбросах от неорганизованных источников загрязнения атмосферы», Донецьк, 2001 р., п. 2.3.1.1 за формулою:

 $M = 2,52 * V_{x} * P_{s(38)} * M_{\pi} * (K_{5x} + K_{5t}) * K_{6} * K_{7} * (1-n) / 1000000000, кг/год,$ 

де V_ж - річний об'єм дизельного палива, що зливається, м³/рік;

 $P_{s(38)}$  - тиск насичених парів при температурі 38° речовини, відповідно над чистою та багатокомпонентною рідиною, гПа;

Значення тиску насичених парів для нафтопродуктів приймається по таблиці Пб.1 додатку 6, в залежності від значення еквівалентної температури початку кипіння рідини t_{екв}, та визначається за формулою:

$$t_{e\kappa e} = t_{n.\kappa} + ((t_{\kappa.\kappa} - t_{n.\kappa.}) / 8, 8)), \ ^{\circ}C$$

де t_{к.к} - t_{п.к.} - відповідно початок та кінець кипіння рідини. Для дизельного палива, згідно

ДСТУ 3868-99 «Дизельне паливо. Технічні умови»,  $t_{\kappa.\kappa}$  = 370°С,  $t_{п.\kappa.}$  = 151°С.

$$t_{e\kappa e} = 151 + ((370 - 151) / 8, 8)) = 176^{\circ}C$$

отже P_{s(38)} = 4,02 гПа (проведена інтерполяція).

М_п - молекулярна маса парів рідини. Для дизельного палива, в залежності від марки та основних компонентів, молекулярна маса може становити 110-230 г/моль, для зручності, в розрахунках приймаємо тип і основні показники палива, що використовується на підприємстві, тому М_п = 128 г/моль;

К_{5х} та К_{5т} - поправочні коефіцієнти, що залежать від тиску насичених парів і температури газового простору всередині ємності зберігання відповідно в холодний і теплий період року.

Поправочні коефіцієнти для холодного і теплого періоду року приймаються по таблиці ПЗ.6., відповідно до середньої температури атмосферного повітря у відповідний період року - теплий або холодний.

При зберіганні дизельного палива в наземних резервуарах середня температура газового простору всередині приймається рівною середній температурі атмосферного повітря за відповідний період. Приймаємо в холодний період року температуру -0,5°C, в теплий період року 14,5°C. Тому, відповідно до значень, наведених в таблиці,  $K_{5x} = 0,0585$ ,  $K_{5T} = 0,1895$  (проведена інтерполяція).

К₆- поправочний коефіцієнт, що залежить від тиску насичених парів і річного оберту нафтопродуктів.

Поправочний коефіцієнт залежить від розміщення підприємства в тій або іншій кліматичній зоні, тиску насичених парів та річного оберту нафтопродукту. Річний оберт дизельного палива визначається, згідно додатку 4, за формулою:

$$\Pi = \mathbf{V}_{\mathfrak{K}} / \mathbf{V}_{\mathfrak{p}},$$

де V_p - об'єм резервуару,м³.

 $\Pi = 193 \text{ м}^3/\text{рік} / 1000 \text{ м}^3 = 0,193 \text{раз/рік}$ 

Поправочний коефіцієнт визначається по таблиці П4.2, як для середньої кліматичної зони, і становить 1,26.

К₇ - поправочний коефіцієнт, що залежить від технічної оснащеності резервуару і режиму експлуатації.

Поправочний коефіцієнт визначається по таблиці П5.1, в залежності від оснащеності резервуару технічними засобами скорочення втрат і режиму експлуатації. Резервуар обладнується дихальним клапаном НДКМ-200, згідно паспортних даних тиск спрацювання його 1400-1600 Па, що в перерахунку становить 14-16 гПа, тому поправочний коефіцієнт становить 1.

n - коефіцієнт ефективності газовловлюючого устаткування резервуару. В проектуємому резервуарі не передбачається встановлення газовловлюючого устаткування, тому n = 0.

Викиди вуглеводнів насичених C12-C19 (розчинник РПК-26611 і ін.) у перерахунку на сумарний органічний вуглець:

Максимально разові викиди (г/с) визначаються:

 $M_{\text{м.р.}} = 0,000080$  кг/год *1000 / 3600 = 0,00002 г/с

Валові викиди (т/рік) забруднюючих речовин розраховуються виходячи із загального часу зберігання дизельного палива в ємності за формулою:

 $M_{B.B.} = M_{M.p.} * 3600 * T / 1000000,$ 

де Т – загальний час зберігання дизельного палива в резервуарі за вирахуванням часу зливання нафтопродукту з естакади з цистернами, що становить: 8760 год/рік - (193м³/рік / 36 м³/год) = 8754 год/рік. Де 36 м³/год - продуктивність насосів.

Мв.в. = 0,00002 г/с * 3600 * 8754 год/рік / 1000000 = 0,00063 т/рік.

## Естакада зливу дизельного палива (джерела викидів №№ 1005-1008 на Генплані).

Нафтопродукти на територію Хмельницької АЕС доставляються залізничним транспортом в цистернах. Естакада призначена для розвантаження цистерн і зливу дизельного палива в резервуари. Одночасно естакада може приймати 8 од. цистерн, ємністю по 50-60 м³.

Річна кількість надходження дизельного палива становить 600 т/рік, 723 м³/рік (при щільності 830 кг/м³). Додатково планується надходження ще 160 т/рік, 193м³/рік, тому на період експлуатації турбінних цехів енергоблоків №5 і №6, загальна кількість дизельного палива, що буде доставлятись на AEC – 760 т/рік, 916 м³/рік.

Розрахунок викидів забруднюючих речовин виконуються згідно з «Сборник методик по расчету количества загрязняющих веществ в выбросах от неорганизованных источников загрязнения атмосферы», Донецьк, 2001 р., п. 2.3.1.3, за формулою:

 $M = 0,2485 * V_{k} * P_{s(38)} * M_{ii} * (K_{5x} + K_{5t}) / 1000000000, кг/год,$ 

де V_ж - річний об'єм дизельного палива, що зливається, м³/рік;

 $P_{s(38)}$  - тиск насичених парів при температурі 38° речовини, відповідно над чистою та багатокомпонентною рідиною, гПа;

Значення тиску насичених парів для нафтопродуктів приймається по таблиці Пб.1 додатку 6, в залежності від значення еквівалентної температури початку кипіння рідини t_{екв}, та визначається за формулою:

$$t_{e\kappa\theta} = t_{n.\kappa} + ((t_{\kappa.\kappa} - t_{n.\kappa.}) / 8, 8)), \ ^{\circ}C$$

де t_{к.к} - t_{п.к.} - відповідно початок та кінець кипіння рідини. Для дизельного палива, згідно ДСТУ 3868-99 «Дизельне паливо. Технічні умови», t_{к.к} = 370°C, t_{п.к.} = 151°C.

$$t_{e_{KB}} = 151 + ((370 - 151) / 8, 8)) = 176^{\circ}C$$

отже  $P_{s(38)} = 4,02$  гПа (проведена інтерполяція).

М_п - молекулярна маса парів рідини. Для дизельного палива, в залежності від марки та основних компонентів, молекулярна маса може становити 110-230 г/моль, для зручності, в розрахунках приймаємо тип і основні показники палива, що використовується на підприємстві, тому М_п = 128 г/моль, згідно зазначених даних в Звіті по інвентаризації;

К_{5х} та К_{5т} - поправочні коефіцієнти, що залежать від тиску насичених парів і температури газового простору всередині цистерни відповідно в холодний і теплий період року.

Поправочні коефіцієнти для холодного і теплого періоду року приймаються по таблиці ПЗ.6., відповідно до середньої температури атмосферного повітря у відповідний період року - теплий або холодний.

При зливанні дизельного палива з цистерн середня температура газового простору всередині цистерни приймається рівною середній температурі атмосферного повітря за відповідний період. Приймаємо в холодний період року температуру -  $-0.5^{\circ}$ С, в теплий період року -  $14.5^{\circ}$ С (згідно з даними Звіту по інвентаризації). Тому, відповідно до значень наведених в таблиці,  $K_{5x} = 0.0585$ ,  $K_{5T} = 0.1895$  (проведена інтерполяція).

Викиди вуглеводнів насичених C12-C19 (розчинник РПК-26611 і ін.) у перерахунку на сумарний органічний вуглець:

М = 0,2485 * 916 м³/рік * 4,02 гПа * 128 г/моль * (0,0585+ 0,1895) / 1000000000 = 0,000029 кг/год

Максимально разові викиди (г/с) визначаються:

 $M_{\text{м.р.}} = 0,000029 \text{ кг/год } *1000 / 3600 = 0,000008 \text{ г/с.}$ 

Валові викиди (т/рік) забруднюючих речовин розраховуються виходячі із продуктивності насосів для перекачування дизельного палива та загального його об'єму за формулою:

 $M_{B.B.} = M_{M.p.} * 3600 * (V_{\pi} / P) / 1000000,$ 

де Р - продуктивність насосів, згідно даних Звіту по інвентаризації, 36 м³/год.

Мв.в. = 0,000008 г/с * 3600 * (916 м³/рік / 36 м³/год) / 1000000 = 0,0000007 т/рік.

Діаметр дихального клапану та характеристика газоповітряної суміші від проектуємого резервуару приймається аналогічними до клапанів існуючих ємностей і їх параметрів роботи. Висота джерела викиду при зберіганні дизельного палива приймається 12 м, у відповідності до висоти типових резервуарів такого об'єму – 1000 м³.

# Система маслозабезпечення турбінних цехів енергоблоків № 5 і № 6 (джерела викидів №№ 1009-1045 на Генплані).

Система маслозабезпечення енергоблоків призначена для змащування підшипників турбін і генераторів, ущільнень валів генераторів, регулювання та захисту турбін, змащування живильних агрегатів.

Для вентиляції маслосистеми, для створення незначного розрідження в зливних маслопроводах і картерах підшипників призначені витяжні вентилятори.

Витяжний вентилятор складається з кожуха, ротора, на якому закріплено робоче колесо та електродвигун. Масляні пари, що відсмоктуються вентиляторами, під напором надходять у міжтрубний простір маслоохолоджувача, по трубах якого

проходить охолоджуюча циркуляційна вода. Сконденсоване масло зливається в маслобак (99%), а несконденсовані пари викидаються через вихлопну трубу в атмосферне повітря (1%). Викиди відбуваються протягом всього року – 8760 год/рік.

Для турбінних цехів енергоблоків № 5 і № 6 планується встановити ряд обладнання маслосистем. Загальна кількість витяжних вентиляторів буде становити – 36 од. (по 18 од. для кожного цеху).

Викиди несконденсованих парів масла здійснюються через окремі для кожного витяжного вентилятора труби.

Питомий показник викиду парів масла приймається 0,26 г/м³ для застосування норм природного убутку нафтопродуктів при розрахунках викидів забруднюючих речовин.

Згідно з даними Westinghouse, продуктивність одного вентилятора становить 9500 м³/год, 2,64 м³/с, діаметр вихлопної труби - 2,4 м.

Максимально разовий викид забруднюючих речовин (г/с) визначається за формулою:

$$M_{M.p.} = E_M * N * b / 100 \%,$$

де Ем - питомий викид забруднюючої речовини, г/м³;

N - потужність роботи витяжного вентилятора, м³/с;

b - процентне співвідношення несконденсованих парів, що викидаються через вихлопну трубу в атмосферне повітря, по відношенню до загальної кількості відпрацьованого масла, %.

Валові викиди забруднюючих речовин (т/рік) розраховуються за формулою:

Мв.в. = Мм.р. * Т * 3600 / 1000000,

де Т - час роботи обладнання, год/рік.

Викиди масла мінерального нафтового (веретенне, машинне, циліндрове і ін.):

 $M_{\text{m.p.}} = 0,26 \text{ } \text{г/m}^3 * 2,64 \text{ } \text{m}^3/\text{c} * 1 \% / 100 \% = 0,0068 \text{ } \text{r/c};$ 

M_{в.в.} = 0,0068 г/с * 8760 год/рік * 3600 / 1000000 = 0,214 т/рік.

## Будівля ремонтно-механічної майстерні (джерела викидів №№ 1046-1048 на Генплані)

Зварювальна дільниця (джерело викидів № 1046). Зварювальні пости – 3. Вентвикид.

Розрахункові дані прийняті такі самі, як для існуючого на проммайданчику AEC джерела викидів від процесів зварювання в енергоремонтному приміщенні, згідно зі Звітом з інвентаризації викидів ЗР на ВП «Хмельницька AEC» (проммайданчик AEC).

## Ручне дугове електрозварювання.

Витрата зварювального матеріалу:

- електроди УОНІ-13/55 800 кг/рік;
- електроди MP-3 400 кг/рік;
- електроди ЭА-400 250 кг/рік;
- електроди ЭА-395/9 200 кг/рік;
- електроди ЗИО-8 (аналог ЭА-395/9) 50 кг/рік;
- електроди ЦУ-5 (аналог АНО-1М) 300 кг/рік;
- електроди ПТ-30 (аналог НИАТ-3М) 100 кг/рік;
- електроди ОЗБ-38 (аналог Комсомолець-100) 25 кг/рік.

Розрахунок величин викидів проводиться згідно з «Показники емісії (питомі викиди) забруднюючих речовин від процесів електро-, газозварювання, наплавлювання, електро-, газорізання та напилювання метала. ІГМЕ ім. Марзеєва. – К., 2003», табл. 1.2.

Таблиця 1 – Вихідні дані для зварювальної дільниці будівлі ремонтномеханічної майстерні.

Найменування	Забруд	нюючі ре	човини	та пока	зники еміс	ії г/кг звар	ювального	матеріал	ıy	
зварювального	Залізу	Манга-	Хрому	Крем-	Водень	Фториди	Фториди	Азоту	Оксид	Міді
матеріалу,	оксид	ну діо-	три-	нію	фторист.	добре	пог. розч.	діоксид	вуглецю	оксид
витрата на рік		ксид	оксид	оксид		розч.				
1. УОНІ-13/55 —	14,90	1,09	-	1,00	1,26	4,80	2,70	2,70	13,3	-
800 кг/рік										
2. MP-3 – 400	9,70	1,80	-	-	0,4	-	-	-	-	-
кг/рік										
3. ЭА-400 – 250	5,02	0,43	0,25	-	0,54	-	-	-	-	-
кг/рік										
4. ƏA-395/9 —	15,47	1,10	0,43	-	-	-	-	-	-	-
200 кг/рік										
5. 3ИО-8	15,47	1,10	0,43	-	-	-	-	-	-	-
(аналог ЭА-										
395/9) – 50										
кг/рік										
6. ЦУ-5 (аналог	6,67	0,43	-	-	2,13	-	-	-	-	-
AHO-1M) - 300										
кг/рік										
7.ПТ-30	0,10	0,21	-	-	-	-	-	-	-	-
(аналог НИАТ-										
3M) - 100										
кг/рік										
8. ОЗБ-38	2,60	3,90	-	3,50	1,11	-	-	0,76	-	9,80
(аналог										
Комсо-										
молець-100)										
— 25 кг/рік.										

Розрахунок викидів залізу та його сполук (у перерахунку на залізо):

$$\begin{split} \mathbf{M} &= [(14,90*800) + (9,70*400) + (5,02*250) + (15,47*250) + (6,67*300) + (0,10*100) + \\ &\quad (2,60*25)]*10^{-6} = 0,0230 \text{ T/pik.} \end{split}$$

Максимальний викид розрахований за умови одночасної роботи 3-х постів. Максимальна витрата електродів - 1 кг/год.

Mм.р. = 
$$(14,9+9,70+15,47)*1$$
 = 0,011 г/с

3600

Розрахунок викидів мангану та його сполук (у перерахунку на діоксид мангану):

$$M = [(1,09*800)+(1,80*400)+(0,43*250)+(1,10*250)+(0,43*300)+(0,21*100)+(3,90*25)]*10^{-6} = 0,0023 \text{ T/pik.}$$
$$M_{M}.p. = (1,10+1,8+3,90)*1 = 0,0019 \text{ }\Gamma/c$$
$$3600$$

Розрахунок викидів хрому та його сполук (у перерахунку на триоксид хрому):

$$M = [(0,25*250)+(0,43*250)] *10^{-6} = 0,000173 \text{ т/рік.}$$
  
Mм.p. =  $(0,25+0,43+0,43)*1 = 0,0003 \text{ г/с}$   
3600

Розрахунок викидів кремнію діоксиду:

$$M = [(1,0*800)+(3,50*25)] *10^{-6} = 0,00089 \text{ т/рік.}$$
$$M_{M.p.} = \underline{(1,0+3,5)*1} = 0,0013 \text{ г/с}$$
$$3600$$

Розрахунок викидів фтористого водню:

$$M = [(1,26*800)+(0,4*400)+(0,54*250)+(2,13*300)+(1,11*25)]*10^{-6} = 0,0020 \text{ T/pik}.$$
$$M_{M.p.} = (1,26+2,13+1,11)*1 = 0,0012 \text{ F/c}$$

3600

Розрахунок викидів фторидів добре розчинних:

$$M = (4,80*800)*10^{-6} = 0,0038 \text{ т/рік.}$$
  
Mм.p. =  $4,8*1$  = 0,0013 г/с  
3600

Розрахунок викидів фторидів погано розчинних:

$$M = (2,7*800)*10^{-6} = 0,0022 \text{ т/рік.}$$
  
Mм.р. =  $2,7*1$  = 0,0008 г/с  
3600

Розрахунок викидів оксидів азоту (у перерахунку на діоксид азоту):

 $M = [(2,7*800)+(0,76*25)] *10^{-6} = 0,0023$  т/рік.

Мм.р. = 
$$(2,7+0,76)*1$$
 = 0,0010 г/с  
3600

Розрахунок викидів оксиду вуглецю:

$$M = (13,3*800)*10^{-6} = 0,0106 \text{ т/рік.}$$
$$M_{M.p.} = \underline{13,3*1} = 0,0037 \text{ г/с}$$
$$3600$$

Розрахунок викидів міді та її сполук ( у перерахунку на мідь):

$$M = (9,80*25)*10^{-6} = 0,00025 \text{ т/рік.}$$
  
Mм.p. =  $9,8*1 = 0,0027 \text{ г/с}$   
3600

За результатами прямих вимірів найбільша концентрація ЗР у газоповітряній суміші (приведена до н.у.) становить:

- залізо та його сполуки (у перерахунку на залізо) 5,26 мг/м³;
- діоксид мангану 0,84 мг/м³;
- триоксид хрому 0,096 мг/м³;
- оксид вуглецю 1,99 мг/м³;
- оксиди азоту (у перерахунку на діоксид азоту) 0,78 мг/м³;
- мідь та її сполуки (у перерахунку на мідь) 0,93 мг/м³;
- кремнію діоксид 0,50 мг/м³.

Об'єм газоповітряної суміші Vo = 2,58 м³/с.

Величини викидів забруднюючих речовин (г/с) становлять:

- залізо та його сполуки (у перерахунку на залізо) 0,0136 г/с;
- діоксид мангану 0,0022 г/с;
- триоксид хрому 0,00025 г/с;
- оксид вуглецю 0,0051 г/с;
- оксиди азоту (у перерахунку на діоксид азоту) 0,0020 г/с;
- мідь та її сполуки (у перерахунку на мідь) 0,0024 г/с;
- кремнію діоксид 0,0013 г/с.

Таблиця 2 - Результати розрахунку викидів для зварювальної дільниці будівлі ремонтно-механічної майстерні.

Всього по джерелу:	г/с	т/рік
Залізо та його сполуки (у перерахунку на залізо)	0,0136	0,0230
Манган та його сполуки (у перерахунку на діоксид мангану)	0,0022	0,0023
Хром та його сполуки (у перерахунку на триоксид хрому)	0,00025	0,00017
Кремнію діоксид	0,0013	0,00089
Фтористий водень	0,0012	0,0020
Фториди добре розчинні	0,0013	0,0038
Фториди погано розчинні	0,0008	0,0022
Азоту оксиди (у перерахунку на діоксид азоту)	0,0020	0,0023

Продовження таблиці 2

Всього по джерелу:	г/с	т/рік
Оксид вуглецю	0,0051	0,0106
Мідь та її сполуки (у перерахунку на мідь)	0,0024	0,00025

Верстати металообробні (джерела викидів №№ 1047-1048). Вентвикиди. Викиди забруднюючих речовин з приміщення здійснюється за допомогою двох дахових вентиляторів.

Розрахункові дані прийняті такі самі, як для існуючої на проммайданчику АЕС майстерні ремонтно-механічного цеху, де встановлені металообробні верстати, при роботі яких використовується змащувально-охолоджувальна рідина (емульсол), згідно зі Звітом з інвентаризації викидів ЗР на ВП «Хмельницька АЕС» (проммайданчик АЕС).

Розрахунок величин викидів здійснюється згідно з «Удельные показатели образования вредных веществ, выделяющихся в атмосферу от основных видов технологического оборудования предприятий машиностроения и военно-промышленного комплекса», г. Харьков, 1997 г.

Питомі виділення емульсолу прийняті по таблиці 6.2 і становлять:

- токарні, фрезерні, свердлильні верстати 0,000002 г/с на 1 кВт потужності верстата;
- шліфувальні верстати 0,000046 г/с на 1 кВт потужності верстата.

Питомі виділення речовин у вигляді суспендованих твердих частинок від шліфувальних верстатів приймаються по таблиці 6.1 і становлять:

- верстат круглошліфувальний (діаметр 600 мм) 0,065 г/с;
- верстат плоскошліфувальний (діаметр 400 мм) 0,055 г/с;

Результати розрахунку зведено в таблицю.

Таблиця 3 - Вихідні дані для металообробних верстатів будівлі ремонтномеханічної майстерні.

Найменування обладнання, тип, кількість	Потужність,	Час	Викиди ем	мульсолу
	кВт	роботи,	г/с	т/рік
		год/рік		
Плоскошліфувальний верстат, 2 шт., діаметр 400	5,0	600	0,0005	0,0010
мм				
Круглошліфувальний верстат, 1 шт., діаметр 600	6,0	600	0,0003	0,0006
мм				
Верстат фрезерний, 8 шт.	17,4	1200	0,00028	0,0012
Токарно-гвинторізний, 4 шт.	11,88	1200	0,00010	0,0004
Токарно-гвинторізний, 3 шт.	8,2	1200	0,00005	0,0002
Токарно-гвинторізний, 3 шт.	16,22	1200	0,00010	0,0004
Токарно-гвинторізний, 2 шт.	30,7	1200	0,00012	0,0005
Токарно-гвинторізний універсальний, 2 шт.	16,2	1200	0,00006	0,0003
Токарно-карусельний, 1 шт.	59,0	1200	0,00012	0,0005
Верстат розточувальний, 2 шт.	6,25	1200	0,00003	0,0001
Верстат радіально-свердлильний, 2 шт.	8,95	1200	0,00004	0,0002
Верстат вертикально-свердлильний, 1 шт.	7,5	1200	0,00002	0,0001
Верстат настільно-свердлильний, 2 шт.	7,5	1200	0,00003	0,0001
Всього верстатів - 33	559		0,00175	0,0056

Розрахунок викидів емульсолу:

Максимальний викид емульсолу визначається із урахуванням одночасності роботи верстатів. Кодн = 0,75.

Умовно приймається таке розподілення викидів по двох джерелах:

- джерело викидів 1046 0,0007 г/с, 0,0028 т/рік;
- джерело викидів 1047 0,0006 г/с, 0,0028 т/рік;

Розрахунок викидів речовин у вигляді суспендованих твердих частинок:

При обробці металів на шліфувальних верстатах із застосуванням ЗОР виділяється пил у кількості 10% від кількості пилу при сухій обробці (примітка до табл. 6.2).

$$M = 0.012*3600*600*10^{-6} = 0.0259$$
 г/с*0,5 = 0.0130 т/рік

За результатами прямих вимірів найбільша концентрація речовин у вигляді суспендованих твердих частинок пилоповітряній суміші (приведена до н.у.) становить:

- джерело викидів 1046: Co= 5,91 мг/ м³; Vo = 1,27 м³/с; Мм.р. = 0,0075 г/с;
- джерело викидів 1047: Co= 6,12 мг/ м³; Vo = 1,18 м³/с; Мм.р. = 0,0072 г/с

-

Таблиця 4 – Результати розрахунку викидів для металообробних верстатів будівлі ремонтно-механічної майстерні.

Всього по джерелу № 1047:	г/с	т/рік
Речовини у вигляді суспендованих твердих частинок	0,0075	0,0130
Емульсол	0,0007	0,0028
Всього по джерелу № 1048:		
Речовини у вигляді суспендованих твердих частинок	0,0072	0,0130
Емульсол	0,0006	0,0028

## Внутрішній проїзд автомобілей під час заїзду та виїзду з парковки (джерело викидів № 1049 на Генплані).

Довжина внутрішного проїзду до парковки – 150 м. Швидкість руху по території підприємства – 10 км/год (відповідно, тривалість проїзду в межах проммайданчика становить 1 хв.). Розрахунок проведений для двох автомобілей (одночасно один заїжджає, другий виїжджає).

Кількість викидів шкідливих речовин в атмосферу від автомобілей визначено згідно з «Методики расчёта выбросов загрязняющих веществ передвижными источниками. ОАО «УкрНТЭК».– Донецк».

Результати наведені в таблиці.

Найменування забруднюючої	Питомий	Коеф. технічн.	Витрата палива,	Викиди ЗР,
речовини	викид, кг/т	стану транс-порту	т/хв.	$\Gamma/c^*$
Автомобілі на дизпаливі				
Діоксид азоту	32,8	0,95	0,0004	0,010
Сажа	3,85	1,8	0,0004	0,002
Діоксид сірки	5,0	1,0	0,0004	0,0000016
Оксид вуглецю	32,0	1,5	0,0004	0,016
Вуглеводні насичені С12-С19	5,65	1,4	0,0004	0,000002

Таблиця 5 – Результати розрахунку викидів для внутрішнього проїзду автомобілей.

* викиди приведені до 20 хв. періоду усереднення.

# Будівля резервного дизель-генераторів (джерела викидів №№ 1050-1051 на Генплані)

Під час експлуатації енергоблоків № 5 та 6 використовуються два резервних дизель-генератори потужністю 5200 кВт - в якості глибокого захисту установки АР1000 під час втрати електроенергії за межами підприємства (LOOP). Кожен дизель-генератор встановлений в окремій будівлі.

Вентвикид. Викид ЗР від будівлі дизель-генератора відбувається через даховий вентилятор. Продуктивність вентилятора 44,7 м³/с. Висота викиду - 8,5 м. Діаметр вихлопної труби – 4 м.

Викиди забруднюючих речовин від дизель-генераторів прийняті за даними Westinghouse та наведені в таблиці нижче.

Скидання забруднюючих речовин	Річні викиди*
Речовини у вигляді суспендованих твердих частинок недиференційованих за складом	< 800 фунтів (< 363 кг)
Діоксид сірки	< 2500 фунтів (<1134 кг)
Оксид вуглецю	< 1000 фунтів (< 454 кг)
Вуглеводні	< 600 фунтів (< 272 кг)
Оксиди азоту	< 12 000 фунтів (< 5 443 кг)

Таблиця 6 – Вихідні дані для резервних дизель-генераторів (2 од.).

* Викиди базуються на 4 годинах роботи на місяць для кожного з генераторів.

Викиди забруднюючих речовин від одного дизель-генератора під час резервного енергоживлення становлять:

Речовини у вигляді суспендованих твердих частинок недиференційованих за складом:

Mм.p. = 
$$\frac{363*1000}{2*4*12*3600}$$
 = 1,05 г/с

Діоксид сірки:

Оксид вуглецю:

MM.p. = 
$$\frac{454 * 1000}{2*4 * 12*3600} = 1,3 \text{ r/c}$$

Вуглеводні:

Mм.p. = 
$$\frac{272 * 1000}{2*4 * 12*3600} = 0,79$$
 г/с

Оксиди азоту:

Таблиця 7 – Результати розрахунку викидів для резервних дизельгенераторів.

Всього по джерелу № 1050/1051:	г/с	т/рік
Речовини у вигляді суспендованих твердих частинок	1,05	0,181
недиференційованих за складом		
Діоксид сірки	3,25	0,562
Оксид вуглецю	1,3	0,225
Вуглеводні	0,79	0,137
Оксиди азоту	15,75	2,722

## **APPENDIX M**

Characterization of the projected sources of non-radioactive pollutant emissions

		Σ	Координати джерела Характеристика точкового другого Кут пилогазоповіт- або початку			Забру	днююча речовина	Визначена потужність викиду								
ела викидів	Найме- нування дже-	Айме- найме- нування дже- арегра		ліній цеі сим площ	лінійного, центра симетрії пощинного		ного, ина і жина щин- ого	жини площи- ного дже- рела	ата, м3/с	, M/C	а, град	Наймену- вання устаткування		Найменування		
Nº джере	한 рела Э викиду 관 기 2	Висота дже	Діаметр дже	X1	У1	X2	У2	віднос- но ОХ завод- ської сис- теми (град)	об'ємна витр	швидкість	температур	(технологіч- ного процесу)	Код	забруднюючої речовини	г/с	т/рік
1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17
							Про	єктовані	джере	ла вин	идів	енергоблоків N	º 5 та 6			
1001	Дихаль- ний клапан	12	0,38	-500	160	-	-	-	0,25	2,2	23	Резервуар зберігання дизельного палива	2754	Вуглеводні насичені C12-C19 (розчинник РПК-26611 і ін.) у перерахунку на сумарний органічний вуглець	0,00002	0,00063
1002	Дихаль- ний клапан	12	0,38	-550	180	-	-	-	0,25	2,2	23	Резервуар зберігання дизельного палива	2754	Вуглеводні насичені C12-C19 (розчинник РПК-26611 і ін.) у перерахунку на сумарний органічний вуглець	0,00002	0,00063

1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17
1003	Дихаль- ний клапан	12	0,38	-678	200	-	-	-	0,25	2,2	23	Резервуар зберігання дизельного палива	2754	Вуглеводні насичені C12-C19 (розчинник РПК-26611 і ін.) у перерахунку на сумарний органічний вуглець	0,00002	0,00063
1004	Дихаль- ний клапан	12	0,38	-728	220	-	-	-	0,25	2,2	23	Резервуар зберігання дизельного палива	2754	Вуглеводні насичені C12-C19 (розчинник РПК-26611 і ін.) у перерахунку на сумарний органічний вуглець	0,00002	0,00063
1005	Неорг. джерело	5	0,5	-550	165	-	-	-	0,29	1,5	23	Естакада зливу дизельного палива з цистерн	2754	Вуглеводні насичені C12-C19 (розчинник РПК-26611 і ін.) у перерахунку на сумарний органічний вуглець	0,00008	0,000007
1006	Неорг. джерело	5	0,5	-600	185	-	-	-	0,29	1,5	23	Естакада зливу дизельного палива з цистерн	2754	Вуглеводні насичені C12-C19 (розчинник РПК-26611 і ін.) у перерахунку на сумарний органічний вуглець	0,00008	0,0000007
1007	Неорг. джерело	5	0,5	-683	205	-	-	-	0,29	1,5	23	Естакада зливу дизельного палива з цистерн	2754	Вуглеводні насичені C12-C19 (розчинник РПК-26611 і ін.) у перерахунку на сумарний органічний вуглець	0,00008	0,0000007
1008	Неорг. джерело	5	0,5	-733	225	-	-	-	0,29	1,5	23	Естакада зливу дизельного палива з цистерн	2754	Вуглеводні насичені C12-C19 (розчинник РПК-26611 і ін.) у перерахунку на сумарний органічний вуглець	0,00008	0,000007

1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17
1009	Труба	48	2,4	-400	315	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №5	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1010	Труба	48	2,4	-395	310	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №5	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1011	Труба	48	2,4	-390	305	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №5	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1012	Труба	48	2,4	-385	300	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №5	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1013	Труба	48	2,4	-380	295	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №5	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1014	Труба	48	2,4	-405	320	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №5	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1015	Труба	48	2,4	-410	315	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №5	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1016	Труба	48	2,4	-415	310	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №5	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214

1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17
1017	Труба	48	2,4	-420	305	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №5	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1018	Труба	48	2,4	-425	300	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №5	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1019	Труба	48	2,4	-430	295	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №5	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1020	Труба	48	2,4	-435	290	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №5	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1021	Труба	48	2,4	-440	285	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №5	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1022	Труба	48	2,4	-445	280	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №5	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1023	Труба	48	2,4	-410	325	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №5	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1024	Труба	48	2,4	-415	320	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №5	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214

1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17
1025	Труба	48	2,4	-420	315	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №5	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1026	Труба	48	2,4	-425	310	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №5	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1027	Труба	48	2,4	-430	305	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №5	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1028	Труба	48	2,4	-561	390	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №6	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1029	Труба	48	2,4	-556	385	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №6	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1030	Труба	48	2,4	-551	380	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №6	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1031	Труба	48	2,4	-546	375	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №6	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1032	Труба	48	2,4	-541	370	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №6	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214

1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17
1033	Труба	48	2,4	-566	395	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №6	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1034	Труба	48	2,4	-571	390	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №6	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1035	Труба	48	2,4	-576	385	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №6	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1036	Труба	48	2,4	-581	380	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №6	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1037	Труба	48	2,4	-586	375	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №6	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1038	Труба	48	2,4	-591	370	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №6	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1039	Труба	48	2,4	-596	365	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №6	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1040	Труба	48	2,4	-601	360	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №6	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214

1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17
1041	Труба	48	2,4	-606	355	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №6	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1042	Труба	48	2,4	-571	400	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №6	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1043	Труба	48	2,4	-576	395	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №6	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1044	Труба	48	2,4	-581	390	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №6	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
1045	Труба	48	2,4	-586	385	-	-	-	2,64	0,58	23	Вентилятор турбінного цеху енерго- блоку №6	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0068	0,214
													123	Залізо та його сполуки (у перерахунку на залізо)	0,0136	0,0230
												Bourungton	143	Марганець та його сполуки (у перерахунку на двоокис марганцю)	0,0022	0,0023
1046	Труба	14	0,6	-320	400				2,74	9,89	23	зварювальної дільниці РММ	203	Хром та його сполуки (у перерахунку на триоксид хрому)	0,00025	0,00017
													323	Кремнію діоксид	0,0013	0,00089
													342	Фтористий водень	0,0012	0,0020
													343	Фториди добре розчинні	0,0013	0,0038
													344	Фториди погано розчинні	0,0008	0,0022

1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17
													301	Азоту оксиди (у перерахунку на діоксид азоту)	0,0020	0,0023
													337	Оксид вуглецю	0,0051	0,0106
													146	Мідь та її сполуки (у перерахунку на мідь)	0,0024	0,00025
												Вентилятор	10265	Емульсол	0,0007	0,0028
1047	Труба	11	0,5	-330	390				1,34	6,82	23	металообр. верстати РММ	2902	Зважені речовини, недиференційовані за складом	0,0075	0,0130
												Вентилятор	10265	Емульсол	0,0006	0,0028
1048	Труба	11	0,5	-295	385				1,34	6,82	23	металообр. верстати РММ	2902	Зважені речовини, недиференційовані за складом	0,0072	0,0130
													301	Азоту оксиди (у перерахунку на діоксид азоту)	0,010	-
													328	Сажа	0,002	-
														Діоксид сірки (діоксид		
1049	Неорг.	5	-	-420	550	-410	695		0,25	32	250	Внутрішній проїзд	330	та триоксид) у перерахунку на діоксид сірки	0,0000016	-
	джерело											автомобілей	337	Оксид вуглецю	0,016	-
													2754	Вуглеводні насичені C12-C19 (розчинник РПК-26611 і ін.) у перерахунку на сумарний органічний вуглець	0,000002	-
1050	Труба	8,5	4	-500	365	-	-	-	44,7	3,56	55	Резервний дизель- генератор енергоблоку	301	Оксиди азоту (оксид та діоксид азоту) у перерахунку на діоксид азоту	15,75	2,722
												Nº5	330		3,25	0,562

1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17
														Діоксид сірки (діоксид та триоксид) у перерахунку на діоксид сірки		
													337	Оксид вуглецю	1,3	0,225
													2754	Вуглеводні насичені C12-C19 (розчинник РПК-26611 і ін.) у перерахунку на сумарний органічний вуглець	0,79	0,137
													2902	Зважені речовини, недиференційовані за складом	1,05	0,181
													301	Оксиди азоту (оксид та діоксид азоту) у перерахунку на діоксид азоту	15,75	2,722
												Резервний	330	Діоксид сірки (діоксид та триоксид) у перерахунку на діоксид сірки	3,25	0,562
												дизель-	337	Оксид вуглецю	1,3	0,225
1051	Труба	8,5	4	-678	440	-	-	-	44,7	3,56	55	генератор енергоблоку №6	2754	Вуглеводні насичені C12-C19 (розчинник РПК-26611 і ін.) у перерахунку на сумарний органічний вуглець	0,79	0,137
													2902	Зважені речовини, недиференційовані за складом	1,05	0,181

1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17
							lc	нуючі д	жерела	вики	дів ен	ергоблоків № ′	1 та 2			
													301	Азоту оксиди (у перерахунку на діоксид азоту)	5,34	0,0480
													337	Оксид вуглецю	1,3781	0,0291
101	Труба	60	3,2	330	-245				11,4	1,42	118	Пуско- резервна котельня ГТЦ	330	Діоксид сірки (діоксид та триоксид) у перерахунку на діоксид сірки	12,7472	2,6166
													2902	Зважені речовини, недиференційовані за складом	0,3731	0,0735
131	Дихаль- ний клапан	6	0,38	535	-325				0,25	2,20	23	Резервуар зберігання дизельного палива	2754	Вуглеводні насичені C12-C19	0,00002	0,0045
132	Дихаль- ний клапан	6	0,38	550	-325				0,25	2,20	23	Резервуар зберігання дизельного палива	2754	Вуглеводні насичені C12-C19	0,00002	0,0045
133	Дихаль- ний клапан	6	0,38	545	-340				0,25	2,20	23	Резервуар зберігання дизельного палива	2754	Вуглеводні насичені C12-C19	0,00002	0,0045
134	Дихаль- ний клапан	6	0,38	532	-340				0,25	2,20	23	Резервуар зберігання дизельного палива	2754	Вуглеводні насичені C12-C19	0,00002	0,0045
00/				175	400				0.74	0.00	0.0	Вентилятор	123	Залізо та його сполуки (у перерахунку на залізо)	0,0136	0,0230
201	Труба	14	0,6	475	-120				2,74	9,89	23	зварювальної дільниці ЕРП	143	Марганець та його сполуки (у перерахунку на двоокис марганцю)	0,0022	0,0023

1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17
													203	Хром та його сполуки (у перерахунку на триоксид хрому)	0,00025	0,00017
													323	Кремнію діоксид	0,0013	0,00089
													342	Фтористий водень	0,0012	0,0020
													343	Фториди добре розчинні	0,0013	0,0038
													344	Фториди погано розчинні	0,0008	0,0022
													301	Азоту оксиди (у перерахунку на ліоксид азоту)	0,0020	0,0023
													337	Оксид вуглецю	0.0051	0.0106
													146	Мідь та її сполуки (у перерахунку на мідь)	0,0024	0,00025
												Вентилятор	10265	Емульсол	0,0007	0,0028
202	Труба	11	0,5	485	-80				1,34	6,82	23	металообр. верстати ЕРП РМЦ	2902	Зважені речовини, недиференційовані за складом	0,0075	0,0130
												Вентилятор	10265	Емульсол	0,0007	0,0028
203	Труба	11	0,5	480	-90				1,34	6,82	23	металообр. верстати ЕРП РМЦ	2902	Зважені речовини, недиференційовані за складом	0,0075	0,0130
305	Труба	50	0,11	195	50				0,75	79	23	Ексгаустер машзала блоку №1	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,00195	0,0615
306	Труба	50	0,11	200	55				0,75	79	23	Ексгаустер машзала блоку №1	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,00195	0,0615
307	Труба	50	0,15	180	5				1,61	91,1	23	Ексгаустер машзала блоку №1	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0042	0,1325

1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17
308	Труба	50	0,08	180	-5				0,19	37,8	23	Ексгаустер машзала блоку №1	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0049	0,0155
309	Труба	50	0,085	170	-40				0,11	19,4	23	Ексгаустер машзала блоку №1	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,00029	0,0091
310	Труба	50	0,085	170	-30				0,11	19,4	23	Ексгаустер машзала блоку №1	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,00029	0,0091
311	Труба	50	0,15	170	-20				0,11	6,23	23	Ексгаустер машзала блоку №1	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0042	0,1325
312	Труба	50	0,15	210	0				0,28	15,8	23	Ексгаустер машзала блоку №1	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,00073	0,0230
313	Труба	50	0,11	40	95				0,75	79,0	23	Ексгаустер машзала блоку №2	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,00195	0,0615
314	Труба	50	0,11	45	105				0,75	79,0	23	Ексгаустер машзала блоку №2	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,00195	0,0615
315	Труба	50	0,15	30	55				1,61	91,1	23	Ексгаустер машзала блоку №2	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0042	0,1325

1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17
316	Труба	50	0,085	15	15				0,11	19,4	23	Ексгаустер машзала блоку №2	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,00029	0,0091
317	Труба	50	0,085	20	20				0,11	19,4	23	Ексгаустер машзала блоку №2	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,00029	0,0091
318	Труба	50	0,15	20	30				0,11	6,23	23	Ексгаустер машзала блоку №2	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,00029	0,0091
319	Труба	50	0,15	55	55				2,08	118	23	Ексгаустер машзала блоку №2	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0054	0,1703
							Про	єктован	і джере	ла ви	кидів	енергоблоків М	№ 3 та 4			
3001	Дихаль- ний клапан	12	0,38	-270	-50	-	-	-	0,25	2,2	23	Резервуар зберігання дизельного палива	2754	Вуглеводні насичені C12-C19 (розчинник РПК-26611 і ін.) у перерахунку на сумарний органічний вуглець	0,00002	0,00063
3002	Неорг. джерело	5	0,5	-280	-65	-	-	-	0,29	1,5	23	Естакада зливу дизельного палива з цистерн	2754	Вуглеводні насичені C12-C19 (розчинник РПК-26611 і ін.) у перерахунку на сумарний органічний вуглець	0,00008	0,000007
3003	Труба	50	0,11	-140	153	-	-	-	0,75	78,9	23	Ексгаустер 3SE81D02 турбінного цеху енерго- блоку №3	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0020	0,0631
3004	Труба	50	0,15	-155	107	-	-	-	1,61	91,0	23		2735		0,0042	0,1325

1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17
												Ексгаустер 3SC59D01 турбінного цеху енерго- блоку №3		Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)		
3005	Труба	50	0,085	-172	68	-	-	-	0,11	19,3	23	Ексгаустер 3SU53D01 турбінного цеху енерго- блоку №3	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0003	0,0095
3006	Труба	50	0,085	-166	74	-	-	-	0,11	19,3	23	Ексгаустер 3SU52D01 турбінного цеху енерго- блоку №3	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0003	0,0095
3007	Труба	50	0,15	-163	82	-	-	-	0,11	6,2	23	Ексгаустер 3SU51D01 турбінного цеху енерго- блоку №3	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0003	0,0095
3008	Труба	50	0,15	-134	107	-	-	-	2,08	117,5	23	Ексгаустер 3SC50D01 турбінного цеху енерго- блоку №3	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0054	0,1703
3009	Труба	50	0,11	-270	217	-	-	-	0,75	78,9	23	Ексгаустер 4SE82D02 турбінного цеху енерго- блоку №4	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0020	0,0631
3010	Труба	50	0,11	-265	222	-	-	-	0,75	78,9	23	Ексгаустер 4SE81D02 турбінного цеху енерго- блоку №4	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0020	0,0631

1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17
3011	Труба	50	0,15	-280	176	-	-	-	1,61	91,0	23	Ексгаустер 4SC59D01 турбінного цеху енерго- блоку №4	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0042	0,1325
3012	Труба	50	0,085	-297	137	-	-	-	0,11	19,3	23	Ексгаустер 4SU53D01 турбінного цеху енерго- блоку №4	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0003	0,0095
3013	Труба	50	0,085	-291	143	-	-	-	0,11	19,3	23	Ексгаустер 4SU52D01 турбінного цеху енерго- блоку №4	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0003	0,0095
3014	Труба	50	0,15	-288	151	-	-	-	0,11	6,2	23	Ексгаустер 4SU51D01 турбінного цеху енерго- блоку №4	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0003	0,0095
3015	Труба	55	0,15	-259	176	-	-	-	2,08	117,5	23	Ексгаустер 4SC50D01 турбінного цеху енерго- блоку №4	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0054	0,1703
3016	Труба	55	0,15	-256	182	-	-	-	2,08	117,5	23	Ексгаустер 4SC50D01 турбінного цеху енерго- блоку №4	2735	Масло мінеральне нафтове (веретенне, машинне, циліндрове і ін.)	0,0054	0,1703